PROJECT MANUAL

MY MARKETPLAC

SUBMISSION

DIMOR



S

240 Continental Drive, Suite 200 Newark, Delaware 19713 phone: 302.738.7551 fax: 302.454.5989

Brandywine School District 1311 Brandywine Boulevard Wilmington, DE 19809

ADDITION AND RENOVATION

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DELAWARE

New Castle County, Delaware

200-15704-17001

November 17, 2017

ADDITION AND RENOVATIONS AT THE CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DELAWARE

PROJECT TEAM

Wilmington, DE 19809

Tetra Tech, Inc.

240 Continental

Suite 200 Newark, I Phone: 3

Brandywine School District

1311 Brandywine Boulevard

8-7551

OWNER:

ARCHITECT:

MEP ENGINEER:

Furhw Associates, Inc. 206 Society Drive Claymont, DE 19703 Phone: 302-798-3515

CIVIL ENGINEER:

ERVICE DESIGN:

CDA Engineering, Inc. 6 Larch Avenue, Suite 401 Wilmington, DE 19804 Phone: 302-998-9202

Corsi Associates, LLC 1489 Baltimore Pike, Suite 109 Springfield, PA 19064 Phone: 610-541-0822

Fax: 302-454-5989



211

SECTION 00 01 10

PROJECT MANUAL

BIDDING AND CONTRACT REQUIREMENTS

PROJECT TEAM

TABLE OF CONTENTS

LIST OF DRAWINGS

State of Delaware Wage Rates

PROJECT FORWARD

INSTRUCTIONS TO BIDDERS

PROJECT SCHEDULE

Documents to be Submitted with Bid BID FORM BID BOND (State Form)

CONTRACT FORMS

STANDARD FORM OF AGREEMENT NETWEEN OWNER AND CONTRACTOR AIA DOCUMENT A101, 2007 EDITION Supplement to Contract for Construction A. 01-2007

Performance Bond and Payment Lond (State Forms) Change Order (AIA Document (70)) Application and Certification for Payment (AIA Documents G702 & G703) Certificate of Substantial completion (AIA Document G704) Contractor's Afficience Payment of Debts and Claims (AIA Document G706) Contractor's Afficience Payment of Debts and Claims (AIA Document G706) Contractor's Afficience Payment (AIA Document G707) Consent of Surety of Final Payment (AIA Document G707) Architect's Supplemental Instructions (AIA Document G710) Construction Change Directive (AIA Document G714) Supplemental Attachment for *ACORD* Certificate of Insurance (AIA Document G715)

OND TIONS OF THE CONTRACT

AIA Document A201, 2007 Edition

Supplementary General Conditions

Additional Supplementary Conditions

^yC

General Requirements PROJECT MANUAL (Continued)

Drug Testing Forms

DIVISION 01 – General Requirement

| 01 10 00 | Summary |
|-------------|-------------------------------------|
| 01 21 00 | Allowances |
| 01 22 00 | Unit Prices |
| 012300 | Alternates |
| 012500 | Substitution Procedures |
| 012600 | Contract Modifications Procedures |
| 012900 | Payment Procedures |
| 013100 | Project Management and Coordination |
| 013120 | Payroll Reports |
| 013200 | Construction Progress |
| 013300 | Submittal Procedures |
| 013301 | CADD Release |
| 015000 | Temporary Facilities and Controls |
| 017329 | Cutting and Patching |
| 017700 | Closeout Procedures |
| 018000 | Schedule of Special Inspections |
| DIVISION 03 | CONCRETE |
| | |
| 03 30 00 | Cast-In-Place Concrete |
| DIVISION 04 | |

DIVISION 03- CONCRETE

Cast Stone Ma

DIVISION 04- MASONRY

04 72 00

DIVISION 05- META

| 05 | 12 | 00 | |
|----|----|----|--|
| 05 | 31 | 00 | |
| 05 | 40 | 00 | |
| 05 | 50 | 00 | |
| 05 | 51 | JU | |
| 05 | 52 | 00 | |
| 6 | 52 | 1 | |
| | | | |

1 Decking old-Formed Metal Framing letal Fabrication Metal Stairs Metal Railings

uctural Steel Framing

Pipe Tube Railings

VISION 06- WOOD, PLASTICS AND COMPOSITES

- 06 10 00 Rough Carpentry
- 06 13 23 Heavy Timber Construction
- 06 16 00 Sheathing
- 06 40 23 Interior Architectural Cabinets
- Plastic-Laminate-Faced Architectural Cabinets 06 41 16

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

DIVISION 07 – THERMAL AND MOISTURE PROTECTION

- 07 21 00 Thermal Insulation
- 07 31 13 Asphalt Shingles
- 07 62 00 Sheet Metal Flashing and Trim
- **Roof Specialties** 07 71 00
- 07 72 00 **Roof Accessories**
- 07 84 13 Penetration Firestopping
- 07 84 46 **Fire-Resistive Joint Systems**
- 07 92 00 Joint Sealants

DIVISION 08 – OPENINGS

- 08 11 13 Hollow Metal Doors and Frames
- 08 14 16 Flush Wood Doors
- 08 16 13 **Fiberglass Doors**
- 08 31 16 Access Doors and Frames
- Seri 08 41 13 Aluminum-Faced Entrances and Storefronts
- 08 80 00 Glazing
- 08 90 00 Louvers and Vents

DIVISION 09 - FINISHES

- 09 00 00 **Finish Schedule**
- 09 30 13 Ceramic Tile
- 09 51 13 Acoustical Panel Ceiling
- Resilient Base and Acq 09 65 13
- **Resilient Tile Floori** 09 65 19
- Tile Carpeting 09 68 13
- 09 91 23 Interior Paintin

DIVISION 12 – FURNISH NGS

12 24 13 12 36 23.13

Vindow Shades rer stic-L minate-Clad Countertops

DIVISION 21 FIRE SUPPRESSION

21 01 70

Fire Protection Systems

IVISION 22- PLUMBING

- 00 00General Provisions – Plumbing/Fire Protection
- 22 00 10 Basic Materials and Methods - Plumbing
- 22 00 30 Insulation & Covering – Plumbing
- 22 01 10 Drainage Systems – Plumbing
- Domestic Water Systems Plumbing 22 01 20
- Fixtures Plumbing 22 01 40
- Equipment Plumbing 22 01 50

| 22 01 90 | Testing – Plumbing |
|----------|--------------------|
|----------|--------------------|

22 01 91 Balancing - Plumbing

DIVISION 23- HEATING, VENTILATION, AND AIR CONDITIONING

- 23 02 00 General Provisions HVAC
- 23 02 10 Basic Materials and Methods HVAC
- 23 02 15 Valves
- 23 02 30 Insulation & Covering HVAC
- 23 04 10 Heating Generation Auxiliary Equipment
- 23 04 50 Refrigeration Equipment HVAC
- 23 04 51 Variable Refrigerant Flow Split System Heat Recovery with Simultaneous Helting & Cooling
- 23 05 10 Water Treatment HVAC
- 23 06 00 Air Distribution & Accessories HVAC
- 23 06 05 Fans
- 23 07 25 Terminal Heating Units
- 23 07 30 Terminal Heating and Cooling Equipment
- 23 07 60 Air Handling Equipment
- 23 09 00 Automatic Temperature Controls (DDC)
- 23 09 50 Testing & Balancing of Mechanical Systems

DIVISION 26- ELECTRICAL

- 26 00 00 General Provisions Electrical
- 26 00 55 Electrical Identification
- 26 01 10 Raceways
- 26 01 20 Wires and Cables
- 26 01 35 Electrical Boxes & Fi
- 26 01 40 Wiring Devices
- 26 01 55 Motor Starters
- 26 01 60 Panelboards
- 26 01 70 Motor and Sircuit Disconnects
- 26 04 52 Grounding

26 06 12

2607.3

- 26 04 70 Distribution Circuits
- 26 04 71 Feeder Orcuits
- 26 04 72 Brach circuits
- 26 06 01 Lightning Protection Systems
 - Imergency Generator Systems
 - Wireless Clock System
 - Heating Terminals

VISION 31- EARTHWORK

| 31 10 00 | Site Clearing |
|----------|---------------|
| 31 20 00 | Earth Moving |

31 23 19 Dewatering

DIVISION 32- EXTERIOR IMPROVEMENTS

32 12 16 Asphalt Paving Restoration

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- 32 13 13 Portland Cement Concrete Paving
- 32 22 10 Erosion and Sediment Control

DIVISION 33- UTILITIES

- 33 25 10 Water Distribution
- 33 41 00 Storm Drainage & Sanitary Sewer



Str

SECTION 00 01 15

LIST OF DRAWINGS

GENERAL

| G-000 | COVER SHEET |
|-------|---------------|
| G-001 | CODE ANALYSIS |

CIVIL

| C-01 | CIVIL COVER SHEET |
|------|--|
| C-02 | SITE DEMOLITION & PRE-CONSTRUCTION STORMWATER MANAGEMENT |
| | PLAN |
| C-03 | CONSTRUCTION SITE STORMWATER MANAGEMENT PLAN |
| C-04 | CONSTRUCTION SITE NOTES & DETAILS |
| C-05 | CONSTRUCTION SITE NOTES & DETAILS |
| C-06 | GENERAL DEVELOPMENT LINES & GRADES NAN |
| C-07 | SITE CONSTRUCTION, LAYOUT & UTILITY PLAN |
| C-08 | SITE CONSTRUCTION DETAILS |
| C-09 | SITE CONSTRUCTION DETAILS |
| | |

ARCHITECTURAL

| A-001 | ABBREVIATIONS, L | EGENDS & | GÉNERAL NOTES |
|--------|------------------|----------|---------------|
| 11 001 | | | |

- AD-101 FIRST FLOOR DEMOLITION PLAN
- A-100
- A-101
- OVERALL FIRST FLOOR NEW CONSTRUCTION PLAN FIRST FLOOR NEW CONSTRUCTION PLAN LOWER LEVEL DEMOLITION & NEW CONSTRUCTION PLAN A-102
- ROOF PLAN DRUOLITION & NEW WORK A-103
- FIRST FLOOR NEW REFLECTED CEILING PLAN A-104
- A-201 KTEK OV **ELEVATIONS & BUILDING SECTIONS** EX
- A-302 **LESECTIONS**
- TERIOR ELEVATIONS A-401
- A-402 GED PLANS & ELEVATIONS
- ENLARGED STAIT PLANS & DETAILS A-403
- **NARTITION TYPES & DETAILS** A-501
 - DOOR SCHEDULE, WINDOW & DOOR DETAILS
 - FIRST FLOOR FINISH PLAN & SCHEDULE

URAL

| S-001 | STRUCTURAL GENERAL NOTES |
|-------|--|
| S-101 | STRUCTURAL FOUNDATION & FRAMING PLANS |
| S-301 | STRUCTURAL FOUNDATION SECTIONS & DETAILS |
| S-302 | STRUCTURAL FRAMING SECTIONS & DETAILS |
| S-501 | STRUCTURAL FOUNDATION DETAILS |
| S-502 | STRUCTURAL FRAMING ELEVATIONS & DETAILS |

MECHANICAL

- LEGEND, NOTES, & DETAILS MECHANICAL M-100
- MD-101 FIRST & SECOND FLOOR PLANS DEMOLITION- MECHANICAL
- M-101 **BASEMENT DEMOLITION/NEW WORK – MECHANICAL**
- M-102 FIRST FLOOR PLAN – MECHANICAL DUCTWORK
- M-103 FIRST FLOOR PLAN – MEHANICAL PIPING
- M-501 DETAILS – MECHANICAL
- M-502 DETAILS – MECHANICAL
- AIR FLOW DIAGRAMS & ATC SEQUENCE OF OPERATIONS MECHANIC M-503
- SCHEDULES MECHANICAL M-601

FIRE PROTECTION

| FP-100 | LEGEND, NOTES, & DETAILS – FIRE PROTECTION |
|--------|--|
| FP-101 | FIRST FLOOR PLAN DEMOLITION - FIRE PROTECT |

- FIRST FLOOR PLAN-FIRE PROTECTION FP-102

PLUMBING

| P-100 | LEGEND, SCHEDULE, & DETAILS - PLUMBING |
|-------|---|
| P-101 | BASEMENT DEMOLITION/NEW WORK – PLOMBING |
| P-102 | FIRST FLOOR DEMOLITION PLAN - PLYMBIAG |
| P-103 | FIRST FLOOR PLAN – PLUMBING |
| | |

ELECTRICAL



- ED-101 LEGEND, NOTES, & BASILATE DEMOLITION PLAN – ELECTRICAL
- FIRST FLOOR PLAN DEMONITION ELECTRICAL ED-102
- ED-103
- FIRST FLOOR PLANS DEMOLITION ELECTRICAL SECOND FLOOR PLAN DEMOLITION ELECTRICAL ED-104
- E-101 BASEMENT PLAC - FLECTRICAL
- LEGEND, SCHEDULE, & FIRST FLOOR PLAN LIGHTING E-102
- FIRST FLOOR PLAN LIGHTING E-103
- **NE FLOOR PLAN LIGHTING** E-104 SEC
- FIRET FLOOR PLAN POWER E-105
- FLOOR PLANS POWER E-106
- E-107 HEDULE, DETAIL, & SECOND FLOOR PLAN – POWER

END OF SECTION

SECTION 00 73 46

DELAWARE PREVAILING WAGE RATES

State of Delaware, Department of Labor, Prevailing Wages. Provided in accordance with Delaware' Prevailing Wage Regulations.

<



STATE OF DELAWARE DEPARTMENT OF LABOR DIVISION OF INDUSTRIAL AFFAIRS OFFICE OF LABOR LAW ENFORCEMENT PHONE: (302) 451-3423

Mailing Address: 4425 North Market St., 3rd FL Wilmington, DE 19802 Located at: 4425 North Market St., 3rd FL Wilmington, DE 19802

PREVAILING WAGES FOR BUILDING CONSTRUCTION EFFECTIVE MARCH 15, 2017

| CLASSIFICATION | NEW CASTLE | KENT | SUSSEA |
|-----------------------------------|------------|--------|--------|
| ASBESTOS WORKERS | 22.86 | 28.16 | 40.98 |
| BOILERMAKERS | 68.44 | 34.72 | 51.05 |
| BRICKLAYERS | 51.99 | 51.99 | 51.99 |
| CARPENTERS | 53.81 | 53 81) | 42.77 |
| CEMENT FINISHERS | 72.28 | 46. 4 | 22.17 |
| ELECTRICAL LINE WORKERS | 45,47 | 3.99 | 29.73 |
| ELECTRICIANS | 66.85 | 66.85 | 66.85 |
| ELEVATOR CONSTRUCTORS | 90.49 | 64.49 | 31.94 |
| GLAZIERS | 71.20 | 71.20 | 56.66 |
| INSULATORS | 55.48 | 55.48 | 55.48 |
| IRON WORKERS | 62.85 | 62.85 | 62.85 |
| LABORERS | 44.79 | 44.70 | 44.70 |
| MILLWRIGHTS | 69.18 | 69.18 | 55.75 |
| PAINTERS | 8.47 | 48,47 | 48.47 |
| PILEDRIVERS | 75.27 | 39.35 | 31.83 |
| PLASTERERS | 29.84 | 29.84 | 22.12 |
| PLUMBERS/PIPEFITTERS/STEAMFITTERS | 65.95 | 51.49 | 57.01 |
| POWER EQUIPMENT OPERATORS | 67.29 | 67.29 | 43.83 |
| ROOFERS-COMPOSITION | 24.01 | 23.70 | 21.64 |
| ROOFER\$-SHINGLE/SLATE/TILE | 18.39 | 21.86 | 17.19 |
| SHEET METAL WORKERS | 67.03 | 67.03 | 67.03 |
| SOFT FLOOR LAYERS | 51.12 | 51.12 | 51.12 |
| SPRINKLER FITTERS | 57.29 | 57.29 | 57.29 |
| TERRAZZO/MARBLE/TILL PAR. | 57.72 | 57.72 | 47.51 |
| TERRAZZO/MARBLE TIL STRS | 66.02 | 66.02 | 55.02 |
| TRUCK DRIVERS | 28.75 | 27.44 | 28.94 |

CERTIFIED: 09/20/20/7

BY ADMINIS LAW ENFORCEMENT CE

THESE RATES ARE PROMULGATED AND ENFORCED PURSUANT TO THE PREVAILING WAGE REGULATIONS ADOPTED BY THE DEPARTMENT OF LABOR ON APRIL 3, 1992.

CLASSIFICATIONS OF WORKERS ARE DETERMINED BY THE DEPARTMENT OF LABOR. FOR ASSISTANCE IN CLASSIFYING WORKERS, OR FOR A COPY OF THE REGULATIONS OR CLASSIFICATIONS, PHONE (302) 761-8200

NON-REGISTERED APPRENTICES MUST BE PAID THE MECHANIC'S RATE.

PROJECT: Carrcroft Elementary School Addition and Renovations , New Castle County

SECTION 00 22 13

INSTRUCTION TO BIDDERS

OMORUN

TABLE OF ARTICLES

- 1. DEFINITIONS
- 2. BIDDER'S REPRESENTATION
- 3. BIDDING DOCUMENTS
- 4. BIDDING PROCEDURES
- 5. CONSIDERATION OF BIDS
- 6. POST-BID INFORMATION
- 7. PERFORMANCE FOND AND PAYMENT BOND

FORM OF AGREEMENT BETWEEN OWNER AND CONTRACTOR

8.

ARTICLE 1: GENERAL

- 1.1 DEFINITIONS
- 1.1.1 Whenever the following terms are used, their intent and meaning shall be interpreted as follows:
- 1.2 STATE: The State of Delaware.
- 1.3 AGENCY: Contracting State Agency as noted on cover sheet.
- 1.4 DESIGNATED OFFICIAL: The agent authorized to act for the Agency.
- 1.5 BIDDING DOCUMENTS: Bidding Documents include the Bidding Pequirements and the proposed Contract Documents. The Bidding Requirements consist of the Advertisement for Bid, Invitation to Bid, Instructions to Bidders, Supplementer instructions to Bidders (if any), General Conditions, Supplementary General Conditions, General Requirements, Special Provisions (if any), the Bid Form (including the Non-collusion Statement), and other sample bidding and contract forms. The proposel Contract Documents consist of the form of Agreement between the Owner and Contractor, as well as the Drawings, Specifications (Project Manual) and all Addanta issued prior to execution of the Contract.
- 1.6 CONTRACT DOCUMENTS: The Centret Documents consist of the, Instructions to Bidders, Supplementary Instruction, Bidders (if any), General Conditions, Supplementary General Conditions, General Requirements, Special Provisions (if any), the form of agreement between the Owner and the Contractor, Drawings (if any), Specifications (Project Manual), an all addenda.
- 1.7 AGREEMENT: The form of the Agreement shall be AIA Document A101, Standard Form of Agreement between Owner and Contractor where the basis of payment is a STIPULATED SULT. In the case of conflict between the instructions contained therein and the General Requirements herein, these General Requirements shall prevail.
- 1.8 GENERAL REQUIREMENTS (or CONDITIONS): General Requirements (or conditions) are instructions pertaining to the Bidding Documents and to contracts in general. They contain, in summary, requirements of laws of the State; policies of the Agency and instructions to bidders.



SPECIAL PROVISIONS: Special Provisions are specific conditions or requirements peculiar to the bidding documents and to the contract under consideration and are supplemental to the General Requirements. Should the Special Provisions conflict with the General Requirements, the Special Provisions shall prevail.

ADDENDA: Written or graphic instruments issued by the Owner/Architect prior to the execution of the contract which modify or interpret the Bidding Documents by additions, deletions, clarifications or corrections.

- 1.11 BIDDER OR VENDOR: A person or entity who formally submits a Bid for the material or Work contemplated, acting directly or through a duly authorized representative who meets the requirements set forth in the Bidding Documents.
- 1.12 SUB-BIDDER: A person or entity who submits a Bid to a Bidder for materials or labor, or both for a portion of the Work.
- 1.13 BID: A complete and properly executed proposal to do the Work for the sums stipulated therein, submitted in accordance with the Bidding Documents.
- 1.14 BASE BID: The sum stated in the Bid for which the Bidder offers to perform the Work described in the Bidding Documents as the base, to which Work may be added or from which Work may be deleted for sums stated in Alternate Bids (if such are required to be stated in the bid).
- 1.15 ALTERNATE BID (or ALTERNATE): An amount stated in the Did, where applicable, to be added to or deducted from the amount of the Base Bid if the corresponding change in the Work, as described in the Bidding Documents is accepted.
- 1.16 UNIT PRICE: An amount stated in the Bid, where applicable, as a price per unit of measurement for materials, equipment or services or aportion of the Work as described in the Bidding Documents.
- 1.17 SURETY: The corporate body which is bound with and for the Contract, or which is liable, and which engages to be responsible for the Contractor's payments of all debts pertaining to and for his acceptable performance of the Work for which he has contracted.
- 1.18 BIDDER'S DEPOSIT: Intersecurity designated in the Bid to be furnished by the Bidder as a guaranty of good faith to enter into a contract with the Agency if the Work to be performed or the material prequipment to be furnished is awarded to him.
- 1.19 CONTRACT: The written agreement covering the furnishing and delivery of material or work to be performed.
- 1.20 CONTRACTOR: Any individual, firm or corporation with whom a contract is made by the Agency.
- 1.21

SUBCONTRACTOR: An individual, partnership or corporation which has a direct ontract with a contractor to furnish labor and materials at the job site, or to perform construction labor and furnish material in connection with such labor at the job site.

CONTRACT BOND: The approved form of security furnished by the contractor and his surety as a guaranty of good faith on the part of the contractor to execute the work in accordance with the terms of the contract.

ARTICLE 2: BIDDER'S REPRESENTATIONS

2.1 PRE-BID MEETING

- 2.1.1 A pre-bid meeting for this project will be held at the time and place designated. Attendance at this meeting is a pre-requisite for submitting a Bid, unless this requirement is specifically waived elsewhere in the Bid Documents.
- 2.2 By submitting a Bid, the Bidder represents that:
- 2.2.1 The Bidder has read and understands the Bidding Documents and that the Bid is made in accordance therewith.
- 2.2.2 The Bidder has visited the site, become familiar with existing conditions under ranch the Work is to be performed, and has correlated the Bidder's his personal observations with the requirements of the proposed Contract Documents.
- 2.2.3 The Bid is based upon the materials, equipment, and systems required by the Bidding Documents without exception.
- 2.3 JOINT VENTURE REQUIREMENTS
- 2.3.1 For Public Works Contracts, each Joint Venturer shall be qualified and capable to complete the Work with their own forces.
- 2.3.2 Included with the Bid submission, and as a requirement to bid, a copy of the executed Joint Venture Agreement shall be submitted and same a by all Joint Venturers involved.
- 2.3.3 All required Bid Bonds, Performance Bonds, Material and Labor Payment Bonds must be executed by both Joint Venturers and se placed in both of their names.
- 2.3.4 All required insurance certificate shall name both Joint Venturers.
- 2.3.5 Both Joint Ventuers scale and the Bid Form and shall submit a copy of a valid Delaware Business Licence waarheir Bid.
- 2.3.6 Both Joint Venturers shall include their Federal E.I. Number with the Bid.
 - In the event of a mandatory Pre-bid Meeting, each Joint Venturer shall have a representative in attendance.



2.3.7

Due to exceptional circumstances and for good cause shown, one or more of these provisions may be waived at the discretion of the State.

ASSIGNMENT OF ANTITRUST CLAIMS

As consideration for the award and execution by the Owner of this contract, the Contractor hereby grants, conveys, sells, assigns and transfers to the State of Delaware all of its right, title and interests in and to all known or unknown causes of action it presently has or may now or hereafter acquire under the antitrust laws of the United States and the State of Delaware, relating to the particular goods or services purchased or acquired by the Owner pursuant to this contract.

ARTICLE 3: BIDDING DOCUMENTS

3.1 COPIES OF BID DOCUMENTS

- 3.1.1 Bidders may obtain complete sets of the Bidding Documents from the Architectural/Engineering firm designated in the Advertisement or Invitation to Bid in the number and for the deposit sum, if any, stated therein.
- 3.1.2 Bidders shall use complete sets of Bidding Documents for preparation of Bids. The suine Agency nor the Architect assumes no responsibility for errors or misinterpretations resulting from the use of incomplete sets of Bidding Documents.
- 3.1.3 Any errors, inconsistencies or omissions discovered shall be reperted to the Architect immediately.
- 3.1.4 The Agency and Architect may make copies of the Bidding Decements available on the above terms for the purpose of obtaining Bids on the Work. No license or grant of use is conferred by issuance of copies of the Bidding Documents.
- 3.2 INTERPRETATION OR CORRECTION OF BIDLUG DOCUMENTS
- 3.2.1 The Bidder shall carefully study and compare the Bidding Documents with each other, and with other work being bid concurrently or presently under construction to the extent that it relates to the Work for which the Bid is submitted, shall examine the site and local conditions, and shall report any protections inconsistencies, or ambiguities discovered to the Architect.
- 3.2.2 Bidders or Sub-bidders equivalent clarification or interpretation of the Bidding Documents shall make a written request to the Architect at least seven days prior to the date for receipt of Bids. Interpretations, corrections and changes to the Bidding Documents will be made by written Addandare, Interpretations, corrections, or changes to the Bidding Documents made in any other marker shall not be binding.
- 3.2.3 The apparent elence of the specifications as to any detail, or the apparent omission from it of detailed description concerning any point, shall be regarded as meaning that only the best commercial practice is to prevail and only material and workmanship of the first quality are to be used. Proof of specification compliance will be the responsibility of the Bidder.



Unless otherwise provided in the Contract Documents, the Contractor shall provide and pay for all permits, labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services necessary for the proper execution and completion of the Work.

- The Owner will bear the costs for all impact and user fees associated with the project.
- 3.3 SUBSTITUTIONS
- 3.3.1 The materials, products and equipment described in the Bidding Documents establish a standard of quality, required function, dimension, and appearance to be met by any

proposed substitution. The specification of a particular manufacturer or model number is not intended to be proprietary in any way. Substitutions of products for those named will be considered, providing that the Vendor certifies that the function, quality, and performance characteristics of the material offered is equal or superior to that specified. It shall be the Bidder's responsibility to assure that the proposed substitution will not affect the intent of the design, and to make any installation modifications required accommodate the substitution.

- 3.3.2 Requests for substitutions shall be made in writing to the Architect at least ten deus poor to the date of the Bid Opening. Such requests shall include a complete description of the proposed substitution, drawings, performance and test data, explanation of required installation modifications due the substitution, and any other information necessary for an evaluation. The burden of proof of the merit of the proposed substitution is upon the proposer. The Architect's decision of approval or disapproval shall be made. The Architect is to notify Owner prior to any approvals.
- 3.3.3 If the Architect approves a substitution prior to the receipt of Bals, such approval shall be set forth in an Addendum. Approvals made in any other manner shall not be binding.
- 3.3.4 The Architect shall have no obligation to consider any substitutions after the Contract award.
- 3.4 ADDENDA

INSTRUCTIONS TO BIDDER

3.4.1 Addenda will be mailed or delivered to all who are known by the Architect to have received a complete set of the Brading Documents.

00 21 13-5

- 3.4.2 Copies of Addeno will be made available for inspection wherever Bidding Documents are
 - on file for thit purpose.
 - No Add no, will be issued later than 4 days prior to the date for receipt of Bids except an Addendum withdrawing the request for Bids or one which extends the time or changes the location for the opening of bids.



4.1

3.4.3

Each bidder shall ascertain prior to submitting his Bid that they have received all Addenda issued, and shall acknowledge their receipt in their Bid in the appropriate space. Not acknowledging an issued Addenda could be grounds for determining a bid to be non-responsive.

ICLE 4: BIDDING PROCEDURES

- PREPARATION OF BIDS
- 4.1.1 Submit the bids on the Bid Forms included with the Bidding Documents.

- 4.1.2 Submit the original Bid Form for each bid. Bid Forms may be removed from the project manual for this purpose.
- 4.1.3 Execute all blanks on the Bid Form in a non-erasable medium (typewriter or manually in ink).
- 4.1.4 Where so indicated by the makeup on the Bid Form, express sums in both words and figures, in case of discrepancy between the two, the written amount shall govern.
- 4.1.5 Interlineations, alterations or erasures must be initialed by the signer of the Bid
- 4.1.6 BID ALL REQUESTED ALTERNATES AND UNIT PRICES, IF ANY. In there is no change in the Base Bid for an Alternate, enter "No Change". The Centra tor is responsible for verifying that they have received all addenda issued during the birding period. Work required by Addenda shall automatically become part of the Connect.
- 4.1.7 Make no additional stipulations on the Bid Form and do not qualify the Bid in any other manner.
- 4.1.8 Each copy of the Bid shall include the legal name of the Bidder and a statement whether the Bidder is a sole proprietor, a partnership, a corporation, or any legal entity, and each copy shall be signed by the person or persons legally authorized to bind the Bidder to a contract. A Bid by a corporation shall further give the state of incorporation and have the corporate seal affixed. A Bid submitted by an event hall have a current Power of Attorney attached, certifying agent's authority to bind the Bidder.
- 4.1.9 Bidder shall complete the Nen-Collusion Statement form included with the Bid Forms and include it with their Bid
- 4.1.10 In the construction of all Paolic Works projects for the State of Delaware or any agency thereof, preference memployment of laborers, workers or mechanics shall be given to bona fide legal citizens of the State who have established citizenship by residence of at least 90 days in the state.
- 4.1.11 Each bride, shall include in their bid a copy of a valid Delaware Business License.'

4.1.12



Each bidder shall include signed Affidavit(s) for the Bidder and each listed Subcontractor certifying compliance with OMB Regulation 4104- "Regulations for the Drug Testing of Contractor and Subcontractor Employees Working on "Large Public Works Projects." "Large Public Works" is based upon the current threshold required for bidding Public Works as set by the Purchasing and Contracting Advisory Council.

BID SECURITY

All bids shall be accompanied by a deposit of either a good and sufficient bond to the agency for the benefit of the agency, with corporate surety authorized to do business in this State, the form of the bond and the surety to be approved by the agency, or a security of the bidder assigned to the agency, for a sum equal to at least 10% of the bid plus all add alternates, or in lieu of the bid bond a security deposit in the form of a certified check, bank treasurer's check, cashier's check, money order, or other prior approved secured deposit

assigned to the State. The bid bond need not be for a specific sum, but may be stated to be for a sum equal to 10% of the bid plus all add alternates to which it relates and not to exceed a certain stated sum, if said sum is equal to at least 10% of the bid. The Bid Bond form used shall be the standard OMB form (attached).

- 4.2.2 The Agency has the right to retain the bid security of Bidders to whom an award is being considered until either a formal contract has been executed and bonds have been furnished or the specified time has elapsed so the Bids may be withdrawn or all Bids have been rejected.
- 4.2.3 In the event of any successful Bidder refusing or neglecting to execute a formal contract and bond within 20 days of the awarding of the contract, the bid bond of security deposited by the successful bidder shall be forfeited.
- 4.3 SUBCONTRACTOR LIST
- 4.3.1 As required by <u>Delaware Code</u>, Title 29, section 6962(d)(10)b, each Bidder shall submit with their Bid a completed List of Sub-Contractors included with the Bid Form. NAME ONLY ONE SUBCONTRACTOR FOR EACH TRADE. A Bid will be considered nonresponsive unless the completed list is included.
- 4.3.2 Provide the Name and Address for each litted strucontractor. Addresses by City, Town or Locality, plus State, will be acceptable.
- 4.3.3 It is the responsibility of the connector to ensure that their Subcontractors are in compliance with the provisions or his law. Also, if a Contractor elects to list themselves as a Subcontractor for any category, they must specifically name themselves on the Bid Form and be able to document them capability to act as Subcontractor in that category in accordance with this law.

4.4 EQUALITY OF EVEN OYMENT OPPORTUNITY ON PUBLIC WORKS

4.4.1 During the performance of this contract, the contractor agrees as follows:

The Contractor will not discriminate against any employee or applicant for employment because of race, creed, sex, color, sexual orientation, gender identity or national origin. The Contractor will take affirmative action to ensure the applicants are employed, and that employees are treated during employment, without regard to their race, creed, sex, color, sexual orientation, gender identity or national origin. Such action shall include, but not be limited to, the following: Employment, upgrading, demotion or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. The Contractor agrees to post in conspicuous places available to employees and applicants for employment notices to be provided by the contracting agency setting forth this nondiscrimination clause.

B. The Contractor will, in all solicitations or advertisements for employees placed by or on behalf of the Contractor, state that all qualified applicants will receive

consideration for employment without regard to race, creed, sex, color, sexual orientation, gender identity or national origin."

4.5 PREVAILING WAGE REQUIREMENT

- 4.5.1 Wage Provisions: For renovation and new construction projects whose costs exceed the thresholds contained in <u>Delaware Code</u>, Title 29, Section 6960, the minimum wage rates for various classes of laborers and mechanics shall be as determined by the Department of Labor, Division of Industrial Affairs of the State of Delaware.
- 4.5.2 The employer shall pay all mechanics and labors employed directly upon the site of work, unconditionally and not less often than once a week and without subsequent eduction or rebate on any account, the full amounts accrued at time of payment, computed a wage rates not less than those stated in the specifications, regardless of any contractual relationship which may be alleged to exist between the employer and such laborer and mechanics.
- 4.5.3 The scale of the wages to be paid shall be posted by the employer in a prominent and easily accessible place at the site of the work.
- 4.5.4 Every contract based upon these specifications shall omtain a stipulation that sworn payroll information, as required by the Department of Labor, be furnished weekly. The Department of Labor shall keep and maintain the sworn payroll information for a period of 6 months from the last day of the work week covered by the payroll.
- 4.6 SUBMISSION OF BIDS
- 4.6.1 Enclose the Bid, the Bid Security, and any other documents required to be submitted with the Bid in a sealed opaque envelope. Address the envelope to the party receiving the Bids. Identify with the project name, project number, and the Bidder's name and address. If the Bid is sent by mail, enclose the sealed envelope in a separate mailing envelope with the notation "BID aNULOSED" on the face thereof. The State is not responsible for the opening of bids prover bid opening date and time that are not properly marked.
- 4.6.2 Deposit Bias at the designated location prior to the time and date for receipt of bids indicated in the Advertisement for Bids. Bids received after the time and date for receipt of bids will be marked "LATE BID" and returned.



4.7

Bidder assumes full responsibility for timely delivery at location designated for receipt of ids.

Oral, telephonic or telegraphic bids are invalid and will not receive consideration.

Withdrawn Bids may be resubmitted up to the date and time designated for the receipt of Bids, provided that they are then fully in compliance with these Instructions to Bidders.

MODIFICATION OR WITHDRAW OF BIDS

4.7.1 Prior to the closing date for receipt of Bids, a Bidder may withdraw a Bid by personal request and by showing proper identification to the Architect. A request for withdraw by letter or fax, if the Architect is notified in writing prior to receipt of fax, is acceptable. A

fax directing a modification in the bid price will render the Bid informal, causing it to be ineligible for consideration of award. Telephone directives for modification of the bid price shall not be permitted and will have no bearing on the submitted proposal in any manner.

- 4.7.2 Bidders submitting Bids that are late shall be notified as soon as practicable and the bid shall be returned.
- 4.7.3 A Bid may not be modified, withdrawn or canceled by the Bidder during a thirty (3 day period following the time and date designated for the receipt and opening of Pies, and Bidder so agrees in submitting their Bid. Bids shall be binding for 30 days after the date of the Bid opening.

ARTICLE 5: CONSIDERATION OF BIDS

- 5.1 OPENING/REJECTION OF BIDS
- 5.1.1 Unless otherwise stated, Bids received on time will be publicly opened and will be read aloud. An abstract of the Bids will be made available to Bidders.
- 5.1.2 The Agency shall have the right to reject any and all Bids. A Bid not accompanied by a required Bid Security or by other data required by the Bidding Documents, or a Bid which is in any way incomplete or irregular is subject to rejection.
- 5.1.3 If the Bids are rejected, it will be done within thirty (30) calendar day of the Bid opening.
- 5.2 COMPARISON OF BIDS
- 5.2.1 After the Bids have been opered and read, the bid prices will be compared and the result of such comparisons will be made available to the public. Comparisons of the Bids may be based on the Bise on plus desired Alternates. The Agency shall have the right to accept Alternates in any order or combination.
- 5.2.2 The Agency reserves the right to waive technicalities, to reject any or all Bids, or any portion hereof, to advertise for new Bids, to proceed to do the Work otherwise, or to abandon he Work, if in the judgment of the Agency or its agent(s), it is in the best interest of the state.



An increase or decrease in the quantity for any item is not sufficient grounds for an increase or decrease in the Unit Price.

The prices quoted are to be those for which the material will be furnished F.O.B. Job Site and include all charges that may be imposed during the period of the Contract.

No qualifying letter or statements in or attached to the Bid, or separate discounts will be considered in determining the low Bid except as may be otherwise herein noted. Cash or separate discounts should be computed and incorporated into Unit Bid Price(s).

5.3 DISQUALIFICATION OF BIDDERS

- 5.3.1 An agency shall determine that each Bidder on any Public Works Contract is responsible before awarding the Contract. Factors to be considered in determining the responsibility of a Bidder include:
 - A. The Bidder's financial, physical, personnel or other resources including Subcontracts;
 - B. The Bidder's record of performance on past public or private construction projects, including, but not limited to, defaults and/or final adjudication or admission of violations of the Prevailing Wage Laws in Delaware or any other state;
 - C. The Bidder's written safety plan;
 - D. Whether the Bidder is qualified legally to contract with the State;
 - E. Whether the Bidder supplied all necessry information concerning its responsibility; and,
 - F. Any other specific criteria for a particular procurement, which an agency may establish; provided however, that the criteria be set forth in the Invitation to Bid and is otherwise in conformity via State and/or Federal law.
- 5.3.2 If an agency determines that a Bidder is nonresponsive and/or nonresponsible, the determination shall be in writing and set forth the basis for the determination. A copy of the determination shall be sent to the affected Bidder within five (5) working days of said determination.
- 5.3.3 In addition, any one of more of the following causes may be considered as sufficient for the disqualification of a Didder and the rejection of their Bid or Bids.
- 5.3.3.1 More that one Bid for the same Contract from an individual, firm or corporation under the same of lifetone names.

Evidence of collusion among Bidders.



5.3.3.2

Unsatisfactory performance record as evidenced by past experience.

If the Unit Prices are obviously unbalanced either in excess or below reasonable cost analysis values.

If there are any unauthorized additions, interlineation, conditional or alternate bids or irregularities of any kind which may tend to make the Bid incomplete, indefinite or ambiguous as to its meaning.

5.3.3.6 If the Bid is not accompanied by the required Bid Security and other data required by the Bidding Documents.

5.3.3.7 If any exceptions or qualifications of the Bid are noted on the Bid Form.

5.4 ACCEPTANCE OF BID AND AWARD OF CONTRACT

- 5.4.1 A formal Contract shall be executed with the successful Bidder within twenty (20) calendar days after the award of the Contract.
- 5.4.2 Per Section 6962(d)(13) a., Title 29, Delaware Code, "The contracting agency shall award any public works contract within thirty (30) days of the bid opening to the lowest responsive and responsible Bidder, unless the Agency elects to award on the basis of beer value, in which case the election to award on the basis of best value shall be stated in the Invitation To Bid."
- 5.4.3 Each Bid on any Public Works Contract must be deemed responsive by the agency to be considered for award. A responsive Bid shall conform in all material respects to the requirements and criteria set forth in the Contract Documents and specifications.
- 5.4.4 The Agency shall have the right to accept Alternates in any order or combination, and to determine the low Bidder on the basis of the sum of the base Bd, plus accepted Alternates.
- 5.4.5 The successful Bidder shall execute a formal couract, submit the required Insurance Certificate, and furnish good and sufficient bonds, unless specifically waived in the General Requirements, in accordance with the General nequirement, within twenty (20) days of official notice of contract award. The successful Bidder shall provide two business days prior to contract execution, copies to the employee Drug Testing Program for the Bidder and all listed Subcontractors. Bonds shall be for the benefit of the Agency with surety in the amount of 100% of the total contract award. Said Bonds shall be conditioned upon the faithful performance of the contract. Bonds shall remain in affect for period of one year after the date of substantial completion.
- 5.4.6 If the successful Bidder tails to execute the required Contract, Bond and all required information, as the estic, within twenty (20) calendar days after the date of official Notice of the Award of the Contract, their Bid guaranty shall immediately be taken and become the property of the State for the benefit of the Agency as liquidated damages, and not as a forfeiture or as a penalty. Award will then be made to the next lowest qualified Bidder of the Wort orreadvertised, as the Agency may decide.
- 5.4.7



Each older shall supply with its bid its taxpayer identification number (i.e., federal employer identification number or social security number) and a copy of its Delaware business license, and should the vendor be awarded a contract, such vendor shall provide to the agency the taxpayer identification license numbers of such subcontractors. Such numbers shall be provided on the later of the date on which such subcontractor is required to be identified or the time the contract is executed. The successful Bidder shall provide to the agency to which it is contracting, within 30 days of entering into such public works contract, copies of all Delaware Business licenses of subcontractors and/or independent contractors that will perform work for such public works contract. However, if a subcontractor or independent contractor is hired or contracted more than 20 days after the Bidder entered the public works contract the Delaware Business license of such subcontractor or independent contractor shall be provided to the agency within 10 days of being contracted or hired.

5.4.8 The Bid Security shall be returned to the successful Bidder upon the execution of the formal contract. The Bid Securities of unsuccessful bidders shall be returned within thirty (30) calendar days after the opening of the Bids.

ARTICLE 6: POST-BID INFORMATION

- 6.1 CONTRACTOR'S QUALIFICATION STATEMENT
- 6.1.1 Bidders to whom award of a Contract is under consideration shall, if requested by the Agency, submit a properly executed AIA Document A305, Contractor's Qualification Statement, unless such a statement has been previously required and submitted.
- 6.2 BUSINESS DESIGNATION FORM
- 6.2.1 Successful bidder shall be required to accurately complete an Office of Management and Budget Business Designation Form for Subcontractors.

ARTICLE 7: PERFORMANCE BOND AND PAYMENT BODD

7.1 BOND REQUIREMENTS

- 7.1.1 The cost of furnishing the required Bords, that are stipulated in the Bidding Documents, shall be included in the Bid.
- 7.1.2 If the Bidder is required by the Agency to secure a bond from other than the Bidder's usual sources, changes in cost will be adjusted as provide in the Contract Documents.
- 7.1.3 The Performance and Payment Bond forms used shall be the standard OMB forms (attached).
- 7.2 TIME OF DELIVERY AND FORM OF BONDS
- 7.2.1 The bonds analybe dated on or after the date of the Contract.
- 7.2.2

The Diader shall require the attorney-in-fact who executes the required bonds on behalf of the surety to affix a certified and current copy of the power of attorney.

8: FORM OF AGREEMENT BETWEEN AGENCY AND CONTRACTOR

Unless otherwise required in the Bidding Documents, the Agreement for the Work will be written on AIA Document A101, Standard Form of Agreement Between Owner and Contractor Where the Basis of Payment is a Stipulated Sum.

END OF SECTION

PAGE LLY QNA INTENT ANK

SECTION 00 31 13

PROJECT SCHEDULE

1.1 PROJECT SCHEDULE

- A. This Document with its referenced attachments is part of the Contract Documents. The surrand finish dates shown on the schedule are subject to change, but the construction duration day are a requirement of the project. The schedule duration days are business days.
- B. Available Project information includes the following:
 - 1. Project Schedule (attached).
- C. Project schedule includes construction milestones as appended to this Document. Related Requirements:
 - 1. Section 01 10 00 "Summary" for phased construction equirements.

END OF

2. Section 01 32 00 "Construction Progress Documentation" for Contractor's construction schedule requirements.

PAGE INTENT NK





| For Bids Due: (DATE) | To: | (OWNER) |
|--|--|--|
| | | C |
| | | |
| | | \sim |
| Name of Bidder: | | |
| Delaware Business License No.: | Taxpayer 3 | ID No.: |
| (A copy of Bidder's Delaware Busines | ss License must be attached to this form.) | |
| | | \mathbf{N} |
| (Other License Nos.): | | |
| Dhone Not (| For to . | |
| Filolie No.: () | Fax 10.: |) |
| The undersigned, representing that he therewith, that he has visited the site an and that his bid is based upon the mate proposes and agrees to provide all labor work described by the aforesaid docume \$(\$ | has read and understands the Eldding Doc id has familiarized himselfiwith the local con erials, systems and equipment described in th or, materials, plant, equipment, supplies, tran ents for the lump som temized below: | cuments and that this bid is made in accordance additions under which the Work is to be performed, the Bidding Documents without exception, hereby insport and other facilities required to execute the |
| <u>ALTERNATES</u> | 2~ | |
| Alternate prices conform to applicable following Alternates. An 'ADD" or D | EDUCT" amount is indicated by the crossed | specifications for a complete description of the out part that does not apply. |
| ALTERNATE No. 1: | (BRIEF DESCRIPTI | ON) |
| Add/Deduct: | | |
| (\$ |) | |
| ALTERNATE No. 2: | (BRIEF DESCRIPTIC | ON) |
| | | |
| Add/Deluct: | | |

)

Tetra Tech 200-15704-17001

(\$

| | BID FORM | | L'S |
|--|---|--------------------------------|---------------------------|
| UNIT PRICES | | | S ^v |
| Unit prices conform to applicabl Unit Prices: | e project specification section. Refer to the spe | ecifications for a complements | cription of the following |
| UNIT PRICE No. 1: | (BRIEF DESCRIPTION) | _ \$\$ | β |
| UNIT PRICE No. 2: | (BRIEF DESCRIPTION) | | δ |
| UNIT PRICE No. 3: | (BRIEF DESCRIPTION) | \$\$ | δ |
| | Schult | | |

Tetra Tech 200-15704-17001

BID FORM

I/We acknowledge Addendums numbered ______ and the price(s) submitted include any cost/schedule impact they may have

This bid shall remain valid and cannot be withdrawn for thirty (30) days from the date of opening of bids (60 days for School/Districts and Department of Education), and the undersigned shall abide by the Bid Security forfeiture provisions. Bid security cattached to this Bid.

The Owner shall have the right to reject any or all bids, and to waive any informality or irregularity in any bid received.

This bid is based upon work being accomplished by the Sub-Contractors named on the list attached o this bid.

Should I/We be awarded this contract, I/We pledge to achieve substantial completion of all the wards within ______ calendar days of the Notice to Proceed.

The undersigned represents and warrants that he has complied and shall comply with all requirements of local, state, and national laws; that no legal requirement has been or shall be violated in making or accepting this bid, in awarding the contract to him or in the prosecution of the work required; that the bid is legal and firm; that he has not, crectly or indirectly, entered into any agreement, participated in any collusion, or otherwise taken action in restraint of free competitive bidding.

Upon receipt of written notice of the acceptance of this Bid, the Bidder shall, within twenty (20) calendar days, execute the agreement in the required form and deliver the Contract Bonds, and Insurance Certificates, required by the Contract Documents.

| I am / We are an Individual / a Partnership / a Corporation | |
|---|------------------------|
| By | Trading as |
| (Individual's / General Partner's / Corporate same) | |
| (State of Corporation) | |
| Business Address: | |
| | |
| | |
| Witness: | By: |
| | (Authorized Signature) |
| (SEAL) | (Title) |
| \sim | Date: |
| N | |
| | |
| AT A CHMENTS Sub Contractor List | |
| Non-Collusion Statement | |
| Affidavit(s) of Employee Drug Testing Program | |
| Bid Security | |
| (Others as Required by Project Manuals) | |

Tetra Tech 200-15704-17001

Str

BID FORM

SUBCONTRACTOR LIST

In accordance with Title 29, Chapter 6962 (d)(10)b <u>Delaware Code</u>, the following sub-contractor listing must according the bid submittal. The name and address of the subcontractor **must be listed for each category** where the bidder intends to use a sub-contractor to perform that category of work. In order to provide full disclosure and acceptance of the bid by the *Owner*, it is **required that bidders list themselves as being the sub-contractor for all ategories where he/she is qualified and intends to perform such work.** This form must be filled out completely with no additions or deletions. Note that all subcontractors listed below must have a signed Affidavit of Employee Drug Testing Program included with this bid.

| Subcontractor Category | <u>Subcontractor</u> | <u>Address/City & State)</u> | Subcontractors tax payer ID # or Delaware Business license # |
|-------------------------|--|----------------------------------|---|
| 1. | | | |
| 2. | | | |
| 3. | | Y | |
| 4. | \begin{aligned} & & & & & & \\ & & & & & & & & & \\ & \\ & \\ & \\ & \\ & \\ & \\ & \\ & \\ & \\ & \\ & \\ & \\ & \\ & \\ & \\ & & & & & & & & \\ & & & & & & & & & \\ & & & & & & & & & \\ & & & & & & & & & \\ & & & & & & & & & \\ & & & & & & & & & \\ & & & & & & & & & \\ & & & & & & & & & \\ & & & & & & & & & \\ & & & & & & & & & \\ & & & & & & & & & \\ & & & & & & & & & \\ & \\ & \\ & \\ & \\ & \\ & \\ & & | ► | |
| 5. | | | |
| 6. | | | |
| 7. | | | |
| | O | | |
| BID FORM 00 41 13 -4 | | Tetra Tec 200-15704-1700 | h 1 |

BID FORM

NON-COLLUSION STATEMENT

This is to certify that the undersigned bidder has neither directly nor indirectly, entered into any agreement, participated in any collusion or otherwise taken any action in restraint of free competitive bidding in connection with this poposal submitted this date (to the Office of Management and Budget, Division of Facilities Management).

All the terms and conditions of (*Project or Contract Number*) have been thoroughly examined and are understood.

| <u> </u> | | |
|---------------------------------------|--------|--|
| • • • • • • • • • • • • • • • • • • • | | |
| day of | 20 | |
| | | |
| | day of | |

THIS PAGE MUST BE SIGNED AND NOTARIZED FOR YOUR BID TO BE CONSIDERED.
CARRCROFT ELEMENTARY SCHOOL ADDITION AND RENOVATIONS BRANDYWINE SCHOOL DISTRICT WILMINTON, DE 19809 200-15704-17001

| | AFFIDAVIT |
|---|---|
| EMPLO | OYEE DRUG TESTING PROGRAM |
| 4104 Regulations for the Drug Testing of Contractor that Contractors and Subcontractors implement a pr Contracts funded all or in part with public funds. | or and Subcontractor Employees Working on Large Public Werks Projects requors and Subcontractory drug testing for Employees who work on Large Public Wo |
| We hereby certify that we have in place or will implored for our employees on the jobsite that complies with | ement during the entire term of the contract a Manuatory Drug Testing Program this regulation: |
| Contractor/Subcontractor Name: | |
| Contractor/Subcontractor Address: | |
| | <u>()</u> |
| | |

Authorized Representative (typed or printed):

Authorized Representative (signature):

Sworn to and Subscribed before me

Title:

_ NOTARY PUBLIC _____ My Commission expires

nis

SPAGE MUST BE SIGNED AND NOTARIZED FOR YOUR BID TO BE CONSIDERED.

day of

_____20_____.

requires

ublic Works

STATE OF DELAWARE

BID BOND

| | TO ACCOM (Not necessar | PANY PROPOSAL y if security is used) | |
|--|---|--|--|
| WNOW ALL MEN | I DV THESE DDESENT | S That: | |
| KNOW ALL MEN | of | S I fial: | he County of |
| | 01 | 11 U | he County of |
| and State | 10 | as Principa | li, and |
| | 01 | In the Coun | |
| | as Surety , legall | y authorized to do busir | less in the Sale of Delaw |
| ("State"), are held and firm | nly unto the State in the | sum of | |
| D | ollars (\$ |), or per | cent not to exceed |
| | (NT | | |
| of amount of bid on Contra | ict No | , to be part | to the State for the use |
| benefit of | 1 1 . 1 . 1 | (insert State agen | cy name) for which paym |
| well and truly to be made, | , we do bind ourselves, | our and each of our held | executors, administrators, |
| successors, jointly and seve | erally for and in the whol | le firmly by these presents. | |
| who has submitted to the certain proposal to enter in State , shall be awarded thi Contract as may be required | nto this contract for the is Contract, and if said if d by the terms of this C <i>nsert State agency same</i> of the award chereol if else to be anoremain in eal andwated thes | (ins furnishing of certain mate trinsipal shall well and t nthect and approved by the othis Contract to be enter- accordance with the ter full force and virtue. day of | sert State agency name) erial and/or services within ruly enter into and execute ee ed into within twenty days a rms of said proposal, then r |
| | | Name of Bio | lder (Organization) |
| | | | |
| Corporate | By: | | |
| Seal | • — | Author | ized Signature |
| | | | C |
| | | | Title |
| | | | T MC |
| • | | Nan | ne of Surety |
| | | | |
| Witness: | D | | |
| | By: | | |
| | By: | | |
| | By: | | Title |

• PAGE LLY INTENT LANK 4

SECTION 00 52 13

STANDARD FORM OF AGREEMENT BETWEEN OWNER AND CONTRACTOR

The Standard Form of Agreement Between Owner and Contractor is as stated in the American Institute of Architects Document AIA A101 (2007 Edition) entitled <u>Standard Form of Agreement Percent Owner and</u> <u>Contractor</u> and is part of this project manual as if herein written in full.

Republick

A draft copy of this document is included herein as follows.



$\mathbf{W} \mathbf{AIA}^{\circ}$ Document A101TH – 2017

nation)

Standard Form of Agreement Between Owner and Contractor where the basis of payment is a Stipulated Sum

AGREEMENT made as of the day of in the year (In words, indicate day, month and year.)

BETWEEN the Owner: (Name, legal status, address and other information)

and the Contractor: DIMCF (Name, legal status, address and other information)

for the following Project: (Name, location and detailed description)

The Architect: (Name, legal status, address and

The Owner agree as follows. ADDITIONS AND r this do The author added infor eded for its com nor may also hav revised the text of the original ndar form. An Additions and etions Report that notes added formation as well as revisions to standard form text is available rom the author and should be reviewed. A vertical line in the left margin of this document indicates where the author has added necessary information and where the author has added to or deleted from the original AIA text.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

The parties should complete A101[™]–2017, Exhibit A, Insurance and Bonds, contemporaneously with this Agreement. AIA Document A201[™]–2017, General Conditions of the Contract for Construction, is adopted in this document by reference. Do not use with other general conditions unless this document is modified.

AIA Document A101TM - 2017. Copyright © 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1967, 1974, 1977, 1987, 1991, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 13:32:16 on 05/10/2017 under Order No. 3987348344 which expires on 01/23/2018, and is not for resale. User Notes:

TABLE OF ARTICLES

- THE CONTRACT DOCUMENTS 1
- 2 THE WORK OF THIS CONTRACT
- 3 DATE OF COMMENCEMENT AND SUBSTANTIAL COMPLETION
- CONTRACT SUM
- PAYMENTS 5
- **DISPUTE RESOLUTION**
- TERMINATION OR SUSPENSION 7
- **MISCELLANEOUS PROVISIONS** 8
- ENUMERATION OF CONTRACT DOCUMENTS

EXHIBIT A INSURANCE AND BONDS

ARTICLE 1 THE CONTRACT DOCUMENTS

The Contract Documents consist of this Agreement, Conditions the Contract (General, Supplementary, and other Conditions), Drawings, Specifications, Addenda issued prior to exect the of this Agreement, other documents listed in this Agreement, and Modifications issued after execution of this Agreement, all of which form the Contract, and are as fully a part of the Contract as if attached to this Agreement or repeated herein. The Contract represents the entire and integrated agreement between the parties herein and supersedes prior negotiations, representations, or agreements, either written or oral. An enumeration of the Contract Documents, other than a Modification, appears in Article 9.

ARTICLE 2 THE WORK OF THIS CONTRAC

The Contractor shall fully execute the W ibed in the Contract Documents, except as specifically indicated in rk desc the Contract Documents to be the response sibility of others.

ARTICLE 3 DATE OF COMMENC MF ND SUBSTANTIAL COMPLETION d D

§ 3.1 The date of commencement of Work shall be: he (Check one of the follow xes.)

- The date of ment.
- A date th in notice to proceed issued by the Owner. et fo
- ed as follows: Establi

(Insert a date or a means to determine the date of commencement of the Work.)

te of commencement of the Work is not selected, then the date of commencement shall be the date of this Agreement.

§ 3.2 The Contract Time shall be measured from the date of commencement of the Work.

§ 3.3 Substantial Completion

§ 3.3.1 Subject to adjustments of the Contract Time as provided in the Contract Documents, the Contractor shall achieve Substantial Completion of the entire Work:

AIA Document A101™ – 2017. Copyright © 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1967, 1974, 1977, 1987, 1991, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 13:32:16 on 05/10/2017 under Order No. 3987348344 which expires on 01/23/2018, and is not for resale. **User Notes:**

1

Init.

2

 \mathbf{C}

(Check one of the following boxes and complete the necessary information.)

Not later than () calendar days from the date of commencement of the Work.

] By the following date:

§ 3.3.2 Subject to adjustments of the Contract Time as provided in the Contract Documents, if portions of the Work are to be completed prior to Substantial Completion of the entire Work, the Contractor shall achieve Substantial Completion of such portions by the following dates:

Portion of Work Substantial Completion Date

§ 3.3.3 If the Contractor fails to achieve Substantial Completion as provided in this Section 3.3, li uidate damages, if any, shall be assessed as set forth in Section 4.5.

ARTICLE 4 CONTRACT SUM

§ 4.1 The Owner shall pay the Contractor the Contract Sum in current funds for the C performance of the ntr ctor Contract. The Contract Sum shall be (\$), subject to additions and deductions in the Contract Documents.

§ 4.2 Alternates

§ 4.2.1 Alternates, if any, included in the Contract Sum:

Item

§ 4.2.2 Subject to the conditions noted below, the following alternates may be accepted by the Owner following execution of this Agreement. Upon acceptance, the Owner Shan as a Modification to this Agreement. (be net for the Owner to accept the alternate.) (Insert below each alternate and the conditions that mu

Price



Init. 1

AlA Document A101™ – 2017. Copyright © 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1967, 1974, 1977, 1987, 1991, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 13:32:16 on 05/10/2017 under Order No. 3987348344 which expires on 01/23/2018, and is not for resale. User Notes:

ARTICLE 5 PAYMENTS § 5.1 Progress Payments

§ 5.1.1 Based upon Applications for Payment submitted to the Architect by the Contractor and Certificates for Payment issued by the Architect, the Owner shall make progress payments on account of the Contract Sum to the Contractor as provided below and elsewhere in the Contract Documents.

§ 5.1.2 The period covered by each Application for Payment shall be one calendar month ending on the last day of the month, or as follows:

§ 5.1.3 Provided that an Application for Payment is received by the Architect not later than the day of Owner shall make payment of the amount certified to the Contractor not later than the day of the onth. If Application for Payment is received by the Architect after the application date fixed above, payment of b e amount certified shall be made by the Owner not later than () days after the Architect receives the licatio for Payment.

(Federal, state or local laws may require payment within a certain period of time.)

§ 5.1.4 Each Application for Payment shall be based on the most recent schedule of nitted by the Contractor in accordance with the Contract Documents. The schedule of values shall all cate the entire Contract Sum among the various portions of the Work. The schedule of values shall be repared in such form, and supported by such data to substantiate its accuracy, as the Architect may require. The ile f values shall be used as a basis for reviewing the Contractor's Applications for Payment.

§ 5.1.5 Applications for Payment shall show the percentage of completion of each portion of the Work as of the end of the period covered by the Application for Payment.

§ 5.1.6 In accordance with AIA Document A201[™]–2017. General Conditions of the Contract for Construction, and subject to other provisions of the Contract Documenta, the amount of each progress payment shall be computed as follows:

§ 5.1.6.1 The amount of each progress payment s all fir t include:

- .1 That portion of the Contract fum properly allocable to completed Work;
 .2 That portion of the Contract fum properly allocable to materials and equipment delivered and suitably stored at the site for subsequent accorporation in the completed construction, or, if approved in advance by the Owner suitably stored off the site at a location agreed upon in writing; and
- That portion of Converction Change Directives that the Architect determines, in the Architect's .3 professional judgment to be reasonably justified.
- § 5.1.6.2 The amount of Coch propress payment shall then be reduced by:
 - greate of any amounts previously paid by the Owner; The a .1
 - The mount of any, for Work that remains uncorrected and for which the Architect has previously .2 with eld Certificate for Payment as provided in Article 9 of AIA Document A201–2017;
 - Approximation amount for which the Contractor does not intend to pay a Subcontractor or material supplier, pless the Work has been performed by others the Contractor intends to pay;
 - For Work performed or defects discovered since the last payment application, any amount for which the Architect may withhold payment, or nullify a Certificate of Payment in whole or in part, as provided in Article 9 of AIA Document A201-2017; and
 - .5 Retainage withheld pursuant to Section 5.1.7.

§ 5.1.7 Retainage

5.1.7.1 For each progress payment made prior to Substantial Completion of the Work, the Owner may withhold the following amount, as retainage, from the payment otherwise due:

(Insert a percentage or amount to be withheld as retainage from each Application for Payment. The amount of retainage may be limited by governing law.)

Init 1

AIA Document A101™ – 2017. Copyright © 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1967, 1974, 1977, 1987, 1991, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 13:32:16 on 05/10/2017 under Order No 3987348344 which expires on 01/23/2018, and is not for resale. User Notes:

§ 5.1.7.1.1 The following items are not subject to retainage:

(Insert any items not subject to the withholding of retainage, such as general conditions, insurance, etc.)

§ 5.1.7.2 Reduction or limitation of retainage, if any, shall be as follows:

(If the retainage established in Section 5.1.7.1 is to be modified prior to Substantial Completion of the entire Work, including modifications for Substantial Completion of portions of the Work as provided in Section 3.3.2, insel provisions for such modifications.)

§ 5.1.7.3 Except as set forth in this Section 5.1.7.3, upon Substantial Completion of the Work, the Contra r mav submit an Application for Payment that includes the retainage withheld from prior Applications Payn ent ction shan not include pursuant to this Section 5.1.7. The Application for Payment submitted at Substantial Comp retainage as follows:

(Insert any other conditions for release of retainage upon Substantial Completion.

§ 5.1.8 If final completion of the Work is materially delayed through no f tractor, the Owner shall pay the Contractor any additional amounts in accordance with Article 9 of A AΓ cument A201-2017.

§ 5.1.9 Except with the Owner's prior approval, the Contractor shar not make advance payments to suppliers for materials or equipment which have not been delivered and store at the ite.

§ 5.2 Final Payment

§ 5.2.1 Final payment, constituting the entire unpaid balance of the Contract Sum, shall be made by the Owner to the Contractor when

- .1 the Contractor has fully performed are Contract except for the Contractor's responsibility to correct Work as provided in Article 12 of AIA I becument A201–2017, and to satisfy other requirements, if any, which extend beyond final protection it; and
- .2 a final Certificate for Payme t has been issued by the Architect.

§ 5.2.2 The Owner's final paymen to be Contractor shall be made no later than 30 days after the issuance of the Architect's final Certificate for P ent. or as follows:

§ 5.3 Interest

Payments due and unpaid under the Contract shall bear interest from the date payment is due at the rate stated below, or in the e ab. nce t rereof, at the legal rate prevailing from time to time at the place where the Project is located.

terest agreed upon, if any.) (Insert rate of

ARTICLE 6 DISPUTE RESOLUTION 6.1 nitial Decision Maker

The Architect will serve as the Initial Decision Maker pursuant to Article 15 of AIA Document A201–2017, unless the parties appoint below another individual, not a party to this Agreement, to serve as the Initial Decision Maker. (If the parties mutually agree, insert the name, address and other contact information of the Initial Decision Maker, if other than the Architect.)

AIA Document A101[™] – 2017. Copyright © 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1967, 1974, 1977, 1987, 1991, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 13:32:16 on 05/10/2017 under Order No. 3987348344 which expires on 01/23/2018, and is not for resale. User Notes:

§ 6.2 Binding Dispute Resolution

For any Claim subject to, but not resolved by, mediation pursuant to Article 15 of AIA Document A201-2017, the method of binding dispute resolution shall be as follows: (Check the appropriate box.)

- [] Arbitration pursuant to Section 15.4 of AIA Document A201-2017
- Litigation in a court of competent jurisdiction
- Other (Specify) []

If the Owner and Contractor do not select a method of binding dispute resolution, or do not sequently agree in writing to a binding dispute resolution method other than litigation, Claims will be reolve I by litigation in a court of competent jurisdiction.

ARTICLE 7 TERMINATION OR SUSPENSION

§ 7.1 The Contract may be terminated by the Owner or the Contractor as icle 14 of AIA Document A201-2017.

§ 7.1.1 If the Contract is terminated for the Owner's convenience in accordance with Article 14 of AIA Document A201-2017, then the Owner shall pay the Contractor a terminat on fre. follows: (Insert the amount of, or method for determining, the fee, if ayable to the Contractor following a termination for the Owner's convenience.)

§ 7.2 The Work may be suspended by the Owner ided in Article 14 of AIA Document A201-2017. s pro

ARTICLE 8 MISCELLANEOUS PROVISIO

§ 8.1 Where reference is made in this A to a provision of AIA Document A201–2017 or another Contract ion as amended or supplemented by other provisions of the Contract Document, the reference refers to pro Documents.

§ 8.2 The Owner's repres nta ve: (Name, address, email a d other information)

he Contractor's representative: , address, email address, and other information)

Init. 1

AlA Document A101™ – 2017. Copyright © 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1967, 1974, 1977, 1987, 1991, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 13:32:16 on 05/10/2017 under Order No. 3987348344 which expires on 01/23/2018, and is not for resale. **User Notes:**

§ 8.4 Neither the Owner's nor the Contractor's representative shall be changed without ten days' prior notice to the other party.

§ 8.5 Insurance and Bonds

§ 8.5.1 The Owner and the Contractor shall purchase and maintain insurance as set forth in AIA Document A101TM 2017, Standard Form of Agreement Between Owner and Contractor where the basis of payment is a Stipulated Sun Exhibit A, Insurance and Bonds, and elsewhere in the Contract Documents.

§ 8.5.2 The Contractor shall provide bonds as set forth in AIA Document A101™–2017 Exhibit A, and elsev the Contract Documents.

§ 8.6 Notice in electronic format, pursuant to Article 1 of AIA Document A201–2017, may be give with AIA Document E203TM–2013, Building Information Modeling and Digital Data Exhibit, if omple d, or as otherwise set forth below:

(If other than in accordance with AIA Document E203–2013, insert requirements for delivering n electronic format such as name, title, and email address of the recipient and whether and how the will be required to generate a read receipt for the transmission.)

§ 8.7 Other provisions:

ARTICLE 9 ENUMERATION OF CONTRACT DOCUMENTS

§ 9.1 This Agreement is comprised of the following document

- AIA Document A101^{IM}-2017, Standard Form of Agreement Between Owner and Contractor
 AIA Document A101^{IM}-2017, Exhibit A. In Agrance and Bonds
 AIA Document A201^{IM}-2017, Commun.
- AIA Document A201[™]–2017, General Conditions of the Contract for Construction
- AIA Document E203[™]–2013, Building 4 formation Modeling and Digital Data Exhibit, dated as indicated below:

(Insert the date of the E20 orated into this Agreement.)



Portions of Addenda relating to bidding or proposal requirements are not part of the Contract Documents unless the bidding or proposal requirements are also enumerated in this Article 9.

.8 Other Exhibits:

> (Check all boxes that apply and include appropriate information identifying the exhibit where required.)

AIA Document A101™ – 2017. Copyright © 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1967, 1974, 1977, 1987, 1991, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA[®] Document is protected by U.S. Copyright Law and International Treaties. Init. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 13:32:16 on 05/10/2017 under Order No. 3987348344 which expires on 01/23/2018, and is not for resale. **User Notes:**

1

[] AIA Document E204TM–2017, Sustainable Projects Exhibit, dated as indicated below: (Insert the date of the E204-2017 incorporated into this Agreement.)

| Title | Date | Pages | |
|--|---|---|---|
| | Duto | i ugos | |
|] Supplementary and other Condition | ns of the Contract: | | C |
| Document | Title | Date | Pages |
| .9 Other documents, if any, list (List here any additional de Document A201 [™] –2017 p. sample forms, the Contract requirements, and other inj proposals, are not part of t documents should be listed | ed below: focuments that are intended to for rovides that the advertisement of tor's bid or proposal, portions of formation furnished by the Own the Contract Documents unless I here only if intended to be part | orm part of the Contrast Do or invitation coold, Instruct of Addendoceloging to bidd ter in unticipation of receiv numerated on this Agreem of the Contract Document | ocuments. AIA tions to Bidders, ling or proposal ing bids or ent. Any such s.) |
| s Agreement entered into as of the day | and year first written above. | | |
| s Agreement entered into as of the day NNER (Signature) | and year first written above. | DR (Signature) | |
| s Agreement entered into as of the day NNER (Signature) rinted name and title) | c and year first written above. | DR (Signature) ne and title) | |
| is Agreement entered into as of the day NNER (Signature) 'rinted name and title) | and year first written above. | DR (Signature) ne and title) | |
| is Agreement entered into as of the day NNER (Signature) 'rinted name and title) | and year first written above. | DR (Signature) ne and title) | |
| is Agreement entered into as of the day NNER (Signature) 'rinted name and title) | contractor (Printed nar | DR (Signature) ne and title) | |
| is Agreement entered into as of the day WNER (Signature) Printed name and title) | contractor (Printed nar | DR (Signature) ne and title) | |
| is Agreement entered into as of the day NNER (Signature) Printed name and title) | contractor (Printed nar | DR (Signature) ne and title) | |
| is Agreement entered into as of the day WNER (Signature) Printed name and title) | and year first pritten above. | DR (Signature) ne and title) | |
| is Agreement entered into as of the day WNER (Signature) Printed name and title) | and year first pritten above. | DR (Signature) ne and title) | |

AlA Document A101[™] – 2017. Copyright © 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1967, 1974, 1977, 1987, 1991, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AlA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AlA[®] Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 13:32:16 on 05/10/2017 under Order No. 3987348344 which expires on 01/23/2018, and is not for resale. User Notes:

SPEC SECTION 00 54 13

SUPPLEMENT TO AGREEMENT BETWEEN OWNER AND CONTRACTOR A101-2007

The following supplements modify the "Standard Form of Agreement Between Owner and Contractor, Document A101-2007. Where a portion of the Standard Form of Agreement is modified or deleted following, the unaltered portions of the Standard Form of Agreement shall remain in effect.

ARTICLE 5: PAYMENTS

5.1 PROGRESS PAYMENTS

5.1.3 Delete paragraph 5.1.3 in its entirety and replace with the following:

"Provided that a valid Application for Payment is received by the Architect that meets all requirements of the Contract, payment shall be nade by the Owner not later than 30 days after the Owner receives the valid Application for Payment."

ARTICLE 6: DISPUTE RESOLUTION

6.2 BINDING DISPUTE RESOLVTION

Check Other – and add the following sentence:

"Any remedies cyalal le in law or in equity."

ARTICLE 8: MISCELL ANEOUS PROVISIONS

8.2

rt the following:



Payments are due 30 days after receipt of a valid Application for Payment. After that 30 day period, interest may be charged at the rate of 1% per month not to exceed 12% per annum."

Delete paragraph 8.5 in its entirety and replace with the following:

"The Contractor's representative shall not be changed without ten days written notice to the Owner."

END OF SUPPLEMENT TO AGREEMENT BETWEEN OWNER AND CONTRACTOR



CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

STATE OF DELAWARE OFFICE OF MANAGEMENT AND BUDGET

PERFORMANCE BOND

| | Bond Number: | 2 |
|--|---|---|
| KNOW ALL PERSONS BY THESE PRESENTS, that | we, as principal | |
| (" Principal "), and , a | corporation, legally | |
| authorized to do business in the State of Delaware, as s | surety ("Surety"), are held and "inny bound | |
| unto the | ("Owner") (invert Sate agency | |
| name), in the amount of(\$ |), to be pair to Owner , for which | |
| payment well and truly to be made, we do bind ourse | elves, our and each and every of our heirs, | |
| executors, administrations, successors and assigns, jo | bintly and severally, for and in the whole, | |
| firmly by these presents. | | |
| | | |
| Sealed with our seals and dated this day of | 20 . | |
| | | |
| NOW THE CONDITION OF THIS OBLIGATION | SUCH, that if Principal , who has been | |
| awarded by Owner that certain contract known | as Contract No. dated the | |
| day of . 20 (the "Contro | which Contract is incorporated herein by | |
| reference, shall well and truly provide and furnishan | erials, appliances and tools and perform all | |
| the work required under and pursuant to the terms and | conditions of the Contract and the Contract | |
| Documents (as defined in the Contract) or any change | es or modifications thereto made as therein | |
| provided, shall make good and reimburg. Owner suffic | tient funds to pay the costs of completing the | |
| Contract that Owner may sustain by reason of any fail | lure or default on the part of Principal , and | |
| shall also indemnify and save harmess Owner from al | l costs, damages and expenses arising out of | |
| or by reason of the performance of the Contract and fo | or as long as provided by the Contract: then | |
| this obligation shall be void, otherwise to be and remain | in full force and effect. | |
| | | |
| Surety , for value received invelops stipulates and agree | es, if requested to do so by Owner , to fully | |
| perform and complete the work to be performed u | under the Contract pursuant to the terms. | |
| conditions and covenant, thereof, if for any cause Princ | cipal fails or neglects to so fully perform and | |
| complete such work | | |
| compile succession | | |
| Surety or value received for itself and its successors | and assigns hereby stipulates and agrees that | |
| the obligation of Surety and its bond shall be in no w | av impaired or affected by any extension of | |
| time, multification, omission, addition or change in or | to the Contract or the work to be performed | |
| he euger or by any payment thereunder before the ti | me required therein or by any waiver of any | |

be eurcer, or by any payment thereunder before the time required therein, or by any waiver of any previsions thereof, or by any assignment, subletting or other transfer thereof or of any work to be performed or any monies due or to become due thereunder; and **Surety** hereby waives notice of any and all such extensions, modifications, omissions, additions, changes, payments, waivers, assignments, subcontracts and transfers and hereby expressly stipulates and agrees that any and all things done and omitted to be done by and in relation to assignees, subcontractors, and other

Tetra Tech 200-15704-17001

PERFORMANCE BOND 00 61 13.13 - 1 transferees shall have the same effect as to **Surety** as though done or omitted to be done by or in relation to **Principal**.

Surety hereby stipulates and agrees that no modifications, omissions or additions in or to the terms of the Contract shall in any way whatsoever affect the obligation of **Surety** and its bond.

Any proceeding, legal or equitable, under this Bond may be brought in any court of competent jurisdiction in the State of Delaware. Notices to **Surety** or Contractor may be mailed or telivered to them at their respective addresses shown below.

IN WITNESS WHEREOF, **Principal** and **Surety** have hereunto set their hand a d seels, and such of them as are corporations have caused their corporate seal to be hereto affixed and these presents to be signed by their duly authorized officers, the day and year first above writter.

| | PRINCIPAL | 2 |
|-----------------------------|-----------|--------|
| | Name: | |
| Witness or Attest: Address: | | |
| | By: | (SEAL) |
| Name: | Name: | |
| (Corporate Seal) | THE | |
| | SURETY | |
| C | Name: | |
| Witness or Attest: Address | | |
| | By: | (SEAL) |
| Name: | Name: | |
| (Corporate Seal) | Title: | |
| | | |
| \mathcal{V} | | |
| | | |

STATE OF DELAWARE OFFICE OF MANAGEMENT AND BUDGET

PAYMENT BOND

| | Bond Number: |
|--|---|
| KNOW ALL PERSONS BY THESE PRESENTS, that | at we,as principal |
| ("Principal"), and, a | corporation, legally |
| authorized to do business in the State of Delaware, as | surety ("Surety"), are need and firmly bound |
| unto the | ("Owner Vainsert State agency |
| name), in the amount of (\$ | <u>), to be pad to Owner</u> , for which |
| payment well and truly to be made, we do bind ours | selves, our and each and every of our heirs, |
| executors, administrations, successors and assigns, join | the severally for and in the whole firmly |
| by these presents. | |

Sealed with our seals and dated this ______ day of _____, 20__.

NOW THE CONDITION OF THIS OBLIGATION IS SUCH, that if **Principal**, who has been awarded by **Owner** that certain contract knowner Contract No. ______dated the ______dated the ______, 20__ (the "Contract"), which Contract is incorporated herein by reference, shall well and truly pay all and every person termshing materials or performing labor or service in and about the performance of the work unler the Contract, all and every sums of money due him, her, them or any of them, for all such materials, labor and service for which **Principal** is liable, shall make good and reimburse **Owner** sufficient funds to pay such costs in the completion of the Contract as **Owner** may sustain by pason of any failure or default on the part of **Principal**, and shall also indemnify and save nomes **Owner** from all costs, damages and expenses arising out of or by reason of the performance or the Contract and for as long as provided by the Contract; then this obligation shall be roid otherwise to be and remain in full force and effect.

Surety, for value received, for itself and its successors and assigns, hereby stipulates and agrees that the obligation of **Surety** and its bond shall be in no way impaired or affected by any extension of time, modification, omission, addition or change in or to the Contract or the work to be performed thereunder, or by any payment thereunder before the time required therein, or by any waiver of any provisions thereof, or by any assignment, subletting or other transfer thereof or of any work to be performed or any monies due or to become due thereunder; and **Surety** hereby waives notice of any and all such extensions, modifications, omissions, additions, changes, payments, waivers, assignments, subcontracts and transfers and hereby expressly stipulates and agrees that any and all things done and omitted to be done by and in relation to assignees, subcontractors, and other transferees shall have the same effect as to **Surety** as though done or omitted to be done by or in relation to **Principal**.

Surety hereby stipulates and agrees that no modifications, omission or additions in or to the terms of the Contract shall in any way whatsoever affect the obligation of **Surety** and its bond.

Tetra Tech 200-15704-17001

PAYMENT BOND 00 61 13.16 - 1 Any proceeding, legal or equitable, under this Bond may be brought in any court of competent jurisdiction in the State of Delaware. Notices to **Surety** or Contractor may be mailed or delivered to them at their respective addresses shown below.

IN WITNESS WHEREOF, **Principal** and **Surety** have hereunto set their hand and seals, and such of them as are corporations have caused their corporate seal to be hereto affixed and these presents to be signed by their duly authorized officers, the day and year first above written.

| | PRINCIPAL Name: | |
|-----------------------------|--------------------------|--------|
| Witness or Attest: Address: | | 8× |
| Name: (Corporate Seal) | _ By: Name: Title: | (SEAL) |
| Witness or Attest: Address: | Name: | |
| Name: (Corporate Sear) | _ By: Name: Title: | (SEAL) |
| | | |

Marine AIA® Document G701[™] – 2001

Change Order



AlA Document G701[™] – 2001. Copyright © 1979, 1987, 2000 and 2001 by The American Institute of Architects. All rights reserved. WARNING: This AlA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AlA[®] Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AlA software at 13:39:31 on 08/02/2005 under Order No.1000162220_1 which expires on 2/15/2006, and is not for resale. User Notes: (1628415007)

AIA° Document G702^m – 1992

| IO OMMER. | PROJECT: | APPLICATION NO. <u>Distribution to:</u> |
|--|--|---|
| | | PERIOD TO: OWNER: |
| | | |
| FROM | VIA | |
| CONTRACTOR: | ARCHITECT | FIELD: |
| | | OTHER: 🗌 |
| CONTRACTOR'S APPLICATIO | ON FOR PAYMENT | The undersigned Contractor certifies that to the best of the Contractor's knowledge, information |
| Application is made for payment, as shown b Continuation Sheet, AIA Document G703, is | below, in connection with the Contract. s attached. | and belief the work avoid by this Application for Payment has been completed in accordance with the Contrac Documents, that all amounts have been paid by the Contractor for Work for which previous Certificates for Payment were issued and payments received from the Owner, and |
| 1. ORIGINAL CONTRACT SUM | | that current pays ent shown herein is now due. |
| 2. Net change by Change Orders | | CONTRACIOR: |
| 3. CONTRACT SUM TO DATE (Line 1 ± 2) | \$ | By Date: |
| 4. TOTAL COMPLETED & STORED TO DATE (| (Column G on G703) \$ | Some of: |
| 5. RETAINAGE: | | County of: |
| a% of Completed Work | | Subscribed and sworn to before |
| (Column D + E on G703) | \$ | me this day of |
| b. $\underline{\qquad}$ % of Stored Material (Column For G703) | | Notaty Public: |
| Total Poteinage (Lines 5a + 5b or Total in | n Column L of $G703$ | My Commission expires: |
| Total Retainage (Lines 5a + 50 of Total II | | |
| 6. TOTAL EARNED LESS RETAINAGE | | In accordance with the Contract Documents, based on on-site observations and the data comprising |
| (Line 4 Less Line 5 Total) | | this application, the Architect certifies to the Owner that to the best of the Architect's knowledge, |
| (Line 6 from prior Certificate) | •••••••••••••••••••••••••••••••••••••• | information and belief the Work has progressed as indicated, the quality of the Work is in |
| & CLIPPENT DAYMENT DUE | | AMOUNT CERTIFIED. |
| 9 BALANCE TO FINISH INCLUDING RETAINA | AGE | |
| (Line 3 less Line 6) | | (Attach explanation if amount certified differs from the amount applied. Initial all figures on this Application and on the Continuation Sheet that are changed to conform with the amount certified.) |
| CHANGE ORDER SUMMARY | ADDITIONS DEDUC | TIONS ARCHITECT: |
| | | Detai |
| Total changes approved in previous months | by wher \$ \$ | By: Date: |
| Total changes approved in previous months Total approved this Month | by when \$ \$ | By: Date This Certificate is not negotiable. The AMOUNT CERTIFIED is payable only to the Contractor |

AlA Document G702¹ to the pyright © 1953, 1963, 1965, 1978 and 1992 by The American Institute of Architects. All rights reserved. WARNING: This AlA[&] Document is protected by U.S. Copyright Law and International Thaties. Unauthorized reproduction or distribution of this AlA[&] Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible order the law. This document was produced by AlA software at 16:53:47 on 05/26/2005 under Order No.1000162220_3 which expires on 2/15/2006, and is not for resale. User Notes:

| IA Do ontaini n tabul Ise Col | cument G702, APPLICATION Ang Contractor's signed certificat ations below, amounts are stated umn I on Contracts where variab | AND CERTIFICA ion is attached. to the nearest dol ole retainage for li | ATION FOR PAY lar. ne items may app | 'MENT, ly. | | APPLICAT APPLICAT PERICAT ANSHIEC | ION NO. ION DITE: | T NO: | |
|--|--|---|---|------------------------|--|---|---------------------------------------|---------------------------------|------------------------------------|
| A | В | С | D | Е | F | G | | Н | Ι |
| TEM NO. | DESCRIPTION OF WORK | SCHEDULED VALUE | WORK CO FROM PREVIOUS APPLICATION (D + E) | MPLETED THIS PERIOD | MATERIALS PRESENTLY STORED (NOT IN D.OR H | COMPLETED ANI STORED TO DATE (D+E+F) | % (G ÷ C) | BALANCE TO FINISH (C - G) | RETAINAGE (IF VARIABLI RATE) |
| | | | | | \mathcal{G} | | · · · · · · · · · · · · · · · · · · · | | |
| | · · · · · · · · · · · · · · · · · · · | | ~ | O'' | | | | | |
| | | | 0 | | | | | | |
| | | .0 | | | | | | | |
| | GRAND TOTAL | s | \$ | \$ | \$ | \$ | | \$ | \$ |

The Document Gross and International Treaties. Unauthorized reproduction or distribution of this AIA[®] Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 17:14:37 on 05/26/2005 under Order No.1000162220_3 which expires on 2/15/2006, and is not for resale.

Mator and the second s

Certificate of Substantial Completion



AlA Document G704™ – 2000. Copyright © 1963, 1978, 1992 and 2000 by The American Institute of Architects. All rights reserved. WARNING: This AIA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA[®] Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 13:44:37 on 08/02/2005 under Order No.1000162220_1 which expires on 2/15/2006, and is not for resale. User Notes: (2591405616)

Contractor's Affidavit of Payment of Debts and Claims



AIA Document G706[™] – 1994. Copyright © 1970 and 1994 by The American Institute of Architects. All rights reserved. WARNING: This AIA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA[®] Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 13:49:12 on 08/02/2005 under Order No.1000162220_1 which expires on 2/15/2006, and is not for resale. User Notes: (1078981934)

Mate Alter and the second sec

Contractor's Affidavit of Release of Liens

ARCHITECT'S PROJECT NUMBER: **PROJECT:** (Name and address) OWNER ARCHITE CONTRACT FOR: CONTRAC CONTRACT DATED: **TO OWNER:** (Name and address) STATE OF: COUNTY OF: The undersigned hereby certifies that to the best of the undersigned's knowledge, info m non-and belief, except as listed below, the Releases or Waivers of Lien attached hereto include the Contractor all ubcontractors, all suppliers of materials and equipment, and all performers of Work, labor or services who have on nay have liens or encumbrances or the right to assert liens or encumbrances against any proof the owner arising in any manner out of the performance of the Contract referenced above. **EXCEPTIONS:** SUPPORTING DOCUMENTS ATTACHED HERETO TRACTOR: (Name and address) Contractor's Release or Waiver of Liens, 1. conditional upon receipt of final payment, 2. Separate Releases or Waivers of Liens from BY: Subcontractors and material and equip ne (Signature of authorized suppliers, to the extent required v the Owner, representative) accompanied by a list thereof. (Printed name and title) Subscribed and sworn to before me on this date: Notary Public: My Commission Expires:

AlA Document G706A[™] – 1994. Copyright © 1982 and 1994 by The American Institute of Architects. All rights reserved. WARNING: This AlA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AlA[®] Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AlA software at 13:52:05 on 08/02/2005 under Order No.1000162220_1 which expires on 2/15/2006, and is not for resale. User Notes: (669341389)

■AIA^{*} Document G707[™] – 1994

Consent Of Surety to Final Payment



AlA Document G707™ – 1994. Copyright © 1982 and 1994 by The American Institute of Architects. All rights reserved. WARNING: This AlA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AlA[®] Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AlA software at 14:38:45 on 08/02/2005 under Order No.1000162220_1 which expires on 2/15/2006, and is not for resale. User Notes: (2340075428)

$\operatorname{ILA}^{\circ}$ Document G710^{$\circ} – 1992$ </sup>

Architect's Supplemental Instructions

PROJECT (Name and address):

ARCHITECT'S SUPPLEMENTAL INSTRUCTION NO:

OWNER:

FIELD: 🗌 OTHER: 🔲

1

ARCHI CONSUL

CON

OWNER (Name and address):

DATE OF ISSUANCE:

CONTRACT FOR:

CONTRACT DATE:

FROM ARCHITECT (Name and address):

TO CONTRACTOR (Name and address):

ARCHITECT'S PROJECT NUMBER:

The Work shall be carried out in accordance with the following supplementar instructions issued in accordance with the Contract Documents without change in Contract Sum or Contract Time. Proceeding with the Work in accordance with these instructions indicates your acknowledgement that here will be no change in the Contract Sum or Contract Time.

DESCRIPTION:



AlA Document G710TM – 1992. Copyright © 1980 and 1992 by The American Institute of Architects. All rights reserved. WARNING: This AlA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AlA[®] Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AlA software at 14:41:59 on 08/02/2005 under Order No.1000162220_1 which expires on 2/15/2006, and is not for resale. User Notes: (344800501)

$\operatorname{AIA}^{\circ}$ Document G714TM – 2007

Construction Change Directive

| PROJEC1 : (Name and address) | DIRECTIVE NUMBER: DATE: CONTRACT FOR: | |
|---|--|---|
| TO CONTRACTOR: (Name and address) | CONTRACT DATED: ARCHITECT'S PROJECT NUMBER: | |
| You are hereby directed to make the follo (Describe briefly any proposed changes of | owing change(s) in this Contract: or list any attached information in the alte | er ativ |
| PROPOSED ADJUSTMENTS 1. The proposed basis of adjustmen □ • Lump Sum of \$ | nt to the Contract Sum or Guaranced Ma | ximum Price is: |
| \Box • Unit Price of \$ per | . () | |
| \Box • As provided in Section 7. | .3.3 of AIA Document AS21-2007 | |
| \Box • As follows: | | |
| 2. The Contract Time is proposed t | . The proposed adjustment, if any | y, is . |
| When signed by the Owner and Architect and becomes effective IMMEDIATELY as a Con Contractor shall proceed with the change(s) d | received by the Contractor, this document soluction Change Directive (CCD), and the scrient above. | Contractor signature indicates agreement with the proposed adjustments in Contrac Sum and Contract Time set forth in this CCD. |
| ARCHITECT (Firm name) | OWNER (Firm name) | CONTRACTOR (Firm name) |
| ADDRESS | ADDRESS | ADDRESS |
| BY (Signature) | BY (Signature) | BY (Signature) |
| | | |
| (Typea name) | (Typed name) | (Typed name) |

■AIA^{*} Document G715^{**} – 1991

Supplemental Attachment for ACORD Certificate of Insurance 25-S

(This document replaces AIA Document G705, Certificate of Insurance.)

PROJECT (Name and address):

| Does the General Aggregate apply to this I Does this policy include coverage for: a. Premises - Operations? b. Explosion, Collapse and Underground c. Personal Injury Coverage? d. Products Coverage? | Project only? d Hazards? | | 9 | |
|--|--|---|--|---|
| Does this policy include coverage for: a. Premises - Operations? b. Explosion, Collapse and Underground c. Personal Injury Coverage? d. Products Coverage? | d Hazards? | | | _ |
| a. Premises - Operations? b. Explosion, Collapse and Underground c. Personal Injury Coverage? d. Products Coverage? | d Hazards? | | | _ |
| b. Explosion, Collapse and Undergroundc. Personal Injury Coverage?d. Products Coverage? | d Hazards? | | | |
| c. Personal Injury Coverage?d. Products Coverage? | | لسا | | |
| d. Products Coverage? | | | | |
| | | | | |
| e. Completed Operations? | | | | |
| f. Contractual Coverage for the Insured | 's obligations in A221? | | | |
| If coverage is written on a claims-made ba | usis, what is the: | | | |
| a. Retroactive Date? | | | | |
| b. Extended Reporting Date? | | | | |
| rker's Compensation | | | | |
| If the Insured is exempt from Worker's Co | omponsation statutes, does the Insured | _ | | - |
| carry the equivalent Voluntary Compensat | tion coverage? | | | LJ |
| Ts this certificate being furnished in come | ection with the Contractor's request for | | | |
| final payment in accordance with the requ | irements of Sections 9.10.2 and 11.1.3 of | | | |
| AIA Document A201, General Condition | f the Contract for Construction? | | | |
| If so, and if the policy period extends bey | ond termination of the Contract for | | | |
| Construction, is Completed Operations co | verage for this Project continued for the | | | 1 |
| balance of the policy period? | | | لسا | |
| Has each policy shown on the certificate a | and this Supplement been endorsed to | | | |
| provide the holder with 30 days notice of | cancellation and/or expiration? List below | | | |
| any policies which do not contain this not | ice. | | | |
| er Provisions | | | | |
| | | | | |
| | | | | |
| | Authorized Representative | | | |
| | a. Retroactive Date? b. Extended Reporting Date? rker's Compensation If the Insured is exempt from Worker's Cocarry the equivalent Voluntary Compensatial Payment Information Is this certificate being furnished in connectinal payment in accordance with the rout AIA Document A201, General condition If so, and if the policy period extends by Construction, is Completed Operators cobalance of the policy period? mination Provisions Has each policy shown on the certificate a provide the hoter with 30 days notice of any policies which us not contain this not ner Provisions | a. Retroactive Date? b. Extended Reporting Date? rker's Compensation If the Insured is exempt from Worker's Compensation statutes, does the Insured carry the equivalent Voluntary Compensation coverage: al Payment Information Is this certificate being furnished in connection with the Contractor's request for final payment in accordance with the requirements of Sections 9.10.2 and 11.1.3 of AIA Document A201, General conditions of the Contract for Construction? If so, and if the policy period exends byyond termination of the Contract for Construction, is Completed Coverage for this Project continued for the balance of the policy period? mination Provisions Has each policy shown on the certificate and this Supplement been endorsed to provide the holder with 30 days notice of cancellation and/or expiration? List below any policies which are not contain this notice. | a. Retroactive Date? b. Extended Reporting Date? rker's Compensation If the Insured is exempt from Worker's Compensation statutes, does the Insured carry the equivalent Voluntary Compensation coverage? al Payment Information Is this certificate being furnished in connection with the Contractor's request for final payment in accordance with the requirements of Sections 9.10.2 and 11.1.3 of AIA Document A201, General conditions of the Contract for Construction? If so, and if the policy period extends beyond termination of the Contract for Construction, is Completed Operations coverage for this Project continued for the balance of the policy period? mination Provisions Has each policy shown on the certificate and this Supplement been endorsed to provide the holder with 30 days notice of cancellation and/or expiration? List below any policies which does not contain this notice. | a. Retroactive Date? b. Extended Reporting Date? rker's Compensation If the Insured is exempt from Worker's Compensation statutes, does the Insured carry the equivalent Voluntary Compensation coverage? al Payment Information Is this certificate being furnished in connection with the Contractor's request for final payment in accordance with the requirements of Sections 9.10.2 and 11.1.3 of AIA Document A201, General condition of the Contract for Construction, is Completed Operations coverage for this Project continued for the balance of the policy period? Imination Provisions Has each policy shown on the certificate and this Supplement been endorsed to provide the holder with 30 days notice of cancellation and/or expiration? List below any policies which use of contain this notice. |

AlA Document G715[™] – 1991. Copyright © 1991 by The American Institute of Architects. All rights reserved. WARNING: This AIA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA[®] Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 13:34:53 on 08/02/2005 under Order No.1000162220_1 which expires on 2/15/2006, and is not for resale. User Notes: (43415021)

SECTION 00 72 13

GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

The General Conditions of this Contract are as stated in the American Institute of Architects Decrated AIA A201 (2007 Edition) entitled <u>General Conditions of the Contract for Construction</u> and is art of the project manual as if herein written in full.

20°

A draft copy of this document is included herein as follows.

2 BIDDIK



▲IA Document A201[™] – 2007

General Conditions of the Contract for Construction

for the following PROJECT: (Name and location or address)

THE OWNER: (Name, legal status and address)

THE ARCHITECT: (Name, legal status and address)

TABLE OF ARTICLES

- 1 **GENERAL PROVISIONS**
- 2 OWNER
- 3 CONTRACTOR
- ARCHITECT A
- SUBCONTRACTORS 5
- **RATE CONTRACTORS** CONSTRUCTION BY OWNER 6
- **CHANGES IN THE WORK** 7
- 8 TIME

9

- PAYMENTS AND COMPLETION
- PROTICTION OF PERSONS AND PROPERTY 10
- **NSURANCE AND BONDS** 11
 - JNCOVERING AND CORRECTION OF WORK
 - *IISCELLANEOUS PROVISIONS*
 - **TERMINATION OR SUSPENSION OF THE CONTRACT**
 - CLAIMS AND DISPUTES

ADDITIONS AND The author of this do added information needed for its non. The archor may also compl have revised the text of the original ndare form. An Additions and De tions Report that notes added formation as well as revisions to the standard form text is available from me author and should be reviewed. A vertical line in the left margin of this document indicates where the author has added necessary information and where the author has added to or deleted from the original AIA text.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

1

AIA Document A2011 - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA" Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

OWCF

INDEX

Init.

1

(Topics and numbers in bold are section headings.)

Acceptance of Nonconforming Work 9.6.6, 9.9.3, 12.3 Acceptance of Work 9.6.6, 9.8.2, 9.9.3, 9.10.1, 9.10.3, 12.3 Access to Work 3.16, 6.2.1, 12.1 Accident Prevention 10 Acts and Omissions 3.2, 3.3.2, 3.12.8, 3.18, 4.2.3, 8.3.1, 9.5.1, 10.2.5, 10.2.8, 13.4.2, 13.7, 14.1, 15.2 Addenda 1.1.1.3.11.1 Additional Costs, Claims for 3.7.4, 3.7.5, 6.1.1, 7.3.7.5, 10.3, 15.1.4 Additional Inspections and Testing 9.4.2, 9.8.3, 12.2.1, 13.5 Additional Insured 11.1.4 Additional Time, Claims for 3.2.4, 3.7.4, 3.7.5, 3.10.2, 8.3.2, 15.1.5 **Administration of the Contract** 3.1.3, 4.2, 9.4, 9.5 Advertisement or Invitation to Bid 1.1.1 **Aesthetic Effect** 4.2.13 Allowances 3.8, 7.3.8 All-risk Insurance 11.3.1, 11.3.1.1 **Applications for Payment** 6.3, 9.7, 9.10, 11.1.3 4.2.5, 7.3.9, 9.2, 9.3, 9.4, 9 Approvals 12.8, 3.12.9, 3.12.10, 2.1.1, 2.2.2, 2.4, 3.1. 4.2.7, 9.3.2, 13.5.1 Arbitration 8.3.1, 11.3.10, 3 1, 15.3.2, 15.4 ARCHITECT Architect, Definition of 4.1 chitec, Extent of Authority 2 4 12.7, 4.1, 4.2, 5.2, 6.3, 7.1.2, 7.3.7, 7.4, 9.2, 1, 9, 4, 9, 5, 9, 6, 3, 9, 8, 9, 10, 1, 9, 10, 3, 12, 1, 12, 2, 1, 13.5.1, 13.5.2, 14.2.2, 14.2.4, 15.1.3, 15.2.1 Architect, Limitations of Authority and Responsibility 2.1.1, 3.12.4, 3.12.8, 3.12.10, 4.1.2, 4.2.1, 4.2.2, 4.2.3, 4.2.6, 4.2.7, 4.2.10, 4.2.12, 4.2.13, 5.2.1, 7.4, 9.4.2, 9.5.3, 9.6.4, 15.1.3, 15.2 Architect's Additional Services and Expenses 2.4.1, 11.3.1.1, 12.2.1, 13.5.2, 13.5.3, 14.2.4

Architect's Administration of the Contract 3.1.3, 4.2, 3.7.4, 15.2, 9.4.1, 9.5 Architect's Approvals 2.4.1, 3.1.3, 3.5, 3.10.2, 4.2.7 Architect's Authority to Reject Work 3.5, 4.2.6, 12.1.2, 12.2.1 Architect's Copyright 1.1.7, 1.5 Architect's Decisions 3.7.4, 4.2.6, 4.2.7, 4.2.11, 4.2.12, 4.2.13, 7.3.7, 7.3.9, 8.1.3, 8.3.1, 9.2, 9.4.1, 9.5, 13.5.2, 15.2, 15.3 Architect's Inspections 3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.8.3 Architect's Instructions 3.2.4, 3.3.1, 4.2.6, 4.2.7, Architect's Interpretations 4.2.11, 4.2.12 Architect's Project Representative 4.2.10 Architect's Ken tionship with Contractor 1.1.2, 1.5, 1.6, 3.2.2, 3.2.3, 3.2.4, 3.3.1, 3.4.2, 3.5, 3.7.4, 3.7.5, 39.2, 3.9.3, 3.10, 3.11, 3.12, 3.16, 3.18, 2, 4 1, 3, 4.2, 5.2, 6.2.2, 7, 8.3.1, 9.2, 9.3, 9.4, 9.5, 4 , 9.8, 99, 10.2.6, 10.3, 11.3.7, 12, 13.4.2, 13.5, 15.2 iter's Relationship with Subcontractors 2, 4.2.3, 4.2.4, 4.2.6, 9.6.3, 9.6.4, 11.3.7 Architect's Representations 9.4.2, 9.5.1, 9.10.1 Architect's Site Visits 3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.5.1, 9.9.2, 9.10.1, 13.5 Asbestos 10.3.1 Attorneys' Fees 3.18.1, 9.10.2, 10.3.3 Award of Separate Contracts 6.1.1, 6.1.2 Award of Subcontracts and Other Contracts for Portions of the Work 5.2 **Basic Definitions** 1.1 **Bidding Requirements** 1.1.1, 5.2.1, 11.4.1 **Binding Dispute Resolution** 9.7, 11.3.9, 11.3.10, 13.1.1, 15.2.5, 15.2.6.1, 15.3.1, 15.3.2, 15.4.1 **Boiler and Machinery Insurance** 11.3.2 Bonds, Lien 7.3.7.4, 9.10.2, 9.10.3 **Bonds, Performance, and Payment** 7.3.7.4, 9.6.7, 9.10.3, 11.3.9, 11.4 **Building Permit** 3.7.1

AlA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and international Treaties, Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No 2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

Capitalization 1.3 Certificate of Substantial Completion 9.8.3, 9.8.4, 9.8.5 **Certificates for Payment** 4.2.1, 4.2.5, 4.2.9, 9.3.3, 9.4, 9.5, 9.6.1, 9.6.6, 9.7, 9.10.1, 9.10.3, 14.1.1.3, 14.2.4, 15.1.3 Certificates of Inspection, Testing or Approval 13.5.4 Certificates of Insurance 9.10.2, 11.1.3 **Change Orders** 1.1.1, 2.4.1, 3.4.2, 3.7.4, 3.8.2.3, 3.11.1, 3.12.8, 4.2.8, 5.2.3, 7.1.2, 7.1.3, 7.2, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 8.3.1, 9.3.1.1, 9.10.3, 10.3.2, 11.3.1.2, 11.3.4, 11.3.9, 12.1.2, 15.1.3 Change Orders, Definition of 7.2.1 **CHANGES IN THE WORK** 2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 8.3.1, 9.3.1.1, 11.3.9 Claims, Definition of 15.1.1 **CLAIMS AND DISPUTES** 3.2.4, 6.1.1, 6.3, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims and Timely Assertion of Claims 15.4.1 **Claims for Additional Cost** 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 **Claims for Additional Time** 3.2.4, 3.7.46.1.1, 8.3.2, 10.3.2, 15.1.5 **Concealed or Unknown Conditions, Claim** 3.7.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1 Claims Subject to Arbitration 15.3.1, 15.4.1 **Cleaning Up** 3.15, 6.3 Commencement of the Work Conditions Relating to 2.2.1, 3.2.2, 3.4 , 3.71, 3.10, 1, 3.12, 6, 5.2, 1, 5.2, 3, 6.2.2, 8.1.2, 8.2, 7.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Common cement of the Work, Definition of 8.1.2 mmenications Facilitating Contract Iminis ration 2.4 ppletion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9 10, 12 2, 13 7, 14.1.2 **COMPLETION, PAYMENTS AND** 9 Completion, Substantial 4.2.9, 8.1.1, 8.1.3, 8.2.3, 9.4.2, 9.8, 9.9.1, 9.10.3, 12.2, 13.7

Init.

1

Compliance with Laws 1.6.1, 3.2.3, 3.6, 3.7, 3.12.10, 3.13, 4.1.1, 9.6.4, 10.2.2, 11.1, 11.3, 13.1, 13.4, 13.5.1, 13.5.2, 13.6, 14.1.1, 14.2.1.3, 15.2.8, 15.4.2, 15.4.3 Concealed or Unknown Conditions 3.7.4, 4.2.8, 8.3.1, 10.3 Conditions of the Contract 1.1.1, 6.1.1, 6.1.4 Consent, Written 3.4.2, 3.7.4, 3.12.8, 3.14.2, 4.1.2, 9.3.2, 9.8.5 9.10.2, 9.10.3, 11.3.1, 13.2, 13.4.2, 15.4.4 **Consolidation or Joinder** 15.4.4 **CONSTRUCTION BY OWNER OR B** SEPARATE CONTRACTORS 1.1.4, 6 **Construction Change Direct** , Definition of 7.3.1 **Construction Change Directives** 1.1.1, 3.4.2, 3.12.8, 4.2, 8, 7, 1.1, 7.1.2, 7.1.3, 7.3, 9.3.1.1 Construction Schearles, Contractor's 3.10, 3.12. 3 12.2, 6.1.3, 15.1.5.2 **Contingent Assignment of Subcontracts** 5.4, 14.2.2.2 ntinui g Contract Performance С **Contract**, Definition of 1.1.2 CONTRACT, TERMINATION OR SUSPENSION OF THE 5.4.1.1, 11.3.9, 14 Contract Administration 3.1.3, 4, 9.4, 9.5 Contract Award and Execution, Conditions Relating to 3.7.1, 3.10, 5.2, 6.1, 11.1.3, 11.3.6, 11.4.1 Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Documents, Definition of 1.1.1 **Contract Sum** 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5 Contract Sum, Definition of 9.1 **Contract** Time 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7, 10.3.2, 12.1.1, 14.3.2, 15.1.5.1, 15.2.5 Contract Time, Definition of 8.1.1 CONTRACTOR 3 Contractor, Definition of 3.1, 6.1.2

Contractor's Construction Schedules 3.10, 3.12.1, 3.12.2, 6.1.3, 15.1.5.2 Contractor's Employees 3.3.2, 3.4.3, 3.8.1, 3.9, 3.18.2, 4.2.3, 4.2.6, 10.2, 10.3, 11.1.1, 11.3.7, 14.1, 14.2.1.1 **Contractor's Liability Insurance** 11.1 Contractor's Relationship with Separate Contractors and Owner's Forces 3.12.5, 3.14.2, 4.2.4, 6, 11.3.7, 12.1.2, 12.2.4 Contractor's Relationship with Subcontractors 1.2.2, 3.3.2, 3.18.1, 3.18.2, 5, 9.6.2, 9.6.7, 9.10.2, 11.3.1.2, 11.3.7, 11.3.8 Contractor's Relationship with the Architect 1.1.2, 1.5, 3.1.3, 3.2.2, 3.2.3, 3.2.4, 3.3.1, 3.4.2, 3.5, 3.7.4, 3.10, 3.11, 3.12, 3.16, 3.18, 4.1.3, 4.2, 5.2, 6.2.2, 7, 8.3 1, 9 2, 9.3, 9 4, 9.5, 9 7, 9.8, 9.9, 10.2.6, 10.3, 11.3 7, 12, 13.5, 15 1.2, 15 2.1 Contractor's Representations 3.2.1, 3.2.2, 3.5, 3.12.6, 6.2.2, 8.2.1, 9.3.3, 9.8.2 Contractor's Responsibility for Those Performing the Work 3 3 2, 3 18, 5.3.1, 6 1 3, 6 2, 9 5 1, 10.2.8 Contractor's Review of Contract Documents 3.2 Contractor's Right to Stop the Work 9.7 Contractor's Right to Terminate the Contract 14.1.15.1.6 Contractor's Submittals 3.10, 3.11, 3.12.4, 4.2.7, 5.2.1, 5.2.3, 9.2, 9.3, 9.8 983, 99.1, 9102, 9103, 11.1.3, 11.4.2 Contractor's Superintendent 3.9, 10.2.6 Contractor's Supervision and Construct Procedures 1.2.2, 3.3, 3.4, 3.12.10, 4.2.2, 4.2.7, 6. .2.4, 7.1.3, 7.3.5, 7.3.7, 8.2, 10, 12, 14, $\mathbf{5}$ Contractual Liability Insura 11.1.1.8, 11.2 Coordination and Correlation 1.2, 3.2.1, 3.3.1 .10, 6.1.3, 6.2.1 Copies Furnished of Drawings and Specifications 1.5, 2.2.5, 3.11 Copyrig 1.5.3.17 rrect, n of Work .7.3, 9.4.2, 9.8.2, 9.8.3, 9.9.1, 12.1.2, 12.2 , 2.4, or elation and Intent of the Contract Documents Cost, Definition of 7.3.7 Costs 2.4.1, 3.2.4, 3.7.3, 3.8.2, 3.15.2, 5.4.2, 6.1.1, 6.2.3, 7.3.3.3, 7.3.7, 7.3.8, 7.3.9, 9.10.2, 10.3.2, 10.3.6, 11.3, 12.1.2, 12.2.1, 12.2.4, 13.5, 14

11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Damages for Delay 6.1.1, 8.3.3, 9.5.1.6, 9.7, 10.3.2 Date of Commencement of the Work, D fination of 8.1.2 Date of Substantial Completion. Demmion of 8.1.3 Day, Definition of 8.1.4 Decisions of the Architec 3.7.4, 4.2.6, 4.2.1, 4.2.1, 4.2.12, 4.2.13, 15.2, 6.3, 2, 9.2, 9.4, 9.5.1, 9.8.4, 9.9.1, 7.3.7, 7.3.9 5.1. , 8 13.5.2, 14.2, 4.2.4, 15.1, 15.2 Decisions to Withhold Certification 9.4 1, 9.5 9.7, 14.1.1.3 Detective r Nonconforming Work, Acceptance, ejection and Correction of 2.4.1, 3.5, 4.2.6, 6.2.5, 9.5.1, 9.5.2, 9.6.6, 9.8.2, 9.9.3, 9.10.4, 12.2.1 Definitions 1.1, 2.1.1, 3.1.1, 3.5, 3.12.1, 3.12.2, 3.12.3, 4.1.1, 15.1.1, 5.1, 6.1.2, 7.2.1, 7.3.1, 8.1, 9.1, 9.8.1 **Delays and Extensions of Time** 3.2, 3.7.4, 5.2.3, 7.2.1, 7.3.1, 7.4, **8.3**, 9.5.1, 9.7, 10.3.2, 10.4.1, 14.3.2, 15.1.5, 15.2.5 Disputes 6.3, 7.3.9, 15.1, 15.2 Documents and Samples at the Site 3.11 Drawings, Definition of 1.1.5 Drawings and Specifications, Use and Ownership of 3.11 Effective Date of Insurance 8.2.2, 11.1.2 Emergencies 10.4, 14.1.1.2, 15.1.4 Employees, Contractor's 3.3.2, 3.4.3, 3.8.1, 3.9, 3.18.2, 4.2.3, 4.2.6, 10.2, 10.3.3, 11.1.1, 11.3.7, 14.1, 14.2.1.1 Equipment, Labor, Materials or 1.1.3, 1.1.6, 3.4, 3.5, 3.8.2, 3.8.3, 3.12, 3.13.1, 3.15.1, 4.2.6, 4.2.7, 5.2.1, 6.2.1, 7.3.7, 9.3.2, 9.3.3, 9.5.1.3, 9.10.2, 10.2.1, 10.2.4, 14.2.1.1, 14.2.1.2 AIA Document A20111 - 2007. Copyright @ 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which explres (726494055)

Cutting and Patching

Damage to the Work

Damages, Claims for

Damage to Construction of Owner or Separate

3.14.2, 6.2.4, 10.2.1.2, 10.2.5, 10.4, 11.1.1, 11.3,

3.14.2, 9.9.1, 10.2.1.2, 10.2.5, 10.4.1, 11.3.1, 12.2

3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 1

3.14, 6.2.5

Contractors

12.2.4

Init.

1

reproduction or distribution of this AIA®

on 03/09/2012, and is not for resale.

User Notes:

Execution and Progress of the Work 1.1.3, 1.2.1, 1.2.2, 2.2.3, 2.2.5, 3.1, 3.3.1, 3.4.1, 3.5, 3.7.1, 3.10.1, 3.12, 3.14, 4.2, 6.2.2, 7.1.3, 7.3.5, 8.2, 9.5.1, 9.9.1, 10.2, 10.3, 12.2, 14.2, 14.3.1, 15.1.3 Extensions of Time 3.2.4, 3.7.4, 5.2.3, 7.2.1, 7.3, 7.4, 9.5.1, 9.7, 10.3.2, 10.4.1, 14.3, 15.1.5, 15.2.5 **Failure of Payment** 9.5.1.3, 9.7, 9.10.2, 13.6, 14.1.1.3, 14.2.1.2 Faulty Work (See Defective or Nonconforming Work) **Final Completion and Final Payment** 4.2.1, 4.2.9, 9.8.2, 9.10, 11.1.2, 11.1.3, 11.3.1, 11.3.5, 12.3.1, 14.2.4, 14.4.3 Financial Arrangements, Owner's 2.2.1, 13.2.2, 14.1.1.4 Fire and Extended Coverage Insurance 11.3.1.1 **GENERAL PROVISIONS** 1 **Governing Law** 13.1 Guarantees (See Warranty) **Hazardous Materials** 10.2.4, 10.3 Identification of Subcontractors and Suppliers 5.2.1 Indemnification 3 17, 3.18, 9 10.2, 10.3.3, 10.3.5, 10.3.6, 11.3.1.2, 11.3.7 Information and Services Required of the Owner 2 1 2, 2.2, 3.2 2, 3 12 4, 3 12.10, 6.1.3, 6.1.4 961, 9.6.4, 9.9.2, 9.10.3, 10.3.3, 11.2, 11.4 13.5. 13.5.2, 14.1.1.4, 14.1.4, 15.1.3 **Initial Decision** 15.2 Initial Decision Maker, Definition of 1.1.8 Initial Decision Maker, Decision 5.2.3, 15.2.4, 15.2.5 14.2.2, 14.2.4, 15.2.1 Initial Decision Maker, Extent of Authority 14.2.2, 14.2.4, 15, 1.3, 5.2 , 15.2.2, 15.2.3, 15.2.4, 15.2.5 Injury or Damage to Person or Property 10.2.8, 1.4.1 Inspections .5, 5, 3, 3, 3, 7, 1, 4, 2, 2, 4, 2, 6, 4, 2, 9, 9, 4, 2, 9, 8, 3, 3 9 2.2, 9.1.1, 12.2.1, 13.5 instructions to Bidders Instructions to the Contractor 8.2.4, 3.3.1, 3.8.1, 5.2.1, 7, 8.2.2, 12, 13.5.2 Instruments of Service, Definition of 1.1.7 Insurance 3.18.1, 6.1.1, 7.3.7, 9.3.2, 9.8.4, 9.9.1, 9.10.2, 11

Init.

1

Insurance, Boiler and Machinery 11.3.2 Insurance, Contractor's Liability 11.1 Insurance, Effective Date of 8.2.2, 11.1.2 Insurance, Loss of Use 11.3.3 **Insurance**, Owner's Liability 11.2 **Insurance**, **Property** 10.2.5, 11.3 Insurance, Stored Materials 9.3.2 **INSURANCE AND BONDS** 11 Partial Occupancy Insurance Companies, Conse 9.9.1 Intent of the Contract 00 Q. [. 1.2.1, 4.2.7, 4.2.12, 4 7.4 Interest 13.6 Interpretation 1.2.3, **1.4**, **4**, **1**, **5**, **1**, **6**, **1**, **2**, **15**, **1**, **1** Interpretations, Written 4 .11, 4., 12, 15.1.4 udgment on Final Award 15.4.2 Labor and Materials, Equipment 1.1.3, 1.1.6, 3.4, 3.5, 3.8.2, 3.8.3, 3.12, 3.13, 3.15.1, 4.2.6, 4.2.7, 5.2.1, 6.2.1, 7.3.7, 9.3.2, 9.3.3, 9.5.1.3, 9.10.2, 10.2.1, 10.2.4, 14.2.1.1, 14.2.1.2 Labor Disputes 8.3.1 Laws and Regulations 1.5, 3.2.3, 3.6, 3.7, 3.12.10, 3.13.1, 4.1.1, 9.6.4, 9.9.1, 10.2.2, 11.1.1, 11.3, 13.1.1, 13.4, 13.5.1, 13.5.2, 13.6.1, 14, 15.2.8, 15.4 Liens 2.1.2, 9.3.3, 9.10.2, 9.10.4, 15.2.8 Limitations, Statutes of 12.2.5, 13.7, 15.4.1.1 Limitations of Liability 2.3.1, 3.2.2, 3.5, 3.12.10, 3.17, 3.18.1, 4.2.6, 4.2.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 10.2.5, 10.3.3, 11.1.2, 11.2, 11.3.7, 12.2.5, 13.4.2 Limitations of Time 2.1.2, 2.2, 2.4, 3.2.2, 3.10, 3.11, 3.12.5, 3.15.1, 4.2.7, 5.2, 5.3.1, 5.4.1, 6.2.4, 7.3, 7.4, 8.2, 9.2, 9.3.1, 9.3.3, 9.4.1, 9.5, 9.6, 9.7, 9.8, 9.9, 9.10, 11.1.3, 11.3.1.5, 11.3.6, 11.3.10, 12.2, 13.5, 13.7, 14, 15 Loss of Use Insurance 11.3.3 Material Suppliers 1.5, 3.12.1, 4.2.4, 4.2.6, 5.2.1, 9.3, 9.4.2, 9.6, 9.10.5 Materials, Hazardous 10.2.4, 10.3

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved, WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:
Materials, Labor, Equipment and 1.1.3, 1.1.6, 1.5.1, 3.4.1, 3.5, 3.8.2, 3.8.3, 3.12, 3.13.1, 3.15.1, 4.2.6, 4.2.7, 5.2.1, 6.2.1, 7.3.7, 9.3.2, 9.3.3, 9.5.1.3, 9.10.2, 10.2.1.2, 10.2.4, 14.2.1.1, 14.2.1.2 Means, Methods, Techniques, Sequences and Procedures of Construction 3.3.1, 3.12.10, 4.2.2, 4.2.7, 9.4.2 Mechanic's Lien 2.1.2, 15.2.8 Mediation 8.3.1, 10.3.5, 10.3.6, 15.2.1, 15.2.5, 15.2.6, 15.3, 15.4.1 **Minor Changes in the Work** 1.1.1, 3.12.8, 4.2.8, 7.1, 7.4 MISCELLANEOUS PROVISIONS 13 Modifications, Definition of 1.1.1 Modifications to the Contract 1.1.1, 1.1.2, 3.11, 4.1.2, 4.2.1, 5.2.3, 7, 8.3.1, 9.7, 10.3.2, 11.3.1 **Mutual Responsibility** 6.2 Nonconforming Work, Acceptance of 9.6.6, 9.9.3, 12.3 Nonconforming Work, Rejection and Correction of 2.3.1, 2.4.1, 3.5, 4.2.6, 6.2.4, 9.5.1, 9.8.2, 9.9.3, 9.10.4, 12.2.1 Notice 2 2 1, 2 3 1, 2.4 1, 3 2 4, 3.3.1, 3.7.2, 3.12.9, 5.2.1, 9.10, 10.2.2, 11.1.3, 12.2.2.1, 13.3, 13.5.1, 13.5.2 14.1, 14.2, 15.2.8, 15.4.1 Notice, Written 2.3.1, 2.4.1, 3.3.1, 3.9.2, 3.12.9, 3.12.10 9.10, 10.2.2, 10.3, 11.1.3, 11.3.6, 1 15.2.8, 15.4.1 **Notice of Claims** 3.7.4, 10.2.8, 15.1.2, 15.4 Notice of Testing and Insp. 13.5.1, 13.5.2 Observations, Contra ctor's 3.2, 3.7.4 Occupancy 2.2.2, 9.6, 6, 9.8, 3.1.5 Orders Viritten 1.1.1, 2.3, 39.2, 2, 7, 8.2.2, 11.3.9, 12.1, 12.2.2.1, 13.5.2, .3.Γ WNE mer, Definition of 2.1.1 **Owner, Information and Services Required of the** 2.1.2, 2.2, 3.2.2, 3.12.10, 6.1.3, 6.1.4, 6.2.5, 9.3.2, 9.6.1, 9.6.4, 9.9.2, 9.10.3, 10.3.3, 11.2, 11.3, 13.5.1, 13.5.2, 14.1.1.4, 14.1.4, 15.1.3

Init.

1

Owner's Authority 1.5, 2.1.1, 2.3.1, 2.4.1, 3.4.2, 3.8.1, 3.12.10, 3.14.2, 4.1.2, 4.1.3, 4.2.4, 4.2.9, 5.2.1, 5.2.4, 5.4.1, 6.1, 6.3, 7.2.1, 7.3.1, 8.2.2, 8.3.1, 9.3.1, 9.3.2, 9.5.1, 9.6.4, 9.9.1, 9.10.2, 10.3.2, 11.1.3, 11.3.3, 11.3.10, 12.2.2, 12.3.1, 13.2.2, 14.3, 14.4, 15.2.7 **Owner's Financial Capability** 2.2.1, 13.2.2, 14.1.1.4 **Owner's Liability Insurance** 11.2 Owner's Relationship with Subcontractors 1.1.2, 5.2, 5.3, 5.4, 9.6.4, 9.10.2, 14.2.2 **Owner's Right to Carry Out the Worl** 2.4, 14.2.2 **Owner's Right to Clean Up** 6.3 **Owner's Right to Perform** struction and to Award Separate Contract 6.1 Owner's Right to St. p to e Work 2.3 Owner's Right nd the Work 14.3 Owner's Right to Terminate the Contract 14 vnersh p and Use of Drawings, Specifications O nd Other Instruments of Service 1, 1.1.6, 1.1.7, 1.5, 2.2.5, 3.2.2, 3.11.1, 3.17, 4.2.12, 5.3.1 Partial Occupancy or Use 9.6.6, 9.9, 11.3.1.5 Patching, Cutting and 3.14, 6.2.5 Patents 3.17 **Payment, Applications for** 4.2.5, 7.3.9, 9.2, 9.3, 9.4, 9.5, 9.6.3, 9.7, 9.8.5, 9.10.1, 14.2.3, 14.2.4, 14.4.3 **Payment, Certificates for** 4.2.5, 4.2.9, 9.3.3, 9.4, 9.5, 9.6.1, 9.6.6, 9.7, 9.10.1, 9.10.3, 13.7, 14.1.1.3, 14.2.4 Payment, Failure of 9.5.1.3, 9.7, 9.10.2, 13.6, 14.1.1.3, 14.2.1.2 Payment, Final 4.2.1, 4.2.9, 9.8.2, 9.10, 11.1.2, 11.1.3, 11.4.1, 12.3.1, 13.7, 14.2.4, 14.4.3 **Payment Bond, Performance Bond and** 7.3.7.4, 9.6.7, 9.10.3, 11.4 **Payments**, **Progress** 9.3, 9.6, 9.8.5, 9.10.3, 13.6, 14.2.3, 15.1.3 PAYMENTS AND COMPLETION Payments to Subcontractors 5.4.2, 9.5.1.3, 9.6.2, 9.6.3, 9.6.4, 9.6.7, 14.2.1.2 PCB 10.3.1

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

6

Performance Bond and Payment Bond 7.3.7.4, 9.6.7, 9.10.3, 11.4 Permits, Fees, Notices and Compliance with Laws 2.2.2, 3.7, 3.13, 7.3.7.4, 10.2.2 PERSONS AND PROPERTY, PROTECTION OF 10 Polychlorinated Biphenyl 10.3.1 Product Data, Definition of 3.12.2 **Product Data and Samples, Shop Drawings** 3.11, 3.12, 4.2.7 **Progress and Completion** 4.2.2, 8.2, 9.8, 9.9.1, 14.1.4, 15.1.3 **Progress Payments** 9.3, 9.6, 9.8.5, 9.10.3, 13.6, 14.2.3, 15.1.3 **Project**, Definition of 1.1.4 **Project Representatives** 4.2.10 **Property Insurance** 10.2.5, 11.3 PROTECTION OF PERSONS AND PROPERTY 10 **Regulations and Laws** 1.5, 3.2.3, 3.6, 3.7, 3.12.10, 3.13, 4.1.1, 9.6.4, 9.9.1, 10.2.2, 11.1, 11.4, 13.1, 13.4, 13.5.1, 13.5.2, 13.6, 14, 15.2.8, 15.4 **Rejection of Work** 3.5, 4.2.6, 12.2.1 Releases and Waivers of Liens 9.10.2 Representations 3.2.1, 3.5, 3.12.6, 6.2.2, 8.2.1, 9.3.3, 9.4.2 9.10.1 Representatives 2.1.1, 3.1.1, 3.9, 4.1.1, 4.2.1, 4.2.2, 4. 1.1, 5.1.2, 13.2.1 Responsibility for Those Purfor the Work 6.2, 6.3, 9.5.1, 10 3.3.2, 3.18, 4.2.3, 5.3, Retainage 9.3.1, 9.6.2, 9.8 9.9 9.1 0.2, 9.10.3 Review of Congrant Documents and Field **Conditions by Contractor** 3.2, 3.12, 6.1.3 Review of Contractor's Submittals by Owner and chite 10.1, 3 10.2, 3.11, 3.12, 4.2, 5.2, 6.1.3, 9.2, 9.8.2 of Shop Drawings, Product Data and Samples Contractor hi 3.12 **Rights and Remedies** 1.1.2, 2.3, 2.4, 3.5, 3.7.4, 3.15.2, 4.2.6, 5.3, 5.4, 6.1, 6.3, 7.3.1, 8.3, 9.5.1, 9.7, 10.2.5, 10.3, 12.2.2, 12.2.4, 13.4, 14, 15.4 **Royalties, Patents and Copyrights** 3.17

Rules and Notices for Arbitration 15.4.1 Safety of Persons and Property 10.2, 10.4 **Safety Precautions and Programs** 3.3.1, 4.2.2, 4.2.7, 5.3.1, 10.1, 10.2, 10.4 Samples, Definition of 3.12.3 Samples, Shop Drawings, Product Data and 3.11, 3.12, 4.2.7 Samples at the Site, Documents and 3.11 Schedule of Values 9.2, 9.3.1 Schedules, Construction 3.10, 3.12.1, 3.12.2, 6.1.3, 1 Separate Contracts and Contracts 1.1.4, 3.12.5, 3.14.2, 42.4, 1.2.1, 6, 8.3.1, 12.1.2 Shop Drawings, Definition of 3.12.1 Shop Drawings, Product Data and Samples 3.11, **3.12**, Site, Use 3.13, 6.1.1, 6. Sit Inspections 2, 5.3, 3.7.1, 3.7.4, 4.2, 9.4.2, 9.10.1, 13.5 3 Site Views, Architect's 4, 4.2.2, 4.2.9, 9.4.2, 9.5.1, 9.9.2, 9.10.1, 13.5 Special Inspections and Testing 4.2.6, 12.2.1, 13.5 Specifications, Definition of 1.1.6 Specifications 1.1.1, 1.1.6, 1.2.2, 1.5, 3.11, 3.12.10, 3.17, 4.2.14 Statute of Limitations 13.7, 15.4.1.1 Stopping the Work 2.3, 9.7, 10.3, 14.1 Stored Materials 6.2.1, 9.3.2, 10.2.1.2, 10.2.4 Subcontractor, Definition of 5.1.1 SUBCONTRACTORS 5 Subcontractors, Work by 1.2.2, 3.3.2, 3.12.1, 4.2.3, 5.2.3, 5.3, 5.4, 9.3.1.2, 9.6.7 Subcontractual Relations 5.3, 5.4, 9.3.1.2, 9.6, 9.10, 10.2.1, 14.1, 14.2.1 Submittals 3.10, 3.11, 3.12, 4.2.7, 5.2.1, 5.2.3, 7.3.7, 9.2, 9.3, 9.8, 9.9.1, 9.10.2, 9.10.3, 11.1.3 Submittal Schedule 3.10.2, 3.12.5, 4.2.7 Subrogation, Waivers of 6.1.1, **11.3.7**

1

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA[®] Document is protected by U.S. Copyright Law and international Treaties. Unauthorized reproduction or distribution of this AIA[®] Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No 2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

7

Substantial Completion 4.2.9, 8.1.1, 8.1.3, 8.2.3, 9.4.2, 9.8, 9.9.1, 9.10.3, 12.2, 13.7 Substantial Completion, Definition of 9.8.1 Substitution of Subcontractors 5.2.3, 5.2.4 Substitution of Architect 4.1.3 Substitutions of Materials 3.4.2, 3.5, 7.3.8 Sub-subcontractor, Definition of 5.1.2 Subsurface Conditions 3.7.4 Successors and Assigns 13.2 Superintendent 3.9, 10.2.6 **Supervision and Construction Procedures** 1 2 2, 3.3, 3 4, 3.12 10, 4 2 2, 4 2.7, 6.1.3, 6.2.4, 7.1.3, 7.3.7, 8.2, 8.3.1, 9.4.2, 10, 12, 14, 15.1.3 Surety 5.4.1.2, 9.8.5, 9.10.2, 9.10.3, 14.2.2, 15.2.7 Surety, Consent of 9.10.2, 9.10.3 Surveys 2.2.3 Suspension by the Owner for Convenience 14.3 Suspension of the Work 5.4.2, 14.3 Suspension or Termination of the Contract 5.4.1.1, 14 Taxes 3.6, 3.8.2.1, 7.3.7.4 **Termination by the Contractor** 14.1, 15.1.6 ase Termination by the Owne 5.4.1.1, 14.2, 15.1.6 Termination by the Owner for Convenience 14.4 Termination of the Architect 4.1.3 on of the Contractor Termin 14.2.2 **ARM: NATION OR SUSPENSION OF THE** ONTRACT ts and Inspections 3.1.3, 3.3.3, 4.2.2, 4.2.6, 4.2.9, 9.4.2, 9.8.3, 9.9.2, 9 10 1, 10 3 2, 11.4.1.1, 12.2.1, 13.5 TIME 8 Time, Delays and Extensions of 3.2.4, 3.7.4, 5.2.3, 7.2.1, 7.3.1, 7.4, **8.3**, 9.5.1, 9.7, 10.3.2, 10.4.1, 14.3.2, 15.1.5, 15.2.5

Time Limits 2.1.2, 2.2, 2.4, 3.2.2, 3.10, 3.11, 3.12.5, 3.15.1, 4.2, 5.2, 5.3, 5.4, 6.2.4, 7.3, 7.4, 8.2, 9.2, 9.3.1, 9.3.3, 9.4.1, 9.5, 9.6, 9.7, 9.8, 9.9, 9.10, 11.1.3, 12.2, 13.5, 13.7, 14, 15.1.2, 15.4 **Time Limits on Claims** 3.7.4, 10.2.8, 13.7, 15.1.2 Title to Work 9.3.2, 9.3.3 **Transmission of Data in Digital Form** 1.6 UNCOVERING AND CORRECTION OF 12 **Uncovering of Work** 12.1 cale Unforeseen Conditions, Con inknown 3.7.4, 8.3.1, 10.3 Unit Prices 7.3.3.2, 7.3.4 Use of Documents 1.1.1, 1.5, 2.2.5, Use of Site 3.13, 6.1.1, Values, Scheule of 9.2 9.3. iver of Claims by the Architect UZ. ver of Claims by the Contractor 9.10.5, 13.4.2, 15.1.6 Waiver of Claims by the Owner 9.9.3, 9.10.3, 9.10.4, 12.2.2.1, 13.4.2, 14.2.4, 15.1.6 Waiver of Consequential Damages 14.2.4, 15.1.6 Waiver of Liens 9.10.2, 9.10.4 Waivers of Subrogation 6.1.1, 11.3.7 Warranty 3.5, 4.2.9, 9.3.3, 9.8.4, 9.9.1, 9.10.4, 12.2.2, 13.7 Weather Delays 15.1.5.2 Work, Definition of 1.1.3 Written Consent 1.5.2, 3.4.2, 3.7.4, 3.12.8, 3.14.2, 4.1.2, 9.3.2, 9.8.5, 9.9.1, 9.10.2, 9.10.3, 11.4.1, 13.2, 13.4.2, 15.4.4.2 Written Interpretations 4.2.11, 4.2.12 Written Notice 2.3, 2.4, 3.3.1, 3.9, 3.12.9, 3.12.10, 5.2.1, 8.2.2, 9.7, 9.10, 10.2.2, 10.3, 11.1.3, 12.2.2, 12.2.4, 13.3, 14, 15.4.1 Written Orders 1.1.1, 2.3, 3.9, 7, 8.2.2, 12.1, 12.2, 13.5.2, 14.3.1, 15.1.2

init.

1

AIA Document A2011 - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 8 reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

ARTICLE 1 GENERAL PROVISIONS § 1.1 BASIC DEFINITIONS § 1.1.1 THE CONTRACT DOCUMENTS

The Contract Documents are enumerated in the Agreement between the Owner and Contractor (hereinafter the Agreement) and consist of the Agreement, Conditions of the Contract (General, Supplementary and other Conditions), Drawings, Specifications, Addenda issued prior to execution of the Contract, other documents listed in the Agreement and Modifications issued after execution of the Contract. A Modification is (1) a written amendment to the Contract signed by both parties, (2) a Change Order, (3) a Construction Change Directive or (4) a written order for a minor change in the Work issued by the Architect. Unless specifically enumerated in the Agreement, the Contract Documents do not include the advertisement or invitation to bid, Instructions to Bidders, sample forms, other information furnished by the Owner in anticipation of receiving bids or proposals, the Contractor's bid or proposals and the contractor's bid or proposals and the contractor's bid or proposals and the contractor of the contracto portions of Addenda relating to bidding requirements.

§ 1.1.2 THE CONTRACT

The Contract Documents form the Contract for Construction. The Contract represents the entire and integ ated agreement between the parties hereto and supersedes prior negotiations, representations or agreements, either written or oral. The Contract may be amended or modified only by a Modification. The Contract Downents shall not be construed to create a contractual relationship of any kind (1) between the Contractor and the Anihitect or the Architect's consultants, (2) between the Owner and a Subcontractor or a Sub-subcontractor, (3) between the Owner and the Architect or the Architect's consultants or (4) between any persons or entities other than the Owner and the Contractor. The Architect shall, however, be entitled to performance and enforcement of obligations under the Contract intended to facilitate performance of the Architect's duties.

§ 1.1.3 THE WORK

The term "Work" means the construction and services required by the Contract Documents, whether completed or partially completed, and includes all other labor, materials, equipment and services provided or to be provided by the Contractor to fulfill the Contractor's obligations. The Work may constitute the whole or a part of the Project.

§ 1.1.4 THE PROJECT

The Project is the total construction of which the Wark performed under the Contract Documents may be the whole or a part and which may include construction by the wner and by separate contractors.

§ 1.1.5 THE DRAWINGS

The Drawings are the graphic and pictorial portions of the Contract Documents showing the design, location and dimensions of the Work, generally including plans, elevations, sections, details, schedules and diagrams.

§ 1.1.6 THE SPECIFICATIONS

The Specifications are that portion of the Contract Documents consisting of the written requirements for materials, equipment, systems, standards and workmanship for the Work, and performance of related services.

§ 1.1.7 INSTRUMENTE OF SERVICE

Instruments of Savice representations, in any medium of expression now known or later developed, of the tangible and intangible were work performed by the Architect and the Architect's consultants under their respective professional services agreements. Instruments of Service may include, without limitation, studies, surveys, models, sketcher, drawings, specifications, and other similar materials.

1.1.8 NITIAL DECISION MAKER

e Initial Decision Maker is the person identified in the Agreement to render initial decisions on Claims in timice with Section 15.2 and certify termination of the Agreement under Section 14.2.2.

§ 1.2 CORRELATION AND INTENT OF THE CONTRACT DOCUMENTS

§ 1.2.1 The intent of the Contract Documents is to include all items necessary for the proper execution and completion of the Work by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all; performance by the Contractor shall be required only to the extent consistent with the Contract Documents and reasonably inferable from them as being necessary to produce the indicated results.

Init

1

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No 2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

9

§ 1.2.2 Organization of the Specifications into divisions, sections and articles, and arrangement of Drawings shall not control the Contractor in dividing the Work among Subcontractors or in establishing the extent of Work to be performed by any trade.

§ 1.2.3 Unless otherwise stated in the Contract Documents, words that have well-known technical or construction industry meanings are used in the Contract Documents in accordance with such recognized meanings.

§ 1.3 CAPITALIZATION

Terms capitalized in these General Conditions include those that are (1) specifically defined, (2) the titles of num articles or (3) the titles of other documents published by the American Institute of Architects.

§ 1.4 INTERPRETATION

In the interest of brevity the Contract Documents frequently omit modifying words such as "all" and "any and a ticles such as "the" and "an," but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement.

§ 1.5 OWNERSHIP AND USE OF DRAWINGS, SPECIFICATIONS AND OTHER INSTRUMENT OF SERVICE 27

§ 1.5.1 The Architect and the Architect's consultants shall be deemed the authors and owners of their respective Instruments of Service, including the Drawings and Specifications, and will retain all oppmontaw, statutory and other reserved rights, including copyrights. The Contractor, Subcontractors, Sub-subcontractor, and material or equipment suppliers shall not own or claim a copyright in the Instruments of Service. Submittal or distribution to meet official regulatory requirements or for other purposes in connection with this Project not to be construed as publication in derogation of the Architect's or Architect's consultants' reserved rights.

§ 1.5.2 The Contractor, Subcontractors, Sub-subcontractors and material or equipment suppliers are authorized to use and reproduce the Instruments of Service provided to them solely and exclusively for execution of the Work. All copies made under this authorization shall bear the copyright notice, if any, shown on the Instruments of Service. The Contractor, Subcontractors, Sub-subcontractors, and matched or equipment suppliers may not use the Instruments of Contractor, Subcontractors, Sub-subcontractors, and materies Service on other projects or for additions to this Project, uts de the scope of the Work without the specific written consent of the Owner, Architect and the Architect' consummts.

§ 1.6 TRANSMISSION OF DATA IN DIGITAL FORM

If the parties intend to transmit Instruments of Service or any other information or documentation in digital form, they shall endeavor to establish necessary provided in the Agreement or the Contract Docume

ARTICLE 2 OWNER

§ 2.1 GENERAL

§ 2.1.1 The Owner is the possent or untity identified as such in the Agreement and is referred to throughout the Contract Documents as if sing that is number. The Owner shall designate in writing a representative who shall have express authority to bind the Owner with respect to all matters requiring the Owner's approval or authorization. Except as otherwise provided in Section 4.2.1, the Architect does not have such authority. The term "Owner" means the Owner or the Owner Sauthorized representative.

§ 2.1.2 the Owner shall furnish to the Contractor within fifteen days after receipt of a written request, information necessary and relevant for the Contractor to evaluate, give notice of or enforce mechanic's lien rights. Such formation shall include a correct statement of the record legal title to the property on which the Project is located, ually referred to as the site, and the Owner's interest therein.

2 INFORMATION AND SERVICES REQUIRED OF THE OWNER

§ 2.2.1 Prior to commencement of the Work, the Contractor may request in writing that the Owner provide reasonable evidence that the Owner has made financial arrangements to fulfill the Owner's obligations under the Contract. Thereafter, the Contractor may only request such evidence if (1) the Owner fails to make payments to the Contractor as the Contract Documents require; (2) a change in the Work materially changes the Contract Sum; or (3) the Contractor identifies in writing a reasonable concern regarding the Owner's ability to make payment when due. The Owner shall furnish such evidence as a condition precedent to commencement or continuation of the Work or the

Init. 1

AIA Document A2011 - 2007. Copyright @ 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 10 reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No 2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

portion of the Work affected by a material change. After the Owner furnishes the evidence, the Owner shall not materially vary such financial arrangements without prior notice to the Contractor.

§ 2.2.2 Except for permits and fees that are the responsibility of the Contractor under the Contract Documents, including those required under Section 3.7.1, the Owner shall secure and pay for necessary approvals, easements, assessments and charges required for construction, use or occupancy of permanent structures or for permanent changes in existing facilities.

§ 2.2.3 The Owner shall furnish surveys describing physical characteristics, legal limitations and utility location for the site of the Project, and a legal description of the site. The Contractor shall be entitled to rely on the accuracy information furnished by the Owner but shall exercise proper precautions relating to the safe performance of the Work.

§ 2.2.4 The Owner shall furnish information or services required of the Owner by the Contract Docume. reasonable promptness. The Owner shall also furnish any other information or services under the Owner's control and relevant to the Contractor's performance of the Work with reasonable promptness after receiving the Contractor's written request for such information or services.

ine Contractor one copy of § 2.2.5 Unless otherwise provided in the Contract Documents, the Owner shall furnish te the Contract Documents for purposes of making reproductions pursuant to Section

§ 2.3 OWNER'S RIGHT TO STOP THE WORK

If the Contractor fails to correct Work that is not in accordance with the Koujements of the Contract Documents as required by Section 12.2 or repeatedly fails to carry out Work in accordance with the Contract Documents, the Owner may issue a written order to the Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, the right of the Owner to stop the Worl, shail not give rise to a duty on the part of the Owner to exercise this right for the benefit of the Contractor or any other person or entity, except to the extent required by Section 6.1.3.

§ 2.4 OWNER'S RIGHT TO CARRY OUT THE WODER

If the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents and fails within a ten-day period after receipt of written etic from the Owner to commence and continue correction of such default or neglect with diligence and promptiess, the Owner may, without prejudice to other remedies the Owner may have, correct such deficiencies. In such dise in appropriate Change Order shall be issued deducting from payments then or thereafter due the Contractor the reasonable cost of correcting such deficiencies, including Owner's expenses and compensation for the Architecus around a services made necessary by such default, neglect or failure. Such action by the Owner and amounts charged to the Contractor are both subject to prior approval of the Architect. If payments then or thereafter the Contractor are not sufficient to cover such amounts, the Contractor shall pay the difference to the Owner.

CONT ACTO **ARTICLE 3**

§ 3.1 GENERAL

§ 3.1.1 The Contractor is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Contractor shall be lawfully licensed, if required in the jurisdiction where the Project is located. The Contractor shall designate in writing a representative who shall have express authority tobind the Contractor with respect to all matters under this Contract. The term "Contractor" means the intractor or the Contractor's authorized representative.

the Contractor shall perform the Work in accordance with the Contract Documents.

§ 3.1.3 The Contractor shall not be relieved of obligations to perform the Work in accordance with the Contract Documents either by activities or duties of the Architect in the Architect's administration of the Contract, or by tests, inspections or approvals required or performed by persons or entities other than the Contractor.

Init.

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects, All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 11 reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

§ 3.2 REVIEW OF CONTRACT DOCUMENTS AND FIELD CONDITIONS BY CONTRACTOR

§ 3.2.1 Execution of the Contract by the Contractor is a representation that the Contractor has visited the site, become generally familiar with local conditions under which the Work is to be performed and correlated personal observations with requirements of the Contract Documents.

§ 3.2.2 Because the Contract Documents are complementary, the Contractor shall, before starting each portion of the Work, carefully study and compare the various Contract Documents relative to that portion of the Work, as well as the information furnished by the Owner pursuant to Section 2.2.3, shall take field measurements of any existing conditions related to that portion of the Work, and shall observe any conditions at the site affecting it. These obligations are for the purpose of facilitating coordination and construction by the Contractor and are not for the purpose of discovering errors, omissions, or inconsistencies in the Contract Documents; however, the Contracton and promptly report to the Architect any errors, inconsistencies or omissions discovered by or made known to the Contractor as a request for information in such form as the Architect may require. It is recognized that the contractor's review is made in the Contract Documents.

§ 3.2.3 The Contractor is not required to ascertain that the Contract Documents are in accordance with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, but the Contractor shall promptly report to the Architect any nonconformity discovered by or made known to the Contractor as a request for information in such form as the Architect may require.

§ 3.2.4 If the Contractor believes that additional cost or time is involved becau e of charifications or instructions the Architect issues in response to the Contractor's notices or requests for information pursuant to Sections 3.2.2 or 3.2.3, the Contractor shall make Claims as provided in Article 15. If the Contractor fails to perform the obligations of Sections 3.2.2 or 3.2.3, the Contractor shall pay such costs and damages to the Owner as would have been avoided if the Contractor had performed such obligations. If the Contractor performs those obligations, the Contractor shall not be liable to the Owner or Architect for damages resulting from errors, inconsistencies or omissions in the Contract Documents, for differences between field measurements or conductors and the Contract Documents, or for nonconformities of the Contract Documents to applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities.

§ 3.3 SUPERVISION AND CONSTRUCTION PROSEDURES

§ 3.3.1 The Contractor shall supervise and arect the Work, using the Contractor's best skill and attention. The Contractor shall be solely responsible for, and have control over, construction means, methods, techniques, sequences and procedures and for coordinating an orbans of the Work under the Contract, unless the Contract Documents give other specific instructions concerning these matters. If the Contract Documents give specific instructions concerning these matters. If the Contract Documents give specific instructions concerning construction means, methods, techniques, sequences or procedures, the Contractor shall evaluate the jobsite safety thereof and, except as stated oeld w, shall be fully and solely responsible for the jobsite safety of such means, methods, techniques, sequences or procedures that such means, methods, techniques, sequences or procedures may not be sate the Contractor shall give timely written notice to the Owner and Architect and shall not proceed with that potion of the Work without further written instructions from the Architect. If the Contractor is then instructed to proceed with the required means, methods, techniques, sequences or procedures or procedures without acceptance of changes proposed by the Contractor, the Owner shall be solely responsible for any loss or damage arising solely from those Owner-required means, methods, techniques, sequences or procedures.

3.3.2 The Contractor shall be responsible to the Owner for acts and omissions of the Contractor's employees, so occurractors and their agents and employees, and other persons or entities performing portions of the Work for, or behall of, the Contractor or any of its Subcontractors.

3.3 The Contractor shall be responsible for inspection of portions of Work already performed to determine that such portions are in proper condition to receive subsequent Work.

§ 3.4 LABOR AND MATERIALS

§ 3.4.1 Unless otherwise provided in the Contract Documents, the Contractor shall provide and pay for labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services necessary for proper execution and completion of the Work, whether temporary or permanent and whether or not incorporated or to be incorporated in the Work.

1

Init

§ 3.4.2 Except in the case of minor changes in the Work authorized by the Architect in accordance with Sections 3.12.8 or 7.4, the Contractor may make substitutions only with the consent of the Owner, after evaluation by the Architect and in accordance with a Change Order or Construction Change Directive.

§ 3.4.3 The Contractor shall enforce strict discipline and good order among the Contractor's employees and other persons carrying out the Work. The Contractor shall not permit employment of unfit persons or persons not properly skilled in tasks assigned to them.

§ 3.5 WARRANTY

The Contractor warrants to the Owner and Architect that materials and equipment furnished under the Contract of good quality and new unless the Contract Documents require or permit otherwise. The Contractor further w that the Work will conform to the requirements of the Contract Documents and will be free from defects, except for those inherent in the quality of the Work the Contract Documents require or permit. Work, materials, or equipment not conforming to these requirements may be considered defective. The Contractor's warranty excludes remedy for damage or defect caused by abuse, alterations to the Work not executed by the Contractor, improper or insufficient maintenance, improper operation, or normal wear and tear and normal usage. If required by the Architect, the Contractor shall furnish satisfactory evidence as to the kind and quality of materials and equipment.

§ 3.6 TAXES

Init.

The Contractor shall pay sales, consumer, use and similar taxes for the Work provided by the Contractor that are legally enacted when bids are received or negotiations concluded, whether or not t effective or merely scheduled to go into effect.

§ 3.7 PERMITS, FEES, NOTICES AND COMPLIANCE WITH LAW

§ 3.7.1 Unless otherwise provided in the Contract Documents, the Contractor shall secure and pay for the building permit as well as for other permits, fees, licenses, and inspectious by generating agencies necessary for proper after execution of the Contract and legally required execution and completion of the Work that are customarily at the time bids are received or negotiations concluded.

§ 3.7.2 The Contractor shall comply with and give notice required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorizes applicable to performance of the Work.

§ 3.7.3 If the Contractor performs Work Knowing into be contrary to applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of provic authorities, the Contractor shall assume appropriate responsibility for such Work and shall bear the costs attributable to correction.

§ 3.7.4 Concealed or Unknown Conditions. If the Contractor encounters conditions at the site that are (1) subsurface or otherwise concealed physical conditions that differ materially from those indicated in the Contract Documents or (2) unknown physical conditions of an unusual nature, that differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Contract Documents, the Contractor shall formally povide notice to the Owner and the Architect before conditions are disturbed and in no event later than 21 mays after first observance of the conditions. The Architect will promptly investigate such conditions and, it he Architect determines that they differ materially and cause an increase or decrease in the Contracter's cost of, or time required for, performance of any part of the Work, will recommend an equitable adjustment the Contract Sum or Contract Time, or both. If the Architect determines that the conditions at the site are ng materially different from those indicated in the Contract Documents and that no change in the terms of the Contract iustifical, the Architect shall promptly notify the Owner and Contractor in writing, stating the reasons. If either party lise the Architect's determination or recommendation, that party may proceed as provided in Article 15.

§ 3.7.5 If, in the course of the Work, the Contractor encounters human remains or recognizes the existence of burial markers, archaeological sites or wetlands not indicated in the Contract Documents, the Contractor shall immediately suspend any operations that would affect them and shall notify the Owner and Architect. Upon receipt of such notice, the Owner shall promptly take any action necessary to obtain governmental authorization required to resume the operations. The Contractor shall continue to suspend such operations until otherwise instructed by the Owner but shall continue with all other operations that do not affect those remains or features. Requests for adjustments in the Contract Sum and Contract Time arising from the existence of such remains or features may be made as provided in Article 15.

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties, Unauthorized 13 reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. **User Notes:**

§ 3.8 ALLOWANCES

§ 3.8.1 The Contractor shall include in the Contract Sum all allowances stated in the Contract Documents. Items covered by allowances shall be supplied for such amounts and by such persons or entities as the Owner may direct, but the Contractor shall not be required to employ persons or entities to whom the Contractor has reasonable objection.

§ 3.8.2 Unless otherwise provided in the Contract Documents,

- Allowances shall cover the cost to the Contractor of materials and equipment delivered at the site an .1 all required taxes, less applicable trade discounts;
- Contractor's costs for unloading and handling at the site, labor, installation costs, overhead, profit a .2 other expenses contemplated for stated allowance amounts shall be included in the Contract Su not in the allowances; and
- Whenever costs are more than or less than allowances, the Contract Sum shall be adjusted accordingly .3 by Change Order. The amount of the Change Order shall reflect (1) the difference between actual costs and the allowances under Section 3.8.2.1 and (2) changes in Contractor's costs under Section 3.8.2.2.

§ 3.8.3 Materials and equipment under an allowance shall be selected by the Owner with reable promptness.

§ 3.9 SUPERINTENDENT

§ 3.9.1 The Contractor shall employ a competent superintendent and necessary assistants who shall be in attendance at the Project site during performance of the Work. The superintendent shall represent the Contractor, and communications given to the superintendent shall be as binding as if given to the Contractor.

§ 3.9.2 The Contractor, as soon as practicable after award of the Contract, Shall furnish in writing to the Owner through the Architect the name and qualifications of a proposed superintendent. The Architect may reply within 14 days to the Contractor in writing stating (1) whether the Owner or the Architect has reasonable objection to the proposed superintendent or (2) that the Architect requires additional time to review. Failure of the Architect to reply ble bjection. within the 14 day period shall constitute notice of no reas

§ 3.9.3 The Contractor shall not employ a proposed superintendent to whom the Owner or Architect has made reasonable and timely objection. The Contractor hall no change the superintendent without the Owner's consent, which shall not unreasonably be withheld or

§ 3.10 CONTRACTOR'S CONSTRUCTION SCHEDULES

§ 3.10.1 The Contractor, promptly accepted awarded the Contract, shall prepare and submit for the Owner's and Architect's information a Contractor's construction schedule for the Work. The schedule shall not exceed time limits current under the Contract Document, shall be revised at appropriate intervals as required by the conditions of the Work and Project, shall be clared to the entire Project to the extent required by the Contract Documents, and shall provide for expeditious and practice ble execution of the Work.

§ 3.10.2 The Contractor shall prepare a submittal schedule, promptly after being awarded the Contract and thereafter as necessary to maintain a surrent submittal schedule, and shall submit the schedule(s) for the Architect's approval. The Architect approval shall not unreasonably be delayed or withheld. The submittal schedule shall (1) be coordinated with the Contractor's construction schedule, and (2) allow the Architect reasonable time to review submittees. If the Contractor fails to submit a submittal schedule, the Contractor shall not be entitled to any increase in Contract Sum or extension of Contract Time based on the time required for review of submittals.

3.10.3 The Contractor shall perform the Work in general accordance with the most recent schedules submitted to the **Owner and Architect**.

§ 3.11 DOCUMENTS AND SAMPLES AT THE SITE

The Contractor shall maintain at the site for the Owner one copy of the Drawings, Specifications, Addenda, Change Orders and other Modifications, in good order and marked currently to indicate field changes and selections made during construction, and one copy of approved Shop Drawings, Product Data, Samples and similar required submittals. These shall be available to the Architect and shall be delivered to the Architect for submittal to the Owner upon completion of the Work as a record of the Work as constructed.

Init.

1

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved, WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 14 reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. **User Notes:**

§ 3.12 SHOP DRAWINGS, PRODUCT DATA AND SAMPLES

§ 3.12.1 Shop Drawings are drawings, diagrams, schedules and other data specially prepared for the Work by the Contractor or a Subcontractor, Sub-subcontractor, manufacturer, supplier or distributor to illustrate some portion of the Work.

§ 3.12.2 Product Data are illustrations, standard schedules, performance charts, instructions, brochures, diagrams and other information furnished by the Contractor to illustrate materials or equipment for some portion of the Work.

§ 3.12.3 Samples are physical examples that illustrate materials, equipment or workmanship and establish standards by which the Work will be judged.

§ 3.12.4 Shop Drawings, Product Data, Samples and similar submittals are not Contract Documents. Ther pr to demonstrate the way by which the Contractor proposes to conform to the information given and the design concept expressed in the Contract Documents for those portions of the Work for which the Contract Documents require submittals. Review by the Architect is subject to the limitations of Section 4.2.7. Informational submittals apon which the Architect is not expected to take responsive action may be so identified in the Contract Decuments. Submittals that are not required by the Contract Documents may be returned by the Architect without action

§ 3.12.5 The Contractor shall review for compliance with the Contract Documents, approve and submit to the Architect Shop Drawings, Product Data, Samples and similar submittals required by the Contract Documents in accordance with the submittal schedule approved by the Architect or, in the absence of an approved submittal schedule, with reasonable promptness and in such sequence as to cause no selar in Work or in the activities of the Owner or of separate contractors.

§ 3.12.6 By submitting Shop Drawings, Product Data, Samples and similar submittals, the Contractor represents to the Owner and Architect that the Contractor has (1) reviewed and approved them, (2) determined and verified materials, field measurements and field construction criteria related therein, on with do so and (3) checked and coordinated the or of the Work and of the Contract Documents. information contained within such submittals with the require

§ 3.12.7 The Contractor shall perform no portion of me Volution the Contract Documents require submittal and review of Shop Drawings, Product Data, Samples r similar submittals until the respective submittal has been approved by the Architect.

§ 3.12.8 The Work shall be in accordance with approved submittals except that the Contractor shall not be relieved of responsibility for deviations from regarized to of the Contract Documents by the Architect's approval of Shop Drawings, Product Data, Samples of similar submittals unless the Contractor has specifically informed the Architect in writing of such deviation at the time of submittal and (1) the Architect has given written approval to the specific deviation as a minor change in the Work, or (2) a Change Order or Construction Change Directive has been issued authorizing the deviation. Necconsactor shall not be relieved of responsibility for errors or omissions in Shop Drawings, Product Dra, Sum les or similar submittals by the Architect's approval thereof.

§ 3.12.9 The Contractor short direct specific attention, in writing or on resubmitted Shop Drawings, Product Data. Samples or similar submittals, to revisions other than those requested by the Architect on previous submittals. In the absence of such witten notice, the Architect's approval of a resubmission shall not apply to such revisions.

§ 3.12.10 Contractor shall not be required to provide professional services that constitute the practice of architecture of engineering unless such services are specifically required by the Contract Documents for a portion of e Work or unless the Contractor needs to provide such services in order to carry out the Contractor's responsibilities instruction means, methods, techniques, sequences and procedures. The Contractor shall not be required to vide professional services in violation of applicable law. If professional design services or certifications by a design professional related to systems, materials or equipment are specifically required of the Contractor by the Contract Documents, the Owner and the Architect will specify all performance and design criteria that such services must satisfy. The Contractor shall cause such services or certifications to be provided by a properly licensed design professional, whose signature and seal shall appear on all drawings, calculations, specifications, certifications, Shop Drawings and other submittals prepared by such professional. Shop Drawings and other submittals related to the Work designed or certified by such professional, if prepared by others, shall bear such professional's written approval when submitted to the Architect. The Owner and the Architect shall be entitled to rely upon the adequacy, accuracy and

Init.

1

AlA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 15 Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the reproduction or distribution of this AIA® maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

completeness of the services, certifications and approvals performed or provided by such design professionals, provided the Owner and Architect have specified to the Contractor all performance and design criteria that such services must satisfy. Pursuant to this Section 3.12.10, the Architect will review, approve or take other appropriate action on submittals only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Contractor shall not be responsible for the adequacy of the performance and design criteria specified in the Contract Documents.

§ 3.13 USE OF SITE

The Contractor shall confine operations at the site to areas permitted by applicable laws, statutes, ordinances, co rules and regulations, and lawful orders of public authorities and the Contract Documents and shall not unreading encumber the site with materials or equipment.

§ 3.14 CUTTING AND PATCHING

§ 3.14.1 The Contractor shall be responsible for cutting, fitting or patching required to complete the Work or to make its parts fit together properly. All areas requiring cutting, fitting and patching shall be restored to the condi on existing prior to the cutting, fitting and patching, unless otherwise required by the Contract Documents.

§ 3.14.2 The Contractor shall not damage or endanger a portion of the Work or fully or parially completed construction of the Owner or separate contractors by cutting, patching or otherwise allowing such construction, or by excavation. The Contractor shall not cut or otherwise alter such construction by the curve or a separate contractor except with written consent of the Owner and of such separate contractor; such consent shall not be unreasonably withheld. The Contractor shall not unreasonably withhold from the Owner or aseptante contractor the Contractor's consent to cutting or otherwise altering the Work.

§ 3.15 CLEANING UP

§ 3.15.1 The Contractor shall keep the premises and surrounding trea free from accumulation of waste materials or rubbish caused by operations under the Contract. At completion of the Work, the Contractor shall remove waste machinery and surplus materials from and about materials, rubbish, the Contractor's tools, construction equipment the Project.

§ 3.15.2 If the Contractor fails to clean up as provided in the Contract Documents, the Owner may do so and Owner shall be entitled to reimbursement from the Connector

§ 3.16 ACCESS TO WORK

The Contractor shall provide the Owner and Architect access to the Work in preparation and progress wherever located.

§ 3.17 ROYALTIES, PATENZS AND COPYRIGHTS

The Contractor shall pay an regames and license fees. The Contractor shall defend suits or claims for infringement of copyrights and patent right, and shall hold the Owner and Architect harmless from loss on account thereof, but shall not be responsible for such defense or loss when a particular design, process or product of a particular manufacturer or manufacturers is required by the Contract Documents, or where the copyright violations are contained in Drawings, Specifications or other documents prepared by the Owner or Architect. However, if the Contractor has reason to believe that the required design, process or product is an infringement of a copyright or a patent, the Contractor shall be responsible for such loss unless such information is promptly furnished to the Architect.

3.18 INDEMNIFICATION

3.18.1 To the fullest extent permitted by law the Contractor shall indemnify and hold harmless the Owner, Architect, chiect's consultants, and agents and employees of any of them from and against claims, damages, losses and enses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work, provided that such claim, damage, loss or expense is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property (other than the Work itself), but only to the extent caused by the negligent acts or omissions of the Contractor, a Subcontractor, anyone directly or indirectly employed by them or anyone for whose acts they may be liable, regardless of whether or not such claim, damage, loss or expense is caused in part by a party indemnified hereunder. Such obligation shall not be construed to negate, abridge, or reduce other rights or obligations of indemnity that would otherwise exist as to a party or person described in this Section 3.18.

Init.

1

AIA Document A201™ – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 16 reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. **User Notes:**

§ 3.18.2 In claims against any person or entity indemnified under this Section 3.18 by an employee of the Contractor, a Subcontractor, anyone directly or indirectly employed by them or anyone for whose acts they may be liable, the indemnification obligation under Section 3.18.1 shall not be limited by a limitation on amount or type of damages, compensation or benefits payable by or for the Contractor or a Subcontractor under workers' compensation acts, disability benefit acts or other employee benefit acts.

ARTICLE 4 ARCHITECT

§ 4.1 GENERAL

§ 4.1.1 The Owner shall retain an architect lawfully licensed to practice architecture or an entity lawfully practicing architecture in the jurisdiction where the Project is located. That person or entity is identified as the Architecture the Agreement and is referred to throughout the Contract Documents as if singular in number.

§ 4.1.2 Duties, responsibilities and limitations of authority of the Architect as set forth in the Contract Document not be restricted, modified or extended without written consent of the Owner, Contractor and Architect. Onsent shall not be unreasonably withheld.

§ 4.1,3 If the employment of the Architect is terminated, the Owner shall employ a successor chitect as to whom the Contractor has no reasonable objection and whose status under the Contract Documents shall be that of the Architect.

§ 4.2 ADMINISTRATION OF THE CONTRACT

§ 4.2.1 The Architect will provide administration of the Contract as described in the Contract Documents and will be an Owner's representative during construction until the date the Architect issue the Scal Certificate for Payment. The Architect will have authority to act on behalf of the Owner only to the extent provided in the Contract Documents.

§ 4.2.2 The Architect will visit the site at intervals appropriate to the stage of construction, or as otherwise agreed with the Owner, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine in general if the Work observed is being performed in a member indicating that the Work, when fully completed, will be in accordance with the Contract Documents between, the Architect will not be required to make exhaustive or continuous on-site inspections to check the quarty or quantity of the Work. The Architect will not have control over, charge of, or responsibility for, the construction means, methods, techniques, sequences or procedures, or for the safety precautions and programs in connection with the Work, since these are solely the Contractor's rights and responsibilities under the Contract Document, occupied provided in Section 3.3.1.

§ 4.2.3 On the basis of the site visits, the Archiece will keep the Owner reasonably informed about the progress and quality of the portion of the Work competed and report to the Owner (1) known deviations from the Contract Documents and from the most receive onstruction schedule submitted by the Contractor, and (2) defects and deficiencies observed in the Work. The A chitect will not be responsible for the Contractor's failure to perform the Work in accordance with the requirements of the Contract Documents. The Architect will not have control over or charge of and will not be repensible for acts or omissions of the Contractor, Subcontractors, or their agents or employees, or any other persons or entities performing portions of the Work.

§ 4.2.4 COMMUNICATIONS, ACILITATING CONTRACT ADMINISTRATION

Except as other vise provided in the Contract Documents or when direct communications have been specially authorized the Owner and Contractor shall endeavor to communicate with each other through the Architect about matters mising out of or relating to the Contract. Communications by and with the Architect's consultants shall be through the chitect. Communications by and with Subcontractors and material suppliers shall be through the intractor. Communications by and with separate contractors shall be through the Owner.

Based on the Architect's evaluations of the Contractor's Applications for Payment, the Architect will review certify the amounts due the Contractor and will issue Certificates for Payment in such amounts.

§ 4.2.6 The Architect has authority to reject Work that does not conform to the Contract Documents. Whenever the Architect considers it necessary or advisable, the Architect will have authority to require inspection or testing of the Work in accordance with Sections 13.5.2 and 13.5.3, whether or not such Work is fabricated, installed or completed. However, neither this authority of the Architect nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect to the Contractor, Subcontractors, material and equipment suppliers, their agents or employees, or other persons or entities performing portions of the Work.

AlA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 17 reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resate. User Notes:

§ 4.2.7 The Architect will review and approve, or take other appropriate action upon, the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Architect's action will be taken in accordance with the submittal schedule approved by the Architect or, in the absence of an approved submittal schedule, with reasonable promptness while allowing sufficient time in the Architect's professional judgment to permit adequate review. Review of such submittals is not conducted for the purpose of determining the accuracy and completeness of other details such as dimensions and quantities, or for substantiating instructions for installation or performance of equipment or systems, all of which remain the responsibility of the Contractor as required by the Contract Documents. The Architect's review of the Contractor's submittals shall not relieve the Contractor of the obligations under Sections 3.3, 3.5 and 3.12. The Architect's review shall not constitute approval of safety preca or, unless otherwise specifically stated by the Architect, of any construction means, methods, techniques, reque procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the it em is a component.

§ 4.2.8 The Architect will prepare Change Orders and Construction Change Directives, and may authorize minor changes in the Work as provided in Section 7.4. The Architect will investigate and make determinations and recommendations regarding concealed and unknown conditions as provided in Section 3.74.

§ 4.2.9 The Architect will conduct inspections to determine the date or dates of Substantia Completion and the date of final completion; issue Certificates of Substantial Completion pursuant to Section 9.8; eceive and forward to the Owner, for the Owner's review and records, written warranties and related accumenterequired by the Contract and assembled by the Contractor pursuant to Section 9.10; and issue a final Continent for Payment pursuant to Section 9.10.

§ 4.2.10 If the Owner and Architect agree, the Architect will proved one br more project representatives to assist in carrying out the Architect's responsibilities at the site. The duties, responsibilities and limitations of authority of such project representatives shall be as set forth in an exhibit to resince porated in the Contract Documents.

§ 4.2.11 The Architect will interpret and decide maters conversing performance under, and requirements of, the Contract Documents on written request of either Se Owner or Contractor. The Architect's response to such requests will be made in writing within any time limit an ecoup n or otherwise with reasonable promptness.

§ 4.2.12 Interpretations and decisions of the Architect will be consistent with the intent of, and reasonably inferable from, the Contract Documents and wires in tring or in the form of drawings. When making such interpretations and decisions, the Architect will endeave the secure faithful performance by both Owner and Contractor, will not show partiality to either and will not be liable for results of interpretations or decisions rendered in good faith.

§ 4.2.13 The Architect's devisions on matters relating to aesthetic effect will be final if consistent with the intent expressed in the Contract Documents.

§ 4.2.14 The Architect will eview and respond to requests for information about the Contract Documents. The Architect's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness. If appropriate, the Architect will prepare and issue supplemental Drawings and Specifications in response to the requests for information.

SUBCONTRACTORS **TICL** 5 5.1 DE INITIONS

Subcontractor is a person or entity who has a direct contract with the Contractor to perform a portion of the sk at the site. The term "Subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Subcontractor or an authorized representative of the Subcontractor. The term "Subcontractor" does not include a separate contractor or subcontractors of a separate contractor.

§ 5.1.2 A Sub-subcontractor is a person or entity who has a direct or indirect contract with a Subcontractor to perform a portion of the Work at the site. The term "Sub-subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Sub-subcontractor or an authorized representative of the Sub-subcontractor.

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 18 Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the reproduction or distribution of this AIA® maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. **User Notes:**

Init.

§ 5.2 AWARD OF SUBCONTRACTS AND OTHER CONTRACTS FOR PORTIONS OF THE WORK

§ 5.2.1 Unless otherwise stated in the Contract Documents or the bidding requirements, the Contractor, as soon as practicable after award of the Contract, shall furnish in writing to the Owner through the Architect the names of persons or entities (including those who are to furnish materials or equipment fabricated to a special design) proposed for each principal portion of the Work. The Architect may reply within 14 days to the Contractor in writing stating (1) whether the Owner or the Architect has reasonable objection to any such proposed person or entity or (2) that the Architect requires additional time for review. Failure of the Owner or Architect to reply within the 14-day period shall constitute notice of no reasonable objection.

§ 5.2.2 The Contractor shall not contract with a proposed person or entity to whom the Owner or Architect has made reasonable and timely objection. The Contractor shall not be required to contract with anyone to whom the Contract of the Contract with anyone to whom the Contract of the Con cto has made reasonable objection.

§ 5.2.3 If the Owner or Architect has reasonable objection to a person or entity proposed by the C intract Contractor shall propose another to whom the Owner or Architect has no reasonable objection_If the proposed but rejected Subcontractor was reasonably capable of performing the Work, the Contract Sum and Contract Time shall be increased or decreased by the difference, if any, occasioned by such change, and an appropriate change Order shall be issued before commencement of the substitute Subcontractor's Work. However, no in real in the Contract Sum or Contract Time shall be allowed for such change unless the Contractor has acted promote and responsively in submitting names as required.

§ 5.2.4 The Contractor shall not substitute a Subcontractor, person or entited selected if the Owner or Architect makes reasonable objection to such substitution.

§ 5.3 SUBCONTRACTUAL RELATIONS

By appropriate agreement, written where legally required for validity, the Contractor shall require each Subcontractor, to the extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by terms of the Contract Documents, and to assume toward the Contractor all the originations and responsibilities, including the responsibility for safety of the Subcontractor's Work, which the Contractor by these Documents, assumes toward the Owner and Architect, Each subcontract agreement shall preserve and potect the rights of the Owner and Architect under the Contract Documents with respect to the Work to performed by the Subcontractor so that subcontracting thereof will not prejudice such rights, and shall allow to the Subcontractor, unless specifically provided otherwise in the subcontract agreement, the benefit of all rights, remedies and redress against the Contractor that the Contractor, by the Contract Documents, has against the Owner. Where appropriate, the Contractor shall require each Subcontractor to enter into similar agreements with Stor ubcontractors. The Contractor shall make available to each proposed Subcontractor, prior to the execution of the subcontract agreement, copies of the Contract Documents to which the Subcontractor will be bound, and, upon written request of the Subcontractor, identify to the Subcontractor terms and conditions of the proposed subcontract agreement that may be at variance with the Contract Documents. Subcontractors will similarly name opies of applicable portions of such documents available to their respective proposed Sub-subcontractors.

§ 5.4 CONTINGENT ASSIGMENT OF SUBCONTRACTS

§ 5.4.1 Each succentract agreement for a portion of the Work is assigned by the Contractor to the Owner, provided that assignment is effective only after termination of the Contract by the Owner for cause pursuant to Section 14.2 and only for those subcontract agreements that the Owner accepts by notifying the Subcontractor and Contractor in writing; and

> assignment is subject to the prior rights of the surety, if any, obligated under bond relating to the Contract.

en the Owner accepts the assignment of a subcontract agreement, the Owner assumes the Contractor's rights and obligations under the subcontract.

§ 5.4.2 Upon such assignment, if the Work has been suspended for more than 30 days, the Subcontractor's compensation shall be equitably adjusted for increases in cost resulting from the suspension.

§ 5.4.3 Upon such assignment to the Owner under this Section 5.4, the Owner may further assign the subcontract to a successor contractor or other entity. If the Owner assigns the subcontract to a successor contractor or other entity, the

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved, WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 19 Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the reproduction or distribution of this AIA® maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

Init.

Owner shall nevertheless remain legally responsible for all of the successor contractor's obligations under the subcontract.

CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS ARTICLE 6

§ 6.1 OWNER'S RIGHT TO PERFORM CONSTRUCTION AND TO AWARD SEPARATE CONTRACTS § 6.1.1 The Owner reserves the right to perform construction or operations related to the Project with the Owner's own forces, and to award separate contracts in connection with other portions of the Project or other construction or operations on the site under Conditions of the Contract identical or substantially similar to these including those portions related to insurance and waiver of subrogation. If the Contractor claims that delay or additional cost is involved because of such action by the Owner, the Contractor shall make such Claim as provided in Article 1

§ 6.1.2 When separate contracts are awarded for different portions of the Project or other construction or the the site, the term "Contractor" in the Contract Documents in each case shall mean the Contractor who executes, ach separate Owner-Contractor Agreement.

§ 6.1.3 The Owner shall provide for coordination of the activities of the Owner's own forcer and of each separate contractor with the Work of the Contractor, who shall cooperate with them. The Contractor shall participate with other separate contractors and the Owner in reviewing their construction schedules. The Contractor shall make any revisions to the construction schedule deemed necessary after a joint review and mutual agreement inconstruction schedules shall then constitute the schedules to be used by the Contractor, separate contractors and the Owner until subsequently revised.

§ 6.1.4 Unless otherwise provided in the Contract Documents, when the Outer performs construction or operations related to the Project with the Owner's own forces, the Owner shall be deemed to be subject to the same obligations and to have the same rights that apply to the Contractor under the conditions of the Contract, including, without excluding others, those stated in Article 3, this Article 6 and Articles 10, 1 and 12.

§ 6.2 MUTUAL RESPONSIBILITY

§ 6.2.1 The Contractor shall afford the Owner and separate contractors reasonable opportunity for introduction and storage of their materials and equipment and performance of their activities, and shall connect and coordinate the Contractor's construction and operations with there as required by the Contract Documents.

§ 6.2.2 If part of the Contractor's Work depends for proper execution or results upon construction or operations by the Owner or a separate contractor, the Contractor shall, prior to proceeding with that portion of the Work, promptly report to the Architect apparent discrepancies or detects in such other construction that would render it unsuitable for such proper execution and results. Failure of me Contractor so to report shall constitute an acknowledgment that the Owner's or separate contractor's completed or partially completed construction is fit and proper to receive the Contractor's Work, except a to efects not then reasonably discoverable.

§ 6.2.3 The Contractor shart remburse the Owner for costs the Owner incurs that are payable to a separate contractor because of the Contractor's celays, improperly timed activities or defective construction. The Owner shall be responsible to the Connector for costs the Contractor incurs because of a separate contractor's delays, improperly timed activities, damage to the Work or defective construction.

Contractor shall promptly remedy damage the Contractor wrongfully causes to completed or partially § 6.2. completed construction or to property of the Owner or separate contractors as provided in Section 10.2.5.

.2.5 The Owner and each separate contractor shall have the same responsibilities for cutting and patching as are ed for the Contractor in Section 3.14.

§ 6.3 OWNER'S RIGHT TO CLEAN UP

If a dispute arises among the Contractor, separate contractors and the Owner as to the responsibility under their respective contracts for maintaining the premises and surrounding area free from waste materials and rubbish, the Owner may clean up and the Architect will allocate the cost among those responsible.

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 20 Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the reproduction or distribution of this AIA[®] maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

ARTICLE 7 CHANGES IN THE WORK

§ 7.1 GENERAL

§ 7.1.1 Changes in the Work may be accomplished after execution of the Contract, and without invalidating the Contract, by Change Order, Construction Change Directive or order for a minor change in the Work, subject to the limitations stated in this Article 7 and elsewhere in the Contract Documents.

§ 7.1.2 A Change Order shall be based upon agreement among the Owner, Contractor and Architect; a Construction Change Directive requires agreement by the Owner and Architect and may or may not be agreed to by the Contractor an order for a minor change in the Work may be issued by the Architect alone.

§ 7.1.3 Changes in the Work shall be performed under applicable provisions of the Contract Documents, Contractor shall proceed promptly, unless otherwise provided in the Change Order, Construction Change Dir order for a minor change in the Work.

§ 7.2 CHANGE ORDERS

Init.

I

§ 7.2.1 A Change Order is a written instrument prepared by the Architect and signed by the Own **Contractor** and Architect stating their agreement upon all of the following:

- The change in the Work; .1
- The amount of the adjustment, if any, in the Contract Sum; and .2
- .3 The extent of the adjustment, if any, in the Contract Time.

§ 7.3 CONSTRUCTION CHANGE DIRECTIVES

§ 7.3.1 A Construction Change Directive is a written order prepared by the chitect and signed by the Owner and Architect, directing a change in the Work prior to agreement on adjustment, Kany, in the Contract Sum or Contract Time, or both. The Owner may by Construction Change Directive without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of addition deletions or other revisions, the Contract Sum and Contract Time being adjusted accordingly.

§ 7.3.2 A Construction Change Directive shall be used in the bsence of total agreement on the terms of a Change Order.

§ 7.3.3 If the Construction Change Directive an adjustment to the Contract Sum, the adjustment shall be based on one of the following methods:

- Mutual acceptance of a lunin sum operly itemized and supported by sufficient substantiating data to .1 permit evaluation;
- .2 Contract Documents or subsequently agreed upon; Unit prices stated in
- Cost to be determined in a nanner agreed upon by the parties and a mutually acceptable fixed or .3 percentage fee; o
- .4 As provided in section 7.3.7.

§ 7.3.4 If unit prices are stated in the Contract Documents or subsequently agreed upon, and if quantities originally contemplated ar materially changed in a proposed Change Order or Construction Change Directive so that application of Schunit prices to quantities of Work proposed will cause substantial inequity to the Owner or Contractor, the applicable unit prices shall be equitably adjusted.

§ 7.3.5 Upon receipt of a Construction Change Directive, the Contractor shall promptly proceed with the change in the ork involved and advise the Architect of the Contractor's agreement or disagreement with the method, if any, pyided in the Construction Change Directive for determining the proposed adjustment in the Contract Sum or ona del Time.

§ 7.3.6 A Construction Change Directive signed by the Contractor indicates the Contractor's agreement therewith, including adjustment in Contract Sum and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a Change Order.

§ 7.3.7 If the Contractor does not respond promptly or disagrees with the method for adjustment in the Contract Sum, the Architect shall determine the method and the adjustment on the basis of reasonable expenditures and savings of those performing the Work attributable to the change, including, in case of an increase in the Contract Sum, an amount

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 21 reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

for overhead and profit as set forth in the Agreement, or if no such amount is set forth in the Agreement, a reasonable amount. In such case, and also under Section 7.3.3.3, the Contractor shall keep and present, in such form as the Architect may prescribe, an itemized accounting together with appropriate supporting data. Unless otherwise provided in the Contract Documents, costs for the purposes of this Section 7.3.7 shall be limited to the following:

- Costs of labor, including social security, old age and unemployment insurance, fringe benefits required .1 by agreement or custom, and workers' compensation insurance;
- Costs of materials, supplies and equipment, including cost of transportation, whether incorporated or .2 consumed:
- Rental costs of machinery and equipment, exclusive of hand tools, whether rented from the Contracto .3 or others:
- Costs of premiums for all bonds and insurance, permit fees, and sales, use or similar taxes related the .4 Work; and
- Additional costs of supervision and field office personnel directly attributable to the change .5

§ 7.3.8 The amount of credit to be allowed by the Contractor to the Owner for a deletion or change that results in a net decrease in the Contract Sum shall be actual net cost as confirmed by the Architect. When both a dimens and credits covering related Work or substitutions are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change.

§ 7.3.9 Pending final determination of the total cost of a Construction Change Directive in the Owner, the Contractor may request payment for Work completed under the Construction Change Directive in applications for Payment. The Architect will make an interim determination for purposes of monthly certification for payment for those costs and certify for payment the amount that the Architect determines, in the Architect's professional judgment, to be reasonably justified. The Architect's interim determination of cost shall adjust the Contract Sum on the same basis as a Change Order, subject to the right of either party to disagree and assert a Claim in accordance with Article 15.

§ 7.3.10 When the Owner and Contractor agree with a determination made by the Architect concerning the ice each agreement upon the adjustments, such adjustments in the Contract Sum and Contract Time, or one agreement shall be effective immediately and the Architect will prepare a Change Order. Change Orders may be issued for all or any part of a Construction Change Directive

§ 7.4 MINOR CHANGES IN THE WORK

The Architect has authority to order minor changes in the Work not involving adjustment in the Contract Sum or extension of the Contract Time and not inconsistent with the intent of the Contract Documents. Such changes will be effected by written order signed by the architect and shall be binding on the Owner and Contractor.

ARTICLE 8 TIME

§ 8.1 DEFINITIONS

construct Time is the period of time, including authorized adjustments, allotted in § 8.1.1 Unless otherwise privi the Contract Documents for Substantial Completion of the Work.

mercement of the Work is the date established in the Agreement. § 8.1.2 The date of con

§ 8.1.3 The date Substantial Completion is the date certified by the Architect in accordance with Section 9.8.

erm "day" as used in the Contract Documents shall mean calendar day unless otherwise specifically § 8.1.4 The inea

OGRESS AND COMPLETION

2.1 Time limits stated in the Contract Documents are of the essence of the Contract. By executing the Agreement the Contractor confirms that the Contract Time is a reasonable period for performing the Work.

§ 8.2.2 The Contractor shall not knowingly, except by agreement or instruction of the Owner in writing, prematurely commence operations on the site or elsewhere prior to the effective date of insurance required by Article 11 to be furnished by the Contractor and Owner. The date of commencement of the Work shall not be changed by the effective date of such insurance.

1

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 22 Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the reproduction or distribution of this AIA® maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

§ 8.2.3 The Contractor shall proceed expeditiously with adequate forces and shall achieve Substantial Completion within the Contract Time.

§ 8.3 DELAYS AND EXTENSIONS OF TIME

§ 8.3.1 If the Contractor is delayed at any time in the commencement or progress of the Work by an act or neglect of the Owner or Architect, or of an employee of either, or of a separate contractor employed by the Owner; or by changes ordered in the Work; or by labor disputes, fire, unusual delay in deliveries, unavoidable casualties or other causes beyond the Contractor's control; or by delay authorized by the Owner pending mediation and arbitration; or by oth causes that the Architect determines may justify delay, then the Contract Time shall be extended by Change Ord such reasonable time as the Architect may determine.

§ 8.3.2 Claims relating to time shall be made in accordance with applicable provisions of Article 15.

§ 8.3.3 This Section 8.3 does not preclude recovery of damages for delay by either party under oth a proisions of the Contract Documents.

ARTICLE 9 PAYMENTS AND COMPLETION

§ 9.1 CONTRACT SUM

The Contract Sum is stated in the Agreement and, including authorized adjustments, is the totabamount payable by the Owner to the Contractor for performance of the Work under the Contract Document

§ 9.2 SCHEDULE OF VALUES

Where the Contract is based on a stipulated sum or Guaranteed Maximum Price, the Contractor shall submit to the Architect, before the first Application for Payment, a schedule of values allocating the entire Contract Sum to the various portions of the Work and prepared in such form and supported by such data to substantiate its accuracy as the Architect may require. This schedule, unless objected to by the Architect shall be used as a basis for reviewing the Contractor's Applications for Payment.

§ 9.3 APPLICATIONS FOR PAYMENT

§ 9.3.1 At least ten days before the date established toneact progress payment, the Contractor shall submit to the Architect an itemized Application for Payment papared in accordance with the schedule of values, if required under Section 9.2, for completed portions of the Work Such application shall be notarized, if required, and supported by such data substantiating the Contractor's right to pryment as the Owner or Architect may require, such as copies of requisitions from Subcontractors and material suppliers, and shall reflect retainage if provided for in the Contract Documents.

§ 9.3.1.1 As provided in Section 7.3. Sugar applications may include requests for payment on account of changes in the Work that have been properly authorized by Construction Change Directives, or by interim determinations of the Architect, but not yet included an Change Orders.

§ 9.3.1.2 Applications for Payment shall not include requests for payment for portions of the Work for which the Contractor does not in end to pay a Subcontractor or material supplier, unless such Work has been performed by others whom the Contractor intends to pay.

§ 9.3.2 cless otherwise provided in the Contract Documents, payments shall be made on account of materials and equipment elivered and suitably stored at the site for subsequent incorporation in the Work. If approved in advance by the where payment may similarly be made for materials and equipment suitably stored off the site at a location reed upon in writing. Payment for materials and equipment stored on or off the site shall be conditioned upon compliance by the Contractor with procedures satisfactory to the Owner to establish the Owner's title to such materials equipment or otherwise protect the Owner's interest, and shall include the costs of applicable insurance, storage and transportation to the site for such materials and equipment stored off the site.

§ 9.3.3 The Contractor warrants that title to all Work covered by an Application for Payment will pass to the Owner no later than the time of payment. The Contractor further warrants that upon submittal of an Application for Payment all Work for which Certificates for Payment have been previously issued and payments received from the Owner shall, to the best of the Contractor's knowledge, information and belief, be free and clear of liens, claims, security interests or

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treatles. Unauthorized 23 reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. **User Notes:**

encumbrances in favor of the Contractor, Subcontractors, material suppliers, or other persons or entities making a claim by reason of having provided labor, materials and equipment relating to the Work.

§ 9.4 CERTIFICATES FOR PAYMENT

§ 9.4.1 The Architect will, within seven days after receipt of the Contractor's Application for Payment, either issue to the Owner a Certificate for Payment, with a copy to the Contractor, for such amount as the Architect determines is properly due, or notify the Contractor and Owner in writing of the Architect's reasons for withholding certification in whole or in part as provided in Section 9.5.1.

§ 9.4.2 The issuance of a Certificate for Payment will constitute a representation by the Architect to the Owner base on the Architect's evaluation of the Work and the data comprising the Application for Payment, that, to the hest the Architect's knowledge, information and belief, the Work has progressed to the point indicated and that the game the Work is in accordance with the Contract Documents. The foregoing representations are subject to en evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, to results of substantiat tests and inspections, to correction of minor deviations from the Contract Documents prior to completion and t b specific qualifications expressed by the Architect. The issuance of a Certificate for Payment will further constnute a representation that the Contractor is entitled to payment in the amount certified. However, the suance of a Certificate for Payment will not be a representation that the Architect has (1) made exhaustive or continuous n-site inspections to check the quality or quantity of the Work, (2) reviewed construction means, methods, reanniques, sequences or procedures, (3) reviewed copies of requisitions received from Subcontractors and material suppliers and other data requested by the Owner to substantiate the Contractor's right to payment, or (4) hade examination to ascertain how or for what purpose the Contractor has used money previously paid on account on the Contract Sum.

§ 9.5 DECISIONS TO WITHHOLD CERTIFICATION

§ 9.5.1 The Architect may withhold a Certificate for Payment in whole on in part, to the extent reasonably necessary to protect the Owner, if in the Architect's opinion the representations to the Owner required by Section 9.4.2 cannot be made. If the Architect is unable to certify payment in the amount of the Application, the Architect will notify the Contractor and Owner as provided in Section 9.4.1. If the Conar and Architect cannot agree on a revised amount, the Architect will promptly issue a Certificate for Paymen, for the amount for which the Architect is able to make such representations to the Owner. The Architect may also within the a Certificate for Payment or, because of subsequently discovered evidence, may nullify the whole or a part of a Certificate for Payment previously issued, to such extent as may be necessary in the Architect's opinion to protect the Owner from loss for which the Contractor is responsible, including loss resulting from acts and omissions described in Section 3.3.2, because of

- defective Work not remedied; .1
- third party claims file or reasonable evidence indicating probable filing of such claims unless security .2 acceptable to the Owner's provided by the Contractor;
- failure of the Contractor to make payments properly to Subcontractors or for labor, materials or .3 equipment;
- reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum; .4
- .5 damage to the Owner or a separate contractor;
- reasonable evalence that the Work will not be completed within the Contract Time, and that the unpaid .6 be ance would not be adequate to cover actual or liquidated damages for the anticipated delay; or
- .7 peated failure to carry out the Work in accordance with the Contract Documents.

en the above reasons for withholding certification are removed, certification will be made for amounts § 9.5.2 previously withheld.

.5.3 11 the Architect withholds certification for payment under Section 9.5.1.3, the Owner may, at its sole option, subjent checks to the Contractor and to any Subcontractor or material or equipment suppliers to whom the tractor failed to make payment for Work properly performed or material or equipment suitably delivered. If the Owner makes payments by joint check, the Owner shall notify the Architect and the Architect will reflect such payment on the next Certificate for Payment.

§ 9.6 PROGRESS PAYMENTS

§ 9.6.1 After the Architect has issued a Certificate for Payment, the Owner shall make payment in the manner and within the time provided in the Contract Documents, and shall so notify the Architect.

1

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 24 reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No 2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

§ 9.6.2 The Contractor shall pay each Subcontractor no later than seven days after receipt of payment from the Owner the amount to which the Subcontractor is entitled, reflecting percentages actually retained from payments to the Contractor on account of the Subcontractor's portion of the Work. The Contractor shall, by appropriate agreement with each Subcontractor, require each Subcontractor to make payments to Sub-subcontractors in a similar manner.

§ 9.6.3 The Architect will, on request, furnish to a Subcontractor, if practicable, information regarding percentages of completion or amounts applied for by the Contractor and action taken thereon by the Architect and Owner on account of portions of the Work done by such Subcontractor.

§ 9.6.4 The Owner has the right to request written evidence from the Contractor that the Contractor has properly p Subcontractors and material and equipment suppliers amounts paid by the Owner to the Contractor for subcont cted Work. If the Contractor fails to furnish such evidence within seven days, the Owner shall have the right o Subcontractors to ascertain whether they have been properly paid. Neither the Owner nor Architect shall have obligation to pay or to see to the payment of money to a Subcontractor, except as may otherwise be required by law.

§ 9.6.5 Contractor payments to material and equipment suppliers shall be treated in a manner similar to that provided in Sections 9.6.2, 9.6.3 and 9.6.4.

§ 9.6.6 A Certificate for Payment, a progress payment, or partial or entire use or occupancy on the Project by the Owner shall not constitute acceptance of Work not in accordance with the Contract Rocements.

§ 9.6.7 Unless the Contractor provides the Owner with a payment bond in me juli and sum of the Contract Sum, payments received by the Contractor for Work properly performed by Succontractors and suppliers shall be held by the Contractor for those Subcontractors or suppliers who performed Work or furnished materials, or both, under contract with the Contractor for which payment was made by the owner. Nothing contained herein shall require money to be placed in a separate account and not commingled with money of the Contractor, shall create any fiduciary liability or tort liability on the part of the Contractor for breach first of shall entitle any person or entity to an award ements of this provision. of punitive damages against the Contractor for breach of

§ 9.7 FAILURE OF PAYMENT

If the Architect does not issue a Certificate for Payment, prough no fault of the Contractor, within seven days after receipt of the Contractor's Application for Parment, or j the Owner does not pay the Contractor within seven days after the date established in the Contract D cumen's the amount certified by the Architect or awarded by binding dispute resolution, then the Contractor may, by on even additional days' written notice to the Owner and Architect, stop the Work until payment of the ancent awing has been received. The Contract Time shall be extended appropriately and the Contract Sumshart or increased by the amount of the Contractor's reasonable costs of shut-down, delay and start-up, plus increat as provided for in the Contract Documents.

§ 9.8 SUBSTANTIAL COMPLET

§ 9.8.1 Substantial Compution is the stage in the progress of the Work when the Work or designated portion thereof is sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work for its intended use,

§ 9.8.2 When the Contractor considers that the Work, or a portion thereof which the Owner agrees to accept separates is substantially complete, the Contractor shall prepare and submit to the Architect a comprehensive list of items to be completed or corrected prior to final payment. Failure to include an item on such list does not alter the ponsibility of the Contractor to complete all Work in accordance with the Contract Documents.

pon receipt of the Contractor's list, the Architect will make an inspection to determine whether the Work or ignated portion thereof is substantially complete. If the Architect's inspection discloses any item, whether or not included on the Contractor's list, which is not sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work or designated portion thereof for its intended use, the Contractor shall, before issuance of the Certificate of Substantial Completion, complete or correct such item upon notification by the Architect. In such case, the Contractor shall then submit a request for another inspection by the Architect to determine Substantial Completion.

Init.

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 25 reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

§ 9.8.4 When the Work or designated portion thereof is substantially complete, the Architect will prepare a Certificate of Substantial Completion that shall establish the date of Substantial Completion, shall establish responsibilities of the Owner and Contractor for security, maintenance, heat, utilities, damage to the Work and insurance, and shall fix the time within which the Contractor shall finish all items on the list accompanying the Certificate. Warranties required by the Contract Documents shall commence on the date of Substantial Completion of the Work or designated portion thereof unless otherwise provided in the Certificate of Substantial Completion.

§ 9.8.5 The Certificate of Substantial Completion shall be submitted to the Owner and Contractor for their written acceptance of responsibilities assigned to them in such Certificate. Upon such acceptance and consent of surety, jany the Owner shall make payment of retainage applying to such Work or designated portion thereof. Such payment shall be adjusted for Work that is incomplete or not in accordance with the requirements of the Contract Documents

§ 9.9 PARTIAL OCCUPANCY OR USE

§ 9.9.1 The Owner may occupy or use any completed or partially completed portion of the Work at any tage when such portion is designated by separate agreement with the Contractor, provided such occupancy obuse is consented to by the insurer as required under Section 11.3.1.5 and authorized by public authorities having jurisdiction over the Project. Such partial occupancy or use may commence whether or not the portion is substantially complete, provided the Owner and Contractor have accepted in writing the responsibilities assigned to each of the for payments, retainage, if any, security, maintenance, heat, utilities, damage to the Work and insurance, and have agreed in writing concerning the period for correction of the Work and commencement of warranties required by the Contract Documents. When the Contractor considers a portion substantially complete, the Contractor shall prepare and submit a list to the Architect as provided under Section 9.8.2. Consent of the Contractor to partial occupancy or use shall not be unreasonably withheld. The stage of the progress of the Work shall be determined by written agreement between the Owner and Contractor or, if no agreement is reached, by decision of the crchitect.

§ 9.9.2 Immediately prior to such partial occupancy or use, the Owner, Contractor and Architect shall jointly inspect the area to be occupied or portion of the Work to be used in order to determine and record the condition of the Work.

§ 9.9.3 Unless otherwise agreed upon, partial occupancy or use of a portion or portions of the Work shall not constitute acceptance of Work not complying with the requirements of the Contract Documents.

§ 9.10 FINAL COMPLETION AND FINAL PAYMENT

§ 9.10.1 Upon receipt of the Contractor's written natice that the Work is ready for final inspection and acceptance and upon receipt of a final Application for Payment, the Architect will promptly make such inspection and, when the Architect finds the Work acceptable ander the Contract Documents and the Contract fully performed, the Architect will promptly issue a final Certificate for nayment stating that to the best of the Architect's knowledge, information and belief, and on the basis of the Architert's on-site visits and inspections, the Work has been completed in accordance with terms and rong tions of the Contract Documents and that the entire balance found to be due the Contractor and noted in the first Certificate is due and payable. The Architect's final Certificate for Payment will constitute a further remember of that conditions listed in Section 9.10.2 as precedent to the Contractor's being entitled to final payment have been fulfilled.

§ 9.10.2 Neither fural payment nor any remaining retained percentage shall become due until the Contractor submits to the Architect (1) an affidavit that payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which the Owner or the Owner's property might be responsible or encumbered (less amounts withheld by Owner have been paid or otherwise satisfied, (2) a certificate evidencing that insurance required by the Contract Documents to remain in force after final payment is currently in effect and will not be canceled or allowed to expire til at last 30 days' prior written notice has been given to the Owner, (3) a written statement that the Contractor increase f no substantial reason that the insurance will not be renewable to cover the period required by the Contract suments, (4) consent of surety, if any, to final payment and (5), if required by the Owner, other data establishing payment or satisfaction of obligations, such as receipts, releases and waivers of liens, claims, security interests or encumbrances arising out of the Contract, to the extent and in such form as may be designated by the Owner. If a Subcontractor refuses to furnish a release or waiver required by the Owner, the Contractor may furnish a bond satisfactory to the Owner to indemnify the Owner against such lien. If such lien remains unsatisfied after payments are made, the Contractor shall refund to the Owner all money that the Owner may be compelled to pay in discharging such lien, including all costs and reasonable attorneys' fees.

Init.

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA* Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 26 reproduction or distribution of this AIA" Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale User Notes:

§ 9.10.3 If, after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Contractor or by issuance of Change Orders affecting final completion, and the Architect so confirms, the Owner shall, upon application by the Contractor and certification by the Architect, and without terminating the Contract, make payment of the balance due for that portion of the Work fully completed and accepted. If the remaining balance for Work not fully completed or corrected is less than retainage stipulated in the Contract Documents, and if bonds have been furnished, the written consent of surety to payment of the balance due for that portion of the Work fully completed and accepted shall be submitted by the Contractor to the Architect prior to certification of such payment. Such payment shall be made under terms and conditions governing final payment, except that it shall not constitute a waiver of claims.

§ 9.10.4 The making of final payment shall constitute a waiver of Claims by the Owner except those arising

- liens, Claims, security interests or encumbrances arising out of the Contract and unsettled .1
- failure of the Work to comply with the requirements of the Contract Documents; or .2
- terms of special warranties required by the Contract Documents. .3

§ 9.10.5 Acceptance of final payment by the Contractor, a Subcontractor or material supplier shall constitute a waiver sumsettled at the time of of claims by that payee except those previously made in writing and identified by that payee final Application for Payment.

PROTECTION OF PERSONS AND PROPERTY **ARTICLE 10**

§ 10.1 SAFETY PRECAUTIONS AND PROGRAMS

The Contractor shall be responsible for initiating, maintaining and supervi precautions and programs in ing connection with the performance of the Contract.

§ 10.2 SAFETY OF PERSONS AND PROPERTY

§ 10.2.1 The Contractor shall take reasonable precautions for safety of, and shall provide reasonable protection to prevent damage, injury or loss to

- employees on the Work and other persons be affected thereby; .1
- the Work and materials and equipment to be incorporated therein, whether in storage on or off the site, .2 under care, custody or control of the contractor or the Contractor's Subcontractors or Sub-subcontractors; and
- .3 other property at the site or adjacent hereto, such as trees, shrubs, lawns, walks, pavements, roadways, structures and utilities not designated for removal, relocation or replacement in the course of construction.

§ 10.2.2 The Contractor shall compress the notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful order or public authorities bearing on safety of persons or property or their protection from damage, iniary r loss.

§ 10.2.3 The Contractor snall sect and maintain, as required by existing conditions and performance of the Contract, reasonable safeguards for safety and protection, including posting danger signs and other warnings against hazards, promulgating sprety regulations and notifying owners and users of adjacent sites and utilities.

§ 10.2.4 When use or storage of explosives or other hazardous materials or equipment or unusual methods are necessary for execution of the Work, the Contractor shall exercise utmost care and carry on such activities under supervision of properly qualified personnel.

10.2.5 The Contractor shall promptly remedy damage and loss (other than damage or loss insured under property assume required by the Contract Documents) to property referred to in Sections 10.2.1.2 and 10.2.1.3 caused in ple or in part by the Contractor, a Subcontractor, a Sub-subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Contractor is responsible under Sections 10.2.1.2 and 10.2.1.3, except damage or loss attributable to acts or omissions of the Owner or Architect or anyone directly or indirectly employed by either of them, or by anyone for whose acts either of them may be liable, and not attributable to the fault or negligence of the Contractor. The foregoing obligations of the Contractor are in addition to the Contractor's obligations under Section 3.18.

1

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 27 reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

§ 10.2.6 The Contractor shall designate a responsible member of the Contractor's organization at the site whose duty shall be the prevention of accidents. This person shall be the Contractor's superintendent unless otherwise designated by the Contractor in writing to the Owner and Architect.

§ 10.2.7 The Contractor shall not permit any part of the construction or site to be loaded so as to cause damage or create an unsafe condition.

§ 10.2.8 INJURY OR DAMAGE TO PERSON OR PROPERTY

If either party suffers injury or damage to person or property because of an act or omission of the other party, or of others for whose acts such party is legally responsible, written notice of such injury or damage, whether or not insured shall be given to the other party within a reasonable time not exceeding 21 days after discovery. The notice shall provide sufficient detail to enable the other party to investigate the matter.

§ 10.3 HAZARDOUS MATERIALS

§ 10.3.1 The Contractor is responsible for compliance with any requirements included in the Contract Documents regarding hazardous materials. If the Contractor encounters a hazardous material or substance no addressed in the Contract Documents and if reasonable precautions will be inadequate to prevent foreseeable coally injury or death to persons resulting from a material or substance, including but not limited to asbestos or poly chlorinated biphenyl (PCB), encountered on the site by the Contractor, the Contractor shall, upon recognizing are condition, immediately stop Work in the affected area and report the condition to the Owner and Architect inwring.

§ 10.3.2 Upon receipt of the Contractor's written notice, the Owner shall obtain the services of a licensed laboratory to verify the presence or absence of the material or substance reported by the Contractor and, in the event such material or substance is found to be present, to cause it to be rendered harmless. Unless otherwise required by the Contract Documents, the Owner shall furnish in writing to the Contractor and Architect the names and qualifications of persons or entities who are to perform tests verifying the presence or absence of such material or substance or who are to perform tests verifying the presence or absence of such material or substance or who are to perform the task of removal or safe containment of such material or substance. The Contractor and the Architect will promptly reply to the Owner in writing stating whether or no cithe has reasonable objection to the persons or entities proposed by the Owner. If either the Contractor or Architect and the Architect have no reasonable objection. When the material or substance has been rendered harmles, Work in the affected area shall resume upon written agreement of the Owner and Contractor. By Change Order, the Contract Time shall be extended appropriately and the Contract Sum shall be increased in the amount of the Contractor's reasonable additional costs of shut-down, delay and start-up.

§ 10.3.3 To the fullest extent permitted by low, the Owner shall indemnify and hold harmless the Contractor, Subcontractors, Architect, Architect's oncultants and agents and employees of any of them from and against claims, damages, losses and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work in the ane ted area if in fact the material or substance presents the risk of bodily injury or death as described in Section 10.3.1 and has not been rendered harmless, provided that such claim, damage, loss or expense is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property (other than the Work in elf-except to the extent that such damage, loss or expense is due to the fault or negligence of the party seeking indemnity.

§ 10.3.4 the Owner shall not be responsible under this Section 10.3 for materials or substances the Contractor brings to the site unless such materials or substances are required by the Contract Documents. The Owner shall be responsible for materials or substances required by the Contract Documents, except to the extent of the Contractor's fault or nugligence in the use and handling of such materials or substances.

3.5 The Contractor shall indemnify the Owner for the cost and expense the Owner incurs (1) for remediation of a material or substance the Contractor brings to the site and negligently handles, or (2) where the Contractor fails to perform its obligations under Section 10.3.1, except to the extent that the cost and expense are due to the Owner's fault or negligence.

§ 10.3.6 If, without negligence on the part of the Contractor, the Contractor is held liable by a government agency for the cost of remediation of a hazardous material or substance solely by reason of performing Work as required by the Contract Documents, the Owner shall indemnify the Contractor for all cost and expense thereby incurred.

AIA Document A201TM – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA[®] Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes: (726494055)

1

§ 10.4 EMERGENCIES

In an emergency affecting safety of persons or property, the Contractor shall act, at the Contractor's discretion, to prevent threatened damage, injury or loss. Additional compensation or extension of time claimed by the Contractor on account of an emergency shall be determined as provided in Article 15 and Article 7.

ARTICLE 11 **INSURANCE AND BONDS**

§ 11.1 CONTRACTOR'S LIABILITY INSURANCE

§ 11.1.1 The Contractor shall purchase from and maintain in a company or companies lawfully authorized to do business in the jurisdiction in which the Project is located such insurance as will protect the Contractor from clams, forth below which may arise out of or result from the Contractor's operations and completed operations under Contract and for which the Contractor may be legally liable, whether such operations be by the Contractor or Subcontractor or by anyone directly or indirectly employed by any of them, or by anyone for whose acts any of them may be liable:

- Claims under workers' compensation, disability benefit and other similar employee enefit lets that are .1 applicable to the Work to be performed;
- death of the Claims for damages because of bodily injury, occupational sickness or diseas .2 Contractor's employees;
- Claims for damages because of bodily injury, sickness or disease, or death or any person other than the .3 Contractor's employees;
- .4 Claims for damages insured by usual personal injury liability coverage;
- Claims for damages, other than to the Work itself, because a number of destruction of tangible .5 property, including loss of use resulting therefrom;
- Claims for damages because of bodily injury, death of a person or property damage arising out of .6 ownership, maintenance or use of a motor vehicle;
- .7
- Claims for bodily injury or property damage arising out of completed operations; and Claims involving contractual liability insurance opplicable to the Contractor's obligations under .8 Section 3.18.

ritten for not less than limits of liability specified in the § 11.1.2 The insurance required by Section 11.1.1 shan be Contract Documents or required by law, whichever coverage is greater. Coverages, whether written on an occurrence or claims-made basis, shall be maintained without inerroption from the date of commencement of the Work until the date of final payment and termination of any coverage required to be maintained after final payment, and, with respect to the Contractor's completed operations coverage antil the expiration of the period for correction of Work or for such other period for maintenance of compresed operations coverage as specified in the Contract Documents.

§ 11.1.3 Certificates of insurance acceptable to the Owner shall be filed with the Owner prior to commencement of the Work and thereafter upon reveal or replacement of each required policy of insurance. These certificates and the insurance policies required whit Section 11.1 shall contain a provision that coverages afforded under the policies will not be canceled or ane well to expire until at least 30 days' prior written notice has been given to the Owner. An additional certificate evidencing continuation of liability coverage, including coverage for completed operations, shall be submitted with the inal application for Payment as required by Section 9.10.2 and thereafter upon renewal or replacement of uc coverage until the expiration of the time required by Section 11.1.2. Information concerning reduction of cover ge on account of revised limits or claims paid under the General Aggregate, or both, shall be furnished by the Contractor with reasonable promptness.

1.1. The Contractor shall cause the commercial liability coverage required by the Contract Documents to include the Opner, the Architect and the Architect's consultants as additional insureds for claims caused in whole or in part Contractor's negligent acts or omissions during the Contractor's operations; and (2) the Owner as an additional red for claims caused in whole or in part by the Contractor's negligent acts or omissions during the Contractor's completed operations.

§ 11.2 OWNER'S LIABILITY INSURANCE

The Owner shall be responsible for purchasing and maintaining the Owner's usual liability insurance.

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved, WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 29 reproduction or distribution of this AIA" Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

§ 11.3 PROPERTY INSURANCE

§ 11.3.1 Unless otherwise provided, the Owner shall purchase and maintain, in a company or companies lawfully authorized to do business in the jurisdiction in which the Project is located, property insurance written on a builder's risk "all-risk" or equivalent policy form in the amount of the initial Contract Sum, plus value of subsequent Contract Modifications and cost of materials supplied or installed by others, comprising total value for the entire Project at the site on a replacement cost basis without optional deductibles. Such property insurance shall be maintained, unless otherwise provided in the Contract Documents or otherwise agreed in writing by all persons and entities who are beneficiaries of such insurance, until final payment has been made as provided in Section 9.10 or until no person of entity other than the Owner has an insurable interest in the property required by this Section 11.3 to be covered whichever is later. This insurance shall include interests of the Owner, the Contractor, Subcontractors and Sub-subcontractors in the Project.

§ 11.3.1.1 Property insurance shall be on an "all-risk" or equivalent policy form and shall include, without limitation, insurance against the perils of fire (with extended coverage) and physical loss or damage including, with purduplication of coverage, theft, vandalism, malicious mischief, collapse, earthquake, flood, windsturm, falsework, testing and startup, temporary buildings and debris removal including demolition occasioner by inforcement of any applicable legal requirements, and shall cover reasonable compensation for Architect's and Contractor's services and expenses required as a result of such insured loss.

§ 11.3.1.2 If the Owner does not intend to purchase such property insurance required by the Contract and with all of the coverages in the amount described above, the Owner shall so inform the Contractor in writing prior to commencement of the Work. The Contractor may then effect insurance that will protect the interest of the Contractor, Subcontractors and Sub-subcontractors in the Work, and by appropriate Change Order the cost thereof shall be charged to the Owner. If the Contractor is damaged by the failure or neglect of the Owner to purchase or maintain insurance as described above, without so notifying the Contractor in writing, then the Owner shall bear all reasonable costs properly attributable thereto.

§ 11.3.1.3 If the property insurance requires deductibles, shall pay costs not covered because of such deductibles.

§ 11.3.1.4 This property insurance shall cover portions of the Work stored off the site, and also portions of the Work in transit.

§ 11.3.1.5 Partial occupancy or use in accordance with Section 9.9 shall not commence until the insurance company or companies providing property insurance have consented to such partial occupancy or use by endorsement or otherwise. The Owner and the Convacion hall take reasonable steps to obtain consent of the insurance company or companies and shall, without mutual withen consent, take no action with respect to partial occupancy or use that would cause cancellation, lease or reduction of insurance.

§ 11.3.2 BOILER AND MACHINERY INSURANCE

The Owner shall purchase and maintain boiler and machinery insurance required by the Contract Documents or by law, which shall specifically cover such insured objects during installation and until final acceptance by the Owner; this insurance shall include interests of the Owner, Contractor, Subcontractors and Sub-subcontractors in the Work, and the Owner and Contractor shall be named insureds.

§ 11.3.3 LOSS OF USE INSURANCE

e Owner, at the Owner's option, may purchase and maintain such insurance as will insure the Owner against loss of e of the Owner's property due to fire or other hazards, however caused. The Owner waives all rights of action the me Contractor for loss of use of the Owner's property, including consequential losses due to fire or other ards however caused.

§ 11.3.4 If the Contractor requests in writing that insurance for risks other than those described herein or other special causes of loss be included in the property insurance policy, the Owner shall, if possible, include such insurance, and the cost thereof shall be charged to the Contractor by appropriate Change Order.

§ 11.3.5 If during the Project construction period the Owner insures properties, real or personal or both, at or adjacent to the site by property insurance under policies separate from those insuring the Project, or if after final payment

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved, WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 30 reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale User Notes:

Init.

property insurance is to be provided on the completed Project through a policy or policies other than those insuring the Project during the construction period, the Owner shall waive all rights in accordance with the terms of Section 11.3.7 for damages caused by fire or other causes of loss covered by this separate property insurance. All separate policies shall provide this waiver of subrogation by endorsement or otherwise.

§ 11.3.6 Before an exposure to loss may occur, the Owner shall file with the Contractor a copy of each policy that includes insurance coverages required by this Section 11.3. Each policy shall contain all generally applicable conditions, definitions, exclusions and endorsements related to this Project. Each policy shall contain a provision that the policy will not be canceled or allowed to expire, and that its limits will not be reduced, until at least 30 days written notice has been given to the Contractor.

§ 11.3.7 WAIVERS OF SUBROGATION

The Owner and Contractor waive all rights against (1) each other and any of their subcontractors, sub-subcontra ctors. agents and employees, each of the other, and (2) the Architect, Architect's consultants, separate contractors described in Article 6, if any, and any of their subcontractors, sub-subcontractors, agents and employees, for amages caused by fire or other causes of loss to the extent covered by property insurance obtained pursuant to his section 11.3 or other property insurance applicable to the Work, except such rights as they have to proceeds of such insurance held by the Owner as fiduciary. The Owner or Contractor, as appropriate, shall require of the Architect, Architect's consultants, separate contractors described in Article 6, if any, and the subcontractors, sub-subcontractors, agents and employees of any of them, by appropriate agreements, written where legally required for validity sincilar waivers each in favor of other parties enumerated herein. The policies shall provide such waivers of sublegation by endorsement or otherwise. A waiver of subrogation shall be effective as to a person or entity even though that no son or entity would otherwise have a duty of indemnification, contractual or otherwise, did not pay the in surface premium directly or indirectly, and whether or not the person or entity had an insurable interest in the property camaged.

§ 11.3.8 A loss insured under the Owner's property insurance shall be adjusted by the Owner as fiduciary and made payable to the Owner as fiduciary for the insureds, as their interests may appear, subject to requirements of any annlicable mortgagee clause and of Section 11.3.10. The contractor shall pay Subcontractors their just shares of applicable mortgagee clause and of Section 11.3.10. The contractor shall pay Subcontractors their just shares of insurance proceeds received by the Contractor, and by a propriate agreements, written where legally required for validity, shall require Subcontractors to make payment to heir Sub-subcontractors in similar manner.

§ 11.3.9 If required in writing by a party in interest, the owner as fiduciary shall, upon occurrence of an insured loss, give bond for proper performance of the Owner's arties. The cost of required bonds shall be charged against proceeds received as fiduciary. The Owner shall deposit in reseparate account proceeds so received, which the Owner shall distribute in accordance with such as the parties in interest may reach, or as determined in accordance with the method of binding dispute resolution spected in the Agreement between the Owner and Contractor. If after such loss no other special agreement is made and unless the Owner terminates the Contract for convenience, replacement of damaged property shall be parto med by the Contractor after notification of a Change in the Work in accordance with Article 7.

§ 11.3.10 The Owner as fiduciary shall have power to adjust and settle a loss with insurers unless one of the parties in interest shall object in writing within five days after occurrence of loss to the Owner's exercise of this power; if such objection is made interesting the dispute shall be resolved in the manner selected by the Owner and Contractor as the method of binding dispute resolution in the Agreement. If the Owner and Contractor have selected arbitration as the method of binding a spute resolution, the Owner as fiduciary shall make settlement with insurers or, in the case of a dispute over distribution of insurance proceeds, in accordance with the directions of the arbitrators.

11.4 PERFORMANCE BOND AND PAYMENT BOND

The Owner shall have the right to require the Contractor to furnish bonds covering faithful performance of the stract and payment of obligations arising thereunder as stipulated in bidding requirements or specifically required in the Contract Documents on the date of execution of the Contract.

§ 11.4.2 Upon the request of any person or entity appearing to be a potential beneficiary of bonds covering payment of obligations arising under the Contract, the Contractor shall promptly furnish a copy of the bonds or shall authorize a copy to be furnished.

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 31 reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

UNCOVERING AND CORRECTION OF WORK ARTICLE 12 § 12.1 UNCOVERING OF WORK

§ 12.1.1 If a portion of the Work is covered contrary to the Architect's request or to requirements specifically expressed in the Contract Documents, it must, if requested in writing by the Architect, be uncovered for the Architect's examination and be replaced at the Contractor's expense without change in the Contract Time.

§ 12.1.2 If a portion of the Work has been covered that the Architect has not specifically requested to examine prior to its being covered, the Architect may request to see such Work and it shall be uncovered by the Contractor. If such Work is in accordance with the Contract Documents, costs of uncovering and replacement shall, by appropriate, Change Order, be at the Owner's expense. If such Work is not in accordance with the Contract Documents, such contract Doc and the cost of correction shall be at the Contractor's expense unless the condition was caused by the Own 5 0 separate contractor in which event the Owner shall be responsible for payment of such costs.

§ 12.2 CORRECTION OF WORK

§ 12.2.1 BEFORE OR AFTER SUBSTANTIAL COMPLETION

The Contractor shall promptly correct Work rejected by the Architect or failing to conform to the requirements of the Contract Documents, whether discovered before or after Substantial Completion and whether not fabricated, installed or completed. Costs of correcting such rejected Work, including additional testing and inspections, the cost of uncovering and replacement, and compensation for the Architect's services and expenses made necessary thereby, shall be at the Contractor's expense.

§ 12.2.2 AFTER SUBSTANTIAL COMPLETION

§ 12.2.2.1 In addition to the Contractor's obligations under Section 3.5, if, within one year after the date of Substantial Completion of the Work or designated portion thereof or after the date for commencement of warranties established under Section 9.9.1, or by terms of an applicable special warranty, equired by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Contractor shall correct it promptly after receipt of written notice from the Owner to do so uncerthe Owner has previously given the Contractor a written acceptance of such condition. The Owner shall good use promptly after discovery of the condition. During the one-year period for correction of Work, if the Owner fails to notify the Contractor and give the Contractor an opportunity to make the correction, the Owner werves the signits to require correction by the Contractor and to make a claim for breach of warranty. If the Contractor fails to correct nonconforming Work within a reasonable time during that period after receipt of notice from the Owner of Trentect, the Owner may correct it in accordance with Section 2.4.

§ 12.2.2.2 The one-year period for correction of Work shall be extended with respect to portions of Work first performed after Substantial Completion by he period of time between Substantial Completion and the actual completion of that portion of the Work

§ 12.2.2.3 The one-year penad for correction of Work shall not be extended by corrective Work performed by the Contractor pursuant to uns Section 12.2.

§ 12.2.3 The Contractor short remove from the site portions of the Work that are not in accordance with the requirements on the Contract Documents and are neither corrected by the Contractor nor accepted by the Owner.

he Contractor shall bear the cost of correcting destroyed or damaged construction, whether completed or § 12.2.4 partially completed, of the Owner or separate contractors caused by the Contractor's correction or removal of Work it is not in accordance with the requirements of the Contract Documents. th

Nothing contained in this Section 12.2 shall be construed to establish a period of limitation with respect to r obligations the Contractor has under the Contract Documents. Establishment of the one-year period for correction of Work as described in Section 12.2.2 relates only to the specific obligation of the Contractor to correct the Work, and has no relationship to the time within which the obligation to comply with the Contract Documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Contractor's liability with respect to the Contractor's obligations other than specifically to correct the Work.

AIA Document A2011 - 2007. Copyright @ 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 32 Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the reproduction or distribution of this AIA® maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

§ 12.3 ACCEPTANCE OF NONCONFORMING WORK

If the Owner prefers to accept Work that is not in accordance with the requirements of the Contract Documents, the Owner may do so instead of requiring its removal and correction, in which case the Contract Sum will be reduced as appropriate and equitable. Such adjustment shall be effected whether or not final payment has been made.

MISCELLANEOUS PROVISIONS ARTICLE 13

§ 13.1 GOVERNING LAW

The Contract shall be governed by the law of the place where the Project is located except that, if the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section,

§ 13.2 SUCCESSORS AND ASSIGNS

§ 13.2.1 The Owner and Contractor respectively bind themselves, their partners, successors, assigns and egal representatives to covenants, agreements and obligations contained in the Contract Documents. Except as provid ed in Section 13.2.2, neither party to the Contract shall assign the Contract as a whole without written consent of me other. If either party attempts to make such an assignment without such consent, that party shall nevertheress replain legally responsible for all obligations under the Contract.

§ 13.2.2 The Owner may, without consent of the Contractor, assign the Contract to a leader providing construction financing for the Project, if the lender assumes the Owner's rights and obligations under the Contract Documents. The Contractor shall execute all consents reasonably required to facilitate such assignment

§ 13.3 WRITTEN NOTICE

Written notice shall be deemed to have been duly served if delivered in person to the individual, to a member of the firm or entity, or to an officer of the corporation for which it was intended; on if delivered at, or sent by registered or certified mail or by courier service providing proof of delivery to, ne last business address known to the party giving notice.

§ 13.4 RIGHTS AND REMEDIES

§ 13.4.1 Duties and obligations imposed by the Contract Documents and rights and remedies available thereunder connections, rights and remedies otherwise imposed or available by shall be in addition to and not a limitation of duties, law.

§ 13.4.2 No action or failure to act by the Owner, Architect or Contractor shall constitute a waiver of a right or duty afforded them under the Contract, nor shall such action or failure to act constitute approval of or acquiescence in a breach there under, except as may be specifically agreed in writing.

§ 13.5 TESTS AND INSPECTIONS

§ 13.5.1 Tests, inspections and a provals of portions of the Work shall be made as required by the Contract Documents and by applicable kws, statutes, ordinances, codes, rules and regulations or lawful orders of public authorities. Unless otherwise provided, the Contractor shall make arrangements for such tests, inspections and approvals with an independent testing laboratory or entity acceptable to the Owner, or with the appropriate public authority, and shall be call elated costs of tests, inspections and approvals. The Contractor shall give the Architect timely notice on when and where tests and inspections are to be made so that the Architect may be present for such procedures The Owner shall bear costs of (1) tests, inspections or approvals that do not become requirements until after biddare received or negotiations concluded, and (2) tests, inspections or approvals where building codes or applicable have or regulations prohibit the Owner from delegating their cost to the Contractor.

13.5.2 The Architect, Owner or public authorities having jurisdiction determine that portions of the Work require ical testing, inspection or approval not included under Section 13.5.1, the Architect will, upon written prization from the Owner, instruct the Contractor to make arrangements for such additional testing, inspection or approval by an entity acceptable to the Owner, and the Contractor shall give timely notice to the Architect of when and where tests and inspections are to be made so that the Architect may be present for such procedures. Such costs, except as provided in Section 13.5.3, shall be at the Owner's expense.

§ 13.5.3 If such procedures for testing, inspection or approval under Sections 13.5.1 and 13.5.2 reveal failure of the portions of the Work to comply with requirements established by the Contract Documents, all costs made necessary by

Init.

such failure including those of repeated procedures and compensation for the Architect's services and expenses shall be at the Contractor's expense.

§ 13.5.4 Required certificates of testing, inspection or approval shall, unless otherwise required by the Contract Documents, be secured by the Contractor and promptly delivered to the Architect.

§ 13.5.5 If the Architect is to observe tests, inspections or approvals required by the Contract Documents, the Architect will do so promptly and, where practicable, at the normal place of testing.

§ 13.5.6 Tests or inspections conducted pursuant to the Contract Documents shall be made promptly to avoid unreasonable delay in the Work.

§ 13.6 INTEREST

Payments due and unpaid under the Contract Documents shall bear interest from the date paymen is due at such rate as the parties may agree upon in writing or, in the absence thereof, at the legal rate prevailing from time t time at the place where the Project is located.

§ 13.7 TIME LIMITS ON CLAIMS

The Owner and Contractor shall commence all claims and causes of action, whether is comment, tort, breach of warranty or otherwise, against the other arising out of or related to the Contract in accordance with the requirements of the final dispute resolution method selected in the Agreement within the time period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion or the Werk The Owner and Contractor waive all claims and causes of action not commenced in accordance with his Section 13.7.

ARTICLE 14 TERMINATION OR SUSPENSION OF THE CONTRACT

§ 14.1 TERMINATION BY THE CONTRACTOR

§ 14.1.1 The Contractor may terminate the Contract if the Work is d for a period of 30 consecutive days through no act or fault of the Contractor or a Subcontractor, Subcontents of their agents or employees or any other persons or entities performing portions of the Work under direct or indirect contract with the Contractor, for any of the following reasons:

- Issuance of an order of a court or the public authority having jurisdiction that requires all Work to be .1 stopped;
- An act of government, such as a decuration of national emergency that requires all Work to be stopped; .2
- Because the Architect has of hour a Certificate for Payment and has not notified the Contractor of the .3 reason for withholding certification as provided in Section 9.4.1, or because the Owner has not made payment on a Certificate for Payment within the time stated in the Contract Documents; or
- .4 The Owner has failed to furnish to the Contractor promptly, upon the Contractor's request, reasonable evidence as required by Section 2.2.1.

§ 14.1.2 The Contractor may reminate the Contract if, through no act or fault of the Contractor or a Subcontractor, Sub-subcontractor of their agents or employees or any other persons or entities performing portions of the Work under direct or indirect contract with the Contractor, repeated suspensions, delays or interruptions of the entire Work by the Owner as describer in Section 14.3 constitute in the aggregate more than 100 percent of the total number of days scheduled for completion, or 120 days in any 365-day period, whichever is less.

§ 14.1.3 If the of the reasons described in Section 14.1.1 or 14.1.2 exists, the Contractor may, upon seven days' Attennotice to the Owner and Architect, terminate the Contract and recover from the Owner payment for Work secuted including reasonable overhead and profit, costs incurred by reason of such termination, and damages.

14.1.4 If the Work is stopped for a period of 60 consecutive days through no act or fault of the Contractor or a Subcontractor or their agents or employees or any other persons performing portions of the Work under contract with the Contractor because the Owner has repeatedly failed to fulfill the Owner's obligations under the Contract Documents with respect to matters important to the progress of the Work, the Contractor may, upon seven additional days' written notice to the Owner and the Architect, terminate the Contract and recover from the Owner as provided in Section 14.1.3.

1

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 34 reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. **User Notes:**

§ 14.2 TERMINATION BY THE OWNER FOR CAUSE

§ 14.2.1 The Owner may terminate the Contract if the Contractor

- repeatedly refuses or fails to supply enough properly skilled workers or proper materials; .1
- fails to make payment to Subcontractors for materials or labor in accordance with the respective .2 agreements between the Contractor and the Subcontractors;
- .3 repeatedly disregards applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of a public authority; or
- otherwise is guilty of substantial breach of a provision of the Contract Documents. .4

§ 14.2.2 When any of the above reasons exist, the Owner, upon certification by the Initial Decision Maker that sufficient cause exists to justify such action, may without prejudice to any other rights or remedies of the Quant after giving the Contractor and the Contractor's surety, if any, seven days' written notice, terminate employment of the Contractor and may, subject to any prior rights of the surety:

- Exclude the Contractor from the site and take possession of all materials, equipmen, tool .1 construction equipment and machinery thereon owned by the Contractor;
- Accept assignment of subcontracts pursuant to Section 5.4; and .2
- Finish the Work by whatever reasonable method the Owner may deem expeditor Upon written request .3 of the Contractor, the Owner shall furnish to the Contractor a detailed a counting of the costs incurred by the Owner in finishing the Work.

§ 14.2.3 When the Owner terminates the Contract for one of the reasons stated a Section n 14.2.1, the Contractor shall not be entitled to receive further payment until the Work is finished.

§ 14.2.4 If the unpaid balance of the Contract Sum exceeds costs of finishing the Work, including compensation for the Architect's services and expenses made necessary thereby, any other damages incurred by the Owner and not expressly waived, such excess shall be paid to the Contractor. If such costs and damages exceed the unpaid balance, the Contractor shall pay the difference to the Owner. The amount is be paid to the Contractor or Owner, as the case may be, shall be certified by the Initial Decision Maker, up contraction, and this obligation for payment shall survive termination of the Contract.

§ 14.3 SUSPENSION BY THE OWNER FOR CONVENIENCE

§ 14.3.1 The Owner may, without cause, order the Contractor in writing to suspend, delay or interrupt the Work in whole or in part for such period of time as the Owner may determine.

§ 14.3.2 The Contract Sum and Contract Time shall be adjusted for increases in the cost and time caused by suspension, delay or interruption as des much in Section 14.3.1. Adjustment of the Contract Sum shall include profit. No adjustment shall be made to the exent

- that performances, was or would have been so suspended, delayed or interrupted by another cause for .1 which the Contractor is responsible; or
- .2 that an equivable adjustment is made or denied under another provision of the Contract.

§ 14.4 TERMINATION BY THE OWNER FOR CONVENIENCE

§ 14.4.1 The Owner may, at any time, terminate the Contract for the Owner's convenience and without cause.

spon receipt of written notice from the Owner of such termination for the Owner's convenience, the Contractor hall

cease operations as directed by the Owner in the notice;

take actions necessary, or that the Owner may direct, for the protection and preservation of the Work; and

except for Work directed to be performed prior to the effective date of termination stated in the notice, terminate all existing subcontracts and purchase orders and enter into no further subcontracts and purchase orders.

§ 14.4.3 In case of such termination for the Owner's convenience, the Contractor shall be entitled to receive payment for Work executed, and costs incurred by reason of such termination, along with reasonable overhead and profit on the Work not executed.

Init.

1

.3

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved, WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties, Unauthorized 35 reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

ARTICLE 15 CLAIMS AND DISPUTES

§ 15.1 CLAIMS

§ 15.1.1 DEFINITION

A Claim is a demand or assertion by one of the parties seeking, as a matter of right, payment of money, or other relief with respect to the terms of the Contract. The term "Claim" also includes other disputes and matters in question between the Owner and Contractor arising out of or relating to the Contract. The responsibility to substantiate Claims shall rest with the party making the Claim.

§ 15.1.2 NOTICE OF CLAIMS

Claims by either the Owner or Contractor must be initiated by written notice to the other party and to the Initiated Decision Maker with a copy sent to the Architect, if the Architect is not serving as the Initial Decision Maker. by either party must be initiated within 21 days after occurrence of the event giving rise to such Claim of with days after the claimant first recognizes the condition giving rise to the Claim, whichever is later.

§ 15.1.3 CONTINUING CONTRACT PERFORMANCE

Pending final resolution of a Claim, except as otherwise agreed in writing or as provided in Section and Article 14, the Contractor shall proceed diligently with performance of the Contract and the Owner shall continue to make payments in accordance with the Contract Documents. The Architect will prepare Charge Drdes and issue Certificates for Payment in accordance with the decisions of the Initial Decision Make

§ 15.1.4 CLAIMS FOR ADDITIONAL COST

If the Contractor wishes to make a Claim for an increase in the Contract Sun, vri notice as provided herein shall be given before proceeding to execute the Work. Prior notice is not required for Claims relating to an emergency endangering life or property arising under Section 10.4.

§ 15.1.5 CLAIMS FOR ADDITIONAL TIME

e Contract Time, written notice as provided § 15.1.5.1 If the Contractor wishes to make a Claim for an increase nate of cost and of probable effect of delay on herein shall be given. The Contractor's Claim shall include and progress of the Work. In the case of a continuing delay, only one Claim is necessary.

§ 15.1.5.2 If adverse weather conditions are the basis for a Claim for additional time, such Claim shall be documented by data substantiating that weather conditions were abnormal for the period of time, could not have been reasonably anticipated and had an adverse effect on the scheduled construction.

§ 15.1.6 CLAIMS FOR CONSEQUENTIAL DAMAGES

The Contractor and Owner waive Chip's against each other for consequential damages arising out of or relating to this Contract. This mutual waiver includes

- damages increases by the Owner for rental expenses, for losses of use, income, profit, financing, .1 business and resonant, and for loss of management or employee productivity or of the services of such person, and
- damages incurred by the Contractor for principal office expenses including the compensation of .2 rsonial stationed there, for losses of financing, business and reputation, and for loss of profit except nti spated profit arising directly from the Work.

al waiver is applicable, without limitation, to all consequential damages due to either party's termination in This my. accordance with Article 14. Nothing contained in this Section 15.1.6 shall be deemed to preclude an award of liguidated damages, when applicable, in accordance with the requirements of the Contract Documents.

NITIAL DECISION

5.2.1 Claims, excluding those arising under Sections 10.3, 10.4, 11.3.9, and 11.3.10, shall be referred to the Initial Decision Maker for initial decision. The Architect will serve as the Initial Decision Maker, unless otherwise indicated in the Agreement. Except for those Claims excluded by this Section 15.2.1, an initial decision shall be required as a condition precedent to mediation of any Claim arising prior to the date final payment is due, unless 30 days have passed after the Claim has been referred to the Initial Decision Maker with no decision having been rendered. Unless the Initial Decision Maker and all affected parties agree, the Initial Decision Maker will not decide disputes between the Contractor and persons or entities other than the Owner.

Init.

1

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized 36 reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 19:31:24 on 09/27/2011 under Order No.2416304113_1 which expires on 03/09/2012, and is not for resale. User Notes:

§ 15.2.2 The Initial Decision Maker will review Claims and within ten days of the receipt of a Claim take one or more of the following actions: (1) request additional supporting data from the claimant or a response with supporting data from the other party, (2) reject the Claim in whole or in part, (3) approve the Claim, (4) suggest a compromise, or (5) advise the parties that the Initial Decision Maker is unable to resolve the Claim if the Initial Decision Maker lacks sufficient information to evaluate the merits of the Claim or if the Initial Decision Maker concludes that, in the Initial Decision Maker's sole discretion, it would be inappropriate for the Initial Decision Maker to resolve the Claim.

§ 15.2.3 In evaluating Claims, the Initial Decision Maker may, but shall not be obligated to, consult with or seek information from either party or from persons with special knowledge or expertise who may assist the Initial Devision Maker in rendering a decision. The Initial Decision Maker may request the Owner to authorize retention of su persons at the Owner's expense.

§ 15.2.4 If the Initial Decision Maker requests a party to provide a response to a Claim or to furnish additional supporting data, such party shall respond, within ten days after receipt of such request, and shall enter the provide a response on the requested supporting data, (2) advise the Initial Decision Maker when the response or supporting data will be furnished or (3) advise the Initial Decision Maker that no supporting data will be furnished. Upon receipt of the response or supporting data, if any, the Initial Decision Maker will either reject or approve the claim in whole or in part.

§ 15.2.5 The Initial Decision Maker will render an initial decision approving or rejecting the Claim, or indicating that the Initial Decision Maker is unable to resolve the Claim. This initial decision shall (1) be in writing; (2) state the reasons therefor; and (3) notify the parties and the Architect, if the Architect sphere reasons the Initial Decision Maker, of any change in the Contract Sum or Contract Time or both. The mitiga decision shall be final and binding on the parties but subject to mediation and, if the parties fail to resolve their displaye through mediation, to binding dispute resolution.

§ 15.2.6 Either party may file for mediation of an initial decision a any ame, subject to the terms of Section 15.2.6.1.

§ 15.2.6.1 Either party may, within 30 days from the date of an initial decision, demand in writing that the other party file for mediation within 60 days of the initial decision. If such a demand is made and the party receiving the demand fails to file for mediation within the time required then both parties waive their rights to mediate or pursue binding dispute resolution proceedings with respect to the initial decision.

§ 15.2.7 In the event of a Claim against the Contractor, the Owner may, but is not obligated to, notify the surety, if any, of the nature and amount of the Claim. If the Claim relates to a possibility of a Contractor's default, the Owner may, but is not obligated to, notify the streturant request the surety's assistance in resolving the controversy.

§ 15.2.8 If a Claim relates to or is the subject of a mechanic's lien, the party asserting such Claim may proceed in accordance with applicable lay temporary with the lien notice or filing deadlines.

§ 15.3 MEDIATION

§ 15.3.1 Claims, insputes, or other matters in controversy arising out of or related to the Contract except those waived as provided for a Sections 9.10.4, 9.10.5, and 15.1.6 shall be subject to mediation as a condition precedent to binding dispute resolution.

§ 15.3.2 The parties shall endeavor to resolve their Claims by mediation which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry ediatic Procedures in effect on the date of the Agreement. A request for mediation shall be made in writing, derivered to the other party to the Contract, and filed with the person or entity administering the mediation. The request we be made concurrently with the filing of binding dispute resolution proceedings but, in such event, mediation shall proceed in advance of binding dispute resolution proceedings, which shall be stayed pending mediation for a period of 60 days from the date of filing, unless stayed for a longer period by agreement of the parties or court order. If an arbitration is stayed pursuant to this Section 15.3.2, the parties may nonetheless proceed to the selection of the arbitrator(s) and agree upon a schedule for later proceedings.

Init.

1

§ 15.3.3 The parties shall share the mediator's fee and any filing fees equally. The mediation shall be held in the place where the Project is located, unless another location is mutually agreed upon. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction thereof.

§ 15.4 ARBITRATION

§ 15.4.1 If the parties have selected arbitration as the method for binding dispute resolution in the Agreement, any Claim subject to, but not resolved by, mediation shall be subject to arbitration which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Indust Arbitration Rules in effect on the date of the Agreement. A demand for arbitration shall be made in writing, delivered to the other party to the Contract, and filed with the person or entity administering the arbitration. The party and notice of demand for arbitration must assert in the demand all Claims then known to that party on which arbitra on 19 permitted to be demanded.

§ 15.4.1.1 A demand for arbitration shall be made no earlier than concurrently with the filing of a reques mediation, but in no event shall it be made after the date when the institution of legal or equitable proceedings based on the Claim would be barred by the applicable statute of limitations. For statute of limitations, surposes, eccept of a written demand for arbitration by the person or entity administering the arbitration shall constitute the institution of legal or equitable proceedings based on the Claim.

hay be entered upon it in § 15.4.2 The award rendered by the arbitrator or arbitrators shall be final, and judgin accordance with applicable law in any court having jurisdiction thereof.

§ 15.4.3 The foregoing agreement to arbitrate and other agreements to arbitrate with an additional person or entity duly consented to by parties to the Agreement shall be specifically enforceable under applicable law in any court having jurisdiction thereof.

§ 15.4.4 CONSOLIDATION OR JOINDER

§ 15.4.4.1 Either party, at its sole discretion, may consolid then a bitration conducted under this Agreement with other arbitration to which it is a party provided that (1) the arbitration agreement governing the other arbitration bitration conducted under this Agreement with any permits consolidation, (2) the arbitrations to be consolidated substantially involve common questions of law or fact, and (3) the arbitrations employ materially similar proceed ral rules and methods for selecting arbitrator(s).

§ 15.4.4.2 Either party, at its sole discretion may include by joinder persons or entities substantially involved in a common question of law or fact whose presence is equired if complete relief is to be accorded in arbitration, provided that the party sought to be joined consents in writing to such joinder. Consent to arbitration involving an additional person or entity shall not constitute conserved arbitration of any claim, dispute or other matter in question not described in the written consent.

or grant to any person or entity made a party to an arbitration conducted under this § 15.4.4.3 The Owner and Sor isingler or consolidation, the same rights of joinder and consolidation as the Owner and Section 15.4, whether Contractor under thi Agreement.

init.

1

SECTION 00 73 13

SUPPLEMENTARY GENERAL CONDITIONS A201-2007

| The following supplements modify the "General Conditions of the Contract for Construction," AIA Document A201-2007. Where a portion of the General Conditions is modified or deleted by the Supplementary Conditions, the unaltered portions of the General Conditions shall remain in effect. | |
|--|--|
| TABLE OF ARTICLES | |
| 1. | GENERAL PROVISIONS |
| 2. | OWNER |
| 3. | CONTRACTOR |
| 4. | ADMINISTRATION OF THE CONTRACT |
| 5. | SUBCONTRACTORS |
| 6. | CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS |
| 7. | CHANGES IN THE WORK |
| 8. | TIME |
| 9. | PAYMENTS AND COMPLETION |
| 10. | PROTECTION OF PERSONS AND PROPERTY |
| 11. | INSURANCE AND BONDS |
| 12. | UNCOVERING AND CORRECTION OF WORK |
| 16. | MISCELLANEOUS PROVISIONS |
| | TERMINATION OR SUSPENSION OF THE CONTRACT |

ARTICLE 1: GENERAL PROVISIONS

1.1 BASIC DEFINITIONS

1.1.1 THE CONTRACT DOCUMENTS

Delete the last sentence in its entirety and replace with the following:

"The Contract Documents also include Advertisement for Bid, Instructions to Bidde, sample forms, the Bid Form, the Contractor's completed Bid and the Award Letter."

Add the following Paragraph:

1.1.2 In the event of conflict or discrepancies among the Contract Documents, the Documents prepared by the State of Delaware, Division of Facilities Management shall take precedence over all other documents.

1.2 CORRELATION AND INTENT OF THE CONTRACT DOCUMENTS

Add the following Paragraphs:

- 1.2.4 In the case of an incensistency between the Drawings and the Specifications, or within either document not clarified by addendum, the better quality or greater quantity of work shall be provided in accordance with the Architect's interpretation.
- 1.2.5 The word "PLOYIDE" as used in the Contract Documents shall mean "FUR.USH LND INSTALL" and shall include, without limitation, all labor, meterials, equipment, transportation, services and other items required to complete the Work.
- 1.2.6 The word "PRODUCT" as used in the Contract Documents means all materials, systems and equipment.
 - VNERSHIP AND USE OF DRAWINGS, SPECIFICATIONS AND OTHER STRUMENTS OF SERVICE

elete Paragraph 1.5.1 in its entirety and replace with the following:

"All pre-design studies, drawings, specifications and other documents, including those in electronic form, prepared by the Architect under this Agreement are, and shall remain, the property of the Owner whether the Project for which they are made is executed or not. Such documents may be used by the Owner to construct one or more like Projects without the approval of, or additional compensation to, the Architect. The Contractor, Subcontractors, Sub-subcontractors and Material or Equipment Suppliers are authorized to use and reproduce applicable portions of the Drawings, Specifications and other documents prepared by the Architect and the Architect's consultants appropriate to and for use in the execution of their Work under the Contract Documents. They are not to be used by the Contractor or any Subcontractor, Sub-subcontractor or Material and Equipment Supplier on

1.5

other Projects or for additions to this Project outside the scope of the Work without the specific written consent of the Owner, Architect and Architect's consultants.

The Architect shall not be liable for injury or damage resulting from the re-use of drawings and specifications if the Architect is not involved in the re-use Project. Prior to re-use of construction documents for a Project in which the Architect is not also involved, the Ownerwill remove from such documents all identification of the original Architect, including name address and professional seal or stamp."

Delete Paragraph 1.5.2 in its entirety.

ARTICLE 2: OWNER

2.2 INFORMATION AND SERVICES REQUIRED OF THE OWNER

To Subparagraph 2.2.3 – Add the following sentence:

"The Contractor, at their expense shall bear the costs to occurately identify the location of all underground utilities in the area of their excavation and shall bear all cost for any repairs required, out of failure to accurately identify said utilities."

Delete Subparagraph 2.2.5 in its entirety and substitute the following:

2.2.5 The Contractor shall be turnshed free of charge up to three (3) sets of the Drawings and Project Wannals. Additional sets will be furnished at the cost of reproduction postage and handling.

ARTICLE 3: CONTRACTOR

3.2 REVIEW OF CONTRACT DOCUMENTS AND FIELD CONDITIONS BY CONTRACTOP

Amend Paragraph 52.2 to state that any errors, inconsistencies or omissions discovered shall be reported to the Architect and Owner immediately.

Delete the third sentence in Paragraph 3.2.3.

SUPERVISION AND CONSTRUCTION PROCEDURES

Add the following Paragraphs:

3.3.2.1 The Contractor shall immediately remove from the Work, whenever requested to do so by the Owner, any person who is considered by the Owner or Architect to be incompetent or disposed to be so disorderly, or who for any reason is not satisfactory to the Owner, and that person shall not again be employed on the Work without the consent of the Owner or the Architect.

3.3
- 3.3.4 The Contractor must provide suitable storage facilities at the Site for the proper protection and safe storage of their materials. Consult the Owner and the Architect before storing any materials.
- 3.3.5 When any room is used as a shop, storeroom, office, etc., by the Contractor or Subcontractor(s) during the construction of the Work, the Contractor making use of these areas will be held responsible for any repairs, patching or cleaning arising from such use.

3.4 LABOR AND MATERIALS

Add the Following Paragraphs:

3.4.4 Before starting the Work, each Contractor shan creefully examine all preparatory Work that has been executed to receive their Work. Check carefully, by whatever means are required to more that its Work and adjacent, related Work, will finish to proper contours, planes and levels. Promptly notify the General Contractor/Construction Manager of any defects or imperfections in preparator. We which will in any way affect satisfactory completion of its Work. Absence of such notification will be construed as an acceptance of preparatory Work and later claims of defects will not be recognized.

3.4.5

Under no circumstances shall the Contractor's Work proceed prior to preparatory Work having been completely cured, dried and/or otherwise made satisfactory to receive this Work. Responsibility for timely installation of all naterials rests solely with the Contractor responsible for that Work, who shall maintain coordination at all times.

3.5 WARRANTY

3.5.1

3.5.3

Add the following Paragraphs:

The Contractor will guarantee all materials and workmanship against original defects, except injury from proper and usual wear when used for the purpose intended, for two (2) years after Acceptance by the Owner, and will maintain all items in perfect condition during the period of guarantee.

Defects appearing during the period of guarantee will be made good by the Contractor at his expense upon demand of the Owner, it being required that all work will be in perfect condition when the period of guarantee will have elapsed.

In addition to the General Guarantee there are other guarantees required for certain items for different periods of time than the two (2) years as above, and are particularly so stated in that part of the specifications referring to same. The said guarantees will commence at the same time as the General Guarantee.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

3.5.4 If the Contractor fails to remedy any failure, defect or damage within a reasonable time after receipt of notice, the Owner will have the right to replace, repair, or otherwise remedy the failure, defect or damage at the Contractor's expense.

3.11 DOCUMENTS AND SAMPLES AT THE SITE

Add the following Paragraphs:

- 3.11.1 During the course of the Work, the Contractor shall maintain arecord scool drawings on which the Contractor shall mark the actual physical location of all piping, valves, equipment, conduit, outlets, access panels, controls, actuators, including all appurtenances that will be concealed once construction is complete, etc., including all invertelevations.
- 3.11.2 At the completion of the project, the Costractor shall obtain a set of reproducible drawings from the Archiect, and neatly transfer all information outlined in 3.11.1 to provide a complete record of the as-built conditions.
- 3.11.3 The Contractor shall provide two (2) wints of the as-built conditions, along with the reproducible drawings themselves, to the Owner and one (1) set to the Architect. In addition, attach one (1) complete set to each of the Operating and Maintenauce instructions/Manuals.
- 3.17 In the first sentence of the paragraph, insert "indemnify" between "shall" and "hold".

ARTICLE 4: ADMINISTRATION OF THE CONTRACT

4.2 ADMINISTRATION OF THE CONTRACT

Delete the first vertence of Paragraph 4.2.7 and replace with the following:

The Architect will review and approve or take other appropriate action upon the Contractor's submittae such as Shop Drawings, Product Data and Samples for the purpose of checking for conformance with the Contract Documents.

Pelete the second sentence of Paragraph 4.2.7 and replace with the following:



The Architect's action will be taken with such reasonable promptness as to cause no delay in the Work in the activities of the Owner, Contractor or separate Contractors, while allowing sufficient time in the Owner's professional judgment to permit adequate review.

Add the following Paragraph:

4.2.10.1 There will be no full-time project representative provided by the Owner or Architect on this project.

Add to Paragraph 4.2.13 "and in compliance with all local requirements." to the end of the sentence

ARTICLE 5: SUBCONTRACTORS

5.2 AWARD OF SUBCONTRACTS AND OTHER CONTRACTS FOR PORTIONS OF THE WORK

Delete Paragraph 5.2.3 in its entirety and replace with the following:

5.2.3 If the Owner or Architect has reasonable objection to a person or untity proposed by the Contractor, the Contractor shall propose ano her to whom the Owner or Architect has no reasonable objection, subject the statutory requirements of 29 Delaware Code § 6962(d)(10)b.3 and 4.

ARTICLE 6: CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTO

6.1 OWNER'S RIGHT TO PERFORM CONSTRUCTION AND 70 WARD SEPARATE CONTRACTS

Delete Paragraph 6.1.4 in its entirety.

- 6.2 MUTUAL RESPONSIBILITY
 - 6.2.3 In the second sentence, strike the word "shall" and insert the word "may".

ARTICLE 7: CHANGES IN THE WORK

(SEE ARTICLE 7: CHANCES IN WORK IN THE GENERAL REQUIREMENTS)

ARTICLE 8: TIME

8.2 PROGRESS AND COMPLETION

Add the f flowing Maragraphs:

Refer to Specification Section SUMMARY OF WORK for Contract time requirements.

If the Work falls behind the Progress Schedule as submitted by the Contractor, the Contractor shall employ additional labor and/or equipment necessary to bring the Work into compliance with the Progress Schedule at no additional cost to the Owner.

DELAYS AND EXTENSION OF TIME

8.3.1 Strike "arbitration" and insert "remedies at law or in equity".

Add the following Paragraph:

8.3.2.1 The Contractor shall update the status of the suspension, delay, or interruption of the Work with each Application for Payment. (The

SUPPLEMENTARY GENERAL CONDITIONS 00 73 13-6

Tetra Tech 200-15704-17001

Contractor shall report the termination of such cause immediately upon the termination thereof.) Failure to comply with this procedure shall constitute a waiver for any claim for adjustment of time or price based upon said cause. Delete Paragraph 8.3.3 in its entirety and replace with the following: 8.3.3 Except in the case of a suspension of the Work directed by the Oy extension of time under the provisions of Paragraph 8.3.1 shall Contractor's sole remedy in the progress of the Work and there she payment or compensation to the Contractor for any expense or d resulting from the delay. Add the following Paragraph: 8.3.4 By permitting the Contractor to work after the ex time for completion of the project, the Owner does not waive bein lights under the Contract. **ARTICLE 9: PAYMENTS AND COMPLETION** 9.2 SCHEDULE OF VALUES Add the following Paragraphs: 9.2.1 The Schedule of Values shall be submitted using AIA Document G702 and Continuation Shee 703. 9.2.2 Values is to include a line item for Project Closeout The Sche ent Submittal. The value of this item is to be no less than 1% of the Docu initi. 1 co t amount. ntra PAYMENT 9.3 APPLICATIO \mathbf{O} Add the following Paragraph: Application for Payment shall be submitted on AIA Document G702 "Application and Certificate for Payment", supported by AIA Document G703 "Continuation Sheet". Said Applications shall be fully executed and notarized. Add the following Paragraphs: 9.3.4 Until Closeout Documents have been received and outstanding items completed the Owner will pay 95% (ninety-five percent) of the amount due the Contractor on account of progress payments. 9.3.5 The Contractor shall provide a current and updated Progress Schedule to the Architect with each Application for Payment. Failure to provide Schedule will be just cause for rejection of Application for Payment.

9.5 DECISIONS TO WITHHOLD CERTIFICATION Add the following to 9.5.1:

- .8 failure to provide a current Progress Schedule;
- .9 a lien or attachment is filed;
- .10 failure to comply with mandatory requirements for maintaining Record Documents.

9.6 PROGRESS PAYMENTS

Delete Paragraph 9.6.1 in its entirety and replace with the following:

9.6.1 After the Architect has approved and issued a Certificate for Payment, payment shall be made by the Owner within 30 days after Owner's receipt of the Certificate for Payment.

9.7 FAILURE OF PAYMENT

In first sentence, strike "seven" and insert "thirty (3)". Also strike "binding dispute resolution" and insert "remedies at law or in equit.

9.8 SUBSTANTIAL COMPLETION

To Subparagraph 9.8.3- Add the following centerice:

"If the Architect is required to make more than 2 inspections of the same portion of work, the Contractor shall be responsible to all rosts associated with subsequent inspections including but not limited to any Architect's free."

9.8.5 In the econd sentence, strike "shall" and insert "may".

ARTICLE 10: PROTECTION OF PERSONS AND PROPERTY

10.1 SAFETY RECADIONS AND PROGRAMS

Id the following Paragraphs:

- Contractor shall develop a safety program in accordance with the Occupational Safety and Health Act of 1970. A copy of said plan shall be furnished to the Owner and Architect prior to the commencement of that Contractor's Work.
- Each Contractor shall appoint a Safety Representative. Safety Representatives shall be someone who is on site on a full time basis. If deemed necessary by the Owner or Architect, Contractor Safety meetings will be scheduled. The attendance of all Safety Representatives will be required. Minutes will be recorded of said meetings by the Contractor and will be distributed to all parties as well as posted in all job offices/trailers etc.

10.2 SAFETY OF PERSONS AND PROPERTY

10.1.2

Add the following Paragraph:

10.2.4.1As required in the Hazardous Chemical Act of June 1984, all vendors supplying any material that may be defined as hazardous must provide Material Safety Data Sheets for those products. Any chemical product should be considered hazardous if it has a caution warning on the label relating to a potential physical or health hazard, if it is known to be present in the work place, and if employees may be exposed under normal conditions or in fore-seable emergency situations. Material Safety Data Sheets shall be provided directly to the Owner, along with the shipping slips that include those products.

10.3 HAZARDOUS MATERIALS

Delete Paragraph 10.3.3 in its entirety.

- 10.5 Delete Paragraphs 10.3.6 in its entirety.
- ARTICLE 11: INSURANCE AND BONDS
- 11.1 CONTRACTOR'S LIABILITY INSURA CE
 - 11.1.4 Strike "the Owner" immediately following "(1)" and strike "and (2) the Owner as an additional instead for claims caused in whole or in part by the Contractor's negligent acts or omissions during the Contractor's completed operations."
- 11.2 OWNER'S LIABILITY INSTRANCE

Delete Paragraph T.2 h its entirety.

11.3 **PROPERTY INSUPANCE**

Pa agraph 11.3 in its entirety and replace with the following:

The State will not provide Builder's All Risk Insurance for the Project. The Contractor and all Subcontractors shall provide property coverage for their tools and equipment, as necessary. Any mandatory deductible required by the Contractor's Insurance shall be the responsibility of the Contractor.



PERFORMANCE BOND AND PAYMENT BOND

Add the following sentence: "The bonds will conform to those forms approved by the Office of Management and Budget."

ARTICLE 12: UNCOVERING AND CORRECTION OF WORK

12.2.2 AFTER SUBSTANTIAL COMPLETION Add the following Paragraph:

- 12.2.2.1.1 At any time during the progress of the Work, or in any case where the nature of the defects will be such that it is not expedient to have corrected, the Owner, at its option, will have the right to deduct such sum, or sums, of money from the amount of the Contract as it considers justified to adjust the difference in value between the defective work and that required under contract including any damage to the structure.
- 12.2.2.1 Strike "one" and insert "two".
- 12.2.2.2Strike "one" and insert "two".
- 12.2.2.3Strike "one" and insert "two".
- 12.2.5 In second sentence, strike "one" and insert "

ARTICLE 13: MISCELLANEOUS PROVISIONS

13.1 GOVERNING LAW

Strike "except that, if the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section 15.4."

13.6 INTEREST

Strike "the date payment is due at such rate as the parties may agree upon in writing or, in the absence thereof, at the legal rate prevailing from time to time at the place where the Project is located." Insert 30 days of presentment of the authorized Certificate of Payment at the annual rate of 12% or 1% per month.

13.7 TIME LIMITS ON CLAIMS

Strike the last sentence

Add the following Party

13.8

13.8.1

IFL CTS WITH FEDERAL STATUTES OR REGULATIONS

any provision, specifications or requirement of the Contract Documents conflict or is inconsistent with any statute, law or regulation of the government of the United State of America, the Contractor shall notify the Architect and Owner immediately upon discovery.

TICLE 14: TERMINATION OR SUSPENSION OF THE CONTRACT

TERMINATION BY THE OWNER FOR CONVENIENCE

Delete Paragraph 14.4.3 in its entirety and replace with the following:

14.4.3 In case of such termination for the Owner's convenience, the Contractor shall be entitled to receive payment for Work executed, and cost incurred by reason of such termination along with reasonable overhead.

ARTICLE 15: CLAIMS AND DISPUTES

- 15.1.2 Throughout the Paragraph strike "21" and insert "45".
- 15.1.6 CLAIMS FOR CONSEQUENTIAL DAMAGES

Delete Paragraph 15.1.6 in its entirety.

15.2 INITIAL DECISION

Delete Paragraph 15.2.5 in its entirety and replace with the follow

15.2.5 The Architect will approve or reject Claims by written decision, which shall state the reasons therefore and shall notify the parties of any change in the Contract Sum or Contract Time or both. The approval or rejection of a Claim by the Architect shall be subject to mediation and other remedies at law or in equity.

Delete Paragraph 15.2.6 and its subparagraphs in their entirety.

15.3 MEDIATION

- 15.3.1 Strike "binding dispute resolution" and insert "any or all remedies at law or in equity"
 - 15.3.2

ARBITR

FION

In the hist centence, delete "administered by the American Arbitration association in accordance with its Construction Industry Mediation Procedure in effect on the date of the Agreement," Strike "binding dispute resolution" and insert "remedies at law and in equity".

15.4

te Paragraph 15.4 and its sub-sections in its entirety.

END OF SECTION

PAGE INTENT LY NK

SECTION 00 73 14

ADDITIONAL SUPPLEMENTAL GENERAL CONDITIONS

PART 1 GENERAL

1.01 GENERAL CONDITIONS

- A. The General Conditions of the Contract for Construction, AIA Document A201 1997 edition, Articles 1 through 14 inclusive, is part of this contract and is bound herewith.
- B. Reference to Articles herein are to AIA Document A201.

1.02 SUPPLEMENTARY CONDITIONS

A. The following provisions modify, change, delete from or and to AIA Document A201. Where any article of the General Conditions is modified or any paragraph, subparagraph or clause thereof is modified or deleted by these provisions, the unaltered provisions of that article, paragraph, sub-paragraph or clause shall remain in effect.

1.03 REFERENCE TO DIVISION 1 - GENERAL REQUIRE MENTS

- A. Certain provisions of Division A GENERAL REQUIREMENTS supplement the administrative and work-related provisions of the GENERAL CONDITIONS.
- B. Articles affected are cross referenced in the various sections of Division 1.
- C. Throughout the entire document, wherever "Architect" is stated, read "Engineer".

1.04 ARTICLE 1 - CONTRACT DOCUMENTS

A. Paragraph 11 - Basic Definitions

d the jollowing subparagraph:

8 Terms and Definitions

The following definitions apply to the terms listed below as used on the Drawings and in the Project Manual.

- Approved: Accepted by the Engineer or authority enforcing standards.
- Described: Refer to Project Manual.
- Specified: Refer to Project Manual.

Shown: Refer to Drawings.

ADDITION AND RENOVATIONS

1.05 ARTICLE 3 - CONTRACTOR

A. Paragraph 3.4 - Labor and Materials

Subparagraph 3.4.1 - Add the following sentence:

Refer to Division 1 for detailed requirements concerning Temporary Facilities and Equipment.

Subparagraph 3.4.2 - Add the following sentence:

Refer specifically to Division 1 for detailed procedures regarding substitutions of material and/or equipment.

B. Paragraph 3.7 - Permits, Fees and Notices

Add the following:

- 3.7.5 Where local law at the site of the building requires a Certificate of Occupancy, the Contractor shall obtain and pay for this Certificate and deliver it to the Owner.
- 1.06 ARTICLE 5 SUBCONTRACTORS
 - A. Paragraph 5.2 Award of Subcontracts and Other Contracts for Portions of the Work.

Add the following subparagraph.

- 5.2.6 For Public Works Sontracts, the Contractor shall not subcontract, sublet, sell, or transfer work or materials to an organization other than their own without written permission from belotate. In case such permission is given, the Contractor will be permitted to subcontract or sublet a portion thereof but shall perform with their own organization, work amounting to not less than ten percent (10%) of the total contract bid price, exclusive of General Condition Items, Overhead, and Profit.
- 1.07 ARTICLE 2 TIME

A. Par grann 8.1.2



The Contractor shall begin work within ten (10) days following receipt of an Official Purchase Order issued by the State of Delaware.

- ARTICLE 11 INSURANCE
- A. Paragraph 11.5 Performance Bond and Payment Bond

Add the following subparagraphs:

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- 11.5.3 Prior to delivery of the executed contract, the Contractor must deliver to the Owner an executed performance bond in the amount of 100% of the accepted bid as security for the faithful performance of their contract and include the one year guarantee; and an executed labor and material payment bond in the amount of 100% of the accepted bid as security for the payment of all persons performing labor or furnishing materials in connection therewith and conditioned that the Contractor shall well and faithfully pay all daily labor employed by their firm for this contract in full once each week. Performance and payment bond may be in a combined form. The bonding company's standard forms are acceptable provided all coverage requirements are included. Consent of Surety must be included.
- 11.5.4 Bonds are to be in favor of the Owner and shall be paid for by the Contractor and furnished by a surety company licensed in the State of Delayara. The Owner has the right to demand proof that the parties signing the bonds are daly authorized to do so.

1.09 ARTICLE 15 - PREFERENCE FOR DELAWARE LABOR

A. 15.1 The Contractor shall comply with the following previous of **Delaware Code**, Title 29, Chapter 69, Section 6962:

In the construction of all public works for the State or any political subdivision or by persons contracting with the State or any political subdivision thereof, preference in employment of laborers, workme, or mechanics, shall be given to bona-fide legal citizens of the State who have established citizenship by residence of at least 90 days in the State. Each Public works contract for the construction of public works for the State or any political subdivision thereof shall contain a stipulation that any person, company, or corporation who violates this Section shall pay a penalty to the Secretary of Knanceequal to the amount of compensation paid to any person in violation of this section.

- 15.2 **Preference for Delaware Labor:** Surety bonds in the amount of 6% of the contract or subcontract price are required for all non-resident contractors for contracts within the state of Delaware in which either:
 - The single contract or subcontract totals \$20,000 or more; or
 - 2. The contract or sub-contract is a "cost-plus" contract whose estimated cost and profit totals \$20,000 or more; or
 - 3. The aggregate of two or more contracts or subcontracts in a calendar year totals \$20,000 or more.

The Division of Revenue will accept cash bonds, which may be paid by check on contracts not exceeding \$100,000.

The Contractor's bond shall be filed before construction commences on any contract upon which a bond is required, pursuant to Title 30 **Delaware Code** Chapter 375 as outlined above.



For licensing requirements, penalties, definitions, information filing and other requirements, bidders shall refer to Technical Information Memorandum 86-10 and Code Sections 375, 2501 and 2503. For additional information, contact 1-800-292-7826.

1.10 ARTICLE 16 - LICENSE, TAX AND STATE LAW REQUIREMENTS

- A. 16.1 In the performance of this Contract the successful bidder is required to comply with all applicable Federal, State and Local laws, ordinances, codes and regulations. The cost of permits, insurance, taxes and other relevant costs required in the performance of the Contract shall be borne by the successful bidder. All Delayare Lews in reference to construction shall be as binding as though quoted in full herein and their application shall be fully adhered to by all parties affected hereby. The Contractor shall furnish upon request any or all of the referenced items.
- B. 16.2 All bidders shall be properly licensed and authorized a transact business in the State of Delaware as provided for in the Delaware Code Citte 30, Chapter §2502 and in any and all Delaware Municipalities having jurisdiction to require such licensing within the geographic boundaries of the site (s) where the work is to be performed prior to the execution of any contract resulting from this bid. In the case of contracts in excess of \$50,000.00 which are competitively bid, such person shall have initiated the license application procedure required by this subsection with the Division of Revenue prior to, or in conjunction with the submission of a bid on a contract, or in the case of a subcontractor, prior to the submission of a bid by the general contractor. Proof of such dicense compliance shall be as determined by the Owner or their designee.
- C. 16.3 All bidders for public works projects shall insure that they have met all general contractor and/or subcontractor licensing requirements in accord with the requirement of Tule 29 Chapter 6962 prior to submitting a bid.
- 1.11 ARTICLE 19 INDEMNIFICATION BY THE CONTRACTOR AND SUBCONTRACTORS



To the fullest extent permitted by law, the Contractor shall indemnify and hold harmless the Owner, the Architect, the Consulting Engineers and their agents and employees from and against all claims, damages, losses and expenses, including but not limited to attorney's fees, arising out of or resulting from the performance of the work, provided that any such claim, damage, loss or expense (1) is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property (other than the work itself including the loss of use resulting therefrom, and (2) is caused in whole or in part by an negligent act or omission of the Contractor, any Subcontractor, anyone directly or indirectly employed by any of them or anyone for whose acts any of them may be liable, regardless of whether or not it is caused in part by a party indemnified hereunder. Such obligation shall not be construed to negate, abridge, or otherwise reduce any other right or obligation of indemnity which would otherwise exist as to any party or person described in the INDEMNIFICATION BY CONTRACTOR Article.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- 19.1.2 In any and all claims against the Owner, the Architect, and Consulting Engineers or any of their agents or employees by any employee of the Contractor, any Subcontractor, anyone directly or indirectly employed by any of them or anyone for whose acts any of them may be liable, the indemnification obligation shall not be limited in any way by any limitation on the amount or type of damages, compensation or benefits payable by or for the Contractor or Subcontractor under workers' or workmen's compensation acts, disability benefits acts or other employee benefit acts.
- 19.1.3.4The obligations of the Contractor shall not extend to the liability of the Architect, the Consulting Engineers or their agents or employees, arising out of (1) the preparation or approval of maps, drawings, opinions, reports, surveys, change orders, design or specifications or (2) the giving of contre failure to give directions or instructions by the Architect, the Consulting Engineers or their agents or employees provided such giving or failure agree is the primary cause of the injury or damage.

19.2 **INDEMNIFICATION BY THE SUBCONTRACTORS**

- 19.2.1 Contractor hereby covenants and agrees that it will include the following provisions in all subcontracts entered into by the contractor for performance of any and all portions of the work on the project:
 - (1) To the fullest extent permitted by law, the Subcontractor shall indemnify and hord harmless the Owner, the Architect, the Consulting Engineers and their agents and employees from and against all claims, damages, losses and expenses, including but not ited to attorneys' fees, arising out of or resulting from the performance of the work, provided that any such claim, damage, ss or expense (1) is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property (other than the work itself) including the loss of use resulting therefrom, and (2) is caused in whole or in part by any negligent act or omission of the Subcontractor may be liable, regardless of whether or not it is caused in part by a party indemnified hereunder. Such obligation shall not be construed to negate, abridge, or otherwise reduce any other right or obligation or indemnity which would otherwise exist as to any party or person described in the INDEMNIFICATION BY THE SUBCONTRACTORS Article.
 - a. In any and all claims against the Owner, the Architect, the Consulting Engineers or any of their agents or employees by any employee of the Subcontractor, anyone directly or indirectly employed by the Subcontractor or anyone for whose acts the Subcontractor may be liable, the indemnification obligation under this Article shall not be limited in any way by any limitation on the amount or type of damages, compensation or benefits payable by or for the Contractor or the Subcontractor under workers' or workmen's compensation acts, disability benefits acts or other employee benefit acts.

b. The obligation of the Subcontractor under this Article shall not extend to the liability of the Architect, the Consulting Engineers, their agents or employees, arising out of (1) the preparation of approval of maps, drawings, opinions, reports, surveys, change orders, design or specifications, or (2) the giving of or the failure of give direction of instructions by the Architect, the Consulting Engineers, their agents or employees provided such giving or failure to give is the primary cause of the injury or damage.

END OF SECTION

Republic

SECTION 00 81 13 GENERAL REQUIREMENTS

TABLE OF ARTICLES

- 1. GENERAL PROVISIONS
- 2. OWNER
- 3. CONTRACTOR
- 4. ADMINISTRATION OF THE CONTRACT
- 5. SUBCONTRACTORS
- CONSTRUCTION BY OWNER OR BY SEPARATE CONTROL
 CHANCES DUTIE WORK
- 7. CHANGES IN THE WORK
- 8. TIME
- 9. PAYMENTS AND COMPLETION
- 10. PROTECTION OF PERSONS AND PROPERTY
- 11. INSURANCE AND BONDS
- 12. UNCOVERING AND CORRECTION OF WORK
- 13. MISCELLANEOUS PROVISIONS
- 14. TERMINATION OF SUSPENSION OF THE CONTRACT

•

ARTICLE 1: GENERAL

- 1.1 CONTRACT DOCUMENTS
- 1.1.1 The intent of the Contract Documents is to include all items necessary for the proper execution and completion of the Work by the Contractor. The Contract Documents are complementary and what is required by one shall be as binding as if required by all Performance by the Contractor shall be required to an extent consistent with the Contractor Documents and reasonably inferable from them as being necessary to produce the intended results.
- 1.1.2 Work including material purchases shall not begin until the Contractor is in receipt of a bonafide State of Delaware Purchase Order. Any work performed or traterial purchases prior to the issuance of the Purchase Order is done at the Contractor's own risk and cost.
- 1.2 EQUALITY OF EMPLOYMENT OPPORTUNITY ON PUBLIC WORKS
- 1.2.1 For Public Works Projects financed in whole or in part by state appropriation the Contractor agrees that during the performance of this contract:
 - 1. The Contractor will not discriminate against any employee or applicant for employment because of race, creed, sex color, sexual orientation, gender identity or national origin. The Contractor will take positive steps to ensure that applicants are employed and that employee are treated during employment without regard to their race, creed, color, sex or national origin. Such action shall include, but not be limited to, the following: employment, upgrading, demotion or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. The Contractor agrees to post in conspicuous places available to employees and applicants for employment notices to be provided by the contracting agency setting form the nondiscrimination clause.
 - 2. The Contractor will, in all solicitations or advertisements for employees placed by or on behalf of the Contractor, state that all qualified applicants will receive consideration for employment without regard to race, creed, color, sex or national rigin."

ARTICLE 2. OWNER



3.2

(NO ADDITIONAL GENERAL REQUIREMENTS – SEE SUPPLEMENTARY GENERAL CONDITIONS)

CONTRACTOR

Schedule of Values: The successful Bidder shall within twenty (20) days after receiving notice to proceed with the work, furnish to the Owner a complete schedule of values on the various items comprising the work.

Subcontracts: Upon approval of Subcontractors, the Contractor shall award their Subcontracts as soon as possible after the signing of their own contract and see that all

GENERAL REQUIREMENT 00 81 13- 2 Tetra Tech 200-15704-17001

material, their own and those of their Subcontractors, are promptly ordered so that the work will not be delayed by failure of materials to arrive on time.

- 3.3 Before commencing any work or construction, the General Contractor is to consult with the Owner as to matters in connection with access to the site and the allocation of Ground Areas for the various features of hauling, storage, etc.
- 3.4 The Contractor shall supervise and direct the Work, using the Contractor's best shall and attention. The Contractor shall be solely responsible for and have control over construction means, methods, techniques, sequences and procedures and for coordinating at periods of the Work under the Contract, unless the Contract Documents give other specific instructions.
- 3.5 The Contractor shall enforce strict discipline and good order among the Contractor's employees and other persons carrying out the Contract. The Contractor shall not permit employment of unfit persons or persons not skilled in tasks assigned to bem.
- 3.6 The Contractor warrants to the Owner that materials and equipment furnished will be new and of good quality, unless otherwise permitted, and that the work will be free from defects and in conformance with the Contract Documents. Mark not conforming to these requirements, including substitutions not properly approved, may be considered defective. If required by the Owner, the Contractor shall furnish evidence as to the kind and quality of materials and equipment provided.
- 3.7 Unless otherwise provided, the Contractor shall pay all sales, consumer, use and other similar taxes, and shall secure and pay forceq fired permits, fees, licenses, and inspections necessary for proper execution of the Work.
- 3.8 The Contractor shall comply with and give notices required by laws, ordinances, rules, regulations, and lawful orders of public authorities bearing on performance of the Work. The Contractor shall promptly active the Owner if the Drawings and Specifications are observed to be at variance meterna.
- 3.9 The Contractor shar be responsible to the Owner for the acts and omissions of the Contractor's employees, Subcontractors and their agents and employees, and other persons performing portions of the Work under contract with the Contractor.
- 3.10



3.11.1

The Contractor shall keep the premises and surrounding area free from accumulation of vaste materials or rubbish caused by operations under the Contract. At completion of the Vork the Contractor shall remove from and about the Project all waste materials, rubbish, the Contractor's tools, construction equipment, machinery and surplus materials. The Contractor shall be responsible for returning all damaged areas to their original conditions.

STATE LICENSE AND TAX REQUIREMENTS

Each Contractor and Subcontractor shall be licensed to do business in the State of Delaware and shall pay all fees and taxes due under State laws. In conformance with Section 2503, Chapter 25, Title 30, <u>Delaware Code</u>, "the Contractor shall furnish the Delaware Department of Finance within ten (10) days after entering into any contract with a contractor or subcontractor not a resident of this State, a statement of total value of

Tetra Tech 200-15704-17001 GENERAL REQUIREMENT 00 81 13-3 such contract or contracts together with the names and addresses of the contracting parties."

- 3.12. The Contractor shall comply with all requirements set forth in Section 6962, Chapter 69, Title 29 of the <u>Delaware Code</u>.
- 3.13 During the contract Work, the Contractor and each listed Subcontractor, shall implement an Employee Drug Testing Program in accordance with OMB Regulation 4104-"Regulations for the Drug Testing of Contractor and Subcontractor Employees Working on "Large Public Works Projects". "Large Public Works" is based upon the current threshold required for bidding Public Works as set by the Purchasing and Contracting Advisory Council.

ARTICLE 4: ADMINISTRATION OF THE CONTRACT

- 4.1 CONTRACT SURETY
- 4.1.1 PERFORMANCE BOND AND LABOR AND MATERIAL PAYMENT BOND
- 4.1.2 All bonds will be required as follows unless specifically waved elsewhere in the Bidding Documents.
- 4.1.3 Contents of Performance Bonds The bond shall be in the form approved by the Office of Management and Budget. The bond shall be conditioned upon the faithful compliance and performance by the successful hidder of each and every term and condition of the contract and the proposal, plans, peeifications, and bid documents thereof. Each term and condition shall be met at me time and in the manner prescribed by the Contract, Bid documents and the specification, including the payment in full to every person furnishing materiel or performance tabor in the performance of the Contract, of all sums of money due the person for s ch labor and materiel. (The bond shall also contain the successful bidder's guarantee to indemnify and save harmless the State and the agency from all costs, damages and expenses growing out of or by reason of the Contract in accordance with the Contract.)
- 4.1.4 Invoking a region of the other strength of the other strengt

4.1.5



Within wenty (20) days after the date of notice of award of contract, the Bidder to whom the award is made shall furnish a Performance Bond and Labor and Material Payment Bond, ach equal to the full amount of the Contract price to guarantee the faithful performance of all terms, covenants and conditions of the same. The bonds are to be issued by an acceptable Bonding Company licensed to do business in the State of Delaware and shall be issued in <u>duplicate</u>.

Performance and Payment Bonds shall be maintained in full force (warranty bond) for a period of two (2) years after the date of the Certificate for Final Payment. The Performance Bond shall guarantee the satisfactory completion of the Project and that the Contractor will make good any faults or defects in his work which may develop during the period of said guarantees as a result of improper or defective workmanship, material or apparatus, whether furnished by themselves or their Sub-Contractors. The Payment Bond shall guarantee that the Contractor shall pay in full all persons, firms or corporations who furnish labor or

material or both labor and material for, or on account of, the work included herein. The bonds shall be paid for by this Contractor. The Owner shall have the right to demand that the proof parties signing the bonds are duly authorized to do so.

4.2 FAILURE TO COMPLY WITH CONTRACT

4.2.1 If any firm entering into a contract with the State, or Agency that neglects or refuses to perform or fails to comply with the terms thereof, the Agency which signed the Contract may terminate the Contract and proceed to award a new contract in accordance with the Chapter 69, Title 29 of the Delaware Code or may require the Surety on the Performance Bond to complete the Contract in accordance with the terms of the Performance Bond. Nothing herein shall preclude the Agency from pursing additional remedies at otherwise provided by law.

4.3 CONTRACT INSURANCE AND CONTRACT LIABILITY

- 4.3.1 In addition to the bond requirements stated in the Bid Doctments, each successful Bidder shall purchase adequate insurance for the performance of the Contract and, by submission of a Bid, agrees to indemnify and save harmless and to aefend all legal or equitable actions brought against the State, any Agency, officer and/or employee of the State, for and from all claims of liability which is or may be be result of the successful Bidder's actions during the performance of the Contract.
- 4.3.2 The purchase or nonpurchase of such insurance or the involvement of the successful Bidder in any legal or equitable defense of any action brought against the successful Bidder based upon work performed pursuant to the Contract will not waive any defense which the State, its agencies and their respective officers, employees and agents might otherwise have against such claims, specifically including the defense of sovereign immunity, where applicable, and by the terms of this section, the State and all agencies, officers and employees her of shall not be financially responsible for the consequences of work performed nursuant to said contract.
- 4.4 RIGHT TO AUDIARECORDS
- 4.4.1 The Owner shall have the right to audit the books and records of a Contractor or any Subcontractor under any Contract or Subcontract to the extent that the books and records relate to the performance of the Contract or Subcontract.
- 4.4.2 ARTICLE 5:

5.1.1

aid books and records shall be maintained by the Contractor for a period of seven (7) years from the date of final payment under the Prime Contract and by the Subcontractor for a period of seven (7) years from the date of final payment under the Subcontract.

: SUBCONTRACTORS

SUBCONTRACTING REQUIREMENTS

All contracts for the construction, reconstruction, alteration or repair of any public building (not a road, street or highway) shall be subject to the following provisions:

1. A contract shall be awarded only to a Bidder whose Bid is accompanied by a statement containing, for each Subcontractor category, the name and address

Tetra Tech 200-15704-17001

GENERAL REQUIREMENT 00 81 13-5 (city or town and State only – street number and P.O. Box addresses not required) of the subcontractor whose services the Bidder intends to use in performing the Work and providing the material for such Subcontractor category.

- 2. A Bid will not be accepted nor will an award of any Contract be made to any Bidder which, as the Prime Contractor, has listed itself as the Subcontractor for any Subcontractor unless:
 - A. It has been established to the satisfaction of the awarding Agency but the Bidder has customarily performed the specialty work of such Subcontractor category by artisans regularly employed by the Bidder's firm;
 - B. That the Bidder is duly licensed by the State to engage in such specialty work, if the State requires licenses; and
 - C. That the Bidder is recognized in the adustry as a bona fide Subcontractor or Contractor in such specialty work and Subcontractor category.
- 5.1.2 The decision of the awarding Agency as to whether a Bidder who list itself as the Subcontractor for a Subcontractor category shell be final and binding upon all Bidders, and no action of any nature shall lie against any awarding agency or its employees or officers because of its decision in this regard.
- 5.1.3 After such a Contract has been awa decythe successful Bidder shall not substitute another Subcontractor for any Subcontractor whose name was set forth in the statement which accompanied the Bid without he written consent of the awarding Agency.
- 5.1.4 No Agency shall consect to any substitution of Subcontractors unless the Agency is satisfied that the subcontractor whose name is on the Bidders accompanying statement:
 - A. Is unqualitized to perform the work required;

As failed to execute a timely reasonable Subcontract;

Las defaulted in the performance on the portion of the work covered by the Subcontract; or



Is no longer engaged in such business.

Should a Bidder be awarded a contract, such successful Bidder shall provide to the agency the taxpayer identification license numbers of such subcontractors. Such numbers shall be provided on the later of the date on which such subcontractor is required to be identified or the time the contract is executed. The successful Bidder shall provide to the agency to which it is contracting, within 30 days of entering into such public works contract, copies of all Delaware Business licenses of subcontractors and/or independent contractors that will perform work for such public works contract. However, if a subcontractor or independent contractor is hired or contracted more than 20 days after the Bidder entered the public works

B.

contract the Delaware Business license of such subcontractor or independent contractor shall be provided to the agency within 10 days of being contracted or hired.

5.2 PENALTY FOR SUBSTITUTION OF SUBCONTRACTORS

5.2.1 Should the Contractor fail to utilize any or all of the Subcontractors in the Contract statement in the performance of the Work on the public bidding, the Ca actor s all be penalized in the amount of (project specific amount*). The Agency may etermine to deduct payments of the penalty from the Contractor or have the amount paid ly to the Agency. Any penalty amount assessed against the Contractor may be remitted of refunded, in whole or in part, by the Agency awarding the Contract, only if it is estimated to the satisfaction of the Agency that the Subcontractor in question has defaulted on is Longer engaged in such business. No claim for the remission or refund of any penalty shall be granted unless an application is filed within one year after the liability of the successful Bidder accrues. All penalty amounts assessed and not refunded or remined contractor shall be reverted to the State.

*one (1) percent of contract amount not to exceed 10,000

- 5.3 ASBESTOS ABATEMENT
- 5.3.1 The selection of any Contractor to perform asbestos abatement for State-funded projects shall be approved by the Office of Management and Budget, Division of Facilities Management pursuant to Chapter 78 of Title 56.
- 5.4 STANDARDS OF CONSTRUCTION FOR THE PROTECTION OF THE PHYSICALLY AAND SAPPED
- 5.4.1 All Contracts shall conform with the standard established by the Delaware Architectural Accessibility Board unless otherwise exempted by the Board.
 - CONTRACT PERFORMANCE

Any firm entering into a Public Works Contract that neglects or refuses to perform or fails to comply with its terms, the Agency may terminate the Contract and proceed to award a new Contract or may require the Surety on the Performance Bond to complete the Contract in accordance with the terms of the Performance Bond.

E 6: CONSTRUCTION BY OWNER OR SEPARATE CONTRACTORS

The Owner reserves the right to simultaneously perform other construction or operations related to the Project with the Owner's own forces, and to award separate contracts in connection with other portions of the Project or other Projects at the same site.

6.2 The Contractor shall afford the Owner and other Contractors reasonable opportunity for access and storage of materials and equipment, and for the performance of their activities,

Tetra Tech 200-15704-17001

5.5

5.5.1

and shall connect and coordinate their activities with other forces as required by the Contract Documents.

ARTICLE 7: CHANGES IN THE WORK

- 7.1 The Owner, without invalidating the Contract, may order changes in the Work consisting of Additions, Deletions, Modifications or Substitutions, with the Contract Sum and Contract completion date being adjusted accordingly. Such changes in the Work shall be authorized by written Change Order signed by the Professional, as the duly authorized agent, the Contractor and the Owner.
- 7.2 The Contract Sum and Contract Completion Date shall be adjusted only by a funy executed Change Order.
- 7.3 The additional cost, or credit to the Owner resulting from a change on the Work shall be by mutual agreement of the Owner, Contractor and the Architect. It all coses, this cost or credit shall be based on the 'DPE' wages required and the cinvolce price" of the materials/equipment needed.
- 7.3.1 "DPE" shall be defined to mean "direct personnel expense. Direct payroll expense includes direct salary plus customary fringe benefits (prevailing wage rates) and documented statutory costs such as workman's compensation insurance, Social Security/Medicare, and unemployment insurance (a maximum multiplier of 1.35 times DPE).
- 7.3.2 "Invoice price" of materials/equipment shall be defined to mean the actual cost of materials and/or equipment that is paid by the Contractor, (or subcontractor), to a material distributor, direct factory vendor, store, material provider, or equipment leasing entity. Rates for equipment that is leased and/or owned by the Contractor or subcontractor(s) shall not exceed those listed in the latest version of the "Means Building Construction Cost Data" publication.
- In addition to the wow the General Contractor is allowed a fifteen percent (15%) 7.3.3 markup for overhead and profit for additional work performed by the General orces. For additional subcontractor work, the Subcontractor is allowed Contractor's ow a fifteen (15) per overhead and profit on change order work above and beyond the direct costs stated previously. To this amount, the General Contractor will be allowed a ot exceeding seven and one half percent (7.5%) on the subcontractors work. mark-up These mark-ups shall include all costs including, but not limited to: overhead, profit, surance, supervision, etc. No markup is permitted on the work of the bo, ds, j abcontractors subcontractor. No additional costs shall be allowed for changes related to the Contractor's onsite superintendent/staff, or project manager, unless a change in the work changes the project duration and is identified by the CPM schedule. There will be no other costs associated with the change order.

ARTICLE 8:

TIME

Time limits, if any, are as stated in the Project Manual. By executing the Agreement, the Contractor confirms that the stipulated limits are reasonable, and that the Work will be completed within the anticipated time frame.

8.2

If progress of the Work is delayed at any time by changes ordered by the Owner, by labor disputes, fire, unusual delay in deliveries, abnormal adverse weather conditions, unavoidable

casualties or other causes beyond the Contractor's control, the Contract Time shall be extended for such reasonable time as the Owner may determine.

8.3 Any extension of time beyond the date fixed for completion of the construction and acceptance of any part of the Work called for by the Contract, or the occupancy of the building by the Owner, in whole or in part, previous to the completion shall not be deemed a waiver by the Owner of his right to annul or terminate the Contract for abandonment or delay in the matter provided for, nor relieve the Contractor of full responsibility.

8.4 SUSPENSION AND DEBARMENT

AD

AGE

- 8.4.1 Per Section 6962(d)(14), Title 29, Delaware Code, "Any Contractor who fails to perform a public works contract or complete a public works project within the time schedule established by the Agency in the Invitation To Bid, may be subject to Suspension or Debarment for one or more of the following reasons: a) failure couply the adequate labor supply ratio for the project; b) inadequate financial resources; or, γ_{T} for performance on the Project."
- 8.4.2 "Upon such failure for any of the above stated reasons gency that contracted for the th public works project may petition the Director of Monagement and Budget for Suspension or Debarment of the Contractor. The Agency shall send a copy of the petition to the Contractor within three (3) working days of filing with the Director. If the Director concludes that the petition has verif, the Director shall schedule and hold a hearing to determine whether to suspend be Contractor, debar the Contractor or deny the petition. The Agency shall have be burden of proving, by a preponderance of the evidence, that the Contractor failed to perform or complete the public works project within the time schedule est blishel by the Agency and failed to do so for one or more of the following reasons: failure to supply the adequate labor supply ratio for the project; b) inadequate financia resources; or, c) poor performance on the project. Upon a finding in favor of the Agercy, the Director may suspend a Contractor from Bidding on any project funded in whom or in part, with public funds for up to 1 year for a first offense, up to 3 years in a second offense and permanently debar the Contractor for a third he Director shall issue a written decision and shall send a copy to the offense. Contractor 2 ad the Agency. Such decision may be appealed to the Superior Court within 30 days for a review on the record."

8.5



Per Section 6962(d)(5) a.3, Title 29, Delaware Code: The Agency may at the beginning of each public works project establish a time schedule for the completion of the project. If the project is delayed beyond the completion date due to the Contractor's failure to meet their responsibilities, the Agency may forfeit, at its discretion, all or part of the Contractor's retainage.

This forfeiture of retainage also applies to the timely completion of the punchlist. A punchlist will only be prepared upon the mutual agreement of the Owner, Architect and Contractor. Once the punchlist is prepared, all three parties will by mutual agreement, establish a schedule for its completion. Should completion of the punchlist be delayed beyond the established date due to the Contractor's failure to meet their responsibilities, the Agency may hold permanently, at its discretion, all or part of the Contractor's retainage.

Tetra Tech 200-15704-17001

GENERAL REQUIREMENT 00 81 13-9

ARTICLE 9: PAYMENTS AND COMPLETION

9.1 APPLICATION FOR PAYMENT

- 9.1.1 Applications for payment shall be made upon AIA Document G702. There will be a first percent (5%) retainage on all Contractor's monthly invoices until completion of the project. This retainage may become payable upon receipt of all required closeout documentation, provided all other requirements of the Contract Documents have been met.
- 9.1.2 A date will be fixed for the taking of the monthly account of work done. Upon recipt of Contractor's itemized application for payment, such application will be ardited, modified, if found necessary, and approved for the amount. Statement shall be submitted to the Owner.
- 9.1.3 Section 6516, Title 29 of the <u>Delaware Code</u> annualized interactis but to exceed 12% per annum beginning thirty (30) days after the "presentment" (as opposed to the date) of the invoice.
- 9.2 PARTIAL PAYMENTS
- 9.2.1 Any public works Contract executed by any Agency may provide for partial payments at the option of the Owner with respect to materials placed along or upon the sites or stored at secured locations, which are suitable focuse in the performance of the contract.
- 9.2.2 When approved by the agency, called payment may include the values of tested and acceptable materials of a nonperisbable or noncontaminative nature which have been produced or furnished for incorporation as a permanent part of the work yet to be completed, provided acceptable provisions have been made for storage.
- 9.2.2.1 Any allowance made for materials on hand will not exceed the delivered cost of the materials as verified by invoices furnished by the Contractor, nor will it exceed the contract bid processor the material complete in place.
- 9.2.3 If requested by the Agency, receipted bills from all Contractors, Subcontractors, and material, men, etc. for the previous payment must accompany each application for payment. Following such a request, no payment will be made until these receipted bills have been received by the Owner.

SUBSTANTIAL COMPLETION

9.3 9.3.1

When the building has been made suitable for occupancy, but still requires small items of miscellaneous work, the Owner will determine the date when the project has been substantially completed.

If, after the Work has been substantially completed, full completion thereof is materially delayed through no fault of the Contractor, and without terminating the Contract, the Owner may make payment of the balance due for the portion of the Work fully completed and accepted. Such payment shall be made under the terms and conditions governing final payment that it shall not constitute a waiver of claims.

- 9.3.3 On projects where commissioning is included, the commissioning work as defined in the specifications must be complete prior to the issuance of substantial completion.
- 9.4 FINAL PAYMENT
- 9.4.1 Final payment, including the five percent (5%) retainage if determined appropriate, shall be made within thirty (30) days after the Work is fully completed and the Contract fung performed and provided that the Contractor has submitted the following accepted documentation (in addition to any other documentation required elsewhere in the Contract Documents):
- 9.4.1.1 Evidence satisfactory to the Owner that all payrolls, material bills, and other indebtedness connected with the work have been paid,
- 9.4.1.2 An acceptable RELEASE OF LIENS,
- 9.4.1.3 Copies of all applicable warranties,
- 9.4.1.4 As-built drawings,
- 9.4.1.5 Operations and Maintenance Manuals,
- 9.4.1.6 Instruction Manuals,
- 9.4.1.7 Consent of Surety to final payment
- 9.4.1.8 The Owner reserves the right to retain payments, or parts thereof, for its protection until the foregoing conditions have been complied with, defective work corrected and all unsatisfactory conditions remelied.

ARTICLE 10: PROTECTION OF PERSONS AND PROPERTY

10.1



The Centrator shall be responsible for initiating, maintaining, and supervising all safety prediction and programs in connection with the performance of the Contract. The Centractor shall take all reasonable precautions to prevent damage, injury or loss to: workers, per ons rearby who may be affected, the Work, materials and equipment to be incorporated, and existing property at the site or adjacent thereto. The Contractor shall give notices and omply with applicable laws ordinances, rules regulations, and lawful orders of public authorities bearing on the safety of persons and property and their protection from injury, damage, or loss. The Contractor shall promptly remedy damage and loss to property at the site caused in whole or in part by the Contractor, a Subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable.

GRÙ

The Contractor shall notify the Owner in the event any existing hazardous material such as lead, PCBs, asbestos, etc. is encountered on the project. The Owner will arrange with a qualified specialist for the identification, testing, removal, handling and protection against exposure or environmental pollution, to comply with applicable regulation laws and ordinances. The Contractor and Architect will not be required to participate in or to perform this operation. Upon completion of this work, the Owner will notify the Contractor and Architect in writing the area has been cleared and approved by the authorities in order for the

Tetra Tech 200-15704-17001

GENERAL REQUIREMENT 00 81 13-11 work to proceed. The Contractor shall attach documentation from the authorities of said approval.

- 10.3 As required in the Hazardous Chemical Information Act of June 1984, all vendors supplying any materials that may be defined as hazardous, must provide Material Safety Data Sheets for those products. Any chemical product should be considered hazardous if it has a warning caution on the label relating to a potential physical or health hazard, if it is known to be present in the work place, and if employees may be exposed under normal conditions or in any foreseeable emergency situation. Material Safety Data Sheets <u>must</u> be provided <u>weethy</u> to the Owner along with the shipping slips that include those products.
- 10.4 The Contractor shall certify to the Owner that materials incorporated into the Work are free of all asbestos. This certification may be in the form of Material Safety Leta Sheet (MSDS) provided by the product manufacturer for the materials used in construction, as specified or as provided by the Contractor.

ARTICLE 11: INSURANCE AND BONDS

- 11.1 The Contractor shall carry all insurance required by any, such as Unemployment Insurance, etc. The Contractor shall carry such insurance coverage as they desire on their own property such as a field office, storage sheds or other structures erected upon the project site that belong to them and for their own use. The Subtontractors involved with this project shall carry whatever insurance protection they consider necessary to cover the loss of any of their personal property, etc.
- 11.2 Upon being awarded the Contract the Contractor shall obtain a minimum of two (2) copies of all required insurance catificates called for herein, and submit one (1) copy of each certificate, to the Owner, within 20 cays of contract award.
- 11.3 Bodily Injury Liability and Jroperty Damage Liability Insurance shall, in addition to the coverage included therein, include coverage for injury to or destruction of any property arising out of the collapse of or structural injury to any building or structure due to demolition work and evidence of these coverages shall be filed with and approved by the Owner.
- 11.4



The Contractor's Property Damage Liability Insurance shall, in addition to the coverage nord helein, include coverage on all real and personal property in their care, custody and control damaged in any way by the Contractor or their Subcontractors during the entire sonstruction period on this project.

Builders Risk (including Standard Extended Coverage Insurance) on the existing building during the entire construction period, shall not be provided by the Contractor under this contract. The Owner shall insure the existing building and all of its contents and all this new alteration work under this contract during entire construction period for the full insurable value of the entire work at the site. Note, however, that the Contractor and their Subcontractors shall be responsible for insuring building materials (installed and stored) and their tools and equipment whenever in use on the project, against fire damage, theft, vandalism, etc.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- 11.6 Certificates of the insurance company or companies stating the amount and type of coverage, terms of policies, etc., shall be furnished to the Owner, within 20 days of contract award.
- 11.7 The Contractor shall, at their own expense, (in addition to the above) carry the following forms of insurance:



Tetra Tech 200-15704-17001

- 11.7.7.1 With respect to all persons at any time employed by or on the payroll of the Contractor or performing any work for or on their behalf, or in connection with or arising out of the Contractor's business, the Contractor shall accept full and exclusive liability for the payment of any and all contributions or taxes or unemployment insurance, or old age retirement benefits, pensions or annuities now or hereafter imposed by the Government of the United States and the State or political subdivision thereof, whether the same be measured by wage salaries or other remuneration paid to such persons or otherwise.
- 11.7.7.2 Upon request, the Contractor shall furnish Owner such information on payrolls or employment records as may be necessary to enable it to fully comply with the law imposing the aforesaid contributions or taxes.
- 11.7.7.3 If the Owner is required by law to and does pay any and/or all of the aforesard contributions or taxes, the Contractor shall forthwith reimburse the Owner for the energy amount so paid by the Owner.

ARTICLE 12: UNCOVERING AND CORRECTION OF WORK

- 12.1 The Contractor shall promptly correct Work rejected by the Owner or failing to conform to the requirements of the Contract Documents, whether observed before or after Substantial Completion and whether or not fabricated, installed or completed, and shall correct any Work found to be not in accordance with the requirements of the Contract Documents within a period of two years from the date of Substantial Completion, or by terms of an applicable special warranty required by the Contract Documents. The provisions of this Article apply to work done by Subcontractors as well as h. Work done by direct employees of the Contractor.
- 12.2 At any time during the progress of the work, or in any case where the nature of the defects shall be such that it is not expedient to have them corrected, the Owner, at their option, shall have the right to deduct such cam, or sums, of money from the amount of the contract as they consider justified to adjust the difference in value between the defective work and that required under contract including any damage to the structure.

ARTICLE 13: MISCELLA NEOUS PROVISIONS

13.1

TTIN G AND PATCHING

13.1.1

The Contractor shall be responsible for all cutting and patching. The Contractor shall coordinate the work of the various trades involved.

DIMENSIONS

С

All dimensions shown shall be verified by the Contractor by actual measurements at the project site. Any discrepancies between the drawings and specifications and the existing conditions shall be referred to the Owner for adjustment before any work affected thereby has been performed.

13.3 LABORATORY TESTS

- 13.3.1 Any specified laboratory tests of material and finished articles to be incorporated in the work shall be made by bureaus, laboratories or agencies approved by the Owner and reports of such tests shall be submitted to the Owner. The cost of the testing shall be paid for by the Contractor.
- 13.3.2 The Contractor shall furnish all sample materials required for these tests and shall deliver same without charge to the testing laboratory or other designated agency when and when directed by the Owner.

13.4 ARCHAEOLOGICAL EVIDENCE

- 13.4.1 Whenever, in the course of construction, any archaeological evidence is incountered on the surface or below the surface of the ground, the Contractor shall notify the authorities of the Delaware Archaeological Board and suspend work in the immediate area for a reasonable time to permit those authorities, or persons designated by the state of examine the area and ensure the proper removal of the archaeological evidence for state ble preservation in the State Museum.
- 13.5 GLASS REPLACEMENT AND CLEANING
- 13.5.1 The General Contractor shall replace without expense to the Owner all glass broken during the construction of the project. If job conditions warrant, at completion of the job the General Contractor shall have all glass cleaned and polished.
- 13.6 WARRANTY
- 13.6.1 For a period of two (2) years from the date of substantial completion, as evidenced by the date of final acceptance of the work, the contractor warrants that work performed under this contract conforms to the contract requirements and is free of any defect of equipment, material or workmatchip performed by the contractor or any of his subcontractors or suppliers. However, manufacturer's warranties and guarantees, if for a period longer than two (2) years, shift take precedence over the above warranties. The contractor shall remedy, at his own expenses, any such failure to conform or any such defect. The protection of this warrante shall be included in the Contractor's Performance Bond.

ARTICLE 14: TERMINATION OF CONTRACT



If the Contractor defaults or persistently fails or neglects to carry out the Work in accordance with the Contract Documents or fails to perform a provision of the Contract, the Owner, after seven days written notice to the Contractor, may make good such deficiencies and may deduct the cost thereof from the payment then or thereafter due the Contractor. Alternatively, at the Owner's option, and the Owner may terminate the Contract and take possession of the site and of all materials, equipment, tools, and machinery thereon owned by the Contractor and may finish the Work by whatever method the Owner may deem expedient. If the costs of finishing the Work exceed any unpaid compensation due the Contractor, the Contractor shall pay the difference to the Owner.

14.2

"If the continuation of this Agreement is contingent upon the appropriation of adequate state, or federal funds, this Agreement may be terminated on the date beginning on the first fiscal year for which funds are not appropriated or at the exhaustion of the appropriation. The Owner may terminate this Agreement by providing written notice to

Tetra Tech 200-15704-17001

GENERAL REQUIREMENT 00 81 13-15 the parties of such non-appropriation. All payment obligations of the Owner will cease upon the date of termination. Notwithstanding the foregoing, the Owner agrees that it will use its best efforts to obtain approval of necessary funds to continue the Agreement by taking appropriate action to request adequate funds to continue the Agreement."

END OF SECTION Q^C orbund

EMPLOYEE DRUG TESTING REPORT FORM Period Ending:

4104 Regulations for the Drug Testing of Contractor and Subcontractor Employees Working on Large Public Works Projects requires that Contractors and Subcontractors who work on Large Public Works Contracts funded all or in part with public funds submit Testing Report Forms to the Owner no less than quarterly.

| 200-15704-17001 |
|--|
| Carrcroft Elementary School – Addition & Renocations |
| |
| |
| on the jobsite during the report period: |
| dom testing during the report period: |
| Jumber of Positive Results |
| nse to a failed or positive random test: |
| |
| actor/Subcontractor: |
| (typed or printed) |
| actor/Subcontractor: |
| (signature) |
| |

EMPLOYEE DRUG TESTING REPORT OF POSITIVE RESULTS

4104 Regulations for the Drug Testing of Contractor and Subcontractor Employees Working on Large Public Works Projects requires that Contractors and Subcontractors who work on Large Public Works Contracts funded all or in part with public funds to notify the Owner in writing of a positive random drug test.

| Project Number: | 200-15704-17001 |
|-------------------------------------|--|
| Project Name: | Carrcroft Elementary School – Addition & Renocations |
| Contractor/Subcontractor Name: | |
| Contractor/Subcontractor Address: | |
| Name of employee with positive tes | t result: |
| Last 4 digits of employee SSN: | |
| Date test results received: | \rightarrow |
| Action taken on employee in respon | se to a positive test result: |
| Authorized Representative of Contra | actor/Subcontractor: |
| Ruthonzed Representative of Contra | (typed or printed) |
| Authorized Representative of Contra | actor/Subcontractor: |
| Itate: | (signature) |
| | |

This form shall be sent by mail to the Owner within 24 hours of receipt of test results.

Enclose this test results form in a sealed envelope with the notation "Drug Testing Form – DO NOT OPEN" on the face thereof and place in a separate mailing envelope.

SECTION 01 10 00

SUMMARY

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Conditions and other Division 01 Specification Sections, apply to this Section

1.2 SUMMARY

- A. Section Includes:
 - 1. Project information.
 - 2. Work covered by Contract Documents.
 - 3. Phased construction.
 - 4. Work by Owner.
 - 5. Work under separate contracts.
 - 6. Future work.
 - 7. Purchase contracts.
 - 8. Owner-furnished products.
 - 9. Contractor-furnished, Owner-installed products.
 - 10. Access to site.
 - 11. Coordination with occ pants
 - 12. Work restrictions.
 - 13. Specification and drawing conventions.
 - 14. Miscellaneous previsions.
- B. Related Requirements
 - 1. Section 1 50 00 "Temporary Facilities and Controls" for limitations and procedures governing temporary use of Owner's facilities.

1.3 **PROJECT INFORMATION**

Project Identification: Carrcroft Elementary School Addition and Renovations

- 1. Project Location: 503 Crest Road, Wilmington, DE 19803
- B. Owner: Brandywine School District
 - 1. Owner's Address: 3305 Green Street, Claymont DE 19703
- C. Architect: Tetra Tech

Tetra Tech 200-15704-17001

- Architect's Consultants: The Architect has retained the following design professionals who have D. prepared designated portions of the Contract Documents:
 - Furlow Associates, MEP & FP 1.
 - 2. CDA Engineering, Civil Design
 - 3. Corsi Associates, Food Service

1.4 WORK COVERED BY CONTRACT DOCUMENTS

- The Work of Project is defined by the Contract Documents and consists of the following A.
 - Create a secure entrance to the school by relocating the administration of 1. enception function to be adjacent to the main entry in the existing cafeteria location
 - 2. Construct an addition to house the new cafeteria.
 - Convert existing cafeteria to new secure entrance as well 3. new administration offices.
 - The existing administration offices will be renovated into the new classrooms. 4.

WORK UNDER SEPARATE CONTRACTS 1.5

A. General: Cooperate fully with separate contractors of work on those contracts may be carried out smoothly, without interfering with or delaying work under this Contract or other contracts. Coordinate the Work of this Contract with work performed under separate contracts.

ACCESS TO SITE 1.6

- General: Contractor shall ha use of Project site for construction operations as A. 'in. 10 Contract limits and as indicated by requirements of this Section. indicated on Drawings by the
- Β. Use of Site: Limit up o Project site to work in areas indicated. Do not disturb portions of Project site beyond area in which the Work is indicated.
 - 1. Limits: Cor the construction operations to Limits of Disturbance.
 - Walkways and Entrances: Keep driveways and entrances serving premises 2. Dri ar and available to Owner, Owner's employees, and emergency vehicles at all times. cl not use these areas for parking or storage of materials.



- Schedule deliveries to minimize space and time requirements for storage of
- materials and equipment on-site.

COORDINATION WITH OCCUPANTS

Partial Owner Occupancy: Owner will occupy the premises during entire construction period, A. with the exception of areas under construction. Cooperate with Owner during construction operations to minimize conflicts and facilitate Owner usage. Perform the Work so as not to interfere with Owner's operations. Maintain existing exits unless otherwise indicated.

SUMMARY 01 10 00-2

- 1. Maintain access to existing walkways, corridors, and other adjacent occupied or used facilities. Do not close or obstruct walkways, corridors, or other occupied or used facilities without written permission from Owner and authorities having jurisdiction.
- Provide not less than 72 hours' notice to Owner of activities that will affect Owner's 2. operations.
- Owner Limited Occupancy of Completed Areas of Construction: Owner reserves the right B. occupy and to place and install equipment in completed portions of the Work, provide Substantial Completion of the Work, provided such occupancy does not interfer completion of the Work. Such placement of equipment and limited occupancy shall constitute acceptance of the total Work.
 - 1. Architect will prepare a Certificate of Substantial Completion for each pecifi portion of the Work to be occupied prior to Owner acceptance of the completed Vol-
 - Obtain a Certificate of Occupancy from authorities having jurisdiction before limited 2. Owner occupancy.
 - Before limited Owner occupancy, mechanical and electrical systems shall be fully 3. operational, and required tests and inspections shall be uccessfully completed. On occupancy, Owner will operate and maintain mechanical and electrical systems serving occupied portions of Work.
 - On occupancy, Owner will assume responsibility 4. r maintenance and custodial service for occupied portions of Work.

1.8 WORK RESTRICTIONS

- Work Restrictions, General: Comply with ctions on construction operations. A.
 - Comply with limitations 1. of public streets and with other requirements of On USE authorities having juri dictio
- On-Site Work Hours: Input Nork in the existing building to normal business working hours of Β. 7:00 a.m. to 3:00 p.m. Money through Friday, unless otherwise indicated.
 - Weeken a H urs: Must be arranged in advance. 1.
 - Early Morang Hours: Provide 72 hours' notice. Hears for Gility Shutdowns: 2.
 - 3.
 - Hours for Core Drilling: Provide 72 hours' notice. 4.
- C. Existing Utility Interruptions: Do not interrupt utilities serving facilities occupied by Owner or others unless permitted under the following conditions and then only after providing temporary tility services according to requirements indicated:
 - Notify Architect or Owner not less than three (3) days in advance of proposed utility interruptions.
 - 2. Obtain Architect and Owner written permission before proceeding with utility interruptions.
- Noise, Vibration, and Odors: Coordinate operations that may result in high levels of noise and D. vibration, odors, or other disruption to Owner occupancy with Owner.
- 1. Notify Architect and Owner not less than two (2) days in advance of proposed disruptive operations.
- 2. Obtain Architect and Owner written permission before proceeding with disruptive operations.
- E. Controlled Substances: Use of tobacco products and other controlled substances Project site not permitted.
- F. Employee Identification: Provide identification tags for Contractor personnel wor Project site. Require personnel to use identification tags at all times.
- G. Employee Screening: Comply with Owner's requirements drug and background screening of Contractor personnel working on Project site.
 - 1. Maintain list of approved screened personnel with Owner's representative

1.9 SPECIFICATION AND DRAWING CONVENTIONS

- A. Specification Content: The Specifications use certain convertions for the style of language and the intended meaning of certain terms, words, and phase when used in particular situations. These conventions are as follows:
 - 1. Imperative mood and streamlined language are generally used in the Specifications. The words "shall," "shall be," or "shall comply with," depending on the context, are implied where a colon (:) is used within a set ence or phrase.
 - 2. Specification requirements are to be performed by Contractor unless specifically stated otherwise.
- B. Division 01 General Requirements. Requirements of Sections in Division 01 apply to the Work of all Sections in the Specifications
- C. Drawing Coordination: requirements for materials and products identified on Drawings are described in detail in the Specifications. One or more of the following are used on Drawings to identify materials and products:
 - 1. Terminology: Materials and products are identified by the typical generic terms used in the individual Specifications Sections.
 - 2. Abbreviations: Materials and products are identified by abbreviations published as part of the U.S. National CAD Standard and scheduled on Drawings.
 - Reynoting: Materials and products are identified by reference keynotes referencing Specification Section numbers found in this Project Manual.

END OF SECTION

SECTION 01 21 00

ALLOWANCES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Conditions and other Division 01 Specification Sections, apply to this Section

1.2 SUMMARY

- A. Section includes administrative and procedural requirements governing allowances.
 - 1. Certain items are specified in the Contract Docusiens by allowances. Allowances have been established in lieu of additional requirements and to defer selection of actual materials and equipment to a later date when direction will be provided to Contractor. If necessary, additional requirements will be assure by Change Order.
- B. Types of allowances include the following
 - 1. Lump-sum allowances.
 - 2. Contingency allowances.

1.3 COORDINATION

A. Coordinate allowance items with other portions of the Work. Furnish templates as required to coordinate installation.

1.4 CONTINGENCY ALLOWANCES

A. Use the contingency allowance only as directed by Architect for Owner's purposes and only by Change Orders that indicate amounts to be charged to the allowance.

Contractor's overhead, and profit ordered by Owner under the contingency allowance are already included in the allowance and are part of the Contract Sum.

At Project closeout, credit unused amounts remaining in the contingency allowance to Owner by Change Order.

PART 2 - PRODUCTS (Not Used)

PART 3 - EXECUTION

3.1 EXAMINATION

A. Examine products covered by an allowance promptly on delivery for damage or defects. Return damaged or defective products to manufacturer for replacement.

3.2 PREPARATION

A. Coordinate materials and their installation for each allowance with related materials and installations to ensure that each allowance item is completely integrated and interfaced with related work.

3.3 SCHEDULE OF ALLOWANCES

A. Allowance No. 1: Contingency Allowance: Include a general contingency allowance of \$10,000.00.

SECTION 01 22 00

UNIT PRICES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Conditions and other Division 01 Specification Sections, apply to this Section

1.2 SUMMARY

- A. Section includes administrative and procedural requirements for unit prices.
- B. Related Requirements:
 - 1. Section 01 21 00 "Allowances" for procedures for using unit prices to adjust quantity allowances.
 - 2. Section 01 26 00 "Contract Modification Procedures" for procedures for submitting and handling Change Orders.
 - 3. Section 01 40 00 "Quality Requirements" for field testing by an independent testing agency.

1.3 DEFINITIONS

PROCEDU

- A. Unit price is an amount incorporated into the Agreement, applicable during the duration of the Work as a price per unit of measurement for materials, equipment, or services, or a portion of the Work, added to or deducted from the Contract Sum by appropriate modification, if the scope of Work or estimated quantities of Work required by the Contract Documents are increased or decreased.
- 1.4
- Unit prices include all necessary material, plus cost for delivery, installation, insurance, pplicable taxes, overhead, and profit.
- Measurement and Payment: See individual Specification Sections for work that requires establishment of unit prices. Methods of measurement and payment for unit prices are specified in those Sections.
- C. Owner reserves the right to reject Contractor's measurement of work-in-place that involves use of established unit prices and to have this work measured, at Owner's expense, by an independent surveyor acceptable to Contractor.

Tetra Tech 200-15704-17001

UNIT PRICES 01 22 00 -1 D. List of Unit Prices: A schedule of unit prices is included in Part 3. Specification Sections referenced in the schedule contain requirements for materials described under each unit price.

PART 2 - PRODUCTS (Not Used)

PART 3 - EXECUTION

3.1 SCHEDULE OF UNIT PRICES

- A. Unit Price No. 1: Removal of unsatisfactory soil and replacement with satisfactory soil material.
 - 1. Description: Unsatisfactory soil excavation and disposal off-site and replacement with satisfactory fill material or engineered fill from off-site as required, according to Section 31 20 00 "Earth Moving."
 - 2. Unit of Measurement: Cubic yard of soil excavated, based on implace surveys of volume before and after removal.
 - 3. Quantity Allowance: Coordinate unit price with allowance adjustment requirements in Section 01 21 00 "Allowances."
- B. Unit Price No. 2: Rock excavation and replacement with satisfactory soil material.
 - 1. Description: Classified rock excaption and disposal off-site and replacement with satisfactory fill material or engineered fill from off-site, as required, according to Section 31 20 00 "Earth Moving."
 - 2. Unit of Measurement: Cubic yard of rock excavated, based on survey of in-place surveys volume of before and after rem. vol.
 - 3. Quantity Allowance: Coordinate unit price with allowance adjustment requirements in Section 01 21 00 "All waves."
- C. Unit Price No. 3: Cuttine and patching of concrete slabs-on-grade.
 - 1. Description Cutting of new or existing concrete slabs-on-grade up to 6 inches thick, removed and excavation as required, and subsequent backfill, compaction, and patching of concrete according to Section 01 73 00 "Execution." not otherwise indicated in the Contract Documents.
 - 2. Unit of Measurement: Square feet of concrete removed.

END OF SECTION

SECTION 01 23 00

ALTERNATES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Conditions and other Division 01 Specification Sections, apply to this Section

1.2 SUMMARY

A. Section includes administrative and procedural requirements for attenutes.

1.3 DEFINITIONS

- A. Alternate: An amount proposed by bidders and stated on the Bid Form for certain work defined in the bidding requirements that may be added to or deducted from the base bid amount if Owner decides to accept a corresponding charge either in the amount of construction to be completed or in the products, materials equipment, systems, or installation methods described in the Contract Documents.
 - 1. Alternates described in this Section are part of the Work only if enumerated in the Agreement.
 - 2. The cost or credit for each adernate is the net addition to or deduction from the Contract Sum to incorporate adernate into the Work. No other adjustments are made to the Contract Sum.

1.4 PROCEDURES

A. Coordination: Devise or adjust affected adjacent work as necessary to completely integrate work of the alternate into Project.



Include as part of each alternate, miscellaneous devices, accessory objects, and similar items incidental to or required for a complete installation whether or not indicated as part of alternate.

Notification: Immediately following award of the Contract, notify each party involved, in writing, of the status of each alternate. Indicate if alternates have been accepted, rejected, or deferred for later consideration. Include a complete description of negotiated revisions to alternates.

C. Execute accepted alternates under the same conditions as other work of the Contract.

D. Schedule: A schedule of alternates is included at the end of this Section. Specification Sections referenced in schedule contain requirements for materials necessary to achieve the work described under each alternate.

PART 2 - PRODUCTS (Not Used)

PART 3 - EXECUTION

3.1 SCHEDULE OF ALTERNATES

- A. Alternate No. 1: Emergency Generator:
 - 1. Base Bid: <Insert brief description of base-bid requirement> [as indicated on Sheet <Insert title of sheet>] [and] [as specified in Section insert Section number> ''<Insert Section title>.'']
 - 2. Alternate: <Insert brief description of alternate requirement> [as indicated on Sheet <Insert title of sheet>] [and] [as specified in Section <Insert Section number> ''<Insert Section title>.'']
- B. Alternate No. 2: Corridor & Gymnasium Lighting
 - 1. Base Bid: <Insert brief description of base-bid requirement> [as indicated on Sheet <Insert title of sheet>] [and] [as specified in Section <Insert Section number> ''<Insert Section title>.'']
 - 2. Alternate: <Insert brief description of alternate requirement> [as indicated on Sheet <Insert title of sheet] [and] [as specified in Section <Insert Section number> "<Insert Section titles "]

END OF SECTION

SECTION 01 25 00

SUBSTITUTION PROCEDURES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Requests for substitution must be made ten days prior to bid. This specification section ap to extra-ordinary conditions that could not be requested during the bidding period.
- B. Drawings and general provisions of the Contract, including General and Supprement Conditions and other Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. This Section includes administrative and procedural requirements for handling requests for substitutions made after award of the Contract, but no later than 60 cays after commencement of the Work.
- B. Related Sections: The following Divisions contain requirements that relate to this Section:
 - 1. Division 01 specifies the applicability of i dustry standards to products specified.
 - 2. Division 01 specifies requirements for summary the Contractor's Construction Schedule and the Submittal Schedule.
 - 3. Division 01 specifies requirements coverning the Contractor's selection of products and product options.

1.3 DEFINITIONS

- A. Definitions in this Article do not enange or modify the meaning of other terms used in the Contract Documents.
- B. Substitutions: changes in products, materials, equipment, and methods of construction required by the Contract Documents proposed by the Contractor after award of the Contract are considered to be requests for substitutions. The following are not considered to be requests for substitutions:
 - . Substitutions requested during the bidding period, and accepted by Addendum prior to ward of the Contract, are included in the Contract Documents and are not subject to requirements specified in this Section for substitutions.
 - Revisions to the Contract Documents requested by the Owner or Architect.
 - Specified options of products and construction methods included in the Contract Documents.
 - 4. The Contractor's determination of and compliance with governing regulations and orders issued by governing authorities.

1.4 SUBMITTALS

A. Substitution Request Submittal: Substitution requests will only be considered during the bidding period. Substitutions will not be considered after the bids are accepted.

- 1. Submit three copies of each request for substitution for consideration. Submit requests in the form and according to procedures required for change order proposals and utilizing the CSI Substitution Request Form 13.1A (sample attached to Project Manual). The contractor is solely responsible for obtaining the required forms to submit before the stated time period expires.
- 2. Identify the product or the fabrication or installation method to be replaced in each request. Include related Specification Section and Drawing numbers.
- 3. Provide complete documentation showing compliance with the requirements for substitutions, and the following information, as appropriate:
 - a. Coordination information, including a list of changes or modifications needed to other parts of the Work and to construction performed by the Owner and separate contractors that will be necessary to accommodate the proposed substitution.
 - b. A detailed comparison of significant qualities of the proposed abstitution with those of the Work specified. Significant qualities may include elements, such as performance, weight, size, durability, and visual effect.
 - c. Product Data, including Drawings and descriptions of products and fabrication and installation procedures.
 - d. Samples, where applicable or requested.
 - e. A statement indicating the substitution's effect on the Contractor's Construction Schedule compared to the schedule without approval of the substitution. Indicate the effect of the proposed substitution or overall Contract Time.
 - f. Cost information, including a proposal of the net change, if any in the Contract Sum.
 - g. The Contractor's certification that the proposed substitution conforms to requirements in the Contrast Pocuments in every respect and is appropriate for the applications indicated
 - h. The Contractor's war er of aghts to additional payment or time that may subsequently become necessary because of the failure of the substitution to perform adequately.
- 4. Architect's Actions Thecessary, the Architect will request additional information or documentation for evaluation within one week of receipt of a request for substitution. The Architect will notify the Contractor of acceptance or rejection of the substitution within two weeks of receipt of the request, or one week of receipt of additional information or documentation, whichever is later.

Use the product specified if the Architect cannot make a decision on the use of a proposed substitute within the time allocated.

PT 2 - IRODUCTS

SUBSTITUTIONS

Conditions: The Architect will receive and consider the Contractor's request for substitution when the following conditions are satisfied, as determined by the Architect. If the following conditions are not satisfied, the Architect will return the requests without action except to record noncompliance with these requirements.

- 1. Revisions to the Contract Documents are not required.
- 2. Proposed changes are in keeping with the general intent of the Contract Documents.
- 3. The request is timely, fully documented, and properly submitted.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- 4. The specified product or method of construction cannot be provided within the Contract Time. The Architect will not consider the request if the product or method cannot be provided as a result of failure to pursue the Work promptly or coordinate activities properly.
- 5. The requested substitution offers the Owner a substantial advantage, in cost, time, energy conservation, or other considerations, after deducting additional responsibilities the Owner must assume. The Owner's additional responsibilities may include compensation to the Architect for redesign and evaluation services, increased cost of other construction by the Owner, and similar considerations.
- 6. The specified product or method of construction cannot receive necessary approvaloy a governing authority, and the requested substitution can be approved.
- 7. The specified product or method of construction cannot be provided in a manyer that is compatible with other materials and where the Contractor certification the substitution will overcome the incompatibility.
- 8. The specified product or method of construction cannot be coord nated with other materials and where the Contractor certifies that the proposed substitution can be coordinated.
- B. The Contractor's submittal and the Architect's acceptance of Shop Drawings, Product Data, or Samples for construction activities not complying with the Contract Documents do not constitute an acceptable or valid request for substruction, nor do they constitute approval.

SECTION

PART 3 - EXECUTION (Not Applicable)



SECTION 01 26 00

CONTRACT MODIFICATION PROCEDURES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. This Section specifies administrative and procedural requirements for haddling and processing Contract modifications.
- B. Related Sections include the following:
 - 1. Division 01 Section "Allowances" for procedural requirements for handling and processing allowances.

1.3 MINOR CHANGES IN THE WORK

A. The Architect will issue supplemental instructions acanorizing Minor Changes in the Work, not involving adjustment to the Contract Sum of the Contract Time, on AIA Document G710, "Architect's Supplemental Instructions".

1.4 PROPOSAL REQUESTS

- A. Owner-Initiated Proposal Requests: The Architect will issue a detailed description of proposed changes in the Work the may require adjustment to the Contract Sum or the Contract Time. If necessary, the description will include supplemental or revised Drawings and Specifications.
 - 1. Proposal Requests issued by the Architect are for information only. Do not consider them inspections either to stop work in progress or to execute the proposed change.
 - 2. Within the specified in Proposal Request after receipt of Proposal Request, submit a quatation estimating cost adjustments to the Contract Sum and the Contract Time necessary to execute the change. Refer to procedures outlined in the *Supplementary Conditions* of the Contract.

Contractor-Initiated Proposals: If latent or unforeseen conditions require modifications to the Contract, Contractor may propose changes by submitting a request for a change to the Architect. Refer to Procedures outlined in the *Supplementary Conditions* of the Contract.

ALLOWANCES

A. Submit claims for increased costs because of a change in scope or nature of the allowance described in the Contract Documents, whether for the Purchase Order amount or Contractor's handling, labor, installation, overhead, and profit. Submit claims within 14 days of receipt of the Change Order or Construction Change Directive authorizing work to proceed. Owner will reject claims submitted later than 21 days after such authorization.

Tetra Tech 200-15704-17001

- 1. Do not include Contractor's or subcontractor's indirect expense in the Change Order cost amount unless it is clearly shown that the nature or extent of work has changed from what could have been foreseen from information in the Contract Documents.
- 2. No change to Contractor's indirect expense is permitted for selection of higher- or lowerpriced materials or systems of the same scope and nature as originally indicated.

1.6 CHANGE ORDER PROCEDURES

- A. On Owner's approval of a Proposal Request, the Architect will issue a Change Order for signatures of Owner and Contractor on AIA Document G701.
- 1.7 CONSTRUCTION CHANGE DIRECTIVE
 - A. Work Change Directive: The Architect may issue a Work Change Directive on AIA Document G714. Work Change Directive instructs Contractor to proceed with a change in the Work, for subsequent inclusion in a Change Order.
 - 1. Work Change Directive contains a complete description of change in the Work. It also designates method to be followed to determine change in the Contract Sum or the Contract Time.
 - B. Documentation: Maintain detailed records on a time and material basis of work required by the Work Change Directive.
 - 1. After completion of change, submit to new zed account and supporting data necessary to substantiate cost and time adjustments to the Contract.

PART 2 - PRODUCTS (Not Used)



END OF SECTION

SECTION 01 29 00

PAYMENT PROCEDURES

PART 1 – GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementar Conditions and other Division 01 Specification Sections, apply to this Section

1.2 SUMMARY

- A. This Section specifies administrative and procedural requirements over sing the Contractor's Applications for Payment.
- B. This Section specifies administrative and procedural requirement governing each prime contractor's Applications for Payment.
 - 1. Coordinate the Schedule of Values and Applications for Payment with the Contractor's Construction Schedule, Submittal Schedule, and List of Subcontracts.
- C. Related Sections: The following Sections contain requirements that relate to this Section.
 - 1. Schedules: The Contractor's Construction Schedule and Submittal Schedule are specified in Division 01 vection AConstruction Progress Documentation.
 - 2. Division 01 Section "Contrast Modification Procedures" for administrative procedures for handling changes to the Contract.
 - 3. Division 01 Section "Submittal Procedures" for administrative requirements governing the preparation and ubmittal of the submittal schedule.

1.3 DEFINITIONS

A. Schedule of values: A statement furnished by Contractor allocating portions of the Contract Sum o various portions of the Work and used as the basis for reviewing Contractor's Applications for Payment

1.4 SCHEDULE OF VALUES

Coordination: Coordinate preparation of the Schedule of Values with preparation of the Contractor's Construction Schedule.

- 1. Correlate line items in the Schedule of Values with other required administrative schedules and forms, including:
 - a. Contractor's Construction Schedule.
 - b. Application for Payment forms, including Continuation Sheets.
 - c. List of subcontractors.
 - d. Schedule of allowances.
 - e. Schedule of alternates.

Tetra Tech 200-15704-17001

- f. List of products.
- List of principal suppliers and fabricators. g.
- Schedule of submittals. h.
- 2. Submit the Schedule of Values to the Architect at the earliest possible date but no later than 14 days before the date scheduled for submittal of the initial Applications for Payment.
- Β. Format and Content: Use the Project Manual table of contents as a guide to establish format for the Schedule of Values. Provide at least one line item for each Specificati Section.
 - 1. Identification: Include the following Project identification on the of Values: dule
 - Project name and location. a.
 - Name of the Architect. b.
 - Project number. c.
 - Contractor's name and address. d.
 - Date of submittal. e.
 - 2. Arrange the Schedule of Values in tabular form who separate columns to indicate the following for each item listed:
 - Related Specification Section of a. Division
 - Description of Work. b.
 - Name of subcontrac c.
 - Name of manufacturer of fabricator. d.
 - e.
 - Name of suppliers Change Orders (numbers) that affect value. f.
 - Dollar value of tazterial cost. g.
 - Dolla, value on labor cost. h.
 - alue total material and labor cost. Dollar i.

Percentage of Contract Sum to nearest one-hundredth percent, adjusted to total 100 percent.

de a breakdown of the Contract Sum in sufficient detail to facilitate continued evaluation of Applications for Payment and progress reports. Coordinate with the Project Manual table of contents. Break principal subcontract amounts down into several line items.

- Include a like item for "Closeout Documents". This amount shall equal three percent (3%) of the total contract amount.
- 5. Round amounts to nearest whole dollar; the total shall equal the Contract Sum.
- 6. Provide a separate line item in the Schedule of Values for each part of the Work where Applications for Payment may include materials or equipment, purchased or fabricated and stored, but not yet installed.
 - Differentiate between items stored on-site and items stored off-site. Include a. requirements for insurance and bonded warehousing, if required.

- 7. Provide separate line items on the Schedule of Values for initial cost of the materials, for each subsequent stage of completion, and for total installed value of that part of the Work.
- 8. Margins of Cost: Show line items for indirect costs and margins on actual costs only when such items are listed individually in Applications for Payment. Each item in the Schedule of Values and Applications for Payment shall be complete. Include the total cost and proportionate share of general overhead and profit margin for each item.
 - a. Temporary facilities and other major cost items that are not direct cost of a ual work-in-place may be shown either as separate line items in the Schedulerst Values or distributed as general overhead expense, at the Contractor's option.
- 9. Schedule Updating: Update and resubmit the Schedule of Values prior to the next Applications for Payment when Change Orders or Construction Change Directives result in a change in the Contract Sum.

1.5 APPLICATIONS FOR PAYMENT

- A. Each Application for Payment shall be consistent with previous applications and payments as certified by the Architect and paid for by the Owner. Sam Application for Payment shall be accompanied with copies of the Weekly Certified Payrol Reports as submitted to the Department of Labor for review by the Architect and Owner. This is in addition to the submission of the weekly Certified Payrol Reports to the State of Delaware.
 - 1. The initial Application for Payment, the Application for Payment at time of Substantial Completion, and the final Application for Payment involve additional requirements.
- B. Payment-Application Timer. Eccorrogress-payment date is indicated in the Agreement. The period of construction Work covered by each Application for Payment is the period indicated in the Agreement.
- C. Payment-Application Forms: Use AIA Document G702 and Continuation Sheets G703 as the form for Applications for Payment. Electronic versions shall only be acceptable if they are identical incorpat to the G702 and G703 forms.
- D. Application Preparation: Complete every entry on the form. Include notarization and execution by a person authorized to sign legal documents on behalf of the Contractor. The availated will return incomplete applications without action.
 - Entries shall match data on the Schedule of Values and the Contractor's Construction Schedule. Use updated schedules if revisions were made.
 - Include amounts of Change Orders and Construction Change Directives issued prior to the last day of the construction period covered by the application.
 - Transmittal: Submit 5 signed and notarized original copies of each Application for Payment to the Architect by a method ensuring receipt within 24 hours. One copy shall be complete, including waivers of lien and similar attachments, when required.
 - 1. Transmit each copy with a transmittal form listing attachments and recording appropriate information related to the application, in a manner acceptable to the Architect.

Tetra Tech 200-15704-17001

1.

2.

- F. Waivers of Mechanics Lien: With each Application for Payment, submit waivers of mechanics liens from subcontractors, sub-subcontractors and suppliers for the construction period covered by the previous application.
 - 1. Submit partial waivers on each item for the amount requested, prior to deduction for retainage, on each item.
 - 2. When an application shows completion of an item, submit final or full waivers.
 - 3. The Owner reserves the right to designate which entities involved in the Work maint waivers.
 - 4. Waiver Delays: Submit each Application for Payment with the Contractor's waiver of mechanics lien for the period of construction covered by the application.
 - a. Submit final Applications for Payment with or preceded by final waivers from every entity involved with performance of the Work covered by the application who is lawfully entitled to a lien.
 - 5. Waiver Forms: Submit waivers of lien on forms, and executed in a manner, acceptable to the Owner.
- G. Initial Application for Payment: Administrative actions and submittals, that must precede or coincide with submittal of the first Application for Ayment, include the following:
 - 1. List of subcontractors.
 - 2. List of principal suppliers and fabricators.
 - 3. Schedule of Values.
 - 4. Contractor's Construction Schecule (preliminary if not final).
 - 5. Schedule of principal produ
 - 6. Schedule of unit prices.
 - 7. Submittal Schedur (preliminary if not final).
 - 8. List of Contractor's stoff assignments.
 - 9. List of Contra tor's principal consultants.
 - 10. Copie of building permits.
 - 11. Copies computerizations and licenses from governing authorities for performance of the Work.
 - 12. Initial progress report.
 - Report of preconstruction meeting.
 - Certificates of insurance and insurance policies.
 - 15. Performance and payment bonds.
 - 16. Data needed to acquire the Owner's insurance.
 - 17. Initial settlement survey and damage report, if required.

Application for Payment at Substantial Completion: Following issuance of the Certificate of Substantial Completion, submit an Application for Payment.

- 1. This application shall reflect Certificates of Partial Substantial Completion issued previously for Owner occupancy of designated portions of the Work.
- 2. Administrative actions and submittals that shall precede or coincide with this application include:
 - a. Occupancy permits and similar approvals.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- b. Warranties (guarantees) and maintenance agreements.
- c. Test/adjust/balance records.
- d. Maintenance instructions.
- e. Meter readings.
- f. Startup performance reports.
- g. Changeover information related to Owner's occupancy, use, operation, and maintenance.
- h. Final cleaning.
- i. Application for reduction of retainage and consent of surety.
- j. Advice on shifting insurance coverages.
- k. Final progress photographs.
- 1. List of incomplete Work, recognized as exceptions to Architect's Certificate of Substantial Completion.
- I. Final Payment Application: Administrative actions and submittal that hust precede or coincide with submittal of the final Application for Payment include the following:
 - 1. Completion of Project closeout requirements.
 - 2. Completion of items specified for completion after Jud tratial Completion.
 - 3. Ensure that unsettled claims will be settled.
 - 4. Ensure that incomplete Work is not accepted and will be completed without undue delay.
 - 5. Transmittal of required Project construction records to the Owner.
 - 6. Certified property survey.
 - 7. Proof that taxes, fees, and similar obligations were paid.
 - 8. Removal of temporary facilities and services.
 - 9. Removal of surplus materials, rubbish, and similar elements.
 - 10. Change of door locks to Owner's access.
- 2 PRODUCTS (Not Applicabl
- 3 EXECUTION (Not Applicable)

END OF SECTION



SECTION 01 31 00

PROJECT MANAGEMENT & COORDINATION

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. This Section includes administrative provisions for coordinating construction operations on the Project including, but not limited to, the following:
 - 1. General project coordination procedures.
 - 2. Coordination Drawings.
 - 3. Administrative and supervisory personnel.
 - 4. Project meetings.
- B. Each contractor shall participate in coordination requirements. Certain areas of responsibility will be assigned to a specific contractor.
- C. Related Sections: The following Sections contain requirements that relate to this Section:
 - 1. Division 01 Section "Closecut Procedures" for coordinating Contract closeout.

1.3 COORDINATION

- A. Coordination: Coordinate construction operations included in various Sections of the Specifications to ensure affident and orderly installation of each part of the Work. Coordinate construction operations, he juded in different Sections that depend on each other for proper installation, come ation, and operation.
 - 1. Schedule construction operations in sequence required to obtain the best results where installation of one part of the Work depends on installation of other components, before or after its own installation.
 - Coordinate installation of different components with other contractors to ensure maximum accessibility for required maintenance, service, and repair.
 - Make adequate provisions to accommodate items scheduled for later installation.

B. I

If necessary, prepare memoranda for distribution to each party involved, outlining special procedures required for coordination. Include such items as required notices, reports, and list of attendees at meetings.

1. Prepare similar memoranda for the Owner and separate contractors if coordination of their Work is required.

- C. Administrative Procedures: Coordinate scheduling and timing of required administrative procedures with other construction activities and activities of other contractors to avoid conflicts and to ensure orderly progress of the Work. Such administrative activities include, but are not limited to, the following:
 - 1. Preparation of the Contractor's Construction Schedule.
 - 2. Preparation of the Schedule of Values.
 - 3. Installation and removal of temporary facilities and controls.
 - 4. Delivery and processing of submittals.
 - 5. Progress meetings.
 - 6. Preinstallation conferences.
 - 7. Project closeout activities.
- D. Conservation: Coordinate construction activities to ensure that operations are carried out with consideration given to conservation of energy, water, and materials.

1.4 SUBMITTALS

- A. Coordination Drawings: Prepare Coordination Drawings if limited space availability necessitates maximum utilization of space for efficient installation of different components or if coordination is required for installation of products and materials fabricated by separate entities.
 - 1. Indicate relationship of components show on se arate Shop Drawings.
 - 2. Indicate required installation sequences.
- B. Staff Names: Within 15 days of starting construction operations, submit a list of principal staff assignments, including superintend of and other personnel in attendance at the Project site. Identify individuals and their duties and risponsibilities; list addresses and telephone numbers, including home and office telephone numbers. Provide names, addresses, and telephone numbers of individuals assigned as standbys in the absence of individuals assigned to the Project.
 - 1. Post copies of listin the Project meeting room, in temporary field office, and by each temporary telephone.
- 1.5 PROJECT INCENINGS
 - A. General: the Architectural/Engineering Consultant shall Schedule and conduct meetings and conductnees at the Project site, unless otherwise indicated.



- Attendees: Inform participants and others involved, and individuals whose presence is required, of date and time of each meeting. Notify the Owner and the Contractor of scheduled meeting dates and times.
- Agenda: Prepare the meeting agenda. Distribute the agenda to all invited attendees.
- Minutes: Record significant discussions and agreements achieved. Distribute the meeting minutes to everyone concerned, including the Owner and the Architect, within 3 days of the meeting.
- B. Preconstruction Conference: The Architectural/Engineering Consultant shall Schedule a preconstruction conference before starting construction, at a time convenient to the Owner and the Architect, but no later than 15 days after execution of the Agreement. Hold the conference

at the Project site or another convenient location. Conduct the meeting to review responsibilities and personnel assignments.

- 1. Attendees: Authorized representatives of the Owner, the Contractor, and their consultants; the Contractor and its superintendent; major subcontractors; manufacturers; suppliers; and other concerned parties shall attend the conference. All participants at the conference shall be familiar with the Project and authorized to conclude matters relating to the Work.
- 2. Agenda: Discuss items of significance that could affect progress, including the following:
 - a. Tentative construction schedule.
 - b. Phasing.
 - c. Critical work sequencing.
 - d. Designation of responsible personnel.
 - e. Procedures for processing field decisions and Change Ord
 - f. Procedures for processing Applications for Payment
 - g. Distribution of the Contract Documents.
 - h. Submittal procedures.
 - i. Preparation of Record Documents.
 - j. Use of the premises.
 - k. Responsibility for temporary facilities and controls.
 - 1. Parking availability.
 - m. Office, work, and storage areas.
 - n. Equipment deliveries and price
 - o. First aid.
 - p. Security.
 - q. Progress cleaning.
 - r. Working hours.
- C. Progress Meetings: The Architectural/Engineering Consultant shall conduct progress meetings at bimonthly intervals. Coordinate dates of meetings with preparation of payment requests.
 - 1. Agenda: review and correct or approve minutes of previous progress meeting. Review other items of significance that could affect progress. Include topics for discussion as appropriate to status of the Project.

Contractor's Construction Schedule: Review progress since the last meeting. Determine whether each activity is on time, ahead of schedule, or behind schedule, in relation to the Contractor's Construction Schedule. Determine how construction behind schedule will be expedited; secure commitments from parties involved to do so. Discuss whether schedule revisions are required to ensure that current and subsequent activities will be completed within the Contract Time.

- b. Review present and future needs of each entity present, including the following:
 - 1) Interface requirements.
 - 2) Sequence of operations.
 - 3) Status of submittals.
 - 4) Deliveries.
 - 5) Off-site fabrication.
 - 6) Access.

- 7) Site utilization.
- 8) Temporary facilities and controls.
- 9) Work hours.
- 10) Hazards and risks.
- 11) Progress cleaning.
- 12) Quality and work standards.
- 13) Change Orders.

RBIC

- 14) Documentation of information for payment requests.
- 2. Reporting: The Architectural/Engineering Consultant shall distribute minute of the meeting to each party present and to parties who should have been present. Include brief summary, in narrative form, of progress since the previous meeting, and report.

END OF

a. Schedule Updating: Revise the Contractor's Construction Schedule after each progress meeting where revisions to the schedule have been 1 ade or recognized. Issue revised schedule concurrently with the report of each meeting.

PART 2 - PRODUCTS (Not Used)

PART 3 - EXECUTION (Not Used)

SECTION 01 31 20

PAYROLL REPORTS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementat Conditions and other Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. This Section includes administrative and procedural requirements for schedules and reports required for proper performance of the Work, including:
 - 1. State of Delaware Payroll Reports.
- B. Related Sections: The following Sections contain requirements that relate to this Section:
 - 1. Division 01 Section "Applications for laymen" specifies requirements for submittal of the Schedule of Values.
 - 2. Division 01 Section "Project Meetings" pecifies requirements for submittal and distribution of meeting and conference minutes.

1.3 SUBMITTAL PROCEDURES

A. Coordination: Coordinate repartition and processing of schedules and reports with performance of other construction activities.

1.4 PAYROLL REPORTS

A. State of Denware Payroll Reports: As required by the State of Delaware, Section 6960, Title 29, of the Delaware Code, payroll wages shall be reported weekly to the Delaware Department of Labor, Division of Industrial Affairs, 4425 North Market Street, Wilmington, DE 19892 phone 302-761-8200. Forms shall be available at the above address. A sample curv of the form is attached under contract forms, State of Delaware Payroll Report.

PRODUCTS (Not Applicable)

3 - EXECUTION (Not Applicable)

END OF SECTION



SECTION 01 32 00

CONSTRUCTION PROGRESS DOCUMENTATION

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 1 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. This Section includes administrative and procedural requirements for the currenting the progress of construction during performance of the Work, including the following
 - 1. Preliminary Construction Schedule.
 - 2. Contractor's Construction Schedule.
 - 3. Submittals Schedule.
 - 4. Daily construction reports.
 - 5. Material location reports.
 - 6. Field condition reports.
 - 7. Construction photographs.
- B. Related Sections include the following:
 - 1. Division 01 Section "Paymen Procedures" for submitting the Schedule of Values.
 - 2. Division 01 Section "Project Management and Coordination" for submitting and distributing meeting and conference minutes.
 - 3. Division 01 Section "Submittal Procedures" for submitting schedules and reports.
 - 4. Division 01 Section "Iloseout Procedures" for submitting photographic negatives as Project Record Downents at Project closeout.
- 1.3 DEFINITION
 - A. Activity: A discrete part of a project that can be identified for planning, scheduling, monitoring, and controlling the construction project. Activities included in a construction schedule consume time and resources.



Critical activities are activities on the critical path. They must start and finish on the planned early start and finish times.

- Predecessor activity is an activity that must be completed before a given activity can be started.
- CPM: Critical path method, which is a method of planning and scheduling a construction project where activities are arranged based on activity relationships. Network calculations determine when activities can be performed and the critical path of Project.
- C. Critical Path: The longest continuous chain of activities through the network schedule that establishes the minimum overall Project duration and contains no float.

- D. Event: The starting or ending point of an activity.
- E. Float: The measure of leeway in starting and completing an activity.
 - 1. Float time is not for the exclusive use or benefit of either Owner or Contractor, but is a jointly owned, expiring Project resource available to both parties as needed to meet schedule milestones and Contract completion date.
 - 2. Free float is the amount of time an activity can be delayed without adversely affecting the early start of the following activity.
 - 3. Total float is the measure of leeway in starting or completing an activity without adversely affecting the planned Project completion date.
- F. Major Area: A story of construction, a separate building, or a similar significant construction element.
- G. Milestone: A key or critical point in time for reference or measurement.
- H. Network Diagram: A graphic diagram of a network schedule, showing activities and activity relationships.
- 1.4 SUBMITTALS
 - A. Qualification Data: For firms and persons specified in Quality Assurance" Article and inhouse scheduling personnel to demonstrate their constrainties and experience. Include lists of completed projects with project names and addresses, names and addresses of architects and owners, and other information specified.
 - B. Submittals Schedule: Submit three copie of schedule. Arrange the following information in a tabular format:
 - 1. Scheduled date for first submittal.
 - 2. Specification Section number and title.
 - 3. Submittal category (action or informational).
 - 4. Name of an contractor.
 - 5. Description of the Work covered.
 - C. Contractor's Construction Schedule: Submit three printed copies of initial schedule, one a reproducible paint and one a blue- or black-line print, large enough to show entire schedule for entire construction period.

CPM Reports: Concurrent with CPM schedule, submit three printed copies of each of the following computer-generated reports. Format for each activity in reports shall contain activity number, activity description, original duration, remaining duration, early start date, early finish date, late start date, late finish date, and total float.

- 1. Activity Report: List of all activities sorted by activity number and then early start date, or actual start date if known.
- 2. Logic Report: List of preceding and succeeding activities for all activities, sorted in ascending order by activity number and then early start date, or actual start date if known.
- 3. Total Float Report: List of all activities sorted in ascending order of total float.

- E. Daily Construction Reports: Submit two copies at weekly intervals.
- F. Material Location Reports: Submit two copies at weekly intervals.
- G. Field Condition Reports: Submit two copies at weekly intervals.
- 1.5 QUALITY ASSURANCE
 - A. Scheduling Consultant Qualifications: An experienced specialist in CPM scheduling and reporting.
- 1.6 COORDINATION
 - A. Coordinate preparation and processing of schedules and reports with performance construction activities and with scheduling and reporting of separate contractors.
 - B. Coordinate Contractor's Construction Schedule with the Schedule of values, list of subcontracts, Submittals Schedule, progress reports, payment requests, and other required schedules and reports.
 - 1. Secure time commitments for performing critical examents of the Work from parties involved.
 - 2. Coordinate each construction activity in the network with other activities and schedule them in proper sequence.

PART 2 - PRODUCTS

2.1 SUBMITTALS SCHEDULE

- A. Preparation: Submit a schedule of submittals, arranged in chronological order by dates required by construction schedule. Include time required for review, resubmittal, ordering, manufacturing, fabrication, and delivery when establishing dates.
 - 1. Coordinate submittals Schedule with list of subcontracts, the Schedule of Values, and Contractor's Construction Schedule.
 - 2. Final Submittal: Submit concurrently with the first complete submittal of Contractor's Constraction Schedule.
- .2 CONTRACTOR'S CONSTRUCTION SCHEDULE (CPM SCHEDULE)

CM Schedule: Prepare Contractor's Construction Schedule using a CPM network analysis diagram.

- 1. Develop network diagram in sufficient time to submit CPM schedule so it can be accepted for use no later than 15 days after date established for the Notice to Proceed..
- 2. Establish procedures for monitoring and updating CPM schedule and for reporting progress. Coordinate procedures with progress meeting and payment request dates.
- 3. Use "one workday" as the unit of time.

- B. CPM Schedule Preparation: Prepare a list of all activities required to complete the Work. Using the preliminary network diagram, prepare a skeleton network to identify probable critical paths.
 - 1. Activities: Indicate the estimated time duration, sequence requirements, and relationship of each activity in relation to other activities. Include estimated time frames for the following activities:
 - a. Preparation and processing of submittals.
 - b. Purchase of materials.
 - c. Delivery.
 - d. Fabrication.
 - e. Installation.
 - 2. Processing: Process data to produce output data or a computer draw a, time-scaled network. Revise data, reorganize activity sequences, and reproduce as often as necessary to produce the CPM schedule within the limitations of the Cov ract Time.
 - 3. Format: Mark the critical path. Locate the critical path near tenter of network; locate paths with most float near the edges.
 - a. Subnetworks on separate sheets are permissible for activities clearly off the critical path.

2.3 REPORTS

3

- A. Daily Construction Reports: Prepare a daily construction report recording the following information concerning events at Project site:
 - 1. List of subcontractors a Projectore
 - 2. List of separate contractors at Project site.
 - 3. Approximate count of personnel at Project site.
 - 4. High and low temperatures and general weather conditions.
 - 5. Accidents.
 - 6. Meetings and significant decisions.
 - 7. Unusual events (refer to special reports).
 - 8. Stoppages, delays, shortages, and losses.
 - 9. Meter readings and similar recordings.
 - 10. Emergency procedures.
 - 11. Orders and requests of authorities having jurisdiction.
 - 12. Change Orders received and implemented.
 - Construction Change Directives received.
 - Services connected and disconnected.
 - 15. Equipment or system tests and startups.
 - 16. Partial Completions and occupancies.
 - 17. Substantial Completions authorized.
- B. Material Location Reports: At monthly intervals, prepare a comprehensive list of materials delivered to and stored at Project site. List shall be cumulative, showing materials previously reported plus items recently delivered. Include with list a statement of progress on and delivery dates for materials or items of equipment fabricated or stored away from Project site.

9^C

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

C. Field Condition Reports: Immediately on discovery of a difference between field conditions and the Contract Documents, prepare a detailed report. Submit with a request for information. Include a detailed description of the differing conditions, together with recommendations for changing the Contract Documents.

PART 3 - EXECUTION (not used)

END OF SECTION

Reput

<



SECTION 01 33 00

SUBMITTAL PROCEDURES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 01 Specification Sections, apply to this Section

1.2 SUMMARY

- A. Section includes requirements for the administrative and procedural requirements for submitting Shop Drawings, Product Data, Samples, and other submittals.
 - 1. Process designated submittals for the Project endronically through designated email system.

1.3 DEFINITIONS

- A. Action Submittals: Written and graphic information and physical samples that require Architect's responsive action. Action s bmittals are those submittals indicated in individual Specification Sections as "action submittals."
- B. Informational Submittals: Whiter and graphic information and physical samples that do not require Architect's responsive action. Submittals may be rejected for not complying with requirements. Informational submittals are those submittals indicated in individual Specification Sections as "informational submittals."
- C. Portable Document Format (PDF): An open standard file format licensed by Adobe Systems used for representing documents in a device-independent and display resolution-independent fixed tay ut document format.
- D. Eman System: A method to transmit certain electronic submittals between the Contractor, Architect, and Owner, via email.
 - For consistency, the standard file format will be PDF. Convert paper originals and other file formats to PDF prior to submission.
 - 2. In the event of system malfunction, submittals shall be processed in accordance with the Architect's instructions, until the system malfunction has been corrected.
 - 3. For this Project, process the following submittal types through the designated email system:
 - a. Product Data.

Tetra Tech 200-15704-17001

- b. Shop Drawings.
- c. Product Schedules.
- d. Qualification Data.
- e. Certificates (Welding, Installer, Manufacturer, Product, and Material, as applicable).
- f. Test Reports (Material, Product, Preconstruction, Compatibility, and Field, applicable).
- g. Research Reports.
- h. Warranty (sample).
- i. Design Data, including calculations.
- j. Coordination Drawings.
- k. Delegated-Design Services Certifications.
- 4. For Samples, provide electronic submittal of Sample cover sheet, den fying location and actual delivery date of Samples. Deliver Samples to location (Arcs tect's office, Project site, etc.) as directed by the Architect.

1.4 SUBMITTAL ADMINISTRATIVE REQUIREMENTS

- A. Coordination: Coordinate preparation and processing of submittals with performance of construction activities.
 - 1. Coordinate each submittal with factication, purchasing, testing, delivery, other submittals, and related activities that require sequential activity.
 - 2. Where indicated, submit all summal items required for each Specification Section concurrently.
 - 3. Coordinate transmittal of different types of submittals for related parts of the Work so processing will pot be delayed because of need to review submittals concurrently for coordination.
 - a. Aremeet reserves the right to withhold action on a submittal requiring commution with other submittals until related submittals are received.
- B. Processing Time: Allow sufficient time for submittal review, including time for resubmittals. No extension of the Contract Time will be authorized because of failure to transmit submittals enough in advance of the Work to permit processing, including resubmittals.

Electronic Submittals: Identify and incorporate information in each electronic submittal file as follows:

- I. Include a cover sheet on each submittal item for identification. Do not combine different submittals under same cover sheet; only one submittal is to be provided per email.
 - a. Cover Sheet: Use PDF version of sample form included in Project Manual. Complete each item on form, sign and date. Architect will furnish PDF version of sample form.
- 2. Name submittal file as directed by Architect.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- 3. Transmit each submittal via email using subject line as directed by Architect.
- 4. Send submittal to designated Project-specific email address:
 - a. Use the following email address: <u>IER.Brandywine@tetratech.com</u>
- 5. Contractor must have a color printer and copier in order to use the electronic submitted process.
- D. Resubmittals: Make resubmittals in same form and, for non-electronic submittals in the s number of copies as initial submittal.
 - 1. Note date and content of revision in label or title block and clearly indicate extent of revision.
 - 2. Resubmit submittals until they are marked with approval notation from Architect.
 - 3. Refer to Supplementary Conditions for provisions and wing Owner to obtain reimbursement from the Contractor for amounts paid to the Architect for evaluation of certain resubmittals.
- E. Distribution: Furnish copies of final submittals to manufacturers, subcontractors, suppliers, fabricators, installers, authorities having jurisdiction, and others as necessary for performance of construction activities.
- F. Use for Construction: Retain complete topics of submittals on Project site. Use only final action submittals that are marked with approval notation from Architect.

PART 2 - PRODUCTS

- 2.1 SUBMITTAL PROCEDURES, GENERAL
 - A. General Submittal Procedure Requirements: Prepare and submit submittals required by individual Specification Sections. Types of submittals are indicated in individual Specification Sections.
- 2.2 ELECTRONIC SUBMITTAL PROCEDURES

Use the designated email system for submittals in this Article.

- Submit electronic submittals via email as PDF electronic files.
 - a. Architect will return annotated file. Annotate and retain one copy of file as an electronic Project record document file.
- 2. Contractor must have a color printer and copier in order to use the electronic submittal process.

- B. Product Data: Collect information into a single submittal for each element of construction and type of product or equipment.
 - 1. Mark submittal to show which products and options are applicable.
 - 2. Include the following information, as applicable:
 - a. Manufacturer's catalog cuts.
 - b. Statement of compliance with specified referenced standards.
 - c. Testing by recognized testing agency.
 - 3. For equipment, include the following in addition to the above, as applicate:
 - a. Printed performance curves.
 - b. Clearances required to other construction, if not indicated on accompanying Shop Drawings.
- C. Shop Drawings: Prepare Project-specific information, drawn accurately to scale. Do not base Shop Drawings on reproductions of the Contract Documents or sandard printed data, unless submittal based on Architect's digital data drawing files is otherwise permitted.
 - 1. Preparation: Fully illustrate requirements in the Contract Documents. Include the following information, as applicable:
 - a. Identification of products.
 - b. Schedules.
 - c. Compliance with specified standards.
 - d. Notation of dimensions established by field measurement.
 - e. Relationship and attaching a to adjoining construction clearly indicated.
 - f. Seal and signature of p of essional engineer if specified.
- D. Product Schedule: A creatized in individual Specification Sections, prepare a written summary indicating types of product required for the Work and their intended location. Include the following information in tabular form:
 - 1. Type of product. Include unique identifier for each product indicated in the Contract Decuments or assigned by Contractor if none is indicated.
 - 2. Manufacturer and product name, and model number if applicable.
 - 3. Vumber and name of room or space.

Qualification Data: Prepare written information that demonstrates capabilities and experience of firm or person. Include lists of completed projects with project names and addresses, contact information of architects and owners, and other information specified.

Certificates:

1. Welding Certificates: Prepare written certification that welding procedures and personnel comply with requirements in the Contract Documents. Submit record of Welding Procedure Specification and Procedure Qualification Record on AWS forms. Include names of firms and personnel certified.

- 2. Installer Certificates: Submit written statements on manufacturer's letterhead certifying that Installer complies with requirements in the Contract Documents and, where required, is authorized by manufacturer for this specific Project.
- 3. Manufacturer Certificates: Submit written statements on manufacturer's letterhead certifying that manufacturer complies with requirements in the Contract Document Include evidence of manufacturing experience where required.
- 4. Product Certificates: Submit written statements on manufacturer's letterhead ce that product complies with requirements in the Contract Documents.
- 5. Material Certificates: Submit written statements on manufacturer's letterhead certifying that material complies with requirements in the Contract Documents
- G. Test Reports:
 - 1. Material Test Reports: Submit reports written by a qualific testing agency, on testing agency's standard form, indicating and interpreting test results of material for compliance with requirements in the Contract Documents.
 - 2. Product Test Reports: Submit written reports indicating that current product produced by manufacturer complies with requirements in the Contract Documents. Base reports on evaluation of tests performed by manufacture and witnessed by a qualified testing agency, or on comprehensive tests performed by a qualified testing agency.
 - 3. Preconstruction Test Reports: Submit reports written by a qualified testing agency, on testing agency's standard form, indusating and interpreting results of tests performed before installation of product, for compliance with performance requirements in the Contract Documents.
 - 4. Compatibility Test Robots. Submit reports written by a qualified testing agency, on testing agency's structure form, indicating and interpreting results of compatibility tests performed before insullation of product. Include written recommendations for primers and substate preparation needed for adhesion.
 - 5. Field Sesa Reports: Submit written reports indicating and interpreting results of field texts performed either during installation of product or after product is installed in its final location, for compliance with requirements in the Contract Documents.
- H. Research Reports: Submit written evidence, from a model code organization acceptable to authorities having jurisdiction, that product complies with building code in effect for Project.

Warranty: Submit sample warranties as required in individual Specification Sections.

Design Data: Prepare and submit written and graphic information, including, but not limited to, performance and design criteria, list of applicable codes and regulations, and calculations. Include list of assumptions and other performance and design criteria and a summary of loads. Include load diagrams if applicable. Provide name and version of software, if any, used for calculations. Include page numbers.
- K. Coordination Drawing Submittals: Comply with requirements specified in Division 01 Section "Project Management and Coordination."
- L. Delegated-Design Services Certification: Submit certificate, signed and sealed by the responsible design professional, for each product and system specifically assigned to Contractor to be designed or certified by a design professional.
 - 1. Indicate that products and systems comply with performance and design criteria in the Contract Documents. Include list of codes, loads, and other factors used in performing these services.
 - 2. In addition, for a project in New Jersey, provide three paper copies of certificate, signed and sealed (with raised seal) by the responsible design professional.

2.3 NON-ELECTRONIC SUBMITTAL PROCEDURES

- A. Samples: Submit Samples for review of kind, color, pattern, and texture for a check of these characteristics with other elements and for a comparison of these characteristics between submittal and actual component as delivered and installed.
 - 1. Transmit Samples that contain multiple, related components such as accessories together in one submittal package.
 - 2. Identification: Attach label on unexpected ide of Samples that includes the following:
 - a. Generic description of bancple
 - b. Product name and name of manufacturer.
 - c. Sample source.
 - d. Number and title of applicable Specification Section.
 - 3. Disposition: Maintein sets of approved Samples at Project site, available for qualitycontrol comparisons throughout the course of construction activity. Sample sets may be used to determine haal acceptance of construction associated with each set.
 - 4. Samples for Initial Selection: Submit manufacturer's color charts consisting of units or sections of units showing the full range of colors, textures, and patterns available.

Number of Samples: Submit two full sets of available choices where color, pattern, texture, or similar characteristics are required to be selected from manufacturer's product line. Architect will return one submittal with options selected.

Samples for Verification: Submit full-size units or Samples of size indicated, prepared from same material to be used for the Work, cured and finished in manner specified, and physically identical with material or product proposed for use, and that show full range of color and texture variations expected. Samples include, but are not limited to, the following: partial sections of manufactured or fabricated components; small cuts or containers of materials; complete units of repetitively used materials; swatches showing color, texture, and pattern; color range sets; and components used for independent testing and inspection.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- a. Number of Samples: Submit two sets of Samples. Architect will return one set.
 - 1) If variation in color, pattern, texture, or other characteristic is inherent in material or product represented by a Sample, submit at least three sets of paired units that show approximate limits of variations.
- B. Shop Drawing Submissions:
 - 1. Submit eight (8) copies for review.
- C. Subcontract List: Prepare a written summary identifying individuals or firms proposed for each portion of the Work, including those who are to furnish products or equipment fabricated to a special design. Include the following information in tabular form:
 - 1. Name, address, and telephone number of entity performing subcontract or supplying products.
 - 2. Number and title of related Specification Section(s) evered by subcontract.
 - 3. Submit subcontract list in the following format:
 - a. Number of Copies: Four paper copies of subcontractor list, unless otherwise indicated. Architect will return one copy
- D. Key Personnel Names: No later than 15 lay after date of Notice of Award, submit a list of key personnel assignments, including superintendent and other personnel in attendance at Project site.
 - 1. Identify individuals and ther duties and responsibilities; list addresses and telephone numbers, including memory, office, and cellular telephone numbers and email addresses.
 - a. Numer of copies: Four paper copies of key personnel list, unless otherwise indicated
- E. Closeout Submittals and Maintenance Material Submittals: Comply with requirements specified a Division 01 Section "Closeout Procedures."
- F. Maintenance Data: Comply with requirements specified in Division 01 Section "Operation and Maintenance Data."

DELEGATED-DESIGN SERVICES

- Performance and Design Criteria: Where professional design services or certifications by a design professional are specifically required of Contractor by the Contract Documents, provide products and systems complying with specific performance and design criteria indicated.
- 1. If criteria indicated are not sufficient to perform services or certification required, submit a written request for additional information to Architect.

Tetra Tech 200-15704-17001

PART 3 - EXECUTION

3.1 CONTRACTOR'S REVIEW

- A. Action and Informational Submittals: Review each submittal and check for coordination with other Work of the Contract and for compliance with the Contract Documents. Note corrections and field dimensions. Identify any deviations from Contract Document requirements Mark cover sheet with approval before submitting to Architect.
 - 1. Sign and date statement certifying that submittal has been reviewed, checked, and approved for compliance with the Contract Documents.

3.2 ARCHITECT'S ACTION

- A. General: Architect will not review submittals that do not bean Covaractor's approval and will return them without action.
- B. Action Submittals: Architect will review each submitter, make marks to indicate corrections or revisions required, and return it. Architect will mark submittal appropriately to indicate action, as follows:
 - 1. Final Unrestricted Release: Where the submattal is marked "Approved," the Work covered by the submittal may proceed provided it complies with the Contract Documents. Final acceptance will depend on that compliance.
 - 2. Final-but-Restricted Release. Where the submittal is marked "Approved as Noted," the Work covered by the submittal may proceed provided it complies both with Architect's notations and corrections on the submittal and the Contract Documents. Final acceptance will depend on that compliance.
 - 3. Resubmit: Where the submittal is marked "Approved, Revise and Return Corrected Copies," the Work covered by the submittal may proceed provided it complies both with Architect's notations and corrections on the submittal and the Contract Documents. Revise submittal according to Architect's notations and corrections and return corrected copies. Final acceptance will depend on that compliance.
 - . Rejected: Where the submittal is marked "Rejected," do not proceed with the Work overed by the submittal. Prepare a new submittal for a product that complies with the Contract Documents.
 - Incomplete Resubmit: Where the submittal is marked "Incomplete, Submit Additional Information," do not proceed with the Work covered by the submittal. Prepare additional information requested, or required by the Contract Documents, that indicates compliance with requirements, and resubmit.
- C. Informational Submittals: Architect will review each submittal and will not return it, or will return it if it does not comply with requirements.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- D. Incomplete submittals are unacceptable, will be considered nonresponsive, and will be returned for resubmittal without review.
- E. Limit information submitted to specific products indicated. Do not submit extraneous matter. Submittals containing excessive extraneous matter will be returned for resubmittal without review.
- F. Submittals not required by the Contract Documents may be returned by the Architect vithout action.

Attachment: Cover Sheet

END OF SECTION

<

ADDITION AND RENOVATIONS

| CONTRACTOR: | SUBMITTAL DATE / / |
|--|---|
| | Check following as applicable: |
| ARCHITECT: Tetra Tech Inc. | |
| PROJECT IDENTIFICATION | RESERVED FOR USE BY TETRA TECH |
| Architect's Project No.: <u>200-15704-17001</u> | |
| Proj. Name: Addition & Renovations | |
| Location: Carrcroft Elementary School | |
| PRODUCT IDENTIFICATION | Approved, Revise and Return Corrected Copies |
| Specification Section No. | Rejected |
| Submittal No. | |
| Name of Product: | Incomplete, Submit Additional Informatio |
| | |
| Name of Manufacturer: | No Action Taken |
| SUBCONTRACTOR | Returned for Resubmittal |
| <u>UPPLIER</u> | Date: Reviewed only for the limited purpose of checking for conformance |
| RELATIONSHIP TO STRUCTURE Building Name | with information given and the design concept expressed in the Con- tract Documents. Review not conducted for the purpose of determin- ing the accuracy and completeness of other details such as dimension and quantities, or for substantiating instructions for installation or performance of equipment or systems, all of which remain the respo |
| (Room #) (Room me) | sibility of the Contractor as required by the Contract Documents. Review shall not constitute approval of safety precautions or of any |
| Contract Drawing No.: | construction means, methods, techniques, sequences or procedures. |
| DEVIATION FROM CONTINUE DODUMENTS | : |
| | |
| CONTRACTOR COMMENTS: | |
| | |
| RCH/FECT'S COMMENTS | |
| | |
| CONTRACTOR'S STAMP | CONTRACTOR'S CERTIFICATION |
| | I CERTIFY THAT THIS SUBMITTAL HAS BEEN REVIEWED AND APPROVED BY THE CONTRACTOR IN ACCORDANCE WITH THE GENERAL CONDITIONS. |

ВҮ _____

SECTION 01 33 01 CADD RELEASE

PART 1 - GENERAL (Not Applicable)

PART 2 - PRODUCTS (Not Applicable)

- PART 3 PART 3 EXECUTION
- 3.1 USE AND INDEMNIFICATION AGREEMENT
 - A. Instructions:
 - 1. Please be aware that Tetra Tech charges contractor(s) for electronic files (this applies to files in AutoCAD (or similar) format).
 - a. PDF's, which are simply an electronic scan of the arawings, do not require the use of the indemnification form; however we charge \$50 per PDF to cover our expenses. Tetra Tech must receive the contractor' check prior to sending PDF's.
 - 2. For AutoCAD type files, the cost is \$100 per electronic drawing, regardless of the number of drawings they are requesting. The Use and Indemnification Agreement is to be signed by the Prime Contractor. Should a subcentractor, such as a steel fabricator, ductwork detailer, desire electronic files, they would need to pursue this request through their Prime Contractor who has the contract with the Slight.
 - 3. Due to the inherent value to be company of our typical details and our other standards, we limit the drawings type, that we will release via this indemnification form to plan type drawings. Typical detail seets are not to be released in the form of an electronic AutoCAD drawing file.
 - 4. In addition, or internal individual Base Plans will not be released; we limit what the contractor can parchase to the actual individual contract drawings.

after the Prime Contractor has determined the number of drawings that they will need, fill out the following two pages. The second page of the form, marked Use and Indemnification Agreement – Business Office, needs to be sent to the Business Office with the Contractor check made out to Tetra Tech. We will not release electronic files until we receive this form and the check.

END OF SECTION

5.



Use and Indemnification Agreement

240 Continental Drive, Suite 200 Newark, Delaware 19713 Tel. (302) 738-7551 Fax (302) 454-5980

Re: Carrcroft Elementary School – Addition and Renovation

Tt Project No. 200-15704-17001

Whereas, _______ (hereinafter the "Contractor"), acknowledges that it has requested certain electronic files and/or media of the Drawings and/or Specifications for the above-referenced Project which are the property of Tetra Tech Engineers, Architects & Landscape Architects, P.C. d/b/a Tetra Tech Architects & Engineers (hereinafter "Tetra Tech").

Whereas, Contractor further acknowledges all requests for electronic files require a pre-payment of \$100/file (Each individual bawin in the set of Contract Documents represents 1 file), regardless of the number of files requested, prior to receiving said files from retra Tech

Now, therefore, Contractor hereby warrants and covenants that it will abide by the following provisions:

A. Indemnification

- 1. In consideration of permission to use electronic files or media, including but not limited to electronic files or dawings created by use of computer, for the Work of this Project only, and which the Contractor has requested from Tetra Tech, the Contractor to be fullest extent permitted by law, hereby agrees to indemnify and hold harmless Tetra Tech, its agents, employees, officers, director undecare hants from and against any and all claims, damages, losses and expenses, including any attorneys' fees, arising out of, resulting from or meconnection with any and all use of said electronic materials, but only if such claim, damage, loss or expense is caused in whole or in part by the contractor, its employees, agents, officers, directors, or any other party directly or indirectly employed by any of them or any party for worse as any of them may be liable, regardless of whether or not it is caused by a party indemnified hereunder. Such obligation shall not be consume to reduce or negate any other right or obligation of indemnification that would otherwise exist as to any party hereto. This is demnification shall not apply to the liability of the indemnifiee arising out of its own negligence. This indemnification shall not be limited in any way because of any limitation on damages, compensation or benefits under any statute, law or governmental requirements any sort.
- 2. The following shall be included within the definition of "expenses" herein: (a) any time expended by the indemnified party of its employees, agents, officers and directors at their usual and customary billing rates, as well as all out-of-pocket expenses such as long-distance telephone calls, costs of reproduction, expenses of travel and lodging; (b) all costs as row, these of experts, consultants, engineers, and any other party retained by the indemnified party reasonably required to defend the claim; (c) all costs, including reasonable attorneys' fees, incurred in bringing any action to enforce the provisions of this indemnification. The following shall be included within the definition of "action" herein: any case brought in any state or federal court, any arbitration, any mediation, and my similar forum for resolution of any dispute herein, and shall also include any counterclaim or third-party action in any such forum.

B. <u>Use and Compatibility</u>

- 1. Tetra Tech' instruments of service are furnished we hour guarantee of compatibility with the Contractor's software or hardware, and Tetra Tech' sole responsibility for the electronic media is a furnish a replacement for defective disks within thirty (30) days after delivery to Contractor.
- 2. Because data stored on electronic media can exeriorate undetected or be modified without Tetra Tech' knowledge, the Contractor agrees that Tetra Tech will not be held liable for the completeness or correctness of the electronic media after an acceptance period of thirty (30) days after delivery of the electronic files. Tetra Tech does chafirm the accuracy of the final sealed hard copy drawings, previously submitted pursuant to the Prime Agreement for this Project
- 3. The electronic files are submitted to the Contractor for a thirty (30) day acceptance period. During this period, the Contractor may review and examine these files, and any more detected during this time will be corrected by Tetra Tech. Any changes requested after the acceptance period will be considered period after the performed on a time and materials basis, at Tetra Tech's standard cost plus terms and conditions.
- 4. Tetra Tetra Tetra tetra sownership of the printed hard copy Drawings and Specifications and the electronic media. The Contractor is granted a license for their userbut only in the operation and maintenance of the Project. Use of these materials for modification, extension, or expansion of this Project on any other project, unless under the direction of Tetra Tech, shall be without liability to Tetra Tech and Tetra Tech's consultants.

| INWITNESS W | WHEREOF: |
|---------------|----------|
| Signed name: | |
| Printed Name: | |
| Title: | |
| Date: | |
| | |



240 Continental Drive, Suite 200 Newark, Delaware 19713 Tel. (302) 738-7551 Fax (302) 454-5980





SECTION 01 50 00

TEMPORARY FACILITIES AND CONTROLS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 01 Specification Sections, may apply to this Section.

1.2 SUMMARY

A. Section includes requirements for temporary utilities, support facilities, and security and protection facilities.

1.3 USE CHARGES

- A. General: Installation and removal of and use charges for temporary facilities shall be included in the Contract Sum unless otherwise indicated. Allow other entities to use temporary services and facilities without cost, including, but not limited to, Owner's construction forces, Architect, Engineer, testing agencies, and authorities having jurisdiction.
- B. Sewer Service: Owner will not provide sewer-service for the project. Contractor shall provide all required temporary sanitary facilities.
- C. Water Service: Owner will provide on-site access to water service used by all entities for construction operation.
- D. Electric Power Service: Owner will provide on-site access to electric-power-service used by all entities for construction operations. Contractor shall furnish all necessary equipment to facilitate connection to power including coordination of permits with any inspection agency and or installation.
- E. Upon completion of the project and prior to demobilization the Contractor shall at his or her sole expense, remove any temporary utility service equipment and restore the service location to as pre-construction condition.



INFORMATIONAL SUBMITTALS

- A. Site Plan: Show temporary facilities, utility hookups, staging areas, and parking areas for construction personnel.
- B. Dust-Control Plan: Submit coordination drawing and narrative that indicates the dust-control measures proposed for use, proposed locations, and proposed time frame for their operation.

Identify further options if proposed measures are later determined to be inadequate. Include the following:

- 1. Locations of dust-control partitions at each phase of work.
- 2. Waste handling procedures.
- 3. Other dust-control measures.

1.5 QUALITY ASSURANCE

- A. Electric Service: Comply with NECA, NEMA, and UL standards and regulations for temporary electric service. Install service to comply with NFPA 70.
- B. Tests and Inspections: Arrange for authorities having jurisdiction to test and inspect each temporary utility before use. Obtain required certifications and permits
- C. Accessible Temporary Egress: Comply with applicable provisions in ADA-ABA Accessibility Guidelines.

1.6 PROJECT CONDITIONS

A. Temporary Use of Permanent Facilities: Engage Instater of each permanent service to assume responsibility for operation, maintenance, and platection of each permanent service during its use as a construction facility before Overer's acceptance, regardless of previously assigned responsibilities.

PART 2 - PRODUCTS

- 2.1 TEMPORARY FACILITIES
 - A. Field Offices, General: Shall be set-up in the building.
 - B. Keep office clear and orderly. Furnish and equip offices as follows:
 - 1. Funiture required for Project-site documents including file cabinets, plan tables, plan racks, and bookcases.
 - Conference room of sufficient size to accommodate meetings of 6 individuals. Provide electrical power service and 120-V ac duplex receptacles, with no fewer than one receptacle on each wall. Furnish room with conference table, chairs, and 4-foot-square tack and marker boards.
 - Drinking water and private toilet.
 - . Coffee machine and supplies.
 - 5. Heating and cooling equipment necessary to maintain a uniform indoor temperature of 68 to 72 deg F.
 - 6. Lighting fixtures capable of maintaining average illumination of 20 fc at desk height.

- C. Storage or Fabrication Sheds: Provide sheds sized, furnished, and equipped or fenced around to accommodate materials and equipment for construction operations as needed for secure area for equipment.
 - 1. Store combustible materials apart from building.

2.2 EQUIPMENT

- A. Fire Extinguishers: Portable, UL rated; with class and extinguishing agent as required locations and classes of fire exposures.
- B. HVAC Equipment: Unless Owner authorizes use of permanent HVAC system, provide vented, self-contained, liquid-propane-gas or fuel-oil heaters with individual space thermostatic control.
 - 1. Use of gasoline-burning space heaters, open-flame heaters or alan ander-type heating units is prohibited.
 - 2. Heating Units: Listed and labeled for type of fuel being comuned, by a qualified testing agency acceptable to authorities having jurisdiction and marked for intended location and application.
- C. Obtain permission from Owner to us existing Having and Cooling equipment. Change filters monthly. Provide a set of clean filters when done and type (2) sets of spare filters.

PART 3 - EXECUTION

3.1 INSTALLATION, GENERAL

- A. Locate facilities where they well serve Project adequately and result in minimum interference with performance of the Work.
 - 1. Locate acil ies to limit site disturbance as specified in Section 01 10 00 "Summary."
 - 2. Area Avilable: Designated area as shown on Sheet CC-01.
- B. Provide each facility ready for use when needed to avoid delay. Do not remove until facilities are no longer needed and remove prior to demobilization.

TEMPORARY UTILITY INSTALLATION

General: Install temporary service or connect to existing service.

- 1. Arrange with utility company, Owner, and existing users for time when service can be interrupted, if necessary, to make connections for temporary services.
- B. Water Service: Connect to Owner's existing water service facilities. Clean and maintain water service facilities in a condition acceptable to Owner. At Substantial Completion, restore these facilities to condition existing before initial use.

- C. Sanitary Facilities: Provide temporary toilets, wash facilities, and drinking water for use of construction personnel. Comply with requirements of authorities having jurisdiction for type, number, location, operation, and maintenance of fixtures and facilities.
- D. Isolation of Work Areas in Occupied Facilities: Prevent dust, fumes, and odors from entering occupied areas.
 - 1. Prior to commencing work if necessary, isolate the HVAC system in area where york is to be performed according to coordination drawings.
 - a. Disconnect supply and return ductwork in work area from HVAc systems servicing occupied areas.
 - b. Maintain negative air pressure within work area using HEA-eolipped airfiltration units, starting with commencement of temporary partition construction, and continuing until removal of temporary partitions is complete.
 - 2. Maintain dust partitions during the Work if needed. Use vacuum collection attachments on dust-producing equipment. Isolate limited work within occupied areas using portable dust-containment devices.
 - 3. Perform daily construction cleanup and final cleanup using approved, HEPA-filterequipped vacuum equipment.
- E. Electric Power Service: Provide electric power service and distribution system of sufficient size, capacity, and power characteristics required for construction operations.
 - 1. Connect temporary service to Owner's existing power source, as directed by Owner.
- F. Lighting: Provide temporary lighting with local switching that provides adequate illumination for construction operations, observations, inspections, and traffic conditions.
 - 1. Install and operate temporary lighting that fulfills security and protection requirements without operating entire system.
- G. Telephone Service Provide temporary telephone service in common-use facilities for use by all construction personnel. Install a minimum of one telephone line for each field office.
 - Provide additional telephone lines for the following:

Provide a dedicated telephone line for each facsimile machine in each field office.

At each telephone, post a list of important telephone numbers.

- a. Police and fire departments.
- b. Ambulance service.

1.

- c. Contractor's home office.
- d. Contractor's emergency after-hours telephone number.
- e. Architect's office.
- f. Engineers' offices.
- g. Owner's office.
- h. Principal subcontractors' field and home offices.

- 3. Provide superintendent with cellular telephone or portable two-way radio for use when away from field office.
- H. Electronic Communication Service: Provide a desktop computer in the primary field office adequate for use by Architect and Owner to access Project electronic documents and maintain electronic communications. Equip computer with not less than the following:
 - 1. Processor: Intel Pentium D or Intel CoreDuo, 3.0 GHz processing speed.
 - 2. Memory: 4 gigabyte.
 - 3. Disk Storage: 300 gigabyte hard-disk drive and combination DVD-RW/CD-FW
 - 4. Display: 22-inch LCD monitor with 256-Mb dedicated video RAM.
 - 5. Full-size keyboard and mouse.
 - 6. Network Connectivity: 10/100BaseT Ethernet.
 - 7. Operating System: Microsoft Windows XP Professional or Microsoft Windows Vista Business.
 - 8. Productivity Software:
 - a. Microsoft Office Professional, XP or higher, including Word, Excel, and Outlook.
 - b. Adobe Reader 7.0 or higher.
 - c. WinZip 7.0 or higher.
 - 9. Printer: "All-in-one" unit equipped with printer server, combining color printing, photocopying, scanning, and faxing, or separate units for each of these three functions.
 - 10. Internet Service: Broadband modem, route and ISP, equipped with hardware firewall, providing minimum 384 Kbps uplow and Mbps download speeds at each computer.
 - 11. Internet Security: Integrated software, providing software firewall, virus, spyware, phishing, and spam protection in a combined application.
 - 12. Backup: External hard drive minimum 40 gigabyte, with automated backup software providing daily backups.

3.3 SUPPORT FACILITIES INSTALLATION

- A. Parking: Use designated areas of Owner's existing parking areas for construction personnel as directed.
- B. Project Signs: Irovide Project signs as need and approved by the DEARNG. Unapproved signs are not permitted.



- Identification Signs: Provide Project identification signs as indicated on Drawings. Temporary Signs: Provide other signs as indicated and as required to inform public and individuals seeking entrance to Project or directions to construction field office.
- a. Provide temporary, directional signs for construction personnel and visitors.
- 3. Maintain and touchup signs so they are legible at all times.
- C. Waste Disposal Facilities: Comply with requirements specified in Section 01 74 19 "Construction Waste Management and Disposal."

- D. Waste Disposal Facilities: Provide waste-collection containers in sizes adequate to handle waste from construction operations. Comply with requirements of authorities having jurisdiction. Comply with progress cleaning requirements in Section 01 73 00 "Execution."
- E. Lifts and Hoists: Provide facilities necessary for hoisting materials and personnel.
 - 1. Truck cranes and similar devices used for hoisting materials are considered "tools and equipment" and not temporary facilities.
- F. Temporary Stairs: Until permanent stairs are available, provide temporary stairs where lac are not adequate.
- G. Existing Stair Usage: Use of Owner's existing stairs will be permitted, provider stairs are cleaned and maintained in a condition acceptable to Owner. At Substantial Compretion, restore stairs to condition existing before initial use.
 - 1. Provide protective coverings, barriers, devices, signs, or other procedures to protect stairs and to maintain means of egress. If stairs become damaged, restore damaged areas so no evidence remains of correction work.
- H. Temporary Use of Permanent Stairs: Use of new stairs for construction traffic will be permitted, provided stairs are protected and finishes restored to new condition at time of Substantial Completion.

3.4 SECURITY AND PROTECTION FACE INSTALLATION

- A. Protection of Existing Facilities: Protect existing vegetation, equipment, structures, utilities, and other improvements at Project size and on adjacent properties, except those indicated to be removed or altered. Repair damage to existing facilities.
- B. Environmental Protectio: Provide protection, operate temporary facilities, and conduct construction as require to comply with environmental regulations and that minimize possible air, waterway, and subsolucontamination or pollution or other undesirable effects.
 - 1. Comply with work restrictions specified in Section 01 10 00 "Summary."
- C. Temporary Erosion and Sedimentation Control: Comply with requirements of 2003 EPA Construction General Permit by following applicable requirements as stipulated on the Erosion and Sediment Plan Sheets CE-01 to CE-04 approved by DNREC, including CCR if required by DNREC, or authorities having jurisdiction, whichever is more stringent and requirements specified in Section 31 10 00 "Site Clearing."

Temporary Erosion and Sedimentation Control: Provide measures to prevent soil erosion and discharge of soil-bearing water runoff and airborne dust to undisturbed areas and to adjacent properties and walkways, according to Erosion and Sediment Plan Sheets CE-01 to CE-04 approved by DNREC and requirements of 2003 EPA Construction General Permit or authorities having jurisdiction, whichever is more stringent.

1. Verify that flows of water redirected from construction areas or generated by construction activity do not enter or cross tree- or plant- protection zones.

- 2. Inspect, repair, and maintain erosion- and sedimentation-control measures during construction until permanent vegetation has been established.
- 3. Clean, repair, and restore adjoining properties and roads affected by erosion and sedimentation from Project site during the course of Project.
- 4. Remove erosion and sedimentation controls and restore and stabilize areas disturbed during removal.
- E. Stormwater Control: Comply with requirements of authorities having jurisdiction including DNREC and CCR, if required by DNREC. Provide barriers in and around excavation and subgrade construction to prevent flooding by runoff of stormwater from heavy rains
- F. Pest Control: Engage pest-control service to recommend practices to minimize avraction and harboring of rodents, roaches, and other pests and to perform exterminition and control procedures at regular intervals so Project will be free of pests and their resulues at Substantial Completion. Perform control operations lawfully, using environmentally size materials.
- G. Site Enclosure Fence: Before construction operations begin, forms, and install site enclosure fence in a manner that will prevent people and animals from eavily entering site except by existing entrance gates at south entrance.
 - 1. Extent of Fence: As required to enclose entire Project site or portion determined sufficient to accommodate construction operations.
 - 2. Maintain security of construction area by finiting number of keys and restricting distribution to authorized personnel. Fundshere set of keys to Owner.
- H. Security Enclosure and Lockup: Install temperary enclosure around partially completed areas of construction. Provide lockable entrances to prevent unauthorized entrance, vandalism, theft, and similar violations of security. Lock entrances at end of each work day.
- I. Barricades, Warning Signs, and Lights: Comply with requirements of authorities having jurisdiction for erecting structurally adequate barricades, including warning signs and lighting.
- J. Temporary Egress: Man tair temporary egress from existing occupied facilities as indicated and as required by amprities having jurisdiction.
- K. Temporary Exclosures: Provide temporary enclosures for protection of construction, in progress and completed from exposure, foul weather, other construction operations, and similar activities. Provide temporary weathertight enclosure for building exterior.
 - Where heating or cooling is needed and permanent enclosure is incomplete, insulate temporary enclosures.

Temporary Partitions: If necessary provide floor-to-ceiling dustproof partitions to limit dust and dirt migration and to separate areas occupied by Owner and tenants from fumes and noise.

- 1. Construct dustproof partitions with gypsum wallboard with joints taped on occupied side, and fire-retardant-treated plywood on construction operations side.
- 2. Construct dustproof partitions with two layers of 6-mil polyethylene sheet on each side. Cover floor with two layers of 6-mil polyethylene sheet, extending sheets 18 inches up the sidewalls. Overlap and tape full length of joints. Cover floor with fire-retardanttreated plywood.

- a. Construct vestibule and airlock at each entrance through temporary partition with not less than 48 inches between doors. Maintain water-dampened foot mats in vestibule.
- 3. Where fire-resistance-rated temporary partitions are indicated or are required by authorities having jurisdiction, construct partitions according to the rated assemblies.
- 4. Insulate partitions to control noise transmission to occupied areas.
- 5. Seal joints and perimeter. Equip partitions with gasketed dustproof doors and security locks where openings are required.
- 6. Protect air-handling equipment.
- 7. Provide walk-off mats at each entrance through temporary partition.
- M. Temporary Fire Protection: If necessary install and maintain temporary fire-protection facilities of types needed to protect against reasonably predictable and controlleale the tosses. Comply with NFPA 241; manage fire-prevention program.
 - 1. Prohibit smoking in construction areas.
 - 2. Supervise welding operations, combustion-type temporary heating units, and similar sources of fire ignition according to requirements of cuthorities having jurisdiction.
 - 3. Develop and supervise an overall fire-prevention and -protection program for personnel at Project site. Review needs with local fire department and establish procedures to be followed. Instruct personnel in methods and procedures. Post warnings and information.
 - followed. Instruct personnel in methods and procedures. Post warnings and information.
 Provide temporary standpipes and hoses for interprotection. Hang hoses with a warning sign stating that hoses are for fire-protection purposes only and are not to be removed. Match hose size with outlet size and equip with suitable nozzles.

3.5 MOISTURE AND MOLD CONTROL

- A. Contractor's Moisture-Protection Itan: Avoid trapping water in finished work. Document visible signs of mold that may appear during construction.
- B. Items C, D, and E below apply only to existing facilities to remain.
- C. Exposed Construction Phase: Before installation of weather barriers, when materials are subject to wetting anti-exposure and to airborne mold spores, protect as follows:
 - 1. Project orous materials from water damage.
 - 2. Protect stored and installed material from flowing or standing water.
 - Yeep porous and organic materials from coming into prolonged contact with concrete.
 - Remove standing water from decks.
 - Keep deck openings covered or dammed.

- Partially Enclosed Construction Phase: After installation of weather barriers but before full enclosure and conditioning of building, when installed materials are still subject to infiltration of moisture and ambient mold spores, protect as follows:
- 1. Do not load or install drywall or other porous materials or components, or items with high organic content, into partially enclosed building.
- 2. Keep interior spaces reasonably clean and protected from water damage.
- 3. Periodically collect and remove waste containing cellulose or other organic matter.

- 4. Discard or replace water-damaged material.
- 5. Do not install material that is wet.
- 6. Discard, replace, or clean stored or installed material that begins to grow mold.
- 7. Perform work in a sequence that allows any wet materials adequate time to dry before enclosing the material in drywall or other interior finishes.
- E. Controlled Construction Phase of Construction: After completing and sealing of the building enclosure but prior to the full operation of permanent HVAC systems, maintain as follow.
 - 1. Control moisture and humidity inside building by maintaining effective dry-i condi
 - 2. Use permanent HVAC system to control humidity.
 - 3. Comply with manufacturer's written instructions for temperature, relative humany, and exposure to water limits.
 - a. Hygroscopic materials that may support mold growth, including wood and gypsum-based products, that become wet during the course of construction and remain wet for 48 hours are considered defective.
 - b. Measure moisture content of materials that have been exposed to moisture during construction operations or after installation. Record readings beginning at time of exposure and continuing daily for 48 hours. Identify materials containing moisture levels higher than allowed. Report findings in writing to Architect.
 - c. Remove materials that cannot be completely restored to their manufactured moisture level within 48 hours.

3.6 OPERATION, TERMINATION, AND NEMOVAL

- A. Supervision: Enforce strict discipline in use of temporary facilities. To minimize waste and abuse, limit availability of temporary collities to essential and intended uses.
- B. Maintenance: Maintain facilities is good operating condition until removal.
 - 1. Maintain operation of temporary enclosures, heating, cooling, humidity control, ventilation, and similar facilities on a 24-hour basis where required to achieve indicated results and composibility of damage.
- C. Temporary Facility Changeover: Change over from using temporary security and protection facilities to permanent facilities must occur before Demobilization and prior to Substantial Completion.



Termination and Removal: Remove each temporary facility when need for its service has ended, when it has been replaced by authorized use of a permanent facility, or no later than Substantial Completion. Complete or, if necessary, restore permanent construction that may have been delayed because of interference with temporary facility. Repair damaged Work, clean exposed surfaces, and replace construction that cannot be satisfactorily repaired.

- 1. Materials and facilities that constitute temporary facilities are property of Contractor. Owner reserves right to take possession of Project identification signs.
- 2. Remove temporary roads and paved areas not intended for or acceptable for integration into permanent construction. Where area is intended for landscape development, remove soil and aggregate fill that do not comply with requirements for fill or subsoil. Remove

materials contaminated with road oil, asphalt and other petrochemical compounds, and other substances that might impair growth of plant materials or lawns. Repair or replace street paving, curbs, and sidewalks at temporary entrances, as required by authorities having jurisdiction.

3. At Substantial Completion, repair, renovate, and clean permanent facilities used during construction period. Comply with final cleaning requirements specified in Section 01 00 "Closeout Procedures."

END OF SECTION

SECTION 01 73 29

CUTTING & PATCHING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementar Conditions and other Division 01 Specification Sections, apply to this Section

1.2 SUMMARY

- A. This Section includes procedural requirements for cutting and patch
- B. Related Sections include the following:
 - 1. Divisions 03 through 09 Sections for specific requirements and limitations applicable to cutting and patching individual parts of the Work.

1.3 DEFINITIONS

- A. Cutting: Removal of in-place construction necessary to permit installation or performance of other Work.
- B. Patching: Fitting and repar work required to restore surfaces to original conditions after installation of other Work.

1.4 QUALITY ASSURANCE

- A. Structural Elements: Do not cut and patch structural elements in a manner that could change their load-carrying capacity or load-deflection ratio.
- B. Operational Elements: Do not cut and patch operating elements and related components in a manner that results in reducing their capacity to perform as intended or that result in increased maintenance or decreased operational life or safety. Operating elements include the following:
 - Fire-suppression systems.
 - Mechanical systems piping and ducts.
 - 3. Control systems.
 - 4. Communication systems.
 - 5. Conveying systems.
 - 6. Electrical wiring systems.

2.

- C. Visual Requirements: Do not cut and patch construction in a manner that results in visual evidence of cutting and patching. Do not cut and patch construction exposed on the exterior or in occupied spaces in a manner that would, in Architect's opinion, reduce the building's aesthetic qualities. Remove and replace construction that has been cut and patched in a visually unsatisfactory manner.
- D. Cutting and Patching Conference: Before proceeding, meet at Project site with parties involved in cutting and patching, including mechanical and electrical trades. Review areas of potential interference and conflict. Coordinate procedures and resolve potential conflicts before proceeding.

PART 2 - PRODUCTS

2.1 MATERIALS

- A. General: Comply with requirements specified in other Sections.
- B. In-Place Materials: Use materials identical to in-place materials. For exposed surfaces, use materials that visually match in-place adjacent surfaces to the fullest extent possible.
 - 1. If identical materials are unavailable or cannot occused, use materials that, when installed, will match the visual and functional performance of in-place materials.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine surfaces to be cut and patched and conditions under which cutting and patching are to be performed.
 - 1. Comparbility: Before patching, verify compatibility with and suitability of substrates, including compatibility with in-place finishes or primers.
 - Proceed with installation only after unsafe or unsatisfactory conditions have been conjected.

2 PREPARATION

2.

Temporary Support: Provide temporary support of Work to be cut.

Protection: Protect in-place construction during cutting and patching to prevent damage. Provide protection from adverse weather conditions for portions of Project that might be exposed during cutting and patching operations.

C. Adjoining Areas: Avoid interference with use of adjoining areas or interruption of free passage to adjoining areas.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

D. Existing Utility Services and Mechanical/Electrical Systems: Where existing services/systems are required to be removed, relocated, or abandoned, bypass such services/systems before cutting to prevent interruption to occupied areas.

3.3 PERFORMANCE

- A. General: Employ skilled workers to perform cutting and patching. Proceed with cutting and patching at the earliest feasible time, and complete without delay.
 - 1. Cut in-place construction to provide for installation of other components or performance of other construction, and subsequently patch as required to restore surfaces to their original condition.
- B. Cutting: Cut in-place construction by sawing, drilling, breaking, chipping grinding, and similar operations, including excavation, using methods least likely to damage elements retained or adjoining construction. If possible, review proposed procedures while original Installer; comply with original Installer's written recommendations.
 - 1. In general, use hand or small power tools designed for sawing and grinding, not hammering and chopping. Cut holes and slots is small as possible, neatly to size required, and with minimum disturbance of adjacent surfaces. Temporarily cover openings when not in use.
 - 2. Finished Surfaces: Cut or drill from the expected or finished side into concealed surfaces.
 - 3. Concrete or Masonry: Cut using a multiple machine, such as an abrasive saw or a diamond-core drill.
 - 4. Mechanical and Electrical Services: Cut off pipe or conduit in walls or partitions to be removed. Cap, valve, or pipe and seal remaining portion of pipe or conduit to prevent entrance of moisture or other foreign matter after cutting.
 - 5. Proceed with patching after construction operations requiring cutting are complete.
- C. Patching: Patch constructed by filling, repairing, refinishing, closing up, and similar operations following performance of other Work. Patch with durable seams that are as invisible as possible. Provide materials and comply with installation requirements specified in other Sections.
 - 1. Inspection: Where feasible, test and inspect patched areas after completion to demonstrate integrity of installation.
 - exposed Finishes: Restore exposed finishes of patched areas and extend finish astoration into retained adjoining construction in a manner that will eliminate evidence of patching and refinishing.
 - a. Clean piping, conduit, and similar features before applying paint or other finishing materials.
 - b. Restore damaged pipe covering to its original condition.
 - 3. Floors and Walls: Where walls or partitions that are removed extend one finished area into another, patch and repair floor and wall surfaces in the new space. Provide an even surface of uniform finish, color, texture, and appearance. Remove in-place floor and wall coverings and replace with new materials, if necessary, to achieve uniform color and appearance.

- a. Where patching occurs in a painted surface, apply primer and intermediate paint coats over the patch and apply final paint coat over entire unbroken surface containing the patch. Provide additional coats until patch blends with adjacent surfaces.
- 4. Ceilings: Patch, repair, or rehang in-place ceilings as necessary to provide an even-plane surface of uniform appearance.
- 5. Exterior Building Enclosure: Patch components in a manner that restores enclosure to a weathertight condition.
- D. Cleaning: Clean areas and spaces where cutting and patching are performed. Completely remove paint, mortar, oils, putty, and similar materials.

END OF SECTION

SECTION 01 77 00

CLOSEOUT PROCEDURES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Conditions and other Division 01 Specification Sections, apply to this Section

1.2 SUMMARY

A. Section includes administrative and procedural requirements for contract closeout, including, but not limited to, the following:

- 1. Substantial Completion procedures.
- 2. Final completion procedures.
- 3. Warranties.
- 4. Final cleaning.
- 5. Repair of the Work.
- B. Related Requirements:
 1. Section 01 73 00 "Execution" for progress cleaning of Project site.
- C. Delaware Department of Transportation (DelDOT) Standard Specifications.
- D. Delaware Department of Natural Resources and Environmental Control (DNREC) Erosion and Sediment Control Han book
- 1.3 ACTION SUB ATTALS
 - A. Product Data: For cleaning agents.
 - B. Conductor's List of Incomplete Items: Initial submittal at Substantial Completion.

Certified List of Incomplete Items: Final submittal at Final Completion.

CLOSEOUT SUBMITTALS

- A. Certificates of Release: From authorities having jurisdiction.
- B. Certificate of Insurance: For continuing coverage.
- C. Field Report: For pest control inspection.

ADDITION AND RENOVATIONS

1.5 MAINTENANCE MATERIAL SUBMITTALS

A. Schedule of Maintenance Material Items: For maintenance material submittal items specified in other Sections.

1.6 SUBSTANTIAL COMPLETION PROCEDURES

- A. Contractor's List of Incomplete Items: Prepare and submit a list of items to be completed and corrected (Contractor's punch list), indicating the value of each item on the list and easons vity the Work is incomplete.
- B. Submittals Prior to Substantial Completion: Complete the following a minimum of 10 days prior to requesting inspection for determining date of Substantial Completion. List items below that are incomplete at time of request.
 - 1. Certificates of Release: Obtain and submit releases from autorities having jurisdiction permitting Owner unrestricted use of the Work and access to services and utilities. Include occupancy permits, operating certificates, and similar releases.
 - 2. Submit closeout submittals specified in other division of Sections, including project record documents, operation and maintenance maguals, final completion construction photographic documentation, damage or settlement surveys, property surveys, and similar final record information.
 - 3. Submit closeout submittals specified in and sideal Sections, including specific warranties, workmanship bonds, maintenance couries agreements, final certifications, and similar documents.
 - 4. Submit maintenance material submit als specified in individual Sections, including tools, spare parts, extra materials, and similar items, and deliver to location designated by Architect. Label with panua curver's name and model number where applicable.
 - a. Schedule of Maintanance Material Items: Prepare and submit schedule of maintenance material submittal items, including name and quantity of each item and name and number of related Specification Section. Obtain Architect's signature for releipt of submittals.
 - 5. Submittes (adjust/balance records.
 - 6. Submit sustainable design submittals not previously submitted.
 - 7. Subsit changeover information related to Owner's occupancy, use, operation, and maintenance.

Procedures Prior to Substantial Completion: Complete the following a minimum of 10 days prior to requesting inspection for determining date of Substantial Completion. List items below that are incomplete at time of request.

- 1. Advise Owner of pending insurance changeover requirements.
- 2. Make final changeover of permanent locks and deliver keys to Owner. Advise Owner's personnel of changeover in security provisions.
- 3. Complete startup and testing of systems and equipment.
- 4. Perform preventive maintenance on equipment used prior to Substantial Completion.
- 5. Instruct Owner's personnel in operation, adjustment, and maintenance of products, equipment, and systems.

CLOSEOUT PROCEDURES 01 77 00-2

Tetra Tech 200-15704-17001

- 6. Advise Owner of changeover in heat and other utilities.
- 7. Participate with Owner in conducting inspection and walkthrough with local emergency responders.
- 8. Terminate and remove temporary facilities from Project site, along with mockups, construction tools, and similar elements.
- 9. Complete final cleaning requirements, including touchup painting.
- 10. Touch up and otherwise repair and restore marred exposed finishes to eliminate visual defects.
- D. Inspection: Submit a written request for inspection to determine Substantial Completion a minimum of 10 days prior to date the work will be completed and ready for final inspection and tests. On receipt of request, Architect will either proceed with inspection or notify contractor of unfulfilled requirements. Architect will prepare the Certificate of Substantial Completion after inspection or will notify Contractor of items, either on Contractor's last or additional items identified by Architect, that must be completed or corrected before certificate will be issued.
 - 1. Reinspection: Request reinspection when the Work identified in previous inspections as incomplete is completed or corrected.
 - 2. Results of completed inspection will form the basis of requirements for final completion.

1.7 FINAL COMPLETION PROCEDURES

- A. Submittals Prior to Final Completion: Before requesting final inspection for determining final completion, complete the following:
 - 1. Certified List of Incomplete Iters: Submit certified copy of Architect's Substantial Completion inspection list of reads to be completed or corrected (punch list), endorsed and dated by Architect Certified copy of the list shall state that each item has been completed or otherwise resorved for acceptance.
 - 2. Certificate of ansurance Submit evidence of final, continuing insurance coverage complying with a surance requirements.
 - 3. Submit postcontrol final inspection report.
- B. Inspection subhit a written request for final inspection to determine acceptance a minimum of 10 days prior to date the work will be completed and ready for final inspection and tests. On receipt on request, Architect will either proceed with inspection or notify Contractor of unfolfined requirements. Architect will prepare a final Certificate for Payment after inspection or will notify Contractor of construction that must be completed or corrected before certificate will be issued.
 - Reinspection: Request reinspection when the Work identified in previous inspections as incomplete is completed or corrected.

LIST OF INCOMPLETE ITEMS (PUNCH LIST)

A. Organization of List: Include name and identification of each space and area affected by construction operations for incomplete items and items needing correction including, if necessary, areas disturbed by Contractor that are outside the limits of construction.

1.

ADDITION AND RENOVATIONS

- 1. Organize list of spaces in sequential order.
- 2. Include the following information at the top of each page:
 - a. Project name.
 - b. Date.
 - c. Name of Architect.
 - d. Name of Contractor.
 - e. Page number.
- 3. Submit list of incomplete items in the following format:
 - a. MS Excel electronic file. Architect will return annotated file.
 - b. PDF electronic file. Architect will return annotated file.
 - c. Three paper copies. Architect will return two copies.

1.9 SUBMITTAL OF PROJECT WARRANTIES

- A. Time of Submittal: Submit written warranties on request of Architect for designated portions of the Work where commencement of warranties other man date of Substantial Completion is indicated, or when delay in submittal of warranties might unit Owner's rights under warranty.
- B. Partial Occupancy: Submit properly executed was arties within 15 days of completion of designated portions of the Work that are completed and occupied or used by Owner during construction period by separate agreement with contractor.
- C. Organize warranty documents into an order v sequence based on the table of contents of Project Manual.
 - 1. Bind warranties and bonds it heavy-duty, three-ring, vinyl-covered, loose-leaf binders, thickness as necessary to accommodate contents, and sized to receive 8-1/2-by-11-inch paper.
 - 2. Provide heavy paper dividers with plastic-covered tabs for each separate warranty. Mark tab to identify the product or installation. Provide a typed description of the product or installation including the name of the product and the name, address, and telephone number or installer.
 - 3. Identify each binder on the front and spine with the typed or printed title "W. PDANTIES," Project name, and name of Contractor.
 - Warranty Electronic File: Scan warranties and bonds and assemble complete warranty and bond submittal package into a single indexed electronic PDF file with links enabling navigation to each item. Provide bookmarked table of contents at beginning of document.

Provide additional copies of each warranty to include in operation and maintenance manuals.

PART 2 - PRODUCTS

PART 3 - EXECUTION

3.1 FINAL CLEANING

- A. General: Perform final cleaning. Conduct cleaning and waste-removal operations to comply with local laws and ordinances and Federal and local environmental and ampoliation regulations.
- B. Cleaning: Employ experienced workers or professional cleaners for final cleaning. Clean each surface or unit to condition expected in an average commercial functing meaning and maintenance program. Comply with manufacturer's written instructions.
 - 1. Complete the following cleaning operations before requesting aspection for certification of Substantial Completion for entire Project or for a designated portion of Project:
 - a. Clean Project site, yard, and grounds, in areas distuibed by construction activities, including landscape development areas, of thosish, waste material, litter, and other foreign substances.
 - b. Sweep paved areas broom clean. Lemmy petrochemical spills, stains, and other foreign deposits.
 - c. Rake grounds that are neither punted nor paved to a smooth, even-textured surface.
 - d. Remove tools, construction quipment, machinery, and surplus material from Project site.
 - e. Leave Project clear and easy for occupancy.
- C. Construction Waste Disposal: "Comply with waste disposal requirements in Section 01 50 00 "Temporary Facilities and Controls" and Section 01 74 19 "Construction Waste Management and Disposal."
- 3.2 REPAIR OF THE WORK
 - A. Complete repair and restoration operations before requesting inspection for determination of Substantial Completion.



Repair or remove and replace defective construction. Repairing includes replacing defective parts, refinishing damaged surfaces, touching up with matching materials, and properly adjusting operating equipment. Where damaged or worn items cannot be repaired or restored, provide replacements. Remove and replace operating components that cannot be repaired. Restore damaged construction and permanent facilities used during construction to specified condition.

ADDITION AND RENOVATIONS

3.3 SUMMARY OF CLOSEOUT DOCUMENTS

- A. Contractor's Affidavit of Payment of Debts and Claims (AIA Document G706)
- B. Contractor's Consent of Surety Company to Final Payment (AIA Document G707) (one copy)
- C. Contractor's Affidavit of Release of Liens (AIA Document G706A) (one copy)
- D. Copy of Letter of Guarantee and Warranty Information (three copies)
- E. Balancing Reports
- F. Subcontractor's Release of Liens had been submitted with each previous Application of Payment (AIA Document G706A) (one copy)
- G. Operation and Maintenance Manuals
- H. Record Shop Drawings and Submittals
- I. As-built Drawings: All construction changes should be clouled and marked.
 - 1. Updated CAD files to reflect changes and arbuilt conditions; AutoCadd dwg file 2010 to 2014 format.
 - 2. Three (3) hard copies of As-builts.
- J. Affidavit of Discharge of State Tax Dabhity (Furnish an affidavit from the State Tax Department that all liabilities thereunder have been discharged by the Contractor and all subcontractors. (Delaware Division of Pevenue, Mr. William Kirby, 302-577-8259).
- K. Punch List Closeout Letter
- L. Electrical Inspection Cretifica
- M. Bond Certification
- N. Boiler Startup and Combustion Reports

END OF SECTION

Schedule of Special Inspection Edit Notes:

- 1. This template reflects the minimum IBC2012 required inspections and frequency of inspection.
- 2. This document is intended to be tailored by the Designer of Record (DOR) to accurately reflect project specific special inspections.
- 3. Tailoring the Schedule of Special Inspection involves the following
 - Deleting pages of the table for work that is clearly not part of a project. For example,
 Masonry Special Inspection Tables should be deleted for a project that does not contain any masonry work.
 - b. Selecting the check box in the first column for those items on each table the will require the special inspection for the project.
 - c. Examining the project for highly critical items that may war antiadoing additional special inspections beyond the IBC minimum or more frequent inspections of more critical items. For Design-Bid-Build project, discuss potential additional inspection with the Government Design Manager to determine if additional inspections should be included.
- 4. DOR is not authorized to delete or reduce the nequency of Special Inspections, where work requiring those inspections is clearly part of the project.
- 5. Items identified with (AISC 341) in the task column are required only where the project is designed per AISC 341 Seismic Provisions for Structural Steel Buildings.
- 6. * indicates Special Inspections, equired for wind resistance in the following areas:
 - a. In wind Exposure Category 5, where V_{asd} is 120 miles per hour or greater.
 - b. In wind Exposure Category CorD, where V_{asd} is 110 miles per hour or greater.
- 7. ****** -indicates Special Inspections required for seismic resistance in the following areas:
 - a. Seismic force-resisting systems in structures assigned to Seismic Design Category C, D, E or F.
 - b. Designated seismic systems in structures assigned to Seismic Design Category C, D, E or F.
 - c. Nechanical and electrical components in structures assigned to Seismic Design Calegory C, D, E or F.
 - Electrical components not part of emergency or standby power systems in structures assigned to Seismic Design Category E or F.
 - Storage racks in structures assigned to Seismic Design Category D, E or F.
 - *** indicates documentation only required for projects designed under AISC 341. For projects only designed under AISC 360, remove "D" designation.
 - # indicates Special Inspections required for progressive collapse resistance.
- 10. Hidden text indicates where each inspection item is located in the building codes.

SCHEDULE OF SPECIAL INSPECTIONS

P – Perform these Special Inspections tasks for each welded joint or member. (AISC 360 & AISC 341)

O – Observe these Special Inspections items on a random daily basis. Operations need not be delayed pending these inspections. (AISC 360 & AISC 341)

D – Document, with a report, that the work has been performed in accordance with the contract documents. (AISC 341)

C – Continuous Special Inspections is the constant monitoring of specific tasks by a special inspector. These inspections must be carried out continuously over the duration of the particular tasks. (Be

P – Periodic Special Inspections is Special Inspections by the special inspector who is intermittently present where the work to be inspected has been or is being performed. (IBC)

STRUCTURAL STEEL

| Required | Task | Perform | Observe | Description |
|----------|---|------------|---------|--|
| | Verify welding procedures (WPS) and consumable certificates | Р | 70 | |
| | Material identification (Type/Grade) | | о | |
| | 3. Welder identification system |) - | 0 | A system shall be maintained by which a welder who has welded a joint or member can be identified. Stamps, if used, shall be the low-stress dye type. |
| | 4. Fit-up groove welds (including jointweometry) | - | 0 | Joint preparation Dimensions (alignment, root opening, root face, bevel) Cleanliness (condition of steel surface) Tacking (tack weld quality and locatio) Backing type and fit (if applicable) |
| | 5. Configuration and finish of access holes | - | 0 | |
| | Fit-up of fillet welds | - | 0 | Dimensions (alignment, gaps at root) Cleanliness (condition of steel surface Tacking (tack weld quality and locatio) |

| | Task | Perform | Observe | Description |
|---|-----------------------------|---------------|----------|--|
| | 1. Use of qualified welders | - | 0 | |
| | 2. Control and handling of | | 0 | Packaging |
| | welding consumables | - | | • Exposure control. |
| | 3. No welding over cracked | _ | 0 | |
| | tack welds | _ | 0 | |
| | 4. Environmental conditions | - | 0 | Wind speed within limit |
| | | | Ŭ | Precipitation and temperature |
| | 5. WPS followed | | | Settings on celding equipment |
| | | | | • Travel speed |
| | | | | Selected welding materials |
| | | | | Shielding gas type/flow rate |
| | | - | 0 | Phyleat upplied |
| _ | | | | • Uterpass temperature maintained |
| | | | | (min./max.) |
| | | | | • Proper position (F, V, H, OH) |
| | | | | Intermix of filler metals avoided unle |
| | | | | approved |
| _ | 6. Welding techniques | | | Interpass and final cleaning |
| | | | 0 | • Each pass within profile limitations |
| | | | | |
| | | \rightarrow | | Each pass meets quality requirement |
| | | 3 | <u> </u> | Each pass meets quality requirement |
| | LOR BI | 3 | <u> </u> | Each pass meets quality requirement |
| | | 3 | | • Each pass meets quality requirement |
| | | 3 | | • Each pass meets quality requirement |

| кequired | Task | Perform | Observe | Description |
|----------|---|---------|---------|---|
| | 1. Welds cleaned | - | 0 | |
| | 2. Size, length, and location of welds | Р | - | |
| | 3. Welds meet visual acceptance criteria | P/D*** | - | Crack prohibition Weld/base-metal fusion Crater cross section Weld profiles Weld size Undercut Porosity |
| | 4. Arc strikes | Р | - | |
| | 5. k-area | р | - (^ | When welcong of doubler plates, continuity plates or stiffeners has been performed in the k-area, visually inspec the web k-area for cracks within 3 in. of the weld. |
| | 6. Backing removed and weld tabs removed (if required) | Р | 7 | |
| | Backing removed, weld tabs removed and finished, and fillet welds added (if required) | P/D | - | |
| | 8. Placement of reinforcing or contouring fillet we ds (if required) | P/D | - | |
| | 9. Repair activities | P/D*** | - | |
| | 10. Document acceptance or rejection of welded join (member | Ρ | - | |

| Required | Task | Perform | Observe | Description |
|----------|--|---------|---------|--|
| | 1. CJP welds (Risk Cat. II) | - | 0 | Ultrasonic testing shall be performed on 10% of CJP groove welds in butt, T- and corner joints subject to transversely applied tension loading in materials 5.16- inch thick or greater. Testing ratemust be increased if > 5% of welds terced have unacceptable defects |
| | 2. CJP welds (Risk Cat. III, IV or V) | - | 0 | Ultrasonic testing shall be performed on all CJP groove whos in butt, T- and corner joints subject to transversely applied tension loading in materials 5/16-inch thick or gnater. |
| | 3. CJP welds | - | Ĉ | Uncasonic cesting shall be performed on 100% of CJP groove welds in materials 5/16-inch or greater. Magnetic particle testing shall be performed on 25% of all beam-to-column CJP groove welds. |
| | 4. Access holes (flange > 2") | | 0 | Thermally cut surfaces of access holes shall be MT or PT when the flange thickness exceeds 2 in. for rolled shapes, or when the web thickness exceeds 2 in. for built-up shapes. Any cracks shall be deemed unacceptable regardless of size or location. |
| | 5. Welded joints subject to fatigue | - | 0 | Radiographic or Ultrasonically inspect welded joints identified on the contract documents to be subject to fatigue per sections 5.1, 5.2, 5.3, 5.4, 6.1, 6.2, and 6.3 of Table A-3 1, AISC 360-10 |

| Required | Task | Perform | Observe | Description |
|----------|--|---------|---------|--|
| | 6. K-area NDT | Р | - | Where welding of doubler plates, continuity plates or stiffeners has been performed in the k-area, the web shall be tested for cracks using magnetic particle testing (MT). The MT inspection area shall include the k-area base metal within 3-inches of the weld. The NT shall be performed no sconer them 48 hours following completion of the welding. |
| | 7. Base metal NDT for lamellar tearing and laminations | - | ° C | After joint completion, base metal thicke than 1 \$\2 nc loaded in tension in the through-thickness direction in tee and comenioints, where the connected material is greater than 3/4 in. and contains CJP groove welds, shall be ultrasonically tested for discontinuities behind and adjacent to the fusion line of such welds. |
| | 8. Beam cope and access hole | | 0 | At welded splices and connections, thermally cut surfaces of beam copes and access holes shall be tested using magnetic particle testing or penetrant testing, when the flange thickness exceeds 1 1/2 in. for rolled shapes, or when the web thickness exceeds 1 1/2 in for built-up shapes. |
| | 9. Reduced beamsection repair | - | 0 | Magnetic particle testing shall be performed on any weld and adjacent area of the reduced beam section (RBS) cut surface that has been repaired by welding, or on the base metal of the RBS cut surface if a sharp notch has been removed by grinding. |
| Ó | Q. Weld tab removal sites | - | 0 | At the end of welds where weld tabs have been removed, magnetic particle testing shall be performed on the same beam-to-column joints receiving UT. |

| Required | Task | Perform | Observe | Description |
|----------|------------------------------|---------------|---------|-------------|
| | 1. Manufacture's | | | |
| | certification available for | · P | - | |
| | fastener materials | | | |
| | 2. Fasteners marked in | | | |
| | accordance with ASTM | - | 0 | |
| | requirements | | | \sim |
| | 3. Proper fasteners selected | d | | |
| | for joint detail (grade, | | | |
| | type, bolt length if | | 0 | |
| | threads are to be | _ | 0 | |
| | excluded from shear | | | |
| | plane | | | |
| | 4. Proper bolting procedure | 2 | 0 | \sim |
| | selected for joint detail | _ | | |
| | 5. Connecting elements, | | | |
| | including appropriate | | | |
| | faying surface condition | | | |
| | and hole preparation, if | | | |
| | specified, meet | | | |
| | applicable requirements | | | |
| | 6. Pre-installation | | | |
| | verification testing by | $\frown \lor$ | | |
| | installation personnel | | | |
| | observed and | | 0/D*** | |
| | documented for astem | | 0,0 | |
| | assemblies and methors | | | |
| | used | | | |
| | | | | |
| | 7. Prorier storage provided | | | |
| | tor balts, ruts, washers, | - | 0 | |
| - | and other fastener | | _ | |
| | components | | | |
STRUCTURAL STEEL

| DURING BO | OLTING (TABLE N5.6-2, AISC 360 | -10 & TABLE J | 7-2, AISC 3 | 41-10): |
|-----------|--|----------------|-------------|--|
| Required | Task | Perform | Observe | Description |
| | Fastener assemblies of suitable condition, paced in all holes and washers (if required) are positioned as required | - | 0 | |
| | 2. Joint brought to the snug- tight condition prior to pretensioning operations | - | Ο | |
| | 3. Fastener component not turned by the wrench prevented from rotating | - | 0 | Å X |
| | Fasteners are pretensioned in accordance with RCSC Specification, progressing systematically from the most rigid point toward the free edges | - | Ŝ | |
| AFTER BOL | TING (TABLE N5.6-3, AISC 360-1 | 0 & TABLE JA | 3, AISC 341 | -10): |
| Required | Task | Perform | Observe | Description |
| | Document acceptance or rejection of bolted connections | /D*** | - | |
| OTHER STE | EL INSPECTIONS (SECTOR 15.7 | , AISC 360-10) |): | |
| Required | Task | Perform | Observe | Description |
| | Anchor rods and other embedment's supporting structural steel | р | - | Verify the diameter, grade, type, and length of the anchor rod or embedded item, and the extent or depth of embedment prior to placement of concrete. |
| S S | Farricated steel or erected steel frame | - | 0 | Verify compliance with the details shown on the construction documents, such as braces, stiffeners, member locations and proper application of joint details at each connection. |

STRUCTURAL STEEL

| Required | Task | Perform | Observe | Description |
|------------|-------------------------------|---------|----------|------------------------------------|
| | 1. Reduced beam sections | | | Contour and finish |
| | (RBS) | P/D | - | Dimensional tolerances |
| | 2 Protected zones | | | No holes or unapproved attachments |
| | 2. FIOLECIEU ZOIIES | P/D | - | made by fabricator or erector |
| | 3 H-niles | | | No holes or unapproved attachments |
| | S. It piles | P/D | - | made by the responsible contractor |
| STEEL ELEN | | | | |
| (TABLE N6 | .1. AISC 360-10): | | | |
| Required | Task | Perform | Observe | Description |
| | 1. Placement and | | | |
| | installation of steel deck | Р | - | |
| | 2. Placement and | | | |
| | installation of steel | Р | - 4 | $h \mathcal{V}$ |
| | headed stud anchors | - | | |
| | 3. Document acceptance or | | | |
| | rejection of steel | Р | | • |
| | elements | • | | |
| COMPOSIT | E STRUCTURES PRIOR TO CONC | | ENT | I |
| (TABLE J9- | 1, AISC 341-10): | | | |
| Required | Task | Perform | Observe | Description |
| | 1. Material identification of | | | |
| | reinforcing steel | | 0 | |
| | (Type/Grade) | | | |
| | 2. Determination of appril | V | | |
| | equivalent for ren forcing | | | |
| | steel other than AST | - | 0 | |
| | A706 | | | |
| | 3. Proper rain arcing steel | | | |
| | size spacing and | - | 0 | |
| | prienation | | | |
| | 4. Peinforcing steel has not | | 0 | |
| | been rebent in the field | - | 0 | |
| | 5. Reinforcing steel has | | | |
| | been tied and supported | - | 0 | |
| .() | as required | | | |
| | 6. Required reinforcing steel | | | |
| | clearances have been | - | 0 | |
| | provided | | | |
| _ | 7. Composite member has | | <u> </u> | |
| | | | | |

STRUCTURAL STEEL

| COMPOSIT (TABLE J9- | TE STRUCTURES DURING CONCR 2. AISC 341-10): | ETE PLACEME | NT | |
|------------------------|---|-------------|---------|----------------|
| Required | Task | Perform | Observe | Description |
| | Concrete: Material identification (mix design, compressive strength, maximum large aggregate size, maximum slump) | - | O/D | S |
| | 2. Limits on water added at the truck or pump | - | O/D | |
| | 3. Proper placement techniques to limit segregation | - | 0 | R ^X |
| ý | | | S | |

STEEL CONSTRUCTION OTHER THAN STRUCTURAL STEEL

| STEEL ROO | F AND FLOOR DECKS (IBC TABLE | 1705.2.2): | | |
|-----------|--|----------------|----------|---|
| Required | Task | Continuous | Periodic | Description |
| | Material verification of cold-formed steel deck | - | р | Confirm that identification markings are provided to conform to ASTM standards specified on approved construction documents. Verify material with manufacturer's certified test reports. |
| | 2. Floor and roof deck welds | - | Р | Visual inspection to confilm that welds meet acceptance price of AWS D1.3 and verify werder qualifications. |
| | OF REINFORCING STEEL (IBC TAB | BLE 1705.2.2): | | |
| Required | Task | Continuous | Periodic | Description |
| | 1. Verification of weldability | - | P | Venfy we dability of reinforcing steel, other than ASTM A 706 based upon carbon equivalent and in accordance with AWS D1.4. |
| | Reinforcing steel resisting flexural and axial forces in intermediate or special moment fames, and boundary elements of special structural walls | ON | | Visually inspect all welds in accordance with AWS D1.4. |
| | 3. Shear reinforcement | с | - | Visually inspect all welds in accordance with AWS D1.4. |
| | 4. Other reinforcingstel | - | Р | Visually inspect all welds in accordance with AWS D1.4. |
| Ś | | | | |

STEEL CONSTRUCTION OTHER THAN STRUCTURAL STEEL

| Required Task Continuous Periodic Description Image: Description 1. Trusses spanning 60-feet or greater - P Verify that temporary and permane truss bracing is installed in accordate with approved truss package. Image: Description - P Verify that temporary and permane truss bracing is installed in accordate with approved truss package. Image: Description - P Visually inspect all welds within the main wind force or seisnic force resisting system in accordate or with AWS D1.4. Image: Description - - P Visually inspect all science or seisnic force resisting system in accordate or with AWS D1.4. Image: Description - - P Visually inspect all science or seisnic force or seisnic forcor seisnic force or seisnic force or seisnic force or seisnic forc |
|--|
| 1. Trusses spanning 60-feet - P Verify that temporary and permanent truss bracing is installed in accordant with approved truss package. 2. Cold-formed steel light-frame construction - P Visually inspect all welds within the main wind force or seisnic force resisting system in accordance with AWS D1.4. 3. Cold-formed steel light-frame construction - P Visually inspect all science and approved trust package. 3. Cold-formed steel light-frame construction - P Visually inspect all science and approved trust package. 3. Cold-formed steel light-frame construction - P Visually inspect all science and approved trust package. bolting, anchoing and other fasteni - - P Visually inspect all science and approved trust package. |
| Cold-formed steel light- frame construction welded connections (*, **) Cold-formed steel light- frame construction Cold-formed |
| 3. Cold-formed steel light- frame construction mechanical connections |
| (*, **) P forcetar susmic force resisting system including shear walls, braces, diap reasons, collectors (drag struts) bold-downs. |
| 4. Cold-formed steel connections (#) - P - P |

CONCRETE CONSTRUCTION

IBC TABLE 1705.3, 1705.12.1:

| Required | Task | Continuous | Periodic | Description |
|----------|---|----------------|----------|--|
| | Reinforcing steel, including prestressing tendons | - | р | Verify prior to placing concrete that reinforcing is of specified type, grade and size; that it is free of oil, our and rust; that it is located and spaced properly; that hooks, be do cos, stirrups and supplemental reinforcement are placed correctly; that lap lengths, spagner and offsets are provided; anothat all mechanical connections are lostalled per the manufacturer's instructions and/or evaluation report. |
| | 2. Anchors cast in concrete | - | Р | Context of the placing concrete that cost in anchors have proper mbedment, spacing and edge distance. |
| | 3. Post-installed anchors or dowels | C | 2 | Inspect all post-installed anchors/dowels as required by the approved ICC-ES report. |
| | 4. Use of required mix design | | Р | Verify that all mixes used comply with the approved construction documents |
| | 5. Concrete slump, air content, and temperature | D ^c | - | At the time fresh concrete is sampled to fabricate specimens for strength test verify these tests are performed. |
| | 6. Concrete & shottcetr placement | с | - | Verify proper application techniques are used during concrete conveyance and depositing avoids segregation or contamination. Verify that concrete is properly consolidated. |
| | 7. Caring temperature and techniques | - | Р | Inspect curing , cold weather protection and hot weather protection procedures. |
| Ó | 8. Pre-stressed concrete | с | - | Verify application of prestressing forces and grouting of bonded prestressing tendons in the seismic force-resisting system. |

CONCRETE CONSTRUCTION

| Required Task Cor 9. Erection of precast concrete 9. 10. In-situ concrete strength verification 10. 11. Formwork 11. | - | Periodic P P | Description Verify that all precast elements are lifted, assembled and braced in accordance with the approved construction documents. Prior to the removal of shores and forms or the stressing of post |
|--|---|--------------------|--|
| 9. Erection of precast concrete 10. In-situ concrete strength verification 11. Formwork | - | P | Verify that all precast elements are lifted, assembled and braced in accordance with the approved construction documents. Prior to the removal of shores and forms or the stressing of post |
| 10. In-situ concrete strength verification 11. Formwork | - | Ρ | Prior to the removal of shores and forms or the stressing of post |
| 11. Formwork | | | tensioned tendons verify that adequate strength has been achieved. |
| | - | Ρ | Inspect the forms to ensure that they are placed pumband conform to the shapes, in es, and dimensions of the members as required by the approved construction documents. |
| 12. Reinforcement complying with ASTM A 615 in special moment frames, special structural walls and coupling beams (**) | - | | Verify that ASTM A 615 reinforcing teel used in these areas complies with ACI 318: 21.1.5.2 by means of certified mill test reports. If this reinforcing steel is to be welded chemical tests shall be performed in accordance with ACI 318: 3.5.2. |
| 13. Reinforcement placement within progressive collapse resisting system (#) | c | - | Visual inspect reinforcing steel placement with a particular emphasis on reinforcing steel anchorages, laps and other details within the progressive collapse resisting system, including horizontal tie force elements, vertical tie force elements and bridging elements. |

| Required | Task | Continuous | Periodic | Description |
|-----------|--|-------------|----------|---|
| | Review material certificates, mix designs, test results and construction procedures | - | Ρ | Verify that materials conform to the requirements of the approved construction documents. |
| AS CONSTR | RUCTION BEGINS (TABLE 1.19.2, | TMS-402/ACI | 530-11): | |
| Required | Task | Continuous | Periodic | Description |
| | Proportions of site- prepared mortar | - | Ρ | Verify that mortar is of the type and color specified on the construction documents, that it conforms to ASTM C 270, and chan it is mixed in accordance with Article 2.0 A of TMS-602/ACI 530.1. |
| | Construction of mortar joints | - | Р | Verify that mortar joints comply with Article 3.3 B of TMS-602/ACI 530.1. |
| | Grade and size of prestressing tendons and anchorages | | Ś | Verify that prestressing tendons comply with Article 2.4 B of TMS- 602/ACI 530.1 and that anchorages, couplers, and end blocks comply with Article 2.4 H. |
| | Location of reinforcement, connectors, and prestressing tendons and anchorages | | Р | Verify that reinforcement is placed in accordance with Article 3.4 of TMS- 602/ACI 530.1. Prestressing tendons shall be placed per Article 3.6 A. |
| | 5. Prestressing technique | - | Ρ | Verify that prestressing technique complies with Article 3.6 B of TMS- 602/ACI 530.1. |
| | 6. Properties of this bed mortar tos ALC masonry | С | Р | Verify that mortar complies with Article 2.1 C of TMS-602/ACI 530.1. Continuous inspection for the first 5000 square feet of wall and periodic for all following applications. |

| | ask | Continuous | Periodic | Description |
|---|--|------------|----------|--|
| | . Grout space | - | Ρ | Verify that grout space is free of martar droppings, debris, loose aggregate, and other deleterious materials and that cleanouts are provided per Article 3.2 D and 3.2 F of TMS-602/At1520. |
| 2 | Grade, type, and size of reinforcement and anchor bolts, and prestressing tendons and anchorages | - | Ρ | Verify that reinforcement, joint reinforcement, was ties, anchor bolts and veneer anchors comply with the approved construction documents and Section 2.16 of TMS 402/ACI 530. |
| 3 | Placement of reinforcement, connectors, and prestressing tendons and anchorages | - | ₽ < | Verify the reinforcement, joint reinforcement, wall ties, anchor bolts and genrer anchors are installed in accordance with the approved onstruction documents and Articles 3.2 E, 3.4, and 3.6 A of TMS 602/ACI 530.1. |
| 4 | Proportions of site- prepared grout and prestressing grout for bonded tendons | 5 | Р | Verify that grout is proportioned per ASTM C 476 and has a slump between 8-11 inches. Self-consolidated grout shall not be proportioned onsite. |
| 5 | Construction of mortar joints | | Ρ | Verify that mortar joints are placed in accordance with Article 3.3 B of TMS 602/ACI 530.1. |

| Required | Task | Continuous | Periodic | Description |
|----------|--|------------|-----------------------|--|
| | Size and location of structural elements | - | Ρ | Verify the locations of structural elements with respect to the approace plans and confirm that toleratices mee the requirements of Article 3.3 cof TMS 602/ACI 530.1. |
| | Type, size, and location of anchors, including other details of anchorage of masonry to structural members, frames, or other construction. | - | Ρ | Verify that correct and regression connections are provided ber the approved plans and Sections 1.16.4.3 and 1.17.1 of tWs 402/ACI 530. |
| | 3. Welding of reinforcement | с | - | Varify wolded reinforcement meet the requirements of Section 2.1.7.7.2, 3.3.3.4(c), and 8.3.3.4(b) of TMS 1.02/ACI 530. |
| | Preparation, construction, and protection of masonry during cold weather (<40°F) or hot weather (>90°F). | | S ^b | Verify that cold-weather construction performed in accordance with Article 1.8 C of TMS 602/ACI 530.1 and hot weather construction per Article 1.8 D of TMS 602/ACI 530.1. |
| | 5. Application and measurement of prestressing force | C | - | Verify the proper prestressting force is applied per Article 3.6 B of TMS 602/ACI 530.1. |
| | 6. Placement of grout and prestressing grout for bonded tendons is in compliance | с | - | Verify placement of grout is done in accordance with Article 3.5 of TMS 602/ACI 530.1 and placement of grout for bonded tendons is in accordance with Article 3.6 C of TMS 602/ACI 530.1. |
| Ö. | 7. Placement of AAC mesonry units and construction of thin-bed mortar joints | с | р | Verify that mortar is placed in accordance with Article 3.3 B.8 of TMS 602/ACI 530.1. Continuous inspection for the first 5000 square feet of wall and periodic for all following applications. |
| | 8. Observation of grout specimens, mortar specimens, and/or prisms | - | Р | Confirm that specimens/prisms are performed as required by Article 1.4 o TMS-602/ACI 530.1. |

| Required | Task | Continuous | Periodic | Description |
|----------|--|------------|----------|---|
| | 1. Verification of Slump Flow and Visual Stability Index (VSI) for self- | - | - | Compressive strength tests should be performed in accordance with ASTN - 1019 for slump flow and ASTN C 1611 |
| | 2. Verification of f'm and f'AAC | - | - | for VSI. Determine the compressive an ength for each wythe by the unit strength method" or by the 'prism test method as specified in Article 1 - B of TMS 602/ACI 530.1 point to construction. |
| | | | | JA |
| | | • | 6 | × |
| | | | | |
| | | 01 | 7 | |
| | | JOI | 7 | |
| | 2Bi | JUI | | |
| | OR DI | | | |
| | - CR-Bi | 501 | | |

| Doguinad | Took | Continueur | Doriodic | Description |
|----------|-------------------------------|---------------|----------|---|
| Required | | Continuous | Periodic | Description |
| _ | Review material certificates, | | - | Verify that materials conform to the |
| | mix designs, test results and | - | Р | requirements of the approved |
| | construction procedures | | | construction documents. |
| AS CONST | RUCTION BEGINS (TABLE 1.19.3, | TMS-402/ACI | 530-11): | |
| Required | Task | Continuous | Periodic | Description |
| | 1. Proportions of site-mixed | | | Verify that mortar is of the type and |
| | mortar | | | color specified on the construction |
| | | | D | documents, that it conforms to ASTM |
| | | _ | ſ | 270, and that it is mixed in accordance |
| | | | | with Article 2.0-A of TMS-602/ACI |
| | | | | 530.1. |
| | 2. Placement of masonry | | | Verify that mortar joints comply with |
| | units & construction of | - | Р 🗸 | Article 3.3 B of TMS-602/ACI 530.1. |
| | mortar joints | | | |
| | 3. Properties of thin-bed | C | | Verify that mortar complies with Artic |
| | mortar for AAC masonry | C | | 2.1 C of TMS-602/ACI 530.1. |
| PRIOR TO | GROUTING (TABLE 1.19.3, TMS-4 | 402/ACI 520-1 | 1) | |
| Required | Task | Continuous | Periodic | Description |
| | 1. Grout space prior to | | | Verify that grout space is free of mort |
| | grouting | | | droppings, debris, loose aggregate, an |
| | | | - | other deleterious materials and that |
| | | | | cleanouts are provided per Article 3.2 |
| | | | | and 3.2 F of TMS-602/ACI 530.1. |
| | 2. Grade, type, and Size of | | | Verify that reinforcement, joint |
| | reinforcement and | | | reinforcement, wall ties, anchor bolts |
| | anchor buits, and | - | Р | and veneer anchors comply with the |
| | prestructing tendons and | | | approved construction documents and |
| | anclorage | | | Section 1.16 of TMS 402/ACI 530. |
| | 3. roportions of site-mixed | | | Verify that grout is proportioned per |
| | yout and prestressing | | р | ASTM C 476 and has a slump betweer |
| | grout for bonded tendons | - | Р | 8-11 inches. Self-consolidated grout |
| | | | | shall not be proportioned onsite. |
| | A Placement of | | | Verify that reinforcement, joint |
| () | reinforcement, | | | reinforcement, wall ties, anchor bolts |
| \sim | connectors, and | | | and veneer anchors are installed in |
| | prestressing tendons and | С | - | accordance with the approved |
| | anchorages | | | construction documents and Articles |
| | | | | 3.2 E, 3.4, and 3.6 A of TMS 602/ACI |
| | | | | 530.1. |

| DURING M | ASONRY CONSTRUCTION (TAB | LE 1.19.3, TMS | -402/ACI 53 | 30-11): |
|------------|--|----------------|-------------|--|
| Required | Task | Continuous | Periodic | Description |
| | Size and location of structural elements | - | Ρ | Verify the locations of structural elements with respect to the approved plans and confirm that tolerances meet the requirements of Article 2.3 s of TMS 602/ACI 530.1-11. |
| | 2. Type, size, and location of anchors, including other details of anchorage of masonry to structural members, frames, or other construction. | с | - | Verify that correct anon rages and connections are provided ber the approved plans and socions 1.16.4.3 and 1.17.1 of two 402/ACI 530-11. |
| | 3. Welding of reinforcement | с | - | Verify wilded reinforcement meet the requirements of Section 2.1.7.7.2, 3.3.3.4(c), and 8.3.3.4(b) of TMS 102/ACI 530. |
| | Preparation, construction, and protection of masonry during cold weather (<40°F) or hot weather (>90°F). | | S | Verify that cold-weather construction is performed in accordance with Article 1.8 C of TMS 602/ACI 530.1 and hot weather construction per Article 1.8 D of TMS 602/ACI 530.1. |
| | 5. Application and measurement of prestressing force | C | - | Verify the proper prestressting force is applied per Article 3.6 B of TMS 602/ACI 530.1. |
| | 6. Placement of grout and prestressing grout for bonded tendons is in compliance | с | - | Verify placement of grout is done in accordance with Article 3.5 of TMS 602/ACI 530.1 and placement of grout for bonded tendons is in accordance with Article 3.6 C of TMS 602/ACI 530.1. |
| | 7. Placement of AAC masonry units and construction of thin-bed mortar joints | с | - | Verify that mortar is placed in accordance with Article 3.3 B.8 of TMS- 602/ACI 530.1-11. |
| $\sqrt{2}$ | 8. Observation of grout specimens, mortar specimens, and/or prisms | с | - | Confirm that specimens/prisms are performed as required by Article 1.4 of TMS-602/ACI 530.1-11. |

| Required | Task | Continuous | Periodic | Description |
|----------|---|-----------------|----------|---|
| | Verification of Slump Flow and Visual Stability Index (VSI) for self- consolidating grout | - | - | Compressive strength tests should be performed in accordance with ASTM 1019 for slump flow and ASTM C 7611 for VSI. |
| | 2. Verification of f'm and f'AAC | - | - | Determine the compressive attendth for each wythe by the upit strength method" or by the (prismitest method as specified in Article 1 + B of TMS 602/ACI 530.1 prior to construction ar every 5000 square feet during construction. |
| | Verification of proportions of materials in premixed or pre- blended mortar and grout | - | | Varify that proportions for mortar me ASTALC270 and proportions for grout meet ASTM C 476. |
| | blended mortar and grout | | G | |
| | | \mathcal{O}'' | | |
| | RBI | | | |
| | | | | |

WOOD CONSTRUCTION

| IBC 1705.5 | IBC 1705.5, 1705.10.1 & 1705.11.2 | | | |
|------------|---|------------|----------|---|
| Required | Task | Continuous | Periodic | Description |
| | 1. High-load diaphragms | - | Ρ | Verify thickness and grade of sheathing, size of framing members at panel edges, nail/staple diameter and length, and the number of fast over lines and fastener spacing are per approved plans. |
| | Metal-plate-connected wood trusses spanning 60 feet or greater | - | Ρ | Verify that temporing and permanent truss bracing in instance of accordance with approvectouss package. |
| | 3. Field Gluing (*, **) | С | - | Inspect all field gruing of structural woodele cent within the main wind- force resisting system. |
| | Nailing, bolting, anchoring and other fastening of components (*, **) | - | Ś | infastep rs within the main wind-force resisting system are spaced less than 4- nches, verify that proper nailing, bolting, anchoring and other fastening of shear walls, diaphragms, drag struts, braces, and holdowns. |
| | Nailing, bolting, anchoring and other fastening of components (#) | | Ρ | Verify proper nailing, bolting, anchoring, and other fastening components within the progressive collapse resisting system, including horizontal tie force elements, vertical tie force elements and bridging elements. |
| Ś | | | | |

SOILS CONSTRUCTION

| IBC TABLE | 1705.6 | | | | |
|-----------|--|-----------------------------|------------|----------|--|
| Required | Task | | Continuous | Periodic | Description |
| | 1. Foundation capacity | bearing | - | Ρ | Verify the materials below foundations are adequate to achieve the design bearing capacity. |
| | 2. Excavations | | - | Ρ | Verify the excavations are extended to the proper depth and have cached proper material. |
| | Perform clas testing of co materials | sification and mpacted fill | - | Ρ | |
| | 4. Compacted f | fill material | С | - | Verify the use of proper materials, densities and lift thicknesses during placement and compaction of compacted fill. |
| | 5. Subgrade | | - | ₽ | Prior to placement of compacted fill, observe sub-grade and verify that site has been prepared properly. |
| | | 8 | JDN. | • | |
| , O | . { 0 | | | | |

DRIVEN DEEP FOUNDATIONS

| Required | Task | Continuous | Periodic | Description |
|----------|---|------------|----------|---|
| | 1. Verify element materials, sizes and lengths comply with the construction documents | С | - | |
| | Determine capacities of test elements and conduct necessary load tests, as required | С | - | |
| | Observe drilling operations and maintain complete and accurate records for each element | С | - | |
| | 4. Verify placement locations & plumbness, confirm type & size of hammer, record number of blows per foot of penetration, determine required penetrations to achieve design capacity, record tip and butt elevations and document any damage to foundation element | | S | |
| | 5. Steel piles | - | - | Inspect per STRUCTURAL STEEL schedule |
| | 6. Concrete dements and concrete-hucd elements. | - | - | Inspect per CONCRETE CONSTRUCTION schedule |
| | 7. Spenalty kiles | - | - | Perform additional inspection as determined by the DOR. |

CAST-IN-PLACE DEEP FOUNDATIONS

| IBC TABLE 1705.8 | | | | |
|------------------|---|------------|------------|--|
| Required | Task | Continuous | Periodic | Description |
| | Observe drilling operations and maintain complete and accurate records for each element | С | - | |
| | 2. Verify placement locations and plumbness, confirm element diameters, bell diameters (if applicable), lengths, embedment into bedrock (if applicable) and adequate end-bearing strata capacity. Record concrete or grout volumes. | С | - | 2 URROS |
| | Perform additional inspections for concrete elements. | - | \bigcirc | Inspect per CONCRETE CONSTRUCTION schedule |

HELICAL PILE FOUNDATIONS

| IBC 1705.9 | | J | | |
|------------|--|------------|----------|-------------|
| Required | Task | Continuous | Periodic | Description |
| | Record installation equipment used, nile dimensions, to elevations, final depth, and final installation torque | С | - | |
| | | | | |

SPRAYED FIRE-RESISTANT MATERIALS (SFRM)

| IBC 1705.13 | | | | | |
|-------------|-----------------------|------------|----------|---|--|
| Required | Task | Continuous | Periodic | Description | |
| | 1. Surface condition | - | Ρ | Prior to application confirm that surface has been prepared per the approved fire-resistance design and manufacturer's instruction | |
| | 2. Application | - | Ρ | Prior to application conferment the substrate meets the minimum ambient temperature per the approved fire- resistance design incomanufacturer's instructions. | |
| | 3. Material thickness | - | Р | Verify that the thickness of the SFRM to structural elements is not less than the thickness require by the fire-resistant design in more than 10 percent of the preasurement, but in no case less than minimum allowable thickness required by 1705.13.4.1. | |
| | 4. Material density | | | Verify that the density of the SFRM to structural elements is not less than the density specified in the fire-resistant design. | |
| | 5. Bond strength | <u>5</u> . | Р | Verify cohesive/adhesive bond strength of the cured SFRM applied to the structural elements is not less than 150 psf. | |

MASTIC AND INTUMESCENT FIRE-RESISTANT COATINGS (IBC 1705.14)

| IBC 1705.1 | 4 | | | |
|------------|------------------------|------------|----------|---|
| Required | Таях | Continuous | Periodic | Description |
| | 1. Surface preparation | - | Ρ | Prior to application confirm that surface temperature and substrate are acceptable in accordance with AWCI 12-B. |
| 2 | 2. Thickness | - | Ρ | Final thickness of coating must be verified in multiple locations prior to applying top coat per AWCI 12-B. |

EXTERIOR INSULATION AND FINISH SYSTEMS (EIFS)

| IBC 1705.15 | | | | |
|-------------|---------------------------|------------|----------|--|
| Required | Task | Continuous | Periodic | Description |
| | Material and installation | С | - | Verify that water-resistive barrier, complying with ASTM E 2570, is installed appropriately over a sherthing substrate. |

FIRE-RESISTANT PENETRATIONS AND JOINTS

| IBC 1705.1 | 6 | | | Ň |
|------------|--|------------|----------|---|
| Required | Task | Continuous | Periodic | Description |
| | 1. Penetration firestops | - | Р | Listed systems shall be inspected in accordance with ASTM E 2174. |
| | Fire-resistant joint systems | - | Р | Lited systems shall be inspected in accordance with ASTM E 2393. |

SMOKE CONTROL

| IBC 1705.1 | 7 | | | |
|--------------------------|--|------------|----------|---|
| Required | Task | Continuous | Periodic | Description |
| | Verify device locations and perform leakage testing | | Ρ | Perform during erection of ductwork and prior to concealment. |
| | 2. Pressure difference testing, flow measurements and detection and control verification | - | Р | Perform prior to occupancy and after sufficient completion. |
| | verificatio | | | |
| $\overline{\mathcal{O}}$ | • | | | |

ARCHITECTURAL COMPONENTS

| IBC 1705.1 | 0.3, 1705.11.5 & 1705.11.7 | | | |
|------------|---|------------|----------|--|
| Required | Task | Continuous | Periodic | Description |
| | 1. Roof and wall cladding (*) | - | Ρ | Verify appropriate materials, fasteners and attachment at commencement of work and at completion. |
| | Erection and fastening of exterior cladding or interior and exterior veneers (**) | - | Ρ | Verify appropriate materials, facteners and attachment at commencement of work and at completions (Not required if height is less than 30 feat or weight is less than 5psf) |
| | Erection and fastening of interior and exterior nonbearing walls (**) | - | Ρ | Verify appropriate materials, fasteners and attrachment at commencement of workshout completion. (Not required if height is less than 30 feet or weight is hess than 15psf). |
| | 4. Access floors (**) | - | C | Verify that anchorage complies with opproved construction documents. Inspection of post-installed anchors shall comply with approved ICC-ES report. |
| | 5. Storage racks (**) | | Р | Verify that anchorage complies with approved construction documents. Inspection of post-installed anchors shall comply with approved ICC-ES |

SCHEDULE OF SPECIAL INSPECTION 01 80 00 -28

MECHANICAL & ELECTRICAL COMPONENTS

| IBC 1705.1 | 1.4, 1/05.11.0 & 1/05.12.3 and | UFC 3-301-01 | | |
|----------------------|--|--------------|-----------------------|--|
| Required | Task | Continuous | Periodic | Description |
| | Anchorage of emergency or standby power systems (**) | - | Р | Verify that anchorage complies with approved construction documents. |
| | Anchorage of electrical equipment not part of emergency or standby power systems (**) | - | Ρ | Verify that anchorage complies with approved construction documents. |
| | Installation of piping systems carrying hazardous materials and their associated mechanical units (**) | - | Ρ | Verify that installation and restraint comply with a proved construction documents. |
| | Installation of HVAC ductwork containing hazardous materials (**) | - | Р | works that installation and restraint comply with approved construction documents. |
| | Installation of vibration isolation systems having a clearance of less than 1/4 inch between the equipment support frame and restraint | | S ^b | Verify that installation complies with approved construction documents and manufacturer's recommendations. |
| | 6. Designated seismic systems | | р | Confirm that manufacturer's certificat of compliance conforms to the requirements of Section 13.2 of ASCE Verify that the label, anchorage or mounting conforms to the manufacturer's certificate of compliance. |
| | 7. Designated seismic system equipment verification | - | р | Verify model number and serial number are in conformance with th Project Specific Seismic Qualificatio (PSSQ). Verify Tag ID is correct and installed per specifications. |
| $\mathbf{\tilde{5}}$ | Designated seismic system equipment mounting | - | Р | Verify that anchor base bolting is installed per PSSQ. Verify that equipment bracing is installed per PSSQ. Verify that bracing attachments are installed per PSSQ. |

MECHANICAL & ELECTRICAL COMPONENTS

| Required | Task | Continuous | Periodic | Description |
|-----------------------|--|------------|----------|--|
| <u>nequirea</u> | Designated seismic system utility conduit/piping | - | P | Verify that conduit/piping is connected to the equipment per PSSQ (flex or rigid) Verify that conduit/piping is seismically supported independently of equipment and in coerdance with PSSQ support requirements. |
| | 10. Designated seismic system clearance | - | Ρ | Adjacent equipment – Verify that there is adequate gap to eliminate the possibility of pounding. Conduct/piping – Verify that there is adequate gap to eliminate possibility or pounding. |
| | | RES | | |
| BC 1705.1 | 1.8 | | 5 | |
| BC 1705.1 Required | 1.8 Task | Continuous | Periodic | Description |

| Signature | Date | Signature | Date |
|--|--|--|---|
| | | | |
| Owner's authorization: | | Building Official's A | cceptance: |
| Signature | | Date | Design Professional Seal |
| | | | |
| Andrew T. Coats, PE (type or print name) | | | |
| Q- | • | | |
| Prepared by: | `` | | |
| Interim Report Frequency: | \sim | | or $oxtimes$ per attached schedule. |
| Job site safety and means and me | thods of construc | tion are solely the respor | nsibility of the Contractor. |
| A Final Report of Special Inspection correction of any discrepancies no Use and Occupancy | ons documenting ted in the inspec | completion of all require ions shall be submitted p | d Special Inspections, testing and prior to issuance of a Certificate of |
| Responsible Charge. | eu io ine dulla | | gistered Design Professional In |
| the Registered Design Professiona the Contractor of his or her respon | al in Responsible sibilities. | Charge. The Spheral In: | spection program does not relieve |
| The Special Inspection Coordinate the Building Official and the I discrepancies shall be brought discrepancies are not corrected, th | r shall keep reco Registered Desig to the immediat he discrepancies | rds of all inspections and on Professional in Re e attention of the on shall be brought o the all charge. The Decision | Asha (furnish inspection reports to spensible Charge. Discovered ntractor for correction. If such tention of the Building Official and |
| Str Are | chitectural | Mechanical/Electrical/ | Plumbirt |
| This Statement of Special Inspect Special Inspection and Structural Inspection services applicable to t the identity of other approved ag Statement of Special Inspections e | <i>ions</i> is submitted Festing requirement this project as we gencies to be ret | as a condition for permi ents of the Building Code Il as the name of the S ained for conducting th lowing disciplines: | it issuance in accordance with the . It includes a schedule of Special pecial Inspection spordinator and ese inspections and tests. This |
| Design Professional in Respon | sible Charge: | | |
| Owner: Brandywine School Dis | strict | | , C |
| Location: Carrcroft Elementary | v School | | |
| | | | |

Schedule of Inspection and Testing Agencies

This Statement of Special Inspections / Quality Assurance Plan includes the following building systems:

| Soils and Foundations Cast-in-Place Concre Precast Concrete Masonry Structural Steel Cold-Formed Steel Fr | s Spray Fire te Solution Spray Fire Wood Cons Exterior Ins Mechanical Architectura raming Special Cas | Resistant Material struction sulation and Finish System I & Electrical Systems al Systems ses |
|--|---|--|
| Special Inspection Agencies | Firm | Address, Telephone, e-mail |
| 1. Special Inspection Coordinator | | |
| 2. Inspector | | |
| 3. Inspector | | |
| 4. Testing Agency | | |
| 5. Testing Agency | | |
| 6. Other | | |
| Note: The inspectors and testing age the Contractor or Subcontractor whether the Contractor | gencies shall be engaged by the Own nose work is to be inspected or tes | ner or the Owner's Agent, and not by ted. Any conflict of interest must be |

CASE Form 101 • Statement of Special Inspections • ©CASE 2004

losed to the Building Official, prior to commencing work.

SECTION 03 30 00

CAST-IN-PLACE CONCRETE

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and S Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section includes cast-in-place concrete, including formwork, reinforcement, concrete materials, mixture design, placement procedures, and finishes.
- B. Related Requirements:
 - 1. See Civil Drawings General Notes, Section "Each Moving" for drainage fill under slabson-grade.
 - 2. See Civil Drawings General Notes. Section "Concrete Paving" for concrete pavement and walks.

1.3 DEFINITIONS

- A. Cementitious Materials: Forthand rement alone or in combination with one or more of the following: blended branula cement, fly ash, and other pozzolans, materials subject to compliance with requirements.
- B. W/C Ratio: The ratio by weight of water to cementitious materials.

1.4 PREINSTALLATION MEETINGS

Prein callation Conference: Conduct conference at Project site.

Before submitting design mixtures, review concrete design mixture and examine procedures for ensuring quality of concrete materials. Require representatives of each entity directly concerned with cast-in-place concrete to attend, including the following:

- a. Contractor's superintendent.
- b. Independent testing agency responsible for concrete design mixtures.
- c. Ready-mix concrete manufacturer.
- d. Concrete Subcontractor.
- e. Special concrete finish Subcontractor.

2. Review special inspection and testing and inspecting agency procedures for field quality control, concrete finishes and finishing, cold- and hot-weather concreting procedures, curing procedures, construction contraction and isolation joints, and joint-filler strips, semi-rigid joint fillers, forms and form removal limitations, shoring and reshoring procedures, vapor-barrier installation, anchor rod and anchorage device installation tolerances, steel reinforcement installation, methods for achieving specified floor and slab flatness and levelness floor and slab flatness measurement, concrete repurprocedures, and concrete protection.

1.5 ACTION SUBMITTALS

- A. Product Data: For each type of product.
- B. Design Mixtures: For each concrete mixture. Submit alternate design mixtures when characteristics of materials, Project conditions, weather, test results or ther circumstances warrant adjustments.
 - 1. Indicate amounts of mixing water to be withheld for later addition at Project site.
- C. Steel Reinforcement Shop Drawings: Placing Drawings that detail fabrication, bending, and placement. Include bar sizes, lengths, material, scale, bar schedules, stirrup spacing, bent bar diagrams, bar arrangement, splices and laps, me handed connections, tie spacing, hoop spacing, and supports for concrete reinforcement.
- D. Construction Joint Layout: Indicate proposed construction joints required to construct the structure.
 - 1. Location of construction joints is abject to approval of the Architect.
- E. Slab on Grade Contraction Control Joint Layout: Indicate proposed Slab on Grade Control Joints which meet specified requirements.
 - 1. Location of all Slavon Grade Control Joint are subject to approval of the Architect.
 - 2. Provide dimensions to all Slab on Grade Control Joint locations.
 - 3. Submittan to indicate all items associated with Slab on Grade Control Joints, such as:
 - Second-Pour "Diamonds" around all column pier locations inside Column Isolation Joints.

Additional rebar at mid-depth of slab at re-entrant corners, door openings, slab opening, etc.

Housekeeping pad layout

INFORMATIONAL SUBMITTALS

- A. Qualification Data: For Installer, manufacturer and testing agency.
- B. Welding certificates.
- C. Material Certificates: For each of the following, signed by manufacturers:

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- 1. Cementitious materials.
- 2. Admixtures.
- 3. Form materials and form-release agents.
- 4. Steel reinforcement and accessories.
- 5. Fiber reinforcement.
- 6. Curing compounds.
- 7. Floor and slab treatments.
- 8. Bonding agents.
- 9. Adhesives.
- 10. Vapor barrier.
- 11. Semirigid joint filler.
- 12. Joint-filler strips.
- 13. Repair materials.
- D. Material Test Reports: For the following, from a qualified testing agence
 - 1. Aggregates
- E. Formwork Shop Drawings: Prepared by or under the supervision of a qualified professional engineer, detailing fabrication, assembly, and support of formwork.
 - 1. Shoring and Reshoring: Indicate proposed renedule and sequence of stripping formwork, shoring removal, and reshoring installation and removal.
- F. Floor surface flatness and levelness measurements indicating compliance with specified tolerances.
- G. Field quality-control reports.
- H. Minutes of preinstallation confere
- 1.7 QUALITY ASSURANCE
 - A. Installer Qualifications: A qualified installer who employs on Project personnel qualified as ACI-certified. Natwork Technician and Finisher and a supervisor who is an ACI-certified Concrete Flatwork Technician.
 - B. Manufacturer Qualifications: A firm experienced in manufacturing ready-mixed concrete products and that complies with ASTM C 94 requirements for production facilities and equipment.
 - Manufacturer certified according to NRMCA's "Certification of Ready Mixed Concrete Production Facilities."
 - Testing Agency Qualifications: An independent agency, qualified according to ASTM C 1077 and ASTM E 329 for testing indicated.
 - 1. Personnel conducting field tests shall be qualified as ACI Concrete Field Testing Technician, Grade 1, according to ACI CP-1 or an equivalent certification program.

- 2. Personnel performing laboratory tests shall be ACI-certified Concrete Strength Testing Technician and Concrete Laboratory Testing Technician, Grade I. Testing agency laboratory supervisor shall be an ACI-certified Concrete Laboratory Testing Technician, Grade II.
- D. Welding Qualifications: Qualify procedures and personnel according to AWS D1.4.

1.8 PRECONSTRUCTION TESTING

A. Preconstruction Testing Service: Engage a qualified testing agency to perform precenstruction testing on concrete mixtures.

1.9 DELIVERY, STORAGE, AND HANDLING

A. Steel Reinforcement: Deliver, store, and handle steel reinforcement to prevent bending and damage. Avoid damaging coatings on steel reinforcement.

1.10 FIELD CONDITIONS

- A. Cold-Weather Placement: Comply with ACI 30.1 and as follows. Protect concrete work from physical damage or reduced strength that could be clused by frost, freezing actions, or low temperatures.
 - 1. When average high and low competatore is expected to fall below 40 deg F for three successive days, maintain delivered concrete mixture temperature within the temperature range required by ACI 301
 - 2. Do not use frozen materials or materials containing ice or snow. Do not place concrete on frozen subgrade or of subgrade containing frozen materials.
 - 3. Do not use calcium coloride, salt, or other materials containing antifreeze agents or chemical accelerators inless otherwise specified and approved in mixture designs.
- B. Hot-Weather Liacement: Comply with ACI 301 and as follows:
 - 1. Maintail concrete temperature below 90 deg F at time of placement. Chilled mixing water or chopped ice may be used to control temperature, provided water equivalent of ice is calculated to total amount of mixing water. Using liquid nitrogen to cool concrete is Contractor's option.

Fog-spray forms, steel reinforcement, and subgrade just before placing concrete. Keep subgrade uniformly moist without standing water, soft spots, or dry areas.

PRODUCTS

CONCRETE, GENERAL

A. ACI Publications: Comply with the following unless modified by requirements in the Contract Documents:

CAST-IN-PLACE CONCRETE 03 30 00 - 4

- 1. ACI 301.
- 2. ACI 117.

2.2 FORM-FACING MATERIALS

- A. Smooth-Formed Finished Concrete: Form-facing panels that provide continuous, true, and smooth concrete surfaces. Furnish in largest practicable sizes to minimize number of joint.
 - 1. Plywood, metal, or other approved panel materials.
 - 2. Exterior-grade plywood panels, suitable for concrete forms, complying with POOPS 1, and as follows:
 - a. High-density overlay, Class 1 or better.
 - b. Medium-density overlay, Class 1 or better; mill-release areat treated and edge sealed.
 - c. Structural 1, B-B or better; mill oiled and edge sealed.
 - d. B-B (Concrete Form), Class 1 or better; mill oiled and edge sealed.
 - 3. Overlaid Finnish birch plywood.
- B. Rough-Formed Finished Concrete: Plywood, lumber, meal, or another approved material. Provide lumber dressed on at least two edges and one side for tight fit.
- C. Pan-Type Forms: Glass-fiber-reinforced plastic or formed steel, stiffened to resist plastic concrete loads without detrimental deformation.
- D. Chamfer Strips: Wood, metal, PVC or rubber strips, 3/4 by 3/4 inch, minimum.
- E. Rustication Strips: Wood, metal, PVC, or rubber strips, kerfed for ease of form removal.
- F. Form-Release Agent: commercially formulated form-release agent that does not bond with, stain, or adversely an et obscrete surfaces and does not impair subsequent treatments of concrete surfaces.
 - 1. Formulat form-release agent with rust inhibitor for steel form-facing materials.
- G. Form Tits: Factory-fabricated, removable or snap-off glass-fiber-reinforced plastic or metal form tics designed to resist lateral pressure of fresh concrete on forms and to prevent spalling of concrete on removal.



Furnish units that leave no corrodible metal closer than 1 inch to the plane of exposed concrete surface.

STEEL REINFORCEMENT

- A. Reinforcing Bars: ASTM A 615, Grade 60, deformed.
- B. Galvanized Reinforcing Bars: ASTM A 615, Grade 60, deformed bars, ASTM A 767, Class I zinc coated after fabrication and bending.

- C. Plain-Steel Wire: ASTM A 1064/A 1064M, as drawn.
- D. Deformed-Steel Wire: ASTM A 1064/A 1064M.

2.4 REINFORCEMENT ACCESSORIES

- A. Joint Dowel Bars: ASTM A 615, Grade 60, plain-steel bars, cut true to length with ends equare, and free of burrs.
- B. Zinc Repair Material: ASTM A 780.
- C. Bar Supports: Bolsters, chairs, spacers, and other devices for spacing, supporting, and fastening reinforcing bars in place precast concrete according to CRSI's "Manual of Standard Practice," of greater compressive strength than concrete and as follows:
 - 1. For concrete surfaces exposed to view, where legs of wire bas supports contact forms, use CRSI Class 1 plastic-protected steel wire or CRSI Class 2 sum as-steel bar supports.
 - 2. For zinc-coated reinforcement, use galvanized wire or dielectric-polymer-coated wire bar supports.

2.5 CONCRETE MATERIALS

- A. Source Limitations: Obtain each type or class of cementitious material of the same brand from the same manufacturer's plant, obtain agree are from single source, and obtain admixtures from single source from single manufacture.
- B. Cementitious Materials:
 - 1. Portland Cement: ASTMC 750, Type I; Type II or Type I/II gray.
 - 2. Fly Ash: ASTMC 118, Class F or C.
 - 3. Blended Hydraulic Cement: ASTM C 595, Type IS, portland blast-furnace slag, Type IP, portland-pezzolan, Type IL, portland-limestone, or Type IT, ternary blended cement.
- C. Normal-Weight Aggregates: ASTM C 33, Class 3S coarse aggregate or better, graded. Provide aggregates from a single source with documented service record data of at least 10 years' satisfactory service in similar applications and service conditions using similar aggregates and cententious materials. ASTM C33, Class 4S coarse aggregate or better, graded, for exterior concrete.
 - Maximum Coarse-Aggregate Size:
 - a. 1-1/2 inches nominal, for slabs on grade.
 - b. All other concrete: 1 inch nominal.
 - 2. Fine Aggregate: Free of materials with deleterious reactivity to alkali in cement.
- D. Air-Entraining Admixture: ASTM C 260.
- E. Chemical Admixtures: Certified by manufacturer to be compatible with other admixtures and that do not contribute water-soluble chloride ions exceeding those permitted in hardened concrete. Do not use calcium chloride or admixtures containing calcium chloride.

CAST-IN-PLACE CONCRETE 03 30 00 - 6

- 1. Water-Reducing Admixture: ASTM C 494/C 494M, Type A.
- 2. Retarding Admixture: ASTM C 494/C 494M, Type B.
- 3. Water-Reducing and Retarding Admixture: ASTM C 494/C 494M, Type D.
- 4. High-Range, Water-Reducing Admixture: ASTM C 494/C 494M, Type F.
- 5. High-Range, Water-Reducing and Retarding Admixture: ASTM C 494/C 494M, Type G.
- 6. Plasticizing and Retarding Admixture: ASTM C 1017/C 1017M, Type II.
- F. Water: ASTM C 94 and potable.

2.6 FIBER REINFORCEMENT

A. Synthetic Macro-Fiber: Polyolefin macro-fibers engineered and designed for use in concrete, complying with ASTM C 1116, Type III, 1 to 2-1/4 inches long.

2.7 VAPOR BARRIER

- A. Vapor Barrier: Water-vapor transmission rate (permeance) less than 0.015 perms (gr/ft2/hr/in-Hg), in accordance with ASTM E 1745. The product must meet water-vapor transmission rate (0.01 perms) requirement for both the new materia and the ASTM E 1745 mandatory conditioning test (ASTM E 1745; paragraph 7.12 through 7.15.) Provide all manufacturers' accessories required for complete installation in Judire mastic and seam tape. Seam tape to be provided with a water-vapor transmission rate of 0.3 perms or lower.
 - 1. Products: Subject to compliance with requirements, provide one of the following:
 - a. Layfield Construction Materials; VaporFlex 15.
 - b. Reef Industries, Inc.; Sriffolyn Vaproguard.
 - c. Stego Industries. LES: Stego Wrap 15 mil Class A.

2.8 LIQUID FLOOR TREATMENTS

A. Penetrating Liorid Flow Treatment: Clear, chemically reactive, waterborne solution of inorganic silicate or siliconate materials and proprietary components; odorless; that penetrates, hardens, and densifies concrete surfaces.

2.9

CURING MATERIALS

Evaporation Retarder: Waterborne, monomolecular film forming, manufactured for application in fresh concrete.

Moisture-Retaining Cover: ASTM C 171, polyethylene film or white burlap-polyethylene sheet.

Water: Potable.

D. Clear, Waterborne, Membrane-Forming Curing Compound: ASTM C 309, Type 1, Class B, nondissipating, certified by curing compound manufacturer to not interfere with bonding of floor covering.

Tetra Tech 200-15704-17001

E. Clear, Waterborne, Membrane-Forming Curing and Sealing Compound: ASTM C 1315, Type 1, Class A.

2.10 RELATED MATERIALS

- A. Expansion- and Isolation-Joint-Filler Strips: ASTM D 1751, asphalt-saturated cellulosic filer or ASTM D 1752, cork or self-expanding cork.
- B. Bonding Agent: ASTM C 1059, Type II, nonredispersible, acrylic emulsion or butadiene.
- C. Epoxy Bonding Adhesive: ASTM C 881, two-component epoxy resin, capable of humid curing and bonding to damp surfaces, of class suitable for application temperature and of grade to suit requirements, and as follows:
 - 1. Types I and II, nonload bearing or Types IV and V, load bearing, for bonding hardened or freshly mixed concrete to hardened concrete.
- D. Reglets: Fabricate reglets of not less than 0.022-inch-thick, alvanzed-steel sheet. Temporarily fill or cover face opening of reglet to prevent intrusion of concrete or debris.
- E. Dovetail Anchor Slots: Hot-dip galvanized-steel sheet, not less than 0.034 inch thick, with bent tab anchors. Temporarily fill or cover face opening of slots to prevent intrusion of concrete or debris.

2.11 REPAIR MATERIALS

- A. Repair Underlayment: Cemant-based, polymer-modified, self-leveling product that can be applied in thicknesses from 1/o inch and that can be feathered at edges to match adjacent floor elevations.
 - 1. Cement Binder: AXAM C 150, portland cement or hydraulic or blended hydraulic cement as defined it ASTM C 219.
 - 2. Primer: roduct of underlayment manufacturer recommended for substrate, conditions, and appreadon.
 - 3. Aggregate: Well-graded, washed gravel, 1/8 to 1/4 inch or coarse sand as recommended by underlayment manufacturer.
 - Compressive Strength: Not less than 4100 psi at 28 days when tested according to ASTM C 109/C 109M.

B.

Repair Overlayment: Cement-based, polymer-modified, self-leveling product that can be applied in thicknesses from 1/4 inch and that can be filled in over a scarified surface to match adjacent floor elevations.

- 1. Cement Binder: ASTM C 150, portland cement or hydraulic or blended hydraulic cement as defined in ASTM C 219.
- 2. Primer: Product of topping manufacturer recommended for substrate, conditions, and application.

- 3. Aggregate: Well-graded, washed gravel, 1/8 to 1/4 inch or coarse sand as recommended by topping manufacturer.
- 4. Compressive Strength: Not less than 5000 psi at 28 days when tested according to ASTM C 109/C 109M.

2.12 CONCRETE MIXTURES, GENERAL

- A. Prepare design mixtures for each type and strength of concrete, proportioned on the cas laboratory trial mixture or field test data, or both, according to ACI 301.
 - 1. Use a qualified independent testing agency for preparing and reporting proposed mixture designs based on laboratory trial mixtures.
- B. Cementitious Materials: Use fly ash, and pozzolan, as needed to reduce the total amount of portland cement, which would otherwise be used, by not less than 40 percent. Limit percentage, by weight, of cementitious materials other than portland cement in concrete as follows:
 - 1. Fly Ash: 25 percent.
 - 2. Combined Fly Ash and Pozzolan: 25 percent.
- C. Admixtures: Use admixtures according to manufacturer's written instructions.
 - 1. Use water-reducing, high-range water-reducing or plasticizing admixture in concrete, as required, for placement and workability
 - 2. Use water-reducing and -retarding admixture when required by high temperatures, low humidity, or other adverse placement conditions.
 - 3. Use water-reducing admixture in numped concrete, and concrete with a w/c ratio below 0.50.
 - 4. Use corrosion-inhibiting admixture in concrete mixtures where indicated.

2.13 CONCRETE MIXTURYS FOR BUILDING ELEMENTS

- A. Footings and Courtation Walls: Normal-weight concrete.
 - 1. Minimum Compressive Strength 4,000 psi at 28 days.
 - 2. Maximum W/C Ratio: 0.50.
 - 3. Stump Limit:4 inches, 8 inches for concrete with verified slump of 2 to 4 inches before adding high-range water-reducing admixture or plasticizing admixtureplus or minus 1 inch.

Air Content: 6 percent, plus or minus 1.5 percent at point of delivery for 1-inchnominal maximum aggregate size.

Slabs-on-Grade: Normal-weight concrete.

- 1. Minimum Compressive Strength: 4,000 psi at 28 days.
- 2. Maximum W/C Ratio: 0.50.
- 3. Minimum Cementitious Materials Content: 470 lb/cu. yd..
- 4. Slump Limit: 4 inches plus or minus 1 inch.
- 5. Air Content: Do not allow air content of trowel-finished floors to exceed 3 percent.

- C. Suspended Slabs: Normal-weight concrete.
 - 1. Minimum Compressive Strength: 4000 psi.
 - 2. Maximum W/C Ratio: 0.45.
 - 3. Minimum Cementitious Materials Content: 520 lb/cu. yd..
 - 4. Slump Limit: 4 inches, 8 inches for concrete with verified slump of 2 to 4 inches before adding high-range water-reducing admixture or plasticizing admixture, plus or minut 1 inch.
 - 5. Air Content: Do not allow air content of trowel-finished floors to exceed 3 percent
 - 6. Synthetic Macro-Fiber: Uniformly disperse in concrete mixture at manufact recommended rate, but not less than a rate of 3.0 lb/cu. yd..

2.14 FABRICATING REINFORCEMENT

A. Fabricate steel reinforcement according to CRSI's "Manual of Standard Practice.

2.15 CONCRETE MIXING

- A. Ready-Mixed Concrete: Measure, batch, mix, and deliver concrete according to ASTM C 94 and ASTM C 1116, and furnish batch ticket information.
 - 1. When air temperature is between 85 and 91 deer F, reduce mixing and delivery time from 1-1/2 hours to 75 minutes; when air tem, erature is above 90 deg F, reduce mixing and delivery time to 60 minutes.

PART 3 - EXECUTION

1. 2.

- 3.1 FORMWORK INSTACLATIC
 - A. Design, erect, shore, brace, and maintain formwork, according to ACI 301, to support vertical, lateral, static, and dynamic loads, and construction loads that might be applied, until structure can support such loads.
 - B. Construct forp work so concrete members and structures are of size, shape, alignment, elevation, and position indicated, within tolerance limits of ACI 117.

Limit concrete surface irregularities, designated by ACI 347 as abrupt or gradual, as follows:

- Class A, 1/8 inch for smooth-formed finished surfaces.
- Class B, 1/4 inch for rough-formed finished surfaces.

Construct forms tight enough to prevent loss of concrete mortar.

- E. Construct forms for easy removal without hammering or prying against concrete surfaces. Provide crush or wrecking plates where stripping may damage cast-concrete surfaces. Provide top forms for inclined surfaces steeper than 1.5 horizontal to 1 vertical.
 - 1. Install keyways, reglets, recesses, and the like, for easy removal.

- 2. Do not use rust-stained steel form-facing material.
- Set edge forms, bulkheads, and intermediate screed strips for slabs to achieve required F. elevations and slopes in finished concrete surfaces. Provide and secure units to support screed strips; use strike-off templates or compacting-type screeds.
- Provide temporary openings for cleanouts and inspection ports where interior area of formwo G. is inaccessible. Close openings with panels tightly fitted to forms and securely braced to p loss of concrete mortar. Locate temporary openings in forms at inconspicuous locations
- H. Chamfer exterior corners and edges of permanently exposed concrete.
- Form openings, chases, offsets, sinkages, keyways, reglets, blocking, screets, and I. bulkheads required in the Work. Determine sizes and locations from trades providing such
- J. Clean forms and adjacent surfaces to receive concrete. Remove chos, wood sawdust, dirt, and other debris just before placing concrete.
- K. Retighten forms and bracing before placing concrete, as required, b prevent mortar leaks and maintain proper alignment.
- Coat contact surfaces of forms with form-releasengent, according to manufacturer's written L. instructions, before placing reinforcement.

EMBEDDED ITEM INSTALLATION 3.2

- Place and secure anchorage device and ther embedded items required for adjoining work that A. is attached to or supported by case in place concrete. Use setting drawings, templates, diagrams, instructions, and directions family with items to be embedded.
 - Install anchor room accurately located, to elevations required and complying with tolerances in Section 15 of AISC 303. Install dometail anchor slots in concrete structures as indicated. 1.
 - 2.

REMOVING AND REUSING FORMS 3.3

- Geveral: Formwork for sides of beams, walls, columns, and similar parts of the Work that does A. not support weight of concrete may be removed after cumulatively curing at not less than 50 deg F for 24 hours after placing concrete. Concrete has to be hard enough to not be damaged by rm-removal operations, and curing and protection operations need to be maintained.
 - Remove forms only if shores have been arranged to permit removal of forms without loosening or disturbing shores.
 - Clean and repair surfaces of forms to be reused in the Work. Split, frayed, delaminated, or otherwise damaged form-facing material are not acceptable for exposed surfaces. Apply new form-release agent.
- C. When forms are reused, clean surfaces, remove fins and laitance, and tighten to close joints. Align and secure joints to avoid offsets. Do not use patched forms for exposed concrete surfaces unless approved by Architect.
3.4 SHORING AND RESHORING INSTALLATION

- A. Comply with ACI 318 and ACI 301 for design, installation, and removal of shoring and reshoring.
 - 1. Do not remove shoring or reshoring until measurement of slab tolerances is complete.
- B. In multistory construction, extend shoring or reshoring over a sufficient number of stores to distribute loads in such a manner that no floor or member will be excessively loaded or will induce tensile stress in concrete members without sufficient steel reinforcement.
- C. Plan sequence of removal of shores and reshore to avoid damage to connete. Let and provide adequate reshoring to support construction without excessive stress or deflection.

3.5 VAPOR-BARRIER INSTALLATION

- A. Sheet Vapor Barriers: Place, protect, and repair sheet vapor barrier according to ASTM E 1643 and manufacturer's written instructions.
 - 1. Lap joints 6 inches and seal with manufacturer's recommended tape.

3.6 STEEL REINFORCEMENT INSTALLATION

- A. General: Comply with CRSI's "Manual of Standard Practice" for fabricating, placing, and supporting reinforcement.
 - 1. Do not cut or puncture report barrier. Repair damage and reseal vapor barrier before placing concrete.
- B. Clean reinforcement or hose rust and mill scale, earth, ice, and other foreign materials that reduce bond to concrete
- C. Accurately position, support, and secure reinforcement against displacement. Locate and support reinforcement with bar supports to maintain minimum concrete cover. Do not tack weld crossing reinforcing bars.

1. Weld reinforcing bars according to AWS D1.4/D 1.4M, where indicated.

Set wire ties with ends directed into concrete, not toward exposed concrete surfaces.

Zinc-Coated Reinforcement: Repair cut and damaged zinc coatings with zinc repair material according to ASTM A 780/A 780M. Use galvanized-steel wire ties to fasten zinc-coated steel reinforcement.

JOINTS

37

A. General: Construct joints true to line with faces perpendicular to surface plane of concrete.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- B. Construction Joints: Install so strength and appearance of concrete are not impaired, at locations indicated or as approved by Architect.
 - 1. Place joints perpendicular to main reinforcement. Continue reinforcement across construction joints unless otherwise indicated. Do not continue reinforcement through sides of strip placements of floors and slabs.
 - 2. Form keyed joints as indicated. Embed keys at least 1-1/2 inches into concrete.
 - 3. Locate horizontal joints in walls and columns at the top of footings or floor slabs.
 - 4. Space vertical joints in walls as indicated. Locate joints beside piers integral with v near corners, and in concealed locations where possible.
 - 5. Use a bonding agent at locations where fresh concrete is placed against hardered or partially hardened concrete surfaces.
 - 6. Use epoxy-bonding adhesive at locations where fresh concrete is placed against hardened or partially hardened concrete surfaces.
- C. Contraction/Control Joints in Slabs-on-Grade: Form weakened-plane contraction/control joints, sectioning concrete into areas as approved in Contractor Landshed Slab on Grade Contraction/Control Joint Layout Shop Drawings. Construct contraction/control joints for a depth equal to at least one-fourth of concrete thickness as follows:
 - 1. Grooved Joints: Form contraction/control joints after initial floating by grooving and finishing each edge of joint to a particle of 1/8 inch. Repeat grooving of contraction/control joints after applying surface traishes. Eliminate groover tool marks on concrete surfaces.
 - 2. Sawed Joints: Form contraction control joints with power saws equipped with shatterproof abrasive or diamond timbed blades. Cut 1/8-inch-wide joints into concrete when cutting action does not tear, abrade, or otherwise damage surface and before concrete develops random contraction cracks.
 - 3. Install Contraction/Control Johns at maximum 15 feet in on center in both directions. Install on column lines wherever possible.
- D. Isolation Joints in Skos- a Grade: After removing formwork, install joint-filler strips at slab junctions with vertical surfaces, such as column pedestals, foundation walls, grade beams, and other locations as indicated.
 - 1. Extend iont-filler strips full width and depth of joint, terminating flush with finished concrete surface unless otherwise indicated.
 - 2. Tensingle full-width joint-filler strips not less than 1/2 inch or more than 1 inch below innished concrete surface where joint sealants, specified in Section 07 92 00 "Joint sealants," are indicated.

Install joint-filler strips in lengths as long as practicable. Where more than one length is required, lace or clip sections together.

Doweled Joints: Install dowel bars and support assemblies at joints where indicated. Lubricate or asphalt coat one-half of dowel length to prevent concrete bonding to one side of joint.

3.8 CONCRETE PLACEMENT

A. Before placing concrete, verify that installation of formwork, reinforcement, and embedded items is complete and that required inspections are completed.

- B. Do not add water to concrete during delivery, at Project site, or during placement unless approved by Architect.
- C. Before test sampling and placing concrete, water may be added at Project site, subject to limitations of ACI 301.
 - Do not add water to concrete after adding high-range water-reducing admixtures to 1. mixture.
- Deposit concrete continuously in one layer or in horizontal layers of such thickness mat no D. concrete is placed on concrete that has hardened enough to cause seams or planes weatness. If a section cannot be placed continuously, provide construction joints as indicated beposit concrete to avoid segregation.
 - 1. Deposit concrete in horizontal layers of depth not to exceed form rk design pressures and in a manner to avoid inclined construction joints.
 - 2.
 - Consolidate placed concrete with mechanical vibrating equipment according to ACI 301. Do not use vibrators to transport concrete inside forms. Insert and withdraw vibrators 3. vertically at uniformly spaced locations to rapidly pendtrate placed layer and at least 6 inches into preceding layer. Do not insert vibrato s into lower layers of concrete that have begun to lose plasticity. At each insertion, limit due tion of vibration to time necessary to consolidate concrete and complete embeddent of reinforcement and other embedded items without causing mixture constituent, to segregate.
- Deposit and consolidate concrete for floor and sabs in a continuous construction joints, until placement of a panel or section is complete. E. d tabs in a continuous operation, within limits of
 - Consolidate concrete during placement operations, so concrete is thoroughly worked 1. around reinforcement and other embedded items and into corners.
 - Maintain reinforcement in polition on chairs during concrete placement. 2.
 - Screed slab surfaces with araightedge and strike off to correct elevations. 3.
 - Slope surfaces mit real to drains where required. 4.
 - Begin initial flowing using bull floats or darbies to form a uniform and open-textured surface plane, before excess bleedwater appears on the surface. Do not further disturb 5. of ore starting finishing operations. slab surface

G FORMED SURFACES 3.9 FIN'SHI

> Rough Formed Finish: As-cast concrete texture imparted by form-facing material with the holes and defects repaired and patched. Remove fins and other projections that exceed specified limits o formed-surface irregularities.

- Apply to concrete surfaces not exposed to public view.
- Smooth-Formed Finish: As-cast concrete texture imparted by form-facing material, arranged in an orderly and symmetrical manner with a minimum of seams. Repair and patch tie holes and defects. Remove fins and other projections that exceed specified limits on formed-surface irregularities.

1.

- 1. Apply to concrete surfaces exposed to public view; to receive a rubbed finish; or to be covered with a coating or covering material applied directly to concrete.
- C. Rubbed Finish: Apply the following to smooth-formed-finished as-cast concrete where indicated:
 - 1. Smooth-Rubbed Finish: Not later than one day after form removal, moisten concrete surfaces and rub with carborundum brick or another abrasive until producing a unform color and texture. Do not apply cement grout other than that created by the lubbing process.
- D. Related Unformed Surfaces: At tops of walls, horizontal offsets, and similar unforced ourfaces adjacent to formed surfaces, strike off smooth and finish with a texture matching adjacent formed surfaces. Continue final surface treatment of formed surfaces upformally across adjacent unformed surfaces unless otherwise indicated.

3.10 FINISHING FLOORS AND SLABS

- A. General: Comply with ACI 302.1R recommendations for screeding, restraightening, and finishing operations for concrete surfaces. Do not wet concrete surfaces.
- B. Scratch Finish: While still plastic, texture concrete surface that has been screeded and bullfloated or darbied. Use stiff brushes, brooms, while so produce a profile amplitude of 1/4 inch in one direction.
 - 1. Apply scratch finish to surfaces to receive concrete floor toppings.
- C. Trowel Finish: After applying heat finish, apply first troweling and consolidate concrete by hand or power-driven trowel. Continue troweling passes and restraighten until surface is free of trowel marks and uniform in texture and appearance. Grind smooth any surface defects that would telegraph through applied coatings or floor coverings.
 - 1. Apply a travel finite to surfaces exposed to view or to be covered with resilient flooring, carpet, cerabic or quarry tile set over a cleavage membrane, paint, or another thin-film-finite soating system.
 - 2. Finish surfaces to the following tolerances, according to ASTM E 1155, for a randomly trafficked floor surface:

Specified overall values of flatness, F(F) 35; and of levelness, F(L) 25; with minimum local values of flatness, F(F) 24; and of levelness, F(L) 17; for slabs-on-grade.

Broom Finish: Apply a broom finish to exterior concrete platforms, steps, ramps, and elsewhere as indicated.

1. Immediately after float finishing, slightly roughen trafficked surface by brooming with fiber-bristle broom perpendicular to main traffic route. Coordinate required final finish with Architect before application.

3.11 MISCELLANEOUS CONCRETE ITEM INSTALLATION

- A. Filling In: Fill in holes and openings left in concrete structures after work of other trades is in place unless otherwise indicated. Mix, place, and cure concrete, as specified, to blend with inplace construction. Provide other miscellaneous concrete filling indicated or required to complete the Work.
- B. Curbs: Provide monolithic finish to interior curbs by stripping forms while concrete green and by steel-troweling surfaces to a hard, dense finish with corners, intersection terminations slightly rounded.
- C. **Equipment Bases and Foundations:**
 - Coordinate sizes and locations of concrete bases with actual equiption p 1.
 - Construct concrete bases 6 inches high unless otherwise indicated and extend base not 2. less than 6 inches in each direction beyond the maximum dimensions of supported equipment unless otherwise indicated or unless required for stignic anchor support.
 - 3.
 - Minimum Compressive Strength: 4000 psi at 28 days. Install dowel rods to connect concrete base to concrete floor. Unless otherwise indicated, 4. install dowel rods on 18-inch centers around the fall parimeter of concrete base.
 - For supported equipment, install epoxy-coated and or bolts that extend through concrete 5. base and anchor into structural concrete substrate.
 - Prior to pouring concrete, place and secure an horage devices. Use setting drawings, templates, diagrams, instructions, and direction furnished with items to be embedded. 6.
 - tell unchor bolts to elevations required for proper 7. Cast anchor-bolt insert into bases. attachment to supported equipment

CONCRETE PROTECTING 3.12

- General: Protect freshly placed oncrete from premature drying and excessive cold or hot A. temperatures. Comply with ACI 306.1 for cold-weather protection and ACI 301 for hot-weather protection during curin,
- Evaporation Ketarier: Apply evaporation retarder to unformed concrete surfaces if hot, dry, or Β. windy conditions cause moisture loss approaching 0.2 lb/sq. ft. x h before and during finishing operations. Apply according to manufacturer's written instructions after placing, screeding, and bull foating or darbying concrete, but before float finishing.
- Formel Surfaces: Cure formed concrete surfaces. If forms remain during curing period, moist cure after loosening forms. If removing forms before end of curing period, continue curing for remainder of curing period.

Unformed Surfaces: Begin curing immediately after finishing concrete. Cure unformed surfaces, including floors and slabs, concrete floor toppings, and other surfaces.

- Cure concrete according to ACI 308.1, by one or a combination of the following methods:
 - 1. Moisture Curing: Keep surfaces continuously moist for not less than seven days with the following materials:

CAST-IN-PLACE CONCRETE 03 30 00 - 16

E.

- a. Water.
- b. Continuous water-fog spray.
- c. Absorptive cover, water saturated, and kept continuously wet. Cover concrete surfaces and edges with 12-inch lap over adjacent absorptive covers.
- 2. Moisture-Retaining-Cover Curing: Cover concrete surfaces with moisture-retaining cover for curing concrete, placed in widest practicable width, with sides and ends lapped at least 12 inches, and sealed by waterproof tape or adhesive. Cure for not less than seven days. Immediately repair any holes or tears during curing period, using cover material and waterproof tape.
 - a. Moisture cure or use moisture-retaining covers to cure concrete surfaces to receive floor coverings.
 - b. Moisture cure or use moisture-retaining covers to cure concrete surfaces to receive penetrating liquid floor treatments.
 - c. Cure concrete surfaces to receive floor coverings with either a moisture-retaining cover or a curing compound that the manufacturer tend does not interfere with bonding of floor covering used on Project.
- 3. Curing Compound: Apply uniformly in continuous operation by power spray or roller according to manufacturer's written instructions. Repeat areas subjected to heavy rainfall within three hours after initial application. Maintain continuity of coating and repair damage during curing period.
 - a. Removal: After curing period bay elapsed, remove curing compound without damaging concrete surfaces av method recommended by curing compound manufacturer unless panulactorer certifies curing compound does not interfere with bonding of floor overing used on Project.
- 4. Curing and Sealing Compound: Apply uniformly to floors and slabs indicated in a continuous operation by cower spray or roller according to manufacturer's written instructions. Recompress subjected to heavy rainfall within three hours after initial application. Report pocess 24 hours later and apply a second coat. Maintain continuity of coating and repair damage during curing period.

3.13 LIQUID FLOOR TREATMENT APPLICATION

- A. Penetraing Liquid Floor Treatment: Prepare, apply, and finish penetrating liquid floor treatment according to manufacturer's written instructions.
- 2.
- Remove curing compounds, sealers, oil, dirt, laitance, and other contaminants and complete surface repairs.
- Apply liquid until surface is saturated, scrubbing into surface until a gel forms; rewet; and repeat brooming or scrubbing. Rinse with water; remove excess material until surface is dry. Apply a second coat in a similar manner if surface is rough or porous.
- B. Sealing Coat: Uniformly apply a continuous sealing coat of curing and sealing compound to hardened concrete by power spray or roller according to manufacturer's written instructions.

3.14 JOINT FILLING

- A. Prepare, clean, and install joint filler according to manufacturer's written instructions.
 - 1. Defer joint filling until concrete has aged at least two month(s). Do not fill joints until construction traffic has permanently ceased.
- B. Remove dirt, debris, saw cuttings, curing compounds, and sealers from joints; leave cont faces of joints clean and dry.
- C. Install semirigid joint filler full depth in saw-cut joints and at least 2 inches deep in firmed joints. Overfill joint and trim joint filler flush with top of joint after hardening

3.15 CONCRETE SURFACE REPAIRS

- A. Defective Concrete: Repair and patch defective areas when approved by Architect. Remove and replace concrete that cannot be repaired and patched to Architect's approval.
- B. Patching Mortar: Mix dry-pack patching mortar, consisting of 1 part portland cement to 2-1/2 parts fine aggregate passing a No. 16 sieve, using only enough water for handling and placing.
- C. Repairing Formed Surfaces: Surface defects include color and texture irregularities, cracks, spalls, air bubbles, honeycombs, rock pockes, fins and other projections on the surface, and stains and other discolorations that cannot be removed by cleaning.
 - 1. Immediately after form removal currous honeycombs, rock pockets, and voids more than 1/2 inch in any dimension to solid concrete. Limit cut depth to 3/4 inch. Make edges of cuts perpendicular to concrete surface. Clean, dampen with water, and brush-coat holes and voids with bonding agent. Fill and compact with patching mortar before bonding agent has dried. Fill formatic voids with patching mortar or cone plugs secured in place with bonding agent.
 - 2. Repair defects on surfaces exposed to view by blending white portland cement and standard restland cement so that, when dry, patching mortar matches surrounding color. Patch a test area at inconspicuous locations to verify mixture and color match before proceeding with patching. Compact mortar in place and strike off slightly higher than surrounding surface.
 - 3. Recair defects on concealed formed surfaces that affect concrete's durability and ructural performance as determined by Architect.

Repairing Unformed Surfaces: Test unformed surfaces, such as floors and slabs, for finish and wrify surface tolerances specified for each surface. Correct low and high areas. Test surfaces sloped to drain for trueness of slope and smoothness; use a sloped template.

- 1. Repair finished surfaces containing defects. Surface defects include spalls, popouts, honeycombs, rock pockets, crazing and cracks in excess of 0.01 inch wide or that penetrate to reinforcement or completely through unreinforced sections regardless of width, and other objectionable conditions.
- 2. After concrete has cured at least 14 days, correct high areas by grinding.

- 3. Correct localized low areas during or immediately after completing surface finishing operations by cutting out low areas and replacing with patching mortar. Finish repaired areas to blend into adjacent concrete.
- 4. Correct other low areas scheduled to receive floor coverings with a repair underlayment. Prepare, mix, and apply repair underlayment and primer according to manufacturer's written instructions to produce a smooth, uniform, plane, and level surface. Feather edge to match adjacent floor elevations.
- 5. Correct other low areas scheduled to remain exposed with a repair topping. Cut cat low areas to ensure a minimum repair topping depth of 1/4 inch to match adjacent ploor elevations. Prepare, mix, and apply repair topping and primer according to manufacture s written instructions to produce a smooth, uniform, plane, and level surface.
- 6. Repair defective areas, except random cracks and single holes 1 inch or less in diameter, by cutting out and replacing with fresh concrete. Remove defective areas with clean, square cuts and expose steel reinforcement with at least a 3/4-in m clearance all around. Dampen concrete surfaces in contact with patching concrete and opply bonding agent. Mix patching concrete of same materials and mixture as original concrete, except without coarse aggregate. Place, compact, and finish to blend with edjacent finished concrete. Cure in same manner as adjacent concrete.
- 7. Repair random cracks and single holes 1 inch or loss in diameter with patching mortar. Groove top of cracks and cut out holes to sound concrete and clean off dust, dirt, and loose particles. Dampen cleaned concrete surfaces and apply bonding agent. Place patching mortar before bonding agent has dried. Compact patching mortar and finish to match adjacent concrete. Keep patched are continuously moist for at least 72 hours.
- E. Perform structural repairs of concrete, subject to chitect's approval, using epoxy adhesive and patching mortar.
- F. Repair materials and installation not specified above may be used, subject to Architect's approval.
- 3.16 FIELD QUALITY CONTROL
 - A. Special Inspection: Owner will engage a special inspector and qualified testing and inspecting agency to perform field tests and inspections and prepare test reports.
 - B. Testing Agence: Engage a qualified testing and inspecting agency to perform tests and inspection and to submit reports.
 - C. Inspections:
 - Steel reinforcement placement.
 - 2. Steel reinforcement welding.
 - 3. Headed bolts and studs.
 - 4. Verification of use of required design mixture.
 - 5. Concrete placement, including conveying and depositing.
 - 6. Curing procedures and maintenance of curing temperature.
 - 7. Verification of concrete strength before removal of shores and forms from beams and slabs.

- D. Concrete Tests: Testing of composite samples of fresh concrete obtained according to ASTM C 172 shall be performed according to the following requirements:
 - 1. Testing Frequency: Obtain one composite sample for each day's pour of each concrete mixture exceeding 5 cu. yd., but less than 25 cu. yd., plus one set for each additional 50 cu. yd. or fraction thereof.
 - 2. Testing Frequency: Obtain at least one composite sample for each 100 cu. yd. or fraction thereof of each concrete mixture placed each day.
 - a. When frequency of testing provides fewer than five compressive-strength tests for each concrete mixture, testing shall be conducted from at least five randomly selected batches or from each batch if fewer than five are used.
 - 3. Slump: ASTM C 143; one test at point of placement for each composite sample, but not less than one test for each day's pour of each concrete mixture. For orm additional tests when concrete consistency appears to change.
 - 4. Concrete Temperature: ASTM C 1064; one test hourly when ar temperature is 40 deg F and below or 80 deg F and above, and one test for each composite sample.
 - 5. Compression Test Specimens: ASTM C 31.
 - a. Cast and laboratory cure two sets of two candard cylinder specimens for each composite sample.
 - b. Cast and field cure two sets of two tandahl cylinder specimens for each composite sample.
 - 6. Compressive-Strength Tests: ASSM C 39/C 39M; test one set of two laboratory-cured specimens at 7 days and one set of two specimens at 28 days.
 - a. Test one set of two field cared specimens at 7 days and one set of two specimens at 28 days.
 - b. A compressive strength test shall be the average compressive strength from a set of two specime coblained from same composite sample and tested at age indicated.
 - 7. When strength of held-cured cylinders is less than 85 percent of companion laboratorycured cylinders. Contractor shall evaluate operations and provide corrective procedures for acceeding and curing in-place concrete.
 - 8. Stengthof each concrete mixture will be satisfactory if every average of any three concernive compressive-strength tests equals or exceeds specified compressive strength and no compressive-strength test value falls below specified compressive strength by nore than 500 psi.

Test results shall be reported in writing to Architect, concrete manufacturer, and Contractor within 48 hours of testing. Reports of compressive-strength tests shall contain Project identification name and number, date of concrete placement, name of concrete testing and inspecting agency, location of concrete batch in Work, design compressive strength at 28 days, concrete mixture proportions and materials, compressive breaking strength, and type of break for both 7- and 28-day tests.

- 10. Nondestructive Testing: Impact hammer, sonoscope, or other nondestructive device may be permitted by Architect but will not be used as sole basis for approval or rejection of concrete.
- 11. Additional Tests: Testing and inspecting agency shall make additional tests of concrete when test results indicate that slump, air entrainment, compressive strengths, or other



Tetra Tech 200-15704-17001

requirements have not been met, as directed by Architect. Testing and inspecting agency may conduct tests to determine adequacy of concrete by cored cylinders complying with ASTM C 42 or by other methods as directed by Architect.

- 12. Additional testing and inspecting, at Contractor's expense, will be performed to determine compliance of replaced or additional work with specified requirements.
- 13. Correct deficiencies in the Work that test reports and inspections indicate do not complex with the Contract Documents.
- E. Measure floor and slab flatness and levelness according to ASTM E 1155 within 24 finishing.

3.17 PROTECTION OF LIQUID FLOOR TREATMENTS

RBIDD

A. Protect liquid floor treatment from damage and wear during the remainder of construction period. Use protective methods and materials, including temporary covering, recommended in writing by liquid floor treatments installer.

END OF SECTION

SPEC SECTION 04 72 00

CAST STONE MASONRY

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Cast stone trim including the following:
 - a. Window sills.

1.3 SUBMITTALS, GENERAL

A. General: Submit all action submittals (except Samples for Verification) and informational submittals required by this Section concurrently.

1.4 ACTION SUBMITTALS

- A. Product Data: For each type of product indicated.
- B. Shop Drawings: Show fabrication and installation details for cast stone units. Include dimensions, devias or reinforcement and anchorages if any, and indication of finished faces.
 - 1. Include huilding elevations showing layout of units and locations of joints and anchors.
- C. Samples for Verification:
 - For each color and texture of cast stone required, 10 inches square in size.

Full-Size Samples: For each shape of cast stone unit required.

- 1. Make available for Architect's review at Project site.
- 2. Make Samples from materials to be used for units used on Project.
- 3. Approved Samples may be installed in the Work.

1.5 INFORMATIONAL SUBMITTALS

- A. Qualification Data: For manufacturer.
- B. Material Test Reports: For each mix required to produce cast stone, based on testing according to ASTM C 1364, including test for resistance to freezing and thawing.
 - 1. Provide test reports based on testing within previous two years.

1.6 QUALITY ASSURANCE

- A. Manufacturer Qualifications: A qualified manufacturer of cast stone unit similar to those indicated for this Project, that has sufficient production capacity to manufacture required units, and is a plant certified by the Cast Stone Institute.
- B. Source Limitations for Cast Stone: Obtain cast stone units through single source from single manufacturer.
- C. Mockups: Furnish cast stone for installation in mockup specified in Division 04 Section "Unit Masonry."
- 1.7 DELIVERY, STORAGE, AND HANDLING
 - A. Coordinate delivery of cast stone with unit reasonry work.
 - B. Pack, handle, and ship cast stone wits in uitable packs or pallets.
 - 1. Lift with wide-belt slives; do not use wire rope or ropes that might cause staining. Move cast stone units, if required, using dollies with wood supports.
 - 2. Store cast stone units on wood skids or pallets with nonstaining, waterproof covers, securely tied. Wrange to distribute weight evenly and to prevent damage to units. Ventilate under covers to prevent condensation.
- PART 2 PRODUCTS

2.1

CASESTONE MATERIALS

eneral: Comply with ASTM C 1364 and the following:

Portland Cement: ASTM C 150, Type I or Type III, containing not more than 0.60 percent total alkali when tested according to ASTM C 114. Provide natural color or white cement as required to produce cast stone color indicated.

- C. Coarse Aggregates: Granite, quartz, or limestone complying with ASTM C 33; gradation and colors as needed to produce required cast stone textures and colors.
- D. Fine Aggregates: Natural sand or crushed stone complying with ASTM C 33, gradation and colors as needed to produce required cast stone textures and colors.

- E. Color Pigment: ASTM C 979, synthetic mineral-oxide pigments or colored water-reducing admixtures; color stable, nonfading, and resistant to lime and other alkalis.
- F. Admixtures: Use only admixtures specified or approved in writing by Architect.
 - 1. Do not use admixtures that contain more than 0.1 percent water-soluble chloride ions be mass of cementitious materials. Do not use admixtures containing calcium chloride.
 - 2. Use only admixtures that are certified by manufacturer to be compatible with cement and other admixtures used.
 - 3. Air-Entraining Admixture: ASTM C 260. Add to mixes for units exposed to the exterior at manufacturer's prescribed rate to result in an air content of 4 to 6 percent, except to not add to zero-slump concrete mixes.
 - 4. Water-Reducing Admixture: ASTM C 494/C 494M, Type A.
 - 5. Water-Reducing, Retarding Admixture: ASTM C 494/C 494M, T/pe
 - 6. Water-Reducing, Accelerating Admixture: ASTM C 494/C 494M, Type E.
- G. Reinforcement: Deformed steel bars complying with ASTM 4 647/A 615M, Grade 60. Use galvanized or epoxy-coated reinforcement when covered with less than 1-1/2 inches of cast stone material.
 - 1. Epoxy Coating: ASTM A 775/A 775M.
 - 2. Galvanized Coating: ASTM A 767/A 767
- H. Embedded Anchors and Other Inserts: Publicated from stainless steel complying with ASTM A 240/A 240M, ASTM A 276, or CCTM A 666, Type 304.

2.2 CAST STONE UNITS

- A. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to, the following:
 - 1. Cast Stone Systems Inc
 - 2. Continent d Cast Stone East by Russell.
 - 3. Corinth an Cost Stone Inc.
 - 4. Howle Store Products.
 - 5. Sun Precust Company.
- B. Regional Materials: Cast stone units shall be manufactured within 500 miles of Project site from aggregates and cement that have been extracted, harvested, or recovered, as well as manufactured, within 500 miles of Project site.

Provide cast stone units complying with ASTM C 1364 using either the vibrant dry tamp or wet-cast method.

- 1. Provide units that are resistant to freezing and thawing as determined by laboratory testing according to ASTM C 666/C 666M, Procedure A, as modified by ASTM C 1364.
- D. Fabricate units with sharp arris and accurately reproduced details, with indicated texture on all exposed surfaces unless otherwise indicated.

- 1. Slope exposed horizontal surfaces 1:12 to drain unless otherwise indicated.
- 2. Provide drips on projecting elements unless otherwise indicated.
- Fabrication Tolerances: E.
 - 1. Variation in Cross Section: Do not vary from indicated dimensions by more than 1 inch.
 - 2. Variation in Length: Do not vary from indicated dimensions by more than 1/360 of the length of unit or 1/8 inch, whichever is greater, but in no case by more than 1/4 in
 - 3. Warp, Bow, and Twist: Not to exceed 1/360 of the length of unit or 1/8 incl., which is greater.
 - Location of Grooves, False Joints, Holes, Anchorages, and Similar Features: Do 4. not varv from indicated position by more than 1/8 inch on formed surfaces of units and 3/8 inch on unformed surfaces.
- F. Cure units as follows:
 - Cure units in enclosed moist curing room at 95 to 100 percent relative humidity and 1. temperature of 100 deg F for 12 hours or 70 deg F for 1 hour
 - Keep units damp and continue curing to comply with the of the following: 2.
 - a. No fewer than five days at mean dail, remperature of 70 deg F or above.
 - b.
 - No fewer than six days at mean daily temperature of 60 deg F or above. No fewer than seven days at mean daily temperature of 50 deg F or above. с.
 - d. No fewer than eight days at e doily temperature of 45 deg F or above.
- ment film from surfaces to be exposed to view. Acid etch units after curing to remove G.
- H. Color and Texture: Provide ine-grained texture and buff color resembling Indiana limestone.
- 2.3 MORTAR MATERIA
 - erials that comply with Division 04 Section "Unit Masonry." A. Provide morta ma
- 2.4 ACCES

A

- Anchers: Type and size indicated, fabricated from Type 304 stainless steel complying with ASTM A 240/A 240M, ASTM A 276, or ASTM A 666.
 - Dowels: 1/2-inch-diameter, round bars, fabricated from Type 304 stainless steel complying with ASTM A 240/A 240M, ASTM A 276, or ASTM A 666.
- MORTAR MIXES
- A. Comply with requirements in Division 04 Section "Unit Masonry" for mortar mixes.
- Β. Comply with ASTM C 270, Proportion Specification.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- 1. For setting mortar, use Type N.
- For pointing mortar, use Type N. 2.

PART 3 - EXECUTION

3.1 **EXAMINATION**

- Examine substrates and conditions, with Installer present, for compliance with requirem A. installation tolerances and other conditions affecting performance of work.
- Proceed with installation only after unsatisfactory conditions have been corrected. B.
- Beginning installation constitutes Contractor's acceptance of substrates C. conditions.

3.2 SETTING CAST STONE IN MORTAR

- A. Install cast stone units to comply with requirements in Divisi ection "Unit Masonry."
- Set cast stone as indicated on Drawings. Set units accurately in locations indicated with edges Β. and faces aligned according to established relationships and indicated tolerances.
 - and other attachments indicated or necessary to Install anchors, supports, fasteners 1. secure units in place.
 - with installation of flashing specified in other 2. Coordinate installation of c Sections.
- Wet joint surfaces thoroughly before applying mortar or setting in mortar. C.
- D. Set units in full bed of no tar with full head joints unless otherwise indicated.
 - Set units with joints //4 to 3/8 inch wide unless otherwise indicated. 1.
 - Build archo's and ties into mortar joints as units are set. 2.
 - Fill dowcholes and anchor slots with mortar. Fill collar joints solid as units are set. 3.
 - 4.
 - Build concealed flashing into mortar joints as units are set. 5.
 - Keep nead joints in coping and other units with exposed horizontal surfaces open to 6. receive sealant.



ake out joints for pointing with mortar to depths of not less than 3/4 inch. Rake joints to uniform depths with square bottoms and clean sides. Scrub faces of units to remove excess mortar as joints are raked.

Point mortar joints by placing and compacting mortar in layers not greater than 3/8 inch. Compact each layer thoroughly and allow it to become thumbprint hard before applying next layer.

G. Tool exposed joints slightly concave when thumbprint hard, using a jointer larger than joint thickness unless otherwise indicated.

- H. Provide sealant joints at copings and other horizontal surfaces, at expansion, control, and pressure-relieving joints, and at locations indicated.
 - 1. Keep joints free of mortar and other rigid materials.
 - 2. Build in compressible foam-plastic joint fillers where indicated.
 - 3. Form joint of width indicated, but not less than 3/8 inch.
 - 4. Prime cast stone surfaces to receive sealant and install compressible backer rod in joil ts before applying sealant unless otherwise indicated.
 - 5. Prepare and apply sealant of type and at locations indicated to comply with ap requirements in Division 07 Section "Joint Sealants."

3.3 INSTALLATION TOLERANCES

- A. Variation from Plumb: Do not exceed 1/8 inch in 10 feet, 1/4 inch in 20 feet, or 1/2 inch maximum.
- B. Variation from Level: Do not exceed 1/8 inch in 10 feet, 1/4 inch in 20 feet, or 1/2 inch maximum.
- C. Variation in Joint Width: Do not vary joint thickness nice than 1/8 inch in 36 inches or one-fourth of nominal joint width, whichever is less.
- D. Variation in Plane between Adjacent Surfaces (sipping): Do not vary from flush alignment with adjacent units or adjacent surfaces in licetee to be flush with units by more than 1/16 inch, except where variation is due to warpage of units within tolerances specified.

3.4 ADJUSTING AND CLEANING

- A. Remove and replace stained ad otherwise damaged units and units not matching approved Samples. Cast stone may be required if methods and results are approved by Architect.
- B. Replace units in commencement results in cast stone matching approved Samples, complying with other requirements and showing no evidence of replacement.
- C. In-Progress Cleaning: Clean cast stone as work progresses.
 - . Demove mortar fins and smears before tooling joints.
 - Remove excess sealant immediately, including spills, smears, and spatter.

END OF SECTION

SECTION 05 12 00

STRUCTURAL STEEL FRAMING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section Includes:
 - 1. Structural steel.
 - 2. Grout.
- B. Related Requirements:
 - 1. Section 05 50 00 "Metal Fabrications" for steel lintels and shelf angles not attached to structural-steel frame, miscellaneous steel fabrications and other steel items not defined as structural steel.
 - 2. Section 09 91 13 "Exterior Peinting" and Section 09 91 23 "Interior Painting" and for surface-preparation and printing requirements.
- 1.3 DEFINITIONS
 - A. Structural Steel: Elements of the structural frame indicated on Drawings and as described in AISC 303, "Code of Standard Practice for Steel Buildings and Bridges."

1.4 COOPDINATION

Α

Coordinate selection of shop primers with topcoats to be applied over them. Comply with paint and coaring manufacturers' written recommendations to ensure that shop primers and topcoats re compatible with one another.

Coordinate installation of anchorage items to be embedded in or attached to other construction without delaying the Work. Provide setting diagrams, sheet metal templates, instructions, and directions for installation.

- 1.5 PREINSTALLATION MEETINGS
 - A. Preinstallation Conference: Conduct conference at Project site.

1.6 ACTION SUBMITTALS

- A. Product Data: For each type of product.
- B. Shop Drawings: Show fabrication of structural-steel components.
 - 1. Include details of cuts, connections, splices, camber, holes, and other pertinent data.
 - 2. Include embedment Drawings.
 - 3. Indicate welds by standard AWS symbols, distinguishing between shop and field wilds, and show size, length, and type of each weld. Show backing bars that are to be rem weld and supplemental fillet welds where backing bars are to remain.
 - 4. Indicate type, size, and length of bolts, distinguishing between shor and field bolts. Identify pretensioned and slip-critical, high-strength bolted connections
- C. Welding Procedure Specifications (WPSs) and Procedure Qualification Records (PQRs): Provide according to AWS D1.1/D1.1M, "Structural Welding Cone - Steel," for each welded joint whether prequalified or qualified by testing, including the following:
 - 1. Power source (constant current or constant voltage)
 - 2. Electrode manufacturer and trade name, for demaid criticar welds.
- D. Delegated-Design Submittal: For structural-steel connections indicated to comply with design loads, include analysis data signed and sealed by the qualified professional engineer licensed in the State of Delaware, responsible for their propartion.

1.7 INFORMATIONAL SUBMITTAL

- A. Qualification Data: For Installer Abreator and testing agency.
- B. Welding certificates.
- C. Paint Compatibility Conficites: From manufacturers of topcoats applied over shop primers, certifying that shop primes are compatible with topcoats.
- D. Mill test reports or structural steel, including chemical and physical properties.
- E. Product Test Reports: For the following:
 - Bolts, nuts, and washers including mechanical properties and chemical analysis.
 - Direct-tension indicators.
 - Tension-control, high-strength, bolt-nut-washer assemblies.
 - Shop primers.
 - Nonshrink grout.
- F. Survey of existing conditions.

5

- G. Source quality-control reports.
- H. Field quality-control and special inspection reports.

1.8 QUALITY ASSURANCE

- A. Fabricator Qualifications: A qualified fabricator that participates in the AISC Quality Certification Program and is designated an AISC-Certified Plant, Category STD.
- B. Installer Qualifications: A qualified installer who participates in the AISC Quality Certification Program and is designated an AISC-Certified Erector, Category ACSE.
- C. Shop-Painting Applicators: Qualified according to AISC's Sophisticated Paint Endorsement P and Endorsement P2 or to SSPC-QP 3, "Standard Procedure for Evaluating Qualincations of Shop Painting Applicators."
- D. Welding Qualifications: Qualify procedures and personnel according to AWS D.1/D1.1M, "Structural Welding Code Steel."
 - 1. Welders and welding operators performing work on botom flame, demand-critical welds shall pass the supplemental welder qualification testing, as required by AWS D1.8/D1.8/D. FCAW-S and FCAW-G shall be considered separate processes for welding personnel qualification.
- E. Comply with applicable provisions of the following specifications and documents:
 - 1. AISC 303.
 - 2. AISC 341 and AISC 341s1.
 - 3. AISC 360.
 - 4. RCSC's "Specification for Structural Joints Using ASTM A 325 or A 490 Bolts."

1.9 DELIVERY, STORAGE, AND TANDLING

- A. Store materials to permit eavy access for inspection and identification. Keep steel members off ground and spaced by using ballets, dunnage, or other supports and spacers. Protect steel members and packaged naterials from corrosion and deterioration.
 - 1. Do not store materials on structure in a manner that might cause distortion, damage, or overlead o members or supporting structures. Repair or replace damaged materials or structure as directed.
- B. Store fisteners in a protected place in sealed containers with manufacturer's labels intact.



- Fasteners may be repackaged provided Owner's testing and inspecting agency observes repackaging and seals containers.
- Clean and relubricate bolts and nuts that become dry or rusty before use.
- Comply with manufacturers' written recommendations for cleaning and lubricating ASTM F 1852 fasteners and for retesting fasteners after lubrication.

PART 2 - PRODUCTS

2.1 PERFORMANCE REQUIREMENTS

- A. Connections: Provide details of connections required by the Contract Documents to be selected or completed by structural-steel fabricator, including comprehensive engineering analysis by a qualified professional engineer, to withstand loads indicated and comply with other information, and restrictions indicated.
 - 1. Select and complete connections using schematic details indicated and AISC
 - 2. Use Allowable Stress Design; data are given at service-load level.
- B. Moment Connections: Type PR, partially restrained.
- C. Construction: Combined system of moment frame and braced frame.

2.2 STRUCTURAL-STEEL MATERIALS

- A. W-Shapes: ASTM A 992/A 992M.
- B. Channels, Angles: ASTM A 572/A 572M, Grade 50
- C. Plate and Bar: ASTM A 36/A 36M.
- D. Cold-Formed Hollow Structural Sections: USIM A 500/A 500M, Grade C, structural tubing.
- E. Welding Electrodes: Comply with AVS equirements.

2.3 BOLTS, CONNECTORS AND ANCHORS

- A. High-Strength Bolts, Nuts, and Washers: ASTM A 325, Type 1, heavy-hex steel structural bolts; ASTM 4 56), Grade C, heavy-hex carbon-steel nuts; and ASTM F 436, Type 1, hardened carbon-steel wasters; all with plain finish.
 - 1. Direct-Tinsion Indicators: ASTM F 959, Type 325, compressible-washer type with plain finish.
- B. High-Strength Bolts, Nuts, and Washers: ASTM A 490, Type 1, heavy-hex steel structural olts; ASTM A 563, Grade DH, heavy-hex carbon-steel nuts; and ASTM F 436, Type 1, hardened carbon-steel washers with plain finish.
 - Direct-Tension Indicators: ASTM F 959, Type 490, compressible-washer type with plain finish.
- C. Tension-Control, High-Strength Bolt-Nut-Washer Assemblies: ASTM F 1852, Type 1, heavyhex head assemblies consisting of steel structural bolts with splined ends, heavy-hex carbonsteel nuts, and hardened carbon-steel washers.
 - 1. Finish: Plain.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- D. Unheaded Anchor Rods: ASTM F 1554, Grade 55, weldable.
 - 1. Configuration: Straight.
 - 2. Nuts: ASTM A 563 heavy-hex carbon steel.
 - 3. Plate Washers: ASTM A 36/A 36M carbon steel.
 - 4. Washers: ASTM F 436, Type 1, hardened carbon steel.
 - 5. Finish: Plain.
- E. Headed Anchor Rods: ASTM F 1554, Grade 55, weldable, straight.
 - 1. Nuts: ASTM A 563 heavy-hex carbon steel.
 - 2. Plate Washers: ASTM A 36/A 36M carbon steel.
 - 3. Washers: ASTM F 436, Type 1, hardened carbon steel.
 - 4. Finish: Plain.
- F. Threaded Rods: ASTM A 36/A 36M.
 - 1. Nuts: ASTM A 563 heavy-hex carbon steel.
 - 2. Washers: ASTM F 436, Type 1, hardened carbon stee
 - 3. Finish: Plain.
- G. Clevises and Turnbuckles: Made from cold finished arbon steel bars, ASTM A 108, Grade 1035.

2.4 PRIMER

- A. Primer: SSPC-Paint 25, Type I, and ovide, alkyd, linseed oil primer and compatible with topcoat.
- 2.5 GROUT
 - A. Nonmetallic, Shrinkaga Resistant Grout: ASTM C 1107/C 1107M, factory-packaged, nonmetallic argretate grout, noncorrosive and nonstaining, mixed with water to consistency suitable for appreadom and a 30-minute working time.

2.6 FASRICATION

Structural Steel: Fabricate and assemble in shop to greatest extent possible. Fabricate according AISC 303, "Code of Standard Practice for Steel Buildings and Bridges," and to AISC 360.

- Camber structural-steel members where indicated.
- 2. Fabricate beams with rolling camber up.
- 3. Identify high-strength structural steel according to ASTM A 6/A 6M and maintain markings until structural steel has been erected.
- 4. Mark and match-mark materials for field assembly.
- 5. Complete structural-steel assemblies, including welding of units, before starting shoppriming operations.
- B. Thermal Cutting: Perform thermal cutting by machine to greatest extent possible.

- 1. Plane thermally cut edges to be welded to comply with requirements in AWS D1.1/D1.1M.
- C. Bolt Holes: Cut, drill or punch standard bolt holes perpendicular to metal surfaces.
- D. Finishing: Accurately finish ends of columns and other members transmitting bearing loads.
- E. Cleaning: Clean and prepare steel surfaces that are to remain unpainted according to SP 3, "Power Tool Cleaning."
- F. Steel Wall-Opening Framing: Select true and straight members for fabricating cheel wallopening framing to be attached to structural-steel frame. Straighten as required to provide uniform, square, and true members in completed wall framing. Build up welled framing, weld exposed joints continuously, and grind smooth.
- G. Welded Door Frames: Build up welded door frames attached to structural steel frame. Weld exposed joints continuously and grind smooth. Plug-weld fixed steel bar stops to frames. Secure removable stops to frames with countersunk machine screws, unformly spaced not more than 10 inches o.c. unless otherwise indicated.
- H. Holes: Provide holes required for securing other work to structural steel and for other work to pass through steel members.
 - 1. Cut, drill, or punch holes perpendiculat to steel surfaces. Do not thermally cut bolt holes or enlarge holes by burning.
 - 2. Baseplate Holes: Cut, drill, mechannally thermal cut, or punch holes perpendicular to steel surfaces.
 - 3. Weld threaded nuts to framing and other specialty items indicated to receive other work.

2.7 SHOP CONNECTIONS

- A. High-Strength Bolts: Stop Install high-strength bolts according to RCSC's "Specification for Structural Jointe Using XSTM A 325 or A 490 Bolts" for type of bolt and type of joint specified.
 - 1. Joint Type: Snug tightened or Pretensioned.
- B. Wed connections: Comply with AWS D1.1/D1.1M for tolerances, appearances, welding procedure specifications, weld quality, and methods used in correcting welding work.

SHOP PRIMING

Shop prime steel surfaces except the following:

- 1. Surfaces embedded in concrete or mortar. Extend priming of partially embedded members to a depth of 2 inches.
- 2. Surfaces to be field welded.
- 3. Surfaces of high-strength bolted, slip-critical connections.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- B. Surface Preparation: Clean surfaces to be painted. Remove loose rust and mill scale and spatter, slag, or flux deposits. Prepare surfaces according to the following specifications and standards:
 - 1. SSPC-SP 3, "Power Tool Cleaning."
- C. Priming: Immediately after surface preparation, apply primer according to manufacturer's written instructions and at rate recommended by SSPC to provide a minimum dry film thickness of 1.5 mils. Use priming methods that result in full coverage of joints, corners, edges, and exposed surfaces.
 - 1. Stripe paint corners, crevices, bolts, welds, and sharp edges.
 - 2. Apply two coats of shop paint to surfaces that are inaccessible after assembly of election. Change color of second coat to distinguish it from first.
- D. Painting: Prepare steel and apply a one-coat, nonasphaltic primer complying with SSPC-PS Guide 7.00, "Painting System Guide 7.00: Guide for Selecting One-Coat Shop Painting Systems," to provide a dry film thickness of not less than 1.5 mile.

2.9 SOURCE QUALITY CONTROL

- A. Testing Agency: Owner will engage a qualified testing agency to perform shop tests and inspections.
 - 1. Provide testing agency with access to places where structural-steel work is being fabricated or produced to perform tests and inspections.
- B. Bolted Connections: Inspect and test shop-bolted connections according to RCSC's "Specification for Structural Joint, ASTM A 325 or A 490 Bolts."
- C. Welded Connections: Viscell, inspect shop-welded connections according to AWS D1.1/D1.1M and the following inspection procedures, at testing agency's option:
 - 1. Liquid Perstrant In pection: ASTM E 165.
 - 2. Magnet c Particle Inspection: ASTM E 709; performed on root pass and on finished weld. Cracks or zones of incomplete fusion or penetration are not accepted.
 - 3. U rason c Inspection: ASTM E 164.
 - 4. Raliographic Inspection: ASTM E 94.
- D. Prepare test and inspection reports.

- EXECUTION

EXAMINATION

A. Verify, with certified steel erector present, elevations of concrete- and masonry-bearing surfaces and locations of anchor rods, bearing plates, and other embedments for compliance with requirements.

Tetra Tech 200-15704-17001

- 1. Prepare a certified survey of existing conditions. Include bearing surfaces, anchor rods, bearing plates, and other embedments showing dimensions, locations, angles, and elevations.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.

3.2 PREPARATION

A. Provide temporary shores, guys, braces, and other supports during erection to keep structual steel secure, plumb, and in alignment against temporary construction loads and loads equal in intensity to design loads. Remove temporary supports when permanent structure steel, connections, and bracing are in place unless otherwise indicated.

3.3 ERECTION

- A. Set structural steel accurately in locations and to elevations indicated and according to AISC 303 and AISC 360.
- B. Baseplates and Leveling Plates: Clean concrete- and masonry-bearing surfaces of bond-reducing materials, and roughen surfaces prior to setting places. Clean bottom surface of plates.
 - 1. Set plates for structural members on wedges, shins, or setting nuts as required.
 - 2. Weld plate washers to top of baseplate
 - 3. Snug-tighten anchor rods after supported members have been positioned and plumbed. Do not remove wedges or shiras buy if protruding, cut off flush with edge of plate before packing with grout.
 - 4. Promptly pack grout sender between bearing surfaces and plates so no voids remain. Neatly finish exposed surfaces; protect grout and allow to cure. Comply with manufacturer's writte installation instructions for shrinkage-resistant grouts.
- C. Maintain erection toler acces of structural steel within AISC 303, "Code of Standard Practice for Steel Buildings and Bridges."
- D. Align and edjust various members that form part of complete frame or structure before permanently fastening. Before assembly, clean bearing surfaces and other surfaces that are in permanent costact with members. Perform necessary adjustments to compensate for discrepancies in elevations and alignment.

F.

Level and plumb individual members of structure.

Make allowances for difference between temperature at time of erection and mean temperature when structure is completed and in service.

Splice members only where indicated.

- Do not use thermal cutting during erection.
- G. Do not enlarge unfair holes in members by burning or using drift pins. Ream holes that must be enlarged to admit bolts.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

3.4 FIELD CONNECTIONS

- A. High-Strength Bolts: Install high-strength bolts according to RCSC's "Specification for Structural Joints Using ASTM A 325 or A 490 Bolts" for type of bolt and type of joint specified.
 - 1. Joint Type: Snug tightened or Pretensioned.
- B. Weld Connections: Comply with AWS D1.1/D1.1M for tolerances, appearances, we procedure specifications, weld quality, and methods used in correcting welding work.
 - 1. Comply with AISC 303 and AISC 360 for bearing, alignment, adequacy of temporary connections, and removal of paint on surfaces adjacent to field welds.
 - 2. Remove backing bars or runoff tabs, back gouge, and grind steel stoo

3.5 FIELD QUALITY CONTROL

- A. Special Inspections: Owner will engage a qualified special hyperborner to perform the following special inspections:
 - 1. Verify structural-steel materials and inspect steel frame joint details.
 - 2. Verify weld materials and inspect welds.
 - 3. Verify connection materials and inspect high-strength bolted connections.
- B. Testing Agency: Owner will engage a qualitied testing agency to perform tests and inspections.
- C. Bolted Connections: Inspect and test bolted connections according to RCSC's "Specification for Structural Joints Using ASTM: 22.5 or A 490 Bolts."
- D. Welded Connections: Visually respect field welds according to AWS D1.1/D1.1M.
 - 1. In addition to visual inspection, test and inspect field welds according to AWS D11/D1.1M and the following inspection procedures, at testing agency's option:
 - a. Lucid Penetrant Inspection: ASTM E 165.
 - Mignetic Particle Inspection: ASTM E 709; performed on root pass and on finished weld. Cracks or zones of incomplete fusion or penetration are not accepted.
 - Ultrasonic Inspection: ASTM E 164.
 - Radiographic Inspection: ASTM E 94.

REPAIRS AND PROTECTION

Touchup Painting: Immediately after erection, clean exposed areas where primer is damaged or missing and paint with the same material as used for shop painting to comply with SSPC-PA 1 for touching up shop-painted surfaces.

1. Clean and prepare surfaces by SSPC-SP 2 hand-tool cleaning or SSPC-SP 3 power-tool cleaning.

Tetra Tech 200-15704-17001

- B. Touchup Painting: Cleaning and touchup painting are specified in Section 09 91 13 "Exterior Painting" and Section 09 91 23 "Interior Painting."
- C. Touchup Priming: Cleaning and touchup priming are specified in Section 09 96 00 "High-Performance Coatings."

END OF SECTION

Republic

<

SECTION 05 31 00

STEEL DECKING

PART 1 - GENERAL

1.1 **RELATED DOCUMENTS**

Drawings and general provisions of the Contract, including General an A. Conditions and Division 01 Specification Sections, apply to this Section.

1.2 **SUMMARY**

- Section Includes: A.
 - 1. Roof deck.
 - 2. Acoustical cellular roof deck.
- Β. **Related Requirements:**
 - 1.
 - Section 05 12 00 "Structural Steel Framing" for shop- and field-welded shear connectors. Section 05 50 00 "Metal Fabrications" for framing deck openings with miscellaneous 2. steel shapes.
 - Section 09 91 13 "Exterior Paining" for repair painting of primed deck and finish 3. painting of deck.
 - or Painting" for repair painting of primed deck and finish painting Section 09 91 23 "Inte 4. of deck.
- ACTION SUBMITTA 1.3
 - Product Data: 1 each type of deck, accessory, and product indicated. A.
 - Β. Shop Dr ving

include layout and types of deck panels, anchorage details, reinforcing channels, pans, cat deck openings, special jointing, accessories, and attachments to other construction.

INFORMATIONAL SUBMITTALS

- Welding certificates.
- B. Product Certificates: For each type of steel deck.
- C. Product Test Reports: For tests performed by a qualified testing agency, indicating that each of the following complies with requirements:

Tetra Tech 200-15704-17001 STEEL DECKING 05 31 00 - 1

- 1. Power-actuated mechanical fasteners.
- 2. Acoustical roof deck.
- D. Evaluation Reports: For steel deck, from ICC-ES.
- E. Field quality-control reports.

1.5 QUALITY ASSURANCE

- A. Testing Agency Qualifications: Qualified according to ASTM E 329 for testing indexed.
- B. Welding Qualifications: Qualify procedures and personnel according to AWS D.3/D1.3M, "Structural Welding Code Sheet Steel."
- C. FM Global Listing: Provide steel roof deck evaluated by FM Global and Inted in its "Approval Guide, Building Materials" for Class 1 fire rating and Class 1-90 wardstorm ratings.

1.6 DELIVERY, STORAGE, AND HANDLING

- A. Protect steel deck from corrosion, deformation, and other damage during delivery, storage, and handling.
- B. Stack steel deck on platforms or pallets and slope to provide drainage. Protect with a waterproof covering and ventilate to avoid condensation
 - 1. Protect and ventilate acoustical cellular roof deck with factory-installed insulation to maintain insulation free construe.

PART 2 - PRODUCTS

- 2.1 PERFORMANCE REQUIREMENTS
 - A. AISI Specifications: Comply with calculated structural characteristics of steel deck according to AISI's 'North American Specification for the Design of Cold-Formed Steel Structural Members.'
 - B. Fire-Resistance Ratings: Comply with ASTM E 119; testing by a qualified testing agency. dentify products with appropriate markings of applicable testing agency.
 - Indicate design designations from UL's "Fire Resistance Directory" or from the listings of another qualified testing agency.

ROOF DECK

A. Manufacturers: Subject to compliance with requirements, provide products by one of the following:

STEEL DECKING 05 31 00 - 2 Tetra Tech 200-15704-17001

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- 1. ASC Profiles, Inc.
- 2. Canam Steel Corporation; Canam Group, Inc.
- 3. CMC Joist & Deck.
- 4. Consolidated Systems, Inc.
- 5. Cordeck.
- 6. DACS, Inc.
- 7. Epic Metals Corporation.
- 8. Marlyn Steel Decks, Inc.
- 9. New Millennium Building Systems, LLC.
- 10. Nucor Corp.
- 11. Roof Deck, Inc.
- 12. Valley Joist.
- 13. Verco Decking, Inc., a Nucor company.
- B. Roof Deck: Fabricate panels, without top-flange stiffening groover, to comply with "SDI Specifications and Commentary for Steel Roof Deck," in SDI Publication No. 31, and with the following:
 - 1. Galvanized-Steel Sheet: ASTM A 653/A 653Mc Structural Steel (SS), Grade 33, G90 zinc coating.
 - 2. Deck Profile: As indicated.
 - 3. Profile Depth: As indicated.
 - 4. Design Uncoated-Steel Thickness: As held
 - 5. Span Condition: As indicated.
 - 6. Side Laps: Overlapped or interlocking seam at Contractor's option.

2.3 ACOUSTICAL ROOF DECK

- A. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
 - 1. ASC Profiles, Inc.
 - 2. Canam Steel Corporation; Canam Group, Inc.
 - 3. CMC Joist & Deck.
 - Consolicated Systems, Inc.
 - 5. CoN

4.

- 6. CSi Metal Dek Group.
- ACS, Inc.
 - Epic Metals Corporation.
- Marlyn Steel Decks, Inc.
- 10. New Millennium Building Systems, LLC.
- 11. Nucor Corp.
- 12. Roof Deck, Inc.
- B. Acoustical Roof Deck: Fabricate panels, without top-flange stiffening grooves, to comply with "SDI Specifications and Commentary for Steel Roof Deck," in SDI Publication No. 31, and with the following:

Tetra Tech 200-15704-17001

- 1. Galvanized and Shop-Primed Steel Sheet: ASTM A 653/A 653M, Structural Steel (SS), Grade 33, G60 zinc coating; cleaned, pretreated, and primed with manufacturer's standard baked-on, rust-inhibitive primer.
 - a. Color: Manufacturer's standard.
- 2. Cellular Deck Profile: As indicated, with bottom plate.
- 3. Profile Depth: As indicated.
- 4. Design Uncoated-Steel Thicknesses; Deck Unit/Bottom Plate: As indicated.
- 5. Span Condition: Simple span.
- 6. Side Laps: Overlapped or interlocking seam at Contractor's option.
- 7. Acoustical Perforations: Cellular deck units with manufacturer's standard performed flatbottom plate welded to ribbed deck.
- 8. Sound-Absorbing Insulation: Manufacturer's standard premolded coll or surp of glass or mineral fiber.
 - a. Factory install sound-absorbing insulation into cells of calular deck.
 - b. Installation of sound-absorbing insulation.
- 9. Acoustical Performance: NRC 0.65/ 0.75/ 0.00/ 0.85/ 0.90, tested according to ASTM C 423.

2.4 ACCESSORIES

- A. General: Provide manufacturer's scandard accessory materials for deck that comply with requirements indicated.
- B. Mechanical Fasteners: Conssion-resistant, low-velocity, power-actuated or pneumatically driven carbon-steel fasteners or seaf-drilling, self-threading screws.
- C. Side-Lap Fasteners: Corroston-resistant, hexagonal washer head; self-drilling, carbon-steel screws, No. 10 prinimum lameter.
- D. Flexible Cloure Strips: Vulcanized, closed-cell, synthetic rubber.
- E. Miscellar sous Sheet Metal Deck Accessories: Steel sheet, minimum yield strength of 33,000 psi, not less than 0.0359-inch design uncoated thickness, of same material and finish as deck; of profile indicated or required for application.



column Closures, End Closures, Z-Closures, and Cover Plates: Steel sheet, of same material, finish, and thickness as deck unless otherwise indicated.

Galvanizing Repair Paint: ASTM A 780/A 780M.

H. Repair Paint: Manufacturer's standard rust-inhibitive primer of same color as primer.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine supporting frame and field conditions for compliance with requirements for installation tolerances and other conditions affecting performance of the Work.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.

3.2 INSTALLATION, GENERAL

- A. Install deck panels and accessories according to applicable specifications and commentary in SDI Publication No. 31, manufacturer's written instructions, and requirements in this Section.
- B. Install temporary shoring before placing deck panels if required to the truckection limitations.
- C. Locate deck bundles to prevent overloading of supporting members.
- D. Place deck panels on supporting frame and adjust to final position with ends accurately aligned and bearing on supporting frame before being permanently fastened. Do not stretch or contract side-lap interlocks.
 - 1. Align cellular deck panels over full lengt of cell runs and align cells at ends of abutting panels.
- E. Place deck panels flat and square and faster to supporting frame without warp or deflection.
- F. Cut and neatly fit deck parels and accessories around openings and other work projecting through or adjacent to deck
- G. Provide additional reprocement and closure pieces at openings as required for strength, continuity of deck and support of other work.
- H. Comply with XWS requirements and procedures for manual shielded metal arc welding, appearance and quality of welds, and methods used for correcting welding work.
- I. Mechanical fasteners may be used in lieu of welding to fasten deck. Locate mechanical fasteners and install according to deck manufacturer's written instructions.

ROOF-DECK INSTALLATION

Fasten roof-deck panels to steel supporting members by arc spot (puddle) welds of the surface diameter indicated or arc seam welds with an equal perimeter that is not less than 1-1/2 inches long, and as follows:

- 1. Weld Diameter: 5/8 inch, nominal.
- 2. Weld Spacing: Weld edge and interior ribs of deck units with a minimum of two welds per deck unit at each support. Space welds 12 inches apart in the field of roof and 6

Tetra Tech 200-15704-17001

STEEL DECKING 05 31 00 - 5 inches apart in roof corners and perimeter, based on roof-area definitions in FMG Loss Prevention Data Sheet 1-28.

- B. Side-Lap and Perimeter Edge Fastening: Fasten side laps and perimeter edges of panels between supports, at intervals not exceeding the lesser of one-half of the span or 18 inches, and as follows:
 - 1. Mechanically fasten with self-drilling, No. 10 diameter or larger, carbon-steel screy s
 - 2. Mechanically clinch or button punch.
 - 3. Fasten with a minimum of 1-1/2-inch-long welds.
- C. End Bearing: Install deck ends over supporting frame with a minimum end bearing of 1-1/2 inches, with end joints as follows:
 - 1. End Joints: Lapped 2 inches minimum.
- D. Miscellaneous Roof-Deck Accessories: Install ridge and valle, plates, finish strips, end closures, and reinforcing channels according to deck manufactures's written instructions. Weld or mechanically fasten to substrate to provide a complete deck installation.
 - 1. Weld cover plates at changes in direction of roof-dick panels unless otherwise indicated.
- E. Flexible Closure Strips: Install flexible closure strips over partitions, walls, and where indicated. Install with adhesive according to manufacture's written instructions to ensure complete closure.
- F. Sound-Absorbing Insulation: Installance into topside ribs of deck.

3.4 FIELD QUALITY CONTROL

- A. Testing Agency: Owner vill engage a qualified testing agency to perform tests and inspections.
- B. Field welds will be subject to inspection.
- C. Prepare test and enspection reports.

3.5 PROTECTION

Galvanizing Repairs: Prepare and repair damaged galvanized coatings on both surfaces of deck with galvanized repair paint according to ASTM A 780/A 780M and manufacturer's written instructions.

Repair Painting: Wire brush and clean rust spots, welds, and abraded areas on both surfaces of prime-painted deck immediately after installation, and apply repair paint.

- 1. Apply repair paint, of same color as adjacent shop-primed deck, to bottom surfaces of deck exposed to view.
- 2. Wire brushing, cleaning, and repair painting of bottom deck surfaces are included in Section 09 91 13 "Exterior Painting" and Section 09 91 23 "Interior Painting."

STEEL DECKING 05 31 00 - 6 Tetra Tech 200-15704-17001

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

C. Repair Painting: Wire brushing, cleaning, and repair painting of rust spots, welds, and abraded areas of both deck surfaces are included in Section 09 91 13 "Exterior Painting" and Section 09 91 23 "Interior Painting."

END OF SECTION Repland

PAGE INTEN BLANK

SPEC SECTION 05 40 00 COLD-FORMED METAL FRAMING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supremints Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. This Section includes the following cold-formed metal framing ("CEVAT) elements:
 - 1. Exterior non-load-bearing wall framing.
 - 2. Roof rafter framing.
 - 3. Ceiling Joist framing.

1.3 PERFORMANCE REQUIREMENTS

- A. Structural Performance: Provide cold for need metal framing capable of withstanding design loads and all combinations of these loads, as equired by applicable codes and as specified on Drawings, within limits and under conditions indicated.
 - 1. Design Loads: As indicated in Drawings.
 - a. Dead Loads, Weights of materials and construction.
 - 2. Deflection Limits. Design framing systems to withstand design loads without deflections greater than he following:
 - a. Exterior Non-Load-Bearing Framing: Horizontal deflection of 1/240 for metal wall panels, 1/360 for EIFS and 1/600 for masonry veneer of the wall height.
 - **B** of Rafter Framing: Horizontal deflection of 1/240 and 1/360 of the horizontally projected span.



Design framing systems to provide for movement of framing members without damage or overstressing, sheathing failure, connection failure, undue strain on fasteners and anchors, or other detrimental effects when subject to a maximum ambient temperature change of 120 deg F.

- Design framing system to maintain clearances at openings, to allow for construction tolerances, and to accommodate live load deflection of primary building structure as follows:
 - a. Upward and downward movement of 1/2 inch.

- B. Cold-Formed Steel Framing, General: Design according to AISI's "Standard for Cold-Formed Steel Framing General Provisions."
 - 1. Headers: Design according to AISI's "Standard for Cold-Formed Steel Framing Header Design."
 - 2. Design exterior non-load-bearing wall framing to accommodate horizontal deflection without regard for contribution of sheathing materials.
- C. Minimum Thickness and Maximum Spacing Requirements: Comply with the requirements of this specification for specific requirements for minimum thickness and maximum spacing for cold-formed metal framing components.

1.4 SUBMITTALS

- A. General: Submit all action submittals and informational submittals required by this Section concurrently.
- B. Action Submittals:
 - 1. Product Data: For each type of cold-formed total framing product and accessory indicated.
 - 2. LEED Submittals:
 - a. Product Data for Credit M2.4: For products having recycled content, documentation indicating percentages by weight of postconsumer and preconsumer recycled content. Include statement indicating cost for each product having recycled content
 - 3. Shop Drawings: Show labout, spacings, sizes, thicknesses, and types of cold-formed metal framing; rablication; and fastening and anchorage details, including mechanical fasteners. Show einforcing channels, opening framing, supplemental framing, strapping, bracing, bridging, oplices, accessories, connection details, and attachment to adjoining work.
 - For cold-formed metal framing indicated to comply with design loads, include structural analysis data signed and sealed by the qualified professional engineer responsible for their preparation.

Informational Submittals:

- Welding certificates.
- Qualification Data: For testing agency and installer. Submit written certification or similar documentation signed by applicable subcontractor, or Contractor indicating compliance with applicable "Qualifications" requirements specified below in "Quality Assurance" article.
- 3. Product Test Reports: From a qualified testing agency, unless otherwise stated, indicating that each of the following complies with requirements, based on evaluation of comprehensive tests for current products:
CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- a. Steel sheet.
- b. Expansion anchors.
- c. Power-actuated anchors.
- d. Mechanical fasteners.
- e. Vertical deflection clips.
- f. Horizontal drift deflection clips
- g. Miscellaneous structural clips and accessories.

1.5 QUALITY ASSURANCE

- A. Engineering Responsibility: Preparation of Shop Drawings, design calculation and other structural data by a qualified professional engineer.
- B. Cold-formed Metal Framing Installer Qualifications: At least 5 years experience in construction of cold-formed metal framing systems similar in scope to application shownon Documents with record of successful in-service performance for prior projects.
- C. Product Tests: Mill certificates or data from a qualified independent testing agency indicating steel sheet complies with requirements, including base metal thickness, yield strength, tensile strength, total elongation, chemical requirements, and metalic-coating thickness.
- D. Welding: Qualify procedures and personnel according to AWS D1.1/D1.1M, "Structural Welding Code--Steel," and AWS D1.3, "Structural Welding Code--Sheet Steel."
- E. AISI Specifications and Standards: Comply with AISI's "North American Specification for the Design of Cold-Formed Steel Structure) Members" and its "Standard for Cold-Formed Steel Framing General Provisions."
- F. Preinstallation Conference: Conduct conference at Project site to comply with requirements in Division 01 Section "Project Management and Coordination."

1.6 DELIVERY, STORAGE, AND HANDLING

- A. Packing and Shapping: Deliver materials and components to Site in manufacturer's unopened package, containers or bundles, fully identified with brand name, type, grade and identification of multi-sture or supplier.
- B. Protect cold-formed metal framing from corrosion, deformation, and other damage during delivery, storage, and handling.

Store cold-formed metal framing, protect with a waterproof covering, and ventilate to avoid condensation.

ADDITION AND RENOVATIONS

PART 2 - PRODUCTS

2.1 MANUFACTURERS

- A. Manufacturers: Subject to compliance with requirements, available manufacturers offering cold-formed metal framing that may be incorporated into the Work include, but are not limited to, the following:
 - 1. AllSteel Products, Inc.
 - 2. ClarkWestern Building Systems, Inc.
 - 3. Dietrich Metal Framing; a Worthington Industries Company.
 - 4. MarinoWare; a division of Ware Industries.
 - 5. MBA Building Supplies, Inc.
 - 6. Super Stud Building Products, Inc.
 - 7. United Metal Products, Inc.
 - 8. Telling Industries, LLC.

2.2 MATERIALS

- A. Steel Sheet: ASTM A 1003/A 1003M, Structural Grade, Type H, metallic coated, of grade and coating weight as follows:
 - 1. Grade: As required by structural performance
 - 2. Coating: G60, (Z180) A60, (ZF100), AZ50, or GF30 (ZGF90).

2.3 EXTERIOR NON-LOAD-BEARING WALL FRAMING

- A. Steel Studs: Manufacturer, standard C-shaped steel studs, of web depths indicated, punched, with stiffened flanges and as follows:
 - 1. Minimum Base-Model Thickness: 33 mils standard or 54 mils for masonry veneer backup.
 - 2. Minimum Prange Width: 1-5/8 inches (41 mm).
- B. Steel Track: Manufacturer's standard U-shaped steel track, of web depths indicated, unpunched, with testiffened flanges, and as follows:



Minimum Base-Metal Thickness: 33 mils or at least matching steel studs. Minimum Flange Width: 1-1/4 inches. (32mm).

Single Deflection Track: Manufacturer's single, deep-leg, U-shaped steel track; unpunched, with unstiffened flanges, of web depth to contain studs while allowing free vertical movement, with flanges designed to support horizontal and lateral loads and transfer them to the primary structure, and as follows:

- 1. Minimum Base-Metal Thickness: 54 mils.
- 2. Flange Width: 1 inch (25 mm) plus the design gap for 1-story structures and 1 inch 925 mm) plus twice the design gap for other applications.

2.4 ROOF-RAFTER FRAMING

- A. Steel Rafters: Manufacturer's standard C-shaped steel sections, of web depths indicated, unpunched, with stiffened flanges, and as follows:
 - 1. Minimum Base-Metal Thickness: 33 mils.
 - 2. Minimum Flange Width: 2 inches (51 mm).
- B. Built-up Members: Built-up members of manufacturer's standard C-shaped steel section, wi stiffened flanges, nested into a U-shaped steel section joist track, with unstiffered flange unpunched; of web depths indicated; and as follows:
 - 1. Minimum Base-Metal Thickness: 33 mils and at least matching steel ratters
 - 2. Minimum Flange Width: 2 inches.

2.5 CEILING JOIST FRAMING

- A. Steel Ceiling Joists: Manufacturer's standard C-shaped steel vectors, of web depths indicated, unpunched, with stiffened flanges, and as follows:
 - 1. Minimum Base-Metal Thickness: 33 mils.
 - 2. Minimum Flange Width: 2 inches.
 - 3.

2.6 FRAMING ACCESSORIES

- A. Fabricate steel-framing accessories from steel sheet, ASTM A 1003/A 1003M, Structural Grade, Type H, metallic coated, of same grade and coating weight used for framing members.
- B. Provide accessories of munul cturer's standard thickness and configuration, unless otherwise indicated, as follows:
 - 1. Supplementary framing.
 - 2. Bracing, ridging, and solid blocking.
 - 3. Web stilleners.
 - 4. Anshor clips.
 - 5. Find clips.
 - Stud kickers, knee braces, and girts.
 - Backer plates.

ANCHORS, CLIPS, AND FASTENERS

Steel Shapes and Clips: ASTM A 1003/A 1003M, Structural Grade, Type H, metallic coated, of grade and coating weight matching associated member.

B. Anchor Bolts: ASTM F 1554, Grade 36, threaded carbon-steel hex-headed bolts and carbonsteel nuts; and flat, hardened-steel washers; zinc coated by hot-dip process according to ASTM A 153/A 153M, Class C.

- C. Expansion Anchors: Fabricated from corrosion-resistant materials, with capability to sustain, without failure, a load equal to 5 times design load, as determined by testing per ASTM E 488 conducted by a qualified independent testing agency.
- D. Power-Actuated Anchors: Fastener system of type suitable for application indicated, fabricated from corrosion-resistant materials, with capability to sustain, without failure, a load equal to 10 times design load, as determined by testing per ASTM E 1190 conducted by a qualified independent testing agency.
- E. Mechanical Fasteners: ASTM C 1513, corrosion-resistant-coated, self-drilling, self-tag steel drill screws.
 - 1. Head Type: Low-profile head beneath sheathing, manufacturer's standard elsewhere.
- F. Welding Electrodes: Comply with AWS standards.

2.8 MISCELLANEOUS MATERIALS

- A. Galvanizing Repair Paint: ASTM A 780.
- B. Nonmetallic, Nonshrink Grout: Premixed, nonmetallic, noncorrosive, nonstaining grout containing selected silica sands, portland generic shrinkage-compensating agents, and plasticizing and water-reducing agents, complying with ASTM C 1107, with fluid consistency and 30-minute working time.
- C. Shims: Load bearing, high-density matim nomer plastic, nonleaching.
- D. Sealer Gaskets: Closed-cell neuronne foam, 1/4 inch thick, selected from manufacturer's standard widths to match wigh of byttom track or rim track members.

2.9 FABRICATION

- A. Fabricate cold for aed metal framing and accessories plumb, square, and true to line, and with connections set trely fastened, according to referenced AISI specifications and standards, manufacturer's written instructions, and requirements in this Section.
 - . Pabricate framing assemblies using jigs or templates.

Cut framing members by sawing or shearing; do not torch cut.

- Fasten cold-formed metal framing members by welding, screw fastening, clinch fastening, or riveting as standard with fabricator. Wire tying of framing members is not permitted.
- a. Comply with AWS D1.3 requirements and procedures for welding, appearance and quality of welds, and methods used in correcting welding work.
- b. Locate mechanical fasteners and install according to Shop Drawings, with screw penetrating joined members by not less than three exposed screw threads.
- 4. Splices in axially loaded members not acceptable.



- B. Reinforce, stiffen, and brace framing assemblies to withstand handling, delivery, and erection stresses. Lift fabricated assemblies to prevent damage or permanent distortion.
- C. Fabrication Tolerances: Fabricate assemblies level, plumb, and true to line to a maximum allowable tolerance variation of 1/8 inch in 10 feet and as follows:
 - 1. Spacing: Space individual framing members no more than plus or minus 1/8 inch from plan location. Cumulative error shall not exceed minimum fastening requirements of sheathing or other finishing materials.
 - 2. Squareness: Fabricate each cold-formed metal framing assembly to a maximum square tolerance of 1/8 inch.
- D. Touch-up all welds with galvanizing primer.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine supporting substrates and abutting structural framing for compliance with requirements for installation tolerances and other conditions affecting performance.
 - 1. Proceed with installation only after unsatisfactory conditions have been corrected.
- B. Beginning installation constitutes contractor's acceptance of substrates and conditions.

3.2 PREPARATION

A. Install load bearing shows on grout between the underside of wall bottom track or rim track and the top of foundation wall or slab at stud or joist locations to ensure a uniform bearing surface on supporting concrete or masonry construction.

3.3 INSTALLATION, GENERAL

. Cold ormed metal framing may be shop or field fabricated for installation, or it may be field assembled.

Install cold-formed metal framing according to AISI's "Standard for Cold-Formed Steel Framing - General Provisions", manufacturer's written instructions, Drawings, and approved Shop Drawings unless more stringent requirements are indicated.

- Install shop- or field-fabricated, cold-formed framing and securely anchor to supporting structure.
 - 1. Screw, bolt, or weld wall panels at horizontal and vertical junctures to produce flush, even, true-to-line joints with maximum variation in plane and true position between fabricated panels not exceeding 1/16 inch.

- D. Install cold-formed metal framing and accessories plumb, square, and true to line, and with connections securely fastened.
 - 1. Cut framing members by sawing or shearing; do not torch cut.
 - 2. Fasten cold-formed metal framing members by welding, screw fastening, clinch fastening, or riveting. Wire tying of framing members is not permitted.
 - a. Comply with AWS D1.3 requirements and procedures for welding, appearance and quality of welds, and methods used in correcting welding work.
 - b. Locate mechanical fasteners and install according to Shop Drawings, and complying with requirements for spacing, edge distances, and screw percent on.
- E. Install framing members in one-piece lengths unless splice connections are indicated for track or tension members.
- F. Install temporary bracing and supports to secure framing and support hads comparable in intensity to those for which structure was designed. Maintain brees and supports in place, undisturbed, until entire integrated supporting structure has been completed and permanent connections to framing are secured.
- G. Do not bridge building expansion and control joint with cold-formed metal framing. Independently frame both sides of joints.
- H. Install insulation, specified in Division 07 Sectorn "thermal Insulation," in built-up exterior framing members, such as headers, sills, the real joists, and multiple studs at openings, that are inaccessible on completion of framing work.
- I. Fasten hole reinforcing plate over web punctrations that exceed size of manufacturer's standard punched openings.
- J. Erection Tolerances: Install cold formed metal framing level, plumb, and true to line to a maximum allowable tolerance variation of 1/8 inch in 10 feet and as follows:
 - 1. Space individual having members no more than plus or minus 1/8 inch from plan location dumulative error shall not exceed minimum fastening requirements of sheathing or other finishing materials.
- 3.4 EXCEPTOR NON-LOAD-BEARING WALL INSTALLATION

Install continuous tracks sized to match studs. Align tracks accurately and securely anchor to supporting structure as indicated.

Fasten both flanges of studs to bottom track, unless otherwise indicated. Space studs as follows:

- 1. Maximum Stud Spacing: 16 inches.
- C. Set studs plumb, except as needed for diagonal bracing or required for nonplumb walls or warped surfaces and similar requirements.

- D. Isolate non-load-bearing steel framing from building structure to prevent transfer of vertical loads while providing lateral support.
 - 1. Install single deflection tracks and anchor to building structure. Do not fasten studs to deflection track.
- E. Install headers and sills for wall openings wider than stud spacing. Locate headers and sills at openings as indicated. Fabricate headers of compound shapes indicated or required to transfer. load to supporting studs, complete with clip-angle connectors, web stiffeners, or gusset patrs.
 - 1. Frame wall openings with not less than a double stud at each jamb of frame condicated on approved Shop Drawings. Fasten jamb members together to uniformly distribute loads.
 - 2. Install runner tracks and jack studs above and below wall openings. Anonor tracks to jamb studs with clip angles or by welding, and space jack studs sure as full-height wall studs.
- F. Install horizontal bridging in wall studs, spaced in rows indicated on Shop Drawings but not more than 48 inches apart. Fasten at each stud intersection
 - 1. Top Bridging for Single Deflection Track: Install ow of horizontal bridging within 12 inches of single deflection track. Install a combination of flat, taut, steel sheet straps of width and thickness indicated and stue or stud-track solid blocking of width and thickness matching studs. Fasten flat straps to stud flanges and secure solid blocking to stud webs or flanges.
 - a. Install solid blocking at certen indicated on approved Shop Drawings.
 - 2. Bridging may be one of the following types:
 - a. Cold-rolled steel channel, welded or mechanically fastened to webs of punched studs.
 - b. Combination of flat, taut, steel sheet straps of width and thickness indicated and stud rack sold blocking of width and thickness to match studs. Fasten flat straps to stud flanges and secure solid blocking to stud webs or flanges.
 - c. Proprietary bridging bars installed according to manufacturer's written instructions.
- G. Instal macellineous framing and connections, including stud kickers, web stiffeners, clip angles continuous angles, anchors, fasteners, and stud girts, to provide a complete and stable wall-huming system.



Install perimeter joist track sized to match joists. Align and securely anchor or fasten track to supporting structure at corners, ends, and spacings indicated on approved Shop Drawings.

- B. Install joists bearing on supporting frame, level, straight, and plumb; adjust to final position, brace, and reinforce. Fasten joists to both flanges of joist track.
 - 1. Install joists over supporting frame with a minimum end bearing of 1-1/2 inches.

- 2. Reinforce ends and bearing points of joists with web stiffeners, end clips, joist hangers, steel clip angles, or steel-stud sections as indicated on approved Shop Drawings.
- C. Space joists not more than 2 inches from abutting walls, and as follows:
 - 1. Joist Spacing: As indicated on approved Shop Drawings.
- D. Frame openings with built-up joist headers consisting of joist and joist track, nesting joints, another combination of connected joists as indicated on approved Shop Drawings.
- E. Install joist reinforcement at interior supports with single, short length of joist section located directly over interior support, with lapped joists of equal length to joist reinforcement, or as indicated on approved Shop Drawings.
 - 1. Install web stiffeners to transfer axial loads of walls above.
- F. Install bridging at intervals indicated on approved Shop Drawings, install prior to application of loads. Fasten bridging at each joist intersection.
 - 1. Bridging may be one of the following:
 - a. Joist-track solid blocking of width and thickness indicated, secured to joist webs.
 - b. Combination of flat, taut, steel sheet staars of width and thickness indicated and joist-track solid blocking of width and thickness indicated. Fasten flat straps to bottom flange of joists and scoure solid blocking to joist webs.
 - 2. Provide end blocking where joint ranger ends are not restrained against rotation.
- G. Secure joists to load-bearing increases to prevent lateral movement of bottom flange.
- H. Provide additional joist under parallel, non-load bearing partitions when partition length exceeds one half span
- I. Do not install web knock outs within 1.5 times depth of member from edge of bearing, unless necessary. Where equired, reinforce web.
- J. Install miscellaneous joist framing and connections, including web stiffeners, closure pieces, clip angles, cortinuous angles, hold-down angles, anchors, and fasteners, to provide a complete and stable joist-framing assembly.

NELD QUALITY CONTROL

Testing: Owner will engage a qualified independent testing and inspecting agency to perform field tests and inspections and prepare test reports. Contractor to provide access to construction for inspections to verify conformance with provisions of Contract Documents and approved Shop Drawings. Inspections to include but not limited to, the following:

- 1. Verify member sizes, configuration and spacing.
- 2. Inspect screwed and welded connections.
- 3. Inspect weld quality.
- 4. Verify field welder's certification.

- 5. Verify galvanizing primer has been applied to all welds and damaged galvanized surfaces.
- 6. Verify compliance with details shown on approved drawings such as bracing, stiffening, member location, and proper application of joint details at each connection.
- B. Field and shop welds will be subject to testing and inspecting.
- C. Testing agency will report test results promptly and in writing to Contractor and Architect
- D. Remove and replace work where test results indicate that it does not comply win requirements.
- E. Additional testing and inspecting, at Contractor's expense, will be performed to determine compliance of replaced or additional work with specified requirements.

3.7 REPAIRS AND PROTECTION

- A. Galvanizing Repairs: Prepare and repair damaged galvanized loatings on fabricated and installed cold-formed metal framing with galvanized repair plaint according to ASTM A 780 and manufacturer's written instructions.
- B. Provide final protection and maintain conditions, in a manner acceptable to manufacturer and Installer, that ensure that cold-formed metal transing is without damage or deterioration at time of Substantial Completion.

ECTION



SPEC SECTION 05 50 00

METAL FABRICATIONS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section Includes:
 - 1. Steel framing and supports for operable partitions
 - 2. Steel framing and supports for overhead doors and calles.
 - 3. Steel framing and supports for mechanical and electrical equipment.
 - 4. Steel framing and supports for applications where framing and supports are not specified in other Sections.
 - 5. Elevator beams.
 - 6. Steel shapes for supporting elevator oper snis.
 - 7. Metal ladders.
 - 8. Miscellaneous steel trim.
 - 9. Loose bearing and leveling protector applications where they are not specified in other Sections.

1.3 PERFORMANCE REDUIR IMENTS

- A. Delegated Design Design ladders, including comprehensive engineering analysis by a qualified professional engineer, using performance requirements and design criteria indicated.
- B. Thermal Movements: Allow for thermal movements from ambient and surface temperature charger acting on exterior metal fabrications by preventing buckling, opening of joints, overstressing of components, failure of connections, and other detrimental effects.

Temperature Change: 120 deg F, ambient; 180 deg F, material surfaces.

SUBMITTALS, GENERAL

. General: Submit all action submittals and informational submittals required by this Section concurrently.

1.5 ACTION SUBMITTALS

- A. Product Data: For the following:
 - 1. Paint products.
 - 2. Grout.
- B. LEED Submittals:
 - 1. Product Data for Credit MR 4: For products having recycled content, document tion indicating percentages by weight of postconsumer and preconsumer recyclea content. Include statement indicating cost for each product having recycled content.
- C. Shop Drawings: Show fabrication and installation details for metal fabrications
 - 1. Include plans, elevations, sections, and details of metal fabrications and their connections. Show anchorage and accessory items.
- D. Delegated-Design Submittal: For installed products indicated to comply with performance requirements and design criteria, including analysis data signed and sealed by the qualified professional engineer responsible for their preparation.
- 1.6 INFORMATIONAL SUBMITTALS
 - A. Welding certificates.
 - B. Paint Compatibility Certificates: From manufacturers of topcoats applied over shop primers certifying that shop primers are empetible with topcoats.
- 1.7 QUALITY ASSURANCE
 - A. Welding Qualifications: Qualify procedures and personnel according to AWS D1.1/D1.1M, "Structural Wading Code Steel."
 - B. Welding Qualifications: Qualify procedures and personnel according to the following:
 - . AWS D1.1/D1.1M, "Structural Welding Code Steel."
 - AWS D1.6, "Structural Welding Code Stainless Steel."

PROJECT CONDITIONS

Field Measurements: Verify actual locations of walls and other construction contiguous with metal fabrications by field measurements before fabrication.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

1.9 COORDINATION

- A. Coordinate selection of shop primers with topcoats to be applied over them. Comply with paint and coating manufacturers' written recommendations to ensure that shop primers and topcoats are compatible with one another.
- B. Coordinate installation of anchorages. Furnish setting drawings, templates, and directions or installing anchorages, including sleeves, concrete inserts, anchor bolts, and items with integral anchors, that are to be embedded in concrete or masonry. Deliver such items to Projectsite in time for installation.

PART 2 - PRODUCTS

2.1 METALS, GENERAL

A. Metal Surfaces, General: Provide materials with smooth, flat surfaces unless otherwise indicated. For metal fabrications exposed to view in the completed Work, provide materials without seam marks, roller marks, rolled trade names, or olerhistics.

2.2 FERROUS METALS

- A. Recycled Content of Steel Products: Postconsumer recycled content plus one-half of preconsumer recycled content not less than 5 percent.
- B. Steel Plates, Shapes, and Bars: ASTM A36/A 36M.
- C. Stainless-Steel Bars and Shares: ATM A 276, Type 304.
- D. Rolled-Stainless-Steel Appr Pate: ASTM A 793.
- E. Steel Tubing: ASTM A 300, cold-formed steel tubing.
- F. Steel Pipe: ASYM A 33/A 53M, standard weight (Schedule 40) unless otherwise indicated.

2.3 FASTENER.

General: Unless otherwise indicated, provide Type 304 stainless-steel fasteners for exterior use nd zinc-plated fasteners with coating complying with ASTM B 633 or ASTM F 1941, Class Fe/Zn 5, at exterior walls. Select fasteners for type, grade, and class required.

MISCELLANEOUS MATERIALS

- A. Welding Rods and Bare Electrodes: Select according to AWS specifications for metal alloy welded.
- B. Alkyd Primer: Modified-alkyd primer compatible with topcoat.

- 1. Products: Subject to compliance with requirements, available products that may be incorporated into the Work include, but are not limited to, the following:
 - a. Tnemec Company, Inc.; Series 10: 10-1009 Gray.
- C. Galvanizing Repair Paint: High-zinc-dust-content paint complying with SSPC-Paint 20 and compatible with paints specified to be used over it.
- D. Nonshrink, Nonmetallic Grout: Factory-packaged, nonstaining, noncorrosive, nongaseous gro complying with ASTM C 1107. Provide grout specifically recommended by manufacture of interior and exterior applications.

2.5 FABRICATION, GENERAL

- A. Shop Assembly: Preassemble items in the shop to greatest extent possible. Disassemble units only as necessary for shipping and handling limitations. Use connections that maintain structural value of joined pieces. Clearly mark units for reastembly and coordinated installation.
- B. Cut, drill, and punch metals cleanly and accurately. Remove burrs and ease edges to a radius of approximately 1/32 inch unless otherwise indicated. Remove sharp or rough areas on exposed surfaces.
- C. Form bent-metal corners to smallest redius cossible without causing grain separation or otherwise impairing work.
- D. Form exposed work with accurate ingles and surfaces and straight edges.
- E. Weld corners and seams continuously to comply with the following:
 - 1. Use materials and methods that minimize distortion and develop strength and corrosion resistance of basement ls.
 - 2. Obtain fusion without undercut or overlap.
 - 3. Remove we ding flux immediately.
 - 4. At exposed connections, finish exposed welds and surfaces smooth and blended so no roughness shows after finishing.
- F. For exposed connections with hairline joints, flush and smooth, using concealed fasteners or welds where possible. Where exposed fasteners are required, use Phillips flat-head (countersunk) fasteners unless otherwise indicated. Locate joints where least conspicuous.

Fabricate seams and other connections that will be exposed to weather in a manner to exclude water. Provide weep holes where water may accumulate.

- Cut, reinforce, drill, and tap metal fabrications as indicated to receive finish hardware, screws, and similar items.
- I. Provide for anchorage of type indicated; coordinate with supporting structure. Space anchoring devices to secure metal fabrications rigidly in place and to support indicated loads.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON. DE

1. Where units are indicated to be cast into concrete or built into masonry, equip with integrally welded steel strap anchors, 1/8 by 1-1/2 inches, with a minimum 6-inch embedment and 2-inch hook, not less than 8 inches from ends and corners of units and 24 inches o.c., unless otherwise indicated.

2.6 MISCELLANEOUS FRAMING AND SUPPORTS

- General: Provide steel framing and supports not specified in other Sections as not A. complete the Work.
- Fabricate units from steel shapes, plates, and bars of welded construction and B. erwise indicated. Fabricate to sizes, shapes, and profiles indicated and as necessary o receive adjacent construction.
 - Furnish inserts for units installed after concrete is placed. 1.
- Fabricate supports for operable partitions from continuous steel deans of sizes indicated with attached bearing plates, anchors, and braces as recommended by partition manufacturer. Drill C. or punch bottom flanges of beams to receive partition track hanger rods; locate holes where indicated on operable partition Shop Drawings.

2.7 METAL LADDERS

- A. General:
 - 1. Comply with ANSI A14.3 caless wise indicated.
 - 2. For elevator pit ladders th ASME A17.1.

Β. Steel Ladders:

- incluse apart unless otherwise indicated. 1. Space siderails
- 2. Siderails: Continues, 1/2-by-2-1/2-inch steel flat bars, with eased edges.
- 3.
- Rungs: 6/4 nch-diameter steel bars. Fit rungs: a centerline of siderails; plug-weld and grind smooth on outer rail faces. 4.
- Provide nonslip surfaces on top of each rung by coating with abrasive material 5. meallically bonded to rung.
- Apport each ladder at top and bottom and not more than 48 inches o.c. with welded or bolted steel brackets.
 - Galvanize ladders, including brackets and fasteners.

METAL FLOOR PLATE

Fabricate from rolled-stainless-steel floor plate of thickness indicated below:

- 1. Thickness: 1/4 inch.
- Provide stainless-steel angle supports as indicated. Β.

ADDITION AND RENOVATIONS

2.9 MISCELLANEOUS STEEL TRIM

- A. Unless otherwise indicated, fabricate units from steel shapes, plates, and bars of profiles shown with continuously welded joints and smooth exposed edges. Miter corners and use concealed field splices where possible.
- B. Provide cutouts, fittings, and anchorages as needed to coordinate assembly and installation with other work.
 - 1. Provide with integrally welded steel strap anchors for embedding in concrete or mas construction.

2.10 LOOSE BEARING AND LEVELING PLATES

A. Provide loose bearing and leveling plates for steel items bearing in hasonry or concrete construction. Drill plates to receive anchor bolts and for grouting.

2.11 LOOSE STEEL LINTELS

- A. Fabricate loose steel lintels from steel angles and shapes of size indicated for openings and recesses in masonry walls and partitions at locations adjusted. Fabricate in single lengths for each opening unless otherwise indicated. Wello adjoining members together to form a single unit where indicated.
- B. Size loose lintels to provide bearing length at each side of openings equal to 1/12 of clear span but not less than 8 inches unless obserwise indicated.

2.12 STEEL WELD PLATES AND ANGLES

A. Provide steel weld place and angles not specified in other Sections, for items supported from concrete construction as needed to complete the Work. Provide each unit with no fewer than two integrally welled steel strap anchors for embedding in concrete.

2.13 FINISHES, GENERAL

A. Compy with NAAMM's "Metal Finishes Manual for Architectural and Metal Products" for recommendations for applying and designating finishes.

Finish metal fabrications after assembly.



Finish exposed surfaces to remove tool and die marks and stretch lines, and to blend into surrounding surface.

2.14 STEEL AND IRON FINISHES

A. Galvanizing: Hot-dip galvanize items as indicated to comply with ASTM A 153/A 153M for steel and iron hardware and with ASTM A 123/A 123M for other steel and iron products.

- 1. Do not quench or apply post galvanizing treatments that might interfere with paint adhesion.
- B. Shop prime iron and steel items not indicated to be galvanized unless they are to be embedded in concrete or masonry, or unless otherwise indicated.
 - 1. Metal Fabrications: For all iron and steel items, shop prime with alkyd primer.
- C. Preparation for Shop Priming: Prepare surfaces to comply with SSPC-SP 6/NACL "Commercial Blast Cleaning."
- D. Shop Priming: Apply shop primer to comply with SSPC-PA 1, "Paint Application Specification No. 1: Shop, Field, and Maintenance Painting of Steel," for shop painting.
 - 1. Stripe paint corners, crevices, bolts, welds, and sharp edges.

PART 3 - EXECUTION

- 3.1 INSTALLATION, GENERAL
 - A. Beginning installation constitutes Contractor's a ceptace of substrates and conditions.
 - B. Cutting, Fitting, and Placement: Perform cutting, drilling, and fitting required for installing metal fabrications. Set metal fabrications accurately in location, alignment, and elevation; with edges and surfaces level, plumb, true, and free of rack; and measured from established lines and levels.
 - C. Fit exposed connections accurately together to form hairline joints. Weld connections that are not to be left as exposed joints but cannot be shop welded because of shipping size limitations. Do not weld, cut, or yor de surfaces of exterior units that have been hot-dip galvanized after fabrication and are for batted or screwed field connections.
 - D. Field Welding Comply with the following requirements:
 - 1. Use materials and methods that minimize distortion and develop strength and corrosion resistance of base metals.
 - 2. Constant fusion without undercut or overlap.
 - Remove welding flux immediately.

At exposed connections, finish exposed welds and surfaces smooth and blended so no roughness shows after finishing and contour of welded surface matches that of adjacent surface.

Fastening to In-Place Construction: Provide anchorage devices and fasteners where metal fabrications are required to be fastened to in-place construction. Provide threaded fasteners for use with concrete and masonry inserts, toggle bolts, through bolts, lag screws, wood screws, and other connectors.

F. Provide temporary bracing or anchors in formwork for items that are to be built into concrete, masonry, or similar construction.

ADDITION AND RENOVATIONS

3.2 INSTALLING MISCELLANEOUS FRAMING AND SUPPORTS

- A. General: Install framing and supports to comply with requirements of items being supported, including manufacturers' written instructions and requirements indicated on Shop Drawings.
- B. Anchor supports for operable partitions securely to and rigidly brace from building structure.

3.3 INSTALLING BEARING AND LEVELING PLATES

- A. Clean concrete and masonry bearing surfaces of bond-reducing materials, and redgren to improve bond to surfaces. Clean bottom surface of plates.
- B. Set bearing and leveling plates on wedges, shims, or leveling nuts. After oearing members have been positioned and plumbed, tighten anchor bolts. Do not remove wedges or shims but, if protruding, cut off flush with edge of bearing plate before packing with grott
 - 1. Use nonshrink grout, either metallic or nonmetallic, in concelled locations where not exposed to moisture; use nonshrink, nonmetallic grout in exposed locations unless otherwise indicated.
 - 2. Pack grout solidly between bearing surfaces and places to ensure that no voids remain.

3.4 ADJUSTING AND CLEANING

- A. Touchup Painting: Immediately after erection, clean field welds, bolted connections, and abraded areas. Paint uncoated and abraded areas with the same material as used for shop painting to comply with SSPC-PA4 for truching up shop-painted surfaces.
 - 1. Apply by brush or spray to provide a minimum 2.0-mil dry film thickness.
- B. Galvanized Surfaces: Cean field welds, bolted connections, and abraded areas and repair galvanizing to comply each JSTM A 780.

END OF SECTION

SPEC SECTION 05 51 00

METAL STAIRS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section Includes:
 - 1. Preassembled steel stairs with concrete-filled treads.
 - 2. Steel tube railings attached to metal stairs.
 - 3. Steel tube handrails attached to walls adjacent to metal stairs.

1.3 PERFORMANCE REQUIREMENTS

- A. Delegated Design: Design metal steirs, including comprehensive engineering analysis by a qualified professional engineer, using performance requirements and design criteria indicated.
- B. Structural Performance of Sairs: Metal stairs shall withstand the effects of gravity loads and the following loads and structures within limits and under conditions indicated.
 - 1. Uniform Load: 100 Inf/sq. ft.
 - 2. Concentrated Loak 500 lbf applied on an area of 4 sq. in.
 - 3. Uniform and concentrated loads need not be assumed to act concurrently.
 - 4. Stair Fraging: Capable of withstanding stresses resulting from railing loads in addition to foads uppelified above.
 - 5. Light deflection of treads, platforms, and framing members to L/360 or 1/4 inch, whichever is less.

Structural Performance of Railings: Railings shall withstand the effects of gravity loads and the following loads and stresses within limits and under conditions indicated.

Handrails and Top Rails of Guards:

- a. Uniform load of 50 lbf/ ft. applied in any direction.
- b. Concentrated load of 200 lbf applied in any direction.
- c. Uniform and concentrated loads need not be assumed to act concurrently.
- 2. Infill of Guards:
 - a. Concentrated load of 50 lbf applied horizontally on an area of 1 sq. ft..

b. Infill load and other loads need not be assumed to act concurrently.

1.4 SUBMITTALS, GENERAL

A. General: Submit all action submittals and informational submittals required by this Section concurrently.

1.5 ACTION SUBMITTALS

- A. Product Data: For metal stairs and the following:
 - 1. Paint products.
 - 2. Grout.
- B. LEED Submittals:
 - 1. Product Data for Credit MR 4: For products having recycled content, documentation indicating percentages by weight of postconsumer and perconsumer recycled content. Include statement indicating cost for each product avoing recycled content.
- C. Shop Drawings: Include plans, elevations, sections, deails, and attachments to other work.
- D. Delegated-Design Submittal: For installed products indicated to comply with performance requirements and design criteria, including analysis data signed and sealed by the qualified professional engineer responsible for their preparation.

1.6 INFORMATIONAL SUBMET

- A. Welding certificates.
- B. Paint Compatibility Certificates: From manufacturers of topcoats applied over shop primers certifying that shop primers are compatible with topcoats.

1.7 QUALITY ASSURANCE

A. Instater Qualifications: Fabricator of products.

NAAMM Stair Standard: Comply with "Recommended Voluntary Minimum Standards for Fixed Metal Stairs" in NAAMM AMP 510, "Metal Stairs Manual," for class of stair designated, unless more stringent requirements are indicated.

- 1. Preassembled Stairs: Commercial class.
- C. Welding Qualifications: Qualify procedures and personnel according to the following:
 - 1. AWS D1.1/D1.1M, "Structural Welding Code Steel."
 - 2. AWS D1.3, "Structural Welding Code Sheet Steel."

METAL STAIRS 05 51 00 - 2

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

1.8 COORDINATION

- A. Coordinate selection of shop primers with topcoats to be applied over them. Comply with paint and coating manufacturers' written recommendations to ensure that shop primers and topcoats are compatible with one another.
- B. Coordinate installation of anchorages for metal stairs. Furnish setting drawings, templates, add directions for installing anchorages, including sleeves, concrete inserts, anchor bolts, and items, with integral anchors, that are to be embedded in concrete or masonry. Deliver such terms to Project site in time for installation.

PART 2 - PRODUCTS

2.1 METALS, GENERAL

A. Metal Surfaces, General: Provide materials with smooth, flat surfaces unless otherwise indicated. For components exposed to view in the completed Worl, provide materials without seam marks, roller marks, rolled trade names, or blemistes.

2.2 FERROUS METALS

- A. Recycled Content of Steel Products: Postconsumer recycled content plus one-half of preconsumer recycled content not less than 5 percent.
- B. Steel Plates, Shapes, and Bars: ASTM A36/A 36M.
- C. Steel Tubing: ASTM A 500 cold rerned).
- D. Uncoated, Cold-Rolled Strel Sheet: ASTM A 1008/A 1008M, structural steel, Grade 25, unless another grade is required by lesign loads; exposed.
- 2.3 FASTENERS
 - A. General: Provide zinc-plated fasteners with coating complying with ASTM B 633 or ASTM F 1941, Class Fe/Zn 12 for exterior use, and Class Fe/Zn 5 where built into exterior walls Select fasteners for type, grade, and class required.

MSCELLANEOUS MATERIALS

Welding Rods and Bare Electrodes: Select according to AWS specifications for metal alloy welded.

- B. Alkyd Primer: Modified-alkyd primer compatible with topcoat.
 - 1. Products: Subject to compliance with requirements, available products that may be incorporated into the Work include, but are not limited to, the following:

ADDITION AND RENOVATIONS

- a. Tnemec, Inc.; Series 10: 10-1009 Gray.
- C. Nonshrink, Nonmetallic Grout: Factory-packaged, nonstaining, noncorrosive, nongaseous grout complying with ASTM C 1107. Provide grout specifically recommended by manufacturer for interior and exterior applications.
- D. Concrete Materials and Properties: Comply with requirements in Division 03 Section "Cast-n-Place Concrete" for normal-weight, air-entrained, ready-mix concrete with a minimum 28-day compressive strength of 3000 psi unless otherwise indicated.
- E. Welded Wire Fabric: ASTM A 185/A 185M, 6 by 6 inches, W1.4 by W1.4, unless otherwise indicated.

2.5 FABRICATION, GENERAL

- A. Provide complete stair assemblies, including metal framing, harvers, struts, railings, clips, brackets, bearing plates, and other components necessary to support and anchor stairs and platforms on supporting structure.
 - 1. Join components by welding unless otherwise indicated.
 - 2. Use connections that maintain structural values of joined pieces.
- B. Preassembled Stairs: Assemble stairs in short, greatest extent possible. Disassemble units only as necessary for shipping and handling limitations. Clearly mark units for reassembly and coordinated installation.
- C. Cut, drill, and punch metals cleanic and accurately. Remove burrs and ease edges to a radius of approximately 1/32 inch unless otherwise indicated. Remove sharp or rough areas on exposed surfaces.
- D. Form bent-metal corders to smallest radius possible without causing grain separation or otherwise impairing work.
- E. Form exposed work with accurate angles and surfaces and straight edges.
- F. Weld connections to comply with the following:
 - Use materials and methods that minimize distortion and develop strength and corrosion resistance of base metals.
 - Obtain fusion without undercut or overlap.
 - Remove welding flux immediately.
 - Weld exposed corners and seams continuously unless otherwise indicated.
 - At exposed connections, finish exposed welds to comply with NOMMA's "Voluntary Joint Finish Standards" for Type 1 welds: no evidence of a welded joint.
- G. Form exposed connections with hairline joints, flush and smooth, using concealed fasteners where possible. Where exposed fasteners are required, use Phillips flat-head (countersunk) screws or bolts unless otherwise indicated. Locate joints where least conspicuous.

2.6 STEEL-FRAMED STAIRS

- A. Stair Framing:
 - 1. Fabricate stringers of steel tubes.
 - a. Provide closures for exposed ends of tube stringers.
 - 2. Construct platforms of steel channel headers and miscellaneous framing member indicated.
 - 3. Weld stringers to headers; weld framing members to stringers and headers.
 - 4. Where masonry walls support metal stairs, provide temporary supporting stats designed for erecting steel stair components before installing masonry.
- B. Metal-Pan Stairs: Form risers, subtread pans, and subplatforms to configurations shown from steel sheet of thickness needed to comply with performance requirements but not less than 0.075 inch.
 - 1. Steel Sheet: Uncoated cold-rolled steel sheet.
 - 2. Attach risers and subtreads to stringers with brackets hade of steel angles or bars. Weld brackets to stringers and attach metal pans to brackets by welding, riveting, or bolting.
 - 3. Shape metal pans to include nosing integral with riser.
 - 4. Provide subplatforms of configuration indicated or, if not indicated, the same as subtreads. Weld subplatforms to platform saming.

2.7 STAIR RAILINGS

- A. Comply with applicable requirements in Division 05 Section "Metal Railings."
- B. Steel Tube Railings: Fabricate tailings to comply with requirements indicated for design, dimensions, details, finish, and member sizes, including wall thickness of tube, post spacings, and anchorage, but not ass than that needed to withstand indicated loads.
- C. Welded Connections: Fabricate railings with welded connections. Cope components at connections to provide close fit, or use fittings designed for this purpose. Weld all around at connections, in luding at fittings.
 - 1. Unish welds to comply with NOMMA's "Voluntary Joint Finish Standards" for Type 1 welds: no evidence of a welded joint.

orm changes in direction of railings as follows:

- By bending.
- Form simple and compound curves by bending members in jigs to produce uniform curvature for each repetitive configuration required; maintain cross section of member throughout entire bend without buckling, twisting, cracking, or otherwise deforming exposed surfaces of components.
- F. Close exposed ends of railing members with prefabricated end fittings.

Tetra Tech 200-15704-17001

METAL STAIRS 05 51 00 - 5

- G. Provide wall returns at ends of wall-mounted handrails unless otherwise indicated. Close ends of returns unless clearance between end of rail and wall is 1/4 inch or less.
- H. Brackets, Flanges, Fittings, and Anchors: Provide wall brackets, end closures, flanges, miscellaneous fittings, and anchors for interconnecting components and for attaching to other work. Furnish inserts and other anchorage devices for connecting to concrete or masonry work
 - 1. Connect posts to stair framing by direct welding unless otherwise indicated.
 - 2. For nongalvanized railings, provide nongalvanized ferrous-metal fittings, bac fasteners, and sleeves, except galvanize anchors embedded in exterior nasonry concrete construction.
- I. Fillers: Provide fillers made from steel plate, or other suitably crush-resistant material, where needed to transfer wall bracket loads through wall finishes to structural supports. Size fillers to suit wall finish thicknesses and to produce adequate bearing area to prevent oracket rotation and overstressing of substrate.

2.8 FINISHES

- A. Comply with NAAMM's "Metal Finishes Manual for Architectural and Metal Products" for recommendations for applying and designating finishes.
- B. Finish metal stairs after assembly.
- C. Preparation for Shop Priming: Prepare unceated ferrous-metal surfaces to comply with SSPC-SP 6/NACE No. 3, "Commercial Blass Cleaning."
- D. Apply shop primer to uncoacel surfaces of metal stair components, except those with galvanized finishes and three to be embedded in concrete or masonry unless otherwise indicated. Comply with SSPC PA 1, "Paint Application Specification No. 1: Shop, Field, and Maintenance Painting of Steel," for shop painting.
 - 1. Stripe paint corners, crevices, bolts, welds, and sharp edges.
- PART 3 EXECUTION

3.1

INSTALLATION, GENERAL

eginning installation constitutes Contractor's acceptance of substrates and conditions.

Fastening to In-Place Construction: Provide anchorage devices and fasteners where necessary for securing metal stairs to in-place construction. Include threaded fasteners for concrete and masonry inserts, through-bolts, lag bolts, and other connectors.

C. Cutting, Fitting, and Placement: Perform cutting, drilling, and fitting required for installing metal stairs. Set units accurately in location, alignment, and elevation, measured from established lines and levels and free of rack.

- D. Install metal stairs by welding stair framing to steel structure or to weld plates cast into concrete unless otherwise indicated.
- E. Provide temporary bracing or anchors in formwork for items that are to be built into concrete, masonry, or similar construction.
- F. Fit exposed connections accurately together to form hairline joints. Weld connections that are not to be left as exposed joints but cannot be shop welded because of shipping size limit dions. Do not weld, cut, or abrade surfaces of exterior units that have been hot-dip galvanized after fabrication and are for bolted or screwed field connections.
- G. Field Welding: Comply with requirements for welding in "Fabrication, Generation"
- H. Place and finish concrete fill for treads and platforms to comply with Division 03 Section "Castin-Place Concrete."

3.2 INSTALLING METAL STAIRS WITH GROUTED BASEPLATIS

- A. Clean concrete and masonry bearing surfaces of bord-reducing materials, and roughen to improve bond to surfaces. Clean bottom surface of basepares.
- B. Set steel stair baseplates on wedges, shims, or leveling nuts. After stairs have been positioned and aligned, tighten anchor bolts. Do not remove wedges or shims but, if protruding, cut off flush with edge of bearing plate before packing with grout.
 - 1. Use nonmetallic, nonshrink growtuness otherwise indicated.
 - 2. Pack grout solidly between cearing surfaces and plates to ensure that no voids remain.

3.3 INSTALLING RAILINGS

- A. Adjust railing systems before anchoring to ensure matching alignment at abutting joints. Space posts at spacing indicated or, if not indicated, as required by design loads. Plumb posts in each direction. Secure posts and rail ends to building construction as follows:
 - 1. Archor losis to steel by welding directly to steel supporting members.
 - 2. Anshor landrail ends to concrete and masonry with steel round flanges welded to rail side and anchored with postinstalled anchors and bolts.



4 ADJUSTING AND CLEANING

A. Touchup Painting: Immediately after erection, clean field welds, bolted connections, and abraded areas of shop paint, and paint exposed areas with same material as used for shop painting to comply with SSPC-PA 1 for touching up shop-painted surfaces.

Tetra Tech 200-15704-17001

METAL STAIRS 05 51 00 - 7

1. Apply by brush or spray to provide a minimum 2.0-mil dry film thickness.

END OF SECTION

Republic

SPEC SECTION 05 52 00

METAL RAILINGS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Steel tube railings.

1.3 PERFORMANCE REQUIREMENTS

- A. Delegated Design: Design railings, including comprehensive engineering analysis by a qualified professional engineer, using performance requirements and design criteria indicated.
- B. General: In engineering railings to with stand structural loads indicated, determine allowable design working stresses of railing materials based on the following:
 - 1. Steel: 72 percent of minimum yield strength.
- C. Structural Performance: realings shall withstand the effects of gravity loads and the following loads and stresses within lights and under conditions indicated:
 - 1. Handrah, and Top Rails of Guards:
 - Uniform load of 50 lbf/ ft. applied in any direction.
 - concentrated load of 200 lbf applied in any direction.
 - Uniform and concentrated loads need not be assumed to act concurrently.

Infill of Guards:

a.

- Concentrated load of 50 lbf applied horizontally on an area of 1 sq. ft..
- b. Infill load and other loads need not be assumed to act concurrently.
- Control of Corrosion: Prevent galvanic action and other forms of corrosion by insulating metals and other materials from direct contact with incompatible materials.

ADDITION AND RENOVATIONS

1.4 SUBMITTALS, GENERAL

A. General: Submit all action submittals and informational submittals required by this Section concurrently.

1.5 ACTION SUBMITTALS

- A. Product Data: For the following:
 - 1. Railing brackets.
 - 2. Grout and paint products.
- B. LEED Submittals:



- C. Shop Drawings: Include plans, elevations, sections, detens, and anachments to other work.
- D. Delegated-Design Submittal: For installed products indicated to comply with performance requirements and design criteria, including analysis data signed and sealed by the qualified professional engineer responsible for their preparation

1.6 INFORMATIONAL SUBMITTALS

- A. Welding certificates.
- B. Paint Compatibility Certificate: From manufacturers of topcoats applied over shop primers certifying that shop primes an compatible with topcoats.

1.7 QUALITY ASSULANCE

- A. Source Limitation: Obtain each type of railing from single source from single manufacturer.
- B. We ding Qualifications: Qualify procedures and personnel according to AWS D1.1/D1.1M, "Structural Welding Code Steel."

PROJECT CONDITIONS

Field Measurements: Verify actual locations of walls and other construction contiguous with metal fabrications by field measurements before fabrication.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

1.9 COORDINATION AND SCHEDULING

- A. Coordinate selection of shop primers with topcoats to be applied over them. Comply with paint and coating manufacturers' written recommendations to ensure that shop primers and topcoats are compatible with one another.
- B. Coordinate installation of anchorages for railings. Furnish setting drawings, templates, and directions for installing anchorages, including sleeves, concrete inserts, anchor bolts, and items, with integral anchors, that are to be embedded in concrete or masonry. Deliver such terms to Project site in time for installation.
- C. Schedule installation so wall attachments are made only to completed walls. Donot support railings temporarily by any means that do not satisfy structural performance requirements.

PART 2 - PRODUCTS

2.1 METALS, GENERAL

- A. Metal Surfaces, General: Provide materials with smooth surfaces, without seam marks, roller marks, rolled trade names, stains, discolorations, orblemistics.
- B. Brackets, Flanges, and Anchors: Cast or former metal of same type of material and finish as supported rails unless otherwise indicated

2.2 STEEL AND IRON

- A. Recycled Content of Steel Products: Postconsumer recycled content plus one-half of preconsumer recycled content nut less than 25 percent.
- B. Tubing: ASTM A 500 (cold formed).
- C. Plates, Shapes and Bars: ASTM A 36/A 36M.
- D. General: Provide the following:
 - . Ungarvanized-Steel Railings: Plated steel fasteners complying with ASTM B 633 or ASTM F 1941, Class Fe/Zn 5 for zinc coating.



asteners for Anchoring Railings to Other Construction: Select fasteners of type, grade, and class required to produce connections suitable for anchoring railings to other types of construction indicated and capable of withstanding design loads.

Fasteners for Interconnecting Railing Components:

1. Provide concealed fasteners for interconnecting railing components and for attaching them to other work, unless otherwise indicated.

ADDITION AND RENOVATIONS

2.3 MISCELLANEOUS MATERIALS

- A. Welding Rods and Bare Electrodes: Select according to AWS specifications for metal alloy welded.
- B. Alkyd Primer: Modified-alkyd primer compatible with topcoat.
 - 1. Products: Subject to compliance with requirements, available products that princorporated into the Work include, but are not limited to, the following:
 - a. Tnemec, Inc.; Series 10: 10-1009 Gray.
- C. Nonshrink, Nonmetallic Grout: Factory-packaged, nonstaining, noncorrosive, nongateous grout complying with ASTM C 1107. Provide grout specifically recommended by headfacturer for interior and exterior applications.

2.4 FABRICATION

- A. General: Fabricate railings to comply with requirements indicated for design, dimensions, member sizes and spacing, details, finish, and anchorate, but not less than that required to support structural loads.
- B. Assemble railings in the shop to greatest exact possible to minimize field splicing and assembly. Disassemble units only as necessary for shipping and handling limitations. Clearly mark units for reassembly and coordinated installation. Use connections that maintain structural value of joined pieces.
- C. Cut, drill, and punch metals cleanly and accurately. Remove burrs and ease edges to a radius of approximately 1/32 inch unless otherwise indicated. Remove sharp or rough areas on exposed surfaces.
- D. Form work true to line and level with accurate angles and surfaces.
- E. Cut, reinforce drill and tap as indicated to receive finish hardware, screws, and similar items.
- F. Connections: Nabicate railings with welded connections unless otherwise indicated.
- G. We ded Connections: Cope components at connections to provide close fit, or use fittings designed for this purpose. Weld all around at connections, including at fittings.



- Use materials and methods that minimize distortion and develop strength and corrosion resistance of base metals.
- Obtain fusion without undercut or overlap.
- B. Remove flux immediately.
- 4. At exposed connections, finish exposed surfaces smooth and blended so no roughness shows after finishing and welded surface matches contours of adjoining surfaces.
- H. Form changes in direction as follows:
 - 1. By bending.

METAL RAILINGS 05 52 00 - 4

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- I. Bend members in jigs to produce uniform curvature for each configuration required; maintain cross section of member throughout entire bend without buckling, twisting, cracking, or otherwise deforming exposed surfaces of components.
- J. Close exposed ends of railing members with prefabricated end fittings.
- K. Provide wall returns at ends of wall-mounted handrails unless otherwise indicated. Close ends of returns unless clearance between end of rail and wall is 1/4 inch or less.
- L. Brackets, Flanges, Fittings, and Anchors: Provide wall brackets, flanges, miscellaneous fitting and anchors to interconnect railing members to other work unless otherwise indicated
 - 1. At brackets and fittings fastened to gypsum board partitions, provide crush-resistant fillers, or other means to transfer loads through wall finishes to arrutual supports and prevent bracket or fitting rotation and crushing of substrate.
- M. Provide inserts and other anchorage devices for connecting reilliers to concrete or masonry work. Fabricate anchorage devices capable of withstanding loals imposed by railings. Coordinate anchorage devices with supporting structure.
- N. Toe Boards: Where indicated, provide toe boards at ratings around openings and at edge of open-sided floors and platforms. Fabricate to dimensions and details indicated.

2.5 FINISHES, GENERAL

A. Comply with NAAMM's "Metal Finishes Manual for Architectural and Metal Products" for recommendations for applying and designating finishes.

2.6 STEEL AND IRON FINISNES

- A. For nongalvanized such rillings, provide nongalvanized ferrous-metal fittings, brackets, fasteners, and sheves, succept galvanize anchors to be embedded in exterior concrete or masonry.
- B. Preparation for Shop Priming: Prepare uncoated ferrous-metal surfaces to comply with SSPC-SP 674A E Nr. 3, "Commercial Blast Cleaning."
- C. Prime Application: Apply shop primer to prepared surfaces of railings unless otherwise indicated. Comply with requirements in SSPC-PA 1, "Paint Application Specification No. 1: shop, Field, and Maintenance Painting of Steel," for shop painting. Primer need not be applied to surfaces to be embedded in concrete or masonry.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine gypsum board assemblies, where reinforced to receive anchors, to verify that locations of concealed reinforcements have been clearly marked for Installer. Locate reinforcements and mark locations if not already done.
- B. Beginning installation constitutes Contractor's acceptance of substrates and condition

3.2 INSTALLATION, GENERAL

- A. Fit exposed connections together to form tight, hairline joints.
- B. Perform cutting, drilling, and fitting required for installing railing, oscillating accurately in location, alignment, and elevation; measured from established line and levels and free of rack.
 - 1. Do not weld, cut, or abrade surfaces of railing components that have been coated or finished after fabrication and that are intended to field connection by mechanical or other means without further cutting or fitting.
 - 2. Set posts plumb within a tolerance of 1/16 inch 3 feet.
 - 3. Align rails so variations from level for horizontal members and variations from parallel with rake of steps and ramps for sloping members do not exceed 1/4 inch in 12 feet.
- C. Adjust railings before anchoring to ensure patching alignment at abutting joints.
- D. Fastening to In-Place Construction: Use anchorage devices and fasteners where necessary for securing railings and for properly transferring loads to in-place construction.

3.3 RAILING CONNECTORS

A. Welded Connections: Use fully welded joints for permanently connecting railing components. Comply with reduirements for welded connections in "Fabrication" Article whether welding is performed in the shop or in the field.

ANCLORING POSTS



3.4

form or core-drill holes not less than 5 inches deep and 3/4 inch larger than OD of post for installing posts in concrete. Clean holes of loose material, insert posts, and fill annular space between post and concrete with nonshrink, nonmetallic grout, mixed and placed to comply with anchoring material manufacturer's written instructions.

- . Cover anchorage joint with flange of same metal as post, welded to post after placing anchoring material.
- C. Anchor posts to metal surfaces with oval flanges, angle type, or floor type as required by conditions, connected to posts and to metal supporting members as follows:

METAL RAILINGS 05 52 00 - 6

1. For steel pipe railings, weld flanges to post and bolt to metal supporting surfaces.

3.5 ATTACHING RAILINGS

- A. Anchor railing ends at walls with round flanges anchored to wall construction and welded to railing ends.
- B. Anchor railing ends to metal surfaces with flanges bolted to metal surfaces and we railing ends.
- C. Attach railings to wall with wall brackets. Provide brackets with 1-1/2-inch chartene from inside face of handrail and finished wall surface. Locate brackets as indicated or, if not indicated, at spacing required to support structural loads.
 - 1. Use type of bracket with flange tapped for concealed anchorage to threaded hanger bolt.
 - 2. Locate brackets as indicated or, if not indicated, at spacing localized to support structural loads.
- D. Secure wall brackets and railing end flanges to building construction as required to comply with performance requirements.

3.6 ADJUSTING AND CLEANING

- A. Touchup Painting: Immediately after erection, clean field welds, bolted connections, and abraded areas of shop paint, and paintexposed areas with the same material as used for shop painting to comply with SSPC-PA4 for truching up shop-painted surfaces.
 - 1. Apply by brush or spray to provide a minimum 2.0-mil dry film thickness.

END OF SECTION



SPEC SECTION 05 52 13

PIPE TUBE RAILINGS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. This Section includes the following:
 - 1. Stainless Steel pipe and tube railings.
- B. Related Sections include the following:
 - Division 02 Section Site work Sections. SECTION 32 12 16 – Hot Mix Asphalt Paying SECTION 32 13 13 – Portland Come at Concrete Paying SECTION 03 30 00 - Cast in Place Concrete SECTION 04 20 00 - Unit Masonn

1.3 PERFORMANCE REQUIREMENT

- A. General: In engineer or ratings to withstand structural loads indicated, determine allowable design working stresses of ailing materials based on the following:
 - 1. Steel: 72 percent of minimum yield strength.
- B. Structura Performance: Provide railings capable of withstanding the effects of gravity loads and the following loads and stresses within limits and under conditions indicated:
 - Handrails:
 - a. Uniform load of 50 lbf/ ft. (0.73 kN/m) applied in any direction.
 - b. Concentrated load of 200 lbf (0.89 kN) applied in any direction.
 - c. Uniform and concentrated loads need not be assumed to act concurrently.
 - 2. Top Rails of Guards:
 - a. Concentrated load of 200 lbf (0.89 kN) applied in any direction.
 - b. Uniform and concentrated loads need not be assumed to act concurrently.
 - 3. Infill of Guards:

- a. Concentrated load of 200 lbf (0.89 kN) applied horizontally on an area of 1 sq. ft. (0.093 sq. m).
- b. Uniform load of 25 lbf/sq. ft. (1.2 kN/sq. m) applied horizontally.
- c. Infill load and other loads need not be assumed to act concurrently.
- C. Thermal Movements: Provide exterior railings that allow for thermal movements resulting from the following maximum change (range) in ambient and surface temperatures by preventing buckling, opening of joints, overstressing of components, failure of connections, and other detrimental effects. Base engineering calculation on surface temperatures of materials due to both solar heat gain and nighttime-sky heat loss.
 - 1. Temperature Change (Range): 120 deg F (67 deg C), ambient; 180 deg F 160 deg C), material surfaces.
- D. Control of Corrosion: Prevent galvanic action and other forms of corrosion by insulating metals and other materials from direct contact with incompatible materials

1.4 SUBMITTALS

- A. Product Data: For the following:
 - 1. Manufacturer's product lines of embedded and in chanically connected railings.
 - 2. Grout, anchoring cement, stainless steer and paint products.
- B. Shop Drawings: Include plans, elevations, sections, details, and attachments to other work.
 - 1. For installed products indicated to comply with design loads, include structural analysis data signed and sealed by the qualified professional engineer responsible for their preparation.
- C. Welding certificates.
- D. Qualification Date: For professional engineer.
- E. Product Test Reports: Based on evaluation of comprehensive tests performed by a qualified testing a ency, according to ASTM E 894 and ASTM E 935.

1.5 QUANTY ASSURANCE

ource Limitations: Obtain each type of railing through one source from a single manufacturer.

Welding: Qualify procedures and personnel according to the following:

1. AWS D1.1, "Structural Welding Code--Steel."
CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

1.6 PROJECT CONDITIONS

- A. Field Measurements: Verify actual locations of walls and other construction contiguous with railings by field measurements before fabrication and indicate measurements on Shop Drawings.
 - 1. Established Dimensions: Where field measurements cannot be made without delaying the Work, establish dimensions and proceed with fabricating railings without field measurements. Coordinate wall and other contiguous construction to ensure that actual dimensions correspond to established dimensions.
 - 2. Provide allowance for trimming and fitting at site.

1.7 COORDINATION AND SCHEDULING

- A. Coordinate installation of anchorages for railings. Furnish setting orawags, templates, and directions for installing anchorages, including sleeves, concrete insides, anchor bolts, and items with integral anchors, that are to be embedded in concrete or maxons. Deliver such items to Project site in time for installation.
- B. Schedule installation so wall attachments are made only o completed walls. Do not support railings temporarily by any means that do not satisfy structural performance requirements.

PART 2 - PRODUCTS

2.1 METALS, GENERAL

- A. Metal Surfaces, General: Provide materials with smooth surfaces, without seam marks, roller marks, rolled trade names, sain, descolorations, or blemishes.
- B. Brackets, Flanges, and anchors: Cast or formed metal of same type of material and finish as supported rails, peless otherwise indicated.

2.2 STEEL

- A. Tulang ASTM A 500 (cold formed) or ASTM A 513, Type 5 (mandrel drawn).
- B. Pipe: ASTM A 53/A 53M, Type F or Type S, Grade A, Standard Weight (Schedule 40), unless nother grade and weight are required by structural loads.
 - 1. Provide stainless steel satin finish for exterior installations and where indicated.
 - Plates, Shapes, and Bars: ASTM A 36/A 36M.

2.3 FASTENERS

A. General: Provide the following:

- 1. Steel Railings: Stainless steel sleeves and fasteners.
- B. Fasteners for Anchoring Railings to Other Construction: Select fasteners of type, grade, and class required to produce connections suitable for anchoring railings to other types of construction indicated and capable of withstanding design loads.
- C. Fasteners for Interconnecting Railing Components:
 - 1. Provide concealed fasteners for interconnecting railing components and for an them to other work, unless otherwise indicated.
 - 2. Provide concealed fasteners for interconnecting railing components and the attaching them to other work, unless exposed fasteners are unavoidable or are the bandard fastening method for railings indicated.
 - 3. Provide tamper-resistant flat-head machine screws for expressed fasteners, unless otherwise indicated.
- D. Anchors: Provide cast-in-place or torque-controlled expansion anchors, fabricated from corrosion-resistant materials with capability to sustain, without fabrica load equal to six times the load imposed when installed in unit masonry and equal to four times the load imposed when installed in concrete, as determined by testing per 1STNI E 488 conducted by a qualified independent testing agency.

2.4 MISCELLANEOUS MATERIALS

- A. Shop Primer for Galvanized Steel: Zine-duct, zinc-oxide primer formulated for priming zinccoated steel and for compatibility with this paint systems indicated, and complying with SSPC-Paint 5.
- B. Galvanizing Repair Paint: High-kinc-dust-content paint for regalvanizing welds in steel, complying with SSPC-Paint 20.
- C. Nonshrink, Nonmetallin Groat: Factory-packaged, nonstaining, noncorrosive, nongaseous grout complying with teSTM Ci107. Provide grout specifically recommended by manufacturer for interior and exterior applications.
- D. Anchoring Cement: Factory-packaged, nonshrink, nonstaining, hydraulic-controlled expansion cemeat formulation for mixing with water at Project site to create pourable anchoring, patching, and grouting compound.



Water-Resistant Product: At exterior locations and where indicated provide formulation that is resistant to erosion from water exposure without needing protection by a sealer or waterproof coating and that is recommended by manufacturer for exterior use.

FABRICATION

A. General: Fabricate railings to comply with requirements indicated for design, dimensions, member sizes and spacing, details, finish, and anchorage, but not less than that required to support structural loads.

- B. Assemble railings in the shop to greatest extent possible to minimize field splicing and assembly. Disassemble units only as necessary for shipping and handling limitations. Clearly mark units for reassembly and coordinated installation. Use connections that maintain structural value of joined pieces.
- C. Cut, drill, and punch metals cleanly and accurately. Remove burrs and ease edges to a radius of approximately 1/32 inch (1 mm), unless otherwise indicated. Remove sharp or rough areas an exposed surfaces.
- D. Form work true to line and level with accurate angles and surfaces.
- E. Fabricate connections that will be exposed to weather in a manner to exclude water provide weep holes where water may accumulate.
- F. Cut, reinforce, drill, and tap as indicated to receive finish hardware, screws and similar items.
- G. Connections: Fabricate railings with welded connections, unless on a wise indicated.
- H. Welded Connections: Cope components at connections to provide close fit, or use fittings designed for this purpose. Weld all around at connections, including at fittings.
 - 1. Use materials and methods that minimize distortion and develop strength and corrosion resistance of base metals.
 - 2. Obtain fusion without undercut or overlap.
 - 3. Remove flux immediately.
 - 4. At exposed connections, finish exposed surfaces smooth and blended so no roughness shows after finishing and webice surface matches contours of adjoining surfaces.
- I. Form changes in direction as f no vs.
 - 1. By radius bends of radius indicated or by inserting prefabricated elbow fittings of radius indicated.
- J. Form simple and composed curves by bending members in jigs to produce uniform curvature for each repetitive configuration required; maintain cross section of member throughout entire bend without euckling, twisting, cracking, or otherwise deforming exposed surfaces of components.
- K. Close exposed ends of railing members with prefabricated end fittings.

Provide wall returns at ends of wall-mounted handrails, unless otherwise indicated. Close ends of returns unless clearance between end of rail and wall is 1/4 inch (6 mm) or less.

Brackets, Flanges, Fittings, and Anchors: Provide anchoring sleeves, wall brackets, flanges, miscellaneous fittings, and anchors to interconnect railing members to other work, unless otherwise indicated.

1. At brackets and fittings fastened to plaster or gypsum board partitions, provide fillers made from crush-resistant material, or other means to transfer wall loads through wall finishes to structural supports and prevent bracket or fitting rotation and crushing of substrate.

- N. Provide inserts and other anchorage devices for connecting railings to concrete or masonry work. Fabricate anchorage devices capable of withstanding loads imposed by railings. Coordinate anchorage devices with supporting structure.
- O. For railing posts set in concrete, provide steel sleeves not less than 6 inches (150 mm) long with inside dimensions not less than 1/2 inch (13 mm) greater than outside dimensions of post, with steel plate forming bottom closure.

2.6 FINISHES, GENERAL

- A. Comply with NAAMM's "Metal Finishes Manual for Architectural and Metal Products" for recommendations for applying and designating finishes.
- B. Provide exposed fasteners with finish matching appearance, including color and texture, of railings.

2.7 STEEL FINISHES

- A. Galvanized Railings:
 - 1. Stainless Steel exterior satin railings, including hyrdware, after fabrication.
 - 2. Comply with ASTM A 123/A 123M for het-dip galvanized railings.
 - 3. Comply with ASTM A 153/A 153M for hyt-dip galvanized hardware.
- B. Fill vent and drain holes that will be supposed in the finished Work, unless indicated to remain as weep holes, by plugging with zinc solder and filing off smooth.
- C. For galvanized railings, provide het-dip galvanized fittings, brackets, fasteners, sleeves, and other ferrous components.
- D. Preparation for Shop Arimine: After galvanizing, thoroughly clean railings of grease, dirt, oil, flux, and other foreign matter, and treat with metallic-phosphate process.
- E. Preparation for Shop Priming: Prepare uncoated ferrous-metal surfaces to comply with minimum requirements indicated below for SSPC surface preparation specifications and environmental exposure conditions of installed railings:



Exterior Railings (SSPC Zone 1B):SSPC-SP 6/NACE No. 3, "Commercial Blast
Cleaning."Interior Railings (SSPC Zone 1A):SSPC-SP 7/NACE No. 4, "Brush-off Blast

Interior Railings (SSPC Zone IA): SSPC-SP //NACE No. 4, "Brush-off Blast Cleaning."

Apply shop primer to prepared surfaces of railings, unless otherwise indicated. Comply with requirements in SSPC-PA 1, "Paint Application Specification No. 1: Shop, Field, and Maintenance Painting of Steel," for shop painting. Primer need not be applied to surfaces to be embedded in concrete or masonry.

1. Stripe paint corners, crevices, bolts, welds, and sharp edges.

PART 3 - EXECUTION

EXAMINATION 3.1

Examine plaster and gypsum board assemblies, where reinforced to receive anchors, to verify A. that locations of concealed reinforcements have been clearly marked for Installer. Loc te. reinforcements and mark locations if not already done.

3.2 INSTALLATION, GENERAL

- Fit exposed connections together to form tight, hairline joints. A.
- B. Perform cutting, drilling, and fitting required for installing railings. alings accurately in location, alignment, and elevation; measured from established lines and levels and free of rack.
 - Do not weld, cut, or abrade surfaces of railing components hat have been coated or 1. finished after fabrication and that are intended for full connection by mechanical or other means without further cutting or fitting.
 - 2.
 - Set posts plumb within a tolerance of 1/16 inch in Speet (2 mm in 1 m). Align rails so variations from level for horizontal mombers and variations from parallel 3. with rake of steps and ramps for sloping mergers do not exceed 1/4 inch in 12 feet (5 mm in 3 m).
- C. Corrosion Protection: Coat concealed surfaces of aluminum that will be in contact with grout, concrete, masonry, wood, or dissimilar meals, with a heavy coat of bituminous paint.
- Adjust railings before anchoring e matching alignment at abutting joints. D. <u>e</u>
- Fastening to In-Place Construct or. Use anchorage devices and fasteners where necessary for E. securing railings and for properly transferring loads to in-place construction.

RAILING CONNECTIONS 3.3

Welded connections: Use fully welded joints for permanently connecting railing components. Α. Compared with equirements for welded connections in Part 2 "Fabrication" Article whether we ting is performed in the shop or in the field.



Expansion Joints: Install expansion joints at locations indicated but not farther apart than quired to accommodate thermal movement. Provide slip-joint internal sleeve extending 2 inches (50 mm) beyond joint on either side, fasten internal sleeve securely to 1 side, and locate joint within 6 inches (150 mm) of post.

ANCHORING POSTS

Use steel pipe sleeves preset and anchored into concrete for installing posts. After posts have A. been inserted into sleeves, fill annular space between post and sleeve with nonshrink, nonmetallic grout, mixed and placed to comply with anchoring material manufacturer's written instructions.

Tetra Tech 200-15704-17001

- B. Form or core-drill holes not less than 5 inches (125 mm) deep and 3/4 inch (20 mm) larger than OD of post for installing posts in concrete. Clean holes of loose material, insert posts, and fill annular space between post and concrete with nonshrink, nonmetallic grout, mixed and placed to comply with anchoring material manufacturer's written instructions.
- C. Cover anchorage joint with flange of same metal as post, welded to post after placing anchoring material.
- D. Leave anchorage joint exposed; wipe off surplus anchoring material; and leave 1/8-inch 3buildup, sloped away from post.
- E. Anchor posts to metal surfaces with oval flanges, angle type, or floor type as required by conditions, connected to posts and to metal supporting members as follows:
 - 1. For steel pipe railings, weld flanges to post and bolt to metal supporting surfaces.

3.5 ANCHORING RAILING ENDS

- A. Anchor railing ends to concrete and masonry with rourd flanges connected to railing ends and anchored to wall construction with anchors and bolts.
- B. Anchor railing ends to metal surfaces with flanger tolted to metal surfaces and welded to railing ends.

3.6 ATTACHING HANDRAILS TO WILLS

- A. Attach handrails to wall with wan brackets. Provide brackets with 1-1/2-inch (38-mm) clearance from inside face of handrail and finished wall surface.
- B. Locate brackets as indicated on if not indicated, at spacing required to support structural loads.
- C. Secure wall brackets to building construction as follows:
 - 1. For concrete and solid masonry anchorage, use drilled-in expansion shields and hanger or lac bolts
 - 2. For hollow masonry anchorage, use toggle bolts.
 - 3. For steel-framed gypsum board partitions, use hanger or lag bolts set into fire-retardanttreated wood backing between studs. Coordinate with stud installation to locate backing members.

ADJUSTING AND CLEANING

- Clean aluminum and stainless steel by washing thoroughly with clean water and soap and rinsing with clean water.
- B. Touchup Painting: Immediately after erection, clean field welds, bolted connections, and abraded areas of shop paint, and paint exposed areas with same material as used for shop painting to comply with SSPC-PA 1 for touching up shop-painted surfaces.

- C. Touchup Painting: Cleaning and touchup painting of field welds, bolted connections, and abraded areas of shop paint are specified in Division 09 painting Sections.
- D. Galvanized Surfaces: Clean field welds, bolted connections, and abraded areas and repair galvanizing to comply with ASTM A 780.

3.8 PROTECTION

- A. Protect finishes of railings from damage during construction period with temporary protective coverings approved by railing manufacturer. Remove protective coverings of time of Substantial Completion.
- B. Restore finishes damaged during installation and construction period so to evidence remains of correction work. Return items that cannot be refinished in the field to the snop; make required alterations and refinish entire unit, or provide new units.

END OF SECTION

RBIDDIK

<



SPEC SECTION 06 10 00

ROUGH CARPENTRY

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section Includes:
 - 1. Miscellaneous lumber.
 - 2. Rooftop equipment bases and support curbs.
 - 3. Wood blocking and nailers.
 - 4. Wood furring and grounds.
 - 5. Wood sleepers.
 - 6. Plywood backing panels.

1.3 DEFINITIONS

- A. Dimension Lumber: Lumber of 2 nches nominal or greater but less than 5 inches nominal in least dimension.
- B. Lumber grading agencie, and the abbreviations used to reference them, include the following:
 - 1. NeLMA. Northeastern Lumber Manufacturers' Association.
 - 2. NLGA: Vational Lumber Grades Authority.
 - 3. SPIB: the Southern Pine Inspection Bureau.
 - 4. WCLIB: West Coast Lumber Inspection Bureau.
 - 5. WWFA: Western Wood Products Association.

SUBMITTALS, GENERAL



General: Submit all action submittals and informational submittals required by this Section concurrently.

ACTION SUBMITTALS

A. Product Data: For each type of process and factory-fabricated product. Indicate component materials and dimensions and include construction and application details.

Tetra Tech 200-15704-17001

- 1. Include data for wood-preservative treatment from chemical treatment manufacturer and certification by treating plant that treated materials comply with requirements. Indicate type of preservative used and net amount of preservative retained.
- 2. Include data for fire-retardant treatment from chemical treatment manufacturer and certification by treating plant that treated materials comply with requirements. Include physical properties of treated materials based on testing by a qualified independent testing agency.
- 3. For fire-retardant treatments, include physical properties of treated lumber both before and after exposure to elevated temperatures, based on testing by a qualified independent testing agency according to ASTM D 5664.
- 4. For products receiving a waterborne treatment, include statement that moisture content of treated materials was reduced to levels specified before shipment to Project see.
- 5. Include copies of warranties from chemical treatment manufacturers for each type of treatment.

1.6 INFORMATIONAL SUBMITTALS

- A. Evaluation Reports: For the following:
 - 1. Wood-preservative-treated wood.
 - 2. Fire-retardant-treated wood.

1.7 DELIVERY, STORAGE, AND HANDLE

A. Stack lumber flat with spacers bereat, and between each bundle to provide air circulation. Protect lumber from weather by covering with waterproof sheeting, securely anchored. Provide for air circulation around stacks and under coverings.

PART 2 - PRODUCTS

- 2.1 WOOD PROFUCTS, GENERAL
 - A. Certified Wood: Materials shall be produced from wood obtained from forests certified by an FSC accordited certification body to comply with FSC STD-01-001, "FSC Principles and Criteriz for Forest Stewardship."

Lumber: DOC PS 20 and applicable rules of grading agencies indicated. If no grading agency indicated, provide lumber that complies with the applicable rules of any rules-writing agency certified by the ALSC Board of Review. Provide lumber graded by an agency certified by the ALSC Board of Review to inspect and grade lumber under the rules indicated.

- 1. Factory mark each piece of lumber with grade stamp of grading agency.
- 2. For exposed lumber indicated to receive a stained or natural finish, mark grade stamp on end or back of each piece.
- 3. Where nominal sizes are indicated, provide actual sizes required by DOC PS 20 for moisture content specified. Where actual sizes are indicated, they are minimum dressed sizes for dry lumber.

- 4. Provide dressed lumber, S4S, unless otherwise indicated.
- C. Maximum Moisture Content of Lumber: 19 percent unless otherwise indicated.

2.2 WOOD-PRESERVATIVE-TREATED LUMBER

- A. Preservative Treatment by Pressure Process: AWPA U1; Use Category UC2 for interior construction not in contact with the ground and Use Category UC3b for exterior construction not in contact with the ground.
 - 1. Preservative Chemicals: Acceptable to authorities having jurisdiction and containing no arsenic or chromium.
 - 2. For exposed items indicated to receive a stained or natural finish, use chemical formulations that do not require incising, contain colorants, bleet through, or otherwise adversely affect finishes.
- B. Kiln-dry lumber after treatment to a maximum moisture content of 19 percent. Do not use material that is warped or that does not comply with requirements for untreated material.
- C. Mark lumber with treatment quality mark of an inspection agency approved by the ALSC Board of Review.
 - 1. For exposed lumber indicated to receive a stained or natural finish, mark end or back of each piece.
- D. Application: Treat items indicated or Drawings, and the following:
 - 1. Wood nailers, curbs, ecomposet support bases, blocking, stripping, and similar members in connection with rooking, flashing, vapor barriers, and waterproofing.
 - 2. Wood sills, sleepers, blocking, furring, stripping, and similar concealed members in contact with maconty of concrete.

2.3 FIRE-RETAR OAP T-TREATED MATERIALS

A. General: Where fire-retardant-treated materials are indicated, use materials complying with requirements in this article, that are acceptable to authorities having jurisdiction, and with fire-test response characteristics specified as determined by testing identical products per test method indicated by a qualified testing agency.



Fire-Retardant-Treated Lumber and Plywood by Pressure Process: Products with a flame spread index of 25 or less when tested according to ASTM E 84, and with no evidence of significant progressive combustion when the test is extended an additional 20 minutes, and with the flame front not extending more than 10.5 feet beyond the centerline of the burners at any time during the test.

- 1. Use treatment that does not promote corrosion of metal fasteners.
- 2. Exterior Type: Treated materials shall comply with requirements specified above for fire-retardant-treated lumber and plywood by pressure process after being subjected to

Tetra Tech 200-15704-17001

ROUGH CARPENTRY 06 10 00 - 3 accelerated weathering according to ASTM D 2898. Use for exterior locations and where indicated.

- 3. Interior Type A: Treated materials shall have a moisture content of 28 percent or less when tested according to ASTM D 3201 at 92 percent relative humidity. Use where exterior type is not indicated.
- 4. Design Value Adjustment Factors: Treated lumber shall be tested according ASTM D 5664 and design value adjustment factors shall be calculated according ASTM D 6841. For enclosed roof framing, framing in attic spaces, and where high, temperature fire-retardant treatment is indicated, provide material with adjustment tor of not less than 0.85 modulus of elasticity and 0.75 for extreme fiber in bending Project's climatological zone.
- C. Kiln-dry lumber after treatment to a maximum moisture content of 19 ercei Kiln-dry plywood after treatment to a maximum moisture content of 15 percent.
- Identify fire-retardant-treated wood with appropriate classification marking of qualified testing D. agency.
 - For exposed lumber indicated to receive a stained, tural finish, mark end or back of 1. each piece.
- E. For exposed items indicated to receive a stained or natural finish, use chemical formulations that do not bleed through, contain colorants, or otherwise adversely affect finishes.
- F. Application: Treat items indicated on Dra nd the following:
 - Framing for raised platforms 1.
 - Concealed blocking. 2.
 - 3. Plywood backing pane
- 2.4 **MISCELLANEOUS**
 - niscella lumber indicated and lumber for support or attachment of other A. General: Provid construction. ing the following:
 - 1. В bckir 2

 - ooftop equipment bases and support curbs. 3
 - Furring.
 - Grounds.



For items of dimension lumber size, provide Construction or No. 2 grade lumber and any of the following species:

- Hem-fir (north); NLGA. 1.
- 2. Mixed southern pine: SPIB.
- 3. Spruce-pine-fir; NLGA.
- Hem-fir; WCLIB or WWPA. 4.
- 5. Spruce-pine-fir (south); NeLMA, WCLIB, or WWPA.

ROUGH CARPENTRY 06 10 00 - 4

C. For blocking and nailers used for attachment of other construction, select and cut lumber to eliminate knots and other defects that will interfere with attachment of other work.

2.5 PLYWOOD BACKING PANELS

- A. Equipment Backing Panels: DOC PS 1, Exposure 1, C-D Plugged, fire-retardant treated, in thickness indicated or, if not indicated, not less than 3/4-inch nominal thickness.
 - 1. Plywood shall comply with the testing and product requirements of the Caliberra Department of Health Services' "Standard Practice for the Testing of Volaile Oganic Emissions from Various Sources Using Small-Scale Environmental Chamber,"

2.6 FASTENERS

- A. General: Provide fasteners of size and type indicated that comply wan requirements specified in this article for material and manufacture.
 - 1. Where rough carpentry is exposed to weather, in ground contact, pressure-preservative treated, or in area of high relative humidity, provide asteners of Type 304 stainless steel.
- B. Nails, Brads, and Staples: ASTM F 1667.
- C. Power-Driven Fasteners: NES NER-272.
- D. Wood Screws: ASME B18.6.1.
- E. Lag Bolts: ASME B18.2.1.
- F. Bolts: Steel bolts complying with ASTM A 307, Grade A; with ASTM A 563 hex nuts and, where indicated, flat washers.
- G. Expansion Anchors: Are nor bolt and sleeve assembly of material indicated below with capability to sistan, without failure, a load equal to six times the load imposed when installed in unit masonry assemblies and equal to four times the load imposed when installed in concrete as determined by testing per ASTM E 488 conducted by a qualified independent testing and inspecting agercy.

Material: Stainless steel with bolts and nuts complying with ASTM F 593 and ASTM F 594, Alloy Group 1 or 2.

MISCELLANEOUS MATERIALS

Water-Repellent Preservative: NWWDA-tested and -accepted formulation containing 3-iodo-2propynyl butyl carbamate, combined with an insecticide containing chloropyrifos as its active ingredient.

ADDITION AND RENOVATIONS

PART 3 - EXECUTION

3.1 INSTALLATION, GENERAL

- A. Set rough carpentry to required levels and lines, with members plumb, true to line, cut, and fitted. Fit rough carpentry to other construction; scribe and cope as needed for accurate ut. Locate furring, nailers, blocking, grounds, and similar supports to comply with requirements for attaching other construction.
- B. Framing Standard: Comply with AF&PA's WCD 1, "Details for Conventional Vod Frame Construction," unless otherwise indicated.
- C. Install plywood backing panels by fastening to studs; coordinate bearing with utilities requiring backing panels. Install fire-retardant treated plywood backing panels with classification marking of testing agency exposed to view.
- D. Do not splice structural members between supports unless otherwise indicated.
- E. Provide blocking and framing as indicated and as required to support facing materials, fixtures, specialty items, and trim.
- F. Provide fire blocking in furred spaces, stud spaces, and other concealed cavities as indicated. and as follows:
 - 1. Fire block furred spaces of walls, at each floor level, at ceiling, and at not more than 96 inches o.c. with solid wood blocking or noncombustible materials accurately fitted to close furred spaces.
 - 2. Fire block concealed spaces between floor sleepers with same material as sleepers to limit concealed space to not more than 100 sq. ft. and to solidly fill space below partitions.
 - 3. Fire block concealed spaces behind combustible cornices and exterior trim at not more than 20 feet o.c.
- G. Sort and select lumber so that natural characteristics will not interfere with installation or with fastening other materials to lumber. Do not use materials with defects that interfere with function of member or pieces that are too small to use with minimum number of joints or optimum joint arrangement.
- H. Compay with AWPA M4 for applying field treatment to cut surfaces of preservative-treated lumber.
 - Securely attach rough carpentry work to substrate by anchoring and fastening as indicated, complying with the following:
 - 1. NES NER-272 for power-driven fasteners.
 - 2. Table 2304.9.1, "Fastening Schedule," in building code in effect for Project.
- J. Use steel common nails unless otherwise indicated. Select fasteners of size that will not fully penetrate members where opposite side will be exposed to view or will receive finish materials.

ROUGH CARPENTRY 06 10 00 - 6 Tetra Tech 200-15704-17001

Make tight connections between members. Install fasteners without splitting wood. Drive nails snug but do not countersink nail heads unless otherwise indicated.

3.2 WOOD GROUND, SLEEPER, BLOCKING, AND NAILER INSTALLATION

- A. Install where indicated and where required for screeding or attaching other work. Form to shapes indicated and cut as required for true line and level of attached work. Coordinate locations with other work involved.
- B. Attach items to substrates to support applied loading. Recess bolts and nuts flush with surfaces unless otherwise indicated.
- C. Where wood-preservative-treated lumber is installed adjacent to netal backing, install continuous flexible flashing separator between wood and metal decking.

3.3 WOOD FURRING INSTALLATION

8-81

A. Install level and plumb with closure strips at edges and openings. Shim with wood as required for tolerance of finish work.

END OF SECTION



SECTION 06 13 23

HEAVY TIMBER CONSTRUCTION

PART 1 - GENERAL

1.1 **RELATED DOCUMENTS**

- A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 **SUMMARY**
 - Section includes truss framing using timbers. A.
- 1.3 DEFINITIONS
 - Timbers: Lumber of 5 inches nominal or greater in left dimension. A.
 - Inspection agencies, and the abbreviations used to reference them, include the following: B.
 - NeLMA Northeastern Lumber Manufacturers Association. 1.
 - NHLA National Hardwood Lumeer Association. NLGA National Lumber Grades Authority. 2.
 - 3.
 - SPIB Southern Pine Aspection Bureau. 4.
 - WCLIB West Coast Lumber Inspection Bureau. 5.
 - 6. WWPA - Wester Products Association.
- 1.4 **SUBMITTAL**
 - Product Vata: For preservative-treated wood products and timber connectors. A.
 - preservative-treated wood products, include chemical treatment manufacturer's written instructions for handling, storing, installing, and finishing treated material. For timber connectors, include installation instructions.



Shop Drawings: For heavy timber construction. Show layout, dimensions of each member, and details of connections.

Samples: Not less than 6 inches wide by 24 inches long, showing the range of variation to be expected in appearance, including surface texture, of wood products. Apply a coat of penetrating sealer to Samples.

D. Material Certificates:

- 1. For heavy timber construction specified to comply with minimum allowable unit stresses. Indicate species and grade selected for each use and design values approved by ALSC's Board of Review.
- E. Certificates of Inspection: Issued by lumber grading agency for exposed timber not marked with grade stamp.
- 1.5 QUALITY ASSURANCE
 - A. Timber Standard: Comply with AITC 108, "Standard for Heavy Timber Construction
 - B. Forest Certification: Provide wood products obtained from forests certified by an FSCaccredited certification body to comply with FSC STD-01-001, "FSC Principles and Criteria for Forest Stewardship."

1.6 DELIVERY, STORAGE, AND HANDLING

- A. Schedule delivery of heavy timber construction to avoir extended on-site storage and to avoid delaying the Work.
- B. Store materials under cover and protected from veather and contact with damp or wet surfaces. Provide for air circulation within and around stacks and under temporary coverings.

PART 2 - PRODUCTS

2.1 TIMBER

- A. General: Comply with DOC 2S 20 and with grading rules of lumber grading agencies certified by ALSC's Board of Nevewas applicable.
 - 1. Factory nark each item of timber with grade stamp of grading agency.
 - 2. For exposed timber indicated to receive a stained or natural finish, apply grade stamps to surfaces that will not be exposed to view, or omit grade stamps and provide certificates of grade compliance issued by grading agency.
- B. Timber Species and Grade: Douglas fir-larch, No. 2, NeLMA, NLGA, SPIB, WCLIB, or WWPA.
 - Allowable Stress Ratings for 12-Inch Nominal Depth: As required by engineered truss design.

Moisture Content: Provide timber with 19 percent maximum moisture content at time of dressing.

D. Dressing: Provide dressed timber (S4S) unless otherwise indicated.

- E. End Sealer: Manufacturer's standard, transparent, colorless wood sealer that is effective in retarding the transmission of moisture at cross-grain cuts and is compatible with indicated finish.
- F. Penetrating Sealer: Manufacturer's standard, transparent, penetrating wood sealer that is compatible with indicated finish.

2.2 TIMBER CONNECTORS

- A. General: Unless otherwise indicated, fabricate from the following materials:
 - 1. Structural-steel shapes, plates, and flat bars complying with ASTM A 3
 - 2. Round steel bars complying with ASTM A 575, Grade M 1020.
- B. Fabricate decorative gussets from structural steel, size as indicated in drawings.
- C. Fabricate tie rods from round steel bars with upset threads connected with forged-steel turnbuckles complying with ASTM A 668/A 668M.
- D. Provide bolts, 3/4 inch unless otherwise indicated, comparing with ASTM A 307, Grade A ; provide nuts complying with ASTM A 563; and, where indicated, provide flat washers.
- E. Finish steel assemblies and fasteners in shop win pairing system specified in section 09 91 23 INTERIOR PAINTING for "interior exposed mixcellaneous metals".

2.3 FABRICATION

- A. Shop fabricate members by stitting and restoring exposed surfaces to match specified surfacing. Finish exposed surfaces to narrow planing or surfacing marks, and to provide a finish equivalent to that produced by machine sanding with No. 120 grit sandpaper.
- B. Predrill for fasteners and exembly of units.
- C. Coat crosscuts with end sealer.
- D. Seal Coa: After fabricating and surfacing each unit, apply a saturation coat of penetrating sealer on surfaces of each unit except for treated wood where the treatment included a water repellent.

- EXECUTION

INSTALLATION

- A. General: Erect heavy timber construction true and plumb. Provide temporary bracing to maintain lines and levels until permanent supporting members are in place.
 - 1. Install heavy timber construction to comply with Shop Drawings.

Tetra Tech 200-15704-17001

- 2. Provide continuous members unless otherwise indicated; tie together over supports if not continuous.
- 3. Handle and temporarily support heavy timber construction to prevent surface damage, compression, and other effects that might interfere with indicated finish.
- B. Framing Built into Concrete: Provide 1/2-inch clearance at tops, sides, and ends of memberabuilt into masonry, bevel cut ends 3 inches; do not embed more than 4 inches unless otherwise indicated.
- C. Cutting: Avoid extra cutting after fabrication. Where field fitting is unavoidable, one requirements for shop fabrication.
- D. Fit members by cutting and restoring exposed surfaces to match specified surfacing. Predrill for fasteners and assembly of units.
 - 1. Finish exposed surfaces to remove planing or surfacing marks and to provide a finish equivalent to that produced by machine sanding with No. 120 and sandpaper.
 - 2. Coat crosscuts with end sealer.
- E. Install timber connectors as indicated.
 - 1. Unless otherwise indicated, install bolts with same orientation within each connection and in similar connections.
 - 2. Install bolts with orientation as indicated on if not indicated, as directed by Architect.

3.2 ADJUSTING

A. Repair damaged surfaces and huistes after completing erection. Replace damaged heavy timber construction if repairs are no approved by Architect.

END OF SECTION

SPEC SECTION 06 16 00

SHEATHING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Wall sheathing.
 - 2. Composite nail base insulated roof sheathing.
 - 3. Sheathing joint and penetration treatment.

1.3 SUBMITTALS, GENERAL

A. General: Submit all action submittals required by this Section concurrently.

1.4 ACTION SUBMITTALS

- A. Product Data: For each type of process and factory-fabricated product. Indicate component materials and dimensioner and include construction and application details.
- 1.5 DELIVERY, STORAGE, AND HANDLING
 - A. Stack papels f ht with spacers beneath and between each bundle to provide air circulation. Project sheading from weather by covering with waterproof sheeting, securely anchored. Provide for air circulation around stacks and under coverings.

2 - PRODUCTS

PERFORMANCE REQUIREMENTS

A. Fire-Test-Response Characteristics: For assemblies with fire-resistance ratings, provide materials and construction identical to those of assemblies tested for fire resistance per ASTM E 119 by a testing and inspecting agency acceptable to authorities having jurisdiction.

Tetra Tech 200-15704-17001

1. Fire-Resistance Ratings: Indicated by design designations from UL's "Fire Resistance Directory.".

2.2 WOOD PANEL PRODUCTS

- A. Emissions: Products shall meet the testing and product requirements of the California Department of Health Services' "Standard Practice for the Testing of Volatile Organic Emissions from Various Sources Using Small-Scale Environmental Chambers."
- B. Certified Wood: For the following wood products, provide materials produced from wood obtained from forests certified by an FSC-accredited certification body to comply with FSC STD-01-001, "FSC Principles and Criteria for Forest Stewardship":
 - 1. Oriented strand board.
- C. Oriented Strand Board: DOC PS 2.
- D. Thickness: As needed to comply with requirements specified, jut not less than thickness indicated.
- E. Factory mark panels to indicate compliance with applicable tandard.

2.3 WALL SHEATHING

- A. Glass-Mat Gypsum Wall Sheathing: STM O1177/1177M.
 - 1. Basis-of-Design Product. Subject to compliance with requirements, provide G-P Gypsum Corporation; Dens-Class Exterior Sheathing, or comparable product.
 - 2. Type and Thickness. Type **X**, 5/8 inch thick.

2.4 COMPOSITE NAIL BASY INSULATED ROOF SHEATHING

A. Vented, Orienter-Strand-Board-Surfaced, Polyisocyanurate-Foam Sheathing: Rigid, cellular, polyisocranurae thermal insulation complying with ASTM C 1289, Type II, Class 1, with oriented strand poard adhered to spacers on one face.

Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to, the following:

- a. Cornell Corporation.
- b. Rmax, Inc.
- 2. Polyisocyanurate-Foam Thickness: 4 inches.
- 3. Oriented-Strand-Board Nominal Thickness: 7/16 inch.
- 4. Spacers: Wood furring strips or blocks not less than 3/4 inch thick and spaced not more than 24 inches o.c.

SHEATHING 06 16 00 - 2

2.5 FASTENERS

- A. General: Provide fasteners of size and type indicated that comply with requirements specified in this article for material and manufacture.
 - 1. For roof and wall sheathing, provide fasteners of Type 304 stainless steel.
- B. Nails, Brads, and Staples: ASTM F 1667.
- C. Power-Driven Fasteners: NES NER-272.
- D. Wood Screws: ASME B18.6.1.
- E. Screws for Fastening Gypsum Sheathing to Cold-Formed Metal Framine Ceel dull screws, in length recommended by sheathing manufacturer for thickness of sheathing to be attached, with organic-polymer or other corrosion-protective coating having a sat-spray resistance of more than 800 hours according to ASTM B 117.
 - 1. For steel framing less than 0.0329 inch thick, use crews that comply with ASTM C 1002.
 - 2. For steel framing from 0.033 to 0.112 inch blck, use screws that comply with ASTM C 954.
- F. Screws for Fastening Oriented-Strand-Board-Sterfaced, Polyisocyanurate-Foam Sheathing to Metal Roof Deck: Steel drill screws in type and length recommended by sheathing manufacturer for thickness of sheathing to be attached, with organic-polymer or other corrosion-protective coating having a solt-spray resistance of more than 800 hours according to ASTM B 117. Provide washers or plates if recommended by sheathing manufacturer.

2.6 SHEATHING JOINT-AND PENETRATION TREATMENT MATERIALS

- A. Sealant for Glass-Met Gyosum Sheathing: Silicone emulsion sealant complying with ASTM C 834, compatible with sheathing tape and sheathing and recommended by tape and sheathing manufacturers for use with glass-fiber sheathing tape and for covering exposed fasteners.
 - 1. Sheathing Tape: Self-adhering glass-fiber tape, minimum 2 inches wide, 10 by 10 or 10 by 20 threads/inch, of type recommended by sheathing and tape manufacturers for use with silicone emulsion sealant in sealing joints in glass-mat gypsum sheathing and with a history of successful in-service use.

- EXECUTION

INSTALLATION, GENERAL

A. Beginning installation constitutes Contractor's acceptance of substrates and conditions.

- B. Do not use materials with defects that impair quality of sheathing or pieces that are too small to use with minimum number of joints or optimum joint arrangement. Arrange joints so that pieces do not span between fewer than three support members.
- C. Cut panels at penetrations, edges, and other obstructions of work; fit tightly against abutting construction unless otherwise indicated.
- D. Securely attach to substrate by fastening as indicated, complying with the following:
 - 1. NES NER-272 for power-driven fasteners.
 - 2. Table 2304.9.1, "Fastening Schedule," in building code in effect for Project.
- E. Coordinate wall and roof sheathing installation with flashing and joint-sealant installation so these materials are installed in sequence and manner that prevent exterior most as from passing through completed assembly.
- F. Coordinate sheathing installation with installation of materials betalled over sheathing so sheathing is not exposed to precipitation or left exposed at end of the workday when rain is forecast.

3.2 GYPSUM SHEATHING INSTALLATION

- A. Comply with GA-253 and with manufacturer whitten instructions.
 - 1. Fasten gypsum sheathing to cold formed metal framing with screws.
 - 2. Install boards with a 3/8-inch g n where non-load-bearing construction abuts structural elements.
 - 3. Install boards with a 1/+ inclusion where they abut masonry or similar materials that might retain moisture, o prevent wicking.
- B. Apply fasteners so here's lear ightly against face of sheathing, but do not cut into facing.
- C. Horizontal Installation: Unstall sheathing with V-grooved edge down and tongue edge up. Interlock tongue with groove to bring long edges in contact with edges of adjacent boards without foreing. Abut ends of boards over centers of studs, and stagger end joints of adjacent boards not less than one stud spacing. Attach boards at perimeter and within field of board to each steel stud

Space fasteners approximately 8 inches o.c. and set back a minimum of 3/8 inch from edges and ends of boards.

Vertical Installation: Install board vertical edges centered over studs. Abut ends and edges of each board with those of adjacent boards. Attach boards at perimeter and within field of board to each stud.

- 1. Space fasteners approximately 8 inches o.c. and set back a minimum of 3/8 inch from edges and ends of boards.
- E. Seal sheathing joints according to sheathing manufacturer's written instructions.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

1. Apply glass-fiber sheathing tape to glass-mat gypsum sheathing joints and apply and trowel silicone emulsion sealant to embed entire face of tape in sealant. Apply sealant to exposed fasteners with a trowel so fasteners are completely covered. Seal other penetrations and openings.

END OF SECTION Reput <



SPEC SECTION 06 40 23

INTERIOR ARCHITECTURAL WOODWORK

PART 1 - GENERAL

- 1.1 RELATED DOCUMENTS
 - A. Drawings and general provisions of the Contract, including General and Supplementar Conditions and Division 1 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. This Section includes the following:
 - 1. Plastic-laminate cabinets.
 - 2. Plastic-laminate countertops.
 - 3. Solid-surfacing-material countertops.
 - 4. Closet and utility shelving.
 - 5. Interior wood window sill, jamb and apron,
 - 6. Interior standing and running trim.

1.3 DEFINITIONS

A. Interior architectural woodwork includes woodfurring, blocking, shims, and hanging strips for installing woodwork items unless concealed within other construction before woodwork installation.

1.4 SUBMITTALS

- A. Product Data: For high pressure decorative laminate, adhesive for bonding plastic laminate, solid-surfacing meterial, croinet hardware and accessories, and finishing materials and processes.
 - 1. Include tata for fire-retardant treatment from chemical treatment manufacturer and certification by treating plant that treated materials comply with requirements.
- B. Shop grawings: Show location of each item, dimensioned plans and elevations, large-scale details, attachment devices, and other components.
 - Show locations and sizes of furring, blocking, and hanging strips, including concealed blocking and reinforcement specified in other Sections.
 - 2. Show locations and sizes of cutouts and holes for plumbing fixtures, faucets, soap dispensers, and other items installed in architectural woodwork.
 - 3. APPLY AWI-CERTIFIED COMPLIANCE LABEL TO FIRST PAGE OF SHOP DRAWINGS.
 - 4. Show compliance with interior woodwork grade specified in Section 2, <u>ON FRONT</u> <u>OVER OF SUBMITTAL.</u>
- C. Samples for Initial Selection:

- 1. Plastic laminates.
- 2. PVC edge material.
- 3. Solid-surfacing materials.
- 4. Exposed cabinet hardware and accessories, one unit for each type and finish.
- D. Samples for Verification:
 - 1. Plastic laminates, 8 by 10 inches, for each type, color, pattern, and surface finish, with sample applied to core material and specified edge material applied to 1 edge.
 - 2. Solid-surfacing materials, 6 inches square.
 - 3. Exposed cabinet hardware and accessories, one unit for each type and finish.
 - 4. Shop applied transparent finishes on 6"x 6" sample of oak window sill and on growth wood paneling.
- E. Casework Samples:
 - 1. Base cabinet: Cabinet conforming to specifications, with drawer and door.
 - 2. Wall cabinet: Cabinet conforming to specifications, with dowr.
 - 3. Cabinet samples shall be complete with specified hardware for doors, drawers and shelves.
 - 4. Component samples: Two sets of samples for each of the following for each color specified in section:
 - a. Decorative laminate color charts
 - b. PVC and ABS edgings.
- F. Woodwork Quality Standard Compliance Certificates: AWI Quality Certification Program certificates.
- G. Qualification Data: For Instance and faoricator.
- 1.5 QUALITY ASSURANCE
 - A. Fabricator Qualifications from that employs skilled workers who custom-fabricate products similar to those required for this Project and whose products have a record of successful inservice performance. Shop is a certified participant in AWI's Quality Certification Program.
 - B. Installer Qualifications: Certified participant in AWI's Quality Certification Program.
 - C. Quarty Standard: Unless otherwise indicated, comply with AWI's "Architectural Woodwork Quality Standards" for grades of interior architectural woodwork indicated for construction, inishes, installation, and other requirements.
 - 1. Provide AWI Quality Certification Program labels and certificates indicating that woodwork, including installation, complies with requirements of grades specified.
 - . Fire-Test-Response Characteristics: Where fire-retardant materials or products are indicated, provide materials and products with specified fire-test-response characteristics as determined by testing identical products per test method indicated by UL, ITS, or another testing and inspecting agency acceptable to authorities having jurisdiction. Identify with appropriate markings of applicable testing and inspecting agency in the form of separable paper label or, where required by authorities having jurisdiction, imprint on surfaces of materials that will be concealed from view after installation.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

1.6 DELIVERY, STORAGE, AND HANDLING

- A. Do not deliver woodwork until painting and similar operations that could damage woodwork have been completed in installation areas. If woodwork must be stored in other than installation areas, store only in areas where environmental conditions comply with requirements specified in "Project Conditions" Article.
- B. Protect work from moisture damage according to QSI, Section 1700, "Installation."

1.7 PROJECT CONDITIONS

- A. Environmental Limitations: Do not deliver or install woodwork until building is endowed, we work is complete, and HVAC system is operating and maintaining temperature and relative humidity at occupancy levels during the remainder of the construction period.
- B. Environmental Limitations: Do not deliver or install woodwork up a bilding is enclosed, wet work is complete, and HVAC system is operating and maintaining temperature between 60 and 90 deg F (16 and 32 deg C) and relative humidity between 25 and 35 percent during the remainder of the construction period.
- C. Field Measurements: Where woodwork is indicated to fixe other construction, verify dimensions of other construction by field measurements before fabrication, and indicate measurements on Shop Drawings. Coordinate fabrication schedule with construction progress to avoid delaying the Work.
 - 1. Locate concealed framing, blocking, and reinforcements that support woodwork by field measurements before being paceases and indicate measurements on Shop Drawings.

1.8 COORDINATION

A. Coordinate sizes and locations of framing, blocking, furring, reinforcements, and other related units of Work specification other Sections to ensure that interior architectural woodwork can be supported and installed as indicated.

PART 2 - PRODUCTS

Α.

2.1 WOODWORK FABRICATORS

Fabricators who are members in good standing of the Architectural Woodwork Institute (AWI) and are familiar with this Standard.

MATERIALS

General: Provide materials that comply with requirements of AWI's quality standard for each type of woodwork and quality grade specified, unless otherwise indicated.

- 1. Particleboard: ANSI A208.1, Grade M-2.
- B. Thermoset Decorative Panels: Particle board finished with thermally fused, melamineimpregnated decorative paper complying with LMA SAT-1.
 - 1. Provide PVC or polyester edge banding complying with LMA EDG-1 on components with exposed or semiexposed edges.

- C. High –Pressure Decorative Laminate: NEMA LD 3, grades as indicated or, if not indicated, as required by woodwork quality standard.
 - 1. Manufacturer: Subject to compliance with requirements, provide high pressure decorative laminates by one of the following:
 - a. Formica Corporation;
 - b. Nevamar Company, LLC, Decorative Products Div.;
 - c. Wilsonart International, Div. Of Premark International, Inc.; or
- D. Solid-Surfacing Material: Homogeneous solid sheets of filled plastic resin complyin ISSFA-2.
 - 1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
 - a. Du Pont Corian.
 - b. Wilsonart International; Div. of Premark International In-
 - c. Approved Equal.

2.3 FIRE-RETARDANT-TREATED MATERIALS

- A. General: Where fire-retardant-treated materials are indicated, use materials complying with requirements in this Article that are acceptable to sub-rities having jurisdiction and with fire-test-response characteristics specified.
 - 1. Do not use treated materials that to not comply with requirements of referenced woodworking standard or that are warped, discolored, or otherwise defective.
 - 2. Use fire-retardant-treatment formulations that do not bleed through or otherwise adversely affect finishes. Do not use colorants to distinguish treated materials from untreated materials.
 - 3. Identify fire-retardant treated materials with appropriate classification marking of UL, U.S. Testing, Theber Products Inspection, or another testing and inspecting agency acceptable to authorities having jurisdiction.
- B. Fire-Retardart-Reated Lumber and Plywood by Pressure Process: Comply with performance requirements of AWPA C20 (lumber) and AWPA C27 (plywood). Use the following treatment type



Aill lumber after treatment within limits set for wood removal that do not affect listed fire-test-response characteristics, using a woodworking plant certified by testing and inspecting agency.

Fire-Retardant Particleboard: Panels complying with the following requirements, made from softwood particles and fire-retardant chemicals mixed together at time of panel manufacture to achieve flame-spread index of 25 or less and smoke-developed index of 25 or less per ASTM E 84.

1. For panels 3/4 inch (19 mm) thick and less, comply with ANSI A208.1 for Grade M-2 except for the following minimum properties: modulus of rupture, 1600 psi (11 MPa); modulus of elasticity, 300,000 psi (2070 MPa); internal bond, 80 psi (550 kPa); and screw-holding capacity on face and edge, 250 and 225 lbf (1100 and 1000 N),

Tetra Tech 200-15704-17001

2. Product: Subject to compliance with requirements, provide "Duraflake FR" by Weyerhaeuser.

2.4 CABINET HARDWARE AND ACCESSORIES

- A. General: Provide cabinet hardware and accessory materials associated with architectural cabinets.
- B. Butt Hinges: 2-3/4-inch (70-mm), 5-knuckle steel hinges made from 0.095-inch- (2.4-m thick metal, and as follows:
 - 1. Semiconcealed Hinges for Overlay Doors: BHMA A156.9, B01521.
- C. Wire Pulls: Back mounted, solid metal, 4 inches (100 mm) long, 5/16 in (10 mm) in diameter.
- D. Catches: Roller catches, BHMA A156.9, B03071.
- E. Shelf Rests: BHMA A156.9, B04013; metal, two-pin type with shelf cold-down clip.
- F. Aluminum Slides for Sliding Glass Doors: BHMA A15...9, 30700
- G. Drawer Slides: BHMA A156.9, B05091.
 - 1. Heavy Duty (Grade 1HD-100 and Grade ND-200): Side mounted; full-extension type; zinc-plated steel ball-bearing slides
 - 2. Box Drawer Slides: Grade 1HD 400, for drawers not more than 6 inches (150 mm) high and 24 inches (600 mm) wide
 - 3. File Drawer Slides: Grade (HD-100; for drawers more than 6 inches (150 mm) high or 24 inches (600 mm) wide.
 - 4. Pencil Drawer Slides: Grade 2; for drawers not more than 3 inches (75 mm) high and 24 inches (600 mm) wide.
- H. Door Locks: BHMA 156.1, E07121. Provide locks at all cabinets.
- I. Drawer Locks, BJ MA A156.11, E07041. Provide locks at all drawers
- J. Grommers for Vable Passage through Countertops: 3-inch OD, as selected by architect from available standard colors, molded-plastic grommets and matching plastic caps with slot for wire passage.



Product: Subject to compliance with requirements, provide "SG series" by Doug Mockett & Company, Inc.

Exposed Hardware Finishes: For exposed hardware, provide finish that complies with BHMA A156.18 for BHMA finish number indicated.

- 1. Satin Stainless Steel: BHMA 630.
- L. For concealed hardware, provide manufacturer's standard finish that complies with product class requirements in BHMA A156.9.

ADDITION AND RENOVATIONS

2.5 MISCELLANEOUS MATERIALS

- A. Furring, Blocking, Shims, and Hanging Strips: Fire-retardant-treated softwood lumber, kiln dried to less than 15 percent moisture content.
- B. Anchors: Select material, type, size, and finish required for each substrate for secure anchorage Provide nonferrous-metal or hot-dip galvanized anchors and inserts on inside face of exterior walls and elsewhere as required for corrosion resistance. Provide toothed-steel or lead expansion sleeves for drilled-in-place anchors.
- C. Adhesives, General: Do not use adhesives that contain urea formaldehyde.
- D. VOC Limits for Installation Adhesives and Glues: Use installation adhesives that comply with the following limits for VOC content when calculated according to 40 CDR 39, Subpart D (EPA Method 24):
 - 1. Wood Glues: 30 g/L.
 - 2. Contact Adhesive (Multipurpose Construction Adhesiyes): 250 g/L.
- E. Adhesive for Bonding Plastic Laminate: Unpigmented contact cement
 - 1. Adhesive for Bonding Edges: Hot-melt adhesive or adhesive specified above for faces.

2.6 FABRICATION, GENERAL

- A. Interior Woodwork Grade: Unless otherwise indicated, provide <u>Premium-grade</u> interior woodwork complying with referenced quality standard.
- B. Wood Moisture Content: Compay with requirements of referenced quality standard for wood moisture content in relation to ambient relative humidity during fabrication and in installation areas.
- C. Sand fire-retardant-treated wood lightly to remove raised grain on exposed surfaces before fabrication.
- D. Fabricate wood ork to dimensions, profiles, and details indicated. Ease edges to radius indicated for the following:
 - . Corners of Cabinets and Edges of Solid-Wood (Lumber) Members and Rails: 1/16 inch (1.5 mm).



Complete fabrication, including assembly, finishing, and hardware application, to maximum extent possible before shipment to Project site. Disassemble components only as necessary for shipment and installation. Where necessary for fitting at site, provide ample allowance for scribing, trimming, and fitting.

- Shop-cut openings to maximum extent possible to receive hardware, appliances, plumbing fixtures, electrical work, and similar items. Locate openings accurately and use templates or roughing-in diagrams to produce accurately sized and shaped openings. Sand edges of cutouts to remove splinters and burrs.
 - 1. Seal edges of openings in countertops with a coat of varnish.

2.7 PLASTIC-LAMINATE CABINETS

- A. Grade: <u>Premium.</u>
- B. AWI Type of Cabinet Construction: Flush overlay.
- C. Laminate Cladding for Exposed Surfaces: High-pressure decorative laminate complying with the following requirements:
 - 1. Horizontal Surfaces Other Than Tops: Grade HGS.
 - 2. Postformed Surfaces: Grade HGP.
 - 3. Vertical Surfaces: Grade HGS.
 - 4. Edges: PVC T-mold matching laminate in color, pattern, and finish
- D. Materials for Semiexposed Surfaces:
 - 1. Surfaces Other Than Drawer Bodies: High-pressure decorative familiate, Grade VGS.
 - a. For semiexposed backs of panels with exposed postic-aminate surfaces, provide surface of high-pressure decorative laminate, G ade +GS or CLS.
 - 2. Drawer Sides and Backs: Thermoset decorative panel.
 - 3. Drawer Bottoms: Thermoset decorative pinels
- E. Concealed Backs of Panels with Exposed Plastic Laminate Surfaces: High-pressure decorative laminate, Grade BKL.
- F. Colors, Patterns, and Finishes: Provide materials and products that result in colors and textures of exposed laminate surfaces complying with the following requirements:
 - 1. As selected by Archiect iron laminate manufacturer's full range in the following categories:
 - a. Solid colors, natte finish.
 - b. Solid colors with core same color as surface, matte finish.
 - c. Paterns, matte finish.

2.8 PLASTIC LAMINATE COUNTERTOPS

A. Grade <u>Premium</u>.

Countertop Thickness: 1 ¹/₂" inch.

High-Pressure Decorative Laminate Grade: HGS.

Colors, Patterns, and Finishes: Provide materials and products that result in colors and textures of exposed laminate surfaces complying with the following requirements:

- 1. As selected by Architect from manufacturer's full range in the following categories:
 - a. Solid colors, matte finish.
 - b. Solid colors with core same color as surface, matte finish.
 - c. Patterns, matte finish.

- E. Edge Treatment: Lumber edge for transparent finish matching wood species and cut on cabinet surfaces.
- F. Core Material: Particleboard, Fire-retardant particleboard.
- G. Core Material at Sinks: Particleboard made with exterior glue.
- H. Backer Sheet: Provide plastic-laminate backer sheet, Grade BKL, on underside of countertop substrate.
- 2.9 SOLID-SURFACING-MATERIAL COUNTERTOPS
 - A. Grade: Premium.
 - B. Solid-Surfacing-Material Thickness: 3/4 inch (19 mm).
 - C. Colors, Patterns, and Finishes: Provide materials and products that your in colors of solidsurfacing material complying with the following requirements:
 - 1. As selected by Architect from manufacturer's full care. Frovide a minimum of Price Group F as it relates to Corian's price grouping of caual grouping from accepted manufacturers as mentioned previously.
 - D. Fabricate tops in one piece, unless otherwise indicated. Comply with solid-surfacing-material manufacturer's written recommendations for addresives, sealers, fabrication, and finishing.
 - E. Install integral sink bowls in countertops in Nurses Suite.
 - F. Drill holes in countertops for plansbing fatings in shop.
- 2.10 INTERIOR STANDING AND RUNNING TRIM FOR TRANSPARENT FINISH
 - A. Grade: Premium
 - B. Wood Species . d Cut: Red oak, quarter sawn.
 - C. For trim tems wider than available lumber, use veneered construction. Do not glue for width.
 - D. For hils wider or thicker than available lumber, use veneered construction. Do not glue for width or thickness.
 - Backout or groove backs of flat trim members and kerf backs of other wide, flat members, except for members with ends exposed in finished work.
 - Assemble casings in plant except where limitations of access to place of installation require field assembly.

2.11 STAIRWORK

A. Grade: Premium.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- B. Wood Species and Cut for Transparent Finish: Red oak, quarter sawn.
- C. Finishes for Stair Parts: As follows:
 - 1. Treads: Transparent.
 - 2. Risers: Transparent.
- 2.12 INTERIOR WOOD SILL, JAMB AND HEAD TRIM
 - A. Quality Standard: comply with AWI Section 300.
 - B. Grade: Premium.
 - C. Boards to be solid in width. Do not glue for width.
 - D. Wood Species: Quarter Sawn Red Oak.
 - E. Finishing: Seal and varnish exposed and semi-exposed to view surfaces. Brush apply only.
- 2.13 CLOSET AND UTILITY SHELVING
 - A. Quality Standard: comply with AWI Section 600
 - B. Grade: Premium.
 - C. Shelf Material: Veneer faced particle beard with veneer edge banding
 - D. Locations: Provide Shelf and Rod at all closet locations.
- 2.14 SHOP FINISHING
 - A. General: Shop finish transparent finished interior architectural woodwork at fabrication shop as specified in this Section Refer to Division 9 Section "Painting" for finishing opaque finished architectural woodwork.
 - B. Transparent Finan: Comply with requirements indicated below for grade, finish system, staining, and sizes, with sheen measured on 60-degree gloss meter per ASTM D 523:
 - . Crade. Premium.
 - AWI Finish System: Catalyzed polyurethane.
 - Staining: Match approved sample for color.
 - Sheen: Gloss, 80-100 gloss units.
 - Color: As selected by Architect from the full range of colors available in finish system specified.

PART 3 - EXECUTION

3.1 PREPARATION

A. Before installation, condition woodwork to average prevailing humidity conditions in installation areas.

B. Before installing architectural woodwork, examine shop-fabricated work for completion and complete work as required, including removal of packing and backpriming.

3.2 INSTALLATION

- A. Grade: Install woodwork to comply with requirements for the same grade specified in Part 2 or fabrication of type of woodwork involved.
- B. Assemble woodwork and complete fabrication at Project site to comply with requirement fabrication in Part 2, to extent that it was not completed in the shop.
- C. Install woodwork level, plumb, true, and straight. Shim as required with concelled shims. Install level and plumb (including tops) to a tolerance of 1/8 inch in 96 ip ne.
- D. Scribe and cut woodwork to fit adjoining work, refinish cut surfaces, and repair damaged finish at cuts.
- E. Fire-Retardant-Treated Wood: Handle, store, and install fire-retardant-treated wood to comply with chemical treatment manufacturer's written instructions increasing those for adhesives used to install woodwork.
- F. Anchor woodwork to anchors or blocking built in or anectly attached to substrates. Secure with countersunk, concealed fasteners and blind milling as required for complete installation. Use fine finishing nails or finishing screws for exposed fastening, countersunk and filled flush with woodwork and matching final finish if thense arent finish is indicated.
- G. Standing and Running Trim: Install with minimum number of joints possible, using full-length pieces (from maximum length of lumber available) to greatest extent possible. Do not use pieces less than 60 inches long, except where shorter single-length pieces are necessary. Scarf running joints and stagger madjacent and related members.
 - 1. Fill gaps, if any, between top of base and wall with plastic wood filler, sand smooth, and finish same as wood base if finished.
 - 2. Install call ailings on indicated metal brackets securely fastened to wall framing.
 - 3. Install standing and running trim with no more variation from a straight line than 1/8 inch in 96 inches.
- H. Panaling: Anchor paneling to supporting substrate with concealed panel-hanger clips. Do not use face fastening, unless covered by trim or[otherwise indicated.

Install flush paneling with no more than 1/16 inch in 96-inch vertical cup or bow and 1/8 inch in 96-inch horizontal variation from a true plane.

Stairs: Securely anchor carriages to supporting substrates. Install stairs with treads and risers no more than 1/8 inch from indicated position.

- J. Cabinets: Install without distortion so doors and drawers fit openings properly and are accurately aligned. Adjust hardware to center doors and drawers in openings and to provide unencumbered operation. Complete installation of hardware and accessory items as indicated.
 - 1. Install cabinets with no more than 1/8 inch in 96-inch sag, bow, or other variation from a straight line.
CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- 2. Fasten wall cabinets through back, near top and bottom, at ends and not more than 16 inches o.c. with No. 10 wafer-head screws sized for 1-inch penetration into wood framing, blocking, or hanging strips.
- K. Countertops: Anchor securely by screwing through corner blocks of base cabinets or other supports into underside of countertop.
 - 1. Align adjacent solid-surfacing-material countertops and form seams to comple with manufacturer's written recommendations using adhesive in color to match countertop. Carefully dress joints smooth, remove surface scratches, and clean entire surface.
 - 2. Install countertops with no more than 1/8 inch in 96-inch sag, bow, or other valiation from a straight line.
 - 3. Secure backsplashes to tops with concealed metal brackets at 16 inche o.c. and to walls with adhesive.
 - 4. Caulk space between backsplash and wall with sealant specified in Division 07 Section "Joint Sealants."
- L. Touch up finishing work specified in this Section after installation of woodwork. Fill nail holes with matching filler where exposed.
- M. Refer to Division 09 Sections for final finishing of estalled architectural woodwork not indicated to be shop finished.

3.3 ADJUSTING AND CLEANING

- A. Repair damaged and defective wordwork, where possible, to eliminate functional and visual defects; where not possible to repair replace woodwork. Adjust joinery for uniform appearance.
- B. Clean, lubricate, and adjust hard var
- C. Clean woodwork on exposed and semiexposed surfaces. Touch up shop-applied finishes to restore damaged a soiled areas.

END OF SECTION



SECTION 06 41 16

PLASTIC-LAMINATE-FACED ARCHITECTURAL CABINETS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Plastic-laminate-faced architectural cabinets.
 - 2. Wood furring, blocking, shims, and hanging stript for installing plastic-laminate-faced architectural cabinets unless concealed within other construction before cabinet installation.
 - 3. Reception Desk and Cubbies

B. Related Requirements:

- 1. Section 06 10 00 "Rough Carpenery" for wood furring, blocking, shims, and hanging strips required for installing abouts and concealed within other construction before cabinet installation.
- 2. Section 12 36 23.13 Plastic Laminate-Clad Countertops."
- 3. Section 12 36 6... Sold Surfacing Countertops.

1.3 PREINSTALLATION MEETINGS

A. Preinstal ation Conference: Conduct conference at Project site.

1.4 ACTION SUBMITTALS

roduct Data: For each type of product, including panel products high-pressure decorative laminate cabinet hardware and accessories.

- 1. Include data for fire-retardant treatment from chemical-treatment manufacturer and certification by treating plant that treated materials comply with requirements.
- B. Shop Drawings: Show location of each item, dimensioned plans and elevations, large-scale details, attachment devices, and other components.
 - 1. Show details full size.

- 2. Show locations and sizes of furring, blocking, and hanging strips, including concealed blocking and reinforcement specified in other Sections.
- 3. Show locations and sizes of cutouts and holes for electrical switches and outlets and other items installed in architectural plastic-laminate cabinets.
- 4. Apply WI Certified Compliance Program label to Shop Drawings.
- 5. Apply AWI Quality Certification Program label to Shop Drawings.
- C. Samples for Initial Selection:
 - 1. Plastic laminates.
 - 2. PVC edge material.
 - 3. Thermoset decorative panels.
- D. Samples for Verification:
 - 1. Plastic laminates, 8 by 10 inches, for each type, color, pattern, and subjace finish.
 - 2. Wood-grain plastic laminates, 12 by 24 inches, for each type, attern and surface finish.
 - 3. Thermoset decorative panels, 8 by 10 inches, for each color, pattern, and surface finish.
 - 4. Corner pieces as follows:
 - a. Cabinet-front frame joints between stiles and rails, as well as exposed end pieces, 18 inches high by 18 inches wide by o inches deep.
 - b. Miter joints for standing trim.
 - 5. Exposed cabinet hardware and acceleration one unit for each type and finish.

1.5 INFORMATIONAL SUBMITTAL

- A. Qualification Data: For Instaner/fablicator.
- B. Product Certificates: For each type of product the following:
 - 1. Composite yood and agrifiber products.
 - 2. Thermoset seconative panels.
 - 3. High pressure decorative laminate.
 - 4. Glass.
 - 5. Accesives
- C. Woodwork Quality Standard Compliance Certificates: AWI Quality Certification Program certificates WI Certified Compliance Program certificates.

Evaluation Reports: For fire-retardant-treated materials, from ICC-ES.

QUALITY ASSURANCE

A. Fabricator Qualifications: Shop that employs skilled workers who custom fabricate products similar to those required for this Project and whose products have a record of successful inservice performance. Shop is a certified participant in AWI's Quality Certification Program. Shop is a licensee of WI's Certified Compliance Program.

- B. Installer Qualifications: Fabricator of products Certified participant in AWI's Quality Certification Program Licensee of WI's Certified Compliance Program.
- C. Testing Agency Qualifications: For testing agency providing classification marking for fireretardant-treated material, an inspection agency acceptable to authorities having jurisdiction that periodically performs inspections to verify that the material bearing the classification marking representative of the material tested.

1.7 DELIVERY, STORAGE, AND HANDLING

A. Do not deliver cabinets until painting and similar operations that could damage woodwork have been completed in installation areas. If cabinets must be stored in other than installation areas, store only in areas where environmental conditions comply with requirements specified in "Field Conditions" Article.

1.8 FIELD CONDITIONS

- A. Environmental Limitations: Do not deliver or install abirets until building is enclosed, wet work is complete, and HVAC system is operating and maintaining temperature and relative humidity at occupancy levels during the remainder of the construction period.
- B. Environmental Limitations: Do not deliver or install cabinets until building is enclosed, wet work is complete, and HVAC system is operating and maintaining temperature between 60 and 90 deg F and relative humidity between 25 and 55 percent during the remainder of the construction period.
- C. Field Measurements: Where canintry are indicated to fit to other construction, verify dimensions of other construction by field measurements before fabrication, and indicate measurements on Shop Drawing a Coordinate fabrication schedule with construction progress to avoid delaying the Wark.
 - 1. Locate conrealed naming, blocking, and reinforcements that support cabinets by field measurements before being enclosed, and indicate measurements on Shop Drawings.
- D. Established Dimensions: Where cabinets are indicated to fit to other construction, establish dimensions for areas where cabinets are to fit. Provide allowance for trimming at site, and coordinate construction to ensure that actual dimensions correspond to established dimensions.

CORDINATION

Coordinate sizes and locations of framing, blocking, furring, reinforcements, and other related units of Work specified in other Sections to ensure that cabinets can be supported and installed as indicated.

B. Hardware Coordination: Distribute copies of approved hardware schedule specified in Section 08 71 11 "Door Hardware (Descriptive Specification)" to fabricator of architectural woodwork; coordinate Shop Drawings and fabrication with hardware requirements.

ADDITION AND RENOVATIONS

PART 2 - PRODUCTS

2.1 PLASTIC-LAMINATE-FACED ARCHITECTURAL CABINETS

- A. Quality Standard: Unless otherwise indicated, comply with the "Architectural Woodworks Standards" for grades of architectural plastic-laminate cabinets indicated for construction finishes, installation, and other requirements.
 - 1. Provide labels and certificates from AWI WI certification program indicating the woodwork, including installation, complies with requirements of grades specified.
 - 2. The Contract Documents contain selections chosen from options in the quality standard and additional requirements beyond those of the quality standard. Comply with those selections and requirements in addition to the quality standard.
- B. Grade: Premium.
- C. Type of Construction: Frameless.
- D. Cabinet, Door, and Drawer Front Interface Style: Flush over ay
- E. Reveal Dimension: 1/2 inch.
- F. High-Pressure Decorative Laminate: NEMALD 3, grades as indicated or if not indicated, as required by woodwork quality standard.
 - 1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
 - a. Basis of Design Wilsonart. (See Finish Schedule for style and color)
 - b. Formica Corporation
 - c. Pionite; Palelan Industries International, Inc. brand.
- G. Laminate Cladding for Exposed Surfaces: (Including Student Cubbies).
 - 1. Horizonta Surfaces: Grade HGS.
 - 2. Vertical Surfaces: Grade HGS.
 - Edges: oMM PVC edge banding, selected from a minimum of 12 colors. 3MM to be olid, high impact, purified, color-thru, acid resistant, pre-lamination primed edging, nachine-applied with hot melt adhesives, automatically trimmed, inside/outside length-radiused for uniform appearance, buffed and corner-radiused for consistent design.
 Pattern Direction: Vertically for drawer fronts, doors, and fixed panels.

Materials for Semiexposed Surfaces:

- 1. Surfaces Other Than Drawer Bodies: High-pressure decorative laminate, NEMA LD 3, Grade VGS.
 - a. Edges of Plastic-Laminate Shelves: PVC edge banding, 0.12 inch thick, matching laminate in color, pattern, and finish.
 - b. Edges of Thermoset Decorative Panel Shelves: PVC or polyester edge banding.

Tetra Tech 200-15704-17001

- c. For semiexposed backs of panels with exposed plastic-laminate surfaces, provide surface of high-pressure decorative laminate, NEMA LD 3, Grade VGS.
- 2. Drawer Sides and Backs: Thermoset decorative panels with PVC or polyester edge banding.
- 3. Drawer Bottoms: Thermoset decorative panels.
- I. Dust Panels: 1/4-inch plywood or tempered hardboard above compartments and drawers unless located directly under tops.
- J. Concealed Backs of Panels with Exposed Plastic-Laminate Surfaces: High-pressure decrative laminate, NEMA LD 3, Grade BKL.
- K. Drawer Construction: Fabricate with exposed fronts fastened to subfrom with mounting screws from interior of body.
 - 1. Join subfronts, backs, and sides with glued rabbeted joints supplemented by mechanical fasteners.
- L. Colors, Patterns, and Finishes: Provide materials and products that result in colors and textures of exposed laminate surfaces complying with the following requirements:
 - 1. As indicated on Finish Schedule.

2.2 WOOD MATERIALS

- A. Wood Products: Provide materials that comply with requirements of referenced quality standard for each type of woodwork and quality grade specified unless otherwise indicated.
 - 1. Wood Moisture Content: 5.0 10 percent.
- B. Composite Wood and rigrif ber Products: Provide materials that comply with requirements of referenced quality standard for each type of woodwork and quality grade specified unless otherwise indicate

2.3 CABINET HARDWARE AND ACCESSORIES

- General: Provide cabinet hardware and accessory materials associated with architectural cabinets except for items specified in Section 08 71 11 "Door Hardware (Descriptive Specification)."
 - Frameless Concealed Hinges (European Type): BHMA A156.9, B01602, 135 degrees of opening.
- C. Wire Pulls: Back mounted, solid metal, 4 inches long, 5/16 inch in diameter.
- D. Catches: Magnetic catches, BHMA A156.9, B03141.
- E. Adjustable Shelf Standards and Supports: BHMA A156.9, B04071; with shelf rests, B04081.

- F. Shelf Rests: BHMA A156.9, B04013; metal metal, two-pin type with shelf hold-down clip.
- G. Drawer Slides: BHMA A156.9.
 - 1. Grade 1 and Grade 2: Side mounted; full-extension type; zinc-plated steel with polymer rollers.
 - 2. Grade 1HD-100 and Grade 1HD-200: Side mounted; full-extension type; zinc-plated steel ball-bearing slides.
 - 3. For drawers not more than 3 inches high and not more than 24 inches wide, Grade 1.
 - 4. For drawers more than 3 inches high but not more than 6 inches high and not more than 24 inches wide, provide Grade 1.
 - 5. For drawers more than 6 inches high or more than 24 inches wide, provide Grade 1HD-100.
 - 6. For computer keyboard shelves, provide Grade 1.
 - 7. For trash bins not more than 20 inches high and 16 inches wide, provide Grade 1HD-100.
- H. Aluminum Slides for Sliding Glass Doors: BHMA A156.9, 80706
- I. Door Locks: BHMA A156.11, E07121.
- J. Drawer Locks: BHMA A156.11, E07041.
- K. Door and Drawer Silencers: BHMA A156.16, 10.01
- L. Exposed Hardware Finishes: For exposed hardware, provide finish that complies with BHMA A156.18 for BHMA finish runn er adicated.
- M. For concealed hardware, provide hean facturer's standard finish that complies with product class requirements in BHMA A1569.

2.4 MISCELLANEOUS MATERIALS

- A. Furring, Blocking Chims, and Hanging Strips: Softwood or hardwood lumber, kiln dried to less than 15 parent hoisture content.
- B. Anchors: Select material, type, size, and finish required for each substrate for secure anchorage. Provide metal expansion sleeves or expansion bolts for post-installed anchors. Use nonferrousmetal or hot-dip galvanized anchors and inserts at inside face of exterior walls and at floors.

hesive for Bonding Plastic Laminate: Unpigmented contact cement.

Adhesive for Bonding Edges: Hot-melt adhesive or adhesive specified above for faces.

FABRICATION

A. Sand fire-retardant-treated wood lightly to remove raised grain on exposed surfaces before fabrication.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- B. Fabricate cabinets to dimensions, profiles, and details indicated.
- C. Complete fabrication, including assembly and hardware application, to maximum extent possible before shipment to Project site. Disassemble components only as necessary for shipment and installation. Where necessary for fitting at site, provide ample allowance for scribing, trimming, and fitting.
 - 1. Notify Architect seven days in advance of the dates and times woodwork fabrication will be complete.
 - 2. Trial fit assemblies at fabrication shop that cannot be shipped completely assembled. Install dowels, screws, bolted connectors, and other fastening devices that can be removed after trial fitting. Verify that various parts fit as intended and check measurements of assemblies against field measurements before disassembling for shipment.
- D. Shop-cut openings to maximum extent possible to receive hardware, and liances, electrical work, and similar items. Locate openings accurately and use templates or roughing-in diagrams to produce accurately sized and shaped openings. Sand edges of cut uts to remove splinters and burrs.
- E. Install glass to comply with applicable requirements in Section 08 80 00 "Glazing" and in GANA's "Glazing Manual." For glass in wood frames, secure glass with removable stops.

PART 3 - EXECUTION

3.1 PREPARATION

- A. Before installation, condition cabinets to average prevailing humidity conditions in installation areas.
- B. Before installing cabines, examine shop-fabricated work for completion and complete work as required.

3.2 INSTALLATION

A. Grade install cabinets to comply with same grade as item to be installed.

Assemble cabinets and complete fabrication at Project site to the extent that it was not empleted in the shop.

Install cabinets level, plumb, true, and straight. Shim as required with concealed shims. Install level and plumb to a tolerance of 1/8 inch in 96 inches.

D. Scribe and cut cabinets to fit adjoining work, refinish cut surfaces, and repair damaged finish at cuts.

- E. Anchor cabinets to anchors or blocking built in or directly attached to substrates. Secure with countersunk, concealed fasteners and blind nailing. Use fine finishing nails or finishing screws for exposed fastening, countersunk and filled flush with woodwork.
 - 1. Use filler matching finish of items being installed.
- F. Cabinets: Install without distortion so doors and drawers fit openings properly and are accurately aligned. Adjust hardware to center doors and drawers in openings and to provide unencumbered operation. Complete installation of hardware and accessory items as indicated.
 - 1. Install cabinets with no more than 1/8 inch in 96-inch sag, bow, or other variation from a straight line.
 - 2. Fasten wall cabinets through back, near top and bottom, and at ends not more than 16 inches o.c. with No. 10 wafer-head sheet metal screws through netal backing or metal framing behind wall finish.

3.3 ADJUSTING AND CLEANING

A. Repair damaged and defective cabinets, where possible, to eliminate functional and visual defects; where not possible to repair, replace woodwork. A fjust joinery for uniform appearance.

ID OI SECTION

- B. Clean, lubricate, and adjust hardware.
- C. Clean cabinets on exposed and semiexposed and

SPEC SECTION 07 21 00

THERMAL INSULATION

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Glass-fiber blanket insulation.
 - 2. Spray polyurethane foam insulation.
 - 3. Vapor retarders.
- 1.3 SUBMITTALS, GENERAL
 - A. General: Submit all submittals required by the Section concurrently.

1.4 ACTION SUBMITTALS

- A. Product Data: For each ope of product indicated.
- 1.5 QUALITY ASSULANCE
 - A. Surface-Furning Characteristics: As determined by testing identical products according to ASTM E84 by a qualified testing agency. Identify products with appropriate markings of applicable using agency.

DELIVERY, STORAGE, AND HANDLING

Protect insulation materials from physical damage and from deterioration due to moisture, soiling, and other sources. Store inside and in a dry location. Comply with manufacturer's written instructions for handling, storing, and protecting during installation.

ADDITION AND RENOVATIONS

PART 2 - PRODUCTS

2.1 GLASS-FIBER BLANKET INSULATION

- A. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to, the following
 - 1. CertainTeed Corporation.
 - 2. Guardian Building Products, Inc.
 - 3. Johns Manville.
 - 4. Knauf Insulation.
 - 5. Owens Corning.
- B. Recycled Content: Postconsumer recycled content plus one-half of perconsumer recycled content not less than 20 percent.
- C. Unfaced, Glass-Fiber Blanket Insulation: ASTM C 665, Type I, with maximum flame-spread and smoke-developed indexes of 25 and 50, respectively, per ASTM E 84; passing ASTM E 136 for combustion characteristics.

2.2 LOOSE-FILL INSULATION

- A. Glass-Fiber Loose-Fill Insulation: ASTM C 761 Type I for pneumatic application or Type II for poured application; with maximum flame-spread and smoke-developed indexes of 5, per ASTM E 84.
 - 1. Recycled Content: Post occur encecycled content plus one-half of preconsumer recycled content not less than 2 percent.

2.3 SPRAY POLYURETRANEFOAM INSULATION

- A. Open-Cell Polyure hane Foam Insulation: Spray-applied polyure hane foam using water as a blowing agent, with maximum flame-spread and smoke-developed indexes of 75 and 450, respectively, per ASTM E 84.
 - . Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to, Icynene Inc.
 - Minimum density of 0.4 lb/cu. ft., thermal resistivity of 3.4 deg F x h x sq. ft./Btu x in. at 75 deg F.

VAPOR RETARDERS

A. Fire-Retardant, Reinforced-Polyethylene Vapor Retarders: Two outer layers of polyethylene film laminated to an inner reinforcing layer consisting of either nonwoven grid of nylon cord or polyester scrim and weighing not less than 22 lb/1000 sq. ft., with maximum permeance rating

THERMAL INSULATION 07 21 00 - 2

Tetra Tech 200-15704-17001

of 0.1317 perm and with flame-spread and smoke-developed indexes of not more than 5 and 60, respectively, per ASTM E 84.

- 1. Products: Subject to compliance with requirements, available products that may be incorporated into the Work include, but are not limited to, the following:
 - a. Raven Industries Inc.; DURA-SKRIM 2FR.
 - b. Reef Industries, Inc.; Griffolyn T-55 FR.
- B. Vapor-Retarder Tape: Pressure-sensitive tape of type recommended by manufacturer for sealing joints and penetrations in vapor retarder.
- C. Vapor-Retarder Fasteners: Pancake-head, self-tapping steel drill screws; with fender washers.
- D. Single-Component Nonsag Urethane Sealant: ASTM C 920, Type L Grade NS, Class 25, Use NT related to exposure, and Use O related to vapor-barrier-related jubstrates."

PART 3 - EXECUTION

- 3.1 PREPARATION
 - A. Clean substrates of substances that are harmful to inculation or that interfere with insulation attachment.
 - B. Beginning installation constitutes Contractor's acceptance of substrates and conditions.

3.2 INSTALLATION, GENERAL

- A. Comply with insulation manufacturer's written instructions applicable to products and applications indicated.
- B. Install insulation that is undamaged, dry, and unsolled and that has not been left exposed to ice, rain, or snow as my time.
- C. Extend insulation to envelop entire area to be insulated. Cut and fit tightly around obstructions and fill roots with insulation. Remove projections that interfere with placement.
 - Provide sizes to fit applications indicated and selected from manufacturer's standard bicknesses, widths, and lengths. Apply single layer of insulation units to produce thickness indicated unless multiple layers are otherwise shown or required to make up total thickness.

Apply insulation units to substrates by method indicated, complying with manufacturer's written instructions. If no specific method is indicated, bond units to substrate with adhesive or use mechanical anchorage to provide permanent placement and support of units.

F. Glass-Fiber Blanket Insulation: Install in cavities formed by framing members according to the following requirements:

- Use insulation widths and lengths that fill the cavities formed by framing members. If 1. more than one length is required to fill the cavities, provide lengths that will produce a snug fit between ends.
- 2. Place insulation in cavities formed by framing members to produce a friction fit between edges of insulation and adjoining framing members.
- 3. Maintain 3-inch clearance of insulation around recessed lighting fixtures not rated for protected from contact with insulation.
- For metal-framed wall cavities where cavity heights exceed 96 inches, support paraced 4. blankets mechanically.
- Apply according to ASTM C 1015 and manufactur G. Loose-Fill Insulation: instructions. Level horizontal applications to uniform thickness as indicated, lightly settle to uniform density, but do not compact excessively.
- anufacturer's written H. Spray-Applied Insulation: Apply spray-applied insulation according to instructions. Do not apply insulation until installation of pipes ducs, conduits, wiring, and electrical outlets is completed and windows, electrical boxes, and ther items not indicated to receive insulation are masked.
- Miscellaneous Voids: Install insulation in miscella eou voids and cavity spaces where I. required to prevent gaps in insulation using the following aterials:
 - Loose-Fill Insulation: Compact to approximately 40 percent of normal maximum 1. volume equaling a density of approximately 2.5 lb/cu. ft. Spray Polyurethane Insulation: Apply according to manufacturer's written instructions.
 - 2.

INSTALLATION OF VAPOR RETARD 3.3

- Place vapor retarders on side of construction indicated on Drawings. Extend vapor A. retarders to extremities of areas to protect from vapor transmission. Secure vapor retarders in place with thesives or other anchorage system as indicated. Extend vapor retarders to cover mice aneous voids in insulated substrates, including those filled with loose-fiber insulation.
- B. Seal verical points in vapor retarders over framing by lapping no fewer than two studs.
 - Perfore installing vapor retarders, apply urethane sealant to flanges of metal framing including runner tracks, metal studs, and framing around door and window openings. Seal overlapping joints in vapor retarders with vapor-retarder tape according to vapor-retarder manufacturer's written instructions. Seal butt joints with vapor-retarder tape. Locate all joints over framing members or other solid substrates.
 - Firmly attach vapor retarders to metal framing and solid substrates with vapor-2. retarder fasteners as recommended by vapor-retarder manufacturer.
- C. Seal joints caused by pipes, conduits, electrical boxes, and similar items penetrating vapor retarders with vapor-retarder tape to create an airtight seal between penetrating objects and vapor retarders.

THERMAL INSULATION 07 21 00 - 4

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

D. Repair tears or punctures in vapor retarders immediately before concealment by other work. Cover with vapor-retarder tape or another layer of vapor retarders."

3.4 **PROTECTION**

A. Protect installed insulation and vapor retarders from damage due to harmful weather exposures physical abuse, and other causes. Provide temporary coverings or enclosures where insulation is subject to abuse and cannot be concealed and protected by permanent construction immediately after installation.

END OF SECTION

Republica



SPEC SECTION 07 31 13

ASPHALT SHINGLES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section Includes:
 - 1. Asphalt shingles.
 - 2. Underlayment.
 - 3. Ridge vents.

1.3 DEFINITION

A. Roofing Terminology: See ASTM D 1012 and glossary of NRCA's "The NRCA Roofing and Waterproofing Manual" for definitions on terms related to roofing work in this Section.

1.4 SUBMITTALS, GENERA

A. General: Submit all action submittals (except Samples for Verification) and informational submittals required by the section concurrently.

1.5 ACTION SUDVINTALS

- A. Product Data. For each type of product indicated.
- B. Sample for Initial Selection: For each type of asphalt shingle indicated.

Samples for Verification: For the following products, of sizes indicated, to verify color selected:

- 1. Asphalt Shingle: Full size.
- D. Warranties: Sample of special warranties.

1.6 INFORMATIONAL SUBMITTALS

A. Qualification Data: For qualified Installer.

1.7 CLOSEOUT SUBMITTALS

- A. Maintenance Data: For asphalt shingles to include in maintenance manuals.
- B. Warranties: Executed special warranties.

1.8 MAINTENANCE MATERIAL SUBMITTALS

- A. Furnish extra materials that match products installed and that are packaged with protective covering for storage and identified with labels describing contents.
 - 1. Asphalt Shingles: 100 sq. ft., in unbroken bundles.

1.9 QUALITY ASSURANCE

- A. Installer Qualifications: Manufacturer's authorized representative who is trained and approved for installation of units required for this Project.
- B. Fire-Resistance Characteristics: Where incicated, provide asphalt shingles and related roofing materials identical to those of assemblies asted for fire resistance per test method below by UL or another testing and inspecting egency acceptable to authorities having jurisdiction. Identify products with appropriate markings of applicable testing agency.
 - 1. Exterior Fire-Test Exposure: Class A; ASTM E 108 or UL 790, for application and roof slopes indicated
- C. Solar Reflectance Index. Not less than 29 when calculated according to ASTM E 1980 based on testing identical products by a qualified testing agency.
- D. Preinstallation Conference: Conduct conference at Project site.

1.10 DED VERY, STORAGE, AND HANDLING



Store roofing materials in a dry, well-ventilated, weathertight location according to asphalt single manufacturer's written instructions. Store underlayment rolls on end on pallets or other raised surfaces. Do not double stack rolls.

- 1. Handle, store, and place roofing materials in a manner to avoid significant or permanent damage to roof deck or structural supporting members.
- B. Protect unused underlayment from weather, sunlight, and moisture when left overnight or when roofing work is not in progress.

ASPHALT SHINGLES 07 31 13 - 2

Tetra Tech 200-15704-17001

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

1.11 PROJECT CONDITIONS

A. Install self-adhering sheet underlayment within the range of ambient and substrate temperatures recommended by manufacturer.

1.12 WARRANTY

- A. Special Warranty: Standard form in which manufacturer agrees to repair or replace asr shingles that fail in materials or workmanship within specified warranty period.
 - 1. Failures include, but are not limited to, the following:
 - a. Manufacturing defects.
 - b. Structural failures including failure of asphalt shingles to self-seal after a reasonable time.
 - 2. Material Warranty Period: 50 years from date of Substantial Completion, prorated, with first 10 years nonprorated.
 - Wind-Speed Warranty Period: Asphalt shingles still tesist blow-off or damage caused by wind speeds up to 90 mph for 10 years from date of substantial Completion.
 Algae-Discoloration Warranty Period: Asphalt shingles will not discolor 10 years from
 - 4. Algae-Discoloration Warranty Period: Asphelt shingles will not discolor 10 years from date of Substantial Completion.
 - 5. Workmanship Warranty Period: 10 years from rate of Substantial Completion.
- B. Special Project Warranty: Roofing Installer, Warranty, or warranty form at end of this Section, signed by roofing Installer, covering the Work of this Section, in which roofing Installer agrees to repair or replace components of asphal shingle roofing that fail in materials or workmanship within specified warranty period.
 - 1. Warranty Period: File year from date of Substantial Completion.
- PART 2 PRODUCTS
- 2.1 GLASS-FIBER-REINFORCED ASPHALT SHINGLES
 - A. Lawingted-Strip Asphalt Shingles: ASTM D 3462, laminated, multi-ply overlay construction, glass siber reinforced, mineral-granule surfaced, and self-sealing.
 - Basis-of-Design Product: Subject to compliance with requirements, provide CertainTeed Corporation; Landmark Solaris or comparable product.
 - 2. Butt Edge: Straight cut.
 - 3. Strip Size: Manufacturer's standard.
 - 4. Algae Resistance: Granules treated to resist algae discoloration.
 - 5. Color and Blends: As selected by Architect from manufacturer's full range.
 - B. Ridge Shingles: Manufacturer's standard units to match asphalt shingles.

ADDITION AND RENOVATIONS

2.2 UNDERLAYMENT MATERIALS

- A. Underlayment: ASTM D 6757, shingle manufacturer's recommended underlayment designed for use on roof decks as a water-resistant layer beneath roofing shingles.
 - 1. Basis-of-Design Product: Subject to compliance with requirements, provide CertainTeer Corporation; Roofers' Select or comparable product.
- B. Self-Adhering Sheet Underlayment, Granular Surfaced: ASTM D 1970, minimum of 55 milthick sheet; glass-fiber-mat-reinforced, SBS-modified asphalt; mineral-granule surfaced; win release paper backing; cold applied. Provide primer for adjoining concrete or masoury surfaces to receive underlayment.
 - 1. Basis-of-Design Product: Subject to compliance with requirements, povide CertainTeed Corporation; WinterGuard or comparable product.

2.3 RIDGE VENTS

- A. Rigid Ridge Vent: Manufacturer's standard, rigid section high-density polypropylene or other UV-stabilized plastic ridge vent with nonwoven geotexite filter strips and external deflector baffles; for use under ridge shingles.
 - 1. Basis-of-Design Product: Subject to compliance with requirements, provide Air Vent, Inc.; a Gibraltar Industries company. Shin, le Vent 2 or comparable product by one of the following:
 - a. GAF Materials Corporation.
 - b. Obdyke, Benjamia Reolporated.
 - 2. Minimum Net Free Area, 12 square inches per lineal foot.
- 2.4 ACCESSORIES
 - A. Asphalt Roofin, Cement: ASTM D 4586, Type II, asbestos free.
 - B. Roofing Vails: ASTM F 1667; aluminum, stainless-steel, copper, or hot-dip galvanized-steel wire slingle nails, minimum 0.120-inch-diameter, barbed shank, sharp-pointed, with a minimum 3/8-inch-diameter flat head and of sufficient length to penetrate 3/4 inch into solid wood decking or extend at least 1/8 inch through OSB or plywood sheathing.
- Where nails are in contact with metal flashing, use nails made from same metal as flashing.
- METAL FLASHING AND TRIM
- A. General: Comply with requirements in Division 07 Section "Sheet Metal Flashing and Trim."

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

B. Fabricate sheet metal flashing and trim to comply with recommendations in SMACNA's "Architectural Sheet Metal Manual" that apply to design, dimensions, metal, and other characteristics of the item.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine substrates, areas, and conditions, with Installer present, for compliance with requirements for installation tolerances and other conditions affecting performance fit. Work.
 - 1. Examine roof sheathing to verify that sheathing joints are supported up traming and blocking or metal clips and that installation is within flatness tolerances.
 - 2. Verify that substrate is sound, dry, smooth, clean, sloped for graining, and completely anchored; and that provision has been made for flashing, and penetrations through asphalt shingles.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.
- C. Beginning installation constitutes Contractor's acceptance of substrates and conditions.

3.2 UNDERLAYMENT INSTALLATION

- A. General: Comply with underlayment manufacturer's written installation instructions applicable to products and applications indicated unless more stringent requirements apply.
- B. Double-Layer Underlayment. Install on roof deck parallel with eaves. Install courses lapping previous courses 19 inches in stingle fashion. Lap ends a minimum of 19 inches. Stagger end laps between succeeding courses at least 72 inches. Fasten with nails.
 - 1. Install underlayment on roof sheathing not covered by self-adhering sheet underlayment. Lap edges over self-adhering sheet underlayment not less than 3 inches in direction to shed wate
 - 2. Terminae underlayment flush against sidewalls, curbs, chimneys, and other roof prejections.
 - 3. Utstall fasteners at no more than 36 inch o.c.

Self-Adhering Sheet Underlayment: Install, wrinkle free, on roof deck. Comply with lowmperature installation restrictions of underlayment manufacturer if applicable. Install at locations indicated below, lapped in direction to shed water. Lap sides not less than 3-1/2 inches. Lap ends not less than 6 inches staggered 24 inches between courses. Roll laps with roller. Cover underlayment within seven days.

- 1. Prime concrete and masonry surfaces to receive self-adhering sheet underlayment.
- 2. Eaves: Extend from edges of eaves 36 inches beyond interior face of exterior wall.
- 3. Rakes: Extend from edges of rake 36 inches beyond interior face of exterior wall.
- 4. Valleys: Extend from lowest to highest point 18 inches on each side.
- 5. Ridges: Extend 36 inches on each side without obstructing continuous ridge vent slot.

Tetra Tech 200-15704-17001 ASPHALT SHINGLES 07 31 13 - 5

- 6. Sidewalls: Extend beyond sidewall 18 inches, and return vertically against sidewall not less than 4 inches.
- 7. Roof-Penetrating Elements: Extend beyond penetrating element 18 inches, and return vertically against penetrating element not less than 4 inches.

3.3 METAL FLASHING INSTALLATION

- A. General: Install metal flashings and other sheet metal to comply with requirement Division 07 Section "Sheet Metal Flashing and Trim."
 - 1. Install metal flashings according to recommendations in ARMA's "Residential Asphalt Roofing Manual" and asphalt shingle recommendations in NRCA's "The NRCA Roofing and Waterproofing Manual."
- B. Apron Flashings: Extend lower flange over and beyond each side of downslope asphalt shingles and up the vertical surface.
- C. Step Flashings: Install with a headlap of 2 inches and exend over the underlying asphalt shingle and up the vertical surface. Fasten to roof deck only
- D. Open-Valley Flashings: Install centered in valleyer lapping ends at least 8 inches in direction to shed water. Fasten upper end of each length to noof deck beneath overlap.
 - 1. Secure hemmed flange edges into restal or ats spaced 12 inches apart and fastened to roof deck.
 - 2. Adhere 9-inch-wide strip of self-adhering sheet to metal flanges and to self-adhering sheet underlayment.
- E. Rake Drip Edges: Install rake drip edge flashings over underlayment and fasten to roof deck.
- F. Eave Drip Edges: Istal eave drip edge flashings below underlayment and fasten to roof sheathing.
- G. Pipe Flashing. Form flashing around pipe penetrations and asphalt shingles. Fasten and seal to asphalt shingles is recommended by manufacturer.

3.4 ASSHALT SHINGLE INSTALLATION

General: Install asphalt shingles according to manufacturer's written instructions, accommendations in ARMA's "Residential Asphalt Roofing Manual," and asphalt shingle recommendations in NRCA's "The NRCA Roofing and Waterproofing Manual."

Install starter strip along lowest roof edge, consisting of an asphalt shingle strip with selfsealing strip face up at roof edge.

- 1. Extend asphalt shingles 3/4 inch over fasciae at eaves and rakes.
- 2. Install starter strip along rake edge.

- C. Install first and remaining courses of asphalt shingles stair-stepping diagonally across roof deck with manufacturer's recommended offset pattern at succeeding courses, maintaining uniform exposure.
- D. Fasten asphalt shingle strips with a minimum of five roofing nails located according to manufacturer's written instructions.
 - 1. When ambient temperature during installation is below 50 deg F, seal asphalt shingle with asphalt roofing cement spots.
- E. Open Valleys: Cut and fit asphalt shingles at open valleys, trimming upper concerled corners of shingle strips. Maintain uniform width of exposed open valley from highest to lower point.
 - 1. Set valley edge of asphalt shingles in a 3-inch-wide bed of asphalt oo ing ement.
 - 2. Do not nail asphalt shingles to metal open-valley flashings.
- F. Ridge Vents: Install continuous ridge vents over asphalt shingles becording to manufacturer's written instructions. Fasten with roofing nails of sufficient length to penetrate sheathing.
- G. Ridge Cap Shingles: Maintain same exposure of cap shingles as roofing shingle exposure. Lap cap shingles at ridges to shed water away from direction of prevailing winds. Fasten with roofing nails of sufficient length to penetrate sheating.
 - 1. Fasten ridge cap asphalt shingles to cover helge ent without obstructing airflow.



SPEC SECTION 07 62 00

SHEET METAL FLASHING AND TRIM

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Steep-slope roof sheet metal fabrications.

1.3 PERFORMANCE REQUIREMENTS

- A. General: Sheet metal flashing and trim assemblies as indicated shall withstand wind loads, structural movement, thermally induced movement, and exposure to weather without failure due to defective manufacture, fabrication instellation, or other defects in construction. Completed sheet metal flashing and trim shall not raile, leak, or loosen, and shall remain watertight.
- B. Thermal Movements: Provice shee metal flashing and trim that allows for thermal movements from ambient and surface temperature changes.
 - 1. Temperature Charge Range): 120 deg F, ambient; 180 deg F, material surfaces.
- 1.4 SUBMITTAL GENERAL
 - A. General: Submit all action submittals (except Samples for Verification) and informational submittals required by this Section concurrently.

ACTION SUBMITTALS

Product Data: For each type of product indicated. Include construction details, material descriptions, dimensions of individual components and profiles, and finishes for each manufactured product and accessory.

B. Shop Drawings: Show fabrication and installation layouts of sheet metal flashing and trim, including plans, elevations, expansion-joint locations, and keyed details. Distinguish between shop- and field-assembled work. Include the following:

- 1. Identification of material, thickness, weight, and finish for each item and location in Project.
- 2. Details for forming sheet metal flashing and trim, including profiles, shapes, seams, and dimensions.
- 3. Details for joining, supporting, and securing sheet metal flashing and trim, including layout of fasteners, cleats, clips, and other attachments. Include pattern of seams.
- 4. Details of termination points and assemblies.
- 5. Details of expansion joints and expansion-joint covers.
- 6. Details of edge conditions.
- 7. Details of special conditions.
- 8. Details of connections to adjoining work.
- C. Samples for Initial Selection: For each type of sheet metal flashing, trin, and accessory indicated with factory-applied color finishes involving color selection.
- D. Samples for Verification: For each type of exposed finish required prepared on Samples of size indicated below:
 - Sheet Metal Flashing: 12 inches long by actual width of wit, including finished seam and in required profile. Include fasteners, cleats, clips closures, and other attachments.
 Trim, Metal Closures, Expansion Joints, Joint Intersections, and Miscellaneous
 - 2. Trim, Metal Closures, Expansion Joints, John Intersections, and Miscellaneous Fabrications: 12 inches long and in required profile. Include fasteners and other exposed accessories.
- E. Warranty: Sample of special warranty.

1.6 INFORMATIONAL SUBMITTAL

- A. Qualification Data: For qualified fabricator.
- 1.7 CLOSEOUT SUBMINIAL
 - A. Maintenance pata. For sheet metal flashing, trim, and accessories to include in maintenance manuals.
 - B. Warranty Executed special warranty.

8 QUALITY ASSURANCE



Fabricator Qualifications: Shop that employs skilled workers who custom fabricate sheet metal flashing and trim similar to that required for this Project and whose products have a record of successful in-service performance.

- B. Sheet Metal Flashing and Trim Standard: Comply with SMACNA's "Architectural Sheet Metal Manual" unless more stringent requirements are specified or shown on Drawings.
- C. Copper Sheet Metal Standard: Comply with CDA's "Copper in Architecture Handbook." Conform to dimensions and profiles shown unless more stringent requirements are indicated.

- D. Preinstallation Conference: Conduct conference at Project site.
 - Meet with Owner, Architect, Installer, and installers whose work interfaces with or 1. affects sheet metal flashing and trim including installers of roofing materials, roof accessories, and roof-mounted equipment.
 - Review methods and procedures related to sheet metal flashing and trim. 2.
 - Examine substrate conditions for compliance with requirements, including flatness 3. attachment to structural members.
 - Review special roof details, roof drainage, roof penetrations, and condition 4. construction that will affect sheet metal flashing.

1.9 DELIVERY, STORAGE, AND HANDLING

- Do not store sheet metal flashing and trim materials in contact with other materials that might A. cause staining, denting, or other surface damage. Store sheet meta flashin, and trim materials away from uncured concrete and masonry.
- Protect strippable protective covering on sheet metal flash hd trim from exposure to B. ing a the period of sheet metal flashing sunlight and high humidity, except to the extent necess and trim installation.

1.10 WARRANTY

- Special Warranty on Finishes: Manufacturer's standard form in which manufacturer agrees to repair finish or replace sheet metal meshing and trim that shows evidence of deterioration of A. factory-applied finishes within specified varianty period.
 - Exposed Panel Finish. Deterpration includes, but is not limited to, the following: 1.
 - Color facing monothan 5 Hunter units when tested according to ASTM D 2244. a.
 - king a excess of a No. 8 rating when tested according to ASTM D 4214. ing, checking, peeling, or failure of paint to adhere to bare metal. b. Chalking
 - c.
 - 2. ranty Period: 20 years from date of Substantial Completion. Fini

PART 2 - PROD

HEET METALS

General: Protect mechanical and other finishes on exposed surfaces from damage by applying a strippable, temporary protective film before shipping.

- Copper Sheet: ASTM B 370, cold-rolled copper sheet, H00 or H01 temper.
 - 1. Non-Patinated Exposed Finish: Mill.
- C. Aluminum Sheet: ASTM B 209, alloy as standard with manufacturer for finish required, with temper as required to suit forming operations and performance required.

Tetra Tech 200-15704-17001

B.

- 1. Surface: Smooth, flat.
- 2. Exposed Coil-Coated Finishes:
 - a. Two-Coat Fluoropolymer: AAMA 620. Fluoropolymer finish containing not less than 70 percent PVDF resin by weight in color coat. Prepare, pretreat, and apply coating to exposed metal surfaces to comply with coating and resin manufacturer written instructions.
- 3. Color: As selected by Architect from manufacturer's full range.

2.2 UNDERLAYMENT MATERIALS

- A. Self-Adhering, High-Temperature Sheet: Minimum 30 to 40 mils thick, consisting of slipresisting polyethylene-film top surface laminated to layer of butyl of SES-modified asphalt adhesive, with release-paper backing; cold applied. Provide primer when recommended by underlayment manufacturer.
 - 1. Products: Subject to compliance with requirements, available products that may be incorporated into the Work include, but are not lighter to, me following:
 - a. Carlisle Coatings & Waterproofing Iren CCW WIP 300HT.
 - b. Grace Construction Products, a unit of V.R. Grace & Co.; Ultra.
 - c. Henry Company; Blueskin PE202 HT

2.3 MISCELLANEOUS MATERIALS

- A. General: Provide materials and types of fasteners, solder, welding rods, protective coatings, separators, sealants, and other miscellaneous items as required for complete sheet metal flashing and trim installation and recommended by manufacturer of primary sheet metal or manufactured item unless otherwise indicates.
- B. Fasteners: Wood screws annular threaded nails, self-tapping screws, self-locking rivets and bolts, and other suitable fasteners designed to withstand design loads and recommended by manufactures of arimary sheet metal or manufactured item.
 - 1. General: Blind fasteners or self-drilling screws, gasketed, with hex-washer head.

Exposed Fasteners: Heads matching color of sheet metal using plastic caps or factory-applied coating.

- Blind Fasteners: Stainless-steel rivets suitable for metal being fastened.
- Fasteners for Copper Sheet: Copper, hardware bronze or Series 300 stainless steel. Fasteners for Aluminum Sheet: Series 300 stainless steel.
- Solder:

3

1. For Copper: ASTM B 32, Grade Sn50, 50 percent tin and 50 percent lead.

- D. Elastomeric Sealant: ASTM C 920, elastomeric polymer sealant; low modulus; of type, grade, class, and use classifications required to seal joints in sheet metal flashing and trim and remain watertight.
- E. Epoxy Seam Sealer: Two-part, noncorrosive, aluminum seam-cementing compound, recommended by aluminum manufacturer for exterior nonmoving joints, including riveter joints.

2.4 FABRICATION, GENERAL

- A. General: Custom fabricate sheet metal flashing and trim to comply with recommendations in SMACNA's "Architectural Sheet Metal Manual" that apply to design, dimensions) geometry, metal thickness, and other characteristics of item indicated. Fabricate items at the shop to greatest extent possible.
 - 1. Fabricate sheet metal flashing and trim in thickness or we put needed to comply with performance requirements, but not less than that specified for each application and metal.
 - 2. Obtain field measurements for accurate fit before shop t brick ion.
 - 3. Form sheet metal flashing and trim without excessive on canning, buckling, and tool marks and true to line and levels indicated, with exposed edges folded back to form hems.
 - 4. Conceal fasteners and expansion provisions where possible. Exposed fasteners are not allowed on faces exposed to view.
- B. Fabrication Tolerances: Fabricate sheet neal flashing and trim that is capable of installation to a tolerance of 1/4 inch in 20 feet on supe and location lines as indicated and within 1/8-inch offset of adjoining faces and of alignment of matching profiles.
- C. Sealed Joints: Form nonexpansion but movable joints in metal to accommodate elastomeric sealant.
- D. Fabricate cleats and attachment devices of sizes as recommended by SMACNA's "Architectural Sheet Metal Manul" for application, but not less than thickness of metal being secured.
- E. Seams: Febrican nonmoving seams with flat-lock seams. Tin edges to be seamed, form seams, and sold r.
- F. Seams or Aluminum: Fabricate nonmoving seams with flat-lock seams. Form seams and seal with expoxy seam sealer. Rivet joints where necessary for strength.

J.C.

not use graphite pencils to mark metal surfaces.

STEEP-SLOPE ROOF SHEET METAL FABRICATIONS

- A. Apron, Step, Cricket, and Backer Flashing: Fabricate from the following materials:
 - 1. Copper: 16 oz./sq. ft.
- B. Valley Flashing: Fabricate from the following materials:

QC

Str

- 1. Copper: 16 oz./sq. ft.
- C. Drip Edges: Fabricate from the following materials:
 - 1. Aluminum: 0.032 inch thick.
- D. Eave, Rake Flashing: Fabricate from the following materials:
 - 1. Aluminum: 0.032 inch thick.
- E. Counterflashing: Fabricate from the following materials:
 - 1. Copper: 16 oz./sq. ft.
- F. Roof-Penetration Flashing: Fabricate from the following materials:
 - 1. Copper: 16 oz./sq. ft.

PART 3 - EXECUTION

- 3.1 EXAMINATION
 - A. Examine substrates, areas, and conditions, with installer present, to verify actual locations, dimensions and other conditions affecting performance of the Work.
 - 1. Verify compliance with requirements for installation tolerances of substrates.
 - 2. Verify that substrate is could day, smooth, clean, sloped for drainage, and securely anchored.
 - B. Proceed with installation nly ofter unsatisfactory conditions have been corrected.
 - C. Beginning installation constitutes Contractor's acceptance of substrates and conditions.
- 3.2 UNDERLAYNENT INSTALLATION
 - A. General: Install underlayment as indicated on Drawings.



Self-Achering Sheet Underlayment: Install self-adhering sheet underlayment, wrinkle free. Apply primer if required by underlayment manufacturer. Comply with temperature restrictions of underlayment manufacturer for installation; use primer rather than nails for installing underlayment at low temperatures. Apply in shingle fashion to shed water, with end laps of not less than 6 inches staggered 24 inches between courses. Overlap side edges not less than 3-1/2 inches. Roll laps with roller. Cover underlayment within 14 days.

- 3.3 INSTALLATION, GENERAL
 - A. General: Anchor sheet metal flashing and trim and other components of the Work securely in place, with provisions for thermal and structural movement. Use fasteners, solder, welding

en

rods, protective coatings, separators, sealants, and other miscellaneous items as required to complete sheet metal flashing and trim system.

- 1. Install sheet metal flashing and trim true to line and levels indicated. Provide uniform, neat seams with minimum exposure of solder, welds, and sealant.
- 2. Install sheet metal flashing and trim to fit substrates and to result in watertight performance. Verify shapes and dimensions of surfaces to be covered before fabricating sheet metal.
- 3. Space cleats not more than 12 inches apart. Anchor each cleat with two fasteners tabs over fasteners.
- 4. Install exposed sheet metal flashing and trim without excessive oil canning, backling, and tool marks.
- 5. Torch cutting of sheet metal flashing and trim is not permitted.
- 6. Do not use graphite pencils to mark metal surfaces.
- B. Metal Protection: Where dissimilar metals will contact each oner or corrosive substrates, protect against galvanic action by painting contact surfaces with bit annous coating or by other permanent separation as recommended by SMACNA.
 - 1. Underlayment: Where installing metal flashing cirectly on cementitious or wood substrates, install a course of self-adhering underlayment.
- C. Expansion Provisions: Provide for thermal expansion of exposed flashing and trim. Space movement joints at a maximum of 10 feet with ne joints allowed within 24 inches of corner or intersection. Where lapped expansion provisions cannot be used or would not be sufficiently watertight, form expansion joints of intermeshing hooked flanges, not less than 1 inch deep, filled with sealant concealed within joints.
- D. Seal joints as shown and as required by watertight construction.
 - 1. Prepare joints and apply scalants to comply with requirements in Division 07 Section "Joint Sealants
- E. Soldered Joints: clean surfaces to be soldered, removing oils and foreign matter. Pre-tin edges of sheets to be soldered to a width of 1-1/2 inches, except reduce pre-tinning where pre-tinned surface work show in completed Work.
 - Donot use torches for soldering. Heat surfaces to receive solder and flow solder into joint. Fill joint completely. Completely remove flux and spatter from exposed surfaces.
 Copper Soldering: Tin edges of uncoated copper sheets using solder for copper.



ROOF FLASHING INSTALLATION

General: Install sheet metal flashing and trim to comply with performance requirements, sheet metal manufacturer's written installation instructions, and SMACNA's "Architectural Sheet Metal Manual." Provide concealed fasteners where possible, set units true to line, and level as indicated. Install work with laps, joints, and seams that will be permanently watertight and weather resistant.

- B. Counterflashing: Coordinate installation of counterflashing with installation of base flashing. Insert counterflashing in reglets or receivers and fit tightly to base flashing. Extend counterflashing 4 inches over base flashing. Lap counterflashing joints a minimum of 4 inches and bed with sealant. Secure in a waterproof manner by means of interlocking folded seam or blind rivets and sealant.
- C. Roof-Penetration Flashing: Coordinate installation of roof-penetration flashing with installation of roofing and other items penetrating roof. Seal with elastomeric sealant and clamp flashing to pipes that penetrate roof.

3.5 ERECTION TOLERANCES

- A. Installation Tolerances: Shim and align sheet metal flashing and trim winin installed tolerance of 1/4 inch in 20 feet on slope and location lines as indicated and wibin 1/8-inch offset of adjoining faces and of alignment of matching profiles.
- B. Installation Tolerances: Shim and align sheet metal flashing and him within installed tolerances specified in MCA's "Guide Specification for Residential Meta Rooting."

3.6 CLEANING AND PROTECTION

- A. Clean exposed metal surfaces of substances that interfere with uniform oxidation and weathering.
- B. Clean and neutralize flux materials. Chan off excess solder.
- C. Clean off excess sealants.
- D. Remove temporary protective coverings and strippable films as sheet metal flashing and trim are installed unless other vise indicated in manufacturer's written installation instructions. On completion of installation, remove unused materials and clean finished surfaces. Maintain in a clean condition during construction.
- E. Replace sheet in stal flashing and trim that have been damaged or that have deteriorated beyond successful repar by finish touchup or similar minor repair procedures.

END OF SECTION

SPEC SECTION 07 71 00

ROOFING SPECIALTIES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplem Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section Includes:
 - 1. Underlayment Materials.
 - 2. Roof-edge drainage systems.

1.3 PERFORMANCE REQUIREMENTS

- A. General Performance: Roof specialties shall webstand exposure to weather and resist thermally induced movement without failure, rattling, leaking, or fastener disengagement due to defective manufacture, fabrication, installation or other defects in construction.
 - 1.
- B. Thermal Movements: Allow for nermal movements from ambient and surface temperature changes to prevent bucking, opening of joints, hole elongation, overstressing of components, failure of joint sealants failure of connections, and other detrimental effects. Provide clips that resist rotation and avoid blear stress as a result of thermal movements. Base calculations on surface temperatures of materials due to both solar heat gain and nighttime-sky heat loss.
 - 1. Temperature Change (Range): 120 deg F, ambient; 180 deg F, material surfaces.
- 1.4 SUBNITTALS, GENERAL
 - General: Submit all action submittals (except Samples for Verification) required by this Section concurrently.

ACTION SUBMITTALS

A. Product Data: For each type of product indicated. Include construction details, material descriptions, dimensions of individual components and profiles, and finishes.

Tetra Tech 200-15704-17001

- B. Shop Drawings: For roof specialties. Include plans, elevations, expansion-joint locations, keyed details, and attachments to other work. Distinguish between plant- and field-assembled work. Include the following:
 - 1. Details for expansion and contraction; locations of expansion joints, including direction of expansion and contraction.
 - 2. Pattern of seams and layout of fasteners, cleats, clips, and other attachments.
 - 3. Details of termination points and assemblies.
 - 4. Details of special conditions.
- C. Samples for Initial Selection: For sealant and each type of roof specialty indicated wan factoryapplied color finishes.
- D. Samples for Verification: For roof-edge drainage systems made from 2-1 ch lengths of fullsize components including fasteners, cover joints, accessories, and attacheents.
- E. Warranty: Sample of special warranty.

1.6 CLOSEOUT SUBMITTALS

- A. Maintenance Data: For roofing specialties to include in maintenance manuals.
- B. Warranty: Executed special warranty.

1.7 QUALITY ASSURANCE

- A. Preinstallation Conference: Conduct sorrerence at Project site.
 - Meet with Owner, Architex, Installer, and installers whose work interfaces with or affects roof specialies including installers of roofing materials and accessories.
 Examine substrate conditions for compliance with requirements, including flatness and
 - 2. Examine substrate conditions for compliance with requirements, including flatness and attachment to structural members.
 - 3. Review special roof details, roof drainage, and condition of other construction that will affect roo specialties.

1.8 DECIVERY, STORAGE, AND HANDLING

Do not store roof specialties in contact with other materials that might cause staining, denting, or other surface damage. Store roof specialties away from uncured concrete and masonry.

Protect strippable protective covering on roof specialties from exposure to sunlight and high humidity, except to extent necessary for the period of roof specialties installation.

WARRANTY

1.9

A. Special Warranty: Warranty, as part of special warranty in Division 07 Section "EPDM Roofing", in which Installer agrees to repair or replace components that fail in materials or workmanship within specified warranty period.

ROOFING SPECIALTIES 07 71 00 - 2

Tetra Tech 200-15704-17001

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- B. Special Warranty on Painted Finishes: Manufacturer's standard form in which manufacturer agrees to repair finish or replace roof specialties that show evidence of deterioration of factory-applied finishes within specified warranty period.
 - 1. Fluoropolymer Finish: Deterioration includes, but is not limited to, the following:
 - a. Color fading more than 5 Hunter units when tested according to ASTM D 2244.
 - b. Chalking in excess of a No. 8 rating when tested according to ASTM D 421
 - c. Cracking, checking, peeling, or failure of paint to adhere to bare metal.
 - 2. Finish Warranty Period: 20 years from date of Substantial Completion.

PART 2 - PRODUCTS

2.1 EXPOSED METALS

- A. Aluminum Sheet: ASTM B 209, alloy as standard with manufacturer for finish required, with temper to suit forming operations and performance required.
 - 1. Surface: Smooth, flat finish.
 - 2. Exposed Coil-Coated Finishes: Prepare pretreat, and apply coating to exposed metal surfaces to comply with coating and resin nanufacturers' written instructions.
 - a. Two-Coat Fluoropolymer, 1AMA 620. System consisting of primer and fluoropolymer color toperat iontaining not less than 70 percent PVDF resin by weight.

2.2 CONCEALED METALS

- A. Stainless-Steel Sheet: WTN A 240/A 240M or ASTM A 666, Type 304.
- B. Zinc-Coated (calyinized) Steel Sheet: ASTM A 653/A 653M, G90 coating designation.

2.3 UNDERLAYMENT MATERIALS

A. Self-x thering, High-Temperature Sheet: Minimum 30 to 40 mils thick, consisting of slipresisting polyethylene-film top surface laminated to layer of butyl or SBS-modified asphalt dhesive, with release-paper backing; cold applied. Provide primer when recommended by underlayment manufacturer.

- Products: Subject to compliance with requirements, available products that may be incorporated into the Work include, but are not limited to, the following:
 - a. Carlisle Coatings & Waterproofing; CCW WIP 300HT.
 - b. Grace Construction Products, a unit of W. R. Grace & Co.; Ultra.
 - c. Henry Company; Blueskin PE200 HT.

Tetra Tech 200-15704-17001

ADDITION AND RENOVATIONS

2.4 MISCELLANEOUS MATERIALS

- A. General: Provide materials and types of fasteners, protective coatings, sealants, and other miscellaneous items required by manufacturer for a complete installation.
- B. Fasteners: Manufacturer's recommended fasteners, of Series 300 stainless steel, suitable for application and designed to meet performance requirements.
 - 1. Exposed Fasteners: Not permitted.
- C. Elastomeric Sealant: ASTM C 920, elastomeric polymer sealant of type, grade, cros, and use classifications required by roofing-specialty manufacturer for each application
 - 1. Color: As selected by Architect from manufacturer's full range to hatch color of roof specialties, unless noted otherwise.

2.5 ROOF-EDGE DRAINAGE SYSTEMS

- A. Basis-of-Design Product: Subject to compliance with requirements, provide the following:
 - 1. Gutters: Firestone Building Products; Firestone adustrial Gutter System FS-1 or comparable product.
 - 2. Downspouts: Firestone Building Products: Firestone Downspout (Closed Face) or comparable product.
 - 3. Downspout Boots: Jay R. Smith My, Co.; 1785 or 1786 Series Downspout Boots or comparable product.
- B. Gutters: Manufactured in uniform section lengths not exceeding 12 feet, with matching corner units, ends, outlet tubes, and other accessories. Elevate back edge at least 1 inch above front edge. Furnish flat-stock gutte, straps, gutter brackets, expansion joints, and expansion-joint covers fabricated from same metal as gutters.
 - 1. Fabricate from the following exposed metal:

a. For ned Aluminum: 0.063 inch thick.

- 2. Comers: Factory mitered and continuously welded.
- 3. Sutter Supports: Straps with finish matching the gutters.

Downspouts: Plain rectangular complete with mitered elbows, manufactured from the allowing exposed metal. Furnish with metal hangers, from same material as downspouts, and anchors.

- 1. Formed Aluminum: 0.050 inch thick.
- D. Aluminum Finish: Two-coat fluoropolymer.
 - 1. Color: As selected by Architect from manufacturer's full range.
CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

2.6 GENERAL FINISH REQUIREMENTS

- A. Comply with NAAMM's "Metal Finishes Manual for Architectural and Metal Products" for recommendations for applying and designating finishes.
- B. Protect mechanical and painted finishes on exposed surfaces from damage by applying strippable, temporary protective covering before shipping.
- C. Appearance of Finished Work: Noticeable variations in same piece are not accorr Variations in appearance of adjoining components are acceptable if they are within the rang approved Samples and are assembled or installed to minimize contrast.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine substrates, areas, and conditions, with Installer present to verify actual locations, dimensions, and other conditions affecting performance of the Work.
- B. Examine walls, roof edges, and parapets for suitable conditions for roof specialties.
- C. Verify that substrate is sound, dry, smooth, chan, sloped for drainage, and securely anchored.
- D. Proceed with installation only after unsatisfy conditions have been corrected.
- E. Beginning installation constitutes contractor's acceptance of substrates and conditions.

3.2 UNDERLAYMENT INSTALLATION

A. Self-Adhering Sheet Underlayment: Install wrinkle free. Apply primer if required by underlayment manufacturer. Comply with temperature restrictions of underlayment manufacturer or installation; use primer rather than nails for installing underlayment at low temperatures. Apply in shingle fashion to shed water. Overlap edges not less than 3-1/2 inches. Roll laps with ollor. Cover underlayment within 14 days.

INSTALLATION, GENERAL

General: Install roof specialties according to manufacturer's written instructions. Anchor roof specialties securely in place, with provisions for thermal and structural movement. Use fasteners, solder, protective coatings, separators, sealants, and other miscellaneous items as required to complete roof-specialty systems.

- 1. Install roof specialties level, plumb, true to line and elevation; with limited oil-canning and without warping, jogs in alignment, buckling, or tool marks.
- 2. Provide uniform, neat seams with minimum exposure of solder and sealant.
- 3. Install roof specialties to fit substrates and to result in watertight performance. Verify shapes and dimensions of surfaces to be covered before manufacture.
- 4. Torch cutting of roof specialties is not permitted.

Tetra Tech 200-15704-17001

3.3

- 5. Do not use graphite pencils to mark metal surfaces.
- B. Metal Protection: Protect metals against galvanic action by separating dissimilar metals from contact with each other or with corrosive substrates by painting contact surfaces with bituminous coating or by other permanent separation as recommended by manufacturer.
 - 1. Underlayment: Where installing metal directly on cementitious or wood substrates install a course of self-adhering, high-temperature sheet underlayment.
- C. Expansion Provisions: Allow for thermal expansion of exposed roof specialties.
 - 1. Space movement joints at a maximum of 12 feet with no joints within 14 there of corners or intersections unless otherwise shown on Drawings.
 - 2. When ambient temperature at time of installation is between 40 and 70 log F, set joint members for 50 percent movement each way. Adjust setting proportionately for installation at higher ambient temperatures.
- D. Fastener Sizes: Use fasteners of sizes that will penetrate substrate not less than recommended by fastener manufacturer to achieve maximum pull-out resistance.
- E. Seal joints with elastomeric sealant as required by roofing specialty manufacturer.
- F. Seal joints as required for watertight construction. Drate sealant to be completely concealed in joint. Do not install sealants at temperatures below 40 deg F.

3.4 ROOF-EDGE DRAINAGE-SYSTEM INSTALLATION

- A. General: Install components to produce a complete roof-edge drainage system according to manufacturer's written instructions. Coordinate installation of roof perimeter flashing with installation of roof-edge drainage system.
- B. Gutters: Join and searcutter lengths. Allow for thermal expansion. Attach gutters to firmly anchored gutter exports spaced not more than 18 inches apart. Attach ends with rivets and seal with sealant termale watertight. Slope to downspouts.
 - 1. Initial gatter with expansion joints at locations indicated but not exceeding 48 feet apart. Install expansion joint caps.
- C. Downpouts: Join sections with manufacturer's standard telescoping joints. Provide hangers with fasteners designed to hold downspouts securely to walls and 1 inch away from walls; heate fasteners at top and bottom and at approximately 60 inches o.c.
 - Provide downspout boots to connect downspouts to underground drainage system indicated. Secure downspout boots to walls with stainless steel flat head bolts.

3.5 CLEANING AND PROTECTION

A. Remove temporary protective coverings and strippable films as roof specialties are installed. On completion of installation, clean finished surfaces. Maintain roof specialties in a clean condition during construction.

ROOFING SPECIALTIES 07 71 00 - 6 Tetra Tech 200-15704-17001

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

B. Replace roof specialties that have been damaged or that cannot be successfully repaired by finish touchup or similar minor repair procedures.

END OF SECTION

Republic

Tetra Tech 200-15704-17001

4



SPEC SECTION 07 72 00

ROOF ACCESSORIES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section Includes:
 - 1. Roof curbs.
 - 2. Equipment supports.

1.3 PERFORMANCE REQUIREMENTS

A. General Performance: Roof accessories shan withstand exposure to weather and resist thermally induced movement without failure, tattling, leaking, or fastener disengagement due to defective manufacture, fabrication instanction, or other defects in construction.

1.4 SUBMITTALS, GENERA

A. General: Submit all action submittals required by this Section concurrently.

1.5 ACTION SUBMITALS

- A. Product Data: For each type of roof accessory indicated. Include construction details, material descriptions, amensions of individual components and profiles, and finishes.
- B. Shop Dawings: For roof accessories. Include plans, elevations, keyed details, and attachments to other work. Indicate dimensions, loadings, and special conditions.

COORDINATION

- A. Coordinate layout and installation of roof accessories with roofing membrane and base flashing and interfacing and adjoining construction to provide a leakproof, weathertight, secure, and noncorrosive installation.
- B. Coordinate dimensions with rough-in information or Shop Drawings of equipment to be supported.

Tetra Tech 200-15704-17001

ROOF ACCESSORIES 07 72 00 - 1 C. Coordinate construction operations on or adjacent to roof, included in different Sections, which depend on each other for proper installation, connection, and operation.

PART 2 - PRODUCTS

- 2.1 METAL MATERIALS
 - A. Zinc-Coated (Galvanized) Steel Sheet: ASTM A 653/A 653M, G90 coating designation

2.2 MISCELLANEOUS MATERIALS

- A. General: Provide materials and types of fasteners, protective coatings sealants, and other miscellaneous items required by manufacturer for a complete installation.
- B. Glass-Fiber Board Insulation: ASTM C 726, thickness as indicated
- C. Wood Nailers: Softwood lumber, pressure treated with caterborne preservatives for aboveground use, acceptable to authorities having pusdiction, containing no arsenic or chromium, and complying with AWPA C2; not less than 1-1/2 inches thick.
- D. Underlayment:
 - 1. Self-Adhering, High-Temperature Steet: Minimum 30 to 40 mils thick, consisting of slip-resisting polyethylene-film top surface laminated to layer of butyl or SBS-modified asphalt adhesive, with release-paper backing; cold applied. Provide primer when recommended by underlayered monufacturer.
 - a. Products: Subject to compliance with requirements, available products that may be incorporated into the Work include, but are not limited to, the following:
 - Carlist Coatings & Waterproofing; CCW WIP 300HT.
 - Grace Construction Products, a unit of W. R. Grace & Co.; Ultra.
 - Henry Company; Blueskin PE200 HT.
- E. Fastevers, Roof accessory manufacturer's recommended fasteners suitable for application and meals being fastened. Match finish of exposed fasteners with finish of material being fastened. Provide nonremovable fastener heads to exterior exposed fasteners. Furnish the following unless otherwise indicated:

Fasteners: Series 300 stainless steel.

Elastomeric Sealant: ASTM C 920, elastomeric polymer sealant as recommended by roof accessory manufacturer for installation indicated; low modulus; of type, grade, class, and use classifications required to seal joints and remain watertight.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

2.3 ROOF CURBS

- A. Roof Curbs: Internally reinforced roof-curb units capable of supporting superimposed live and dead loads, including equipment loads and other construction indicated on Drawings; with welded or mechanically fastened and sealed corner joints, and integrally formed deck-mounting flange at perimeter bottom.
 - 1. Basis-of-Design Product: Subject to compliance with requirements, provid Company (The); PC-2 or comparable product by one of the following:
 - a. Conn-Fab Sales Incorporated.
 - b. Thybar Corporation.
- B. Size: Coordinate dimensions with roughing-in information or Shop Drawings of equipment to be supported.
- C. Material: Zinc-coated (galvanized) steel sheet, 0.052 inch thick
 - 1. Finish: Mill phosphatized.
- D. Construction:
 - 1. Insulation: Factory insulated with 1-1/2-i ch-tink glass-fiber board insulation.
 - 2. Factory-installed wood nailer at top of turk continuous around curb perimeter.
 - 3. Fabricate curbs to minimum height of 24 tuches unless otherwise indicated.

2.4 EQUIPMENT SUPPORTS

- A. Equipment Supports: Internally reinforced metal equipment supports capable of supporting superimposed live and dead loads including equipment loads and other construction indicated on Drawings; with warded or mechanically fastened and sealed corner joints, and integrally formed deck-mounting fange at perimeter bottom.
 - 1. Basis-or-Design Product: Subject to compliance with requirements, provide Pate Company (The); ES-2 or comparable product by one of the following:
 - a. Conn-Fab Sales Incorporated.b. Thybar Corporation.

. Size: Coordinate dimensions with roughing-in information or Shop Drawings of equipment to be supported.

Material: Zinc-coated (galvanized) steel sheet, 0.052 inch thick.

- 1. Finish: Mill phosphatized.
- D. Construction:
 - 1. Factory-installed continuous wood nailers 3-1/2 inches wide at tops of equipment supports.

Tetra Tech 200-15704-17001

- 2. Metal Counterflashing: Manufacturer's standard, removable, fabricated of same metal and finish as equipment support.
- 3. Fabricate equipment supports to minimum height of 12 inches unless otherwise indicated.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine substrates, areas, and conditions, with Installer present, to verify actual docations, dimensions, and other conditions affecting performance of the Work.
- B. Verify that substrate is sound, dry, smooth, clean, sloped for drainage, and securely anchored.
- C. Verify dimensions of roof openings for roof accessories.
- D. Proceed with installation only after unsatisfactory conditions have been corrected.
- E. Beginning installation constitutes Contractor's acceptance of substrates and conditions.

3.2 INSTALLATION

- A. General: Install roof accessories according to manufacturer's written instructions.
 - 1. Install roof accessories level plume, true to line and elevation, and without warping, jogs in alignment, excessive cil calming, buckling, or tool marks.
 - 2. Anchor roof accessories securely in place so they are capable of resisting indicated loads.
 - 3. Use fasteners, separators sectants, and other miscellaneous items as required to complete installation of root accessories and fit them to substrates.
 - 4. Install roof accessorie to resist exposure to weather without failing, rattling, leaking, or loosening of fasteness and seals.
- B. Metal Protection Protect metals against galvanic action by separating dissimilar metals from contact with each other or with corrosive substrates by painting contact surfaces with bituminous coaing or by other permanent separation as recommended by manufacturer.



Self-Adhering Sheet Underlayment: Install wrinkle free. Apply primer if required by underlayment manufacturer. Comply with temperature restrictions of underlayment manufacturer for installation; use primer rather than nails for installing underlayment at low temperatures. Apply in shingle fashion to shed water. Overlap edges not less than 3-1/2 inches. Roll laps with roller. Cover underlayment within 14 days.

Roof Curb Installation: Install each roof curb so top surface is level.

- D. Equipment Support Installation: Install equipment supports so top surfaces are level with each other.
- E. Seal joints with elastomeric sealant as required by roof accessory manufacturer.

ROOF ACCESSORIES 07 72 00 - 4

3.3 REPAIR AND CLEANING

- A. Galvanized Surfaces: Clean field welds, bolted connections, and abraded areas and repair galvanizing according to ASTM A 780.
- B. Clean exposed surfaces according to manufacturer's written instructions.
- C. Clean off excess sealants.
- D. Replace roof accessories that have been damaged or that cannot be successfully repair finish touchup or similar minor repair procedures.

END OF SECTION

Republic



SPEC SECTION 07 84 13

PENETRATION FIRESTOPPING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Penetrations in fire-resistance-rated walls.
 - 2. Penetrations in horizontal assemblies.
 - 3. Penetrations in smoke barriers.

1.3 SUBMITTALS, GENERAL

A. General: Submit all action submittels (except Samples for Verification) and informational submittals required by this Section concurrently.

1.4 ACTION SUBMITTALS

- A. Product Data: For each type of product indicated.
- B. Product Schedule: For each penetration firestopping system. Include location and design designation of valued testing and inspecting agency.

1.5 INFORMATIONAL SUBMITTALS

Qualification Data: For qualified Installer.

S

Installer Certificates: From Installer indicating penetration firestopping has been installed in compliance with requirements and manufacturer's written recommendations.

Product Test Reports: Based on evaluation of comprehensive tests performed by a qualified testing agency, for penetration firestopping.

ADDITION AND RENOVATIONS

1.6 QUALITY ASSURANCE

- A. Installer Qualifications: A firm experienced in installing penetration firestopping similar in material, design, and extent to that indicated for this Project, whose work has resulted in construction with a record of successful performance. Qualifications include having the necessary experience, staff, and training to install manufacturer's products per specified requirements. Manufacturer's willingness to sell its penetration firestopping products to Contractor or to Installer engaged by Contractor does not in itself confer qualification on layer.
- B. Fire-Test-Response Characteristics: Penetration firestopping shall comply with the follo requirements:
 - 1. Penetration firestopping tests are performed by a qualified testing agency acceptable to authorities having jurisdiction.
 - 2. Penetration firestopping is identical to those tested per testing tendard referenced in "Penetration Firestopping" Article. Provide rated systems complying with the following requirements:
 - a. Penetration firestopping products bear classification marking of qualified testing and inspecting agency.
 - b. Classification markings on penetration fire topping correspond to designations listed by the following:
 - 1) UL in its "Fire Resistance Directory."
- C. Preinstallation Conference: Conduct conference at Project site.

1.7 PROJECT CONDITIONS

- A. Environmental Limitations. Do not install penetration firestopping when ambient or substrate temperatures are outside limit, permitted by penetration firestopping manufacturers or when substrates are wet because o rain, frost, condensation, or other causes.
- B. Install and cure prnetration firestopping per manufacturer's written instructions using natural means of ventila ions or, where this is inadequate, forced-air circulation.

1.8 CCORDINATION

Coordinate construction of openings and penetrating items to ensure that penetration restopping is installed according to specified requirements.

Coordinate sizing of sleeves, openings, core-drilled holes, or cut openings to accommodate penetration firestopping.

C. Notify Owner's testing agency at least seven days in advance of penetration firestopping installations; confirm dates and times on day preceding each series of installations.

PART 2 - PRODUCTS

2.1 MANUFACTURERS

- A. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to, the following
 - 1. A/D Fire Protection Systems Inc.
 - 2. Grace Construction Products.
 - 3. Hilti, Inc.
 - 4. Johns Manville.
 - 5. NUCO Inc.
 - 6. RectorSeal Corporation.
 - 7. Specified Technologies Inc.
 - 8. 3M Fire Protection Products.
 - 9. Tremco, Inc.; Tremco Fire Protection Systems Group.

2.2 PENETRATION FIRESTOPPING

- A. Provide penetration firestopping that is produced and instanted to resist spread of fire according to requirements indicated, resist passage of smoke and other gases, and maintain original fire-resistance rating of construction penetrated. Peretration firestopping systems shall be compatible with one another, with the substrates forming openings, and with penetrating items if any.
- B. Penetrations in Fire-Resistance-Keted Valls: Provide penetration firestopping with ratings determined per ASTM E 814 (1731, 1472), based on testing at a positive pressure differential of 0.01-inch wg.
 - 1. Fire-resistance, atel wals include fire walls, fire-barrier walls, smoke-barrier walls, and fire partitions, as indicated.
 - 2. F-Rating: Not less nan the fire-resistance rating of constructions penetrated.
- C. Penetration in Horizontal Assemblies: Provide penetration firestopping with ratings determined pen ASTM E 814 or UL 1479, based on testing at a positive pressure differential of 0.01 incharg.



- Horizontal assemblies include floors, floor/ceiling assemblies, and ceiling membranes of roof/ceiling assemblies, as indicated.
- F-Rating: At least 1 hour, but not less than the fire-resistance rating of constructions penetrated.
- T-Rating: At least 1 hour, but not less than the fire-resistance rating of constructions penetrated except for floor penetrations within the cavity of a wall.
- D. Penetrations in Smoke Barriers: Provide penetration firestopping with ratings determined per UL 1479.
 - 1. L-Rating: Not exceeding 5.0 cfm/sq. ft. of penetration opening at 0.30-inch wg at both ambient and elevated temperatures.

- E. Exposed Penetration Firestopping: Provide products with flame-spread and smoke-developed indexes of less than 25 and 450, respectively, as determined per ASTM E 84.
- F. Low-Emitting Materials: Penetration firestopping sealants and sealant primers shall comply with the testing and product requirements of the California Department of Health Services' "Standard Practice for the Testing of Volatile Organic Emissions from Various Sources Using Small-Scale Environmental Chambers."
- G. Accessories: Provide components for each penetration firestopping system that are needed to install fill materials and to maintain ratings required. Use only those components specific Usy penetration firestopping manufacturer and approved by qualified testing and inspecting agency for firestopping indicated.
 - 1. Permanent forming/damming/backing materials, including the following
 - a. Slag-wool-fiber or rock-wool-fiber insulation.
 - b. Sealants used in combination with other forming/dataning/backing materials to prevent leakage of fill materials in liquid state.
 - c. Fire-rated form board.
 - d. Fillers for sealants.
 - 2. Temporary forming materials.
 - 3. Substrate primers.
 - 4. Collars.
 - 5. Steel sleeves.

2.3 FILL MATERIALS

- A. Cast-in-Place Firestop Devices: Fictory-assembled devices for use in cast-in-place concrete floors and consisting of a outer metallic sleeve lined with an intumescent strip, a radial extended flange attacked to one end of the sleeve for fastening to concrete formwork, and a neoprene gasket.
- B. Latex Sealant: Seale-component latex formulations that do not re-emulsify after cure during exposure to consure.
- C. Fires op Devices: Factory-assembled collars formed from galvanized steel and lined with international sized to fit specific diameter of penetrant.

Intumescent Composite Sheets: Rigid panels consisting of aluminum-foil-faced elastomeric speet bonded to galvanized-steel sheet.

Intumescent Putties: Nonhardening dielectric, water-resistant putties containing no solvents, inorganic fibers, or silicone compounds.

F. Intumescent Wrap Strips: Single-component intumescent elastomeric sheets with aluminum foil on one side.

- G. Mortars: Prepackaged dry mixes consisting of a blend of inorganic binders, hydraulic cement, fillers, and lightweight aggregate formulated for mixing with water at Project site to form a nonshrinking, homogeneous mortar.
- H. Pillows/Bags: Reusable heat-expanding pillows/bags consisting of glass-fiber cloth cases filled with a combination of mineral-fiber, water-insoluble expansion agents, and fire-retardant additives. Where exposed, cover openings with steel-reinforcing wire mesh to protect pillows/bags from being easily removed.
- I. Silicone Foams: Multicomponent, silicone-based liquid elastomers that, when mixed, ex and cure in place to produce a flexible, nonshrinking foam.
- J. Silicone Sealants: Single-component, silicone-based, neutral-curing elastomeric sealants of grade indicated below:
 - 1. Grade: Pourable (self-leveling) formulation for openings in floors and other horizontal surfaces, and nonsag formulation for openings in vertical and stoped surfaces, unless indicated firestopping limits use of nonsag grade for both opening conditions.

2.4 MIXING

A. For those products requiring mixing before application, comply with penetration firestopping manufacturer's written instructions for accurate proportioning of materials, water (if required), type of mixing equipment, selection of mixer speeds, mixing containers, mixing time, and other items or procedures needed to produce products of uniform quality with optimum performance characteristics for application indicated.

PART 3 - EXECUTION

- 3.1 EXAMINATION
 - A. Examine substrate and conditions, with Installer present, for compliance with requirements for opening configurations, penetrating items, substrates, and other conditions affecting performance of the Work.
 - B. Prosee with installation only after unsatisfactory conditions have been corrected.
 - Beginning installation constitutes Contractor's acceptance of substrates and conditions.

PREPARATION

Surface Cleaning: Clean out openings immediately before installing penetration firestopping to comply with manufacturer's written instructions and with the following requirements:

1. Remove from surfaces of opening substrates and from penetrating items foreign materials that could interfere with adhesion of penetration firestopping.

- 2. Clean opening substrates and penetrating items to produce clean, sound surfaces capable of developing optimum bond with penetration firestopping. Remove loose particles remaining from cleaning operation.
- 3. Remove laitance and form-release agents from concrete.
- B. Priming: Prime substrates where recommended in writing by manufacturer using that manufacturer's recommended products and methods. Confine primers to areas of bond; do not allow spillage and migration onto exposed surfaces.
- C. Masking Tape: Use masking tape to prevent penetration firestopping from contacting adjoining surfaces that will remain exposed on completion of the Work and that would otherwise be permanently stained or damaged by such contact or by cleaning methods used to remove stains. Remove tape as soon as possible without disturbing firestopping's seal with substrate.

3.3 INSTALLATION

- A. General: Install penetration firestopping to comply with manufacturer's written installation instructions and published drawings for products and applications indicated.
- B. Install forming materials and other accessories of types required to support fill materials during their application and in the position needed to produce cross-sectional shapes and depths required to achieve fire ratings indicated.
 - 1. After installing fill materials and allowing them to fully cure, remove combustible forming materials and other access ries not indicated as permanent components of firestopping.
- C. Install fill materials for firestor price by proven techniques to produce the following results:
 - 1. Fill voids and cavities formed by openings, forming materials, accessories, and penetrating items a required to achieve fire-resistance ratings indicated.
 - 2. Apply materials so hey contact and adhere to substrates formed by openings and penetrating items.
 - 3. For fillematorials that will remain exposed after completing the Work, finish to produce smooth uniform surfaces that are flush with adjoining finishes.

3.4 IDINTIFICATION

А,

Identify penetration firestopping with preprinted metal or plastic labels. Attach labels permanently to surfaces adjacent to and within 6 inches of firestopping edge so labels will be visible to anyone seeking to remove penetrating items or firestopping. Use mechanical fasteners or self-adhering-type labels with adhesives capable of permanently bonding labels to surfaces on which labels are placed. Include the following information on labels:

- 1. The words "Warning Penetration Firestopping Do Not Disturb. Notify Building Management of Any Damage."
- 2. Contractor's name, address, and phone number.
- 3. Designation of applicable testing and inspecting agency.
- 4. Date of installation.

- 5. Manufacturer's name.
- 6. Installer's name.

3.5 FIELD QUALITY CONTROL

- A. Owner will engage a qualified testing agency to perform tests and inspections.
- B. Where deficiencies are found or penetration firestopping is damaged or removed becur testing, repair or replace penetration firestopping to comply with requirements.
- C. Proceed with enclosing penetration firestopping with other construction only any impection reports are issued and installations comply with requirements.

3.6 CLEANING AND PROTECTION

- A. Clean off excess fill materials adjacent to openings as the Work progresses by methods and with cleaning materials that are approved in writing by penetration fire topping manufacturers and that do not damage materials in which openings occur.
- B. Provide final protection and maintain conditions during and after installation that ensure that penetration firestopping is without damage or deterioration at time of Substantial Completion. If, despite such protection, damage or deterioration occurs, immediately cut out and remove damaged or deteriorated penetration firestopping and install new materials to produce systems complying with specified requirements.

3.7 PENETRATION FIRESTOPPING SCHUDULE

- A. Where UL-classified systems are indicated, they refer to system numbers in UL's "Fire Resistance Directory" and er product Category XHEZ.
- B. For each location where orire-resistance-rated floor or wall assembly is penetrated, provide a UL-listed penetration firestopping system selected from the applicable UL number range listed in the following schedule that complies with this Section and that is suitable for the penetration conditions indicated for the Project.

Tetra Tech 200-15704-17001



PENETRATION FIRESTOPPING 07 84 13 - 8

ADDITION AND RENOVATIONS

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

| MISCELLANEOUS 7999 7999 F-C-7001- F-B-7001- F-B-7001- F-B-7001- F-B-7001- F-B-7001- F-B-7001- F-B-7001- F-B-7001- F-B-7001- F-C-8001- F-C-8001- F-C-8001- F-C-8001- F-C-8001- F-C-8001- F-C-8001- F-C-8001- B-B-80 | LECTRICAL ENETRANTS | 6999 or F-A-6001- 6999 | 6999 | | | 6999, C-BJ-6001- 6999, or W-BJ-6001- | 6999 |
|--|---|---|---|-------------------|-------------------|---|------------------------------------|
| GROUPINGS OF C-AJ-8001- 8999 F-A-8001- 8999 C-BJ-8001- 8999 F-C-8001- 8999 C-BJ-8001- 8999 C-BJ-8001- 8999 B999 or W-J-8001- 8999 To To To W-J-8001- 8999 END OF SECTION END OF SECTION C-BJ-8001- 8999 C-BJ-8001- 899 C-BJ-8001- 899 C-BJ-8001- 899 C-BJ-8001- 899 C-BJ-8001- 899 C-BJ-8001- 899 C-BJ-8 | 1ISCELLANEOUS 1ECHANICAL 'ENETRANTS | C-AJ-7001- 7999 or F-A-7001- 7999 | C-BJ-7001- 7999 or F-B-7001- 7999 | F-C-7001- 7999 | F-E-7001- 7999 | 6999 C-AJ-7001- 7999, C-BJ-7001- 7999, or W-J-7001- 7999 | W-L-7001- W-M-7091- 7999 - 7979 |
| END OF SECTION | ROUPINGS OF ENETRATIONS | C-AJ-8001- 8999 or F-A-8001- 8999 | C-BJ-8001- 8999 or F-B-8001- 8999 | F-C-8001- 8999 | F-E-8001- 8999 | C-AJ-8001- 8999, C-BJ-8001- 8999, or W-J-8001- 8999 | 11 - 00 - 1 8999 |
| END OF SECTION | ENETRATIONS | or F-A-8001- 8999 | or F-B-8001- 8999 | 8999 | 8999 | 8999, or W-J-8001- 8999 | 8999 |
| | | | | | | . ?` | |
| CR BILL | | | | END (| OF SECTIO | N | |
| | | | | END | OF SECTIO | | |
| | | | | END | OF SECTIO | | |
| | | .0 | 2 | | OF SECTIO | | |



SPEC SECTION 07 84 46

FIRE-RESISTIVE JOINT SYSTEM

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Joints in or between fire-resistance-rated construction
 - 2. Joints in smoke barriers.
- 1.3 SUBMITTALS, GENERAL
 - A. General: Submit all action submittals (except Samples for Verification) and informational submittals required by this Section concurrency.

1.4 ACTION SUBMITTALS

- A. Product Data: For each type of product indicated.
- B. Product Schedule: For each fire-resistive joint system. Include location and design designation of qualified terung agency.

1.5 INFORMATIONAL SUBMITTALS

- Qualification Data: For qualified Installer.
 - Installer Certificates: From Installer indicating fire-resistive joint systems have been installed in compliance with requirements and manufacturer's written recommendations.

Product Test Reports: Based on evaluation of comprehensive tests performed by a qualified testing agency, for fire-resistive joint systems.

1.6 QUALITY ASSURANCE

A. Installer Qualifications: A firm experienced in installing fire-resistive joint systems similar in material, design, and extent to that indicated for this Project, whose work has resulted in

Tetra Tech 200-15704-17001

Α.

construction with a record of successful performance. Qualifications include having the necessary experience, staff, and training to install manufacturer's products per specified requirements. Manufacturer's willingness to sell its fire-resistive joint system products to Contractor or to Installer engaged by Contractor does not in itself confer qualification on buyer.

- B. Fire-Test-Response Characteristics: Fire-resistive joint systems shall comply with the following requirements:
 - 1. Fire-resistive joint system tests are performed by a qualified testing agency accepted authorities having jurisdiction.
 - 2. Fire-resistive joint systems are identical to those tested per testing standard referenced in "Fire-Resistive Joint Systems" Article. Provide rated systems complying with the following requirements:
 - a. Fire-resistive joint system products bear classification marking of qualified testing agency.
 - b. Fire-resistive joint systems correspond to those indicated by reference to designations listed by the following:
 - 1) UL in its "Fire Resistance Directory
- C. Preinstallation Conference: Conduct conference entroject site.

1.7 PROJECT CONDITIONS

- A. Environmental Limitations: Do not insulf re-resistive joint systems when ambient or substrate temperatures are outside limits permitted by fire-resistive joint system manufacturers or when substrates are wet due to rain. For condensation, or other causes.
- B. Install and cure fire-resistive joint systems per manufacturer's written instructions using natural means of ventilation *c*, where this is inadequate, forced-air circulation.

1.8 COORDINATION

- A. Coordinate construction of joints to ensure that fire-resistive joint systems are installed according to specified requirements.
- B. Coordinate sizing of joints to accommodate fire-resistive joint systems.

Notify Owner's testing agency at least seven days in advance of fire-resistive joint system installations; confirm dates and times on day preceding each series of installations.

RT 2 - PRODUCTS

2.1 FIRE-RESISTIVE JOINT SYSTEMS

A. Where required, provide fire-resistive joint systems that are produced and installed to resist spread of fire according to requirements indicated, resist passage of smoke and other gases, and

maintain original fire-resistance rating of assemblies in or between which fire-resistive joint systems are installed. Fire-resistive joint systems shall accommodate building movements without impairing their ability to resist the passage of fire and hot gases.

- B. Joints in or between Fire-Resistance-Rated Construction: Provide fire-resistive joint systems with ratings determined per ASTM E 1966 or UL 2079:
 - 1. Joints include those installed in or between fire-resistance-rated walls, floor/ceiling assemblies, and roofs or roof/ceiling assemblies.
 - 2. Fire-Resistance Rating: Equal to or exceeding the fire-resistance rating of constructive they will join.
 - 3. Manufacturers: Subject to compliance with requirements, available nonsfacturers offering products that may be incorporated into the Work include, but are no limited to, the following:
 - a. A/D Fire Protection Systems Inc.
 - b. Fire Trak Corp.
 - c. Grace Construction Products.
 - d. Hilti, Inc.
 - e. Johns Manville.
 - f. Nelson Firestop Products.
 - g. NUCO Inc.
 - h. Passive Fire Protection Partners.
 - i. RectorSeal Corporation.
 - j. Specified Technologies Inc.
 - k. 3M Fire Protection Products
 - 1. Tremco, Inc.; Tremco, Tre Projection Systems Group.
 - m. USG Corporation.
- C. Joints in Smoke Barriers: Provide fire-resistive joint systems with ratings determined per UL 2079.
 - 1. L-Rating: Not exceeding 5.0 cfm/ft of joint at 0.30 inch wg at both ambient and elevated temperature.
 - 2. Manufacturers Subject to compliance with requirements, available manufacturers off any products that may be incorporated into the Work include, but are not limited to, the following:
 - Grace Construction Products.
 - Hilti, Inc.
 - c. Johns Manville.
 - d. Nelson Firestop Products.
 - e. NUCO Inc.
 - f. Passive Fire Protection Partners.
 - g. Specified Technologies Inc.
 - h. 3M Fire Protection Products.
 - i. Tremco, Inc.; Tremco Fire Protection Systems Group.
- D. Exposed Fire-Resistive Joint Systems: Provide products with flame-spread and smokedeveloped indexes of less than 25 and 450, respectively, as determined per ASTM E 84.

- E. Low-Emitting Materials: Fire-resistive joint system sealants shall comply with the testing and product requirements of the California Department of Health Services' "Standard Practice for the Testing of Volatile Organic Emissions from Various Sources Using Small-Scale Environmental Chambers."
- F. Accessories: Provide components of fire-resistive joint systems, including primers and forming materials, that are needed to install fill materials and to maintain ratings required. Use only components specified by fire-resistive joint system manufacturer and approved by the qualified testing agency for systems indicated.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine substrates and conditions, with Installer present, for compliance with requirements for joint configurations, substrates, and other conditions affecting performance of the Work.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.
- C. Beginning installation constitutes Contractor's acceptance of substrates and conditions.

3.2 PREPARATION

- A. Surface Cleaning: Clean joints immediately before installing fire-resistive joint systems to comply with fire-resistive joint system nanufacturer's written instructions and the following requirements:
 - 1. Remove from surfaces of joint substrates foreign materials that could interfere with adhesion of fill material
 - 2. Clean joint substrates to produce clean, sound surfaces capable of developing optimum bond with fill materials. Remove loose particles remaining from cleaning operation.
 - 3. Remove laitince and form-release agents from concrete.
- B. Priming: Prime substrates where recommended in writing by fire-resistive joint system manufacturer using that manufacturer's recommended products and methods. Confine primers to see of bond; do not allow spillage and migration onto exposed surfaces.



Masking Tape: Use masking tape to prevent fill materials of fire-resistive joint system from contacting adjoining surfaces that will remain exposed on completion of the Work and that would otherwise be permanently stained or damaged by such contact or by cleaning methods used to remove stains. Remove tape as soon as possible without disturbing fire-resistive joint system's seal with substrates.

INSTALLATION

A. General: Install fire-resistive joint systems to comply with manufacturer's written installation instructions and published drawings for products and applications indicated.

- B. Install forming materials and other accessories of types required to support fill materials during their application and in position needed to produce cross-sectional shapes and depths required to achieve fire ratings indicated.
 - 1. After installing fill materials and allowing them to fully cure, remove combustible forming materials and other accessories not indicated as permanent components of fire resistive joint system.
- C. Install fill materials for fire-resistive joint systems by proven techniques to prod following results:
 - 1. Fill voids and cavities formed by joints and forming materials as required to echieve fire-resistance ratings indicated.
 - 2. Apply fill materials so they contact and adhere to substrates former by jo
 - 3. For fill materials that will remain exposed after completing the Work, finish to produce smooth, uniform surfaces that are flush with adjoining finishes.

3.4 IDENTIFICATION

- A. Identify fire-resistive joint systems with preprinted meal or plastic labels. Attach labels permanently to surfaces adjacent to and within 6 inches of joint edge so labels will be visible to anyone seeking to remove or penetrate joint system. One mechanical fasteners or self-adhering-type labels with adhesives capable of permanently bonding labels to surfaces on which labels are placed. Include the following information on labels:
 - 1. The words "Warning Fire-Resistive Joint System Do Not Disturb. Notify Building Management of Any Damage"
 - 2. Contractor's name, address, and phone number.
 - 3. Designation of applicable testing agency.
 - 4. Date of installation.
 - 5. Manufacturer's name
 - 6. Installer's name.
- 3.5 FIELD QUALITY CONTROL
 - A. Inspecting Agency: Owner will engage a qualified testing agency to perform tests and inspections.

Where deficiencies are found or fire-resistive joint systems are damaged or removed due to using, repair or replace fire-resistive joint systems so they comply with requirements.

Proceed with enclosing fire-resistive joint systems with other construction only after inspection reports are issued and installations comply with requirements.

ADDITION AND RENOVATIONS

3.6 CLEANING AND PROTECTING

- A. Clean off excess fill materials adjacent to joints as the Work progresses by methods and with cleaning materials that are approved in writing by fire-resistive joint system manufacturers and that do not damage materials in which joints occur.
- B. Provide final protection and maintain conditions during and after installation that ensure fileresistive joint systems are without damage or deterioration at time of Substantial Completion. If damage or deterioration occurs despite such protection, cut out and remove damage or deteriorated fire-resistive joint systems immediately and install new materials to produce Sizeresistive joint systems complying with specified requirements.

3.7 FIRE-RESISTIVE JOINT SYSTEM SCHEDULE

- A. Where UL-classified systems are indicated, they refer to system humbers in UL's "Fire Resistance Directory" under product Category XHBN or Category XHDG.
 - 1. Systems are as indicated on Drawings.

RBIDDI

END OF SECTION

< fC

SPEC SECTION 07 92 00

JOINT SEALANT

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Silicone joint sealants.
 - 2. Urethane joint sealants.
 - 3. Latex joint sealants.

1.3 SUBMITTALS, GENERAL

A. General: Submit all action submittals (except Samples for Verification) and informational submittals required by this Section concurrently.

1.4 ACTION SUBMITTALS

- A. Product Data: For each join sealant product indicated.
- B. Samples for Initial Selection: Manufacturer's color charts consisting of strips of cured sealants showing the full large of colors available for each product exposed to view.
- C. Samples for Verification: For each kind and color of joint sealant required, provide Samples with joint sealants in 1/2-inch-wide joints formed between two 6-inch-long strips of material matching the appearance of exposed surfaces adjacent to joint sealants.

oint-Sealant Schedule: Include the following information:

- Joint-sealant application, joint location, and designation.
- 2. Joint-sealant manufacturer and product name.
- 3. Joint-sealant formulation.
- 1.5 INFORMATIONAL SUBMITTALS
 - A. Qualification Data: For qualified Installer.

Tetra Tech 200-15704-17001

1.

- B. Product Certificates: For each kind of joint sealant and accessory, from manufacturer.
- C. Product Test Reports: Based on evaluation of comprehensive tests performed by a qualified testing agency, indicating that sealants comply with requirements.

1.6 QUALITY ASSURANCE

- A. Installer Qualifications: Manufacturer's authorized representative who is trained and appr for installation of units required for this Project.
- B. Source Limitations: Obtain each kind of joint sealant from single source from single manufacturer.
- C. Mockups: Install sealant in mockups of assemblies specified in other Sections that are indicated to receive joint sealants specified in this Section. Use materizes and asstallation methods specified in this Section.
- D. Preinstallation Conference: Conduct conference at Project site

1.7 PROJECT CONDITIONS

- A. Do not proceed with installation of joint sealasts under the following conditions:
 - 1. When ambient and substrate temperature conditions are outside limits permitted by jointsealant manufacturer or are below 40 deg F.
 - 2. When joint substrates are wit.
 - 3. Where joint widths are less than those allowed by joint-sealant manufacturer for applications indicated.
 - 4. Where contaminants capable of interfering with adhesion have not yet been removed from joint substrate:

PART 2 - PRODUCTS

2.1 GENERAL MATE

Compatibility: Provide joint sealants, backings, and other related materials that are compatible with one another and with joint substrates under conditions of service and application, as demonstrated by joint-sealant manufacturer, based on testing and field experience.

Α.

Low-Emitting Interior Sealants: Sealants and sealant primers used inside the weatherproofing system shall comply with the testing and product requirements of the California Department of Health Services' "Standard Practice for the Testing of Volatile Organic Emissions from Various Sources Using Small-Scale Environmental Chambers."

C. Liquid-Applied Joint Sealants: Comply with ASTM C 920 and other requirements indicated for each liquid-applied joint sealant specified, including those referencing ASTM C 920 classifications for type, grade, class, and uses related to exposure and joint substrates.

JOINT SEALANT 07 92 00 - 2 Tetra Tech 200-15704-17001

Type S.

- D. Stain-Test-Response Characteristics: Where sealants are specified to be nonstaining to porous substrates, provide products that have undergone testing according to ASTM C 1248 and have not stained porous joint substrates indicated for Project.
- E. Suitability for Contact with Food: Where sealants are indicated for joints that will come in repeated contact with food, provide products that comply with 21 CFR 177.2600.
- F. Colors of Exposed Joint Sealants: As selected by Architect from manufacturer's full rang

2.2 SILICONE JOINT SEALANTS

- A. Single-Component, Nonsag, Neutral-Curing Silicone Joint Sealant: AST4 C 9. Grade NS, Class 100/50, for Use NT.
 - 1. Products: Subject to compliance with requirements, availably products that may be incorporated into the Work include, but are not limited to the following:
 - a. Dow Corning Corporation; 790.
 - b. GE Advanced Materials Silicones; SilPru^eLM SC52700.
 - c. Pecora Corporation; 890.
 - d. Tremco Incorporated; Spectrem 1.
- B. Mildew-Resistant, Single-Component, Acid Cuing Silicone Joint Sealant: ASTM C 920, Type S, Grade NS, Class 25, for Use NT.
 - 1. Products: Subject to compliance with requirements, available products that may be incorporated into the Work include but are not limited to, the following:
 - a. BASF Building Systems; Omniplus.
 - b. Dow Corning For pration; 786 Mildew Resistant.
 - c. GE Advance L Maerials Silicones; Sanitary SCS1700.

2.3 URETHANE OD T SEALANTS

A. Single-Component, Nonsag, Urethane Joint Sealant: ASTM C 920, Type S, Grade NS, Class 25, for U/e NT.

Products: Subject to compliance with requirements, available products that may be incorporated into the Work include, but are not limited to, the following:

- a. BASF Building Systems; Sonolastic NP1.
- b. Pecora Corporation; Dynatrol I-XL.
- c. Sika Corporation, Construction Products Division; Sikaflex 1a.
- d. Tremco Incorporated; Dymonic.
- B. Single-Component, Pourable, Traffic-Grade, Urethane Joint Sealant: ASTM C 920, Type S, Grade P, Class 25, for Use T.
 - 1. Products: Subject to compliance with requirements, available products that may be incorporated into the Work include, but are not limited to, the following:

Tetra Tech 200-15704-17001

- a. BASF Building Systems; Sonolastic SL 1.
- b. Bostik, Inc.; Chem-Calk 950.
- c. Pecora Corporation; Urexpan NR-201.
- d. Tremco Incorporated; Vulkem 45.

2.4 LATEX JOINT SEALANTS

- A. Latex Joint Sealant: Acrylic latex or siliconized acrylic latex, ASTM C 834, Type Grade NF.
 - 1. Products: Subject to compliance with requirements, available products hat may be incorporated into the Work include, but are not limited to, the following
 - a. BASF Building Systems; Sonolac.
 - b. Bostik, Inc.; Chem-Calk 600.
 - c. Pecora Corporation; AC-20+.
 - d. Tremco Incorporated; Tremflex 834.

2.5 JOINT SEALANT BACKING

- A. General: Provide sealant backings of material hat an nonstaining; are compatible with joint substrates, sealants, primers, and other joint filles; and are approved for applications indicated by sealant manufacturer based on field experience and laboratory testing.
- B. Cylindrical Sealant Backings: ASTMES 1.30, Type C (closed-cell material with a surface skin) or Type O (open-cell material), and of size and density to control sealant depth and otherwise contribute to producing optimum real properformance.
- C. Bond-Breaker Tape: Potentylene tape or other plastic tape recommended by sealant manufacturer for preventing sealant from adhering to rigid, inflexible joint-filler materials or joint surfaces at back of oin. Provide self-adhesive tape where applicable.

2.6 MISCELLANE AUS MATERIALS

A. Primer: Material recommended by joint-sealant manufacturer where required for adhesion of sealant to joint substrates indicated, as determined from preconstruction joint-sealant-substrate tests and field tests.



Neaners for Nonporous Surfaces: Chemical cleaners acceptable to manufacturers of sealants and sealant backing materials, free of oily residues or other substances capable of staining or harming joint substrates and adjacent nonporous surfaces in any way, and formulated to promote optimum adhesion of sealants to joint substrates.

2. Masking Tape: Nonstaining, nonabsorbent material compatible with joint sealants and surfaces adjacent to joints.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine joints indicated to receive joint sealants, with Installer present, for compliance with requirements for joint configuration, installation tolerances, and other conditions affecting joint-sealant performance.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.
- C. Beginning installation constitutes Contractor's acceptance of substrates and contractor

3.2 PREPARATION

- A. Surface Cleaning of Joints: Clean out joints immediately before installing joint sealants to comply with joint-sealant manufacturer's written instructions and the following requirements:
 - 1. Remove all foreign material from joint substrate, that could interfere with adhesion of joint sealant, including dust, paints (except for perpanent, protective coatings tested and approved for sealant adhesion and compatibility by sealant manufacturer), old joint sealants, oil, grease, waterproofing, water reperents, water, surface dirt, and frost.
 - Clean porous joint substrate surfaces by rusbing, grinding, mechanical abrading, or a combination of these methods to produce clean, sound substrate capable of developing optimum bond with joint sealants. Remove loose particles remaining after cleaning operations above by vacuuming or bowing out joints with oil-free compressed air.
 - 3. Remove laitance and form-t lease igents from concrete.
 - 4. Clean nonporous joint substrate surfaces with chemical cleaners or other means that do not stain, harm substrates, or leave residues capable of interfering with adhesion of joint sealants.
- B. Joint Priming: Prime yant substrates where recommended by joint-sealant manufacturer or as indicated by preconstruction joint-sealant-substrate tests or prior experience. Apply primer to comply with joint-sealant manufacturer's written instructions. Confine primers to areas of joint-sealant bords do not allow spillage or migration onto adjoining surfaces.
- C. Masting Taper Use masking tape where required to prevent contact of sealant or primer with adjuining surfaces that otherwise would be permanently stained or damaged by such contact or by cleaning methods required to remove sealant smears. Remove tape immediately after tooling without disturbing joint seal.

INSTALLATION OF JOINT SEALANTS

- General: Comply with joint-sealant manufacturer's written installation instructions for products and applications indicated, unless more stringent requirements apply.
- B. Sealant Installation Standard: Comply with recommendations in ASTM C 1193 for use of joint sealants as applicable to materials, applications, and conditions indicated.

Tetra Tech 200-15704-17001

- C. Install sealant backings of kind indicated to support sealants during application and at position required to produce cross-sectional shapes and depths of installed sealants relative to joint widths that allow optimum sealant movement capability.
 - 1. Do not leave gaps between ends of sealant backings.
 - 2. Do not stretch, twist, puncture, or tear sealant backings.
 - 3. Remove absorbent sealant backings that have become wet before sealant application and replace them with dry materials.
- D. Install bond-breaker tape behind sealants where sealant backings are not used between sea and backs of joints.
- E. Install sealants using proven techniques that comply with the following and at the same time backings are installed:
 - 1. Place sealants so they directly contact and fully wet joint substra
 - 2. Completely fill recesses in each joint configuration.
 - 3. Produce uniform, cross-sectional shapes and depths relative to joint widths that allow optimum sealant movement capability.
- F. Tooling of Nonsag Sealants: Immediately after sealan application and before skinning or curing begins, tool sealants according to requirements specified in subparagraphs below to form smooth, uniform beads of configuration indicated, to eliminate air pockets; and to ensure contact and adhesion of sealant with sides of joint.
 - 1. Remove excess sealant from surfaces edjacent to joints.
 - 2. Use tooling agents that are approved in writing by sealant manufacturer and that do not discolor sealants or adjacent surfacts.
 - 3. Provide concave joint profile per Figure 8A in ASTM C 1193, unless otherwise indicated.
- 3.4 CLEANING
 - A. Clean off excess scelant or sealant smears adjacent to joints as the Work progresses by methods and with change materials approved in writing by manufacturers of joint sealants and of products in which joints occur.

3.5 **PROTECTION**

20

Notect joint sealants during and after curing period from contact with contaminating substances and from damage resulting from construction operations or other causes so sealants are without deterioration or damage at time of Substantial Completion. If, despite such protection, damage or deterioration occurs, cut out and remove damaged or deteriorated joint sealants immediately so installations with repaired areas are indistinguishable from original work.

3.6 JOINT-SEALANT SCHEDULE

A. Joint-Sealant Application: Exterior joints up to 1 inch wide in horizontal nontraffic surfaces.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- 1. Joint Locations:
 - a. Control and expansion joints in ceilings and other overhead surfaces.
 - b. Other joints as indicated.
- 2. Urethane Joint Sealant: Single component, nonsag, Class 25.
- B. Joint-Sealant Application: Exterior joints in vertical surfaces.
 - 1. Joint Locations:
 - a. Construction joints in cast-in-place concrete.
 - b. Control and expansion joints in unit masonry.
 - c. Joints in dimension stone cladding.
 - d. Joints in cast stone.
 - e. Joints between metal panels.
 - f. Joints between different materials listed above.
 - g. Perimeter joints between materials listed above and here's of doors, windows and louvers.
 - h. Other joints as indicated.
 - 2. Silicone Joint Sealant: Single component, possag, neutral curing, Class 100/50.
- C. Joint-Sealant Application: Interior joints up to 1 schulide in horizontal traffic surfaces.
 - 1. Joint Locations:
 - a. Isolation joints in cassin-place concrete slabs.
 - b. Control and expansion primes in tile flooring.
 - c. Other joints as a dicate
 - 2. Urethane Joint Lealunt: Single component, pourable, traffic grade.
- D. Joint-Sealant Application, interior joints up to 1 inch wide in horizontal nontraffic surfaces.
 - 1. Joint Locations:

. Control and expansion joints in ceilings and other overhead surfaces. Other joints as indicated.

Urethane Joint Sealant: Single component, nonsag, Class 25.

Joint-Sealant Application: Interior joints in vertical surfaces.

- Joint Locations:
 - a. Control and expansion joints on exposed interior surfaces of exterior walls.
 - b. Perimeter joints of exterior openings.
 - c. Other joints as indicated.
- 2. Silicone Joint Sealant: Single component, nonsag, neutral curing, Class 100/50.

Tetra Tech 200-15704-17001

- F. Joint-Sealant Application: Mildew-resistant interior joints in vertical surfaces.
 - 1. Joint Sealant Location:
 - a. Tile control and expansion joints.
 - b. Other joints as indicated.
 - 2. Silicone Joint Sealant: Single component, nonsag, mildew resistant, acid curing.
- G. Joint-Sealant Application: Interior joints in vertical surfaces.
 - 1. Joint Locations:
 - a. Vertical joints on exposed surfaces of interior unit masonry concrete, and gypsum board walls and partitions.
 - b. Perimeter joints between interior wall surfaces and filmer of interior doors, windows, and elevator entrances.
 - c. Other joints as indicated.

RBIDL

2. Joint Sealant: Latex.

END OF SECTION

<

SPEC SECTION 08 11 13

HOLLOW METAL DOORS AND FRAMES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Custom hollow metal doors and frames.
- 1.3 DEFINITIONS
 - A. Minimum Thickness: Minimum thickness of base metal without coatings.
 - B. Custom Hollow Metal Work: Hollow heta work fabricated according to ANSI/NAAMM-HMMA 861.

1.4 SUBMITTALS, GENERA

A. General: Submit all action submittals required by this Section concurrently.

1.5 ACTION SUBMITTALS

- A. Product Data: For each type of product indicated. Include construction details, material descriptions, core descriptions, fire-resistance rating, temperature-rise ratings, and finishes.
- 3. Shop Dawings: Include the following:
 - Elevations of each door design.
 - 2. Details of doors, including vertical and horizontal edge details and metal thicknesses.
 - 3. Frame details for each frame type, including dimensioned profiles and metal thicknesses.
 - 4. Locations of reinforcement and preparations for hardware.
 - 5. Details of each different wall opening condition.
 - 6. Details of anchorages, joints, field splices, and connections.
 - 7. Details of accessories.
 - 8. Details of moldings, removable stops, and glazing.
- C. Other Action Submittals:

1. Schedule: Provide a schedule of hollow metal work prepared by or under the supervision of supplier, using same reference numbers for details and openings as those on Drawings. Coordinate with door hardware schedule.

1.6 QUALITY ASSURANCE

- A. Source Limitations: Obtain hollow metal work from single source from single manufacturer.
- B. Fire-Rated Door Assemblies: Assemblies complying with NFPA 80 that are listed and laceled by a qualified testing agency, for fire-protection ratings indicated, based on testing at positive pressure according to NFPA 252 or UL 10C.
 - 1. Temperature-Rise Limit: At vertical exit enclosures and exit passegevays, provide doors that have a maximum transmitted temperature end point of not more than 450 deg Fabove ambient after 30 minutes of standard fire-test exposure.
- C. Fire-Rated, Borrowed-Light Frame Assemblies: Assemblies complying with NFPA 80 that are listed and labeled, by a testing and inspecting agency acceptible to authorities having jurisdiction, for fire-protection ratings indicated, based of testing according to NFPA 257. Label each individual glazed lite.
- D. Smoke-Control Door Assemblies: Comply with UL
- E. Preinstallation Conference: Conduct conference at Project site.

1.7 DELIVERY, STORAGE, AND HUNDLING

- A. Deliver hollow metal work calletized, wrapped, or crated to provide protection during transit and Project-site storage. Denoture nonvented plastic.
- B. Deliver welded frames with we removable spreader bars across bottom of frames, tack welded to jambs and multions.
- C. Store hollow me al work under cover at Project site. Place in stacks of five units maximum in a vertical fosition with heads up, spaced by blocking, on minimum 4-inch-high wood blocking. Do not store in a manner that traps excess humidity.

Provide minimum 1/4-inchspace between each stacked door to permit air circulation.

COORDINATION

Coordinate installation of anchorages for hollow metal frames. Furnish setting drawings, templates, and directions for installing anchorages, including sleeves, concrete inserts, anchor bolts, and items with integral anchors. Deliver such items to Project site in time for installation.
PART 2 - PRODUCTS

2.1 MANUFACTURERS

- A. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to, the following
 - 1. Ceco Door Products; an Assa Abloy Group company.
 - 2. Curries Company; an Assa Abloy Group company.
 - 3. Pioneer Industries, Inc.
 - 4. Steelcraft; an Ingersoll-Rand company.

2.2 MATERIALS

- A. Cold-Rolled Steel Sheet: ASTM A 1008/A 1008M, Commercial Steer (CS), Type B; suitable for exposed applications.
- B. Hot-Rolled Steel Sheet: ASTM A 1011/A 1011M, Commercial Steel (CS), Type B; free of scale, pitting, or surface defects; pickled and oiled.
- C. Metallic-Coated Steel Sheet: ASTM A 653/A 653/A 653/A Commercial Steel (CS), Type B; with minimum G60 or A60 metallic coating.
- D. Frame Anchors: ASTM A 591/A 591M Commercial Steel (CS), 40Z coating designation; mill phosphatized.
- E. Inserts, Bolts, and Fasteners: How div ga vanized according to ASTM A 153/A 153M.
- F. Powder-Actuated Fasteners in Concrete: Fastener system of type suitable for application indicated, fabricated from corresion-resistant materials, with clips or other accessory devices for attaching hollow meta-frames of type indicated.
- G. Grout: ASTM C 76, except with a maximum slump of 4 inches, as measured according to ASTM C 143/C 43M.
- H. Minetal-Siber Insulation: ASTM C 665, Type I (blankets without membrane facing); consisting of obert manufactured from slag or rock wool with 6- to 12-lb/cu. ft. density; with maximum flame spread and smoke-development indexes of 25 and 50, respectively; passing ASTM E 136 for combustion characteristics.

20

Glazing: Comply with requirements in Division 08 Section "Glazing."

Bituminous Coating: Cold-applied asphalt mastic, SSPC-Paint 12, compounded for 15-mil dry film thickness per coat. Provide inert-type noncorrosive compound free of asbestos fibers, sulfur components, and other deleterious impurities.

ADDITION AND RENOVATIONS

2.3 CUSTOM HOLLOW METAL DOORS

- A. General: Provide doors not less than 1-3/4 inches thick, of seamless hollow construction unless otherwise indicated. Construct doors with smooth surfaces without visible joints or seams on exposed faces. Comply with ANSI/NAAMM-HMMA 861.
- B. Interior Door Face Sheets: Fabricated from cold-rolled steel sheet unless metallic-coated sheet is indicated, minimum 0.042 inch thick.
- C. Core Construction:
 - 1. Steel-Stiffened Core: 0.026-inch-thick, steel vertical stiffeners of same knaerial as face sheets extending full-door height, with vertical webs spaced not more than 6 inches apart, spot welded to face sheets a maximum of 5 inches o.c. Spaces filled between stiffeners with glass- or mineral-fiber insulation.
 - a. Fire Door Core: As required to provide fin-protection and temperature-rise ratings indicated.
- D. Vertical Edges for Single-Acting Doors: Beveled 148 inch in 2 inches.
- E. Top and Bottom Channels: Closed with continuous channels, minimum 0.053 inchthick, of same material as face sheets and spot welded to both face sheets.
- F. Hardware Reinforcement: Fabricate according to ANSI/NAAMM-HMMA 861 with reinforcing plates from same material as door face sheets.

2.4 CUSTOM HOLLOW METAL VERAMES

- A. General: Fabricate frames of construction indicated. Close contact edges of corner joints tight with faces mitered and steps butted or mitered. Continuously weld faces and soffits and finish faces smooth. Corolly with ANSI/NAAMM-HMMA 861.
 - 1. Dror Frimes for Openings 48 Inches Wide or Less: Fabricated from 0.053-inch-thick steel sheet.
 - 2. Poor Frames for Openings More Than 48 Inches Wide: Fabricated from 0.067-inchthick steel sheet.
 - Sidelight and Transom Frames: Fabricated from same thickness material as adjacent door frame.
 - Borrowed-Light Frames: Fabricated from 0.053-inch-thick steel sheet.

Interior Frames: Fabricated from cold-rolled steel sheet unless metallic-coated sheet is indicated.

- C. Hardware Reinforcement: Fabricate according to ANSI/NAAMM-HMMA 861 with reinforcing plates from same material as frame.
- D. Head Reinforcement: Provide minimum 0.093-inch-thick, steel channel or angle stiffener for opening widths more than 48 inches.

HOLLOW METAL DOORS AND FRAMES 08 11 13 - 4

2.5 FRAME ANCHORS

- A. Jamb Anchors:
 - 1. Masonry Type: Adjustable strap-and-stirrup or T-shaped anchors to suit frame size, not less than 0.042 inch thick, with corrugated or perforated straps not less than 2 incher wide by 10 inches long; or wire anchors not less than 0.177 inch thick.
 - 2. Stud-Wall Type: Designed to engage stud, welded to back of frames; not less than 0.04 inch thick.
- B. Floor Anchors: Formed from same material as frames, not less than 0.042 inch take, and as follows:
 - 1. Monolithic Concrete Slabs: Clip-type anchors, with two holes to receive horeners

2.6 STOPS AND MOLDINGS

- A. Moldings for Glazed Lites in Doors: Minimum 0.032 inch thek, fabricated from same material as door face sheet in which they are installed.
- B. Fixed Frame Moldings: Formed integral with hollow metal frames, a minimum of 5/8 inch high unless otherwise indicated.
- C. Loose Stops for Glazed Lites in Frames. Minimum 0.032 inch thick, fabricated from same material as frames in which they are installed

2.7 ACCESSORIES

- A. Mullions and Transom Bars, Join to adjacent members by welding or rigid mechanical anchors.
- B. Ceiling Struts: Minimum 1/1-inch-thick by 1-inch-wide steel.
- C. Grout Guards: Formed from same material as frames, not less than 0.016 inch thick.

2.8 FABPICATIO

- A. Fabruate hollow metal work to be rigid and free of defects, warp, or buckle. Accurately form metal to required sizes and profiles, with minimum radius for thickness of metal. Where ractical, fit and assemble units in manufacturer's plant. To ensure proper assembly at Project site, clearly identify work that cannot be permanently factory assembled before shipment.
 - Tolerances: Fabricate hollow metal work to tolerances indicated in ANSI/NAAMM-HMMA 861.
- C. Hollow Metal Doors:
 - 1. Glazed Lites: Factory cut openings in doors.

- 2. Astragals: Provide overlapping astragal on one leaf of pairs of doors where required by NFPA 80 for fire-performance rating or where indicated. Extend minimum 3/4 inch beyond edge of door on which astragal is mounted.
- Hollow Metal Frames: Where frames are fabricated in sections due to shipping or handling D. limitations, provide alignment plates or angles at each joint, fabricated of same thickness meta as frames.
 - Welded Frames: Weld flush face joints continuously; grind, fill, dress, and make 1. flush, and invisible.
 - 2. Sidelight and Transom Bar Frames: Provide closed tubular members with no face seams or joints, fabricated from same material as door frame. Fasten nemoers at crossings and to jambs by butt welding.
 - 3. Grout Guards: Weld guards to frame at back of hardware months s frames to be grouted.
 - 4. Floor Anchors: Weld anchors to bottom of jambs and multions with at least four spot welds per anchor.
 - Jamb Anchors: Provide number and spacing of anchors as follow 5.
 - Masonry Type: Locate anchors not more than 18 inches from top and bottom of a. frame. Space anchors not more than 32 inches o.c. and as follows:
 - Two anchors per jamb up to 60 inches high. 1)
 - 2)
 - Three anchors per jamb from 60 to 90 inches high. Four anchors per jamb from 60 to 120 inches high. 3)
 - Stud-Wall Type: Locate archers not more than 18 inches from top and bottom of b. frame. Space anchors not more than 32 inches o.c. and as follows:
 - Three and ors per jamb up to 60 inches high. 1)
 - Four and other jamb from 60 to 90 inches high. 2)
 - Fire a show per jamb from 90 to 96 inches high. 3)
 - Five anciers per jamb plus 1 additional anchor per jamb for each 24 inches 4) or fraction thereof above 96 inches high.
 - anchors per head for frames above 42 inches wide and mounted in metal-stud partitions.
 - encers: Drill stops to receive door silencers as follows. Keep holes clear during onstruction.
 - Single-Door Frames: Drill stop in strike jamb to receive three door silencers.
 - Double-Door Frames: Drill stop in head jamb to receive two door silencers.

Fabricate concealed stiffeners, edge channels, and hardware reinforcement from either cold- or hot-rolled steel sheet.

- 1. Locate hardware as indicated, or if not indicated, according to ANSI/NAAMM-HMMA 861.
- 2. Reinforce doors and frames to receive nontemplated, mortised and surface-mounted door hardware.

a.

b.

- F. Stops and Moldings: Provide stops and moldings around glazed lites where indicated. Form corners of stops and moldings with butted or mitered hairline joints.
 - 1. Single Glazed Lites: Provide fixed stops and moldings welded on secure side of hollow metal work.
 - 2. Multiple Glazed Lites: Provide fixed and removable stops and moldings so that each glazed lite is capable of being removed independently.
 - 3. Provide fixed frame moldings on outside of exterior and on secure side of interior doors and frames.
 - 4. Provide loose stops and moldings on inside of hollow metal work.
 - 5. Coordinate rabbet width between fixed and removable stops with type of grazing and type of installation indicated.

2.9 STEEL FINISHES

- A. Prime Finish: Apply manufacturer's standard primer immediately at a cleaning and pretreating.
 - 1. Shop Primer: Manufacturer's standard, fast-curing, lead- and chromate-free primer complying with ANSI/SDI A250.10 acceptance diterra; recommended by primer manufacturer for substrate; compatible with substrate and field-applied coatings despite prolonged exposure.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine substrates, areas, and conditions, with Installer present, for compliance with requirements for installation to carces and other conditions affecting performance of the Work.
- B. Examine roughing-in the embedded and built-in anchors to verify actual locations before frame installation.
- C. Proceed with in allation only after unsatisfactory conditions have been corrected.
- D. Beginnin installation constitutes Contractor's acceptance of substrates and conditions.

PREPARATION



3.2

Remove welded-in shipping spreaders installed at factory. Restore exposed finish, as required to make repaired area smooth, flush, and invisible on exposed faces.

Prior to installation, adjust and securely brace welded hollow metal frames for squareness, alignment, twist, and plumbness to the following tolerances:

- 1. Squareness: Plus or minus 1/16 inch, measured at door rabbet on a line 90 degrees from jamb perpendicular to frame head.
- 2. Alignment: Plus or minus 1/16 inch, measured at jambs on a horizontal line parallel to plane of wall.

- 3. Twist: Plus or minus 1/16 inch, measured at opposite face corners of jambs on parallel lines, and perpendicular to plane of wall.
- 4. Plumbness: Plus or minus 1/16 inch, measured at jambs on a perpendicular line from head to floor.
- C. Drill and tap frames to receive nontemplated, mortised, and surface-mounted door hardware.

3.3 **INSTALLATION**

- A. General: Install hollow metal work plumb, rigid, properly aligned, and securely place; comply with Drawings and manufacturer's written instructions.
- Hollow Metal Frames: Install hollow metal frames of size and profile in the B. omply with HMMA 840.
 - 1. Set frames accurately in position, plumbed, aligned, and bra securely until permanent anchors are set. After wall construction is complete, remove emporary braces, leaving surfaces smooth and undamaged.
 - a.
 - At fire-protection-rated openings, install frames according to NFPA 80. Where frames are fabricated in metions because of shipping or handling b. limitations, field splice at approve locations by welding face joint continuously; grind, fill, dress, and make splice smooth flush, and invisible on exposed faces. Install frames with removable glazing stops located on secure side of opening. Install door silencers in frames before grouting. Remove temporary brees necessary for installation only after frames have been
 - c.
 - d.
 - e. properly set and secured.
 - Check plumbness, squarpass, and twist of frames as walls are constructed. Shim f. as necessary to comply with installation tolerances.
 - Field apply brunch us coating to backs of frames that are filled with grout containing artifreizing agents. g.
 - Floor Approx: Provide floor anchors for each jamb and mullion that extends to floor, 2. and secure ith postinstalled expansion anchors.

or anchors may be set with powder-actuated fasteners instead of postinstalled bansion anchors if so indicated and approved on Shop Drawings.



Letal-Stud Partitions: Solidly pack mineral-fiber insulation behind frames.

Masonry Walls: Coordinate installation of frames to allow for solidly filling space between frames and masonry with grout.

- Concrete Walls: Solidly fill space between frames and concrete with grout. Take precautions, including bracing frames, to ensure that frames are not deformed or damaged by grout forces.
- Ceiling Struts: Extend struts vertically from top of frame at each jamb to overhead structural supports or substrates above frame unless frame is anchored to masonry or to other structural support at each jamb. Bend top of struts to provide flush contact for securing to supporting construction. Provide adjustable wedged or bolted anchorage to frame jamb members.

- Installation Tolerances: Adjust hollow metal door frames for squareness, alignment, 7. twist, and plumb to the following tolerances:
 - Squareness: Plus or minus 1/16 inch, measured at door rabbet on a line 90 degrees a. from jamb perpendicular to frame head.
 - Alignment: Plus or minus 1/16 inch, measured at jambs on a horizontal lip b. parallel to plane of wall.
 - Twist: Plus or minus 1/16 inch, measured at opposite face corners of jamos с. parallel lines, and perpendicular to plane of wall.
 - d. Plumbness: Plus or minus 1/16 inch, measured at jambs at floor.
- C. Hollow Metal Doors: Fit hollow metal doors accurately in frames, within clearant Decified below. Shim as necessary.
 - Non-Fire-Rated Standard Steel Doors: 1.
 - Jambs and Head: 1/8 inch plus or minus 1/16 inch. a.
 - Between Edges of Pairs of Doors: 1/8 inch plus or maus 1/16 inch. Between Bottom of Door and Top of Threshold: Maximum 3/8 inch. b.
 - c.
 - Between Bottom of Door and Top of Finit a Floor (No Threshold): Maximum 3/4 d. inch.
 - 2. Fire-Rated Doors: Install doors with clearances according to NFPA 80.
 - Smoke-Control Doors: Install doors according to NFPA 105. 3.
- Glazing: Comply with installation requirements in Division 08 Section "Glazing" and with D. hollow metal manufacturer's writter instructions.
 - 1. Secure stops with coupers in flat- or oval-head machine screws spaced uniformly not and not more than 2 inches o.c. from each corner. more than 9 inches o.c

ADJUSTING AND C 3.4

- Final Adjustment Check and readjust operating hardware items immediately before final A. Renove and replace defective work, including hollow metal work that is warped, inspection r other wise unacceptable. bowed,
- Report grout and other bonding material from hollow metal work immediately after Β. install tion.

sime-Coat Touchup: Immediately after erection, sand smooth rusted or damaged areas of prime coat and apply touchup of compatible air-drying, rust-inhibitive primer.

Metallic-Coated Surfaces: Clean abraded areas and repair with galvanizing repair paint according to manufacturer's written instructions.

END OF SECTION



SPEC SECTION 08 14 16

FLUSH WOOD DOORS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Solid-core doors with wood-veneer faces.
 - 2. Factory finishing flush wood doors.
 - 3. Factory fitting flush wood doors to frames and factor, machining for hardware.

1.3 SUBMITTALS, GENERAL

A. General: Submit all action submittals (except Samples for Verification) required by this Section concurrently.

1.4 ACTION SUBMITTALS

- A. Product Data: For earbeyp of door indicated. Include details of core and edge construction and trim for openings. Include factory-finishing specifications.
- B. LEED Submittel.
 - 1. Certificates for Credit MR 7: Chain-of-custody certificates indicating that flush wood doors comply with forest certification requirements. Include documentation that manufacturer is certified for chain of custody by an FSC-accredited certification body. Include statement indicating cost for each certified wood product.

Product Data for Credit IEQ 4.4: For adhesives, documentation indicating that product contains no urea formaldehyde.

Shop Drawings: Indicate location, size, and hand of each door; elevation of each kind of door; construction details not covered in Product Data; location and extent of hardware blocking; and other pertinent data.

- 1. Indicate dimensions and locations of mortises and holes for hardware.
- 2. Indicate dimensions and locations of cutouts.
- 3. Indicate requirements for veneer matching.
- 4. Indicate doors to be factory finished and finish requirements.

Tetra Tech 200-15704-17001

FLUSH WOOD DOORS 08 14 16 - 1

- 5. Indicate fire-protection ratings for fire-rated doors.
- D. Samples for Initial Selection: For factory-finished doors.
- E. Samples for Verification:
 - 1. Factory finishes applied to actual door face materials, approximately 8 by 10 inches, or each material and finish. For each wood species and transparent finish, provide set of three samples showing typical range of color and grain to be expected in the finished work.
- F. Warranty: Sample of special warranty.

1.5 CLOSEOUT SUBMITTALS

- A. Warranty: Executed special warranty.
- 1.6 QUALITY ASSURANCE
 - A. Manufacturer Qualifications: A qualified manufacturer that is certified for chain of custody by an FSC-accredited certification body.
 - B. Source Limitations: Obtain flush wood doors from single manufacturer.
 - C. Quality Standard: In addition to requirements specified, comply with WDMA I.S.1-A, "Architectural Wood Flush Doors."
 - D. Fire-Rated Wood Doors: Hoors complying with NFPA 80 that are listed and labeled by a qualified testing agency, for fire protection ratings indicated, based on testing at positive pressure according to MFPA 252 or UL 10C.
 - 1. Temperature-Rise cimit: At vertical exit enclosures and exit passageways, provide doors that have a maximum transmitted temperature end point of not more than 450 deg F above and tent after 30 minutes of standard fire-test exposure.
 - E. Preinstallation Conference: Conduct conference at Project site.
- 1.7 DELIVERY, STORAGE, AND HANDLING

Comply with requirements of referenced standard and manufacturer's written instructions.

Package doors individually in plastic bags or cardboard cartons.

Mark each door on bottom rail with opening number used on Shop Drawings.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

1.8 PROJECT CONDITIONS

A. Environmental Limitations: Do not deliver or install doors until spaces are enclosed and weathertight, wet work in spaces is complete and dry, and HVAC system is operating and maintaining ambient temperature and humidity conditions at occupancy levels during the remainder of the construction period.

1.9 WARRANTY

- A. Special Warranty: Manufacturer's standard form in which manufacturer agrees to repuir replace doors that fail in materials or workmanship within specified warranty period
 - 1. Failures include, but are not limited to, the following:
 - a. Warping (bow, cup, or twist) more than 1/4 inch in a 42-b -84 inch section.
 - b. Telegraphing of core construction in face veneers excitating 0.01 inch in a 3-inch span.
 - 2. Warranty shall also include installation and finishing that may be required due to repair or replacement of defective doors.
 - 3. Warranty Period for Solid-Core Interior Dorne: Life of installation.

PART 2 - PRODUCTS

2.1 MANUFACTURERS

- A. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to, the following:
 - 1. Algoma Hardwords,
 - 2. Eggers Industries.
 - 3. Marshfield Loor Systems, Inc.
 - 4. Oshkosh, architectural Door Company.
- 2.2 DCOR CONSTRUCTION, GENERAL

Certified Wood: Fabricate doors with not less than 70 percent of wood products produced from wood obtained from forests certified by an FSC-accredited certification body to comply with FSC STD-01-001, "FSC Principles and Criteria for Forest Stewardship."

Low-Emitting Materials: Fabricate doors with adhesives that do not contain urea formaldehyde.

- C. WDMA I.S.1-A Performance Grade: Extra Heavy Duty.
- D. Structural-Composite-Lumber-Core Doors:
 - 1. Structural Composite Lumber: WDMA I.S.10.

Tetra Tech 200-15704-17001

- a. Screw Withdrawal, Face: 700 lbf.
- b. Screw Withdrawal, Edge: 400 lbf.
- E. Fire-Protection-Rated Doors: Provide core specified or mineral core as needed to provide fireprotection rating indicated.
 - 1. Edge Construction: Provide edge construction with intumescent seals concealed by outer stile. Comply with specified requirements for exposed edges.
 - 2. Pairs: Provide fire-retardant stiles that are listed and labeled for applications indicated without formed-steel edges and astragals. Provide stiles with concealed intume cont seals. Comply with specified requirements for exposed edges.
- F. Mineral-Core Doors:
 - 1. Core: Noncombustible mineral product complying with requirements of referenced quality standard and testing and inspecting agency for fire-protection rating indicated.
 - 2. Blocking: Provide composite blocking with improved screw bolding capability approved for use in doors of fire-protection ratings indicated as needed to eliminate through-bolting hardware.
 - 3. Edge Construction: At hinge stiles, provide laminated-edge construction with improved screw-holding capability and split resistance. Couply with specified requirements for exposed edges.

2.3 VENEERED-FACED DOORS FOR TRANSPARENT FINISH

- A. Interior Solid-Core Doors:
 - 1. Grade: Premium, with Grade A faces.
 - 2. Species: Red oak.
 - 3. Cut: Rotary cut.
 - 4. Match between Verleer Leaves: Book match.
 - 5. Assembly of Veneer Leaves on Door Faces: Balance match.
 - 6. Pair and but Match: Provide for doors hung in same opening or separated only by mullions.
 - 7. Expected Vertical and Top Edges: Same species as faces.
 - 8. Cire: Ether glued wood stave or structural composite lumber.
 - 9. Construction: Five plies. Stiles and rails are bonded to core, then entire unit abrasive planed before veneering. Faces are bonded to core using a hot press.

NGHT FRAMES

Wood Beads for Light Openings in Wood Doors: Provide manufacturer's standard wood beads as follows unless otherwise indicated.

- 1. Wood Species: Same species as door faces.
- 2. Profile: Lipped beads, overlapping door face veneer.

B. Metal Frames for Light Openings in Fire-Rated Doors: Manufacturer's standard frame formed of 0.048-inch-thick, cold-rolled steel sheet; with baked-enamel- or powder-coated finish; and approved for use in doors of fire-protection rating indicated.

2.5 FABRICATION

- A. Factory fit doors to suit frame-opening sizes indicated. Comply with clearance requirements of referenced quality standard for fitting unless otherwise indicated.
 - 1. Comply with requirements in NFPA 80 for fire-rated doors.
- B. Factory machine doors for hardware that is not surface applied. Locate hardware to comply with DHI-WDHS-3. Comply with final hardware schedules, door fame shop Drawings, DHI A115-W series standards, and hardware templates.
 - 1. Coordinate with hardware mortises in metal frames to verify almensions and alignment before factory machining.
- C. Openings: Cut and trim openings through doors in factory.
 - 1. Light Openings: Trim openings with moldings of material and profile indicated.
 - 2. Glazing: Factory install glazing in doors indicated to be factory finished. Comply with applicable requirements in Division 08 Section "Glazing."

2.6 FACTORY FINISHING

- A. General: Comply with reference quality standard for factory finishing. Complete fabrication, including fitting doors for opening and machining for hardware that is not surface applied, before finishing.
 - 1. Finish faces, and our edges, edges of cutouts, and mortises. Stains and fillers may be omitted or bottom ages, edges of cutouts, and mortises.
- B. Finish doors at lectory.
- C. Use only paints and coatings that comply with the testing and product requirements of the California Department of Health Services' "Standard Practice for the Testing of Volatile Organic Emissions from Various Sources Using Small-Scale Environmental Chambers."

ransparent Finish:

- Grade: Premium.
- 2. Finish: WDMA TR-6 catalyzed polyurethane.
- 3. Staining: As selected by Architect from manufacturer's full range.
- 4. Effect: Semifilled finish, produced by applying an additional finish coat to partially fill the wood pores.
- 5. Sheen: Semigloss.

1.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine doors and installed door frames before hanging doors.
 - 1. Verify that frames comply with indicated requirements for type, size, location, and wing characteristics and have been installed with level heads and plumb jambs.
 - 2. Reject doors with defects.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.
- C. Beginning installation constitutes Contractor's acceptance of substrates and conditions

3.2 INSTALLATION

- A. Hardware: For installation, see Division 08 Section "Door Hardware".
- B. Installation Instructions: Install doors to comply with reanufacturer's written instructions and the referenced quality standard, and as indicated.
 - 1. Install fire-rated doors in corresponding file-rate frames according to NFPA 80.
- C. Factory-Fitted Doors: Align in frames for builtonn clearance at each edge.
- D. Factory-Finished Doors: Restore unish before installation if fitting or machining is required at Project site.

3.3 ADJUSTING

- A. Operation: Rehang or replace doors that do not swing or operate freely.
- B. Finished Doorse Replace doors that are damaged or that do not comply with requirements. Doors may be repaired or refinished if work complies with requirements and shows no evidence of repair or refinishing.



SPEC SECTION 08 16 13

FIBERGLASS DOORS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General and Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Exterior and interior manual-swing fiberglass doors and a minum door-frame units.
- 1.3 DEFINITIONS
 - A. ADA/ABA Accessibility Guidelines: U.S. Architectural & Transportation Barriers Compliance Board's "Americans with Disability Act, (ADA) and Architectural Barriers Act (ABA) Accessibility Guidelines for Buildings and Facilities."

1.4 PERFORMANCE REQUIREMENTS

- A. General Performance: communi-framed fiberglass door systems shall withstand the effects of the following performance requirements without exceeding performance criteria or failure due to defective manufacture foorication, installation, or other defects in construction:
 - 1. Movements of supporting structure indicated on Drawings including, but not limited to, detection from uniformly distributed and concentrated live loads.
 - 2. Dimensional tolerances of building frame and other adjacent construction.
 - E. Failure includes the following:
 - Deflection exceeding specified limits.
 - b. Thermal stresses transferring to building structure.
 - c. Framing members transferring stresses, including those caused by thermal and structural movements to glazing.
 - d. Noise or vibration created by wind and by thermal and structural movements.
 - e. Loosening or weakening of fasteners, attachments, and other components.
 - f. Sealant failure.
 - g. Failure of operating units.
- B. Delegated Design: Design aluminum-framed fiberglass door systems, including comprehensive engineering analysis by a qualified professional engineer, using performance requirements and design criteria indicated.

Tetra Tech 200-15704-17001

FIBERGLASS DOORS 08 16 13 - 1

- C. Structural Loads:
 - 1. Wind Loads: As indicated on Drawings.
- D. **Deflection of Framing Members:**
 - Deflection Normal to Wall Plane: Limited to edge of glass in a direction perpendiculation 1. glass plane shall not exceed L/175 of the glass edge length for each individual glazing lite or an amount that restricts edge deflection of individual glazing lites to 3 1ch whichever is less.
 - 2. Deflection Parallel to Glazing Plane: Limited to L/360 of clear span inch. whichever is smaller.
- Structural-Test Performance: Provide aluminum-framed fiberglass do E. tems tested according to ASTM E 330 as follows:
 - When tested at positive and negative wind-load design 1. essures, systems do not evidence deflection exceeding specified limits.
 - When tested at 150 percent of positive and negative wind-load design pressures, systems, 2. including anchorage, do not evidence material failure, structural distress, and permanent deformation of main framing members exceeding on percent of span. Test Durations: As required by design wind velocity, out not fewer than 10 seconds.
 - 3.
- Air Infiltration: Provide aluminum-framed filter lass door systems with maximum air leakage F. through fixed glazing and framing areas for cfm/sq. ft. of fixed wall area when tested according to ASTM E 283 at a minimum static-air-pressure difference of 6.24 lbf/sq. ft.
- Water Penetration under Static Pressure Provide aluminum-framed fiberglass door systems G. that do not evidence water price at prthrough fixed glazing and framing areas when tested according to ASTM E 331 are minimum static-air-pressure difference of 20 percent of positive wind-load design pressure, but not less than 6.24 lbf/sq. ft.
- ovide aluminum-framed fiberglass door systems that allow for thermal H. Thermal Movements: movements resulting from the following maximum change (range) in ambient and surface temperatures. Base engineering calculation on surface temperatures of materials due to both n and nighttime-sky heat loss. solar heat
 - Te. pper dure Change (Range): 120 deg F, ambient: 180 deg F, material surfaces. 1
 - Thermal Conductance: Provide aluminum-framed fiberglass door systems with fixed glazing and framing areas having an average U-factor of not more than 0.70 Btu/sq. ft. x h x deg F when texted according to AAMA 1503.

SUBMITTALS, GENERAL

General: Submit all action submittals (except Samples for Verification) and informational submittals required by this Section concurrently.

1.6 ACTION SUBMITTALS

- A. Product Data: For each type of product indicated. Include construction details, material descriptions, dimensions of individual components and profiles, and finishes for aluminum-framed fiberglass door systems.
- B. LEED Submittals:
 - 1. Laboratory Test Reports for Credit IEQ 4: For adhesives and sealants used incide the weatherproofing system, documentation indicating that products comply with the testing and product requirements of the California Department of Health Services' Standard Practice for the Testing of Volatile Organic Emissions from Various Searce. Using Small-Scale Environmental Chambers."
- C. Shop Drawings: For aluminum-framed fiberglass door systems. In one plans, elevations, sections, details, and attachments to other work.
 - 1. Include details of provisions for system expansion and contraction and for drainage of moisture in the system to the exterior.
- D. Samples for Initial Selection: For units with color finishe
- E. Samples for Verification: For each type of expressed mish required, in manufacturer's standard sizes.
- F. Delegated-Design Submittal: For alternative-framed fiberglass door systems indicated to comply with performance requirements and design criteria, including analysis data signed and sealed by the qualified professional engineer responsible for their preparation.
 - 1. Detail fabrication and ssembly of aluminum-framed fiberglass door systems.
 - 2. Include design calculations
- G. Warranties: Sample of Decid warranties.
- 1.7 INFORMATIO. AL SUBMITTALS
 - A. Qualification Data: For qualified Installer.
 - B. Product Test Reports: Based on evaluation of comprehensive tests performed by a qualified testing agency, for aluminum-framed fiberglass door systems, indicating compliance with performance requirements.

CLOSEOUT SUBMITTALS

- A. Maintenance Data: For aluminum-framed fiberglass door systems to include in maintenance manuals.
- B. Warranties: Executed special warranties.

Tetra Tech 200-15704-17001

ADDITION AND RENOVATIONS

1.9 QUALITY ASSURANCE

- A. Installer Qualifications: Manufacturer's authorized representative who is trained and approved for installation of units required for this Project.
- B. Engineering Responsibility: Prepare data for aluminum-framed fiberglass door systems including Shop Drawings, based on testing and engineering analysis of manufacturer's standard units in systems similar to those indicated for this Project.
- C. Product Options: Information on Drawings and in Specifications establishes requirement for systems' aesthetic effects and performance characteristics. Aesthetic effects are scalcated by dimensions, arrangements, alignment, and profiles of components and assembles as they relate to sightlines, to one another, and to adjoining construction. Performance characteristics are indicated by criteria subject to verification by one or more methods including preconstruction testing, field testing, and in-service performance.
 - 1. Do not revise intended aesthetic effects, as judged solely by Architect, except with Architect's approval. If revisions are proposed, submit comprehensive explanatory data to Architect for review.
- D. Accessible Entrances: Comply with applicable provisions in the U.S. Architectural & Transportation Barriers Compliance Board's TDA-ABA Accessibility Guidelines and ICC/ANSI A117.1.
- E. Source Limitations for Aluminum-Framed Fiber lass Door Systems: Obtain from single source from single manufacturer.
- F. Fire-Test-Response Characteristics. Provide fiberglass door FRP panels with finishes meeting the following as determined by testing identical products by UL or another testing and inspecting agency acceptable to authorities having jurisdiction:
 - 1. Surface-Burning Characteristics: As determined by testing per ASTM E 84.
 - a. Flam-Spread Index: 25 or less.
 - b. Smol A Developed Index: 450 or less.
- G. Preinstal ation Conference: Conduct conference at Project site.
 - . Review and discuss the finishing of aluminum framing that is required to be coordinated with the finishing of other aluminum work for color and finish matching.

PROJECT CONDITIONS

Field Measurements: Verify actual locations of structural supports for aluminum-framed fiberglass door systems by field measurements before fabrication and indicate measurements on Shop Drawings.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

1.11 WARRANTY

- Special Warranty: Manufacturer's standard form in which manufacturer agrees to repair or A. replace components of aluminum-framed fiberglass door systems that do not comply with requirements or that fail in materials or workmanship within specified warranty period.
 - 1. Failures include, but are not limited to, the following:
 - Structural failures including, but not limited to, excessive deflection. a.
 - Noise or vibration caused by thermal movements. b.
 - Deterioration of metals, metal finishes, and other materials bey c. weathering.
 - Adhesive or cohesive sealant failures. d.
 - Water leakage through fixed glazing and framing areas. e.
 - f. Failure of operating components.
 - Warranty Period: 10 years from date of Substantial Compl 2.

PART 2 - PRODUCTS

2.1 MATERIALS

- Aluminum: Alloy and temper recommended by manufacturer for type of use and finish A. indicated.
 - 1. Sheet and Plate: ASTM B 29.
 - Extruded Bars, Rods, Profiles, and Tubes: ASTM B Extruded Structural Pipe and Tubes: ASTM B 429. 2. und Tubes: ASTM B 221.
 - 3.
 - Structural Profiles: NSTAL 308/B 308M. 4.
 - Welding Rods and Bare Electrodes: AWS A5.10/A5.10M. 5.
- Steel Reinforcement: Manufacturer's standard zinc-rich, corrosion-resistant primer, complying B. with SSPC S Chide No. 12.00; applied immediately after surface preparation and select surface preparation methods according to recommendations in SSPCpretreatment SP COM and prepare surfaces according to applicable SSPC standard.
 - Tructural Shapes, Plates, and Bars: ASTM A 36/A 36M.
 - Cold-Rolled Sheet and Strip: ASTM A 1008/A 1008M.
 - Hot-Rolled Sheet and Strip: ASTM A 1011/A 1011M.

FRAMING SYSTEMS

Framing Members: Manufacturer's standard extruded-aluminum framing members of thickness required and reinforced as required to support imposed loads.

- 1. Construction: Nonthermal.
- 2. Glazing System: Retained mechanically with gaskets on four sides.
- Glazing Plane: As indicated. 3.

Tetra Tech 200-15704-17001

- 4. Framing Design: 2-inch wide by 6-inch deep, with minimum 0.125-inch-thick, extrudedaluminum members.
- 5. Basis-of Design Product: Subject to compliance with requirements, provide Commercial Door Systems; Model CDS 2600 Series Framing, or comparable product.
- B. Brackets and Reinforcements: Manufacturer's standard high-strength aluminum with nonstaining, nonferrous shims for aligning system components.
- C. Fasteners and Accessories: Manufacturer's standard corrosion-resistant, non nonbleeding fasteners and accessories compatible with adjacent materials.
 - 1. Use self-locking devices where fasteners are subject to loosening or turning but from thermal and structural movements, wind loads, or vibration.
 - 2. Reinforce members as required to receive fastener threads.
 - 3. Use exposed fasteners with countersunk Phillips screw heads, finished to match framing system.
- D. Concrete and Masonry Inserts: Hot-dip galvanized cast-iron, malleable-iron, or steel inserts, complying with ASTM A 123/A 123M or ASTM A 153/A 153M.
- E. Concealed Flashing: Manufacturer's standard corroside-resistant, nonstaining, nonbleeding flashing compatible with adjacent materials.
- F. Framing System Gaskets and Sealants: Manufacturer's standard, recommended by manufacturer for joint type.
 - 1. Sealants used inside the wear property system shall comply with the testing and product requirements of the California Department of Health Services' "Standard Practice for the Testing of Volatile Organic Emissions from Various Sources Using Small-Scale Environmental Chambers."
- 2.3 GLAZING SYSTEMS
 - A. Glazing: As exercised in Division 08 Section "Glazing."
 - B. Glazing Gaskey: Manufacturer's standard compression types; replaceable, molded or extruded, of profile and bardness required to maintain watertight seal.
 - C. Spaces and Setting Blocks: Manufacturer's standard elastomeric type.

FIBERGLASS DOOR SYSTEMS

Fiberglass Doors: Manufacturer's standard glazed fiberglass doors for manual-swing operation.

- 1. Door Construction: 1-3/4-inchoverall thickness, with 0.120-inch-thick pebble-textured, Class A fire-test-response characteristics, fiberglass reinforced polyester (FRP) face sheet interlocked into extruded-aluminum rail and stile members. Mechanically fasten with concealed tie rods. Cavity filled with polystyrene core.
- 2. Door Design: As indicated.

FIBERGLASS DOORS 08 16 13 - 6

- a. Basis-of-Design Products: Subject to compliance with requirements, provide Commercial Door Systems; CDS F500HD Heavy Duty Wide Stile (FRP) Doors, or comparable product.
- b. Accessible Doors: Smooth surfaced for width of door in area within 10 inchesabove floor or ground plane.
- 3. Glazing Stops and Gaskets: Extruded-aluminum stops and preformed gaskets.
- B. Fiberglass Door Hardware: As specified in Division 08 Section "Door Hardware."

2.5 FIBERGLASS DOOR HARDWARE

- A. General: Provide fiberglass door hardware and fiberglass door hardware sets indicated in door and frame schedule for each entrance door to comply with requirements in this Section.
- B. Weather Stripping: Manufacturer's standard replaceable component
 - 1. Compression Type: Made of ASTM D 2000, molecul neoprene, or ASTM D 2287, molded PVC.
 - 2. Sliding Type: AAMA 701, made of wool, polytopylene, or nylon woven pile with nylon-fabric or aluminum-strip backing.
- C. Weather Sweeps: Manufacturer's standard exern r-door bottom sweep with concealed fasteners on mounting strip.
- D. Silencers: BHMA A156.16, Grade
- E. Recessed Pull Handle: Manufacturers standard recessed pull handle complying with applicable accessibility requirements.

2.6 ACCESSORY MATERIA

- A. Joint Sealants. For installation at perimeter of aluminum-framed fiberglass door systems, as specified in Div tion 07 Section "Joint Sealants."
 - 1. Se lants used inside the weatherproofing system shall comply with the testing and product requirements of the California Department of Health Services' "Standard Practice for the Testing of Volatile Organic Emissions from Various Sources Using Small-Scale Environmental Chambers."



Bituminous Paint: Cold-applied, asphalt-mastic paint complying with SSPC-Paint 12 requirements except containing no asbestos; formulated for 30-mil thickness per coat.

FABRICATION

A. Form or extrude aluminum shapes before finishing.

Tetra Tech 200-15704-17001

- B. Weld in concealed locations to greatest extent possible to minimize distortion or discoloration of finish. Remove weld spatter and welding oxides from exposed surfaces by descaling or grinding.
- C. Framing Members, General: Fabricate components that, when assembled, have the following characteristics:
 - 1. Profiles that are sharp, straight, and free of defects or deformations.
 - Accurately fitted joints with ends coped or mitered. 2.
 - 3. Means to drain water passing joints, condensation within framing members, nd mo migrating within the system to exterior.
 - 4. Physical and thermal isolation of glazing from framing members.
 - 5. Accommodations for thermal and mechanical movements of glazin and framing to maintain required glazing edge clearances.
 - Provisions for field replacement of glazing. 6.
 - Fasteners, anchors, and connection devices that are cong ale 7. from view to greatest extent possible.
- D. Mechanically Glazed Framing Members: Fabricate for gla ing with projecting stops.
- E. Door Frames: Reinforce as required to support loads mposed by door operation and for installing door hardware.
 - 1. At exterior doors, provide compression ve her stripping at fixed stops.
 - At interior doors, provide silence 2. ch ps. Install three silencers on strike jamb of single-door frames and two silencers on head of frames for pairs of doors.
- Doors: Reinforce doors as required for installing door hardware. F.
 - At pairs of exterior doors, plovide sliding-type weather stripping retained in adjustable 1. strip and mortised interdorredge. At exterior doors, provide weather sweeps applied to door bottoms.
 - 2.
- Door Hardware installation: Factory install door hardware to the greatest extent possible. Cut, G. drill, and tap for factory-installed entrance door hardware before applying finishes.
- H. After farrication, clearly mark components to identify their locations in Project according to Shop Dra ving

ALUMINUM FINISHES



High-Performance Organic Finish: 2-coat fluoropolymer finish complying with AAMA 2605 and containing not less than 70 percent PVDF resin by weight in color coat. Prepare, pretreat, and apply coating to exposed metal surfaces to comply with coating and resin manufacturers' written instructions.

1. Color and Gloss: As selected by Architect from manufacturer's full range.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine areas and conditions, with Installer present, for compliance with requirements for installation tolerances and other conditions affecting performance of the Work.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.
- C. Beginning installation constitutes Contractor's acceptance of substrates and condition

3.2 INSTALLATION

- A. General:
 - 1. Comply with manufacturer's written instructions.
 - 2. Do not install damaged components.
 - 3. Fit joints to produce hairline joints free of burrs and distortion
 - 4. Rigidly secure nonmovement joints.
 - 5. Install anchors with separators and isolators to prevent metal corrosion and electrolytic deterioration.
 - 6. Seal joints watertight unless otherwise indicated
- B. Metal Protection:
 - 1. Where aluminum will contact desimilar metals, protect against galvanic action by painting contact surfaces with refiner or applying sealant or tape, or by installing nonconductive spacers as recommended by manufacturer for this purpose.
 - 2. Where aluminum whi contact concrete or masonry, protect against corrosion by painting contact surfaces with bluminous paint.
- C. Install components to cann water passing joints, condensation occurring within framing members, and moj ture migrating within the system to exterior.
- D. Set continuous silf members and flashing in full sealant bed as specified in Division 07 Section "Joint Sealants" to produce weathertight installation.
- E. Instal components plumb and true in alignment with established lines and grades, and without warp or rack.

Install glazing as specified in Division 08 Section "Glazing."

Doors: Install doors to produce smooth operation and tight fit at contact points.

- 1. Exterior Doors: Install to produce weathertight enclosure and tight fit at weather stripping.
- 2. Field-Installed Door Hardware: Install surface-mounted door hardware according to door hardware manufacturers' written instructions using concealed fasteners to greatest extent possible.

H. Install perimeter joint sealants as specified in Division 07 Section "Joint Sealants" to produce weathertight installation.

3.3 ERECTION TOLERANCES

- A. Install aluminum-framed fiberglass door systems to comply with the following maximum erection tolerances:
 - 1. Location and Plane: Limit variation from true location and plane to 1/8 inclume 1/4 inch over total length.
 - 2. Alignment:
 - a. Where surfaces abut in line, limit offset from true alignment of Ninch.
 - b. Where surfaces meet at corners, limit offset from true alignment to 1/32 inch.
- B. Diagonal Measurements: Limit difference between diagonal measurements to 1/8 inch.

3.4 ADJUSTING

A. Adjust operating door hardware to function smoothly as recommended by manufacturer.



SPEC SECTION 08 31 13

ACCESS DOORS AND FRAMES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Access doors and frames for walls and ceilings.
 - 2. Floor access doors and frames.

1.3 SUBMITTALS, GENERAL

A. General: Submit all submittals required by this section concurrently.

1.4 ACTION SUBMITTALS

- A. Product Data: For each type of product.
 - 1. Include construction letails, fire ratings, materials, individual components and profiles, and finishes.
- B. Shop Drawing
 - 1. Include plans, elevations, sections, details, and attachments to other work.
 - 2. Detail fabrication and installation of access doors and frames for each type of substrate.
 - Produce Schedule: Provide complete access door and frame schedule, including types, locations, sizes, latching provisions, and other data pertinent to installation.

PRODUCTS

PERFORMANCE REQUIREMENTS

A. Fire-Rated Access Doors and Frames: Units complying with NFPA 80 that are identical to access door and frame assemblies tested for fire-test-response characteristics according to the

Tetra Tech 200-15704-17001

following test method and that are listed and labeled by UL or another testing and inspecting agency acceptable to authorities having jurisdiction:

1. NFPA 252 or UL 10B for fire-rated access door assemblies installed vertically.

2.2 ACCESS DOORS AND FRAMES FOR WALLS AND CEILINGS

- A. Basis-of-Design Product: Subject to compliance with requirements, provide product indice or comparable product.
- B. Source Limitations: Obtain each type of access door and frame from single source from single manufacturer.
- C. Flush Access Doors with Concealed Flanges:
 - 1. Basis-of-Design Product: Karp Associates, Inc.; KDW.
 - 2. Assembly Description: Fabricate door to fit flush to frame. Provide frame with gypsum board beads for concealed flange installation.
 - 3. Locations: Wall and ceiling.
 - 4. Door Size: 14 inches by 14 inches, unless noted otherwise.
 - 5. Uncoated Steel Sheet for Door: Nominal 0.075 inch, 14 gage.
 - a. Finish: Factory prime.
 - 6. Frame Material: Same material and brish as door; nominal 0.060 inch, 16 gage.
 - 7. Hinges: Spring-loaded, concerned-printype or continuous piano.
 - 8. Hardware: Cam latch operated by screwdriver with interior release.
- D. Fire-Rated, Flush Access Doors with Exposed Flanges:
 - 1. Basis-of-Design Produce Karp Associates, Inc.; KRP-150FR.
 - 2. Assembly Description Fabricate door to fit flush to frame, with a core of mineral-fiber insulation inclosed in sheet metal. Provide self-latching door with automatic closer and interior latch release. Provide manufacturer's standard-width exposed flange, properties al to door size.
 - 3. Location: Wall.
 - 4. Dour Size: 14 inches by 14 inches, unless noted otherwise.
 - 5. Vire-Resistance Rating: Not less than 1-1/2 hours.
 - 5. Uncoated Steel Sheet for Door: Nominal 0.036 inch, 20 gage.
 - a. Finish: Factory prime.
 - Frame Material: Same material and finish as door; nominal 0.060 inch, 16 gage.
 - 8. Hinges: Continuous piano.
 - 9. Hardware: Self-latching bolt operated by ring turn with interior release.

2.3 FLOOR ACCESS DOORS AND FRAMES

A. Basis-of-Design Product: Subject to compliance with requirements, provide product indicated or comparable product.

ACCESS DOORS AND FRAMES 08 31 13 - 2

7

- B. Floor Doors, General: Equip each door with adjustable counterbalancing springs, heavy-duty hold-open arm that automatically locks door open at 90 degrees, release handle with red vinyl grip that allows for one-handed closure, and recessed lift handle.
- C. Aluminum Floor Door: Single-leaf opening. Extruded-aluminum angle frame with 1/4-inchthick, diamond-pattern, aluminum tread plate door; nonwatertight; loading capacity to support 300-lbf/sq. ft. pedestrian live load.
 - 1. Basis-of-Design Product: Karp Associates, Inc.; KFD-SD.
 - 2. Door Size: 36 inches by 30 inches, unless noted otherwise.
 - 3. Hinges: Heavy-duty, stainless-steel butt hinges with stainless-steel pins.
 - 4. Latch: Stainless-steel slam latch.
 - 5. Hardware Material: Stainless steel, including latch and lifting mechanism assemblies, hold-open arms, and all brackets, hinges, pins, and fasteners.

2.4 MATERIALS

- A. Steel Sheet: Uncoated or electrolytic zinc coated, ASTM A 879/A 779M, with cold-rolled steel sheet substrate complying with ASTM A 1008/A 1008M, Commercial Steel (CS), exposed.
- B. Aluminum Extrusions: ASTM B 221, Alloy 6063 To.
- C. Aluminum-Alloy Rolled Tread Plate: ASTM R 632/B 632M, Alloy 6061-T6.
- D. Aluminum Sheet: ASTM B 209, alloy and temper recommended by aluminum producer and finisher for type of use and finish milicated, and with not less than strength and durability properties of Alloy 5005-H15; with minimum sheet thickness according to ANSI H35.2.
- E. Frame Anchors: Same type & door face.
- F. Inserts, Bolts, and Ancher Fasteners: Hot-dip galvanized steel according to ASTM A 153/A 153Mer ASTM F 2329.
- 2.5 FABRICATION
 - A. General: Prov de access door and frame assemblies manufactured as integral units ready for installation.

Metal Surfaces: For metal surfaces exposed to view in the completed Work, provide materials with smooth, flat surfaces without blemishes. Do not use materials with exposed pitting, seam marks, roller marks, rolled trade names, or roughness.

Doors and Frames: Grind exposed welds smooth and flush with adjacent surfaces. Furnish attachment devices and fasteners of type required to secure access doors to types of supports indicated.

- 1. Provide mounting holes in frames for attachment of units to metal framing.
- 2. Provide mounting holes in frame for attachment of masonry anchors.

- D. Latching Mechanisms: Furnish number required to hold doors in flush, smooth plane when closed.
- E. Extruded Aluminum: After fabrication, apply manufacturer's standard protective coating on aluminum that will come in contact with concrete.

2.6 FINISHES

- A. Comply with NAAMM's "Metal Finishes Manual for Architectural and Metal Proceeding for applying and designating finishes.
- B. Appearance of Finished Work: Noticeable variations in same piece are not acceptal
- C. Steel and Metallic-Coated-Steel Finishes:
 - 1. Factory Prime: Apply manufacturer's standard, fast-curing read, and chromate-free, universal primer immediately after surface preparation and pretreatment.
- D. Aluminum Finishes:
 - 1. Mill finish.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine substrates for compliance with requirements for installation tolerances and other conditions affecting performance of the Work.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.
- C. Beginning installation constitutes Contractor's acceptance of substrates and conditions.
- 3.2 INSTALLATION
 - Comply with manufacturer's written instructions for installing access doors and frames.
 - Install doors flush with adjacent finish surfaces or recessed to receive finish material.

ADJUSTING

- A. Adjust doors and hardware, after installation, for proper operation.
- B. Remove and replace doors and frames that are warped, bowed, or otherwise damaged.

END OF SECTION

SPEC SECTION 08 41 13

ALUMINUM-FRAMED ENTRANCES AND STOREFRONTS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section Includes:
 - 1. Exterior and interior storefront framing.
 - 2. Storefront framing for punched openings.
 - 3. Exterior and interior manual-swing entrance doors and door-frame units.

1.3 DEFINITIONS

A. ADA/ABA Accessibility Guidelines: U.S. Alchitectural & Transportation Barriers Compliance Board's "Americans with Disability Act (ADA) and Architectural Barriers Act (ABA) Accessibility Guidelines for Buildings and Facilities."

1.4 PERFORMANCE RECORDINENTS

- A. General Performance: A bininum-framed systems shall withstand the effects of the following performance requirements without exceeding performance criteria or failure due to defective manufacture, the action, installation, or other defects in construction:
 - 1. Movements of supporting structure indicated on Drawings including, but not limited to, deflection from uniformly distributed and concentrated live loads.

Dimensional tolerances of building frame and other adjacent construction. Failure includes the following:

- a. Deflection exceeding specified limits.
- b. Thermal stresses transferring to building structure.
- c. Framing members transferring stresses, including those caused by thermal and structural movements to glazing.
- d. Noise or vibration created by wind and by thermal and structural movements.
- e. Loosening or weakening of fasteners, attachments, and other components.
- f. Sealant failure.
- g. Failure of operating units.

- B. Delegated Design: Design aluminum-framed systems, including comprehensive engineering analysis by a qualified professional engineer, using performance requirements and design criteria indicated.
- C. Structural Loads:
 - 1. Wind Loads: As indicated on Drawings.
- D. Deflection of Framing Members:
 - 1. Deflection Normal to Wall Plane: Limited to edge of glass in a direction perpendicular to glass plane shall not exceed L/175 of the glass edge length for each individual glazing lite or an amount that restricts edge deflection of individual glazing lites to 3/4 inch, whichever is less.
 - 2. Deflection Parallel to Glazing Plane: Limited to L/360 of char span or 1/8 inch, whichever is smaller.
- E. Structural-Test Performance: Provide aluminum-framed systems tested according to ASTM E 330 as follows:
 - 1. When tested at positive and negative wind-load design pressures, systems do not evidence deflection exceeding specified limits
 - 2. When tested at 150 percent of positive and negative wind-load design pressures, systems, including anchorage, do not evidence note ial fullures, structural distress, and permanent deformation of main framing members exceeding 0.2 percent of span.
 - 3. Test Durations: As required by devige wind velocity, but not fewer than 10 seconds.
- F. Air Infiltration: Provide aluminum-framed systems with maximum air leakage through fixed glazing and framing areas of 0.36 fra/sq. ft. of fixed wall area when tested according to ASTM E 283 at a minimum static-ar-pressure difference of 6.24 lbf/sq. ft.
- G. Water Penetration under Staic Pressure: Provide aluminum-framed systems that do not evidence water penetration brough fixed glazing and framing areas when tested according to ASTM E 331 at a minimum static-air-pressure difference of 20 percent of positive wind-load design pressure, but not less than 6.24 lbf/sq. ft.
- H. Thermal Movements: Provide aluminum-framed systems that allow for thermal movements resulting from the following maximum change (range) in ambient and surface temperatures. Basic engineering calculation on surface temperatures of materials due to both solar heat gain and nighttime-sky heat loss.



Temperature Change (Range): 120 deg F, ambient; 180 deg F, material surfaces.

Condensation Resistance: Provide aluminum-framed systems with fixed glazing and framing areas having condensation-resistance factor (CRF) of not less than 54 when tested according to AAMA 1503.

J. Thermal Conductance: Provide aluminum-framed systems with fixed glazing and framing areas having an average U-factor of not more than 0.47 Btu/sq. ft. x h x deg F when tested according to AAMA 1503.

Tetra Tech 200-15704-17001

1.5 SUBMITTALS, GENERAL

A. General: Submit all action submittals (except Samples for Verification) and informational submittals required by this Section concurrently.

1.6 ACTION SUBMITTALS

- A. Product Data: For each type of product indicated. Include construction details, wateria descriptions, dimensions of individual components and profiles, and finishes for animous framed systems.
- B. Shop Drawings: For aluminum-framed systems. Include plans, elevations sections, details, and attachments to other work.
 - 1. Include details of provisions for system expansion and coparation and for drainage of moisture in the system to the exterior.
- C. Samples for Initial Selection: For units with factory-applied color fulsibles.
- D. Samples for Verification: For each type of exposed finite required, in manufacturer's standard sizes.
- E. Delegated-Design Submittal: For aluminum framed systems indicated to comply with performance requirements and design criteria. Including analysis data signed and sealed by the qualified professional engineer responsible for their preparation.
 - 1. Detail fabrication and assenally of aluminum-framed systems.
 - 2. Include design calculations

1.7 INFORMATIONAL SOBMITTALS

- A. Qualification Date: For qualified Installer.
- B. Product Test Reports: Based on evaluation of comprehensive tests performed by a qualified testing (genc) for aluminum-framed systems, indicating compliance with performance requirements.

1.8 CLOSEOUT SUBMITTALS

Maintenance Data: For aluminum-framed systems to include in maintenance manuals.

QUALITY ASSURANCE

A. Installer Qualifications: Manufacturer's authorized representative who is trained and approved for installation of units required for this Project.

- B. Engineering Responsibility: Prepare data for aluminum-framed systems, including Shop Drawings, based on testing and engineering analysis of manufacturer's standard units in systems similar to those indicated for this Project.
- C. Product Options: Information on Drawings and in Specifications establishes requirements for systems' aesthetic effects and performance characteristics. Aesthetic effects are indicated by dimensions, arrangements, alignment, and profiles of components and assemblies as they relate to sightlines, to one another, and to adjoining construction. Performance characteristics are indicated by criteria subject to verification by one or more methods including preconstruction testing, field testing, and in-service performance.
 - 1. Do not revise intended aesthetic effects, as judged solely by Architect, except with Architect's approval. If revisions are proposed, submit comprehensive explanatory data to Architect for review.
- D. Accessible Entrances: Comply with applicable provisions in the U.S. Architectural & Transportation Barriers Compliance Board's ADA-ABA Accessibility Guidelines and ICC/ANSI A117.1.
- E. Source Limitations for Aluminum-Framed Systems: Obtain from single source from single manufacturer.
- F. Preinstallation Conference: Conduct conference at Robect site.
 - 1. Review and discuss the finishing of clumerum framing that is required to be coordinated with the finishing of other aluminum work for color and finish matching.

1.10 PROJECT CONDITIONS

A. Field Measurements: Verify actual locations of structural supports for aluminum-framed systems by field measurements before fabrication and indicate measurements on Shop Drawings.

PART 2 - PRODUCTS

2.1 MATERIA

Aluminum: Alloy and temper recommended by manufacturer for type of use and finish adicated.

- 1. Sheet and Plate: ASTM B 209.
- 2. Extruded Bars, Rods, Profiles, and Tubes: ASTM B 221.
- 3. Extruded Structural Pipe and Tubes: ASTM B 429.
- 4. Structural Profiles: ASTM B 308/B 308M.
- 5. Welding Rods and Bare Electrodes: AWS A5.10/A5.10M.
- B. Steel Reinforcement: Manufacturer's standard zinc-rich, corrosion-resistant primer, complying with SSPC-PS Guide No. 12.00; applied immediately after surface preparation and

pretreatment. Select surface preparation methods according to recommendations in SSPC-SP COM and prepare surfaces according to applicable SSPC standard.

- 1. Structural Shapes, Plates, and Bars: ASTM A 36/A 36M.
- 2. Cold-Rolled Sheet and Strip: ASTM A 1008/A 1008M.
- 3. Hot-Rolled Sheet and Strip: ASTM A 1011/A 1011M.

2.2 FRAMING SYSTEMS

- A. Framing Members: Manufacturer's standard extruded-aluminum framing members of mickness required and reinforced as required to support imposed loads.
 - 1. Construction: Thermally improved.
 - 2. Glazing System: Retained mechanically with gaskets on four side
 - 3. Glazing Plane: Center.
 - Framing Design: 2-inch wide by 4-1/2-inch deep, with panimum 0.080-inch-thick, extruded-aluminum members. Poured-in-place, polyurethan thermal barrier.
 Basis-of Design Product: Subject to compliance with requirements, provide Kawneer
 - 5. Basis-of Design Product: Subject to compliance with requirements, provide Kawneer North America; an Alcoa company; Trifab VG 517 Storefront System, or comparable product.
- B. Brackets and Reinforcements: Manufacturer's standard high-strength aluminum with nonstaining, nonferrous shims for aligning system components.
- C. Fasteners and Accessories: Manufacturer's standard corrosion-resistant, nonstaining, nonbleeding fasteners and accessories compatible with adjacent materials.
 - 1. Use self-locking devices where fasteners are subject to loosening or turning out from thermal and structural povenents, wind loads, or vibration.
 - 2. Reinforce members a required to receive fastener threads.
 - 3. Use exposed fasterers with countersunk Phillips screw heads, finished to match framing system.
- D. Concrete and Maconry Inserts: Hot-dip galvanized cast-iron, malleable-iron, or steel inserts, complying with VSTM A 123/A 123M or ASTM A 153/A 153M.
- E. Concealed Flaching: Manufacturer's standard corrosion-resistant, nonstaining, nonbleeding flashing compatible with adjacent materials.

Framing System Gaskets and Sealants: Manufacturer's standard, recommended by nanufacturer for joint type.

1. Sealants used inside the weatherproofing system shall comply with the testing and product requirements of the California Department of Health Services' "Standard Practice for the Testing of Volatile Organic Emissions from Various Sources Using Small-Scale Environmental Chambers."

ADDITION AND RENOVATIONS

2.3 GLAZING SYSTEMS

- A. Glazing: As specified in Division 08 Section "Glazing."
- B. Glazing Gaskets: Manufacturer's standard compression types; replaceable, molded or extruded, of profile and hardness required to maintain watertight seal.
- C. Spacers and Setting Blocks: Manufacturer's standard elastomeric type.

2.4 ENTRANCE DOOR SYSTEMS

- A. Entrance Doors: Manufacturer's standard glazed entrance doors for manual-string or eration.
 - 1. Door Construction: 1-3/4-inchoverall thickness, with minimum 0.125-inch-thick, extruded-aluminum tubular rail and stile members. Mechanically fasten corners with reinforcing brackets that are deeply penetrated and filler wences or that incorporate concealed tie rods.
 - 2. Door Design: Wide stile; 5-inchnominal width.
 - a. Basis-of-Design Products: Subject to compliance with requirements, provide Kawneer North America; an Aleba company; 500 Wide Stile Entrances, or comparable product.
 - b. Accessible Doors: Smooth surfaced for width of door in area within 10 inchesabove floor or ground place.
 - 3. Glazing Stops and Gaskets: Square, snap-on, extruded-aluminum stops and preformed gaskets.
- B. Entrance Door Hardware: As specified in Division 08 Section "Door Hardware."

2.5 ENTRANCE DOOR HAPOWARE

- A. General: Provide entrance door hardware and entrance door hardware sets indicated in door and frame schedule for each entrance door to comply with requirements in this Section.
- B. We the Stripping: Manufacturer's standard replaceable components.
- 1.
- Compression Type: Made of ASTM D 2000, molded neoprene, or ASTM D 2287, molded PVC.
- Sliding Type: AAMA 701, made of wool, polypropylene, or nylon woven pile with nylon-fabric or aluminum-strip backing.

Weather Sweeps: Manufacturer's standard exterior-door bottom sweep with concealed fasteners on mounting strip.

D. Silencers: BHMA A156.16, Grade 1.

2.6 ACCESSORY MATERIALS

- Joint Sealants: For installation at perimeter of aluminum-framed systems, as specified in A. Division 07 Section "Joint Sealants."
 - 1. Sealants used inside the weatherproofing system shall comply with the testing and product requirements of the California Department of Health Services' "Standard Practice for the Testing of Volatile Organic Emissions from Various Sources Using Small Scale Environmental Chambers."
- B. Bituminous Paint: Cold-applied, asphalt-mastic paint complying with nt 12 requirements except containing no asbestos; formulated for 30-mil thickness p

2.7 FABRICATION

- Form or extrude aluminum shapes before finishing. A.
- Weld in concealed locations to greatest extent possible to maximize distortion or discoloration B. om exposed surfaces by descaling or of finish. Remove weld spatter and welding oxides, grinding.
- Framing Members, General: Fabricate comportents out, when assembled, have the following C. characteristics:
 - Profiles that are sharp, straight, and free of defects or deformations. 1.
 - Accurately fitted joints with end, coped or mitered. 2.
 - Means to drain water passing joint, condensation within framing members, and moisture 3. migrating within the system to exterior.
 - 4. Physical and thermal holation of glazing from framing members.
 - Accommodations for thermal and mechanical movements of glazing and framing to maintain required grazing edge clearances. Provisions for the relacement of glazing. 5.
 - 6.
 - nchors, and connection devices that are concealed from view to greatest 7. Fasteners extent coss
- D. Mechanically Clazed Framing Members: Fabricate for flush glazing without projecting stops.
- Storefunt Framing: Fabricate components for assembly using shear-block system. E.
 - Entrance Door Frames: Reinforce as required to support loads imposed by door operation and r installing entrance door hardware.
 - At exterior doors, provide compression weather stripping at fixed stops.
 - 2. At interior doors, provide silencers at stops to prevent metal-to-metal contact. Install three silencers on strike jamb of single-door frames and two silencers on head of frames for pairs of doors.
- Entrance Doors: Reinforce doors as required for installing entrance door hardware. G.

1.

- 1. At pairs of exterior doors, provide sliding-type weather stripping retained in adjustable strip and mortised into door edge.
- 2. At exterior doors, provide weather sweeps applied to door bottoms.
- H. Entrance Door Hardware Installation: Factory install entrance door hardware to the greatest extent possible. Cut, drill, and tap for factory-installed entrance door hardware before applying finishes.
- I. After fabrication, clearly mark components to identify their locations in Project according Shop Drawings.

2.8 ALUMINUM FINISHES

- A. High-Performance Organic Finish: 2-coat fluoropolymer finish complying with AAMA 2605 and containing not less than 70 percent PVDF resin by weight in colo coat. Prepare, pretreat, and apply coating to exposed metal surfaces to comply with coating and resin manufacturers' written instructions.
 - 1. Color and Gloss: As selected by Architect from nanulacturer's full range.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine areas and conditions, with Installer present, for compliance with requirements for installation tolerances and other conditions affecting performance of the Work.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.
- C. Beginning installation enstrutes Contractor's acceptance of substrates and conditions.
- 3.2 INSTALLATION
 - A. General:
 - Comply with manufacturer's written instructions.
 - Do not install damaged components.
 - Fit joints to produce hairline joints free of burrs and distortion.
 - Rigidly secure nonmovement joints.
 - Install anchors with separators and isolators to prevent metal corrosion and electrolytic deterioration.
 - 6. Seal joints watertight unless otherwise indicated.
 - B. Metal Protection:
 - 1. Where aluminum will contact dissimilar metals, protect against galvanic action by painting contact surfaces with primer or applying sealant or tape, or by installing nonconductive spacers as recommended by manufacturer for this purpose.
- 2. Where aluminum will contact concrete or masonry, protect against corrosion by painting contact surfaces with bituminous paint.
- C. Install components to drain water passing joints, condensation occurring within framing members, and moisture migrating within the system to exterior.
- D. Set continuous sill members and flashing in full sealant bed as specified in Division 07 Section "Joint Sealants" to produce weathertight installation.
- E. Install components plumb and true in alignment with established lines and grades, and why warp or rack.
- F. Install glazing as specified in Division 08 Section "Glazing."
- G. Entrance Doors: Install doors to produce smooth operation and tight fit a contact points.
 - 1. Exterior Doors: Install to produce weathertight enclosure and tight fit at weather stripping.
 - 2. Field-Installed Entrance Door Hardware: Install surface-mounted entrance door hardware according to entrance door hardware nanu acturers' written instructions using concealed fasteners to greatest extent possible.
- H. Install perimeter joint sealants as specified in Ilivision 07 Section "Joint Sealants" to produce weathertight installation.

3.3 ERECTION TOLERANCES

- A. Install aluminum-framed systems to comply with the following maximum erection tolerances:
 - 1. Location and Plane: Linit variation from true location and plane to 1/8 inch in 12 feet; 1/4 inch over total enge
 - 2. Alignment:
 - a. Vhere surfaces abut in line, limit offset from true alignment to 1/16 inch.b. Where surfaces meet at corners, limit offset from true alignment to 1/32 inch.
 - 0. Where surfaces meet at corners, mult onset from the angument to 1/32 men.
- B. Diagonal Measurements: Limit difference between diagonal measurements to 1/8 inch.

3.4 ADJUSTING

Adjust operating entrance door hardware to function smoothly as recommended by manufacturer.

END OF SECTION



SPEC SECTION 08 80 00

GLAZING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section includes glazing for the following products and applications including those specified in other Sections where glazing requirements are specified by reference to this Section:
 - 1. Windows.
 - 2. Doors.
 - 3. Storefront framing.
 - 4. Glazed entrances.
 - 5. Interior borrowed lites.

1.3 DEFINITIONS

- A. Glass Manufacturers: Firms that produce primary glass, fabricated glass, or both, as defined in referenced glazing publications
- B. Glass Thicknesses: Understeed by thickness designations in millimeters according to ASTM C 1036.

C. Interspace: Space between lites of an insulating-glass unit.

1.4 PERFORMANCE REQUIREMENTS

General: Installed glazing systems shall withstand normal thermal movement and wind and impact loads (where applicable) without failure, including loss or glass breakage attributable to the following: defective manufacture, fabrication, or installation; failure of sealants or gaskets to remain watertight and airtight; deterioration of glazing materials; or other defects in construction.

- B. Delegated Design: Design glass, including comprehensive engineering analysis according to ASTM E 1300 by a qualified professional engineer, using the following design criteria:
 - 1. Design Wind Pressures: As indicated on Drawings.
 - 2. Design Snow Loads: As indicated on Drawings.

Tetra Tech 200-15704-17001

GLAZING 08 80 00 - 1

- 3. Vertical Glazing: For glass surfaces sloped 15 degrees or less from vertical, design glass to resist design wind pressure based on glass type factors for short-duration load.
- 4. Sloped Glazing: For glass surfaces sloped more than 15 degrees from vertical, design glass to resist each of the following combinations of loads:
 - a. Outward design wind pressure minus the weight of the glass. Base design on glastype factors for short-duration load.
 - b. Inward design wind pressure plus the weight of the glass plus half of the design snow load. Base design on glass type factors for short-duration load.
 - c. Half of the inward design wind pressure plus the weight of the glass plue design snow load. Base design on glass type factors for long-duration load.
- 5. Glass Type Factors for Patterned Glass:
 - a. Short-Duration Glass Type Factor for Patterned Glass: 1
 - b. Long-Duration Glass Type Factor for Patterned Glass 0.6
- 6. Thickness of Patterned Glass: Base design of patterned glass on thickness at thinnest part of the glass.
- 7. Probability of Breakage for Sloped Glazing: For gass surfaces sloped more than 15 degrees from vertical, design glass for a probability of breakage not greater than 0.001.
- 8. Maximum Lateral Deflection: For glass supported on all four edges, limit center-of-glass deflection at design wind pressure to not nore than 1/50 times the short-side length or 1 inch, whichever is less.
- 9. Differential Shading: Design glass to esist thermal stresses induced by differential shading within individual glass line.
- C. Thermal Movements: Allow for thermal movements from ambient and surface temperature changes acting on glass framing moneys and glazing components.
 - 1. Temperature Change. 12, veg F, ambient; 180 deg F, material surfaces.
- 1.5 PRECONSTRUCTION TESTING
 - A. Preconstruction adhesion and Compatibility Testing: Test each glazing material type, tape sealant, tasket, glazing accessory, and glass-framing member for adhesion to and compatibility with classement glazing sealants.



- Testing will not be required if data are submitted based on previous testing of current sealant products and glazing materials matching those submitted.
- Use ASTM C 1087 to determine whether priming and other specific joint-preparation techniques are required to obtain rapid, optimum adhesion of glazing sealants to glass, tape sealants, gaskets, and glazing channel substrates.
- Test no fewer than eight Samples of each type of material, including joint substrates, shims, sealant backings, secondary seals, and miscellaneous materials.
- Schedule sufficient time for testing and analyzing results to prevent delaying the Work.
- 5. For materials failing tests, submit sealant manufacturer's written instructions for corrective measures including the use of specially formulated primers.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

1.6 SUBMITTALS, GENERAL

A. General: Submit all action submittals and informational submittals required by this Section concurrently.

1.7 ACTION SUBMITTALS

- A. Product Data: For each glass product and glazing material indicated.
- B. Glass Samples: For each type of the following products; 12 inches square.
 - 1. Patterned glass.
 - 2. Viewing mirror glass.
 - 3. Fire-resistive glazing products.
 - 4. Insulating glass.
 - 5. Laminated insulating glass.
 - 6. Patterned insulating glass.
- C. Glazing Schedule: List glass types and thicknesses for each size opening and location. Use same designations indicated on Drawings.
- D. Delegated-Design Submittal: For glass indicate to comply with performance requirements and design criteria, including analysis data signed and sealed by the qualified professional engineer responsible for their preparation.
- E. Warranties: Sample of special warranties.

1.8 INFORMATIONAL SUBMET

- A. Qualification Data: For thistaners
- B. Product Certificates: For class and glazing products, from manufacturer.
- C. Preconstruction, dheston and compatibility test report.

1.9 CLOSFOUT SUBMITTALS

Warranties: Executed special warranties.

QUALITY ASSURANCE

Installer Qualifications: A qualified installer who employs glass installers for this Project who are certified under the National Glass Association's Certified Glass Installer Program.

B. Source Limitations for Glass: Obtain tinted float glass, coated float glass, laminated glass, and insulating glass from single source from single manufacturer for each glass type.

- C. Source Limitations for Glazing Accessories: Obtain from single source from single manufacturer for each product and installation method.
- D. Glazing Publications: Comply with published recommendations of glass product manufacturers and organizations below, unless more stringent requirements are indicated. Refer to these publications for glazing terms not otherwise defined in this Section or in referenced standards.
 - 1. GANA Publications: GANA's "Laminated Glazing Reference Manual" and GANA "Glazing Manual."
 - 2. AAMA Publications: AAMA GDSG-1, "Glass Design for Sloped Glazing AAMA TIR-A7, "Sloped Glazing Guidelines."
 - 3. IGMA Publication for Sloped Glazing: IGMA TB-3001, "Guideanes for Sloped Glazing."
 - 4. IGMA Publication for Insulating Glass: SIGMA TM-3000, "North American Glazing Guidelines for Sealed Insulating Glass Units for Commercial and Readential Use."
- E. Safety Glazing Labeling: Where safety glazing labeling is indicated opermanently mark glazing with certification label of the SGCC or another certification agency acceptable to authorities having jurisdiction. Label shall indicate manufacturer's name, type of glass, thickness, and safety glazing standard with which glass complies.
- F. Fire-Protection-Rated Glazing Labeling: Permarchily mark fire-protection-rated glazing with certification label of a testing agency acceptable to autorities having jurisdiction. Label shall indicate manufacturer's name, test standard, whether glazing is for use in fire doors or other openings, whether or not glazing passer hose stream test, whether or not glazing has a temperature rise rating of 450 deg F, anothe Gre-resistance rating in minutes.
- G. Insulating-Glass Certification Program: Permanently marked either on spacers or on at least one component lite of units with appropriate certification label of IGCC.
- H. Preinstallation Conference: Conduct conference at Project site.
 - 1. Review and finance construction schedule and verify availability of materials, Installer's personnel, equipment, and facilities needed.
 - 2. Review temperary protection requirements for glazing during and after installation.

1.11 DEL VELY, STORAGE, AND HANDLING

Protect glazing materials according to manufacturer's written instructions. Prevent damage to glass and glazing materials from condensation, temperature changes, direct exposure to sun, or other causes.

PROJECT CONDITIONS

A. Environmental Limitations: Do not proceed with glazing when ambient and substrate temperature conditions are outside limits permitted by glazing material manufacturers and when glazing channel substrates are wet from rain, frost, condensation, or other causes.

1. Do not install glazing sealants when ambient and substrate temperature conditions are outside limits permitted by sealant manufacturer or below 40 deg F.

1.13 WARRANTY

- A. Manufacturer's Special Warranty for Coated-Glass Products: Manufacturer's standard form in which coated-glass manufacturer agrees to replace coated-glass units that deteriorate within specified warranty period. Deterioration of coated glass is defined as defects developed from normal use that are not attributed to glass breakage or to maintaining and cleaning roated glass contrary to manufacturer's written instructions. Defects include peeling, cracking, and other indications of deterioration in coating.
 - 1. Warranty Period: Five years from date of Substantial Completion,
- B. Manufacturer's Special Warranty on Laminated Glass: Manufacturer's standard form in which laminated-glass manufacturer agrees to replace laminated-glass and that deteriorate within specified warranty period. Deterioration of laminated glass is defined as defects developed from normal use that are not attributed to glass breakage or to maintaining and cleaning laminated glass contrary to manufacturer's written instructions. Defects include edge separation, delamination materially obstructing vision through glass, and blemishes exceeding those allowed by referenced laminated-glass standard.
 - 1. Warranty Period: 10 years from date of Su standal Completion.
- C. Manufacturer's Special Warranty on Insulating Glass: Manufacturer's standard form in which insulating-glass manufacturer agrees to replace insulating-glass units that deteriorate within specified warranty period. Deterioration of insulating glass is defined as failure of hermetic seal under normal use that is no acributed to glass breakage or to maintaining and cleaning insulating glass contrary to manufacturer's written instructions. Evidence of failure is the obstruction of vision by dust measure, or film on interior surfaces of glass.
 - 1. Warranty Period 10 ears from date of Substantial Completion.
- PART 2 PRODUCTS
- 2.1 GLASS PRODUCTS, GENERAL

Thickness: Where glass thickness is indicated, it is a minimum. Provide glass lites in nicknesses as needed to comply with requirements indicated.

Strength: Where fully tempered glass is indicated, provide Kind FT heat-treated float glass.

- Thermal and Optical Performance Properties: Provide glass with performance properties specified, as indicated in manufacturer's published test data, based on procedures indicated below:
- 1. For monolithic-glass lites, properties are based on units with lites 6.0 mm thick.
- 2. For laminated-glass lites, properties are based on products of construction indicated.

- 3. For insulating-glass units, properties are based on units of thickness indicated for overall unit and for each lite.
- 4. U-Factors: Center-of-glazing values, according to NFRC 100 and based on LBL's WINDOW 5.2 computer program, expressed as Btu/sq. ft. x h x deg F.
- 5. Solar Heat-Gain Coefficient and Visible Transmittance: Center-of-glazing values, according to NFRC 200 and based on LBL's WINDOW 5.2 computer program.
- 6. Visible Reflectance: Center-of-glazing values, according to NFRC 300.

2.2 GLASS PRODUCTS

- A. Float Glass: ASTM C 1036, Type I, Quality-Q3, Class I (clear) unless otherwise indice
- B. Heat-Treated Float Glass: ASTM C 1048; Type I; Quality-Q3; Class I (clear) taless otherwise indicated; of kind and condition indicated.
 - 1. Fabrication Process: By horizontal (roller-hearth) process with roll-wave distortion parallel to bottom edge of glass as installed unless otherwise indicated.
 - 2. For uncoated glass, comply with requirements for Condition
 - 3. For coated vision glass, comply with requirement for Condition C (other coated glass).
- C. Uncoated Tinted Float Glass: Class 2, complying with other requirements specified.
 - 1. Basis-of-Design Product: Subject to compliance with requirements, provide PPG Industries, Inc.; Solexia, or comparable product.
 - 2. Tint Color: Light green.
 - 3. Visible Light Transmittance: There in minimum.
- D. Low-E Coated Float Glass: Complying with requirements specified.
 - 1. Basis-of-Design Produce. Subject to compliance with requirements, provide PPG Industries, Inc. 50 prba, 60, or comparable product.
- E. Tempered Patterned Glass. ASTM C 1048, Kind FT (fully tempered), Type II, Class 1 (clear), Form 3; Qualiv-Oa Finish F1 (patterned one side), of pattern indicated.
 - 1. Basis-of Design Product: Subject to compliance with requirements, provide AGC Flat Glass North America; Industrex, or comparable product.
- F. Reflective-Coated Vision Glass (Viewing Mirror): ASTM C 1376, coated by pyrolytic process, and complying with other requirements specified.
 - Basis-of-Design Product: Subject to compliance with requirements, provide Pilkington North America, Inc.; Mirropane T.M. Transparent Mirror or comparable product.
 - 2. Kind: Kind CV (coated vision glass).
 - 3. Coating Color: Silver.
 - 4. Glass: Tinted float.
 - 5. Tint Color: Gray.
 - 6. Visible Light Transmittance: 11 percent.
 - 7. Visible Reflectance: 71 percent.

2.3 LAMINATED GLASS

- A. Laminated Glass: ASTM C 1172, and complying with testing requirements in 16 CFR 1201 for Category II materials, and with other requirements specified. Use materials that have a proven record of no tendency to bubble, discolor, or lose physical and mechanical properties after fabrication and installation.
 - 1. Construction: Laminate glass with polyvinyl butyral interlayer to comply with interlayer, manufacturer's written recommendations.
 - 2. Interlayer Thickness: Provide thickness not less than that indicated and comply with requirements.
 - 3. Interlayer Color: Clear unless otherwise indicated.

2.4 INSULATING GLASS

- A. Insulating-Glass Units: Factory-assembled units consisting of search tes of glass separated by a dehydrated interspace, qualified according to ASTM E 2190, and complying with other requirements specified.
 - 1. Sealing System: Dual seal, with manufacturer's standard primary and secondary.
 - 2. Spacer: Manufacturer's standard spacer matchial and construction.
 - 3. Desiccant: Molecular sieve or silica gel, or bland of both.

2.5 FIRE-PROTECTION-RATED GLAZING

- A. Fire-Protection-Rated Glazing, General: Listed and labeled by a testing agency acceptable to authorities having jurisdiction to the rotection ratings indicated, based on testing according to NFPA 252 for door assemblies and NFPA 257 for window assemblies.
- B. Fire-Protection-Rated reported Glass: 1/4-inch-thick, fire-protection-rated tempered glass, complying with testing equipements in 16 CFR 1201 for Category II materials.
 - 1. Products: subject to compliance with requirements, available products that may be incorporated into the Work include, but are not limited to, the following:

InerEdge, Inc., a subsidiary of AFG Industries, Inc.; PyroEdge-20. Vetrotech Saint-Gobain; SSG Pyroswiss.

Laminated Ceramic Glazing: Laminated glass made from 2 plies of clear, ceramic flat glass; X16-inch total nominal thickness; complying with testing requirements in 16 CFR 1201 for Category II materials.

- 1. Products: Subject to compliance with requirements, available products that may be incorporated into the Work include, but are not limited to, the following:
 - a. Nippon Electric Glass Co., Ltd. (distributed by Technical Glass Products); FireLite Plus.
 - b. Vetrotech Saint-Gobain; SGG Keralite FR-L.

2.6 GLAZING GASKETS

- A. Dense Compression Gaskets: Molded or extruded gaskets of profile and hardness required to maintain watertight seal, made from one of the following:
 - 1. EPDM complying with ASTM C 864.
 - 2. Silicone complying with ASTM C 1115.
 - 3. Thermoplastic polyolefin rubber complying with ASTM C 1115.
- B. Soft Compression Gaskets: Extruded or molded, closed-cell, integral-skinned EPDM, silkore, or thermoplastic polyolefin rubber gaskets complying with ASTM C 509, Type C olak; of profile and hardness required to maintain watertight seal.
 - 1. Application: Use where soft compression gaskets will be compressed by inserting dense compression gaskets on opposite side of glazing or pressure upplied by means of pressure-glazing stops on opposite side of glazing.

2.7 GLAZING SEALANTS

- A. General:
 - Compatibility: Provide glazing sealants that an compatible with one another and with other materials they will contact, including glast products, seals of insulating-glass units, and glazing channel substrates, under conditions of service and application, as demonstrated by sealant manufacturer based on testing and field experience.
 Suitability: Comply with coalign and glass manufacturers' written instructions for
 - 2. Suitability: Comply with reason and glass manufacturers' written instructions for selecting glazing sealants suitable for applications indicated and for conditions existing at time of installation.
 - 3. Sealants used inside the weatherproofing system shall comply with the testing and product requirements of the California Department of Health Services' "Standard Practice for the Testing of Volatle Organic Emissions from Various Sources Using Small-Scale Environmental Clambers."
 - 4. Colors of hyposed glazing Sealants: As selected by Architect from manufacturer's full range.
- B. Glazing sealant Neutral-curing silicone glazing sealant complying with ASTM C 920, Type S, Grade NS, Class 50, Use NT.

C. Glazing Sealants for Fire-Rated Glazing Products: Products that are approved by testing agencies that listed and labeled fire-resistant glazing products with which they are used for applications and fire-protection ratings indicated.



GLAZING TAPES

A. Back-Bedding Mastic Glazing Tapes: Preformed, butyl-based, 100 percent solids elastomeric tape; nonstaining and nonmigrating in contact with nonporous surfaces; with or without spacer rod as recommended in writing by tape and glass manufacturers for application indicated; and complying with ASTM C 1281 and AAMA 800 for products indicated below:

GLAZING 08 80 00 - 8

- 1. AAMA 804.3 tape, where indicated.
- AAMA 806.3 tape, for glazing applications in which tape is subject to continuous 2. pressure.
- 3. AAMA 807.3 tape, for glazing applications in which tape is not subject to continuous pressure.
- Expanded Cellular Glazing Tapes: Closed-cell, PVC foam tapes; factory coated with adhesive B. on both surfaces; and complying with AAMA 800 for the following types:
 - 1.
 - AAMA 810.1, Type 1, for glazing applications in which tape acts as the primary sea AAMA 810.1, Type 2, for glazing applications in which tape is used in combinatio 2. with a full bead of liquid sealant.

MISCELLANEOUS GLAZING MATERIALS 2.9

- General: Provide products of material, size, and shape comply whin referenced glazing A. standard, requirements of manufacturers of glass and other glazing materials for application indicated, and with a proven record of compatibility with surfaces contacted in installation.
- alant or gasket manufacturer. B. Cleaners, Primers, and Sealers: Types recommended by
- C. Setting Blocks: Elastomeric material with a Shore, The A durometer hardness of 85, plus or minus 5.
- Elastomeric blocks or continuous extrusions of hardness required by glass D. Spacers: in lact for installation indicated. manufacturer to maintain glass lites
- Edge Blocks: Elastomeric mac ardness needed to limit glass lateral movement (side E. walking).
- Cylindrical Glazing Scalart Bicking: ASTM C 1330, Type O (open-cell material), of size and density to control glazing sealant depth and otherwise produce optimum glazing sealant F. performance.
- G. Perimeter Insult on for Fire-Resistive Glazing: Product that is approved by testing agency that listed and labered fire-resistant glazing product with which it is used for application and fireprotection rating indicated.

FABRICATION OF GLAZING UNITS 2.10



Fabricate glazing units in sizes required to fit openings indicated for Project, with edge and face clearances, edge and surface conditions, and bite complying with written instructions of product manufacturer and referenced glazing publications, to comply with system performance requirements.

- 2.11 MONOLITHIC-GLASS TYPES
 - A. Glass Type GL-1: Clear fully tempered float glass.

ADDITION AND RENOVATIONS

28^C

14

Ś

- 1. Thickness: 6.0 mm.
- 2. Provide safety glazing labeling.
- B. Glass Type GL-2: Tempered patterned glass.
 - 1. Thickness: 4.0 mm.
 - 2. Provide safety glazing labeling.
- C. Glass Type GL-3: Reflective-coated vision glass, fully tempered float glass.
 - 1. Thickness: 6.0 mm.
 - 2. Coating Location: First surface.
 - 3. Provide safety glazing labeling.

2.12 **INSULATING-GLASS TYPES**

- A. Glass Type GL-4: Low-e-coated, tinted insulating glass.
 - Overall Unit Thickness: 1 inch. 1.
 - 2. Thickness of Each Glass Lite: 6.0 mm.
 - 3. Outdoor Lite: Tinted fully tempered float
 - 4. Interspace Content: Air.
 - 5. Indoor Lite: Clear fully tempered floated
 - Low-E Coating: Pyrolytic or sputtered outhird surface. Visible Light Transmittance: 61 percent minimum. 6.
 - 7.
 - Winter Nighttime U-Factor: 0.22 maximum. 8.
 - Summer Daytime U-Factor. 0.27 naximum. 9.
 - 10. Solar Heat Gain Coefficient maximum.
 - Provide safety glazing abelin 11.
- 2.13 INSULATING-LAM **ATED-GLASS TYPES**
 - Glass Type GK-5: Clear insulating laminated glass. A.
 - Orerall Inft Thickness: 1-1/8 inch. 1.
 - Thekneys of Outdoor Lite: 6.0 mm. 2.
 - Sutdoor Lite: Fully tempered float glass. 3.
 - Interspace Content: Air.

Indoor Lite: Clear laminated glass with two plies of fully tempered float glass.

- Thickness of Each Glass Ply: 5.0 mm. a.
- Interlayer Thickness: 0.030 inch minimum. b.
- 6. Provide safety glazing labeling.

2.14 FIRE-PROTECTION-RATED GLAZING TYPES

A. Glass Type GL-6: 20-minute fire-rated glazing without hose-stream test; fire-protection-rated tempered glass.

GLAZING 08 80 00 - 10

- 1. Provide safety glazing labeling.
- B. Glass Type GL-7: Fire-rated glazing of minute-rating indicated; laminated ceramic glazing.
 - 1. Provide safety glazing labeling.

2.15 PATTERNED INSULATING-GLASS TYPES

- A. Glass Type GL-8: Patterned, tinted insulating glass.
 - 1. Overall Unit Thickness: 1 inch.
 - 2. Thickness of Outdoor Lite: 6.0 mm.
 - 3. Outdoor Lite: Tinted fully tempered float glass.
 - 4. Interspace Content: Air.
 - 5. Thickness of Indoor Lite: 4.0 mm.
 - 6. Indoor Lite: Tempered patterned glass.
 - 7. Provide safety glazing labeling.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine framing, glazing channels, and stops, with Installer present, for compliance with the following:
 - 1. Manufacturing and installation tolerances, including those for size, squareness, and offsets at corners.

211

- 2. Presence and functioning of weep systems.
- 3. Minimum required face and edge clearances.
- 4. Effective sealing between joints of glass-framing members.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.
- C. Beginning instillation constitutes Contractor's acceptance of substrates and conditions.

PRENARATION

3.2

Clean glazing channels and other framing members receiving glass immediately before glazing. Remove coatings not firmly bonded to substrates.

Examine glazing units to locate exterior and interior surfaces. Label or mark units as needed so that exterior and interior surfaces are readily identifiable. Do not use materials that will leave visible marks in the completed work.

3.3 GLAZING, GENERAL

- A. Comply with combined written instructions of manufacturers of glass, sealants, gaskets, and other glazing materials, unless more stringent requirements are indicated, including those in referenced glazing publications.
- B. Adjust glazing channel dimensions as required by Project conditions during installation to provide necessary bite on glass, minimum edge and face clearances, and adequate realant thicknesses, with reasonable tolerances.
- C. Protect glass edges from damage during handling and installation. Remove damage grass from Project site and legally dispose of off Project site. Damaged glass is glass with edge during or other imperfections that, when installed, could weaken glass and impair performance and appearance.
- D. Apply primers to joint surfaces where required for adhesion of sea ants, as determined by preconstruction testing.
- E. Install setting blocks in sill rabbets, sized and located to compty with referenced glazing publications, unless otherwise required by glass manufacturer. Set blocks in thin course of compatible sealant suitable for heel bead.
- F. Do not exceed edge pressures stipulated by glass manufacturers for installing glass lites.
- G. Provide spacers for glass lites where length plus width is larger than 50 inches.
 - Locate spacers directly opposite each other on both inside and outside faces of glass. Install correct size and spacing to preserve required face clearances, unless gaskets and glazing tapes are used that say demonstrated ability to maintain required face clearances and to comply with system performance requirements.
 Provide 1/8-inch minimum other of spacers on glass and use thickness equal to sealant
 - 2. Provide 1/8-inch minimum oite of spacers on glass and use thickness equal to sealant width. With glazing tape, use thickness slightly less than final compressed thickness of tape.
- H. Provide edge flooring where indicated or needed to prevent glass lites from moving sideways in glazing barrel, as recommended in writing by glass manufacturer and according to requirements in referenced glazing publications.
- I. Set eles lites in each series with uniform pattern, draw, bow, and similar characteristics.
 - Set glass lites with proper orientation so that coatings face exterior or interior as specified.



Where wedge-shaped gaskets are driven into one side of channel to pressurize sealant or gasket on opposite side, provide adequate anchorage so gasket cannot walk out when installation is subjected to movement.

2. Square cut wedge-shaped gaskets at corners and install gaskets in a manner recommended by gasket manufacturer to prevent corners from pulling away; seal corner joints and butt joints with sealant recommended by gasket manufacturer.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

3.4 TAPE GLAZING

- A. Position tapes on fixed stops so that, when compressed by glass, their exposed edges are flush with or protrude slightly above sightline of stops.
- B. Install tapes continuously, but not necessarily in one continuous length. Do not stretch tapes to make them fit opening.
- C. Cover vertical framing joints by applying tapes to heads and sills first and then to jambs horizontal framing joints by applying tapes to jambs and then to heads and sills.
- D. Place joints in tapes at corners of opening with adjoining lengths butted together net tapped. Seal joints in tapes with compatible sealant approved by tape manufacturer.
- E. Do not remove release paper from tape until right before each glazing until installed.
- F. Apply heel bead of elastomeric sealant.
- G. Center glass lites in openings on setting blocks and press firmly against tape by inserting dense compression gaskets formed and installed to lock in prace against faces of removable stops. Start gasket applications at corners and work toward centers of openings.
- H. Apply cap bead of elastomeric sealant over exposed case of tape.

3.5 GASKET GLAZING (DRY)

- A. Cut compression gaskets to lengths recommended by gasket manufacturer to fit openings exactly, with allowance for stretch during installation.
- B. Insert soft compression gasket between glass and frame or fixed stop so it is securely in place with joints miter cut and bonded together at corners.
- C. Installation with Drive-in-Wedge Gaskets: Center glass lites in openings on setting blocks and press firmly a ainst soft compression gasket by inserting dense compression gaskets formed and installed to lock in place against faces of removable stops. Start gasket applications at corners and work toward centers of openings. Compress gaskets to produce a weathertight seal without developing berding stresses in glass. Seal gasket joints with sealant recommended by gasket manufacturer.

E.

Installation with Pressure-Glazing Stops: Center glass lites in openings on setting blocks and press firmly against soft compression gasket. Install dense compression gaskets and pressure-glazing stops, applying pressure uniformly to compression gaskets. Compress gaskets to produce a weathertight seal without developing bending stresses in glass. Seal gasket joints with sealant recommended by gasket manufacturer.

Install gaskets so they protrude past face of glazing stops.

ADDITION AND RENOVATIONS

3.6 SEALANT GLAZING (WET)

- A. Install continuous spacers, or spacers combined with cylindrical sealant backing, between glass lites and glazing stops to maintain glass face clearances and to prevent sealant from extruding into glass channel and blocking weep systems until sealants cure. Secure spacers or spacers and backings in place and in position to control depth of installed sealant relative to edge clearance for optimum sealant performance.
- B. Force sealants into glazing channels to eliminate voids and to ensure complete wetting tr of sealant to glass and channel surfaces.
- C. Tool exposed surfaces of sealants to provide a substantial wash away from glass

3.7 CLEANING AND PROTECTION

- A. Protect glass from damage immediately after installation by attaching crossed streamers to framing held away from glass. Do not apply markers to glass surface. Remove nonpermanent labels and clean surfaces.
- B. Protect glass from contact with contaminating substances resulting from construction operations. If, despite such protection, contaminating substances do come into contact with glass, remove substances immediately as recommended in writing by glass manufacturer.
- C. Examine glass surfaces adjacent to or below exterior concrete and other masonry surfaces at frequent intervals during construction, but not less than once a month, for buildup of dirt, scum, alkaline deposits, or stains; remove concommended in writing by glass manufacturer.
- D. Remove and replace glass that is proper, chipped, cracked, or abraded or that is damaged from natural causes, accidents, and vandaism, during construction period.
- E. Wash glass on both exposed surfaces in each area of Project not more than four days before date scheduled for inspect as hat establish date of Substantial Completion. Wash glass as recommended in writing by glass manufacturer.

END OF SECTION



SPEC SECTION 08 90 00

LOUVERS AND VENTS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Fixed, extruded-aluminum louvers.
- 1.3 DEFINITIONS
 - A. Louver Terminology: Definitions of terms for metar louvers contained in AMCA 501 apply to this Section unless otherwise defined in this Section or in referenced standards.
 - B. Horizontal Louver: Louver with barizontal blades; i.e., the axes of the blades are horizontal.
 - C. Drainable-Blade Louver: Louver with blades having gutters that collect water and drain it to channels in jambs and multion, which carry it to bottom of unit and away from opening.

1.4 PERFORMANCE REQUILEMENTS

- A. Delegated Design: Design louvers, including comprehensive engineering analysis by a qualified professional engineer, using structural performance requirements and design criteria indicated
- B. Structural Performance: Louvers shall withstand the effects of gravity loads and the following loads and stresses within limits and under conditions indicated without permanent deformation of louver components, noise or metal fatigue caused by louver blade rattle or flutter, or permanent damage to fasteners and anchors. Wind pressures shall be considered to act normal to the face of the building.
 - 1. Wind Loads: Determine loads based on pressures as indicated on Drawings.
- C. Thermal Movements: Allow for thermal movements from ambient and surface temperature changes, without buckling, opening of joints, overstressing of components, failure of connections, or other detrimental effects.
 - 1. Temperature Change (Range): 120 deg F, ambient; 180 deg F, material surfaces.

Tetra Tech 200-15704-17001

LOUVERS AND VENTS 08 90 00 - 1

- D. Louver Performance Ratings: Provide louvers complying with requirements specified, as demonstrated by testing manufacturer's stock units identical to those provided, except for length and width according to AMCA 500-L.
- 1.5 SUBMITTALS, GENERAL
 - A. General: Submit all action submittals (except Samples for Verification) required by this Section concurrently.

1.6 ACTION SUBMITTALS

- A. Product Data: For each type of product indicated.
 - 1. For louvers specified to bear AMCA seal, include printed catalog pages showing specified models with appropriate AMCA Certified Ratings seas.
- B. Shop Drawings: For louvers and accessories. Include plans elevitions, sections, details, and attachments to other work. Show frame profiles and blace profiles, angles, and spacing.
 - 1. Show weep paths, gaskets, flashing, sealent, and other means of preventing water intrusion.
 - 2. Show mullion profiles and locations.
- C. Samples for Initial Selection: For units with factory-applied color finishes.
- D. Samples for Verification: For each type of metal finish required.
- E. Delegated-Design Submittate For louvers indicated to comply with structural performance requirements and design criteria, including analysis data signed and sealed by the qualified professional engineer response le for their preparation.
- 1.7 QUALITY ASSULANCE

B.

D.

- A. Source Idmitation: Obtain louvers and vents from single source from a single manufacturer where indicated to be of same type, design, or factory-applied color finish.
 - Welding: Qualify procedures and personnel according to the following:
 - AWS D1.2/D1.2M, "Structural Welding Code Aluminum."
 - SMACNA Standard: Comply with recommendations in SMACNA's "Architectural Sheet Metal Manual" for fabrication, construction details, and installation procedures.
 - Preinstallation Conference: Conduct conference at Project site.
 - 1. Review and discuss the finishing of aluminum that is required to be coordinated with the finishing of other aluminum work for color and finish matching.

1.8 PROJECT CONDITIONS

A. Field Measurements: Verify actual dimensions of openings by field measurements before fabrication.

PART 2 - PRODUCTS

2.1 MATERIALS

- A. Aluminum Extrusions: ASTM B 221, Alloy 6063-T5, T-52, or T6.
- B. Aluminum Sheet: ASTM B 209, Alloy 3003 or 5005 with temper as required in forming, or as otherwise recommended by metal producer for required finish.
- C. Aluminum Castings: ASTM B 26/B 26M, Alloy 319.
- D. Fasteners: Use types and sizes to suit unit installation conditions.
 - 1. Use Phillips flat-head screws for exposed fasteners valess otherwise indicated.
 - 2. For fastening aluminum, use aluminum or 300 series stainless-steel fasteners.
 - 3. For color-finished louvers, use fasteners with bards that match color of louvers.
- E. Postinstalled Fasteners for Concrete and Mason v: Torque-controlled expansion anchors, made from stainless-steel components, with capability to sustain, without failure, a load equal to 4 times the loads imposed, for concrete per optimes the load imposed, for masonry, as determined by testing per ASTM E 488, conducted by a qualified independent testing agency.
- F. Bituminous Paint: Cold-applied as that emulsion complying with ASTM D 1187.

2.2 FABRICATION, GENERA

- A. Assemble lowers in factory to minimize field splicing and assembly. Disassemble units as necessary for chapping and handling limitations. Clearly mark units for reassembly and coordinated installation.
- B. VertcabAssemblies: Where height of louver units exceeds fabrication and handling limitations, fabricite units to permit field-bolted assembly with close-fitting joints in jambs and mullions, reinforced with splice plates.
 - Continuous Vertical Assemblies: Fabricate units without interrupting blade-spacing pattern unless horizontal mullions are indicated.

Maintain equal louver blade spacing, including separation between blades and frames at head and sill, to produce uniform appearance.

D. Fabricate frames, including integral sills, to fit in openings of sizes indicated, with allowances made for fabrication and installation tolerances, adjoining material tolerances, and perimeter sealant joints.

- 1. Frame Type: Channel unless otherwise indicated.
- E. Include supports, anchorages, and accessories required for complete assembly.
- F. Provide vertical mullions of type and at spacings indicated, but not more than recommended by manufacturer, or 72 inches o.c., whichever is less.
 - 1. Semirecessed Mullions: Where indicated, provide mullions partly recessed behind louver, blades so louver blades appear continuous. Where length of louver exceeds fabrication and handling limitations, fabricate with interlocking split mullions and close uting blade splices designed to permit expansion and contraction.
- G. Join frame members to each other and to fixed louver blades with fillet wells concealed from view unless otherwise indicated or size of louver assembly makes bolter connections between frame members necessary.

2.3 FIXED, EXTRUDED-ALUMINUM LOUVERS

- A. Horizontal, Continuous-Line, Drainable-Blade Louver, Drainable-blade louver with blade gutters (drains) in rear two-thirds of blades only and win semirecessed mullions capable of collecting and draining water from blades.
 - 1. Basis-of-Design Product: Subject to compliance with requirements, provide Arrow United Industries; a division of Mertek Inc; Model EA415DCL or comparable product.
 - 2. Louver Depth: 4 inches.
 - 3. Frame and Blade Nominal Thick est. Not less than 0.125 inch.
 - 4. Louver Performance Rating.
 - a. Free Area: Not ess than 9 sq. ft. for 48-inch-wide by 48-inch-high louver.
- 2.4 LOUVER SCREENS
 - A. General: Provide creen at each exterior louver.
 - 1. Screen Docation for Fixed Louvers: Interior face.
 - 2. Screening Type: Bird screening.
 - B. Secure screen frames to louver frames with stainless-steel machine screws, spaced a maximum of 6 inches from each corner and at 12 inches o.c.
 - Louver Screen Frames: Fabricate with mitered corners to louver sizes indicated.
 - 1. Metal: Same kind and form of metal as indicated for louver to which screens are attached. Reinforce extruded-aluminum screen frames at corners with clips.
 - 2. Finish: Same finish as louver frames to which louver screens are attached.
 - D. Louver Screening for Aluminum Louvers:
 - 1. Bird Screening: Aluminum, 1/2-inch-square mesh, 0.063-inch wire.

2.5 BLANK-OFF PANELS

- A. Insulated, Blank-Off Panels: Laminated panels consisting of insulating core surfaced on back and front with metal sheets and attached to back of louver.
 - 1. Thickness: 2 inches.
 - 2. Metal Facing Sheets: Aluminum sheet, not less than 0.025-inch nominal thickness.
 - 3. Insulating Core: Extruded-polystyrene foam.
 - 4. Seal perimeter joints between panel faces and louver frames with gaskets or seala
 - 5. Panel Finish: Same type of finish applied to louvers, but black color.
 - 6. Attach blank-off panels with clips.
- 2.6 FINISHES, GENERAL
 - A. Comply with NAAMM's "Metal Finishes Manual for Architectural and Metal Products" for recommendations for applying and designating finishes.

2.7 ALUMINUM FINISHES

- A. Finish louvers after assembly.
- B. High-Performance Organic Finish: 2-coat fixed polymer finish complying with AAMA 2605 and containing not less than 70 percent PVDF risin by weight in color coat. Prepare, pretreat, and apply coating to exposed metal surfaces to comply with coating and resin manufacturers' written instructions.
 - 1. Color and Gloss: As selected by Architect from manufacturer's full range.

PART 3 - EXECUTION

- 3.1 EXAMINATION
 - A. Examine substrates and openings, with Installer present, for compliance with requirements for installation tolerances and other conditions affecting performance.
 - B. Proceed with installation only after unsatisfactory conditions have been corrected.
 - Reginning installation constitutes Contractor's acceptance of substrates and conditions.

PREPARATION

. Coordinate setting drawings, diagrams, templates, instructions, and directions for installation of anchorages that are to be embedded in concrete or masonry construction. Coordinate delivery of such items to Project site.

3.3 INSTALLATION

- A. Locate and place louvers and vents level, plumb, and at indicated alignment with adjacent work.
- B. Use concealed anchorages where possible. Provide brass or lead washers fitted to screws where required to protect metal surfaces and to make a weathertight connection.
- C. Form closely fitted joints with exposed connections accurately located and secured.
- D. Provide perimeter reveals and openings of uniform width for sealants and join indicated.
- E. Repair finishes damaged by cutting, welding, soldering, and grinding. Restore finishes so no evidence remains of corrective work. Return items that cannot be refinished in the field to the factory, make required alterations, and refinish entire unit or provide new parts.
- F. Protect unpainted galvanized and nonferrous-metal surfaces that win be in contact with concrete, masonry, or dissimilar metals from corrosion and galvanic action by applying a heavy coating of bituminous paint or by separating surfaces with waterpoof gaskets or nonmetallic flashing.

3.4 ADJUSTING AND CLEANING

- A. Clean exposed surfaces of lowers and cents that are not protected by temporary covering, to remove fingerprints and son during construction period. Do not let soil accumulate during construction period.
- B. Before final inspection clean exposed surfaces with water and a mild soap or detergent not harmful to finishes. Thoraghly rinse surfaces and dry.
- C. Restore lowers indivents damaged during installation and construction so no evidence remains of corrective pork. If results of restoration are unsuccessful, as determined by Architect, remove a maged units and replace with new units.



Touch up minor abrasions in finishes with air-dried coating that matches color and gloss of, and is compatible with, factory-applied finish coating.

END OF SECTION

BRANDYWINE SCHOOL DISTRICT CARRCROFT ELEMENTARY SCHOOL

SECTION 090000 – FINISH SCHEDULE

Carpet Tile

Basis of Design: Mohawk Group

CPT-1: Manufacturer: Mohawk Group / Denim Collection Style: Hem / GT178 Color: 955 Skinny Size: Modular 24" x 24" / ER3 construction. Installation Method: Brick Ashlar Primary Backing: Ecoflex ICT Construction: Tufted Gauge: 1/12" Pile height average: .187" Fiber System: Dynex SD Nylon (Permanent Stain Resistance) Dye Method: 100% solution dyed Surface Flammability: ASTM E 648 Class 1 (Glue Down) Smoke Density: ASTM E662 Less than 450 CPT-2: Manufacturer: Mohawk Group / Denim Collection Style: Jean / GT177 Color: 955 Skinny Size: Modular 24" x 24" / ER3 construction Installation Method: Monolithic Primary Backing: Ecoflex ICT Construction: Tufted Gauge: 1/12" Pile height average: .187" Fiber System: Dynex SD Lylon Permanent Stain Resistance) Dye Method: 100% solution lye Surface Flammabilit, STALE 648 Class 1 (Glue Down) Smoke Density: AS E 62 Less than 450 **Porcelain Tile** Basis of Design: Gan in State Tile cture : Garden State Tile CT-1: Mahut

C1-1: Manufacture: Garden State Tile Style Sands
Cour: Grey Sand Finish: Natural
Size: 12" x 24"
Pattern: Brick Ashlar (Overlap per manufacturer's requirement) Location: Bathroom Floor Tile Grout Color: Mapei, Sahara Beige
Alternate CT-1: Manufacturer: Daltile Style: Keystones. Mosaic Colorbody Porcelain Color A: Lime Sherbet D619 Color B: Uptown Taupe D132 Finish: Natural

Tetra Tech

BRANDYWINE SCHOOL DISTRICT CARRCROFT ELEMENTARY SCHOOL

Size: 2" x 2" Pattern: See Floor Plan for Pattern Grout Color: Mapei, Sahara Beige Location: Bathroom Floor Tile (See Plan for floor pattern) CT-2: Manufacturer: Garden State Tile Style: Sands Color: Ivory Sand Finish: Semipolished Size: 12" x 24" Pattern: Brick Ashlar Grout Color: Mapei, Alabaster Location: Bathroom Wall Tile Alternate CT-2: Manufacturer: Altro Whiterock Color A: Mocha 206 Color: B: Eau De Chic 6605 (See elevation for pattern.) **Rubber Tile** Size: 24" X 24" X 2 mm, unless otherwise noted. Basis of Design: Nora Rubber Flooring / No wax, no sealer required RT-1: Manufacturer: Nora Style: Nora Plan / Eco Color: Tuna (Lightest Grey) Location: Corridor 154 RT-2: Manufacturer: Nora Style: Nora Plan / Eco Color: Catfish (Light Grey) Location: Corridor 154 / Cafeter RT-3: Manufacturer: Nora Style: Nora Plan / E Color: Hammerhead (Mechum Grey) Location: Corridor 15- Cafeteria 155 RT-4: Manufacturar: Nora Style: Nora Non / Eco Color: Mant. Ray (Dark Grey) Location: Corridor 154 / Cafeteria 155 RT-5: Many facturer: Nora Styre: Nora Plan / Eco Color: Mahi-Mahi (Green) Location: Cafeteria 155 Manufacturer: Nora Style: Nora Plan / Environcare Color: Lace Vine Location: Classrooms 118, 119, 126, 127, 127A RT-7: Manufacturer: Nora Style: Nora Plan / Environcare Color: Silver Crown Location: Classrooms 118, 119, 126, 127, 127A

Tetra Tech

BRANDYWINE SCHOOL DISTRICT CARRCROFT ELEMENTARY SCHOOL

- RT-8: Manufacturer: Nora Style: Nora Plan / Environcare Color: Driftwood Location: Classrooms 118, 119, 126, 127, 127A
- RT-9: Manufacturer: Nora Style: Nora Plan / Environcare Color: Marigold Glow Location: Classrooms 118, 119, 126, 127, 127A
- RT-10: Manufacturer: Nora Style: Norament Saturna Stairtreads & Stair Landing. Color: Callisto

4" Resilient Rubber Cove Wall Base

- Basis of Design: Roppe / Pinnacle RB-1: Manufacturer: Roppe
- Style: Pinnacle Color: Dolphin
- RB-2: Manufacturer: Roppe Style: Pinnacle Color: Pewter

Paint

Basis of Design:

Manufacturer: Sherwin Williams.

- Color: SW 7011 Natural Choice (General P-1: Wan Color)
- P-2: Color: SW 6441 White Mint
- Color: SW 6442 Supreme Gree P-3:
- P-4: Color: SW 6443 Relish
- P-5: Color: SW6652 Flan
- Color: SW 6654 Surpris P-6:
- P-7: Color: Ceiling Whit

Plastic Laminate

Basis of Design: W

- PL-1: Manufacture: Vilsonart or: Slate Grey D91-60 Col cabinets) PL-2
 - Ma ufacturer: Wilsonart
 - Color: Wallaby D439-60
 - (Countertop & Upper Cabinets)
- END OF SECTION 090000

SECTION 09 30 13

CERAMIC TILING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section Includes:
 - 1. Ceramic tile.
 - 2. Stone thresholds.
 - 3. Waterproof membrane.
 - 4. Crack isolation membrane for tile.
 - 5. Surface preparation materials
 - 6. Tile setting mortars and adhesives

B. Related Requirements:

- 1. Section 07 92 00 "Joint Scalants" for sealing of expansion, contraction, control, and isolation joints in tile surfaces.
- 2. Section 09 29 00 "Gypsum Board" for cementitious backer units.

1.3 DEFINITIONS

- A. General: Definitions in the ANSI A108 series of tile installation standards and in ANSI A137.1 apply to Work withis Section unless otherwise specified.
- B. ANSI A108 Series: ANSI A108.01, ANSI A108.02, ANSI A108.1A, ANSI A108.1B, ANSI A108.1C, ANSI A108.4, ANSI A108.5, ANSI A108.6, ANSI A108.8, ANSI A108.9, ANSI A108.10, ANSI A108.11, ANSI A108.12, ANSI A108.13, ANSI A108.14, ANSI A108.15, ANSI A108.16, and ANSI A108.17, which are contained in its "Specifications for Installation of Ceramic Tile."

Module Size: Actual tile size plus joint width indicated.

Face Size: Actual tile size, excluding spacer lugs.

1.4 PERFORMANCE REQUIREMENTS

A. Static Coefficient of Friction: For tile installed on walkway surfaces, provide products meeting code requirements and testing identical products per ASTM C 1028 for the following:

Tetra Tech 200-15704-17001

CERAMIC TILING 09 30 13 - 1

ADDITION AND RENOVATIONS

1. Level Surfaces: Minimum 0.70.

1.5 PREINSTALLATION MEETINGS

- A. Preinstallation Conference: Conduct conference at Project site.
 - 1. Review requirements in ANSI A108.01 for substrates and for preparation by other trades.
 - 2. Meeting Agenda includes but is not limited to;
 - a. Tile and installation material compatibility.
 - b. Grouting procedure.
 - c. Maintenance and cleaning products and methods.
 - d. Surface preparation.

1.6 ACTION SUBMITTALS

- A. Product Data: For each type of product.
 - 1. Ceramic tile.
 - 2. Marble thresholds.
 - 3. Setting Materials
 - 4. Grout
 - 5. Waterproofing membrane
 - 6. Crack solation membrane
 - 7. Primer.
 - 8. Self-leveling underlayment
 - 9. Patching compounds.
 - 10. Tile cleaner.
- B. Shop Drawings: Show locations of each type of tile and tile pattern. Show widths, details, and locations of expansion, control, and isolation joints in tile substrates, finished tile surfaces, locations of of floor drains including sloped slab locations, and marble threshold locations.
- C. Samples for Initial Selection: For tile, grout, and accessories involving color selection.
- D. Samples for Verification:



- Full-size units of each type and composition of tile and for each color and finish required. For ceramic mosaic tile in color blend patterns, provide full sheets of each color blend. Assembled samples mounted on a rigid panel, with grouted joints, for each type and composition of tile and for each color and finish required. Make samples at least 12 inches square, but not fewer than four tiles. Use grout of type and in color or colors approved for completed Work.
- 3. Full-size units of each type of trim and accessory for each color and finish required.
- 4. Stone thresholds in 6-inch lengths.

1.7 INFORMATIONAL SUBMITTALS

- A. Qualification Data: For Installer.
- B. Master Grade Certificates: For each shipment, type, and composition of tile, signed by tile manufacturer and Installer.
- C. Product Certificates: For each type of product.

1.8 MAINTENANCE MATERIAL SUBMITTALS

- A. Furnish extra materials that match and are from same production runs as products i stalled and that are packaged with protective covering for storage and identified with tables describing contents.
 - 1. Tile and Trim Units: Furnish quantity of full-size units equa to 3 percent of amount installed for each type, composition, color, pattern, and size indicated.
 - 2. Grout: Furnish quantity of grout equal to 3 percent of amount installed for each type, composition, and color indicated.

1.9 QUALITY ASSURANCE

- A. Mockups: Build mockups to verify selections made under Sample submittals and to demonstrate aesthetic effects and set quality standards for materials and execution.
 - 1. Build mockup of each type of floor tile installation.
 - 2. Build mockup of each type of valitile installation.
 - 3. Size: 3'-0" x 3'-0" on alyword backing.
- 1.10 DELIVERY, STORANY, AND HANDLING
 - A. Deliver and store rackaged materials in original containers with seals unbroken and labels intact until time of user Comply with requirements in ANSI A137.1 for labeling tile packages.
 - B. Deliver and store materials on site at least 24 hours before work begins in a heated and dry storage facility on site.

Store tile and cementitious materials on elevated platforms, under cover, and in a dry location.

Store aggregates where grading and other required characteristics can be maintained and contamination can be avoided.

Store liquid materials in unopened containers and protected from freezing.

F. Handle tile that has temporary protective coating on exposed surfaces to prevent coated surfaces from contacting backs or edges of other units. If coating does contact bonding surfaces of tile, remove coating from bonding surfaces before setting tile.

Tetra Tech 200-15704-17001

CERAMIC TILING 09 30 13 - 3

1.11 FIELD CONDITIONS

A. Environmental Limitations: Do not install tile until construction in spaces is complete and ambient temperature and humidity conditions are maintained at the levels indicated in referenced standards and manufacturer's written instructions.

PART 2 - PRODUCTS

2.1 MANUFACTURERS

- h from one source or
- A. Source Limitations for Tile: Obtain tile of each type and color or finish from one source or producer.
 - 1. Obtain tile of each type and color or finish from same production ion and of consistent quality in appearance and physical properties for each contigious area.
- B. Source Limitations for Setting and Grouting Materials: Obtain ingredients of a uniform quality for each mortar, adhesive, and grout component from one manufacturer and each aggregate from one source or producer.
- C. Source Limitations for Other Products: Obtain each of the following products specified in this Section from a single manufacturer for each product:
 - 1. Stone thresholds.
 - 2. Waterproof membrane.
 - 3. Crack isolation membrane

2.2 PRODUCTS, GENERAL

- A. ANSI Ceramic Tile Sandard: Provide tile that complies with ANSI A137.1 for types, compositions, and other characteristics indicated.
 - 1. Provide the complying with Standard grade requirements unless otherwise indicated.
- B. ANSI Standards for Tile Installation Materials: Provide materials complying with ANSI A108.02, ANSI standards referenced in other Part 2 articles, ANSI standards referenced by TCNA installation methods specified in tile installation schedules, and other requirements specified.

ISO 13007 Standards for Ceramic Tiles, Grouts and Adhesives: Provide materials complying with ISO 13007-1, 13007-2, 13007-3, 13007-4.

Factory Blending: For tile exhibiting color variations within ranges, blend tile in factory and package so tile units taken from one package show same range in colors as those taken from other packages and match approved Samples.

E. Mounting: For factory-mounted tile, provide back- or edge-mounted tile assemblies as standard with manufacturer unless otherwise indicated.

CERAMIC TILING 09 30 13 - 4

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

1. Where tile is indicated for installation in wet areas, do not use back- or edge-mounted tile assemblies unless tile manufacturer specifies in writing that this type of mounting is suitable for installation indicated and has a record of successful in-service performance.

2.3 TILE PRODUCTS

- A. Tile Type CT-1: Porcelain Tile
 - 1. Basis-of-Design Product: Subject to compliance with requirements, provide State Tile; See Finish Schedule, Spec Section 09 00 00.
 - a. CT-1: Porcelain Tile w/Mix Color Glaze.
 - 2. Module Size: 12" x 24".
 - 3. Thickness: 3/8 inch.
 - 4. Face: rectified edges.
 - 5. Finish: Natural.
 - 6. Tile Color and Pattern: See Finish Schedule. Section 0. 00 0
 - 7. Grout Color: See Finish Schedule Section 09 00.
 - 8. Trim Units: Coordinated with sizes and cruising of adjoining flat tile where applicable and matching characteristics of adjoining flat tile. Provide shapes as follows, Retain shape requirements from options in supparagraphs below that suit installation methods. Revise or supplement subparagraphs to sup Project.
 - a. Internal Corners: Field-batted square corners. For coved base and cap use angle pieces designed to fit with suetcher shapes.

B. Ceramic Tile Type CT-2:

- 1. Basis-of-Design Pipduce Subject to compliance with requirements, provide Garden State Tile; See Finsh Schedule, Spec Section 09 00 00.
 - a. **CI-2** Modern Dimensions.
 - Module Lize: 12" X 24".
- Module Lize: 12" X
 Theknes: 3/8 inch.
- . Vace: Rectified Edges.
- Surface: Semi-polished.
 - Tile Color and Pattern: See Finish Schedule. Section 09 00 00:
 - Grout Color: See Finish Schedule Section 09 00 00.

THRESHOLDS

A. General: Fabricate to sizes and profiles indicated or required to provide transition between adjacent floor finishes.

- 1. Bevel edges at 1:2 slope, with lower edge of bevel aligned with or up to 1/16 inch above adjacent floor surface. Finish bevel to match top surface of threshold. Limit height of threshold to 1/2 inch or less above adjacent floor surface.
- B. Marble Thresholds: ASTM C 503, with a minimum abrasion resistance of 12 per ASTM C 1353 or ASTM C 241 and with honed finish.
 - 1. Description: As selected by Architect from manufacturer's full range.

2.5 SURFACE PREPARATION MATERIALS

- A. General: Manufacturer's standard product, selected from the following, and is recommended by the manufacturer for the application indicated. Include reinforce and accessories recommended by manufacturer.
- B. Reduced-preparation, self-leveling underlayment: for smoothing the repairing interior floors before the installation of floor coverings from feather edge to up to 2
 - 1. Products: Subject to compliance with requirements, provide the following:
 - a. MAPEI Corporation; Ultraplan Easy
 - 1) Requires primer MAPEI-Corporation; Primer T
- C. Cementitious Patching Compound;
 - 1. Products: Subject to compliance with requirements, provide the following:
 - a. MAPEI Corporation, Monecem Quickpatch
- 2.6 WATERPROOF MEMOLAN

a.

- A. General: Manufacturer's chandard product, selected from the following, that complies with ANSI A118.16 and is recommended by the manufacturer for the application indicated. Include reinforcement and accessories recommended by manufacturer.
- B. Fluid Applied Membrane: Liquid-latex rubber or elastomeric polymer.

Products: Subject to compliance with requirements, provide the following:

MAPEI Corporation; Mapelastic HPG

CRACK ISOLATION MEMBRANE

- A. General: Manufacturer's standard product, selected from the following, that complies with ANSI A118.10 and is recommended by the manufacturer for the application indicated. Include reinforcement and accessories recommended by manufacturer.
- B. Fabric-Reinforced, Fluid-Applied Membrane: System consisting of liquid-latex rubber or elastomeric polymer and fabric reinforcement.

CERAMIC TILING 09 30 13 - 6

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

1. Products: Subject to compliance with requirements, provide the following:

MAPEI Corporation; Mapelastic HPG with MAPEI Fiberglass Mesh.

2.8 SETTING MATERIALS

- A. Non-sag, medium-bed and thin-set, Polymer modified single component mortar: ANSI A18.4, A118.11 and ISO 13007 C2TES1P1;
 - Basis-of-Design Product: Subject to compliance with requirements, provide N Corporation; Ultraflex LFT or comparable product by one of the following:
 a. Laticrete International, Inc.
 - b. TEC; a subsidiary of H. B. Fuller Company.

2.9 GROUT MATERIALS

- A. High-Performance, fast-setting, sanded polymer-modified tile grout; ANSI A118.7.
 - 1. Basis of Design Product: Subject to compliance with requirements, provide MAPEI Ultracolor Plus.
 - 2. Colors: See Finish Schedule, Section 09 00 00

2.10 MISCELLANEOUS MATERIALS

- A. Tile Cleaner: A neutral cleaner capable of removing soil and residue without harming tile and grout surfaces, specifically approved for materials and installations indicated by tile and grout manufacturers.
 - 1. Basis of Design product: Subject to compliance with requirements, provide MAPEI UltraCare Concernated Tile & Grout Cleaner.
- 2.11 MIXING MORYARS AND GROUT
 - A. Mix mostars and grouts to comply with referenced standards and mortar and grout manufacturers written instructions.
 - Add materials, water, and additives in accurate proportions.

Obtain and use type of mixing equipment, mixer speeds, mixing containers, mixing time, and other procedures to produce mortars and grouts of uniform quality with optimum performance characteristics for installations indicated.

ADDITION AND RENOVATIONS

PART 3 - EXECUTION

3.1 **EXAMINATION**

- Examine substrates, areas, and conditions where tile will be installed, with Installer present, for A. compliance with requirements for installation tolerances and other conditions affecting performance of the Work.
 - Verify that substrates for setting tile are firm; dry; clean; free of coatings the 1. incompatible with tile-setting materials, including curing compounds and other substances that contain soap, wax, oil, or silicone; and comply with fathe s telerances required by ANSI A108.01 for installations indicated.
 - 2. Verify that concrete substrates for tile floors installed with thing the comply with surface finish requirements in ANSI A108.01 for installations indicated.
 - Verify that surfaces that received a steel trowel final have been mechanically a. scarified.
 - Verify that protrusions, bumps, and ridges have been removed by sanding or b. grinding.
 - 3. Verify that installation of grounds, anchor, recessed frames, electrical and mechanical
 - units of work, and similar items located in or behind tile has been completed. Verify that joints and cracks in tile sub traces are coordinated with tile joint locations; if not coordinated, adjust joint location in acquilation with Architect. 4.
- Proceed with installation only after. tist ctory conditions have been corrected. B.
- C. Beginning installation constitu Contractor's acceptance of substrates and conditions.
- 3.2 PREPARATION
 - Fill cracks, hole and depressions in concrete substrates for tile floors installed with thinset mortar with tow-table leveling and patching compound specifically recommended by tile-A. setting material hanufacturer.
 - Where indicated, prepare substrates to receive waterproofing by applying a reinforced mortar B. bee the complies with ANSI A108.1A and is sloped 1/4 inch per foot toward drains.



Blending: For tile exhibiting color variations, verify that tile has been factory blended and ckaged so tile units taken from one package show same range of colors as those taken from other packages and match approved Samples. If not factory blended, either return to manufacturer or blend tiles at Project site before installing.

CERAMIC TILE INSTALLATION

A. Comply with TCNA's "Handbook for Ceramic, Glass, and Stone Tile Installation" for TCNA installation methods specified in tile installation schedules. Comply with parts of the ANSI A108 series "Specifications for Installation of Ceramic Tile" that are referenced in TCNA

CERAMIC TILING 09 30 13 - 8

installation methods, specified in tile installation schedules, and apply to types of setting and grouting materials used.

- 1. For the following installations, follow procedures in the ANSI A108 series of tile installation standards for providing 95 percent mortar coverage:
 - a. Tile floors in wet areas.
- B. Extend tile work into recesses and under or behind equipment and fixtures to form covering without interruptions unless otherwise indicated. Terminate work near obstructions, edges, and corners without disrupting pattern or joint alignments.
- C. Accurately form intersections and returns. Perform cutting and drilling of tile without marring visible surfaces. Carefully grind cut edges of tile abutting trim, finish, o built-in items for straight aligned joints. Fit tile closely to electrical outlets, piping aixtures, and other penetrations so plates, collars, or covers overlap tile.
- D. Provide manufacturer's standard trim shapes where necessary to enquine te exposed tile edges.
- E. Where accent tile differs in thickness from field tile, very setting-bed thickness so that tiles are flush.
- F. Jointing Pattern: Lay tile in grid pattern unless otherwise indicated. Lay out tile work and center tile fields in both directions in each space of an ach wall area. Lay out tile work to minimize the use of pieces that are less than half of tile. Provide uniform joint widths unless otherwise indicated.
 - 1. For tile mounted in sheets, make points between tile sheets same width as joints within tile sheets so joints between thete are not apparent in finished work.
 - 2. Where adjoining tiles in floor, base, walls, or trim are specified or indicated to be same size, align joints.
 - 3. Where tiles are specified or indicated to be whole integer multiples of adjoining tiles on floor, base, walls or t.im, align joints unless otherwise indicated.
- G. Joint Widths: Unless otherwise indicated, install tile with the following joint widths:
 - 1. Colorbody Porcelain Tile: 1/8".
 - a. Must Meet Manufactures recommendations.
- H. Lay out tile wainscots to dimensions indicated or to next full tile beyond dimensions indicated.

Appansion Joints: Provide expansion joints and other sealant-filled joints, including control, contraction, and isolation joints, where indicated. Form joints during installation of setting materials, mortar beds, and tile. Do not saw-cut joints after installing tiles.

- 1. Where joints occur in concrete substrates, locate joints in tile surfaces directly above them.
- J. Stone Thresholds: Install stone thresholds in same type of setting bed as adjacent floor unless otherwise indicated.

- 1. At locations where mortar bed (thickset) would otherwise be exposed above adjacent floor finishes, set thresholds in latex-portland cement mortar (thin set).
- 2. Do not extend cleavage membrane] waterproofing under thresholds set in latex-portland cement mortar. Fill joints between such thresholds and adjoining tile set on cleavage membrane waterproofing with elastomeric sealant.

3.4 WATERPROOFING INSTALLATION

- A. Install waterproofing to comply with ANSI A108.13 and manufacturer's written in aructions produce waterproof membrane of uniform thickness that is bonded securely to substance.
- B. Allow waterproofing to cure and verify by testing that it is watertight before installing tile or setting materials over it.

3.5 CRACK ISOLATION MEMBRANE INSTALLATION

- A. Install crack isolation membrane to comply with ANSIAP8.17 and manufacturer's written instructions to produce membrane of uniform thickness that p bonded securely to substrate.
- B. Allow crack isolation membrane to cure before installing tile or setting materials over it.

3.6 ADJUSTING AND CLEANING

- A. Remove and replace tile that is damaged or that does not match adjoining tile. Provide new matching units, installed as specified and in a manner to eliminate evidence of replacement.
- B. Cleaning: On completion of placement and grouting, clean all ceramic tile surfaces so they are free of foreign matter.
 - 1. Remove grout revidue from tile as soon as possible.
 - 2. Clean grout smean and haze from tile according to tile and grout manufacturer's written instructions out no sooner than 10 days after installation. Use only cleaners recommended by tile an grout manufacturers and only after determining that cleaners are safe to use by terting on samples of tile and other surfaces to be cleaned. Protect metal surfaces and plumbing fixtures from effects of cleaning. Flush surfaces with clean water before and other cleaning.

PROTECTION

Protect installed tile work with kraft paper or other heavy covering during construction period to prevent staining, damage, and wear. If recommended by tile manufacturer, apply coat of neutral protective cleaner to completed tile walls and floors.

- B. Prohibit foot and wheel traffic from tiled floors for at least seven days after grouting is completed.
- C. Before final inspection, remove protective coverings and rinse neutral protective cleaner from tile surfaces.

CERAMIC TILING 09 30 13 - 10

3.8 INTERIOR CERAMIC TILE INSTALLATION SCHEDULE

- A. Interior Floor Installations, New Concrete Subfloor:
 - 1. Ceramic Tile Installation TCNA F122-14; Thin-set mortar on waterproofing membrane.
 - a. Ceramic Tile Type: CT-1.
 - b. Setting Bed: Premium non-sag, medium-bed and thin-set, Polymer modified single component mortar.
 - c. Grout: High-Performance, fast-setting, sanded polymer-modified tile group
 - d. Crack Isolation Membrane
 - e. Waterproofing Membrane
- B. Interior Floor Installation Existing Concrete Subfloor
 - 1. Ceramic Tile Installation TCNA F205-14 and F205-14A) On Ground Concrete, Cementitious Self Leveling Underlayment Ceramic Tile, with Vaterproofing membrane.
 - a. Ceramic Tile Type: CT-1.
 - b. Setting Bed: Premium non-sag, medium-bed and thin-set, Polymer modified single component mortar.
 - c. Grout: High-Performance, fast-setting, sanded polymer-modified tile grout.
 - d. Crack Isolation Membrane
 - e. Waterproofing Membrane
- C. Interior Wall Installations, Wood or Metal Suds or Furring:
 - 1. Ceramic Tile Installation TCNA W244C-14: Thin-set mortar over Cement Backer Board
 - a. Ceramic Tile Type: C1-2, CT-3.
 - b. Setting Bed: Premium non-sag, medium-bed and thin-set, Polymer modified single component nortal
 - c. Grout: His -Performance, fast-setting, sanded polymer-modified tile grout.

END OF SECTION


SECTION 09 51 13

ACOUSTICAL PANEL CEILINGS

PART 1 - GENERAL

1.1 **RELATED DOCUMENTS**

A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.

1.2 **SUMMARY**

- Section includes acoustical panels and exposed suspension system interior ceilings. A.
- Β. **Related Requirements:** Section 07 92 00 "Joint Sealants" for acoustical 1.
- PREINSTALLATION MEETINGS 1.3
 - Preinstallation Conference: Conduct conference as Project site. A.

ACTION SUBMITTALS 1.4

- Product Data: For each type of A.
 - coustical panel ceiling type ACT-1. 1. Acoustical pan
 - 2. Attachment devic
 - Carrying chinnels. 3.
 - Wire harvers, braces, and ties. Heid-do yn clips. 4.
 - 5.
 - Roll-formed, sheet-metal edge moldings and trim. 6.
 - Extruced-aluminum soffit and perimeter moldings and trim. 7.
 - Samples. For each exposed product and for each color and texture specified, 6 inches in size.

Samples for Initial Selection: For components with factory-applied finishes.

Samples for Verification: For each component indicated and for each exposed finish required, prepared on Samples of sizes indicated below:

- 1. Acoustical Panels: Set of 6-inch square Samples of each type, color, pattern, and texture.
- 2. Exposed Suspension-System Members, Moldings, and Trim: Set of 6-inch long samples of each type, finish, and color.
- Clips: Full-size hold-down clips. 3.

ADDITION AND RENOVATIONS

1.5 INFORMATIONAL SUBMITTALS

- A. Coordination Drawings: Reflected ceiling plans, drawn to scale, on which the following items are shown and coordinated with each other, using input from installers of the items involved:
 - 1. Ceiling suspension-system members.
 - 2. Structural members to which suspension systems will be attached.
 - 3. Method of attaching hangers to building structure.
 - 4. Carrying channels or other supplemental support for hanger-wire attachment v conditions do not permit installation of hanger wires at required spacing.
 - 5. Size and location of initial access modules for acoustical panels.
 - 6. Items penetrating finished ceiling and ceiling-mounted items including the to low
 - a. Lighting fixtures.
 - b. Diffusers.
 - c. Grilles.
 - d. Speakers.
 - e. Sprinklers.
 - f. Access panels.
 - g. Perimeter moldings.
- B. Qualification Data: For manufacturer and installe
 - 1. Submit documentation or certification of each requirement.
- C. Product Test Reports: For each acoustical panel ceiling, for tests performed by manufacturer and witnessed by a qualified testing age cy
- D. Evaluation Reports: For each constitute panel ceiling suspension system from ICC-ES.
- E. Field quality-control reports
- 1.6 CLOSEOUT SUPMITTAL
 - A. Maintenance Data: For finishes to include in maintenance manuals.
- 1.7 MAINTENANCE MATERIAL SUBMITTALS

Furnish extra materials that match products installed and that are packaged with protective overing for storage and identified with labels describing contents.

- 1. Acoustical Ceiling Units: Full-size panels equal to 2 percent of quantity installed.
- 2. Suspension-System Components: Quantity of each exposed component equal to 2 percent of quantity installed.
- 3. Hold-Down Clips: Equal to 2 percent of quantity installed.
- 1.8 QUALITY ASSURANCE
 - A. Manufacturer Qualifications:

- 1. Manufacturer member in good standing of CISCA (Ceiling and Interior Systems Construction Association).
- 2. Single-Source Responsibility: Provide acoustical panel units and grid components by a single manufacturer.
- B. Installer Qualifications:
 - 1. At least 10 completed projects of similar size and scope.
 - 2. Installation to meet ASTM C636.

1.9 DELIVERY, STORAGE, AND HANDLING

- A. Deliver acoustical panels, suspension-system components, and accessories to Deject site and store them in a fully enclosed, conditioned space where they will be protected against damage from moisture, humidity, temperature extremes, direct sunlight surface contamination, and other causes.
- B. Before installing acoustical panels, permit them to reach room temperature and a stabilized moisture content.
- C. Handle acoustical panels carefully to avoid chipping edges or damaging units in any way.

1.10 FIELD CONDITIONS

A. Environmental Limitations: Do not instan acoustical panel ceilings until spaces are enclosed and weathertight, wet-work in spaces is complete and dry, work above ceilings is complete, and ambient temperature and humiling conditions are maintained at the levels indicated for Project when occupied for its intended use.

PART 2 - PRODUCTS

- 2.1 MANUFACTUTERS
 - A. Source Limitations: Obtain each type of acoustical ceiling panel and its supporting suspension system from single source from single manufacturer.

PERFORMANCE REQUIREMENTS

Surface-Burning Characteristics: Comply with ASTM E 84; testing by a qualified testing agency. Identify products with appropriate markings of applicable testing agency.

- 1. Flame-Spread Index: Class A according to ASTM E 1264.
- 2. Smoke-Developed Index: 50 or less.

ADDITION AND RENOVATIONS

2.3 ACOUSTICAL PANELS - ACT-1 (To coordinate with Drawings)

- A. Products: Subject to compliance with requirements, available products that may be incorporated into the Work include, but are not limited to, the following:
 - 1. Armstrong World Industries, Inc.; Ultima, 1910.
 - 2. USG Interiors, Inc.; Mars Panels 86185
 - 3. CertainTeed; Symphony M
- B. Acoustical Panel Standard: Provide manufacturer's standard panels according to ASTM E 224 and designated by type, form, pattern, acoustical rating, and light reflectance unless otherwise indicated.
- C. Classification: Provide panels as follows:
 - 1. Type and Form: TypeIV, mineral base with painted finish, Form water felted with vinyl overlay on face, back, and sealed edges.
 - 2. Pattern: as indicated by manufacturer's designation.
- D. Color: White.
- E. Light Reflectance (LR): Not less than 0.85.
- F. Ceiling Attenuation Class (CAC): Not less than
- G. Noise Reduction Coefficient (NRC): No. les, than 0.70.
- H. Edge/Joint Detail: Square
- I. Thickness: 3/4 inch.

2.4

- J. Modular Size: 24 by 2. mehes
- K. Antimicrobial Treatment of Anufacturer's standard broad spectrum, antimicrobial formulation that inhibits fingue, mold, mildew, and gram-positive and gram-negative bacteria and showing no mold, mildew, or bacterial growth when tested according to ASTM D 3273, ASTM D 3274, or ASTM G 21 and evaluated according to ASTM D 3274 or ASTM G 21.
 - METAL SUSPENSION SYSTEM FOR ACT-1 (To coordinate with Drawings)
 - roducts: Subject to compliance with requirements, available products that may be incorporated into the Work include, but are not limited to, the following:
 - 1. Armstrong World Industries, Inc.; Prelude XL 15/16 inch Exposed Tee.
 - 2. USG Interiors, Inc.; Donn Brand DX Exposed 15/16 inch Exposed Tee
 - 3. Chicago Metallic Corporation; Snap Grid 200 15/16 inch Exposed Tee
 - 4. CertainTeed Ceilings; Classic Stab 15/16 inch Exposed Tee
- B. Wide-Face, Capped, Double-Web, Steel Suspension System: Main and cross runners roll formed from cold-rolled steel sheet; prepainted, electrolytically zinc coated, or hot-dip galvanized, G30 coating designation; with prefinished 15/16-inch wide metal caps on flanges.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- 1. Structural Classification: Intermediate-duty system.
- 2. End Condition of Cross Runners: Override (stepped) or butt-edge type.
- 3. Face Design: Flat, flush.
- 4. Cap Material: Cold-rolled steel.
- 5. Cap Finish: Painted white.

2.5 ACCESSORIES

- A. Attachment Devices: Size for five times the design load indicated in ASTM C (5)C Table 1, "Direct Hung," unless otherwise indicated.
 - 1. Anchors in Concrete: Anchors of type and material indicated below, with holes or loops for attaching hangers of type indicated and with capability to succan without failure, a load equal to five times that imposed by ceiling construction, as determined by testing according to ASTM E 488/E 488M or ASTM E 1512 as applicable, conducted by a qualified testing and inspecting agency.
 - a. Type: Cast-in-place, Postinstalled expansion, Postinstalled bonded anchors.
 - b. Corrosion Protection: Carbon-steel components zinc plated according to ASTM B 633, Class SC 1 (mild) service condition.
 - 2. Power-Actuated Fasteners in Concrete: Fasteners system of type suitable for application indicated, fabricated from corrosion-resistant materials, with clips or other accessory devices for attaching hangers of type incicated and with capability to sustain, without failure, a load equal to 10 times that imposed by ceiling construction, as determined by testing according to ASTM2 1900 conducted by a qualified testing and inspecting agency.
- B. Wire Hangers, Braces, and Tes: Provide wires as follows:
 - 1. Zinc-Coated, Carbon-Seel Wire: ASTM A 641/A 641M, Class 1 zinc coating, soft temper.
 - 2. Size: Wire diameter sufficient for its stress at three times hanger design load (ASTMC 6.5/C 635M, Table 1, "Direct Hung") will be less than yield stress of wire, but not less than 0.106-inch diameter wire.
- C. Hold Down Clops: Manufacturer's standard hold-down clips.

METAL EDGE MOLDINGS AND TRIM

Roll-Formed, Sheet-Metal Edge Moldings and Trim: Type and profile indicated or, if not indicated, manufacturer's standard moldings for edges and penetrations that comply with seismic design requirements; formed from sheet metal of same material, finish, and color as that used for exposed flanges of suspension-system runners.

1. Edge moldings shall fit acoustical panel edge details and suspension systems indicated and match width and configuration of exposed runners unless otherwise indicated.

- 2. For lay-in panels with reveal edge details, provide stepped edge molding that forms reveal of same depth and width as that formed between edge of panel and flange at exposed suspension member.
- B. Extruded-Aluminum Edge Moldings and Trim: Where indicated, provide manufacturer's extruded-aluminum edge moldings and trim of profile indicated or referenced by manufacturer's designations, including splice plates, corner pieces, and attachment and other clips, complying with seismic design requirements.
 - 1. Baked-Enamel or Powder-Coat Finish: Minimum dry film thickness of 1.5 nms. Courty with ASTM C 635/C 635M and coating manufacturer's written instructions in cleaning, conversion coating, and applying and baking finish.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine substrates, areas, and conditions, including atructural framing to which acoustical panel ceilings attach or abut, with Installer present, for compliance with requirements specified in this and other Sections that affect ceiling installation and anchorage and with requirements for installation tolerances and other conditions affecting performance of acoustical panel ceilings.
- B. Examine acoustical panels before installation. Reject acoustical panels that are wet, moisture damaged, or mold damaged.
- C. Proceed with installation only and uncalesfactory conditions have been corrected.
- D. Beginning installation constitutes Contractor's acceptance of substrates and conditions.

3.2 PREPARATION

- A. Measure each or ting area and establish layout of acoustical panels to balance border widths at opposite edges of each ceiling. Avoid using less-than-half-width panels at borders unless otherwise indicated, and comply with layout shown on reflected ceiling plans.
- B. Layor openings for penetrations centered on the penetrating items.

INSTALLATION

Install acoustical panel ceilings according to ASTM C 636/C 636M, and manufacturer's written instructions.

- B. Suspend ceiling hangers from building's structural members and as follows:
 - 1. Install hangers plumb and free from contact with insulation or other objects within ceiling plenum that are not part of supporting structure or of ceiling suspension system.

- 2. Where width of ducts and other construction within ceiling plenum produces hanger spacings that interfere with location of hangers at spacings required to support standard suspension-system members, install supplemental suspension members and hangers in form of trapezes or equivalent devices.
- 3. Secure wire hangers to ceiling-suspension members and to supports above with a minimum of three tight turns. Connect hangers directly to structure or to inserts, every screws, or other devices that are secure and appropriate for substrate and that will not deteriorate or otherwise fail due to age, corrosion, or elevated temperatures.
- 4. Do not support ceilings directly from permanent metal forms or floor deck. Esten hangers to cast-in-place hanger inserts, postinstalled mechanical or adhesive anchor or power-actuated fasteners that extend through forms into concrete.
- 5. When steel framing does not permit installation of hanger wires at spacing required, install carrying channels or other supplemental support for attachment of hanger wires.
- 6. Do not attach hangers to steel deck tabs.
- 7. Do not attach hangers to steel roof deck. Attach hangers to structural members.
- Space hangers not more than 48 inches o.c. along each member supported directly from hangers unless otherwise indicated; provide hangers not nore than 8 inches from ends of each member.
- 9. Size supplemental suspension members and hangers to support ceiling loads within performance limits established by referenced standards.
- C. Secure bracing wires to ceiling suspension members and to supports with a minimum of four tight turns. Suspend bracing from building's structural rembers as required for hangers, without attaching to permanent metal forms, steel deck, so seel deck tabs. Fasten bracing wires into concrete with cast-in-place or postinstalleconclus.
- D. Install edge moldings and trim of type indicated at perimeter of acoustical ceiling area and where necessary to conceal edges of acoustical panels.
 - 1. Apply acoustical sealast in a continuous ribbon concealed on back of vertical legs of moldings before they are hestalled.
 - 2. Screw attach nolding to substrate at intervals not more than 16 inches o.c. and not more than 3 inches from ends. Miter corners accurately and connect securely.
 - 3. Do not use exposed fasteners, including pop rivets, on moldings and trim.
- E. Install supervious system runners so they are square and securely interlocked with one another. Remove and replace dented, bent, or kinked members.
- F. Install acoustical panels with undamaged edges and fit accurately into suspension-system runners and edge moldings. Scribe and cut panels at borders and penetrations to provide precise
- fit. 1. 2.
- For square-edged panels, install panels with edges fully hidden from view by flanges of suspension-system runners and moldings.
- For reveal-edged panels on suspension-system runners, install panels with bottom of reveal in firm contact with top surface of runner flanges.
- 3. For reveal-edged panels on suspension-system members with box-shaped flanges, install panels with reveal surfaces in firm contact with suspension-system surfaces and panel faces flush with bottom face of runners.

- 4. Paint cut edges of panel remaining exposed after installation; match color of exposed panel surfaces using coating recommended in writing for this purpose by acoustical panel manufacturer.
- 5. Install hold-down clips; space according to panel manufacturer's written instructions unless otherwise indicated.
 - a. Hold-Down Clips: Space 24 inches o.c. on all cross runners.

3.4 FIELD QUALITY CONTROL

- A. Special Inspections: Owner will engage a qualified special inspector to perform the Chowing special inspections:
 - 1. Periodic inspection during the installation of suspended ceilling grids according to ASCE/SEI 7.

3.5 CLEANING

- A. Clean exposed surfaces of acoustical panel ceilings, including trim, edge moldings, and suspension-system members. Comply with manufacturer's written instructions for cleaning and touchup of minor finish damage.
- B. Remove and replace ceiling components that cannot be successfully cleaned and repaired to permanently eliminate evidence of damage.

SECTION

SECTION 09 65 13

RESILIENT BASE AND ACCESSORIES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Thermoset-rubber base.
 - 2. Rubber molding accessories.

1.3 ACTION SUBMITTALS

- A. Product Data: For each type of product.
- B. Samples for Initial Selection: For each type of product indicated.
- C. Samples for Verification: For each type of product indicated and for each color, texture, and pattern required in manufacture's standard-size Samples.

1.4 MAINTENANCE MATTPIAL SUBMITTALS

- A. Furnish extra locertais, from the same product run, that match products installed and that are packaged with protective covering for storage and identified with labels describing contents.
 - 1. Eurnich not less than 10 linear feet for every 500 linear feet or fraction thereof, of each type, color, pattern, and size of resilient product installed.



DELIVERY, STORAGE, AND HANDLING

Store resilient products and installation materials in dry spaces protected from the weather, with ambient temperatures maintained within range recommended by manufacturer, but not less than 50 deg F or more than 90 deg F.

1.6 FIELD CONDITIONS

- A. Maintain ambient temperatures within range recommended by manufacturer, but not less than 70 deg F or more than 95 deg F, in spaces to receive resilient products during the following periods:
 - 1. 48 hours before installation.
 - 2. During installation.
 - 3. 48 hours after installation.
- B. After installation and until Substantial Completion, maintain ambient temperatures vianin range recommended by manufacturer, but not less than 55 deg F or more than 95 deg F.
- C. Install resilient products after other finishing operations, including panting, have been completed.

PART 2 - PRODUCTS

- 2.1 THERMOSET-RUBBER BASE: RB-1 & RB-2: See Finen Schedule Spec Section for Basis of Design.
 - A. Product Standard: ASTM F 1861, Type TS (rubber vulcanized thermoset), Group I (solid, homogeneous).
 - 1. Style and Location:
 - a. Style B, Cove: .
 - B. Thickness: 0.125 inch.
 - C. Height: 4 inches As indicated on Drawings.
 - D. Lengths: Coil in hanufacturer's standard length.
 - E. Outside corner: Preformed.
 - F. Inside Corners: Preformed.
 - G. Colors: Per Finish Schedule-Spec Section 09 00 00.

RUBBER MOLDING ACCESSORY

- Description: Rubber carpet edge for glue-down applications, transition strips.
- B. Profile and Dimensions: As indicated on Finish Schedule Section 09 00 00.
- C. Locations: Provide rubber molding accessories in areas indicated.
- D. Colors and Patterns: See Finish Schedule Spec Section 09 00 00.

2.3 INSTALLATION MATERIALS

- A. Trowelable Leveling and Patching Compounds: Latex-modified, portland-cement-based or blended hydraulic-cement-based formulation provided or approved by resilient-product manufacturer for applications indicated.
- B. Adhesives: Water-resistant type recommended by resilient-product manufacturer for resilient products and substrate conditions indicated.
- C. Floor Polish: Provide protective, liquid floor-polish products recommended by realist tread manufacturer.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine substrates, with Installer present, for compliance with equirements for maximum moisture content and other conditions affecting performance of the Work.
 - 1. Verify that finishes of substrates comply with tolerances and other requirements specified in other Sections and that substrates are free of cracks, ridges, depressions, scale, and foreign deposits that might interfere with a thesian of resilient products.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.
 - 1. Installation of resilient products indicates acceptance of surfaces and conditions.

3.2 PREPARATION

- A. Prepare substrates according to manufacturer's written instructions to ensure adhesion of resilient products
- B. Concrete Subscrites for Resilient Stair Accessories: Prepare horizontal surfaces according to ASTM F/10.
 - Yerity that substrates are dry and free of curing compounds, sealers, and hardeners.



- Remove substrate coatings and other substances that are incompatible with adhesives and that contain soap, wax, oil, or silicone, using mechanical methods recommended by manufacturer. Do not use solvents.
- Alkalinity and Adhesion Testing: Perform tests recommended by manufacturer. Proceed with installation only after substrate alkalinity falls within range on pH scale recommended by manufacturer in writing, but not less than 5 or more than 9 pH.
- Moisture Testing: Perform tests so that each test area does not exceed 200 sq. ft., and perform no fewer than three tests in each installation area and with test areas evenly spaced in installation areas.
 - a. Anhydrous Calcium Chloride Test: ASTM F 1869. Proceed with installation only after substrates have maximum moisture-vapor-emission rate of 3 lb of water/1000 sq. ft. in 24 hours.

- b. Relative Humidity Test: Using in-situ probes, ASTM F 2170. Proceed with installation only after substrates have a maximum 75 percent relative humidity level measurement.
- C. Fill cracks, holes, and depressions in substrates with trowelable leveling and patching compound; remove bumps and ridges to produce a uniform and smooth substrate.
- D. Do not install resilient products until materials are the same temperature as space where they a to be installed.
 - 1. At least 48 hours in advance of installation, move resilient products and installation materials into spaces where they will be installed.
- E. Immediately before installation, sweep and vacuum clean substrates to be covered by resilient products.

3.3 RESILIENT BASE INSTALLATION

- A. Comply with manufacturer's written instructions for installing resident base.
- B. Apply resilient base to walls, columns, pilasters, enework and cabinets in toe spaces, and other permanent fixtures in rooms and areas where base is acquired.
- C. Install resilient base in lengths as long contractical without gaps at seams and with tops of adjacent pieces aligned.
- D. Tightly adhere resilient base to substrate throughout length of each piece, with base in continuous contact with horizonal and vertical substrates.
- E. Do not stretch resilient base during installation.
- F. On masonry surfaces which r similar irregular substrates, fill voids along top edge of resilient base with manufacturer's accommended adhesive filler material.
- G. Preformed Corners: Install preformed corners before installing straight pieces.
- 3.4 RETILIENT ACCESSORY INSTALLATION

Comply with manufacturer's written instructions for installing resilient accessories.

Resilient Stair Accessories:

- 1. Use stair-tread-nose filler to fill nosing substrates that do not conform to tread contours.
- 2. Tightly adhere to substrates throughout length of each piece.
- 3. For treads installed as separate, equal-length units, install to produce a flush joint between units.
- C. Resilient Molding Accessories: Butt to adjacent materials and tightly adhere to substrates throughout length of each piece. Install reducer strips at edges of floor covering that would otherwise be exposed.

RESILIENT BASE AND ACCESSORIES 09 65 13 - 4

3.5 CLEANING AND PROTECTION

- A. Comply with manufacturer's written instructions for cleaning and protecting resilient products.
- B. Perform the following operations immediately after completing resilient-product installation:
 - 1. Remove adhesive and other blemishes from surfaces.
 - 2. Sweep and vacuum horizontal surfaces thoroughly.

RBIDD

- 3. Damp-mop horizontal surfaces to remove marks and soil.
- C. Protect resilient products from mars, marks, indentations, and other damage from construction operations and placement of equipment and fixtures during remainder of construction period.
- D. Floor Polish: Remove soil, adhesive, and blemishes from resilient stair reads before applying liquid floor polish.
 - 1. Apply two coat(s).
- E. Cover resilient products subject to wear and foot traffic until Substantial Completion.

END OF SECTION



SECTION 09 65 19

RESILIENT TILE FLOORING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Rubber floor tile.
- 1.3 ACTION SUBMITTALS
 - A. Product Data: For each type of product.
 - B. Shop Drawings: For each type of floer ble. Include floor tile layouts, edges, columns, doorways, enclosing partitions, built-in hermiture, cabinets, and cutouts.
 - 1. Show details of special patterns.
 - C. Samples: Full-size unit of each color and pattern of floor tile required.
 - D. Samples for Initial Selection: For each type of floor tile indicated.
 - E. Samples for Verification: Full-size units of each color and pattern of floor tile required.
 - F. Product Ichedule: For floor tile. Use same designations indicated on Drawings.

1.4 INFORMATIONAL SUBMITTALS

Qualification Data: For Installer.

CLOSEOUT SUBMITTALS

A. Maintenance Data: For each type of floor tile to include in maintenance manuals.

ADDITION AND RENOVATIONS

1.6 MAINTENANCE MATERIAL SUBMITTALS

- A. Furnish extra materials that match products installed and that are packaged with protective covering for storage and identified with labels describing contents.
 - 1. Floor Tile: Furnish one box for every 50 boxes or fraction thereof, of each type, color and pattern of floor tile installed.

1.7 QUALITY ASSURANCE

- A. Installer Qualifications: A qualified installer who employs workers for this Project the are competent in techniques required by manufacturer for floor tile installation and searing method indicated.
 - 1. Engage an installer who employs workers for this Project who are academy trained or certified by floor tile manufacturer for installation techniques required.

1.8 DELIVERY, STORAGE, AND HANDLING

A. Store floor tile and installation materials in dry spaces protected from the weather, with ambient temperatures maintained within range recommendate by manufacturer, but not less than 50 deg F or more than 90 deg F. Store floor tiles on that surfaces.

1.9 FIELD CONDITIONS

- A. Maintain ambient temperatures within range recommended by manufacturer, but not less than 70 deg F or more than 95 deg F, in spaces to receive floor tile during the following time periods:
 - 1. 48 hours before ins alla ion
 - 2. During installation.
 - 3. 48 hours after installation.
- B. After installation and until Substantial Completion, maintain ambient temperatures within range recommended by manufacturer, but not less than 55 deg F or more than 95 deg F.

C. Close spaces to traffic during floor tile installation.

Close spaces to traffic for 48 hours after floor tile installation.

Install floor tile after other finishing operations, including painting, have been completed.

RT 2 - PRODUCTS

D

2.1 PERFORMANCE REQUIREMENTS

A. Fire-Test-Response Characteristics: For resilient tile flooring, as determined by testing identical products according to ASTM E 648 or NFPA 253 by a qualified testing agency.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- 1. Critical Radiant Flux Classification: Class I, not less than 0.45 W/sq. cm.
- B. Rubber Tile to have a five (5) years Wear Layer Warranty of less than .0035 wear per year.
- 2.2 RUBBER FLOOR TILE: RT-1 thru RT-10. See finish schedule and finish plan.
 - A. Manufacturer:
 - 1. Basis of Design Nora System Inc.
 - a. Nora Plan Environcare 2.0 mm.
 - b. Nora Plan Eco 2.0 mm.
 - c. Norament Saturna 2.0 mm.
 - 2. Tarkett
 - 3. Flexco
 - B. Tile Standard: ASTM F 1344, Class I-A, homogeneous rubber the, solid color, with no wear layer, no sealant or finish recommended and no poly urely are coatings.
 - C. Hardness: Manufacturer's standard hardness, neasaned using Shore, Type A durometer per ASTM D 2240.
 - D. Wearing Surface: Smooth.
 - E. Thickness: 2.0 mm.
 - F. Size: 24 by 24 inches.
 - 1. RT-10: Stair Treater Size to be coordinated with Architectural docs.
 - G. Colors and Patterns: See Finish Schedule.
- 2.3 INSTALLATION MATERIALS
 - A. Trowelable Leveling and Patching Compounds: Latex-modified, portland cement based or bleneid hydraulic-cement-based formulation provided or approved by floor tile manufacturer for applications indicated.

Achesives: Water-resistant type recommended by floor tile and adhesive manufacturers to suit floor tile and substrate conditions indicated. (Nora NTX adhesive systems.)

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine substrates, with Installer present, for compliance with requirements for maximum moisture content and other conditions affecting performance of the Work.
 - 1. Verify that finishes of substrates comply with tolerances and other requirements specified in other Sections and that substrates are free of cracks, ridges, depressions, scale, are foreign deposits that might interfere with adhesion of floor tile.
- B. Proceed with installation only after unsatisfactory conditions have been corrected

3.2 PREPARATION

- A. Prepare substrates according to floor tile manufacturer's written hetrections to ensure adhesion of resilient products.
- B. Concrete Substrates: Prepare according to ASTM F 710
 - 1. Verify that substrates are dry and free of chring compounds, sealers, and hardeners.
 - 2. Remove substrate coatings and other substances that are incompatible with adhesives and that contain soap, wax, oil, or silicone, using mechanical methods recommended by floor tile manufacturer. Do not use solvent
 - 3. Alkalinity and Adhesion Testing. Perform tests recommended by floor tile manufacturer. Proceed with installation only after substrate alkalinity falls within range on pH scale recommended by manufactures in writing, but not less than 5 or more than 10 pH.
 - 4. Moisture Testing: Proceed with installation only after substrates pass testing according to floor tile manufacturers written recommendations, but not less stringent than the following:
 - a. Perform anhydrous calcium chloride test according to ASTM F 1869. Proceed with i stal ation only after substrates have maximum moisture-vapor-emission rate of 3 lb cr water/1000 sq. ft. in 24 hours.

Perform relative humidity test using in situ probes according to ASTM F 2170. Proceed with installation only after substrates have a maximum 75 percent relative humidity level.

Access Flooring Panels: Remove protective film of oil or other coating using method access flooring manufacturer.

Fill cracks, holes, and depressions in substrates with trowelable leveling and patching compound; remove bumps and ridges to produce a uniform and smooth substrate.

- E. Do not install floor tiles until they are the same temperature as the space where they are to be installed.
 - 1. At least 48 hours in advance of installation, move resilient floor tile and installation materials into spaces where they will be installed.

F. Immediately before installation, sweep and vacuum clean substrates to be covered by resilient floor tile.

3.3 FLOOR TILE INSTALLATION

- A. Comply with manufacturer's written instructions for installing floor tile.
- B. Lay out floor tiles from center marks established with principal walls, discounting prinor offsets, so tiles at opposite edges of room are of equal width. Adjust as necessary to avoid using cut widths that equal less than one-half tile at perimeter.
 - 1. Lay tiles square with room axis.
- C. Match floor tiles for color and pattern by selecting tiles from cartons in the same sequence as manufactured and packaged, if so numbered. Discard broken, crackel, chipped, or deformed tiles.
 - 1. Lay tiles with grain running in one direction.
- D. Scribe, cut, and fit floor tiles to butt neatly and tighter to vertical surfaces and permanent fixtures including built-in furniture, cabinets, piper putlets, and door frames.
- E. Extend floor tiles into toe spaces, door reveals cosets, and similar openings. Extend floor tiles to center of door openings.
- F. Maintain reference markers, holes, and oppnings that are in place or marked for future cutting by repeating on floor tiles as marked on substrates. Use chalk or other nonpermanent marking device.
- G. Install floor tiles on covers for telephone and electrical ducts, building expansion-joint covers, and similar items in any shee floor areas. Maintain overall continuity of color and pattern between pieces of tile enstelled on covers and adjoining tiles. Tightly adhere tile edges to substrates that above covers and to cover perimeters.
- H. Adhere floor thes to flooring substrates using a full spread of adhesive applied to substrate to produce a completed installation without open cracks, voids, raising and puckering at joints, telegraphing of adhesive spreader marks, and other surface imperfections.
- I. Seamoss Installation:
- Heat-Welded Seams: Comply with ASTM F 1516. Rout joints and heat weld with welding bead to permanently fuse sections into a seamless flooring. Prepare, weld, and finish seams to produce surfaces flush with adjoining flooring surfaces.
- Chemically Bonded Seams: Bond seams with chemical-bonding compound to permanently fuse sections into a seamless flooring. Prepare seams and apply compound to produce tightly fitted seams without gaps, overlays, or excess bonding compound on flooring surfaces.

ADDITION AND RENOVATIONS

3.4 CLEANING AND PROTECTION

- A. Comply with manufacturer's written instructions for cleaning and protecting floor tile.
- B. Perform the following operations immediately after completing floor tile installation and in accordance with manufacturer's installation guide:
 - 1. Remove adhesive and other blemishes from exposed surfaces.
 - 2. Sweep and vacuum surfaces thoroughly.

- 3. Damp-mop surfaces to remove marks and soil, with manufacturer's recomm cleaning equipment.
- C. Protect floor tile from mars, marks, indentations, and other damage from construction operations and placement of equipment and fixtures during remainder of construction period.
- D. Finish Coats: Resilient Tile Flooring is NOT to receive or require any wax or coating. Only water and Pro-clean Pad, or neutral cleaner.

END OF SECTION

SECTION 09 68 13

TILE CARPETING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section includes modular carpet tile.
- B. Related Requirements:
 - 1. Section 09 65 13 "Resilient Base and Accessories" Section 09 65 19 "Resilient Tile Flooring" for resilient wall base and accessorie installed with carpet tile.

1.3 PREINSTALLATION MEETINGS

- A. Preinstallation Conference: Conduct concretice at Project site.
 - 1. Review methods and procedures related to carpet tile installation including, but not limited to, the following.
 - a. Review whiteen, storage, and handling procedures.
 - b. Review amount conditions and ventilation procedures.
 - c. Peviev subfloor preparation procedures.

1.4 ACTION SUBJIITTALS

Product Data: For each type of product.

- Include manufacturer's written data on physical characteristics, durability, and fade resistance.
- Include manufacturer's written installation recommendations for each type of substrate.

Shop Drawings: For carpet tile installation, plans showing the following:

- 1. Columns, doorways, enclosing walls or partitions, built-in cabinets, and locations where cutouts are required in carpet tiles.
- 2. Carpet tile type, color, and dye lot.
- 3. Type of subfloor.
- 4. Type of installation.

ADDITION AND RENOVATIONS

- 5. Pattern of installation.
- 6. Pattern type, location, and direction.
- 7. Pile direction.
- 8. Type, color, and location of insets and borders.
- 9. Type, color, and location of edge, transition, and other accessory strips.
- 10. Transition details to other flooring materials.
- C. Samples: For each of the following products and for each color and texture required. Laber each. Sample with manufacturer's name, material description, color, pattern, and designation indicated on Drawings and in schedules.
 - 1. Carpet Tile: Full-size Sample.
 - 2. Exposed Edge, Transition, and Other Accessory Stripping.
- D. Samples for Verification: For each of the following products and for each color and texture required. Label each Sample with manufacturer's name, material description, color, pattern, and designation indicated on Drawings and in schedules.
 - 1. Carpet Tile: Full-size Sample.
 - 2. Exposed Edge, Transition, and Other Accessory Stripping: 12-inch-long Samples.
- E. Product Schedule: For carpet tile. Use same designations indicated on Drawings.

1.5 INFORMATIONAL SUBMITTALS

- A. Qualification Data: For Installer.
- B. Product Test Reports: For carpet sile, for tests performed by a qualified testing agency.
- C. Sample Warranty: For special warranty.
- 1.6 CLOSEOUT SUPMITTAL
 - A. Maintenance D. a: For carpet tiles to include in maintenance manuals. Include the following:
 - 1. Methods for maintaining carpet tile, including cleaning and stain-removal products and procedures and manufacturer's recommended maintenance schedule.
 - Precautions for cleaning materials and methods that could be detrimental to carpet tile.

MAINTENANCE MATERIAL SUBMITTALS

Furnish extra materials, from the same product run, that match products installed and that are packaged with protective covering for storage and identified with labels describing contents.

1. Carpet Tile: Full-size units equal to 5 percent of amount installed for each type indicated, but not less than 10 sq. yd.

1.8 QUALITY ASSURANCE

- A. Installer Qualifications: An experienced installer who is certified by the International Certified Floorcovering Installers Association at the Commercial II certification level.
- 1.9 DELIVERY, STORAGE, AND HANDLING
 - A. Comply with CRI's "CRI Carpet Installation Standard."
- 1.10 FIELD CONDITIONS
 - A. Comply with CRI's "CRI Carpet Installation Standard" for temperature, bumidity, and ventilation limitations.
 - B. Environmental Limitations: Do not deliver or install carpet tiles bath spaces are enclosed and weathertight, wet-work in spaces is complete and dry, and ambient emperature and humidity conditions are maintained at levels planned for building occupants luring the remainder of the construction period.
 - C. Do not install carpet tiles over concrete slabs until slabs have cured and are sufficiently dry to bond with adhesive and concrete slabs have an range recommended by carpet tile manufacturer.
 - D. Where demountable partitions or other terms are indicated for installation on top of carpet tiles, install carpet tiles before installing them items.

1.11 WARRANTY

2.

- A. Special Warranty for capet Tiles: Manufacturer agrees to repair or replace components of carpet tile installation bar fall in materials or workmanship within specified warranty period.
 - 1. Warranty does not include deterioration or failure of carpet tile due to unusual traffic, failure of cubstrate, vandalism, or abuse.
 - Falures include, but are not limited to, the following:

More than 10 percent edge raveling, snags, and runs.

- Dimensional instability.
- Excess static discharge.
- d. Loss of tuft-bind strength.
- e. Loss of face fiber.
- f. Delamination.
- 3. Warranty Period: 10 years from date of Substantial Completion.

PART 2 - PRODUCTS

- 2.1 CARPET TILE: See Finish Schedule Spec Section 09 00 00 for Basis of Design.
 - A. See Finish Schedule, Spec Section 09 00 00 for Product Specs.

2.2 INSTALLATION ACCESSORIES

- A. Trowelable Leveling and Patching Compounds: Latex-modified, hydraulic-curent based formulation provided or recommended by carpet tile manufacturer.
- B. Adhesives: Water-resistant, mildew-resistant, nonstaining, pressure-sensitive type to suit products and subfloor conditions indicated, that comply with flamma ility requirements for installed carpet tile, and are recommended by carpet tile manufacturer for releasable installation.
- C. Metal Edge/Transition Strips: Extruded aluminum with mill finite opporties and width shown, of height required to protect exposed edge of carpet, and of maximum lengths to minimize running joints.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine substrates, areas, and conditions, with Installer present, for compliance with requirements for maximum mointure content, alkalinity range, installation tolerances, and other conditions affecting carpet the performance.
- B. Examine carpet tile for ty e. Nor, pattern, and potential defects.
- C. Concrete Slabs: Verify that finishes comply with requirements specified in Section 03 30 00 "Cast-in-Place Concrete" and that surfaces are free of cracks, ridges, depressions, scale, and foreign deposit.
 - 1. Maisture Testing: Perform tests so that each test area does not exceed 200 sq. ft., and perform no fewer than three tests in each installation area and with test areas evenly spaced in installation areas.
 - a. Anhydrous Calcium Chloride Test: ASTM F 1869. Proceed with installation only after substrates have maximum moisture-vapor-emission rate of 3 lb of water/1000 sq. ft. in 24 hours.
 - b. Relative Humidity Test: Using in situ probes, ASTM F 2170. Proceed with installation only after substrates have a maximum 75 percent relative humidity level measurement.
 - c. Perform additional moisture tests recommended in writing by adhesive and carpet tile manufacturers. Proceed with installation only after substrates pass testing.
- D. Proceed with installation only after unsatisfactory conditions have been corrected.

TILE CARPETING 09 68 13 - 4

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

3.2 PREPARATION

- A. General: Comply with CRI's "Carpet Installation Standards" and with carpet tile manufacturer's written installation instructions for preparing substrates indicated to receive carpet tile.
- B. Use trowelable leveling and patching compounds, according to manufacturer's written instructions, to fill cracks, holes, depressions, and protrusions in substrates. Fill or level cracks, holes and depressions 1/8 inch wide or wider, and protrusions more than 1/32 inch unless more stringent requirements are required by manufacturer's written instructions.
- C. Concrete Substrates: Remove coatings, including curing compounds, and other subcances that are incompatible with adhesives and that contain soap, wax, oil, or silicore, withcan using solvents. Use mechanical methods recommended in writing by adhesive and carpet tile manufacturers.
- D. Broom and vacuum clean substrates to be covered immediately before installing carpet tile.

3.3 INSTALLATION

- A. General: Comply with CRI's "CRI Carpet Installation Standard," Section 18, "Modular Carpet" and with carpet tile manufacturer's written installation instructions.
- B. Installation Method: As recommended in writing by carpet tile manufacturer.
- C. Maintain dye-lot integrity. Do not mix dye by s in same area.
- D. Maintain pile-direction patterns in cated on Drawings & Random Installation Methods.
- E. Cut and fit carpet tile to butt tightly to vertical surfaces, permanent fixtures, and built-in furniture including cabinets, press outlets, edgings, thresholds, and nosings. Bind or seal cut edges as recommender by carpet tile manufacturer.
- F. Extend carpet tile into toe spaces, door reveals, closets, open-bottomed obstructions, removable flanges, alcoves, and similar openings.
- G. Maintain reference markers, holes, and openings that are in place or marked for future cutting by rapealing of carpet tile as marked on subfloor. Use nonpermanent, nonstaining marking dence
- H. Install pattern parallel to walls and borders.



Access Flooring: Stagger joints of carpet tiles so carpet tile grid is offset from access flooring panel grid. Do not fill seams of access flooring panels with carpet adhesive; keep seams free of adhesive.

- 4 CLEANING AND PROTECTION
- A. Perform the following operations immediately after installing carpet tile:

- 1. Remove excess adhesive and other surface blemishes using cleaner recommended by carpet tile manufacturer.
- 2. Remove yarns that protrude from carpet tile surface.
- 3. Vacuum carpet tile using commercial machine with face-beater element.
- B. Protect installed carpet tile to comply with CRI's "Carpet Installation Standard," Section 20 "Protecting Indoor Installations."
- C. Protect carpet tile against damage from construction operations and placement of equipment and fixtures during the remainder of construction period. Use protection methods indicatel or recommended in writing by carpet tile manufacturer.

END OF SECTION

Repurse

SECTION 09 91 23

INTERIOR PAINTING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Seconditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section includes surface preparation and the application of part systems on the following interior substrates:
 - 1. Concrete masonry units (CMUs).
 - 2. Wood.
 - 3. Gypsum board.

1.3 DEFINITIONS

- A. MPI Gloss Level 1: Not more than five units at 60 degrees and 10 units at 85 degrees, according to ASTM D 523.
- B. MPI Gloss Level 2: Not more than 10 units at 60 degrees and 10 to 35 units at 85 degrees, according to ASTM D 221.
- C. MPI Gloss Level 3: 10 to 25 units at 60 degrees and 10 to 35 units at 85 degrees, according to ASTM D 523.
- D. MPI Gloss Level V: 20 to 35 units at 60 degrees and not less than 35 units at 85 degrees, according to AFTM D 523.
- E. MPI doss Level 5: 35 to 70 units at 60 degrees, according to ASTM D 523.

MPI Gloss Level 6: 70 to 85 units at 60 degrees, according to ASTM D 523.

MPI Gloss Level 7: More than 85 units at 60 degrees, according to ASTM D 523.

ACTION SUBMITTALS

A. Product Data: For each type of product. Include preparation requirements and application instructions.

- 1. Include Printout of current "MPI Approved Products List" for each product category specified, with the proposed product highlighted.
- 2. Indicate VOC content.
- B. Samples for Verification: For each type of paint system and in each color and gloss of topcoat.
 - 1. Submit Samples on rigid backing, 8 inches square.
 - 2. Label each Sample for location and application area.
- C. Product List: Cross-reference to paint system and locations of application areas designations indicated on Drawings and in schedules. Include color designations.

1.5 MAINTENANCE MATERIAL SUBMITTALS

- A. Furnish extra materials, from the same product run, that match products histalled and that are packaged with protective covering for storage and identified with larges describing contents.
 - 1. Paint: 5 percent, but not less than 1 gal. of each material and olor applied.

1.6 QUALITY ASSURANCE

- A. Mockups: Apply mockups of each paint system endicated and each color and finish selected to verify preliminary selections made under Sample submittals and to demonstrate aesthetic effects and set quality standards for materials and execution.
 - 1. Architect will select one surface to represent surfaces and conditions for application of each paint system.
 - a. Vertical and Porizontal Surfaces: Provide samples of at least 100 sq. ft..
 - b. Other Iteras: Architect will designate items or areas required.
 - 2. Final approval of color selections will be based on mockups.
 - a. If preliminary color selections are not approved, apply additional mockups of additional colors selected by Architect at no added cost to Owner.
 - 3. Approval of mockups does not constitute approval of deviations from the Contract Documents contained in mockups unless Architect specifically approves such deviations in writing.

Subject to compliance with requirements, approved mockups may become part of the completed Work if undisturbed at time of Substantial Completion.

DELIVERY, STORAGE, AND HANDLING

- A. Store materials not in use in tightly covered containers in well-ventilated areas with ambient temperatures continuously maintained at not less than 45 deg F.
 - 1. Maintain containers in clean condition, free of foreign materials and residue.
 - 2. Remove rags and waste from storage areas daily.

INTERIOR PAINTING 09 91 23 - 2

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

1.8 FIELD CONDITIONS

- A. Apply paints only when temperature of surfaces to be painted and ambient air temperatures are between 50 and 95 deg F.
- B. Do not apply paints when relative humidity exceeds 85 percent; at temperatures less than deg F above the dew point; or to damp or wet surfaces.

PART 2 - PRODUCTS

2.1 MANUFACTURERS

- A. Basis of Design: Sherwin Williams.
- B. Products: Subject to compliance with requirements, provide, product listed in the Interior Painting Schedule for the paint category indicated, and Basis of Design
- 2.2 PAINT, GENERAL
 - A. MPI Standards: Products shall comply with MII standards indicated and shall be listed in its "MPI Approved Products Lists."
 - B. Material Compatibility:
 - 1. Materials for use within each pant system shall be compatible with one another and substrates indicated, under conditions of service and application as demonstrated by manufacturer, based or testing and field experience.
 - 2. For each coat in a puint system, products shall be recommended in writing by topcoat manufacturers for use in paint system and on substrate indicated.
 - C. Colors: As indicated on Virish Schedule, Section 09 00 00.
 - 1. Thirty piccent of surface area will be painted with deep tones.

2.3 SOURCE QUALITY CONTROL

Testing of Paint Materials: Owner reserves the right to invoke the following procedure:

- Owner will engage the services of a qualified testing agency to sample paint materials. Contractor will be notified in advance and may be present when samples are taken. If paint materials have already been delivered to Project site, samples may be taken at Project site. Samples will be identified, sealed, and certified by testing agency.
- 2. Testing agency will perform tests for compliance with product requirements.
- 3. Owner may direct Contractor to stop applying paints if test results show materials being used do not comply with product requirements. Contractor shall remove noncomplying paint materials from Project site, pay for testing, and repaint surfaces painted with rejected materials. Contractor will be required to remove rejected materials from

Tetra Tech 200-15704-17001

INTERIOR PAINTING 09 91 23 - 3 previously painted surfaces if, on repainting with complying materials, the two paints are incompatible.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine substrates and conditions, with Applicator present, for compliance with require for maximum moisture content and other conditions affecting performance of the Wext.
- B. Maximum Moisture Content of Substrates: When measured with an electronic monture meter as follows:
 - 1. Concrete: 12 percent.
 - 2. Fiber-Cement Board: 12 percent.
 - 3. Masonry (Clay and CMUs): 12 percent.
 - 4. Wood: 15 percent.
 - 5. Gypsum Board: 12 percent.
 - 6. Plaster: 12 percent.
- C. Gypsum Board Substrates: Verify that finishing formound is sanded smooth.
- D. Plaster Substrates: Verify that plaster is fully cut
- E. Spray-Textured Ceiling Substrates: Voify harsurfaces are dry.
- F. Verify suitability of substrates, including surface conditions and compatibility, with existing finishes and primers.
- G. Proceed with coating population only after unsatisfactory conditions have been corrected.
 - 1. Application of coaling indicates acceptance of surfaces and conditions.
- 3.2 PREPARATION
 - A. Complexition manufacturer's written instructions and recommendations in "MPI Architectural Painting Specification Manual" applicable to substrates and paint systems indicated.

J.C.

emove hardware, covers, plates, and similar items already in place that are removable and are not to be painted. If removal is impractical or impossible because of size or weight of item, provide surface-applied protection before surface preparation and painting.

- 1. After completing painting operations, use workers skilled in the trades involved to reinstall items that were removed. Remove surface-applied protection if any.
- C. Clean substrates of substances that could impair bond of paints, including dust, dirt, oil, grease, and incompatible paints and encapsulants.

INTERIOR PAINTING 09 91 23 - 4

- 1. Remove incompatible primers and reprime substrate with compatible primers or apply tie coat as required to produce paint systems indicated.
- D. Concrete Substrates: Remove release agents, curing compounds, efflorescence, and chalk. Do not paint surfaces if moisture content or alkalinity of surfaces to be painted exceeds that permitted in manufacturer's written instructions.
- E. Masonry Substrates: Remove efflorescence and chalk. Do not paint surfaces if moisture content, or alkalinity of surfaces or mortar joints exceeds that permitted in manufacturer's written instructions.
- F. Wood Substrates:
 - 1. Scrape and clean knots, and apply coat of knot sealer before applying rit
 - 2. Sand surfaces that will be exposed to view, and dust off.
 - 3. Prime edges, ends, faces, undersides, and backsides of wood
 - 4. After priming, fill holes and imperfections in the finish surfaces with putty or plastic wood filler. Sand smooth when dried.

3.3 APPLICATION

- A. Apply paints according to manufacturer's written instructions and to recommendations in "MPI Manual."
 - 1. Use applicators and techniques suffect for paint and substrate indicated.
 - 2. Paint surfaces behind movable equipment and furniture same as similar exposed surfaces. Before final installation, part surfaces behind permanently fixed equipment or furniture with prime coat only.
 - Paint front and backsides of access panels, removable or hinged covers, and similar hinged items to matchexpored surfaces.
 - Do not paint over labels of independent testing agencies or equipment name, identification, performance rating, or nomenclature plates.
 Primers stepified in painting schedules may be omitted on items that are factory primed
 - 5. Primers specified in painting schedules may be omitted on items that are factory primed or factory freiched if acceptable to topcoat manufacturers.
- B. Tint each unde coat a lighter shade to facilitate identification of each coat if multiple coats of same material are to be applied. Tint undercoats to match color of topcoat, but provide sufficient difference in shade of undercoats to distinguish each separate coat.

If undercoats or other conditions show through topcoat, apply additional coats until cured film has a uniform paint finish, color, and appearance.

20

Apply paints to produce surface films without cloudiness, spotting, holidays, laps, brush marks, roller tracking, runs, sags, ropiness, or other surface imperfections. Cut in sharp lines and color breaks.

- E. Painting Fire Suppression, Plumbing, HVAC, Electrical, Communication, and Electronic Safety and Security Work:
 - 1. Paint the following work where exposed in equipment rooms:

- a. Equipment, including panelboards.
- b. Uninsulated metal piping.
- c. Uninsulated plastic piping.
- d. Pipe hangers and supports.
- e. Metal conduit.
- f. Plastic conduit.
- g. Tanks that do not have factory-applied final finishes.
- h. Duct, equipment, and pipe insulation having cotton or canvas insulation covering or other paintable jacket material.
- 2. Paint the following work where exposed in occupied spaces:
 - a. Equipment, including panelboards.
 - b. Uninsulated metal piping.
 - c. Uninsulated plastic piping.
 - d. Pipe hangers and supports.
 - e. Metal conduit.
 - f. Plastic conduit.
 - g. Duct, equipment, and pipe insulation having other or canvas insulation covering or other paintable jacket material.
 - h. Other items as directed by Architect.
- 3. Paint portions of internal surfaces of metal durts, without liner, behind air inlets and outlets that are visible from occupied spaces

3.4 FIELD QUALITY CONTROL

- A. Dry Film Thickness Testing Owner may engage the services of a qualified testing and inspecting agency to inspect and test paint for dry film thickness.
 - 1. Contractor shall touch up and restore painted surfaces damaged by testing.
 - 2. If test results show that dry film thickness of applied paint does not comply with paint manufacturer's written recommendations, Contractor shall pay for testing and apply additional costs as needed to provide dry film thickness that complies with paint margin storer's written recommendations.

3.5 CLEANING AND PROTECTION

At end of each workday, remove rubbish, empty cans, rags, and other discarded materials from Roject site.

After completing paint application, clean spattered surfaces. Remove spattered paints by washing, scraping, or other methods. Do not scratch or damage adjacent finished surfaces.

C. Protect work of other trades against damage from paint application. Correct damage to work of other trades by cleaning, repairing, replacing, and refinishing, as approved by Architect, and leave in an undamaged condition.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

D. At completion of construction activities of other trades, touch up and restore damaged or defaced painted surfaces.

3.6 INTERIOR PAINTING SCHEDULE

- A. CMU Substrates:
 - 1. Latex Aggregate System MPI INT 4.2B:
 - a. Prime Coat: Primer for textured coating, latex, flat, as recommended to writing b topcoat manufacturer.
 - b. Intermediate Coat: Intermediate coat for textured coating, lat x, flat, as recommended in writing by topcoat manufacturer.
 - 2. Institutional Low-Odor/VOC Latex System MPI INT 4.2E:
 - a. Block Filler: Block filler, latex, interior/exterior, MPA#
 - 1) Basis of Design; Sherwin Williams.
 - b. Intermediate Coat: Latex, interior, institutional ow odor/VOC, matching topcoat.
 - c. Topcoat: Latex, interior, institutional low odor/VOC (MPI Gloss Level 4), MPI #146.
 - 1) Basis of Design: Sherwin Williams.
- B. Wood Substrates: Glued-laminated construction.
 - 1. Institutional Low-Oder/NOC Latex System MPI INT 6.1Q:
 - a. Prime Cout Primer, latex, for interior wood, MPI #39.
 - Basis of Design; Sherwin Williams.
 - Intermediate Coat: Latex, interior, institutional low odor/VOC, matching topcoat.
 - Topcoat: Latex, interior, institutional low odor/VOC (MPI Gloss Level 4), MPI #146.
 - 1) Basis of Design; Sherwin Williams.
 - d. Topcoat: Latex, interior, institutional low odor/VOC, semi-gloss (MPI Gloss Level 5), MPI #147.
 - 1) Basis of Design; Sherwin Williams.
- C. Wood Substrates: Wood trim and Architectural woodwork.
 - 1. Institutional Low-Odor/VOC Latex System MPI INT 6.3V:

- a. Prime Coat: Primer, latex, for interior wood, MPI #39.
 - 1) Basis of Design; Sherwin Williams.
- b. Intermediate Coat: Latex, interior, institutional low odor/VOC, matching topcoat.
- c. Topcoat: Latex, interior, institutional low odor/VOC (MPI G Level 4), MPI #146.
 - 1) Basis of Design; Sherwin Williams.
- d. Topcoat: Latex, interior, institutional low odor/VOC, semi-gross (NPr Gloss Level 5), MPI #147.
 - 1) Basis of Deign; Sherwin Williams.

D. Gypsum Board Substrates:

- 1. Latex over Latex Sealer System MPI INT 9.2A:
 - a. Prime Coat: Primer sealer, latex, interior, M2/#50.
 - 1) Basis of Design: Sherwin William
 - b. Prime Coat: Latex, interior, matching topcoat.
 - c. Intermediate Coat: Latex, interior, matching topcoat.
 - d. Topcoat: Latex, interfor (MNI Gloss Level 4), MPI #43.
 - 1) Basis of Design Sherwin Williams.
 - e. Topcoat; Latex, Aterior, semi-gloss (MPI Gloss Level 5), MPI #54.
 -) Basis of Design: Sherwin Williams.

END OF SECTION

INTERIOR PAINTING 09 91 23 - 8

SECTION 12 24 13

ROLLER WINDOW SHADES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General an Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Manually operated roller shades with single roller.

1.3 ACTION SUBMITTALS

- A. Product Data: For each type of product.
 - 1. Include construction details, material descriptions, dimensions of individual components and profiles, features, finishes, and operating instructions for roller shades.
- B. Shop Drawings: Show fabrication and installation details for roller shades, including shadeband materials, their orientation o rollers, and their seam and batten locations.
 - 1. Motor-Operates Shares: Include details of installation and diagrams for power, signal, and control wiring
- C. Samples: For even exposed product and for each color and texture specified, 10 inches long.
- D. Samples for Initial Selection: For each type and color of shadeband material.

include Samples of accessories involving color selection.

Samples for Verification: For each type of roller shade.

- Shadeband Material: Not less than 10 inches square. Mark interior face of material if applicable.
- 2. Roller Shade: Full-size operating unit, not less than 16 inches wide by 36 inches long for each type of roller shade indicated.
- 3. Installation Accessories: Full-size unit, not less than 10 inches long.
- F. Product Schedule: For roller shades. Use same designations indicated on Drawings.
ADDITION AND RENOVATIONS

1.4 INFORMATIONAL SUBMITTALS

- A. Qualification Data: For Installer.
- B. Product Certificates: For each type of shadeband material.
- C. Product Test Reports: For each type of shadeband material, for tests performed by manufacturer and witnessed by a qualified testing agency a qualified testing agency.

1.5 CLOSEOUT SUBMITTALS

A. Operation and Maintenance Data: For roller shades to include in maintenance nanu

1.6 MAINTENANCE MATERIAL SUBMITTALS

- A. Furnish extra materials that match products installed and that are packaged with protective covering for storage and identified with labels describing contents.
 - 1. Roller Shades: Full-size units equal to 5 percent of quantity installed for each size, color, and shadeband material indicated, but no fewer than two units.

1.7 QUALITY ASSURANCE

- A. Installer Qualifications: Fabricator of product
- B. Mockups: Build mockups to varify selections made under Sample submittals, to demonstrate aesthetic effects, and to set quality standards for fabrication and installation.
 - 1. Approval of mecaups does not constitute approval of deviations from the Contract Documents conviaed in mockups unless Architect specifically approves such deviations in writing
 - 2. Subject to compliance with requirements, approved mockups may become part of the complete Work if undisturbed at time of Substantial Completion.

1.8 DECIVER 1, STORAGE, AND HANDLING

Deliver roller shades in factory packages, marked with manufacturer, product name, and potential of installation using same designations indicated on Drawings.

FIELD CONDITIONS

- . Environmental Limitations: Do not install roller shades until construction and finish work in spaces, including painting, is complete and dry and ambient temperature and humidity conditions are maintained at the levels indicated for Project when occupied for its intended use.
- B. Field Measurements: Where roller shades are indicated to fit to other construction, verify dimensions of other construction by field measurements before fabrication and indicate

ROLLER WINDOW SHADES 12 24 13 - 2

Tetra Tech 200-15704-17001

measurements on Shop Drawings. Allow clearances for operating hardware of operable glazed units through entire operating range. Notify Architect of installation conditions that vary from Drawings. Coordinate fabrication schedule with construction progress to avoid delaying the Work.

PART 2 - PRODUCTS

- 2.1 MANUFACTURERS
 - A. Source Limitations: Obtain roller shades from single source from single manufactu

2.2 MANUALLY OPERATED SHADES WITH SINGLE ROLLERS

- A. Basis of Desgin: Mecho/5 by Mecho Systems
- B. Chain-and-Clutch Operating Mechanisms: With continuous loop bead chain and clutch that stops shade movement when bead chain is released; perparently adjusted and lubricated.
 - 1. Bead Chains: Manufacturer's standard Staipless steel
 - a. Loop Length: Full length of roller shade,
 - b. Limit Stops: Provide upper and lower ball stops.
 - c. Chain-Retainer Type: Clip, jarb mount.
 - 2. Spring Lift-Assist Mechanisms: Manufacturer's standard for balancing roller shade weight and for lifting hears roller shades.
 - a. Provide for shidebands that weigh more than 10 lb or for shades as recommended by manufact rer, whichever criterion is more stringent.
- C. Rollers: Corrosion-resistant steel or extruded-aluminum tubes of diameters and wall thicknesses required to according dependence operating mechanisms and weights and widths of shadebands indicated without deflection. Provide with permanently lubricated drive-end assemblies and idle-end assemblies designed to facilitate removal of shadebands for service.
 - 1. Poller Drive-End Location: Right side of interior face of shade.
 - Direction of Shadeband Roll: Regular, from back (exterior face) of roller.
 - Shadeband-to-Roller Attachment: Manufacturer's standard method.



Mounting Hardware: Brackets or endcaps, corrosion resistant and compatible with roller assembly, operating mechanism, installation accessories, and mounting location and conditions indicated.

- Roller-Coupling Assemblies: Coordinated with operating mechanism and designed to join up to three inline rollers into a multiband shade that is operated by one roller drive-end assembly.
- F. Shadebands:

3.

Tetra Tech 200-15704-17001

- 1. Shadeband Material: Light-filtering fabric. Basis of Design: Mecho Systems Thermoveil 1500 Series (3% open)
- 2. Shadeband Bottom (Hem) Bar: Steel or extruded aluminum.
 - a. Type: Enclosed in sealed pocket of shadeband material.
 - b. Color and Finish: As selected by Architect from manufacturer's full range.
- G. Installation Accessories:
 - 1. Front Fascia: Aluminum extrusion that conceals front and underside of rolle operating mechanism and attaches to roller endcaps without exposed fasteners
 - a. Shape: L-shaped.
 - b. Height: Manufacturer's standard height required to concert roller and shadeband assembly when shade is fully open, but not less than 4 inches
 - 2. Exposed Headbox: Rectangular, extruded-aluminum enclosurg including front fascia, top and back covers, endcaps, and removable bottom closure.
 - a. Height: Manufacturer's standard height required to enclose roller and shadeband assembly when shade is fully open, but not has than 4 inches.
 - 3. Endcap Covers: To cover exposed endcap.
 - 4. Closure Panel and Wall Clip: Removable aluminum panel designed for installation at bottom of site-constructed ceiling recess or pocket and for snap-in attachment to wall clip without fasteners.
 - a. Closure-Panel Wiann 2 in ne
 - 5. Installation Accessories Color and Finish: As selected from manufacturer's full range.
- 2.3 SHADEBAND MATERNES
 - A. Shadeband Max ial Flame-Resistance Rating: Comply with NFPA 701. Testing by a qualified testing a ency. Identify products with appropriate markings of applicable testing agency.
 - B. Light-Filtering Fabric: Woven fabric, stain and fade resistant.



- Source: Roller shade manufacturer.
- Type: PVC-coated polyester.
- Weave: Basketweave.
- Roll Width: Provide width that maximizes roll without vert. seams at any window.
- 5. Orientation on Shadeband: Up the bolt.
- 6. Openness Factor: 3 percent.
- 7. Color: As selected by Architect from manufacturer's full range.

2.4 ROLLER SHADE FABRICATION

- A. Product Safety Standard: Fabricate roller shades to comply with WCMA A 100.1, including requirements for flexible, chain-loop devices; lead content of components; and warning labels.
- B. Unit Sizes: Fabricate units in sizes to fill window and other openings as follows, measured at 70 deg F:
 - 1. Outside of Jamb Installation: Width and length as indicated, with terminations between shades of end-to-end installations at centerlines of mullion or other defined vertical separations between openings.
- C. Shadeband Fabrication: Fabricate shadebands without battens or seams to extent possible, except as follows:
 - 1. Vertical Shades: Where width-to-length ratio of shadeband is equal to or greater than 1:4, provide battens and seams at uniform spacings along studeband length to ensure shadeband tracking and alignment through its full range of notement without distortion of the material.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine substrates, areas, and conditions, with Installer present, for compliance with requirements for installation tolerances, operational clearances, and other conditions affecting performance of the Work.
- B. Proceed with installation only alter unsatisfactory conditions have been corrected.
- 3.2 ROLLER SHADE INSTALLATION
 - A. Install roller sholes level, plumb, and aligned with adjacent units according to manufacturer's written instructions.
 - B. Electrical Connections: Connect motor-operated roller shades to building electrical system.
 - C. Roller Shade Locations: At exterior windows and As indicated on Drawings.

ADJUSTING

Adjust and balance roller shades to operate smoothly, easily, safely, and free from binding or malfunction throughout entire operational range.

3.4 CLEANING AND PROTECTION

A. Clean roller shade surfaces, after installation, according to manufacturer's written instructions.

- B. Provide final protection and maintain conditions, in a manner acceptable to manufacturer and Installer, that ensure that roller shades are without damage or deterioration at time of Substantial Completion.
- C. Replace damaged roller shades that cannot be repaired, in a manner approved by Architect, before time of Substantial Completion.
- 3.5 DEMONSTRATION
 - A. Engage a factory-authorized service representative to train Owner's maintenance personne adjust, operate, and maintain motor-operated roller shades.

END OF SECTION

2 BIDDIK

208

SECTION 12 36 23.13

PLASTIC-LAMINATE-CLAD COUNTERTOPS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section includes plastic-laminate countertops.
- B. Countertop Assemblies.
- C. Cutouts for sinks, faucets, fittings, and other plumbing and energical fixtures, electrical and mechanical runs and connections and similar items
- D. Materials and devices necessary to make solid connections be existing structure.

1.3 ACTION SUBMITTALS

- A. Product Data: For each type of product including panel products, high-pressure decorative laminate, adhesive for bonding plantic laminate, and fire-retardant-treated materials.
- B. Shop Drawings: Show location on each item, dimensioned plans and elevations, large-scale details, attachment devices, and other components.
 - 1. Show locations and bizes of cutouts and holes for plumbing fixtures, faucets, soap dispensers electrice switches and outlets, and other items installed in plastic-laminate counterpope
- C. Samples or Initial Selection:
 - 1. Plastic taminates.
 - D. Samples for Verification:
 - Plastic laminates, 8 by 10 inches, for each type, color, pattern, and surface finish.
 - Wood-grain plastic laminates, 12 by 24 inches, for each type, pattern and surface finish.

INFORMATIONAL SUBMITTALS

- A. Qualification Data: For Installer and fabricator.
- B. Product Certificates: For each type of product and the following:

Tetra Tech 200-15704-17001

ADDITION AND RENOVATIONS

- 1. Composite wood and agrifiber products.
- 2. High-pressure decorative laminate.
- 3. Adhesives.
- C. Evaluation Reports: For fire-retardant-treated materials, from ICC-ES.

1.5 QUALITY ASSURANCE

- A. Fabricator Qualifications: Shop that employs skilled workers who custom fabricate products similar to those required for this Project and whose products have a record of storess all inservice performance.
- B. Installer Qualifications: Fabricator of products.
- C. Testing Agency Qualifications: For testing agency providing classification marking for fireretardant-treated material, an inspection agency acceptable to autholities having jurisdiction that periodically performs inspections to verify that the material bearing the classification marking is representative of the material tested.

1.6 DELIVERY, STORAGE, AND HANDLING

A. Do not deliver countertops until painting and similar operations that could damage countertops have been completed in installation areas. If countertops must be stored in other than installation areas, store only in areas where environmental conditions comply with requirements specified in "Field Conditions" Article.

1.7 FIELD CONDITIONS

- A. Environmental Limitations: Do not deliver or install countertops until building is enclosed, wet work is complete, and aVLC system is operating and maintaining temperature and relative humidity at occurancy levels during the remainder of the construction period.
- B. Environmental similations: Do not deliver or install countertops until building is enclosed, wet work is complete, and HVAC system is operating and maintaining temperature between 60 and 90 dog F and relative humidity between 25 and 55 percent during the remainder of the construction period.

Field Measurements: Where countertops are indicated to fit to other construction, verify timensions of other construction by field measurements before fabrication, and indicate measurements on Shop Drawings. Coordinate fabrication schedule with construction progress to avoid delaying the Work.

Established Dimensions: Where countertops are indicated to fit to other construction, establish dimensions for areas where countertops are to fit. Provide allowance for trimming at site, and coordinate construction to ensure that actual dimensions correspond to established dimensions.

PART 2 - PRODUCTS

2.1 PLASTIC-LAMINATE COUNTERTOPS

- A. Quality Standard: Unless otherwise indicated, comply with the "Architectural Woodwork Standards" for grades indicated for construction, installation, and other requirements.
- B. Grade: Premium.
- C. High-Pressure Decorative Laminate: NEMA LD 3, Grade HGS.
 - 1. Manufacturers: Basis of Design, Subject to compliance with requirements, provide products by one of the following;
 - a. Basis of Design: Wilsonart Contract Laminate.
- D. Colors, Patterns, and Finishes: Provide materials and products that result in colors and textures of exposed laminate surfaces complying with the following requirements:
 - 1. As indicated in specification section 09 00 00 "Finish Schedule" and per casework notes and details on drawings.
 - 2. Grain Direction: Parallel to cabinet froms.
- E. Edge Treatment: 3-mm PVC edging selected from a minimum of 12 colors. 3MM PVC to be solid, high impact, purified, color-then acid resistant, pre-lamination primed edging, machine-applied with hot melt adhesive, automatically trimmed, inside/outside length-radius for uniform appearance, buffed and corner-radius for consistent design.
- F. Core Material: Particleboard or me num-density fiberboard.
- G. Core Material at Sink medium-density fiberboard made with exterior glue or exterior-grade plywood.
- H. Core Thickness 5/4 inch.
 - 1. Build up countertop thickness to 1-1/2 inches at front, back, and ends with additional layers of core material laminated to top.
 - Backer Sheet: Provide plastic-laminate backer sheet, NEMA LD 3, Grade BKL, on underside of ountertop substrate.



Paper Backing: Provide paper backing on underside of countertop substrate.

WOOD MATERIALS

- A. Wood Products: Provide materials that comply with requirements of referenced quality standard unless otherwise indicated.
 - 1. Wood Moisture Content: 5 to 10 percent.

Tetra Tech 200-15704-17001

- B. Composite Wood and Agrifiber Products: Provide materials that comply with requirements of referenced quality standard for each type of woodwork and quality grade specified unless otherwise indicated.
 - 1. Medium-Density Fiberboard: ANSI A208.2, Grade 130.
 - 2. Particleboard: ANSI A208.1, Grade M-2 or M-2-Exterior Glue.
 - 3. Softwood Plywood: DOC PS 1.

2.3 FIRE-RETARDANT-TREATED MATERIALS

- A. Fire-Retardant-Treated Materials, General: Where fire-retardant-treated materials are tradated, use materials complying with requirements in this article that are acceptable to authorities having jurisdiction and with fire-test-response characteristics specified and ermined by testing identical products per test method indicated by a qualified testing agency.
 - 1. Use treated materials that comply with requirements of referenced woodworking standard. Do not use materials that are warped, discolored, o onerwise defective.
 - 2. Use fire-retardant-treatment formulations that do not bled through or otherwise adversely affect finishes. Do not use colorante to listinguish treated materials from untreated materials.
 - 3. Identify fire-retardant-treated materials with appropriate classification marking of qualified testing agency in the form of renovable paper label or imprint on surfaces that will be concealed from view after instantion
- B. Fire-Retardant-Treated Lumber and Ply pool: Products with a flame-spread index of 25 or less when tested according to ASTM E 64, with no evidence of significant progressive combustion when the test is extended an additional 20 minutes, and with the flame front not extending more than 10.5 feet beyond the centraline of the burners at any time during the test.
 - 1. Kiln dry lumber and plywood after treatment to a maximum moisture content of 19 and 15 percent, respectively.
 - 2. For items indicated b receive a stained or natural finish, use organic resin chemical formulation
 - 3. Mill lumber after treatment within limits set for wood removal that do not affect listed fire text remonse characteristics, using a woodworking shop certified by testing and impecting agency.
 - 4. Multiplet before treatment and implement special procedures during treatment and arying processes that prevent lumber from warping and developing discolorations from trying sticks or other causes, marring, and other defects affecting appearance of treated woodwork.

Fire-Retardant Particleboard: Panels complying with the following requirements, made from softwood particles and fire-retardant chemicals mixed together at time of panel manufacture to achieve flame-spread index of 25 or less and smoke-developed index of 25 or less per ASTM E 84.

1. For panels 3/4 inch thick and less, comply with ANSI A208.1 for Grade M-2 except for the following minimum properties: modulus of rupture, 1600 psi; modulus of elasticity, 300,000 psi; internal bond, 80 psi; and screw-holding capacity on face and edge, 250 and 225 lbf, respectively.

- 2. For panels 13/16 to 1-1/4 inches thick, comply with ANSI A208.1 for Grade M-1 except for the following minimum properties: modulus of rupture, 1300 psi; modulus of elasticity, 250,000 psi; linear expansion, 0.50 percent; and screw-holding capacity on face and edge, 250 and 175 lbf, respectively.
- 3. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
 - a. Flakeboard Company Limited.
 - b. SierraPine.

2.4 ACCESSORIES

- A. Grommets for Cable Passage through Countertops: 2-inch OD, black, model-phone grommets and matching plastic caps with slot for wire passage.
 - 1. Manufacturers: Subject to compliance with requirements provide products by one of the following:
 - a. Doug Mockett & Company, Inc.
- B. Concealed Metal Countertop Support Brackets: Lead Limit Range of 1800 to 7960 lbs per pair. Mounting Hardware includes 3/8"-16 x 3 box as erablies, fully welded construction, with powder coated finish. Color: Selected from manufacturer's full range. See Drawings for support arm lengths.
 - 1. Basis-of-Design Product: Subject to compliance with requirements, provide A&M Hardware, Inc., Concealed bracket"C Bracket", or equal.

2.5 MISCELLANEOUS MATERIAL

- A. Adhesive for Bonding Vasti Laminate: Unpigmented contact cement.
 - 1. Adhesive for Bonding Edges: Hot-melt adhesive or adhesive specified above for faces.

2.6 FABRICATIO

A. Sand Gre-retardant-treated wood lightly to remove raised grain on exposed surfaces before fabrication.

Fabricate countertops to dimensions, profiles, and details indicated. Provide front and end overhang of 1 inch over base cabinets. Ease edges to radius indicated for the following:

- 1. Solid-Wood (Lumber) Members: 1/16 inch unless otherwise indicated.
- C. Complete fabrication, including assembly, to maximum extent possible before shipment to Project site. Disassemble components only as necessary for shipment and installation. Where necessary for fitting at site, provide ample allowance for scribing, trimming, and fitting.

- 1. Notify Architect seven days in advance of the dates and times woodwork fabrication will be complete.
- 2. Trial fit assemblies at fabrication shop that cannot be shipped completely assembled. Install dowels, screws, bolted connectors, and other fastening devices that can be removed after trial fitting. Verify that various parts fit as intended and check measurements of assemblies against field measurements before disassembling for shipment.
- D. Shop cut openings to maximum extent possible to receive appliances, plumbing hydres, electrical work, and similar items. Locate openings accurately and use templates or roughing in diagrams to produce accurately sized and shaped openings. Sand edges of cutous to remove splinters and burrs.
 - 1. Seal edges of openings in countertops with a coat of varnish.

PART 3 - EXECUTION

3.1 PREPARATION

- A. Before installation, condition countertops to merage prevailing humidity conditions in installation areas.
- B. Before installing countertops, examine shop fabricated work for completion and complete work as required, including removal of packing and backpriming.

3.2 INSTALLATION

- A. Grade: Install countertops to comply with same grade as item to be installed.
- B. Assemble countertops and complete fabrication at Project site to the extent that it was not completed in the shop.
 - 1. Provide c touts for appliances, plumbing fixtures, electrical work, and similar items.
 - 2. Seal edg s of cutouts by saturating with varnish.
- C. Field Uniting: Where possible, make in the same manner as shop jointing, using dowels, spline, adhesives, and fasteners recommended by manufacturer. Prepare edges to be joined in shop so Project-site processing of top and edge surfaces is not required. Locate field joints where shown on Shop Drawings.
 - 1. Secure field joints in plastic-laminate countertops with concealed clamping devices located within 6 inches of front and back edges and at intervals not exceeding 24 inches. Tighten according to manufacturer's written instructions to exert a constant, heavy-clamping pressure at joints.
- D. Install countertops level, plumb, true, and straight. Shim as required with concealed shims. Install level and plumb to a tolerance of 1/8 inch in 96 inches.

- E. Scribe and cut countertops to fit adjoining work, refinish cut surfaces, and repair damaged finish at cuts.
- F. Fire-Retardant-Treated Wood: Handle, store, and install fire-retardant-treated wood to comply with chemical treatment manufacturer's written instructions, including those for adhesives used to install woodwork.
- G. Countertops: Anchor securely by screwing through corner blocks of base cabinets or of supports into underside of countertop.
 - 1. Install countertops with no more than 1/8 inch in 96-inch sag, bow, or other valiation from a straight line.
 - 2. Secure backsplashes to tops with concealed metal brackets at 16 inche o.c. and to walls with adhesive.
 - 3. Seal junctures of tops, splashes, and walls with mildew-resistant silicone sealant or another permanently elastic sealing compound recommended by countertop material manufacturer.

3.3 ADJUSTING AND CLEANING

- A. Repair damaged and defective countertops, where possible, to eliminate functional and visual defects; where not possible to repair, replace wordwork. Adjust joinery for uniform appearance.
- B. Clean countertops on exposed and semicroposed surfaces. Touch up shop-applied finishes to restore damaged or soiled areas.

SECTION



SECTION 210170

FIRE PROTECTION SYSTEMS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. The drawings and general provisions of the Contract, including the Conditions of the Contract, (General, Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the work specified in this Section.
- B. Codes and Standards listed below, apply to work indicated on the drawings and in the specifications.
 - 1. National Fire Protection Association (NFPA)
 - 2. Delaware State Fire Prevention Regulations (DSFPR)
 - 3. American National Standards Institute (ANSI)
 - 4. American Society for Testing Materials (ASTM)
 - 5. National Electrical Manufacturer's Association (NEM
 - 6. Underwriters' Laboratories (UL)

1.2 DESCRIPTION OF WORK

- A. This Section shall include all work necessary and/or required and furnish all materials and equipment for construction of a complete automatic spinker system (and fire standpipe system) for the building areas indicated. Such work includes but hand limited to the following:
 - 1. UL/FM labeled equipment.
 - 2. All piping and equipment required for a complete wet sprinkler system on occupied floor levels.
 - 3. All piping and equipment required for a complete dry sprinkler system where shown.
 - 4. Installation of zoned text assemblies.
 - 5. Installation of tamper and flow switches.
 - 6. Installation of riser check valve assembly.
 - 7. All piping and equipment for a complete wet fire standpipe system.



To coordinate with the Mechanical, Plumbing and Electrical Contractors, the installation of the mains and sprinkler piping and supports to allow installation of their work with maximized accessibility for these trades and service requirements for maintenance and repair. Prior to installing any piping or other devices, obtain written conformation from these contractors that requirements, conflicts and coordination issues have been discussed and resolved. Provide system drawings with elevation of any piping or other systems to the Mechanical Contractor so he can prepare the necessary coordination drawings that may be required. No work may be installed until the coordination issues are resolved. Any and all expense relating to coordination drawings.

ADDITION AND RENOVATIONS

1.3 REFERENCE STANDARDS

A. Refer to Section 220000 for a general description of requirements applying to this section.

1.4 QUALITY ASSURANCE

- A. Refer to Section 220010 for a general description of requirements applying to this section.
- B. Sprinkler System Contractor shall provide new separate and complete sprinkler systems (wet and in satisfactory operating condition which shall conform to requirements of the following:
 - 1. NFPA Pamphlet 13
 - 2. New Castle County Fire Marshal's Office
 - 3. Owner's Insurance Agency
 - 4. Owner's Standards
- C. Submit working drawings to the New Castle County Fire Marshal's Office and obtain approval before beginning work.
- D. Sprinkler systems shall be "Light Hazard Occupancy" and shall over all rooms, closets, attic spaces, etc., in the entire building.
 - 1. Design and layout shall be based on Calculated System (Hydraulic).
 - 2. Exact routing of piping shall be governed by structural conditions and obstructions.
 - 3. The Sprinkler Contractor shall coordinate his work with the other trades so as to clear all construction items, lights, ducts, pipe, etc.

1.5 SUBMITTALS

A. Submit shop drawings and product data in accordance with Section 220000.

Submit shop drawings with Fire Marshal's approval and descriptive data, complete with product designation for the following:

- 1. OS&Y Valves
- 2. Sprinkler maan
- 3. Check Valve
 - Fit Department Hose Valves

Backflow Preventer

Submit complete sprinkler layout indicating location of heads by dimensions from walls, pipe size, and locations of valves, fittings and accessories, with Fire Marshal's approval.

Submit manufacturer's product data on sprinkler heads, valves, fire department connections, air compressor, hangers, pipe, and fittings, etc.

- 1.6 WARRANTY/GUARANTEE
 - A. All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, GENERAL REQUIREMENTS.

1.7 TESTS AND INSPECTIONS

- A. Contractor shall arrange and pay for all inspections, examinations and tests required by authorities specified herein and deliver certificates of such inspections to Owner.
- B. Complete sprinkler system shall be tested in accordance with the latest requirements of NFPA Pamphlet 13 and the New Castle County Fire Marshal's Office.
- C. Fire Marshal's acceptance test shall be performed before system is placed in service and not less that five working days after Fire Marshal is notified.
- 1.8 QUALIFICATIONS OF CONTRACTOR
 - A. Contractor for sprinkler installation shall be licensed by the State of Delawar and be regularly engaged in installation of automatic sprinkler systems and other fire protection quipment.
 - B. Consult General Provisions for additional requirement.

PART 2 – PRODUCTS

- 2.1 FIRE PROTECTION PIPING MATERIALS & PRODUCTS
 - A. Provide piping materials and factory-fabricated piping products of sizes, types, pressure ratings, temperature ratings, and capacities as indicated. Where not indicated, provide proper selection as determined by installer to comply with installation equipements. Provide sizes and types matching piping and equipment connections; provide fittings of materials which match pipe materials used in fire protection piping systems. Where more than hype or materials or products are indicated, selection is installer's option.

2.2 BASIC IDENTIFICATION

- A. Provide identification complying with applicable Division 22 sections in accordance with the following listings:
 - 1. Fire Protection Piping. Plastic pipe markers.
 - 2. Fire Protection Valvex Metallic valve tags.
- 2.3 BASIC PIPE AND UTTINGS
 - A. Comply with the weight, size and type of pipe and fittings by the latest issued schedule of NFPA Pamphlet 13, adopted by Authorities having jurisdiction.
 - B. All fit protection piping within the Mechanical Room shall be minimum Schedule 40 black iron pipe.
 - . Plastic piping shall not be permitted.

Uni-Flange type connections shall not be permitted on this project.

BASIC PIPING SPECIALTIES

Provide piping specialties complying with Section 220010 Basic Materials & Methods in accordance with the following listing:

Pipe escutcheons

Dielectric unions

Drip pans

Sleeves

Sleeve seals

- 2.5 BASIC VALVES
 - A. Comply with the latest issue of NFPA 13 adopted by the Authorities having jurisdiction for the following:

Control Valve - OS&Y Valve

Check - Swing Valve

- 2.6 SPECIAL VALVE & ASSEMBLIES
 - A. Provide valves, UL listed, in accordance with the following listing. Provide izes and types which mate and match piping and equipment connections.
 - B. Fire Department Connection: 4" x 2-1/2" x 2-1/2" back outlet, cast brast two-way inlet body with drop clappers, polished bronze plate with letters "AUTO SPKR" to ble female snoots with rigid end threaded to match local fire department, plugs and chain.

2.7 FIRE PROTECTION SPECIALTIES

- A. Provide fire protection specialties, UL listed, in accordance with the following listing. Provide sizes and types which mate and match piping and equipment connections.
- B. Automatic Sprinklers: Sprinkler heads sharebe UL approved for intended use and have temperature ratings as indicated or required for location

Provide the following type sprinklet heads as indicated:

Upright: Viking "Micrometic Model "M" bronze finish, 1/2" orifice.

- Pendent: Viking "Merodatic Model "M" chrome plated, 1/2" orifice and escutcheon plate.
- Pendent: Viking "Microbist Model "M" quick response with all white finish including escutcheon place, ¹/₂ orifice.
- Sidewall: Viking Micromatic Model "M", chrome plate, ¹/₂" orifice.
- Sidewall: ⁷²¹/₂" orifice. ¹/₂" orifice.
- idewall: (Extended Coverage) Viking "Microfast Model "M", quick response, all white finish, extra large orifice.

Semi-Recessed: Viking "Silhouette Model A-1 quick response, chrome plated, ¹/₂" orifice.

Full Concealed: Viking "Horizon-Mirage" large orifice quick response, white cover plate.

Dry Sprinklers:

Upright: Viking "Model "M" plain barrel, ½" orifice bronze finish.

Pendent: Viking "Model "M", adjustable standard, all white finish, ¹/₂" orifice.

Pendent: Viking "Model "M" adjustable semi-recessed, all white finish, ¹/₂" orifice.

Sidewall: Viking "Model "M" adjustable standard, all white finish, ½" orifice.

Sidewall: Viking "Model "M" adjustable semi-recessed, all white finish, ¹/₂" orifice.

- C. <u>Sprinkler Cabinet and Wrench:</u> Furnish steel, baked red enameled, sprinkler box with capacity store 10 sprinklers and wrench sized to sprinklers.
- 2.8 MANUFACTURERS
 - A. Available Manufacturers: Subject to compliance with requirements, manufacturers offering products that may be incorporated into the work include, but are not limited to, the following.
 - B. Manufacturers: Subject to compliance with requirements, provide products by on of the following:
 - 1. Specialty Valves and Devices:

Vicatulic Company

Tyco Fire Suppression & Building Products

Reliable Automatic Sprinkler Co., Inc.

Viking Corp.

2. Water-Flow Indicators and Supervisory Switch

Tyco Fire Suppression & Building Pro-

Potter Electric Signal Co.

Reliable Automatic Sprinkler Co.

Viking Corp.

 Sprinkler, Drain and Anarm Test Fittings: Tyco Fire Suppression & building Products Fire-End and Coller Corp.

File-Elia and Collection Colp.

G Manuacturing

Victavic Company

Sprickler, Branch-Line Test Fittings:

Elkhart Brass Mfg. Co., Inc.

Fire-End and Croker Corp.

Smith Industries, Inc.; Potter-Roemer Div.

- AFG Manufacturing
- Sprinkler, Inspector's Test Fittings: Fire-End and Croker Corp.

G/J Innovations, Inc. Triple R Specialty of Ajax, Inc. AFG Manufacturing 6. Sprinklers: Tyco Fire Suppression & Building Products Reliable Automatic Sprinkler Co., Inc. Viking Corp. Victaulic Company 7. Fire-Protection-Service Valves: PUT Tyco Fire Suppression & Building Products Central Sprinkler Corp. Nibco, Inc. Stockham Valves & Fittings, Inc. Victaulic Company 8. Keyed Couplings for Steel Piping: (Grooved Tyco Fire Suppression & Building Pro Victaulic Company Viking Corp. Anvil International Grulo 9. Fire Protection Back revention Valve Assemblies (including RPZ Type). W: terworks Ames Fire & Company Watts (Conbraco) PART 3 – EX UTION **NSTALLATION OF BASIC IDENTIFICATION** Install fire protection signs on piping in accordance with ANSI/NFPA 13. FIRE SPRINKLER PIPING SYSTEMS Comply with requirements of ANSI/NFPA 13 for installation of fire sprinkler piping materials. Install

fire sprinkler piping products where indicated, in accordance with manufacturer's written instructions, and in accordance with recognized industry practices to ensure that fire sprinkler piping complies with requirements and serves intended purposes.

3.1

- B. Coordinate with ceiling space available, other trades, including plumbing piping, as necessary to interface components of fire sprinkler piping properly with all other items of construction.
- C. Install drain piping at low points of fire sprinkler piping.
- D. Provide auxiliary drains as required.
- E. Install system valve assemblies where indicated.
- F. Install electric weatherproof bell where indicated.
- G. Install siamese connection where indicated.
- H. Install the following type sprinkler head in the following locations:
- I. Exterior Sprinklers: Exterior sprinkler coverage and protection shall be provided iterany canopy or soffit overhang construction.
- J. Install sprinkler in acoustical tile suspended ceilings, in the center of the alco with heads installed in such a way that the requirements for both coverage and symmetry are fulfilled.
- K. Install dry pipe valve complete with circuit closure and all trimmings equired.
- L. Install wall-mounted air compressor and piping to dry pipe Neve.
- M. No portion of the wet sprinkler system shall be installed within the attic and/or potential freezing areas of the building without being provided with freeze protection.
- N. Provide tamper and flow switches where indicated onoted. All switches shall be wired by the Fire Alarm System Contractor. Provide all coordination and communication with the Fire Alarm Contractor for number and relocation of an switches.
- 3.3 INSTALLATION OF SUPPOPERS, INCLORS AND SEALS
 - A. Comply with the latest issue of NSP/r adopted by the Authorities having jurisdiction.
- 3.4 INSTALLATION OF NRUPLOTECTION SPECIALTIES
- A. Comply with the larst issue of NFPA adopted by the Authorities having jurisdiction.
- 3.5 ADJUST & CLEAN
 - A. Sprinkler hiping Flushing: Prior to connecting sprinkler risers for flushing, flush water feed mains, lead in connections and control portions of sprinkler piping. After fire sprinkler piping installation has been completed and before piping is placed in service, flush entire sprinkler system, as required to remove foreign substances, under pressure as specified in ANSI/NFPA 13. Continue flushing until water is clear, and check to ensure that debris has not clogged sprinklers.

FIELD QUALITY CONTROL

Notify the Authority Having Jurisdiction, the Owner's Representative, and Architect & Engineer of time and date of scheduled testing. Provide minimum of 5 day prior notice of testing to allow for witnessing.

B. Perform all required system testing and acceptance requirements on the new (and modified) system installations in accordance with NFPA 13, 14, & 25, the Delaware State Fire Prevention Regulations, the Authorities Having Jurisdictions (AHJ) requirements and all other local codes and ordinances. At a

Tetra Tech 200-15704-17001

minimum, provide hydrostatic pressure testing on new (and modified) above ground systems piping in accordance with NFPA 13. New system shall be tested to 50 psi over normal system working pressure (minimum 200 psi) for 2 hours without leaks.

- C. Provide all required reports, records, and documentation, to the Owner, Engineer, and Authority Having Jurisdiction prior to or at the completion of the project. At a minimum, provide completed and signed "Contractor's Material and Test Certificate for Aboveground Piping" for each system.
- D. As part of the sprinkler system renovations, the fire protection contractor shall perform an internal visual inspection of the sprinkler piping that is opened up as part of the renovation work (the fit system internal pipe inspection in accordance with NFPA 25). This shall also include a visual inspection of the sprinklers that are removed as part of the renovation work. Any material found within the sprinkler piping and sprinklers that could potentially obstruct piping or plug sprinklers shall be reported to Owner for their review.

3.7 EXTRA STOCK

A. For each style and temperature range required, furnish additional sprinkler heads, amounting to 1 unit for every 100 installed units.

RBIDDI

END OF SECTION 21017

SECTION 220000

GENERAL PROVISIONS - PLUMBING/FIRE PROTECTION

PART 1 - GENERAL

- 1.1 RELATED DOCUMENTS
 - A. The general provisions of the Contract, including the Conditions of the Contract (General, Supplementary and other Conditions, if any) and all other applicable Divisions, apply to work of this Section.
 - B. This specification or drawing and the design features or resulting construction disclosed are the property of Furlow Associates, Inc., and shall not be reproduced without written permission.
 - C. All fire protection suppression systems shall be part of and included in all of the following 220000 thru 220191 Sections.

1.2 WORK INCLUDED

- A. Provide labor, materials, equipment and supervision necessary to install complete operating Plumbing and Fire Protection Systems as indicated the drawings and specified herein, including all work at the site and within the proposed construction areas to accomplish the required work.
- 1.3 REGULATIONS, CODES AND STANDARDS
 - A. Work shall be performed in accordance with lass adopted codes, regulations and ordinances by authorities having jurisdiction. Observe all structure gulations.
 - B. Latest editions of any referenced standards shall govern.
 - C. Obtain all municipal and/or the Authorities Laving Jurisdiction permits and inspection certificates and pay all charges.
 - D. Make or arrange for any/or all espection agency reviews or visits and pay all charges. This includes communication with excharge agency and/or utility to verify the project system work, coordination responsibilities, tees, back charges, etc., required.
 - E. All fees and back charges shall be verified during the bidding phase of the work. Any discrepancy of this item between my utility, inspection agency and the Contractor shall be brought to the attention of the A/E prior to tid opening.
 - F. Subrassion of a bid will be deemed evidence of having complied with these requirements.
- 1.4 **KELATED WORK**

Refer to equipment shown or specified in all other applicable Divisions that require Plumbing and Fire Protection services.

Refer to work related to Plumbing and Fire Protection as shown on the following contract drawings:

Architectural & Structural

HVAC

Electrical

1.5 COORDINATION

- A. The Mechanical, Plumbing and Electrical Contractors are responsible to coordinate all manufacturer's recommended circuit breakers, starters, disconnects and fuse sizes for all equipment. Submission of a shop drawing will certify that this has been completed. Any necessary changes required will be included as part of this contract.
- B. Plumbing and Sprinkler Contractors shall coordinate scheduling, submittals and work of the various sections of the Project Manual to ensure efficient and orderly sequence of installation of independent work elements, with provisions to accommodate items that may be installed at a later time.
- C. Plumbing and Sprinkler Contractors shall verify utility requirements and all characteristics of operating equipment are compatible with the building utilities. Coordinate the work on all sections related and required for installing, connection and placing in service of all equipment.
- D. Plumbing and Sprinkler Contractors shall coordinate all space requirements, supports and installation of all mechanical, electrical, plumbing and fire protection work, which are indicated diagrammatically on the Drawings. Verify routing of all pipes, ducts, conduits and equipment connections. Maximize accessibility for other work, and service requirements for maintenance and repairs. Develop overall coordination drawing (all trades) and submit for review prior to herication/installation.
- E. Obtain written confirmation from all related trade Contractor and the Owner or his representative that requirements, conflicts and coordination issues have been discussed and resolved.
- F. Coordination of Trades in the Field: The Sheet Met I Contractor shall take precedence and, therefore, shall develop his shop drawings first. These then will be used to overlay the other trades. Next shall be the mechanical piping, plumbing, fire sprindler and electrical in the order stated. Drawings shall be 3/8" in scale. Initial meeting of contractors shall be convened prior to start of drawings to work out layout, breakdown of building and other details. All drawings shall be completed in CAD with a format compatible and convertible to DWG files. At the end of the effort, each contractor shall provide a full set of shop drawings to each of the other contractors and three sets to the construction manager. Devices requiring access for maintenance shall not be infringed upon by adjacent trades. Coil pull allowances shall be shown on drawings.

1.6 SUBMITTALS

- A. Shop Drawings & Product Data:
 - 1. Shop grawings and product data shall be submitted in accordance with Division 22 specifications except where herein modified.

List ed are the required shop drawings and reports required for this project. The Engineer/Owner shall reserve the right to require additional submissions not listed below:

- All fixtures, equipment and associated devices so listed on the Fixture Schedule on Drawing.
- Insulation
- All specified piping systems.
- All specified valves.
- Medical Gas Systems, certifier and report.
- Gauges and thermometers

- Recirculating pump.
- Hanger and supports including Sumner system.
- Piping labels and identification.
- Sprinkler System and all related data, devices, switches and trimmings.
- Sump pumps.
- Testing reports.
- Sterilization report.
- Operating/Maintenance manuals.
- As-Built Drawings.
- 3. Submittals comprising complete catalog cuts, shop drawings and verformance test data for Plumbing materials and equipment as required by other sections of D vision 22, shall be submitted for review checking. The Contractor shall review these foreconformance to contract documents prior to submission and affix contractor's signature to each submittal certifying that this review has been done. By approving and submitting shop drawing, product data, samples and similar materials, the Contractor represents that the Contractor has determined and verified materials, field measurements and field construction data that relates to the work, and has checked and coordinated this information with all of the requirements contained in the contract documents for the work of all trades.
- 4. All submittals shall have the following identification data, as applicable, contained therein or permanently adhered thereto.
 - a. Project name.
 - b. Project number.
 - c. Sub-contractory, vendor's and/or manufacturer's name and address.
 - d. Product identification
 - e. Identification of deviation from contract documents.
 - f. A plicable contract drawings and specification section number.
 - nop drawing title, drawing number, revision number, and date of drawing and revision.
 - h. Resubmit revised or additional submittals as requested.
 - Wherever shop drawings or vendor's standard data sheets indicate work to be done "by others", it shall be the responsibility of the contractor making the submission to identify by name, the contractor who is to do this work. If the contractor named is other than the contractor making the submission, the shop drawing submission must be reviewed by the named contractor and bear his mark of approval, prior to submission to the Architect/Engineer.
 - j. Where equipment proposed differs from that shown on the drawings or specified, he shall submit for approval drawings showing the manner in which the layout is affected by the substitution.

- k. The Contractor shall keep one copy of approved shop drawings at the job site,, filed in a suitable metal container. The shop drawings shall be cataloged and kept in good repair, and shall be available for use by the Owner, Architect and Engineer.
- 1. No equipment shall be ordered, fabricated, etc., before approval of shop drawings.
- B. Contractor is responsible for the shop drawing coordination and interface with the work of other contracts and adjacent work. The relationship of Contractor's work shall be verified as it relates to adjacent and critical features of the work of this and all contracts and materials.

1.7 WARRANTY/GUARANTEE

- A. All work and materials are subject to the general warranty as described in the General conditions of the Contract and in all other applicable Divisions. In addition, refer to specifications for special guarantees.
- B. Wherever in the specification sections of this division, reference is made to specific warranty period, this warranty shall be in addition to and not a limitation of other rights be towner may have against the Contractor under the contract documents.
- C. Contractor to include an 11 month "walk-thru" of the building systems with representatives of the School District, Architect, Engineer and the Construction Wan ger. The purpose is to establish a list of corrective work that relates to operational issues, material/estallation deficiencies.
- 1.8 SITE INSPECTION
 - A. The Contractor shall visit the site, inspect, and become aware of all conditions which may affect the work during the estimation phase of his work and prior to bid openings. Investigate utilities, protection requirements for adjacent franties, storage locations, and access to the construction area.
 - B. Submission of a bid will be deemed evidence of having complied with this requirement.

1.9 SUBSTITUTIONS

- A. Whenever a material, an cle piece of equipment or system is identified in the following specification or indicated on the drawners by reference to manufacturers' or vendors' names, trade names, catalog numbers or the likes it is so identified for the purpose of establishing the basis of the Bid.
- B. Substitution approval must be obtained and included as an addendum item prior to the submission of the bid. An approved substitution shall not be considered as an approval for the Contractor or an equipment vender to deviate from the written portion of the specifications unless so stated in the addendua.
- C. The drawings illustrate the space allocated for equipment and the Contractor shall install the equipment accordingly. If changes are required in the building or arrangement due to substitution of equipment, the Contractor making the substitution must pay for the necessary modifications.

The listed equivalent or substituted manufacturers along with the bidding related contractor shall be completely responsible to comply with all requirements as indicated on all contract documents and as described within the specifications. This shall include, but shall not be limited to space requirements, code clearances, the type, horsepower, capacities, number and size of services required from other trades, including all required ancillary items furnished and installed by other trades. If the manufacturer or related bidding contractor does not comply with these requirements, then they shall be responsible for any and all additional costs associated with the changes required by other trades.

1.10 LUBRICATION

- A. Furnish, install and maintain all required lubrication of any equipment operated prior to acceptance by the Owner. Lubrication shall be as recommended by the equipment manufacturer.
- B. Provide one year's supply of lubricants to Owner at date of acceptance.
- C. Verify that required lubrication has taken place prior to any equipment start-up.
- 1.11 EQUIPMENT START-UP
 - A. Verify proper installation by manufacturer or his representative.
 - B. Advise General Contractor 2 days prior to actual start-up.
 - C. Verify proper operation. Obtain signed statement by manufacturer or his representative that equipment is operating within warranty requirements. Submit statement to Scheral Contractor.
- 1.12 OPERATION & MAINTENANCE INSTRUCTIONS
 - A. Properly and fully instruct Owner's personnel in the operation and maintenance of all systems and equipment.
 - B. Insure that the Owner's personnel are familiar with all operations to carry on required activities.
 - C. Such instruction shall be for each item of equipment and ach system as a whole.
 - D. Provide report that instruction has taken place. Include in the report the equipment and/or systems instructed, date, contractor, Owner's personnel vender, and that a complete operating and maintenance manual has been reviewed.
 - E. Manual shall include all instructions on operation, maintenance, repair parts list, lubrication requirements, brochures, catalogue cots, using diagrams, piping diagrams, control sequences, service requirements, names and addresses of vendors, suppliers and emergency contacts. Three manuals shall be provided.
 - F. Submit manuals for review prior to operating instruction period. Manuals shall be 8-1/2 x 11" with hard cover, suitably bound.
 - G. Provide to the Owner any special tools necessary for operation and routine maintenance of any of the equipment
- 1.13 TOOLS
 - A. All equipment furnished by the Contractor which requires special tools or devices other than those formally available to the maintenance or operating staff shall be furnished in duplicate to the Owner, sufficiently marked, packed or boxed for staff usage. The tools provided shall be listed by the Contractor identified as to their use or the equipment applicable in a written transmittal to the Owner.

CLEANING AND FINISHING

. After equipment start-up and all operating tests have been made and the system pronounced satisfactory, each respective Contractor shall go over the entire project, clean all equipment, etc., installed by him and leave in a clean and working condition. Any surfaces found marred after this final cleaning shall be refinished or replaced by each Contractor at no cost to the Owner.

- B. Provide for the safety and good condition of all materials and equipment until final acceptance by the Owner. Protect all materials and equipment from damage. Provide adequate and proper storage facilities during the progress of the work. Special care shall be taken to provide protection for bearings, open connections, pipe coils, pumps, compressors and similar equipment.
- C. All NEW fixtures, piping, finished surfaces and equipment installed shall have all grease, adhesive labels and foreign materials removed.
- D. All new piping installed shall be drained and flushed to remove grease and foreign matter. Tress dre regulating assemblies, traps, flush valves and similar items shall be thoroughly cleaned. Remove and thoroughly clean and reinstall all liquid strainer screens after the system has been in operation to (10) days.
- E. Gas piping shall be blown out with clean compressed air or inert gas.
- F. When connections are made to existing systems, the Contractor shall do all chaning and purging of the existing systems required to restore them to the condition existing price to be start of work.
- G. Clean-up: Remove from the premises, all unused material and debris resulting from the performance of work under this section.

PART 2 – PRODUCTS

- 2.1 GENERAL
 - A. All material and equipment shall be new and of preserv day manufacture, and shall conform to accepted standards of the trade where such a standard has been established for the particular type of equipment or material.
 - B. Whenever equipment or material is referred to in the singular, such as "the plumbing fixture", it shall be deemed to apply to as many such items as necessary to complete the work.
- 2.2 PRODUCT DELIVERY, STORAGE AND HANDLING
 - A. During loading, transporting and unloading exercise care to prevent damage to material.
 - B. Store all materials in dry exposures or under protective coverings out of way of work progress.
 - C. Material shall not be allowed to be stored directly on ground.
 - D. Deliver in manufacturer's original cartons or on skids.
 - E. Hand e and protect so as to prevent damage to product or any surrounding material.
- 2.3 CONCRETE

Concrete if used on this project, shall be in accordance with Section 033000.

The 28-day minimum compressive strength shall be 3000 psi.

PRT 3 – EXECUTION

3.1 PROTECTION

- A. Plug or cap open ends of piping systems.
- B. Stored materials shall be covered to prevent damage by inclement weather, sun, dust or moisture.

- C. Protect all installed work until accepted in place by the Owner.
- D. Plates, polished metal escutcheons and other finished devices shall not be installed until masonry, tile, and painting operations are complete unless otherwise protected.
- E. Protect all work from operations which may cause damage such as hauling, welding, solder painting, insulating and covering.
- F. Do not remove protective material until equipment is placed in service.

3.2 WORKMANSHIP

- A. Install all work neat, trim and plumb with building lines.
- B. Install work in spaces allocated.
- C. Cutting and patching shall be performed by skilled tradesmen non-pally employed for the work involved.

3.3 EXCAVATION

- A. The excavation shall be of the open-trench method and to the epths and widths as may be necessary. The Contractor shall do all excavation required in connection with his work. Bottoms of trenches shall be excavated to a uniform grade. All materials excavated shall be deposited on the side of the trenches and beyond the reach of the slides. Excavate, material shall not be piled where it will interfere with traffic. If rock is encountered it shall be removed by the General Contractor. See provisions in Division 2.
- B. No piping shall be bedded directly on rock. They shall be cushioned by a 6-inch layer of crushed stone or gravel of selected grade on izeto pass through 3/4" mesh sieve. Not less than 30% shall be fine which will pass through a 1/8" mesh sieve.

3.4 SHORING AND PUMPING

- A. The Contractor shall provide all shoring, bracing or sheet piling necessary to maintain the banks of his excavation and shall take out same as the work progresses and filling in has been accomplished. Shoring shall be inacceedance with OSHA Standards.
- B. The arran ement of shoring must be such as to prevent any movement of the trench banks and consequent trains on the conduits. Shoring shall be provided to prevent damage to work installed by other traines.
- C. The Contractor shall do all pumping required to keep his excavations free of water. The water shall be conveyed in piping or watertight troughs a sufficient distance that it will flow from the site and not affect other work being performed.

BACKFILLING

- After work in trenches has been completed, they shall be filled with select fill in 8" layers and shall be pneumatically tamped before the next layer of material has been filled in. The backfill shall be free of excavated rock, cinders, stones, brickbats or other debris.
- B. Wherever rock is removed, the Contractor shall secure and fill select clean earth to a minimum depth of 3'-0" above the top of the pipe. Unless otherwise indicated, no rock shall be deposited in the trench

fill. This clean earth fill shall be procured other than from the site unless permission for earth borrow from the site is granted by the Architect. If site borrow is permitted, the topsoil removal, relocation and finished grading will be accomplished as directed by the Architect.

- C. Under no circumstances shall excavated material be left where it will interfere with the Owner's or other Contractor's operations.
- D. All earth and other materials taken from the trenches and not required for backfilling shall deposited where directed, or removed from the premises as directed by the Architect.
- E. Any rock removed from the excavation shall be removed from the project site by the Contra
- F. Trenches which pass under wall footings or within 18" of column footings shall be backfilled with lean concrete. To secure adequate foundation support, the method and depositing of the concrete fill shall be as directed by the Architect. To prevent the concrete from adhering to he pipes, necessary pipe protection shall be applied.

3.6 EQUIPMENT SETTING

- A. Furnish and install as a minimum, a 4 inch concrete pad beneathful floor-mounted equipment. Install anchor bolts in pour.
- B. Furnish and install as a minimum, spring vibration isolation us der any equipment 10 HP and over and rubber in shear vibration isolation on any equipment up to 10 HP.
- C. Concrete shall be 3,000 psi, 28 day compressive treacth is accordance with ACI-613. Reinforce with No. 4 rod 12" on centers both ways or as otherwise detailed.
- 3.7 FASTENERS, HANGERS AND SUPPORTS
 - A. Furnish and install all hangers and supports required to suspend, mount, or hang the work.
 - B. Furnish and install all miscellaneous steel angles, channels, beams, clips, brackets and anchors necessary to hang or support the work. Provide submissions for review.
 - C. Install concrete inserts of the concrete is poured.
 - D. Drilled inserts shall ot be loaded more than 1/4 rated capacity or 200 pounds.
 - E. Power-driver fost pers shall not be allowed for piping larger than 2 inch, or equipment. When used they shall not be paded more than 1/8 rated capacity or 200 pounds.
 - F. All hangers, muscellaneous steel, braces and supports shall be galvanized, cadmium plated, or primed steel. copper tubing shall be supported with copper hangers. No direct contact of dissimilar metals between the piping system and its hanger support shall be permitted.

Piping shall be supported from adjustable clevis type hangers with insulation pipe saddles. Where hangers are 18" or longer, provide lateral bracing at every fourth hanger. See IPC Pipe Support Table below:

| Material | Horizontal Max. Feet | Vertical Max. Feet |
|-------------------------|-------------------------|-----------------------|
| | | |
| Aluminum | 10 | 15 |
| Brass | 10 | 10 |
| Brass Tube up to 1-1/4" | 6 | 10 |
| Brass Tube over 1-1/2" | 10 | 10 |
| Cast Iron | 5 | |
| Copper up to 1-1/4" | 6 | |
| Copper over 1-1/4" | 10 | 10 |
| CPVC Up to 1" | 3 | 10 |
| CPVC Over 1" | 4 | 10 |
| Lead Pipe | Con inuou | 4 |
| PB Pipe/Tubing | 261 (32") | 10 |
| PVC Pipe | | 10 |
| PEX | 2.6 ft. (32") | 10 |
| Steel Tubing | 8 | 10 |
| Steel Pipe | 12 | 15 |

PIPE SUPPORT SPACING

H. Support vertical piping at floor levels using approved riser clamps. Clamp material shall be compatible with pipe material. Maximum vertical spacing shall be10'-0".

3.8 SLEEVES

A. Provide each pipe passing through a masonry or concrete wall, floor or partition with a sleeve made from standard weight steel pipe for pipe with smooth edges, securely and neatly cemented in place. Provide each pipe passing through a frame or metal partition with a sleeve made from No. 22 gauge galvanized sheet metal, securely fastened in place.

Pipe passing through foundation wall or under foundation shall be provided with relieving arch or steel pipe per IPC Section 305.5.

- C. Be responsible for the proper location and alignment of all sleeves.
- D. Provide hydrostatic seals for sleeves passing through outside walls, below grade, or through hydrostatically sealed slabs or floors on grade. Provide fire-rated seals for all other sleeves.

- E. Install both piping and sleeve seals so as to maintain integrity of seals with expansion and contraction of piping.
- F. Set floor sleeves flush with floor surface in finished areas, 1" above the finished floor in kitchens, cafeterias, and similar service areas unless such areas are slab-on-grade; 1" above the floor in mechanical rooms, pipe chases, pipe spaces and other unfinished areas, unless otherwise indicated, and flush with the underside of slabs. Extend wall and partition sleeves through and cut flush with each surface unless otherwise indicated or specified.
- G. Select sleeves two pipe sizes larger than any pipe that is to remain uncovered, unless otherwise required by the sealing method specified. Where pipes are to be covered, provide deves large enough to allow the covering to pass through the sleeves with sufficient clearance for sealing as specified hereinafter. Size sleeves for branch piping from vertical risers large enough to permit vertical expansion at the riser.
- H. Place sleeves imbedded in concrete floors or walls in the forms before concrete s poured; sleeves shall have integral waterstop flanges, where they are to receive either wate tighter hydrostatic seals.
- I. Install sleeves passing through above-grade floors of mechanical rooms, toilet rooms, kitchens or similar service areas where liquid leaks or spillover may occur in a wat rtight manner. Sleeves shall be such that waterproofing membrane can be flashed around and into the sleeve where necessary.
- J. Seal sleeves for pipes passing through ceiling air plenum walls or the floor above air tight in a manner similar to that specified for fire-rated sleeves.
- K. Hydrostatic Sealing Method: Provide compressible symmetic rubber seals, equivalent to LINK SEAL, manufactured by the Thunderline Corporation, or THRUWALL manufactured by O.Z. Gedney. Install seals in accordance with the manufacturer's recommendations to provide air tightness aboveground and hydrostatic sealing below rade. Caulking or other type mastic is not acceptable.
- L. Fire-Rated Sealing Method:
 - 1. Sleeves, openings and sear intschall comply with applicable codes, recommended practices and standards, and manufacturer instructions. Fire sealants shall have ability to prevent spread of flame, smoke or water throughout the penetration and shall pass 3 hour test, UL test ASTM E814 and UL 1479
 - 2. Products: Cruse Corporation CTC PR-855, O. Z. Gedney CRS/CAFS, 3M Electro-Products Division Puty 303 or Caulk CP25 penetration sealing kits, General Electric Company sealants type RCV-850, 6428 or 7403, Thunderline Corporation "Link-Seal Pyro-Pak". Installation and type of sealant to be used as recommended by the manufacturer.

Expansion collars, fire seal/firestop collars – ASTM E814 (UL1479). Spec Seal Corporation, Inc. (plastic pipe).

PLATES

Furnish and install chrome plated plates wherever piping passes into finished area.

- B. Plates shall be securely fastened to piping or building construction.
- C. Floor plates shall cover 1 inch sleeve extension.
- 3.10 OFFSETS, TRANSITIONS, MODIFICATIONS
 - A. Furnish and install all offsets necessary to install the work and to provide clearance for other trades.

GENERAL PROVISIONS – PLUMBING/FIRE PROTECTION 220000-10

- B. Maintain adequate headroom and clearance.
- C. Incidental modifications necessary to the installation of the systems shall be made as necessary and as approved by the Architect.
- 3.11 RECESSES
 - A. Furnish information to the General Contractor as to sizes and locations of recesses required to instrupanels, boxes, and other equipment or devices which are to be recessed in walls.
 - B. Make offsets or modifications as required to suit final locations.
- 3.12 LABELING
 - A. All Plumbing equipment such as pumps, and devices requiring identification for operating procedures shall be provided with permanent black laminated micarta white core labele with 3/8 inch letters.
 - B. This shall also apply to all controllers, remote start/stop pushbuttons and equipment cabinets.
- 3.13 FLASHING AND COUNTERFLASHING
 - A. Roof drains, vents, roof curbs, etc., shall have counterflashing httings. General Contractor shall provide flashing.
 - B. Piping and conduit thru the roof shall be flashed by the General Contractor. Furnish and install counterflashing.
- 3.14 ACCESS
 - A. Locate all equipment, valves, devices and controllers which may need service in accessible places.
 - B. Where access is not available, access panels shall be provided. Furnish access doors to the General Contractor for installation.
 - C. Access doors shall be Elmdor, Kap Co., MIFAB or Controlled Air Manufacturing Limited, with 16 gauge frames and 14 gauge tree cloor, prime painted.
 - D. Maintain required access corrances.
- 3.15 WIRING
 - A. Packaged alumbing system equipment shall be furnished with disconnect switches, and magnetic starters, factory farnished and wired by the unit manufacturer.
 - B. All coverol wiring shall be furnished and installed under this Division of the work.
 - All wiring shall be in accordance with the National Electrical Code and as recommended by the equipment manufacturer.

UTILITIES

- Do not interrupt any utility or service to the Owner without adequate previous notice and schedule.
- B. Arrange and pay for the relocation, disconnection or removal of, or relocate, disconnect or remove existing utilities and services where such work is shown or where such utilities or services interfere with new construction, whether or not shown. Provide all excavation, backfilling and paving required by such work.

C. Perform alteration of utilities and services in accordance with the rules, regulations and requirements of the involved utility companies, regulatory agencies having jurisdiction.

3.17 CUTTING AND PATCHING EXTERIOR SURFACES

- A. This Contractor shall be responsible for returning disturbed paved and/or grass areas to original condition where excavation for utilities has been required.
- B. Cut and patch paved areas to match original surface.
- C. Properly tamp backfill before finishing or repairing disturbed area surfaces.

3.18 OPENINGS - CUTTING, REPAIRING

- A. This contractor shall cooperate with the work to be done under other sections reproviding information as to openings required in walls, slabs and footings for all piping and equipment, including sleeves where required.
- B. Any drilling or cutting required for the performance of won, under this Section, shall be the responsibility of this Contractor and the cost thereof shall be borned, him.
- C. Holes in Concrete: Sleeves shall be furnished, accurately located and installed in forms before pouring of concrete. This contractor shall pay all additional costs for cutting of holes as the result of the incorrect location of sleeves. All holes through existing concrete shall be either core drill or saw cut. All holes required shall have the approval of the Suructural Engineer prior to cutting or drilling.
- D. It shall be the responsibility of this Contractor to ascertain that all chases and openings are properly located.

3.19 GUARANTEE

A. All materials and equipment provided and/or installed under this section of the specifications shall be guaranteed for a period of one year from the date of acceptance of the work by the Owner unless otherwise specified in one applicable Divisions. Should any trouble develop during this period due to defective materials or factory workmanship, the Contractor shall furnish all necessary labor and materials to correct the trouble without any cost to the Owner. Any defective materials or inferior workmanship noticed at time of installation and/or during the guarantee period shall be corrected immediately to the entire satisfaction of the Owner.

In the event of occupancy by the Owner prior to final acceptance of the project, the guarantee date for equipment placed in operation shall be mutually agreed to by the Contractor and the Owner's representative.

Contractor to include an 11 month "walk-thru" of the building system with representatives of the School District, Architect, Engineer and the Construction Manager. The purpose is to establish a list of corrective work that relates to operational issues, material/installation deficiencies.

DRAWINGS

A. The Plumbing and Fire Protection Systems are indicated on the Contract Drawings. Certain pertinent information and details required by the Plumbing and Fire Protection Work appear on the Architectural, Structural and Electrical Drawings; become familiar with all Drawings; and incorporate all pertinent requirements.

- B. Drawings are diagrammatic and indicate the general arrangement of systems and requirements of the Work. Do not scale Drawings. Exact locations of fixtures and equipment, not specifically shown shall be obtained before starting work.
- C. When indicated on the drawings, plumbing riser diagrams are completely diagrammatic and indicate the intent of the work for both the Contractor, L&I review agencies and/or Authorities Having Jurisdiction. Where valves, shock absorbers, incidental equipment, devices, etc., including execution notes are indicated on the riser diagrams, they shall be so required and installed as part of the system work.

3.21 RECORD DRAWINGS

A. As-Built record drawings, showing dimensions, locations and depth of all bured and concealed piping, plugged outlets and equipment shall be kept up to date. Master copy shall be kept on the job. No backfilling of trenches shall be permitted until as-built drawings are approved as up-to-date by the Owner/Representative. No plumbing progress payments shall be approved rules as-built drawings are up- to-date. Depth of sewers shall be from a permanent bench mark an shown on the contract drawings. Refer to project record drawings under General Conditions.

2 BIDDIE

END OF SECTION 220000



SECTION 220010

BASIC MATERIALS AND METHODS – PLUMBING

PART 1 – GENERAL

1.1 RELATED DOCUMENTS

A. The general provisions of the Contract, including the Conditions of the Contract (General, Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the work specified in this Section.

1.2 REFERENCE

A. Install all piping, fixtures, equipment, etc., to meet the requirements of the following

New Castle County Department of License and Inspection

New Castle County Department of Sewers

International Plumbing Code

New Castle County Fire Marshal's Office

NATIONAL Plumbing Code

International Plumbing Code (All applicable section

International Mechanical Code (All applications)

International Fuel Gas Code (All applicable sections)

Water Company

NFPA

OSHA

All requirements of the above governing agencies shall be in compliance with the latest issues, rules or regulations in effect.

- B. Appliances and materials governed by UL requirements shall meet such requirements and bear the label.
- 1.3 QUALITY ASSURANCE
 - A. Provide adequate supervision of labor force to assure all aspects of specifications are being fulfilled.

Inserve that all work and equipment is installed in accordance with manufacturer's warranty requirements.

Replace all pipes and fittings shown to be defective as a result of testing.

1.4 SUBMITTALS

- A. Submit shop drawings and product data in accordance with Section 220000.
- B. Submit the following:
 - 1. Manufacturer's Product Data on all pipe and fittings to be used in project.

Tetra Tech 200-15704-17001
2. Manufacturer's Product Data on all valves to be used in project.

1.5 WARRANTY/GUARANTEE

A. All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, GENERAL REQUIREMENTS.

PART 2 – PRODUCTS

- 2.1 STEEL PIPE & FITTINGS
 - A. Pipe: ASTM A-53, seamless, Schedule 40.
 - B. Fittings:
 - 1. Cast iron, threaded, 175 psi, ANSI B-16.4.
 - 2. Malleable iron, threaded, ASA B 16.3.
 - 3. Steel, socket weld, ASTM A-53.
 - 4. Wrought iron, socket weld, ASTM A-72.
 - C. Thread tape shall be teflon tape, 3 mils minimum thickness. Terlon tape shall not be permitted for use on gas piping systems.
 - D. See Section 220130 for Gas Piping Systems.
- 2.2 CAST IRON PIPE AND FITTINGS

(Note: Any cast iron piping made or marked "CMINA" will NOT be acceptable on this project)

- A. Aboveground:
 - 1. Pipe & Fittings: Hubless cast iron CISPI 301, ASTM A-74 and ASTM A-888 shall be marked with the collective trademark of the Cast Iron Institute (soil pipe).
 - 2. Joints: Neoprene slove and stainless-steel shield and clamp assembly, CISPI 310, ASTM-1277.
- B. Below grade and or lab: (Contractor's Option)
 - 1. Bell and Spiget: Service weight bell and spigot pattern ASTM-74 with compression type neoprine gastets ASTM C-564.
 - 2. Hubless: Hubless cast iron pipe CISPI 301, with heavy duty 3.04.016 stainless steel bands for below-grade installation. Elastomeric seal component ASTM C-564 and CSA B-602.
 - Hubless Joints: Cast iron CISPI 310 and as TM C-1277.
 - PVC DWV pipe and fittings, Schedule 40, ASTM D-2665, D2949, F891 and CSA B181.2.
 - . Corrosion protection shall be in accordance with IPC 305.1. Provide appropriate wrapping or sheathing when pipe is exposed to lime and acid of concrete, cinder or other corrosive materials.
 - 6. Protection of all below-grade storm and sanitary shall be in accordance with IPC Section 305.
 - 7. All Kitchen and Boiler Room below slab piping shall be extra heavy schedule cast iron only. PVC not allowed.

C. Corrosion protection shall be in accordance with IPC 305.1. Provide appropriate wrapping or sheathing when piping is exposed to lime and acid of concrete, cinder or other corrosive materials.

2.3 COPPER TUBING

- A. Domestic hot, cold and recirculated water:
 - 1. Aboveground:
 - a. Tubing: Hard-drawn, seamless ASTM B-88, Type "L".
 - b. Fittings: Solder joint wrought copper ANSI B-16.22.
 - c. Joints: Lead-free solder 410°, ASTM B-32 alloy designation "TC", ASTM B-
 - d. Flux: Non-toxic and non-corrosive, ASTM B-813.
 - 2. Underground:
 - a. Tubing: Soft-drawn, seamless ASTM B-88, Type "K"
 - b. Fittings: Solder joint wrought copper ANSI B-16.22
 - c. Joints: Lead-free solder 410°, ASTM B-32, ASTM B-828.
 - d. Flux: Non-toxic and non-corrosive, ASTM B-8
- B. Drainage and vent piping:
 - 1. Aboveground:
 - a. Tubing: Hard-drawn seamlers ASTM B-88, ASTM B-75, Type "M" and DWV as pipe size permits.
 - b. Fittings: Solder joint cast corper drainage type ANSI B-16.29.
 - c. Joints: Solderer, 9 /5 th-antimony ASTM B-828, ASTM B-32.
 - d. Flux: Non-toxic and non-corrosive, ASTM B-813.
- C. Solder/Flux: Se Paragraph 3.4 of this section for Soldering/Brazing.
- 2.4 DUCTILE IRON PIPE
 - A. Piper Ductile non, ANSI A-21.51, ANSI/AWWA C151.
 - B. Joints: Rubber gasket, ANSI A-21.11, ANSI/AWWA C111.
 - Fixings: Mechanical joint, ANSI/AWWA C110, C153 bolt tolerances AWWA C-111, ASTM A-563.
 - Lining: Cement mortar, ANSI A-21.4, ANSI/AWWA C104.
- 2.5 PVC GRAVITY SEWER PIPE
 - A. Pipe: Unplasticized polyvinyl chloride (PVC) with integral wall bell and spigot joints.
 - B. Material: ASTM D-3034 for SDR 35, colored green for inground identification as sewer pipe.

- C. Joints: Two sections of pipe shall be assembled in accordance with manufacturer's recommendations and tested as per ASTM D 3212 for use with flexible elastomeric seals.
- D. Sizes: For site drainage systems 4" to 15".
- E. Additional compliances:
 - 1. Drop Impact Test ASTM D-2444
 - 2. Pipe Stiffness ASTM D-2412
 - 3. Temperature for Testing Designed to pass all tests at 73 degrees F (+/- 3 degrees E)
- 2.6 POLYVINYL CHLORIDE (PVC) PIPE AND FITTINGS
 - A. Aboveground Drainage & Vent (Sanitary) IPC Table 202.1
 - 1. ASTM D 2665
 - 2. ASTM D 2949
 - 3. CSA CAN/CSA B 181.2
 - 4. ASTM F 1488
 - 5. ASTM F-81
 - B. Underground Drainage & Vent (Sanitary) IPC Table 7/2
 - 1. ASTM D 2665
 - 2. ASTM D 2949
 - 3. ASTM F 891
 - 4. CSA CAN/CSA-B 181.2
 - C. Building Sewer Pipe (Near Water Service) IPC Table 702.3 (DWV)
 - 1. ASTM D 2667
 - 2. ASTM D 2941
 - 3. ASTM D 301
 - 4. ASTM F 891
 - . CSA B182.2

6. CSA B 182.4 (Ribbed Sewer Pipe & Fittings)

Fittings:

- 1. ASTM D 3311
- 2. ASTM D-2665
- 3. ASTM F-1866

E. Solvent Cement: (All Purpose on ABS, PVC and CPVC)

Potable Water, Sewer, Drain Waste and Vent

- 1. ASTM D-2564, D-2235 and F-493
- 2. CSA B137.3
- 3. CSA B181.2 or B182.1 (Sanitary Pipe only)
- 4. ASTM D2855
- 5. CSA B181.1
- F. Primers: (PVC and CPVC)
 - 1. ASTM F 656, purple color, SCAQMD Rule 1168 and OTC Regulations for VOC emission levels. NSF Standard 61 PW, DWV, Sewer.
- G. Uniformity: To insure installation uniformity, all piping components shall be of one manufacturer.
- 2.7 POROUS CONCRETE PIPE
 - A. Pipe & Fittings: Porous concrete drain pipe, A.A.H.O. destruction M176.
 - B. Joints: Interlocking tongue and groove.
- 2.8 REINFORCED CONCRETE PIPE AND FITTING
 - A. Pipe & Fittings: Reinforced concrete, ANSI/A STM C-75, Class 2.
 - B. Joints: Modified tongue and groove with compression gasket, ANSI/ASTM C-443.
- 2.9 POLYPROPYLENE PIPE & F/1 TRVG
 - A. Pipe & Fittings: Polypropylete hum retardant ASTM D-2146 Schedule (40) (80).
 - B. Joints: (Aboveground)
 - 1. Mechanical
 - 2. Fusion welder socket ends.
 - C. Joints: (Below ground) Fusion welded socket ends
- 2.10 PLENUM RATED PVDF PIPE & FITTINGS/CORROSIVE WASTE DRAINAGE SYSTEM
 - A. Pipe & Fittings: Polyvinyllidene fluoride (PVDF), ASTM F-1673, pipe shall be marked with "UL" to indicate compliance with UL723 (ASTM E84).
 - Joints (Aboveground)
 - 1. No hub, plain end, outerban, nuts and bolts per ASTM B117.
 - 2. Socket Fusion: ASTM 2657, ASTM D3222.
- 2.11 FLOWGUARD GOLD CPVC PIPE
 - A. Scope: This specification covers the manufacturing requirements for CPVC SDR 11 Copper Tube Size (CTS) pipe and fittings. Both the pipe and fittings are manufactured in North America and meet

or exceed the requirements set forth by the American Society for Testing Materials (ASTM) and ANSI/NSF Standards 14 and 61.

- B. CPVC Materials: FlowGuard Gold[®] CPVC pipe and fittings are extruded/molded from CPVC compounds manufactured by Lubrizol. The pipe compound meets cell class 24448 and the fitting compound meets cell class 23447 as defined by ASTM D1784. Both the pipe and the fitting compounds are certified by NSF International for use with potable water.
- C. Dimensions And Properties:

Dimensions, tolerances and physical properties meet or exceed the requirements of AS IM

- D. Solvent Cement: All socket type joints shall be assembled employing solvent cements that meet or exceed the requirements of ASTM F493. The standard practice for safe handling of solvent cements shall be in accordance with ASTM F402. Solvent cement shall be listed by MSF international for use with potable water, and approved by the FlowGuard Gold[®] pipe and fittings canufacturers.
- E. Flame And Smoke Requirements: Water filled FlowGuard Gold[®] pipe and mains (1/2" through 2") tested in general accordance with UL 723/ASTM E 84 (NFPA 255 and UBC 8-1) meet the 25/50 flame and smoke requirement and shall be permitted to be installed in return air plenums. Test reports from a third party testing laboratory shall be obtained and made available upon request.
- F. Marking: The marking on the CPVC pipe and fittings meet the requirements of ASTM D2846 and state the pipe/fitting manufacture's name or trademark, the material designation, the size, the NSF mark for potable water and the ASTM designation (ASTM D2846).

2.12 CORZAN CPVC PIPE

- A. Scope: This specification covers the pane facturing requirements for CPVC Schedule 80 Iron Pipe Size (IPS) pipe and fittings. Both the pipe and fittings are manufactured in North America and meet or exceed the requirements set for any be American Society for Testing Materials (ASTM) and ANSI/NSF Standards 14 and 6
- B. CPVC Materials: Corzan CPVC pipe and fittings are extruded/molded from CPVC compounds manufactured by Lubriz V rhepipe compound meets cell class 24448 and the fitting compound meets cell class 23447 as defined to ASTM D1784. Both the pipe and the fitting compounds are certified by NSF International for use with potable water.
- C. Dimensions and Properties: Dimensions, tolerances and physical properties meet or exceed the requirements of ASTM Standards F441 for pipe, F439 for socket fittings and ASTM F437 or F439 for threated fittings. Threaded fittings have taper pipe threads in accordance with ASTM F1498. Unions and flatges meet or exceed the requirements of ASTM F1970.

solvent Cement: All socket type joints shall be assembled employing solvent cements that meet or exceed the requirements of ASTM F493. The standard practice for safe handling of solvent cements shall be in accordance with ASTM F402. Solvent cement shall be listed by NSF International for use with potable water, and approved by the Corzan[®] pipe and fittings manufacturers.

Flame And Smoke Requirements: Water filled Corzan[®] pipe and fittings (1/2" through 6") tested in general accordance with UL 723/ASTM E 84 (NFPA 255 and UBC 8-1) meets the 25/50 flame and smoke requirement and shall be permitted to be installed in return air plenums. Test reports from a third party testing laboratory shall be obtained and made available upon request.

- F. Marking: The marking on the CPVC pipe meet the requirements of ASTM F441 and the marking on the fittings meets the requirements of ASTM Standards F437, F438 or F1970. The pipe and fittings markings state the pipe/fitting manufacture's name or trademark, the material designation, the size, the NSF mark for potable water and the ASTM designation.
- G. To ensure compliance with Green Building Design and Construction under IEQ Credit 4.1 for Low-Emitting Materials (adhesives and sealants), all interior use adhesives, sealants and sealant primers shall comply with the South Coast Air Quality Management District (SCAQMD) Rule #11fe. All primers and cements used for joining CPV and CPVC piping shall comply with the following Votatile Organic Compound (VOC) limits when calculated according to 40 CFR 59, subpart D (VPA Melod 24):

| PVC welding | -510 g/L less water |
|----------------------------|---------------------|
| CPVC welding | -490 g/L less water |
| Adhesive Primer of Plastic | -550 g/L less water |

H. Valves - Plastic PVC & CPVC: Valves listed below shall be for dometric water systems and comply with ASTM Standard D 1784, Rigid PVC and CPVC compounds. Classes 12454 (Type 1, Grade 1) PVC and 23447 (Type 4, Grade 1) CPVC.

All PVC and CPVC Valves are listed by the NSF International to NSF/ANSI Standard 14: Plastic Piping System Components and Related materials. This Adependent third-party agency certifies that products and materials bearing the 'NSF-pw' marking are regularly tested to comply with ASTM F 1970 Standard Specification for Special Engineered Fittings, Appurtenances or Valves for use in Poly (Vinyl Chloride) (PVC) or Chlorinated Poly (Vinyl Chloride) (CPVC) Systems, as well as, certifying these products conform to the requirements of NSF/ANSI Standard 61: Drinking Water System Components-Health Effects.

Ball and Ball-Check Valves are 100% pressure tested for shell and seat leaks in accordance with quality standards established by NIBCO Engineering. The rejection point is a leak greater than 60 cc per minute.

All True Union-Tru-Blog Ben Valves, True Union Check Valves, Model-B Butterfly Valves, Chemcock Valve, Teedle Valve, Angle and Y-Pattern Valves are manufactured and assembled silicone free. Lebricane are occasionally used to assemble these valves, but these lubricants contain no silicone

Butterfly Wlyes shall be assembled using a lubricant containing silicone.

The above listed valves shall be manufactured in an ISO 9001:2000 certified facility.

valves-Plastic Manufacturers - Subject to compliance with requirements, provide valves of one of the following:

Chemtrol

Haywood

Ipex

- 2.13 VALVES (Copper Systems) Solder ends of Threaded
 - A. Valves listed below shall be for domestic water systems and comply with the latest requirements of

NSF 61-8. Refer to individual sections for gas valves.

- B. Ball Valves: NIBCO two piece, full port, 600 psi WOG rated, cold non-shock valve with reinforced TFE seals, 316 stainless steel ball, Eco-brass body, ASTM 584, Alloy C87850, solder ends, or threaded non-blowout stem design. Acceptable NIBCO figure numbers: T/S 685-80-66-LF; T/S 595-Y-66-LF (3 piece).
- C. Check Valves: NIBCO Class 125, Eco-brass body, ASTM 584, Alloy C87850, swing type, Y Pattern threaded cap access. Acceptable NIBCO figure number: T/S 413-LF.
- D. Gate Valves: NIBCO Class 125, Eco-Brass body, ASTM 584, Alloy C87850, I isi Acceptable NIBCO figure number: T/S 113-LF.
- E. Balance Valves: All balance valves shall be provided with a memory stop feature with calibrated name plate to assure specific valve setting. Bronze body/brass ball, carboa filed FrE seat rings. NIBCO, Bell & Gosset, Accu-Flow, Taco or Flow Design "Accusetter" Accuptable NIBCO figure numbers: T/S 1710, F/G 737.
- F. Strainers:
 - Class 125 Bronze Y-Strainer, body to be ASTM B584 or 362 bronze with threaded, solder or female press end connections and .033 inch perforated upp 304 stainless steel screen or 20 mesh type 304 stainless steel screen accessible without removing the strainer from the line. Acceptable Figure numbers: NIBCO Fig. S/T-221, S/T-222, PF 221/222-A,B.
 - 2. Class 125 Flanged Cast Iron Y-Strainer, body to be ASTM A-126 Class B cast iron. End connections to be Class 125 flanged, tap, edited bonnet with plug. Screen shall be .033 inch perforated type 304 stainless steel screen of 20 mesh type 304 stainless steel screen accessible without removing the strainer from the line. Acceptable Figure numbers: NIBCO Fig. F 721-A.
 - 3. Class 250 Threaded Cast Iron & Stepher: Strainer body to be ASTM A-126 Class B cast iron. End connections to be Class 250 threaded, tapped screw-in bonnet with plug. Screen shall be .033 inch perforated type 304 stabless steel screen or 20 mesh type 304 stainless steel screen accessible without removing the strainer from the line. Acceptable Figure numbers: NIBCO Fig. T-751-A
- G. VALVES (Copter Systems) Press Fit
 - 1. Valves lister below shall be for domestic water systems and comply with the latest requirements of NSI-61-8



Inch and Smaller Ball Valves (On/Off):

Ball Valves with male or female press to connect shall be rated at 200 PSI CWP to +225°F maximum. Valves shall be manufactured in accordance with MSS SP-110 and constructed of dezincification resistant cast bronze bodies. Brass with more than 15% zinc shall not be approved. Valve shall have reinforced PTFE Seats, Blow-out Proof Stem, Full Port Ball, Chrome/Nickel Plated or Stainless Steel Ball for aggressive water.

b. 2 Inch and Smaller Check Valves (Swing Type):

Check valves shall be swing type Y pattern with male or female press to connect ends and shall be rated 200 PSI CWP to + 250°F maximum. Valves shall be manufactured in accordance with MSS SP-80. Body & cap shall be manufactured of dezincification resistant cast bronze ASTM B62 or ASTM B584 Alloy C8440. Valves shall have PTFE seat disc.

c. 2 Inch and Smaller Check Valves (Lift or Spring Type):

Incline resilient disc, spring actuated, 250psi rating, non-shock cold working pressure, 2500F maximum working temperature, bronze ASTM B584 alloy C84400. Stainless steel stem and disc holder and spring, EDPM O-ring.

PUF

- H. Insofar as possible, all valves of the same type shall be of the same manufacturer.
- I. Valve Manufacturers: Subject to compliance with requirements, provide valves of one of following:

Apollo/Conbraco Stockham Nibco

Milwaukee

Watts

Hammond

Webstone

- J. System Application:
 - 1. Domestic Water:
 - a. Check Valves 2" & Smaller threaded or soldered.
 - b. Ball Valves 3" & Smaller thread or soldered.
 - c. Balance Valves All sizes breaded.
 - d. Butterfly Valves 4" and loger flanged.
 - e. Butterfly Valves 3" nd smaller wafer type.
- 2.14 THERMOMETURS
 - A. Separable secket inserted into fluid flow, adjustable, hermetically sealed, red mercury, die-cast, baked enamel finish, double strength glass lens, white scale and black graduations.
 - B. Scale. Steet range of thermometer to indicate normal operating temperature at mid-point of scale for domestic water systems.
 - Manufacturer: U.S. Gauge, H.O. Trerice, Moeller, Duro.

GAUGES

Phosphor bronze bourdon tube, polypropylene case, gasketed glass crystal, aluminum dial, black graduations 4-1/2 inch diameter.

- B. Range: 0 to 150 psi, 5 pound intervals, 1/2 pound graduations.
- C. Manufacturers: Danton, U.S. Gauge, H.O. Trerice, Moeller.
- D. Install with bronze gauge cock.

2.16 ISOLATING FITTINGS

- A. Furnish isolating fittings between all sections of dissimilar piping materials or piping, general supports, equipment and supports, including piping hanger and rack supports where one material is ferrous and the other is non-ferrous.
- B. Install copper or brass piping or tubing in such a way as not to touch or come in contact with ferrous metals.
- C. Where ferrous piping or equipment is connected to copper or brass piping, make connection with insulating or dielectric unions to prevent electrolytic action between the ferrous and non-ferrous metals.
- D. Where copper or brass piping, tubing or fittings are anchored to, supported by or may core in contact with ferrous metal construction, provide an insulating nonconductor space of mober, fiber or equivalent material to assure prevention of electrolysis.
- E. Manufacturer: Epco Sales, Inc., or insulated unions by Central Plastic Co.

2.17 ANCHORS AND GUIDES

- A. Anchors and guides shall be provided to support and mainten pixes in position and properly distribute expansion. The anchors and guides must be securely fastenee to the building structure, and must be completely installed before the system is tested.
- B. Guides shall be as manufactured by J.J. McNally, Jne. Fleconics, Inc., Tube-Turns, American District Steam Co.

2.18 UNIONS

- A. Up to and including 2 inch pipe size: Screwed pattern, bronze-to- bronze seat.
- B. Above 2 inch pipe size: 125 Cass Flanged pattern, A.S.A. sweat copper fitting, with gaskets, bolts and nuts.
- C. Copper tubing unions shall have sweated type ends. Flanged unions on copper tubing may be soldered connections.
- D. Materials and pressure rutings shall be the same as specified for the respective pipe and fitting system unless otherwise specified.

2.19 MOTORS

A. All single phase and polyphase motors shall be manufactured to incorporate the latest NEMA standards.

All bingle phase and polyphase motors shall have steel frames with ball bearings and copper windings. All motors to have a Class "F" insulation system with a service factor of 1.15.

All motors shall be 1725 RPM, 4 pole design, unless otherwise noted on the drawings, or in the equipment specifications.

- D. Motors installed indoors and not exposed to moisture shall be open, dripproof, Class B temperature rise based on 40 deg. C maximum ambient temperature.
- E. Motors installed outdoors and exposed to moisture shall be totally enclosed, fan cooled, Class B temperature rise based on 40 deg. C maximum ambient temperature.

| | Nominal Effici | encies for "N | EMA Premiun | n TM " Inductio | on Motors | |
|-----|--|---------------|---------------|----------------------------|-----------|---------|
| | Ra | ted 600 Volts | or Less (Rand | om Wound) | | |
| | Open Drip-Proof Totally Enclosed Fan-Coo | | | -Cooled | | |
| | | | | | | |
| HP | 6-Pole | 4-Pole | 2-Pole | 6-Pole | 4-Pole | 2. Jole |
| | | | | | | |
| 1 | 82.5 | 85.5 | 77.0 | 82.5 | 85.5 | |
| 1.5 | 86.5 | 86.5 | 84.0 | 87.5 | 86.5 | 8.0 |
| 2 | 87.5 | 86.5 | 85.5 | 88.5 | 86.5 | 85.5 |
| 3 | 88.5 | 89.5 | 85.5 | 89.5 | 97 | 86.5 |
| 5 | 89.5 | 89.5 | 86.5 | 89.5 | 89.5 | 88.5 |
| 7.5 | 90.2 | 91.0 | 88.5 | 91.0 | 917 | 89.5 |
| 10 | 91.7 | 91.7 | 89.5 | 91.0 | 71.7 | 90.2 |
| 15 | 91.7 | 93.0 | 90.2 | 91.7 | 92.4 | 91.0 |
| 20 | 92.4 | 93.0 | 91.0 | <u>01.</u> , | 93.0 | 91.0 |
| 25 | 93.0 | 93.6 | 91.7 | 97.0 | 93.6 | 91.7 |
| 30 | 93.6 | 94.1 | 91.7 | 93.0 | 93.6 | 91.7 |
| 40 | 94.1 | 94.1 | 92.4 | 94.1 | 94.1 | 92.4 |
| 50 | 94.1 | 94.5 | 93.0 | 94.1 | 94.5 | 93.0 |
| 60 | 94.5 | 95.0 | 93. | 94.5 | 95.0 | 93.6 |
| 75 | 94.5 | 95.0 | ¥2.0 | 94.5 | 95.4 | 93.6 |
| 100 | 95.0 | 95.4 | 9.6 | 95.0 | 95.4 | 93.6 |
| 125 | 95.4 | 95.4 | 4.1 | 95.0 | 95.4 | 94.1 |
| 150 | 95.4 | 95.8 | 94.1 | 95.8 | 95.8 | 95.0 |
| 200 | 95.4 | 95.8 | 94.1 | 95.8 | 96.2 | 95.0 |

F. Based on NEMA Standards motors shall comply with the following minimum nominal efficiencies at full load.

G. Motor Characteristics: Lefter to Equipment Schedules for specific data.

120/208 Volt System: Motors 1/2HP & Larger - 208V, 3 Phase, 3 Wire

Motors Less than 1/2HP- 120V, 1 Phase, 2 Wire

System: Motors 1/2HP & Larger - 480V, 3 Phase, 3 Wire

Motors Less than 1/2HP-120/277V, 1 Phase, 2 Wire.

H. All motors rated less than 1/2HP shall have thermal protection of the auto-reset type as an integral part of the motor.

All motors rated 1/2HP and larger shall have thermal protection provided by an external device.

POWER FACTOR CORRECTION AT MOTORS

A. All motors 3 HP and above shall be provided with capacitors sized as recommended by the motor manufacturer to correct the full load running current of the motor to 95 PF, or to the maximum PF value as determined by the total capacitance which the particular motor design and duty can safely accommodate. Capacitors shall be wired to the load side of the starting contacts ahead of the overload protective devices so that motor and capacitor are switches simultaneously.

- B. Capacitors for motors with starters located in a motor control center shall be furnished and mounted on the rear of the motor control center by the Electrical Contractor.
- C. Where the motor, starter, and wiring area integral part of a packaged type equipment assembly, the capacitors shall be furnished, wired and mounted on the equipment by the manufacturer.
- D. Multi-speed motors with two or more windings shall have capacitors on each speed (or winding) arranged with auxiliary contactors so that when that winding is not energized the capacitor is disconnected.
- E. For motors connected to high inertia loads, a time delay device shall be provided to prevent restar of that motor until its speed has been reduced and the charge remaining in the capacitor has reached a safe level for restart.
- F. Power capacitors shall be 3 phase, industrial grade, mounted in a dust tight eaclo ure with removable shroud. Sprague Power Capacitors, Square D, General Electric or approved substitute.

PART 3 – EXECUTION

3.1 PIPING SYSTEM INSTALLATION REQUIREMENTS

- A. Drawings are generally diagrammatic and due to small scale, it is impossible to indicate all fittings, valves, gauges and specialties required. Provide complete operating systems and all necessary fittings, valves gauges and specialties whether or not indicated.
- B. Install all piping in accordance with the best practices of the trade and latest code requirements. Use uniform system materials throughout the building. All branch take-offs shall be off the top of the pipe.
- C. Pipe and fittings shall be clean from cutting burrs, foreign materials and defects in structure and threading. Make all cuts square. Ream after cutting. Clean off scale and dirt inside and outside, before assembly. Remove welding slav or other foreign material.
- D. Keep all piping as high as possible, consistent with proper pitch, to maintain maximum headroom. Cut piping accurately to measurements established at the building, work into place without springing, forcing or cutting of the buncing structure, and install as directly as possible between connecting points parallel with or at night angles to building construction, except as required to obtain pitch.
- E. Unless otherwise shown, run piping within the building, concealed in the walls, furred spaces, pipe spaces or above suppended ceilings. Unless otherwise noted, do not build in or bury horizontal piping in partitions. Install all exposed piping as closely as possible to walls, ceilings and columns, consistent with access indepplicable insulation requirements.
- F. This preject includes a return air plenum ceiling. Regardless of materials specified, all system piping and/or materials shall be non-combustible and shall be in full compliance with the requirements set forth in the IPC.

All piping to drain to low points. Low points will be provided with drain valves with hose thread. All piping shall have high points vented with ball valve, nipple and threaded cap.

- H. Do not install trapped lines where water cannot be drained or air can accumulate without being vented.
- I. Piping shall run square with building lines.
- J. Piping shall not be insulated or covered until tested and until building is closed in.

- K. Necessary drains, off-sets, vents and drips shall be provided for coordination of the work as part of the contract.
- L. Piping shall not be installed over electrical transformers, panels, switchgear, substations, and control panels as per the National Electric Code. No piping shall be installed in elevator machine rooms unless it is directly related to the room's system equipment.
- M. Allow clearance for expansion and contraction.
- N. Install isolating fittings between sections of ferrous and non- ferrous pipe or connected equip
- O. Valves shall be installed with stems above horizontal.
- P. Valves shall be installed on all sides of equipment and control valves to allow is lation for repair.
- Q. Do not support piping from other piping, conduits or equipment. Provide additional bracing to prevent movement of trapeze piping, or any singular run of pipe to fixtures. Provide additional bracing on all piping through walls to flush valves to prevent movement during neuroperation or performing maintenance on valves.
- R. Thermometers and gauges shall be installed where indicated cathedrayings, required by equipment specifications and where indicated elsewhere in the specifications. Gauges shall be located at an elevation that can be readable.
- S. Unions shall be provided adjacent to all valves, attequipment connections, and where necessary to facilitate dismantling of the piping system.
- T. Ball valves to be installed with the proper charance for operating the valve handle. A minimum clearance of 10" from center of valve to walknust be maintained for ease of operation.
- U. Thermometers are to be located so the can asily be seen from the floor in front of unit. Make final adjustment by tilting thermometer. Locate bulb in waterway with an oversized tee or elbow fitting.
- V. Install pressure gauges on incoming rervices both domestic water and fire services. Locate pressure gauge after main shut-on varie and ahead of water meter if one is provided within building.
- W. All pipe unions installed shall be accessible. Unions shall not be concealed or located in places where they cannot be maintained.
- X. Support and braving of 4" and above pipe shall be in accordance with the CISPI Standards and IPC Chapter 3.
- 3.2 TAGS CHARTS, AND IDENTIFICATION
 - All piping shall be labeled in accordance with IPC 303.1 and 303.4.
 - Identify each valve in all systems with black, numbered and stamped 1-1/2" brass or aluminum tags fastened to valve by brass chain and S-hook.
 - Piping Identification: Provide identification and safety products, semi-rigid plastic, wraparound pipe markers with flow arrows and conforming to ANSI A13.1. Locate marker at each valve, changes in direction, where pipes pass thru barriers and every 25' of horizontal runs. Lettering on background shall be in accordance with the following colors:

| Legend | Background | Lettering |
|--------|------------|-----------|
| 1. Gas | - Yellow | - Black |

ADDITION AND RENOVATIONS

| 2. Fire Protection | - Red | - White |
|--|----------|---------|
| 3. Domestic Cold Water | - Green | - White |
| 4. Domestic Hot Water (110° ^ 140°) | - Yellow | - Black |
| 5. Domestic Hot Water Return (110° ^ 140°) | - Yellow | - Black |
| 6. Sanitary Drainage | - Green | - White |
| 7. Condensate Drainage | - Yellow | - Black |
| 8. Vent | - Yellow | - Black |
| 9. Storm Drainage | - Green | - White |

- D. Provide 1/8" scale diagrams showing location, number and service or function of cash tagger item.
 - 1. Frame diagrams in approved metal frames with clear acrylic front, hinges and l
 - 2. Secure to wall in Mechanical Room.
 - 3. Provide two additional separate copies permanently covered and our
- E. Furnish and install color coded 1" diameter markers on ceiling tile grid to indicate system and valve locations.
 - 1. Domestic cold water: Green
 - 2. Domestic hot water: Yellow
 - 3. Domestic hot water return: Yellow
 - 4. Gas
- F. Available Manufacturers: Subject to compliance with requirements, manufacturer's offering identification markers which may be incorrorated in the work are limited to the following:

- Yeh

Seton

Brimar

B-Line

Marking Service In

- 3.3 WELDIN
 - A. All concelled and inaccessible black steel piping shall be welded.
 - B. All blacksteel piping larger than 2 inch shall be fusion welded.
 - Anelbows, tees and branch connections shall be made with welding fittings ANSI B16.9.
 - Welding shall be in accordance with the ASME Boiler and Pressure Vessel Code Section IX.
 - Furnish welder test certificate for review. Certificates of successful qualification by the following organizations shall be acceptable.
 - 1. ASME Boiler and Pressure Vessel Code
 - 2. ANSI Code for Pressure Piping
 - 3. National Certified Pipe Welding Bureau

4. Military Specification MIL-STD-248

3.4 SOLDERING/BRAZING

- A. Connections between copper tubing and copper sweat fittings shall be made by soldering using Taramet Sterling or approved substitute. Flux shall be non-corrosive type "Nokorode" or approved substitute or as recommended by the manufacturer of the solder.
- All solder shall be "lead nickel and antimony free" in accordance with the Federal Safe B. Water Act Amendments of 1986 and 1996 as is ASTM B-32 Grade TC.

Composition:

| Tin | 95% |
|---------------------|------------|
| Copper | 4.0 - 5.0% |
| Selenium | .042% |
| Tensile Strength | 7,130 psi |
| Shear Strength | 5,970 psi |
| Melting temperature | 410°F |

- Tubing shall be cut square and then reamed and d burks. End of tubing and inside of fitting cup C. shall be cleaned with steel wool and the flux shall be applied to the clean surface before soldering. After soldering, the excess solder shall be wired of while still plastic.
- D. Silver brazing alloy shall be equal to and U be used for joints in:
 - 1. Medical Gas Piping (All System
 - 2. Medical Vacuum Piping
- E. **Brazed Joints:**
 - 1. All brazed joints share be cleaned. An approved flux shall be applied; joint filler metal shall conform to AVN A5.8.
 - AWS Standard A5.31, Type F83-A or F83-C. 2. Flux shall
- F. 410 solder shall be used for all joints in:
 - nestic cold water Ď
 - Domestic hot water
 - Domestic hot water return
 - Copper drainage piping
 - 5. Plant compressed air
- G. Lead-Tin (50-50) solder or any solder containing lead shall NOT be used or permitted for joint connections on this project.

- H. Where the silver brazing is performed in a confined non-ventilated space, a non-toxic, cadmium-free brazing alloy such as Stay-Brite shall be used instead of Easy-Flo. Bring joint to solder temperature or brazing temperature in as short a time as possible.
- I. Form continuous solder bead or brazing filler bead around entire circumference of joint.
- J. Wipe excess solder from joint area while solder is still plastic.
- K. Solder joints shall be in accordance with IPC Section 605.2, 605.14.3 and ASTM B838. Fbx s conform to ASTM B-813.
- 3.5 PRESS-FIT SYSTEM
 - A. All new domestic water piping installed on this project shall be a solderless, press it, donestic water system. The system shall be Viega/Rigid copper press fitting system. Fitting such a be rated 0□ to 250□ at 200 psi and tested to 600 psi.
 - B. Fittings shall meet ANSI/NSF 61, ASME B-16.22 and ASTM B88. Electomeric seals shall meet ASTM D-2000.
 - C. Mechanical joining shall be recognized by:

IPC International Plumbing Code

SBCCI Standard Plumbing Code

IAPMO Uniform Plumbing Code

PHCC National Standard Plumbing Code

- D. Copper and copper alloy press fittings shall conform to material requirements of ASME B16.18 or ASME B16.22 and performance criteria of IAPMO PS 117. Sealing elements for press fittings shall be EPDM. Sealing elements shall be factory installed or an alternative supplied by fitting manufacturer. Press end shall have SC (Smart Connect) feature design (leakage path). Smart ConnectTM (SC Feature) in Propress ½" to 4" dimensions, the Smart Connect Feature assures leakage of liquids and/or gases how middle the system past the sealing element of an unpressed connection. This feature shall provide the installer quick and easy identification of connections which have not been pressed prior to putting the system into operation.
- E. Press Connectees. Copper press fitting joints shall be made in accordance with the manufacturer's installation instructions. The tubing shall be fully inserted into the fitting and the tubing marked at the shoulder of be fitting. The fitting alignment shall be checked against the mark on the tubing to assure the tubing is fully engaged (inserted) in the fitting. The joints shall be pressed using the tool approved by the manufacturer.

Inseller shall be a qualified installer, licensed within the jurisdiction, and familiar with the installation of ProPress copper press joint systems. ProPress copper press fittings shall be installed using the proper tool, actuator, jaws and rings as instructed by the press fitting manufacturer. The installation of copper tubing for hot and cold-water distribution systems shall conform to the requirements of the ICC International Plumbing Code or IAPMO Uniform Plumbing Code.

G. Note: Viega Press-fit installation shall only be permitted on this project. Push-on shark-teeth, or any type connection fittings that are not Press-Fit, shall NOT be approved.

H. T-drill mechanically formed tee fittings shall be used in conjunction with the ProPress Copper System in accordance with the IPC Chapter 6 Section 605.5.1, 605.5.1.2 and 605.14.1. Use caution around combustible material and follow all safety guidelines for open flame during silver brazing.

END OF SECTION 220010

Republication

Tetra Tech 200-15704-17001



SECTION 220030

INSULATION & COVERING – PLUMBING

PART 1 – GENERAL

1.1 RELATED DOCUMENTS

- A. The general provisions of the Contract, including the Conditions of the Contract (General, Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the Vork specified in this Section.
- 1.2 DESCRIPTION OF WORK
 - A. This section includes insulation and covering furnished and installed on the following piving systems and equipment:
 - 1. Domestic cold water.
 - 2. Domestic hot water supply and return
 - 3. "Primary" Horizontal rainwater conductors including unlerside of roof drains. "Secondary" rainwater systems insulation is not required.
 - 4. Condensate waste piping from air conditioning and
 - 5. Exposed waste, trap and wall supplies at alkhaudicap lavatories.
 - 6. Branch waste lines from all chilled water counterns.
- 1.3 REFERENCE STANDARDS
 - A. Refer to Section 220000 for a general rescription of requirements applying to this section.
 - B. Materials shall conform to the requirements of the NFPA Code.
- 1.4 QUALITY ASSURANCE
 - A. Refer to Section 220010 four general description of requirements applying to this section.
- 1.5 SUBMITTALS
 - A. Submit ship dravings and product data in accordance with Section 220000.
 - B. Subt it the following:

Product data on all insulation and covering.

WARRANTY/GUARANTEE

All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, GENERAL REQUIREMENTS.

PART 2 – PRODUCTS

- 2.1 PIPE INSULATION MATERIAL
 - A. Fiberglass:
 - 1. Material: Preformed fiberglass bonded with resins to form circular pipe sleeves with factory

applied, white all-service jacket bonded to reinforced foil vapor barrier jacketing. The jacket shall have factory-applied double pressure-sensitive adhesive closure and vapor sealing of longitudinal joints. Thermal Conductivity: .25 per inch at 100 degrees F. Flame spread of 25 and developed smoke of 50 or less.

- 2. All Valves and Fittings:
 - a. Class fiber insert and premolded PVC cover, Manville "Zeston" and "Hi-Lo Temp Insert" for valves and fittings.
 - b. Factory molded fibrous glass fitting covering for fittings.
 - c. Mitered sections of pipe covering for valves.
- 3. Manufacturers: Johns-Manville, Certain-Teed, Owens-Corning.
- B. Closed Cell:
 - 1. Material: Flexible elastomeric foamed plastic closed cell structure asulauon 25/50 rated with a flame spread rating of 25 or less and a smoke developed rating of 50 or less.
 - 2. Flexible pipe insulation shall be a foamed plastic closed cell structure material, with a thermal conductivity of not more than 0.27 Btu/Hr./Sq. Ft./Inchesta mean temperature of 75 degrees F. The insulation shall have an average density of at least 2 pounds per cubic foot, shall be self-extinguishing, and shall have a water vapor transmission rating of not more than 0.1 perms. Between temperature limits of -40 degrees F and plas 220 degrees F, the insulation shall not indicate any deviation from its original gate.
 - 3. Manufacturers: Armacel, Insul-Tube Nomeso Insulation.
 - 4. Specification Compliance: (Lates accepted Standards and Codes)
 - IECC 804.5: Insulation thickness for domestic hot and recirculation mains.
 - ASTM-E-84 In me toread and smoke developed.
 - NFPA 255: Sandard method of test of surface burning of building materials.
 - ASTM C174: _____Thermal conductivity.
 - NFPA 90A, 90A: Flame & smoke rating
 - M-54 Type 1 Tubular Grade, Self-Sealing

Factory made air ducts and air connectors. (Armacell UL181 has to do with mold growth)

Test for surface burning characteristics of building materials.

- UL723
- ASTM G21/C1338: Fungi resistance

ASTM G2: Bacterial Resistance

ASTM D1056, 2B1: Standard spec for flexible cellular materials.

MIL-P-15280J, FORMT

MIL-C-3133B (MIL STD 670B) Grade SBE-3

MEA 96-85M

- C. Covering of Pipe Insulation Outdoors:
 - 1. Wrapping: Wrap insulation with embossed .016" aluminum jacket.
 - 2. Fastenings: Cover shall be held in place with soft aluminum bands on 12" centers.
 - 3. Valves and Fittings: Weatherproof all valves and fittings.
 - 4. Manufacturers: Johns-Manville, Certain-Teed, Owens-Corning, Knauf.
- D. Protective cover for foam insulation in wet areas indoors:
 - 1. PVC heavy duty fitting covers and jacketing for kitchen wet areas.
 - 2. Fitting covers shall be glossy white, high impact, UV resistant PVC.
 - 3. Operating Temperature Limit: Up to 150°F.
 - 4. Flame Spread: 25 or less.
 - 5. Smoke Developed: 50 or less.
 - 6. Grade: Weatherable.
 - 7. Color: White
 - 8. Finish: Gloss
 - 9. Fitting covers and jacketing shall be "Zestor" 300 Series PVC, heavy duty covers and "Zestor" PVC jacketing.

PART 3 - EXECUTION

- 3.1 INSTALLATION
 - A. Do not install until systems have been tested and meet requirements.
 - B. Do not install until uilding is closed in.
 - C. Heavy work which may damage insulation shall have been completed in the vicinity of the insulation work.
 - D. All installations shall be made by skilled craftsmen regularly engaged in this type of work.
 - E. Insulation shall be continuous thru-wall, ceiling and floors.
 - Pre and equipment to be clean and dry prior to insulating.

Install all insulation in strict conformance with manufacturer's instructions.

Where "Barrier-free" lavatory supplies and waste are covered with a protective covering or insulation, the insulation must be installed back to wall, flush with wall escutcheon. Escutcheon to be finished flush with wall and wall opening to be smaller than escutcheon plate through entire building.

I. All electrical heat tracing installations shall be coordinated with the electrical contractor. No insulation shall be installed until the heat trace wiring is completely installed, tested and approved. All insulation materials and installation work shall be the responsibility of the Insulation Contractor.

- J. Install pipe insulation by slitting tubular sections and applying onto piping or tubing. Alternately, whenever possible, slide unslit sections over the open ends of piping or tubing. All seams and butt joints shall be adhered and sealed using Armaflex 520 or 520 BLV Adhesive. If when using AP Armaflex SS, only the butt joints shall be adhered using Armaflex 520 or 520 BLV Adhesive, Armaflex HT 625 Adhesive shall be used with HT Armaflex.
- K. Insulation shall be pushed onto the pipe, never pulled. Stretching of insulation may result in open seams and joints.
- L. Tape the ends of the copper tubing before slipping the Armaflex insulation over the property prevent dust from entering the pipe.
- M. All edges shall be clean cut. Rough or jagged edges of the insulation shall not be permitted. Proper tools such as sharp, non-serrated knives must be used.
- N. On cold piping, insulation shall be adhered directly to the piping at the bigh and of the run using a two-inch strip of Armaflex 520 or 520 BLV Adhesive on the ID of the insulation and on the pipe. All exposed end cuts of the insulation shall be coated with Armaflex 520 or 520 BLV Adhesive. All penetrations through the insulation and termination points must be adhered to the substrate to prevent condensation migration.
- O. Sheet insulation shall be used on all pipes larger than 6" IPS. Insulation shall not be stretched around the pipe. On pipes larger than 12" IPS, adhere insulation directly to the pipe on the lower 1/3 of the pipe.
- P. Seams shall be staggered when applying multiple overs of insulation.
- 3.2 VALVES, FLANGES AND FITTINGS:
 - A. All fittings shall be insulated with the same insulation thickness as the adjacent piping. All seams and mitered joints shall be adhered with Annahex 520 or 520 BLV Adhesive. Screwed fittings shall be sleeved and adhered with a minimum 1" overlap onto the adjacent insulation. Armaflex HT 625 Adhesive shall be used with HT Annahex.
 - B. Valves, flanges, strainer and Victaulic couplings shall be insulated using Armaflex donuts that shall then be covered with sheet oversized tubular insulation.
- 3.3 HANGERS
 - A. Support poing system using high density inserts with sufficient compressive strength. The pipe support insolation shall be elastomeric foam with the same or greater thickness than the pipe insulation. All joints shall be sealed with Armaflex 520 or 520 BLV adhesive.
 - B. Standard and split hangers: Piping supported by ring hangers shall have hangers insulated with the same insulation thickness as the adjacent pipe. All seams and butt joints shall be sealed with Armaflex 520 or 520 BLV Adhesive. Armaflex HT 625 Adhesive shall be used with HT Armaflex. Ring nangers may be sleeved using oversized tubular insulation. On cold piping, insulation shall extend up the hanger rod a distance equal to four times the insulation thickness. Insulation tape may be used to a thickness equal to the adjacent insulation thickness.
 - C. Clevis Hangers or other pipe support systems: Saddles shall be installed under all insulated lines at unistrut clamps, clevis hangers or locations where the insulation may be compressed due to the weight of the pipe. All piping shall have wooden dowels or blocks of a thickness equal to the insulation inserted and adhered to the insulation between the pipe and the saddle.

It is highly recommended for continuous insulation protection to use hanger sizes equal to the outer diameter of the pipe plus insulation thickness

- D. Armafix IPH o Armafix NPH can be used to prevent compression of insulation at standard split, clevis hangers or other pipe support systems. To minimize the movement of Armafix, it is recommended that a pair of non-skid pads be adhered to the clamps. In addition, to prevent loosening of the clamps use of an antivibratory fastener, such as a nylon-locking nut, is also recommended.
- 3.4 OUTDOORS EXPOSED PIPING
 - A. All outdoor exposed piping shall be painted with two coats of WB Armaflex Finish. Prior to epplying the Finish, the insulation shall be wiped clean with denatured alcohol. The Finish shell not be inted.
 - B. All outdoor exposed piping shall have the seams located on the lower half of the pipe.

3.5 PIPE COVERING (FOAMED PLASTIC TYPE)

A. All joints and seams shall be sealed with a compatible adhesive. Approved edhesives are as follows:

Armacel

No. 520 (Low VOC use 520 BLV

Benjamin Foster Company No. 85-75 up to 200 degrees

Contractor may use self-sealing insulation in lieu of above.

B. Fitting covers shall be fabricated from the foamed clastic tipe insulation or from sheet insulation of the identical material. The fabrication shall be in accordance with manufacturer's instructions, and all seams mitered joints shall be joined using the albed ives described.

3.6 PIPE INSULATION – TYPES & THICKNESSES

A. Flexible Closed Cell:

| Piping System | Up to 3" | Over 3" to 6" | Over 6" |
|---------------------------------|------------------------|---------------|----------|
| | | | |
| Cold Water | 1/2" | 1/2" | 3/4" |
| Hot Water (120°) | 1" | 1" | 1-1/2" |
| Hot Water Return () 20°) | 1" | 1" | 1-1/2" |
| Hot Water (1408) | 1" | 1" | 1-1/2" |
| Hot Wate, Return (140°) | 1" | 1" | 1-1/2" |
| Condensate Walte | 1/2" | 1/2" | - |
| Hor zop al Storm (Primary) | 1/2" | 1/2" | 3/4" |
| Horizontal Storm (Secondary) | Not Requ | uired | |
| Underside of Roof Drains | 1/2" | 1/2" | 3/4" |
| Branch Waste From EWC's | 1/2" | | |
| Handicap Lav Waste & Water | 1/2" | | |
| Soil/Waste Piping Above Ceiling | 1/2" | 1/2" | 3/4" |
| Soil/Waste Piping Above Ceiling | ⁷ 2 1/2" | 1/2" | 3/4" |

B. Fiberglass:

| Piping System | Up to 3" | Over 3" to 6" | Over 6" |
|---------------|----------|---------------|---------|
| Cold Water | 1/2" | 1/2" | 3/4" |
| Hot Water | 1" | 1" | 1-1/2" |

ADDITION AND RENOVATIONS

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

| Hot Water Return | 1" | 1" | 1-1/2" |
|---------------------------------|------|------|--------|
| Hot Water | 1" | 1" | 1-1/2" |
| Hot Water Return | 1" | 1" | 1-1/2" |
| Condensate Waste | 1/2" | 1/2" | |
| Horizontal Storm (Primary) | 1/2" | 1/2" | 3/4" |
| Horizontal Storm (Secondary) | 1/2" | 1/2" | 3/4" |
| Underside of Roof Drains | 1/2" | 1/2" | |
| Soil/Waste Piping Above Ceiling | 1/2" | 1/2" | 3/4" |

END OF SECTION 220030

Republication

 $\langle \langle \cdot \rangle$

SECTION 220110

DRAINAGE SYSTEMS – PLUMBING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. The general provisions of the Contract, including the Conditions of the Contract (General Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the work specified in this Section.
- 1.2 DESCRIPTION OF WORK
 - A. This section includes:
 - 1. Soil and waste piping system work as indicated on drawings and schedures and by requirements of this section.
 - 2. Applications for soil and waste piping systems include the following
 - a. Above ground soil, waste and vent piping within buildings including soil stacks, vent stacks, horizontal branches, traps and connections to fixtures and drains.
 - b. Underground building drain piping including mains, branches, traps, connections to fixtures and drains, and connections to stacks, extension from the building, terminating at connection to site sewer.
 - 3. Storm water drainage piping as indicated on drawings and by requirements of this section.
 - 4. Applications for storm water drainage signing include the following:
 - a. Roof drains and connections to gutters, with rain water conductors and connections to underground building sorm crains.
 - b. Underground by adding sorm drains, extending and connecting to site drainage system.
 - 5. Insulation for soil and wate and storm water drainage as specified in Section 220030 is included as work of this section.
 - 6. Trenching and backfilling required in conjunction with underground building drainage and site drainage piping as specified in Section 220000 is included as work of this section. Refer to Division I

Installation of detectable metallic underground tape for <u>all</u> exterior buried PVC drainage piping.

REFERENCE STANDARDS

Refer to Section 220000 for a general description of requirements applying to this section.

QUALITY ASSURANCE

A. Refer to Section 220010 for a general description of requirements applying to this section, and a listing of all applicable codes.

1.5 SUBMITTALS

A. Submit shop drawings and product data in accordance with Section 220000.

- B. Submit the following:
 - 1. Product data on all systems equipment.
- C. See requirements for submission of cross referencing information.
- 1.6 WARRANTY/GUARANTEE
 - A. All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, GENERAL REQUIREMENTS.

PART 2 – PRODUCTS

- 2.1 PIPING UNDERGROUND
 - A. Interior:
 - 1. Sanitary, storm water and condensate waste drainage piping within the building and extending beyond the building wall, unless otherwise noted on the plans share beam option selection of a, b, or c below:
 - a. Service weight hub and spigot pattern cast iron soil any and fittings with neoprene gaskets.
 - b. Hubless cast iron soil pipe and fittings with cast iron coupling clamps and gaskets or heavy duty 3.04-.016" thick stainless-steel bands.
 - c. PVC Schedule 40 pipe and fittings with solvent cement joints.
 - 2. Kitchen Sanitary Drainage and/or Mechanical Room Sanitary Drainage: All Kitchen and/or Mechanical Room sanitary below slab proing and fittings shall be service weight cast iron hub and spigot fitting with butyl rubber caskets or hubless fittings with heavy duty couplings (no PVC shall be acceptable).
 - B. Exterior:
 - 1. Stormwater drainage piping 10 inches and smaller, and all sanitary drainage piping unless otherwise noted on to plats, shall be:
 - a. Service reight hub and spigot pattern cast iron soil pipe and fittings, with neoprene gaskets.
 - b. Hubies, cart iron soil pipe and fittings with cast iron coupling clamps and gaskets.
 - c. Unplasticized PVC sewer pipe and fittings SDR-35.
 - Ste mwater drainage piping 12 inches and larger, shall be:
 - a. Reinforced concrete pipe and fittings.
 - b. Corrugated metal pipe.
 - c. As identified on the drawings.
 - 3. Foundation drainage piping shall be:
 - a. Porous concrete pipe and fittings.
 - b. Perforated PVC pipe and fittings Schedule 40.

2.2 PIPING ABOVE GROUND

- A. All above ground storm water, condensate, soil, waste and vent piping shall be:
 - 1. Hubless cast iron soil pipe with cast iron drainage fittings, couplings and stainless steel clamp bands for piping 2" and larger.
 - 2. Copper tubing, type DWV with wrought copper solder type drainage fitting for piping smaller than 2" in size.
 - 3. PVC pipe and fittings type DWV with solvent cement joint connections. (Not permitted in a of plenum rated ceilings.)
- 2.3 CONDENSATE WASTE PIPING SYSTEM
 - A. All aboveground condensate waste piping including connection to equipment shall be
 - 1. PVC pipe and fittings type DWV with solvent cement joint connection). (Not permitted in areas of plenum rated ceilings.)
 - 2. Copper tubing, type DWV with wrought copper solder type trainage fittings.

2.4 FLASHING

- A. All vents extending through the roof shall be flashed by the General Contractor. However, the Plumbing Contractor shall furnish and install the necessary counterflashing consisting of a Jay R. Smith Figure 1748 counterflashing fitting, or approved spostitute as manufactured by Josam or Zurn. Vents shall terminate 18" above the roof.
- 2.5 SPECIAL EXPANSION COMPENSATION
 - A. Special expansion compensation products required for storm, condensate, soil and waste piping systems include the following types:
 - B. Cast Iron Drainage System Expansion Joints: Cast-iron body, adjustable bronze sleeve, bronze bolts with wing nuts; for vertical matanation only.
 - C. PVC Drainage System Expansion Joints: Factory prelubricated "O" ring expansion joint fitting. Installation must be a strict conformance with manufacturer's recommendations.
 - D. Available Manufacturers: Subject to compliance with requirements. Manufacturers offering expansion joints which may be incorporated in the work include:
 - Cyc. Iron Piping Systems J.R. Smith or approved substitute.
 - PVC Piping Systems George Fisher or approved substitute.

BACKWATER VALVES / FLOODGATE

- Backwater valves (BWV) installed in the interior or exterior building sewer extension, shall be all PVC construction materials.
- B. The backwater valve shall be in compliance with the latest edition of IPC and UPC Plumbing Codes.
- C. Description:
 - 1. The BWV shall be as manufactured in accordance with "ICC-ES" criteria.

- 2. The BWV shall be 3- and 4-inch diameter backwater valves, shall meet the performance requirements of CSA B181.1 and CSA B181.2. The factory kit consists of a valve body, disc, disc seat, and upper collar. The factory supplied plastic materials conform to NSF 14-99 and ASME A 112.14.1.
- 3. Removal of the integral lifting device from a buried valve allows above-grade replacement of the disc assembly and reinstallation into the valve body. The integral lifting device is self-aligning self-seating, and provided with an alignment indicator located within 12 inches (305 mm of the upper access opening.
- 4. Floodgate / Backwater Valve Basin

Valve basin must be installed in a ventilated and dry access pit in the horizontal drain line between the building and the sewer main.

Provide access pit of a size that will provide suitable room to inspect and or replace the valve.

Floodgate / backwater valves are approved for storm and sanitary drain systems. Consult with the local plumbing and sewer authority for any additional requirements.

2.7 SYSTEMS EQUIPMENT

- A. Refer to Plumbing Fixture and Equipment Schedule for type number, size and manufacturer of all drainage equipment and accessories.
- B. Available Manufacturers: Subject to compliance with requirements, manufacturers offering drainage equipment which may be incorporated in the work are limited to the following:



| | Josam |
|--------|--------------------------------------|
| | Wade |
| | Watts |
| | Smith |
| | MIFAB |
| | Downspout Shoes |
| | Neenah |
| | Zurn |
| | Josam |
| | Wade |
| | Smith |
| | Watts |
| | MIFAB |
| | Interceptors |
| | Zurn |
| | Josam |
| | Wade |
| | Watts |
| | Smith |
| | Woodford |
| | MIFAB |
| | Backwater warwes & Traps (Cast Iron) |
| | Zurn |
| | Josam |
| | Nade |
| | Watts |
| \sim | Smith |
| | Red Valve Co. |
| | MIFAB |
| | Backwater Valve (PVC) |
| | Clean Check, Inc. |
| | |

Tetra Tech 200-15704-17001 Wall Spout Nozzles

Zurn

Josam

Wade

Watts

Smith

MIFAB

- C. Cross Reference Identification:
 - 1. If the Contractor selects a manufacturer of drainage equipment products the than as identified on the Schedule but is selected from the available manufacturers listed ab vest cover sheet shall be included with the submission of shop drawings indicating the cross recorded manufacturer and model number.
 - 2. Shop drawings shall not be reviewed or accepted if not in on sliphce with this requirement.

PART 3 – EXECUTION

- 3.1 INSTALLATION OF SOIL AND WASTE PIPING
 - A. The Plumbing Contractor shall install a complete visual of sanitary drainage piping as shown on the drawings. All drainage lines shall be properly run, happed and vented in accordance with the local Plumbing Code and all dry vents, back vents, loop vents, revents or special vents required by the Code shall be furnished and installed by the Plumbing Contractor.
 - B. Drainage lines of the sizes shown on the bawings shall be extended within the building with branches connecting to the base of all soil, waite and vent stack, etc., leaving outlets for connection to all fixtures, floor drains, as required.
 - C. All changes in direction of drahage piping shall be installed with "Y" branches and 1/8 bends. All stacks shall be supported with concealed pipe clamps or hangers as required and the openings in the roof for the vencopies will be provided by this Contractor.
 - D. All drainage piping which will be located above suspended ceilings shall be checked for slope to assure positive drainage, prior to installation of the ceilings. Pressure tests for leaks, as hereinafter specified shan also be performed prior to ceiling installation.
 - E. Install so, and vent piping pitched to drain at minimum slope of 1/4" per foot (2%) for piping 3" and snaller, and 1/8" per foot (1%) for piping 4" and larger.
 - Vertical to horizontal change in direction to be made with long radius fittings.

Support all soil and waste piping per IPC Section 308.5, 308.6 and 308.7.

- 3.2 INSTALLATION OF STORM WATER DRAINAGE PIPING
 - A. Connect piping to roof drains and outlets provided in gutters, install rainwater conductors and extend to underground storm building drains as indicated.

- B. Underground storm building drains shall be extended from the building, terminating beyond the building wall.
- C. Provide exterior clean-out on both sanitary and storm drain mains. Minimum size shall be 4" installed within 5 ft. of the building. (Also see Paragraph 3.5).
- D. Connect to exterior downspouts, install cast iron downspout shoes, and extend piping from the building wall.
- E. All changes in direction of drainage piping shall be installed with "Y" branches and 1/8 bends. All stacks shall be supported with concealed pipe clamps or hangers as required, and the openings in the roof for the vent pipes will be provided by this Contractor.
- F. All drainage piping which will be located above suspended ceilings shall be checked for slope to assure positive drainage, prior to installation of the ceilings. Pressure tests for eaks, as hereinafter specified, shall also be performed prior to ceiling installation.
- G. Install storm water drainage piping pitched to drain at minimum slope of 1 or per foot (1%) for piping 4" and larger.
- H. Vertical to horizontal change in direction to be made with long radius fittings.
- 3.3 INSTALLATION OF ACID WASTE & VENT PIPING
 - A. Connect to all laboratory equipment requiring acid wast piping, furnish and install traps and extend piping to sanitary drainage piping as indicated.
 - B. Acid vents shall be extended through the rook independently of the sanitary system vents.
 - C. All acid resistant pipe shall be installed in spice accordance with the manufacturers recommendations. The use of solvent cement to dissimilar PVC pipe materials shall not be permitted.
 - D. All changes in direction of acid drainage piping shall be installed with "Y" branches and 1/8 bends. All stacks shall be supported with corcealed pipe clamps or hangers as required, and the openings in the roof for the vent pipes will be provided by this Contractor.
 - E. All drainage piping which will be located above suspended ceilings shall be checked for slope to assure positive draining, prior to installation of the ceilings. Pressure tests for leaks, as hereinafter specified, shall are be performed prior to ceiling installation.
 - F. Install acid wasterand vent piping pitched to drain at minimum slope of 1/4" per foot (2%) for piping 3" and smaller and 1/8" per foot (1%) for piping 4" and larger.
- 3.4 INSTANLATION OF SPECIAL EXPANSION COMPENSATION PRODUCTS

Expansion Joints: Install expansion joints on vertical risers as indicated, and/or as required by International Plumbing Code.

PVC piping systems in multi-story (four stories or more) shall require "O" ring expansion joints to compensate for length changes in soil, waste and vent stacks. Expansion joints shall be required at every floor level for soil and waste stack and at alternate floors for vent stacks and rainwater conductors.

- 3.5 INSTALLATION OF CLEANOUTS
 - A. Cleanouts: Install in sanitary piping and storm conductor and building drain piping as indicated,

and/or as required by International Plumbing Code; at each change in direction of piping greater than 45 degrees; at minimum intervals of 100' for all size straight run piping; and at base of each conductor. Install floor and wall cleanout covers for concealed piping, select type to match adjacent building finish.

- B. Exterior cleanouts shall be installed with access covers flush to grade. The cleanout shall be installed within a concrete pad, 18"x18"x6" thick.
- 3.6 INSTALLATION OF FLOOR DRAINS (ALL TYPES)
 - A. Install floor drains in accordance with manufacturer's written instructions and in locations in live
 - B. Install floor drains at low points of surface areas to be drained, or as indicated. Set ops of drains flush with finished floor.
 - C. Install drain flashing collar or flange so that no leakage occurs between drain and adjoining flooring. Maintain integrity of waterproof membranes, where penetrated.
 - D. Position drains so that they are accessible and easy to maintain.
 - E. <u>All</u> floor drains shall be provided with trap primer connections. <u>All</u> floor drains shall have a trap primer discharge line connected to the outlet.
 - F. All exposed drainage piping shall be DWV copper pipe and fittings. All piping shall be rigidly supported off the wall with split ring clamps or uni strue.
- 3.7 INSTALLATION OF ROOF DRAINS
 - A. Install drains in accordance with manufacturer swritten instruction and in location indicated.
 - B. Coordinate with roofing as necessary to interface roof drains with roofing work.
 - C. Install drains at low points of surface reas to be drained, or as indicated.
 - D. Install drain flashing collar or kange so that no leakage occurs between drain and adjoining roofing. Maintain integrity of waterwoof membranes, where penetrated.
 - E. Position drains so that they we accessible and easy to maintain.
 - F. The roof drain specified is a combination "Primary and Secondary" arrangement. Verify the correct outlet connections to their respective systems.
- 3.8 INSTALLATION OF DOWNSPOUT SHOES
 - A. Install cownspout shoes in accordance with manufacturer's written instructions and in locations indicated.

Downspout shoes shall be installed with top of shoe located 18 inches above finished grade.

INSTALLATION OF BACKWATER VALVES

A threaded plug, female adaptor and 6-inch diameter (152.4 mm) pipe shall be supplied by others for use as the access sleeve and cover. The access sleeve shall be cut to length in the field and attached to the socket on the top of the valve body. This forms the housing for the removable integral lifting device. The integral lifting device consists, in part, of a 4-inch diameter (101.6 mm) pipe which is supplied by others. The 4-inch diameter (101.6 mm) pipe is cut to length in the field, then joined to the disc assembly on the bottom and the collar on the top.

B. The manufacturer's published installation instructions shall be strictly adhered to and, if requested by the code official, a copy shall be maintained on the job site during installation. A copy of the maintenance instructions shall be left with the Owner.

3.10 INSTALLATION OF INTERCEPTORS

- A. Install interceptors in accordance with manufacturer's written instruction and in location indicated.
- B. Install flow control fitting where indicated on the drawing and/or diagrams including vent relief piping.
- C. Interceptors shall be vented in accordance with the local plumbing code and as indicated of drawings.

3.11 INSTALLATION OF FOUNDATION DRAINAGE

- A. Each section of foundation drainage piping shall be extended up, to a cleanout a grade. Piping shall pitch toward the connection to the stormwater drainage system.
- B. Piping shall be laid on minimum of 6 inches No. 8 aggregate crushed store, and backfill shall be done as detailed on the drawings. Crushed stone shall be washed. Aggregate shall be of the sizes indicated and in accordance with ASTM standards. Stone fill layer shell be carefully placed to the indicated depth without disturbing the pipe alignment and a layer of entrepted building paper or 2 inches of salt hay placed on top of the stone, the full width of the trench. Remaining earth fill shall be mounded at top to allow for settlement.

3.12 UNDERGROUND METALLIC TAPE

- A. <u>All exterior underground PVC drainage piping (sameary, storm, condensate waste) shall be provided</u> with detectable metallic underground tape.
- B. Tape shall be similar to Lineguard Maintenince Systems as provided by Utility Supply of America 800-548-1234 or approved substitute as canufactured by Seton.
- C. Installation shall comply with nature cturer's recommendations and shall be installed in the backfill after refilling the trench opening completely, and allowed to settle to the desired 4" to 6" depth. The Contractor shall install the tare after final lifts in compaction backfilling or unroll it before final restoration or instruction of sod, black dirt, seeding, etc.
- D. The tape system call be installed under the supervision of the Owner's Representative. When the tape system is complete, the Contractor shall provide a test using the tape manufacturer's recommended direction device, to prove the integrity of the installation with the Owner's Representative.

3.13 INVERTS AND ELEVATIONS

Indicated inverts and elevations of existing utilities are approximate and based on the best information available. Upon award of Contract, Contractor shall verify in the field all such information and report any discrepancies to the Engineer before proceeding with work.

PIPING INSTALLED IN FILLED GROUND

- A. Piping located below floor slab in filled areas shall be supported either from the floor slab, or with masonry piers to undisturbed earth. Drainage piping shall be supported at each joint. Exterior piping located in filled areas shall be supported with piers.
- B. Details of supports and method of installation shall meet with the approval of the Engineer.

Tetra Tech 200-15704-17001

3.15 INSPECTION

- A. The Plumbing Contractor shall, upon completion of the drainage systems, secure from the Inspector and/or the Municipality under which the installation was made and inspected, certificates or letters of approval indicating the system has been installed satisfactorily. The Plumbing Contractor shall certify that all inspection fees, permits and charges have been duly paid.
- 3.16 BUILDING DRAINAGE SYSTEM CLEANING AND CAMERA VIDEO TAPE/TRANSMI
 - A. The Plumbing Contractor shall hire the services of Tri-State Grouting (302) 286-0701, Conta Andrus.
 - B. The Camera-Video tape/transmitter work shall be executed for the benefit of the Plambing Contractor to locate all existing below-floor slab sanitary and storm water drainage systems. This work shall be done to verify exact locations where new piping systems shall connect to existing piping systems.
 - C. When all plumbing work is complete and before final test of the drainage systems, the entire building active below-slab storm and sanitary waste systems, including all short ran and small pipe laterals, shall be power flushed and cleaned to clear all or any obstructions in the line. The work shall include all stack systems through each (VTR) vent thru roof and roof drant to the exterior laterals that exit the building.

END OF SECTION 220110

SECTION 220120

DOMESTIC WATER SYSTEMS – PLUMBING

PART 1 – GENERAL

1.1 RELATED DOCUMENTS

- A. The general provisions of the Contract, including the Conditions of the Contract (General Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the vort specified in this Section.
- 1.2 DESCRIPTION OF WORK
 - A. This Section includes:
 - 1. Domestic water piping systems work is indicated on drawings and schedules and by requirements of this section.
 - B. Applications for water piping systems include the following:
 - 1. Exterior water service piping from main at curb to building domestic water piping.
 - 2. Exterior water service piping from connection 5'-0" be ond building foundation to building domestic water piping.
 - 3. Exterior fire/sprinkler service from connection 5⁻-0⁻ beyond building foundation to flanged connection 12" above finished floor slab in Michanical/Boiler Room.
 - 4. Exterior fire/sprinkler service from street main connection to flanged connection 12" above finished floor slab in Mechanical Boiler Room.
 - 5. Exterior domestic water meter and pitanstallation.
 - 6. Domestic cold-water pipt g.
 - 7. Domestic hot-water pir my
 - 8. Domestic recirculating-water piping.
 - D. Complete flow balancing of the entire domestic hot water return system.
 - E. Insulation for dor lestic water piping as specified in Section 220030 is included as work of this section.
 - F. Trenching and backfill required in conjunction with exterior water piping as specified in Section 220000 c included as work of this section. Refer to Division 1.

REFERENCE STANDARDS

Refer to Section 220000 for a general description of requirements applying to this section.

- QUALITY ASSURANCE
- A. Refer to Section 220010 for a general description of requirements applying to this section.
- 1.5 SUBMITTALS
 - A. Submit shop drawings and product data in accordance with Section 220000.

- B. Submit the following:
 - 1. Product data on all specialties and systems equipment.
- 1.6 WARRANTY/GUARANTEE
 - A. All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, GENERAL REQUIREMENTS.

PART 2 – PRODUCTS

2.1 DOMESTIC WATER PIPING MATERIALS AND PRODUCTS

- A. Provide piping materials and factory fabricated piping products of sizes, types presume ratings, temperature ratings, and capacities as indicated. Where not indicated, provide proper election as determined by Installer to comply with installation requirements. Provide eizes and types matching piping and equipment connections; provide fittings of materials which materials used in domestic water piping systems. Where more than 1 type of materials or products are indicated, selection is Installer's option.
- 2.2 BASIC PIPE, TUBE AND FITTINGS
 - A. Provide pipe, tube, and fittings complying with Division 32 basic Materials and Methods section "Pipe, Tube, and Fittings", in accordance with the following lating:

rawn temper.

pper, solder-joints.

B. Interior Domestic Water Piping:

Tube Size 4" and Smaller:

Wall Thickness:

Wall Thickness:

Fittings:

Fittings:

Pipe Size

Pipe Vieig

ittings

Fittings:

E-tonian Wester Commission Distance

e Size 3" and Smaller:

- C. Exterior Water Service Piping
 - Pipe Size 3" and Smaller

Copper tube.

ght-d

Copper tub

Type "

Wro

Type "K" Soft Temper

Wrought copper solder joint.

Ductile-iron pipe with cement-mortar lining, and gasketed joints.

Schedule 150.

Ductile-iron, with cement-mortar lining, mechanical joint.

Polyvinyl Chloride (PVC)

ASTM 1785

Schedule 80.

PVC with Schedule 80, socket-type solvent joints or elastomeric gasketed joints.

Pipe Size 4" to 12":

Wall Thickness:

AWWA C900

Polyvinyl Chloride (PVC)

Pipe Class:

Class 150

Fittings:

Molded pressure Class 150 with AWWA C907, gaskets conforming to ASTM F-477.

2.3 BASIC PIPING SPECIALTIES

A. Provide piping specialties complying with Section 220010 Basic Materials and Methods in accordance with the following listing:

Pipe escutcheons

Dielectric unions

Drip pans

Pipe sleeves

Sleeve seals

- 2.4 SPECIAL PIPING SPECIALTIES
 - A. Water Hammer Arresters: Provide bellows or piston type stater hammer arresters, pressure rated for 250 psi, tested and certified in accordance with PDI Standari WH-201.
- 2.5 BASIC VALVES
 - A. Provide valves complying with applicable Division 22 sections "Valves", in accordance with the following listing:
 - B. Sectional Valves:

2-1/2" and Smaller:

3" and Larger:

Ball Valves. Butterfly Valves.

/alves.

C. <u>Shutoff Valves:</u> 2-1/2" and Small

Ball Valves.

Gate Valves

Ball Valves.

Butterfly Valves.

Hose End Threaded Gate or Ball Valves.

E. <u>Balancing Valves:</u>

All

Drain Valves:

2" and Smaller:

and Larger:

Ball Valves (Circuit Setter Type). (w/ Memory Stop)

Tetra Tech 200-15704-17001
F. Check Valves:

All Sizes:

Swing Check Valves. Horizontal Installations

Spring Check Valves. Vertical Installations

2.6 WATER METER

- A. Provide water meter and related piping conforming to applicable local Utility Company regulations and AWWA Standards.
- B. Water Meter: Provided by Local Utility Company. Provide roughing- in and bypass for accordance with Utility Company requirements.

2.7 SPECIAL VALVES

- A. Special valves required for domestic water piping systems include the following types:
- B. Hose Bibbs: Threaded end, renewable composition disc, tee handle, 34" PT inlet, 3/4" hose outlet with vacuum breaker.
 - 1. Finished Areas: Chrome plated.
 - 2. Unfinished Areas: Bronze finish.
- C. Wall Hydrants: Non-freeze, cast-bronze body, ter handle key, bronze casing, length to suit wall thickness, vacuum breaker, hinged locking cover, 14" inlet, hose outlet.
- D. Lawn Boxes: Non-freeze, bronze ground hyperart with locking cover, rough bronze box, vacuum breaker, 1 inch connection, and suitable for 1-or depth of bury.

2.8 BASIC THERMOMETERS AND CAUGE

A. Provide thermometers and gauges complying with Division 22 Basic Materials and Methods Section "Meters and Gauges", in accordance with the following listing:

Pressure gauges

Glass thermomete

Pressure and temperature connections

2.9 BASIC PUMPS

A.

A. Provide sumps as specified in applicable Section 220150 Equipment - Plumbing. Use inline pumps for hot water recirculating.

2.10 BICKFLOW PREVENTERS

Provide, of the type indicated on the drawing schedule, reduced pressure principal type, blackflow preventers shall consist of an assembly including shutoff valves on inlet and outlet, and strainer on inlet. Backflow preventers shall include test cocks, and pressure-differential relief valve located between two positive seating check valves. Construct in accordance with ASSE Standard.

B. On dead-end services (HVAC make-up) provide a spring-loaded check valve ahead of the backflow preventer assembly.

2.11 SYSTEMS EQUIPMENT MANUFACTURERS

- A. Refer to Plumbing Fixture and Equipment Schedule for type, number, size and manufacturer of all equipment and accessories.
- B. Available Manufacturers: Subject to compliance with requirements, manufacturers offering equipment which may be incorporated in the work are limited to the following:

Shock Absorbers: Zurn Josam Wade BUDNINGRUK Watts Smith PPP Inc. MIFAB Automatic Trap Primers PPP Inc. Sloan Sioux Chief MIFAB Hose Bibbs Nibco Tanner Central Brass Wolverine Iyd Wall libco nner Central Brass Wolverine Wall Hydrants Zurn Josam

| | Wade |
|--------|--------------------------|
| | Watts |
| | Smith |
| | Woodford |
| | MIFAB |
| | Lawn Boxes |
| | Zurn |
| | Josam |
| | Wade |
| | Watts |
| | Smith |
| | Woodford |
| | MIFAB |
| | Backflow Preventers |
| | Conbraco |
| | Febco |
| | Cla-Val |
| | Wilkins |
| | ITT Grinnell |
| | Neptune V |
| | Watts |
| | Relief Valves |
| | Rockwell |
| | Fisher |
| | DeZurik |
| \cap | Pressure Reducing Valves |
| | Conbraco |
| ~ | Jamesbury |
| • | DeZurik |
| | Fisher |
| | ITT Bell & Gossett |
| | |

PART 3 – EXECUTION

3.1 INSTALLATION OF BASIC IDENTIFICATION

- A. Install mechanical identification in accordance with Section 220010 Basic Materials and Methods.
- B. Support vertical piping at floor levels using approved riser clamps. Clamp material shall be compatible with pipe material. Maximum vertical spacing shall be 10'-0". Domestic water piping shall be supported in accordance with the International Mechanical Code, Section 305 and Table 305.4 Spacing Intervals, or in accordance with MSS-SP-69. International Plumbing Code statest edition, Section 308.5, accept as follows:
 - 1. Copper tubing $\frac{1}{2}$ " to 1-1/4" nominal size, not to exceed 6 ft. horizontal intervals.
 - 2. Copper tubing 1-1/2" and larger nominal size, not to exceed 10 ft. horiz and later a
 - 3. Copper tubing $\frac{1}{2}$ " to 1-1/4" nominal size, not to exceed 10 ft. vertical intervals.
 - 4. Copper tubing 1-1/2" and larger nominal size not to exceed 10 ft. Attical intervals.
 - 5. CPVC pipe or tubing $\frac{1}{4}$ " to 1" nominal size, not to exceed x ft. horizontal spacing.
 - 6. CPVC pipe or tubing 1-1/4" and larger nominal size not to exceed 4 ft. horizontal spacing.
 - 7. CPVC pipe or tubing $\frac{1}{4}$ " to 1" nominal size not co exceed 10 ft. vertical.
 - CPVC pipe or tubing 1-1/4" and larger nominal size for to exceed 10 ft. vertical."
 *Mid-Story Guide.
- 3.2 INSTALLATION OF DOMESTIC WATER VERVICE SYSTEM BUILDING
 - A. Install water distribution system is rescondance with Section 220010 Basic Materials and Methods, and the International Mechanical C des Section 305, and Support Intervals under Tables 305.4 and 308.5 or in accordance with MSS-SP-62
- 3.3 INSTALLATION OF LOVELTIC WATER SERVICE SYSTEM SITE
 - A. Install site water conjuct system in compliance with local governing regulations.
 - B. Street Main Connection: Arrange and pay for tap in water main, of size and in location indicated, by Local Water Utility Company.
 - C. Water Service Liping: Extend water service piping from service stop of size and in location indicated to the veter pit and from the meter pit to the water service entrance at building. Provide sleeve in foundation wall for water service entry; make entry watertight. Provide gate valve at water service entry inside building; strainer, pressure gauge, test tee with valve.
 - Ductile-Iron Pipe: Install in accordance with ANSI/AWWA C-60.

Sterilization: At completion of water service line installation, flush and sterilize in conformance with AWWA C-601, to satisfaction of local Authorities having jurisdiction.

- 3.4 INSTALLATION OF PIPING SPECIALTIES
 - A. Install piping specialties in accordance with Section 220010 Basic Materials and Methods.

B. Water Hammer Arresters: Install in upright position, in locations and of sizes in accordance with PDI Standard WH-201, and elsewhere as indicated.

3.5 INSTALLATION OF LAWN BOXES

- A. Lawn boxes shall be provided with 18" x 18" x 6" concrete pad at grade, unless located in paving. Installation shall include 1 cu. ft. of crushed stone at the base of the hydrant, to accept drainage from weep holes.
- 3.6 REACTION BACKING
 - A. All plugs, tees and elbows in the underground piping shall be provided with reaction backing consisting of concrete placed between solid undisturbed earth and the fitting to be archived. Concrete shall be of such bearing area as to assure adequate resistance to the thrust to be encountered. In general, backing shall be so placed that the joint will be accessible for inspection addrepair.

3.7 INSTALLATION OF VALVES

- A. Install valves in accordance with Division 22 Basic Materials and Nethods section, "Valves".
- B. Sectional Valves: Install on each branch and riser, close to man, where branch or riser serves 2 or more fixtures, equipment connections, and elsewhere as indicated.
- C. Shutoff Valves: Install on inlet of each plumbing equipmentitem, and on inlet of each plumbing fixture, and elsewhere as indicated.
- D. Drain Valves: Install on each plumbing equipment hand located to completely drain equipment for service or repair. Install at base of each rise, a close of each rise or drop in piping system, and elsewhere where indicated or required to completely drain domestic water piping system.
- E. Check Valves: Install on discharge side of each pump, and elsewhere as indicated.
- F. Balance Cocks: Install in main recirculating loop and in each branch hot water recirculating loop. Install a ball valve and check valve areach balance valve installation.
- G. Hose Bibbs: Install on xp set piping where indicated, with vacuum breaker.
- 3.8 INSTALLATION OF BACKFLOW PREVENTERS
- A. Install backflow preventers where indicated, and where required by International Plumbing Code. Locate in tame from or area as equipment being protected.
- B. RPZ ype bashflow preventers to be piped from the relief outlet to nearest floor drain.
- C. check alve is required on the upstream side of all RPZ installations.

INSTALLATION OF PRESSURE REGULATING VALVES

Install pressure regulating valves where indicated. Provide inlet and outlet shutoff valves, and ball valve bypass. Provide pressure gauge on valve outlet.

- 3.10 INSTALLATION OF EXPANSION COMPENSATION PRODUCTS
 - A. This project shall require the installation of expansion compensators.
 - B. Furnish and install expansion compensation products in accordance with Section 220210 Basic Materials and Methods HVAC

3.9

3.11 INSTALLATION OF THERMOMETERS AND GAUGES

A. Install thermometers and gauges in accordance with Section 220010 Basic Materials and Methods.

3.12 INSTALLATION OF WATER METER

- A. Install water meter in accordance with Section 220010 Basic Materials and Methods.
- B. Meter shall be supported in accordance with the requirements of the Water Department.

3.13 EQUIPMENT CONNECTIONS

- A. Piping Runouts to Fixtures: Provide hot and cold water piping runouts to fixtures of sizes indicated, but in no case smaller than required by International Plumbing Code.
- B. Equipment furnished by the Owner or Contractors other than this Contractor. A fits, equipment has been set in place, this Contractor shall furnish all labor and material required to make final connections, between roughing-in and the equipment. Install valves, fixings, trip and appurtenances furnished with the equipment. All exposed piping in the kitchen areas shall be chrome plated. Piping in other areas shall be of the same material as the system to which it connects.

3.14 AUTOMATIC TRAP PRIMERS

- A. Install units in accordance with manufacturer's written instructions.
- B. Cap-off all unused tube connections not to be used for any harge procedures.
- C. Units installed in Mechanical Room shall be surfice mounted. All others shall be arranged for recessed installation and shall include a 14"x 1" access door #D-1416SS.
- D. All discharge tubing from the tube connectors to the floor drain connection shall be minimum Type "M" copper for above floor slab instanction. Where discharge piping runs below the floor slab, the piping may be PVC grade water piping. Ensure that all connections are properly made and leak-free.
- E. Verify all electrical connection phot to the start-up of the units.
- F. Shield copper from direct context with concrete, stone and sharp edges below slab.
- 3.15 SPARE PARTS
 - A. Furnish to wher, with receipt, one valve key for each key operated hydrant, bibb, or faucet installed.
- 3.16 INSTACLATION OF EXTERIOR WATER PIPING (5'-0" EXTENSIONS)
 - A. Install xterior water service piping system in compliance with local governing regulations.
 - B. Main Connections: Coordinate all work with the Site Contractor. The Plumbing Contractor shall mare final connections of the domestic and fire protection water services.

Water Service Piping: From the final connection points, extend water service piping of size and in locations indicated to the water service entrance in building. Provide ball value at the domestic water service and a flanged connection with a ³/₄" blow-off value for the fire protection water service (12" A.F.F.).

- D. Ductile-Iron Pipe: Install in accordance with ANSI/AWWA C-60.
- E. Test:

ADDITION AND RENOVATIONS

- 1. Domestic Water Service: Minimum 125 psi.
- 2. Fire Protection Water Service: Minimum 200 psi.
- 3. Combination Domestic & Fire Protection Water Service: Minimum 200 psi.
- F. Sterilization: At completion of water service line installation, flush and sterilize in conformance with AWWA C-601, to satisfaction of local Authorities Having Jurisdiction.
- 3.17 KITCHEN DOMESTIC WATER
 - A. All kitchen domestic water system piping shall be roughed-in and strictly coordinated with the kitchen uppender drawings.
 - B. Provide all rough-in piping and final connections to equipment furnished by the Litcher Equipment Contractor (KEC). This also includes any equipment items furnished by the KEC and are to be completely installed by the Plumbing Contractor.
 - C. Verify all responsibilities during the bid phase of the work.
 - D. All piping shall be supported off the wall with split ring clamps or unistrut.
 - E. All piping shall be insulated and identified.
 - F. Provide shut-off valves and stainless-steel flex here connections to all individual equipment connections.
 - G. All exposed piping shall be chrome plated brass.
- 3.18 DOMESTIC HOT WATER RETURN
 - A. This Contractor shall install complete and operating hot water return system. The system shall be balanced and include a report as expired in HVAC Specification Section 230950.
 - B. Balancing Valves are required in the system as hereinbefore specified. The system shall also include the installation of "air bleed" of "burp" valves to remove any trapped air in the system.
 - C. Where emergency showers are installed with thermostatic mixing valve, they shall require the installation of a bot vater return line as detailed on the drawings.

END OF SECTION 220120

SECTION 220140

FIXTURES – PLUMBING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. The general provisions of the Contract, including the Conditions of the Contract (General Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the V or specified in this Section.

OWN

- 1.2 DESCRIPTION OF WORK
 - A. This Section includes:
 - 1. Plumbing fixtures and trim work as indicated by drawings and schedules, and by requirements of this section.
 - 2. Types of plumbing fixtures required for the project include the Polloving:

Water Closets

Urinals

Lavatories

Countertop Sinks

Service Sinks

Mop Receptors

Showers

Electric Water Coolers

Drinking Fountain/Cu.pid

Sensor-Operated Flush Valves and Faucets

Manually Operated Flush Valves

Handicap La atory Insulation

a atory Shield Enclosure

Refer to Section 220120 for domestic water piping systems used in conjunction with plumbing fixtures; not work of this section.

Refer to Section 220110 for soil and waste piping systems used in conjunction with plumbing fixtures; not work of this section.

5. Refer to Division 26 sections for electrical connections to water coolers and other plumbing fixtures; not work of this section.

1.3 REFERENCE STANDARDS

A. Refer to Section 220000 for a general description of requirements applying to this section.

ADDITION AND RENOVATIONS

1.4 QUALITY ASSURANCE

- A. Refer to Section 220010 for a general description of requirements applying to this section.
- B. Manufacturers: Firms regularly engaged in manufacture of plumbing fixtures of the type, style and configuration required, whose products have been in satisfactory use in similar service for not less than 3 years.
- C. Plumbing Fixture Standards: Comply with applicable portions of International Plumbing Code pertaining to materials and installation of plumbing fixtures.
- D. ANSI Standards: Comply with applicable ANSI standards pertaining to plumbing Catures and systems.
- E. ANSI & ADA Standards: Comply with ANSI A171.1 Standard and the ADA standard pertaining to plumbing fixtures and provisions for handicapped.
 - 1. Water closets shall measure 17" to 19" from the floor to the top of the seat. Bowls shall be elongated type.
 - 2. Flush valve mechanisms shall be on the wide side of the stell, to higher than 44" above the floor.
 - 3. Urinals shall be elongated (14" rim from the wall) mounted no higher than 17" from the floor.
 - 4. Lavatories shall be mounted no higher than 34" from the floor and provide knee clearance using an offset drain assembly with "P" trap set parallel to the fixture supporting wall. Trap and wall supplies shall be installed for clearance required for the installation of lavatory shield enclosures.
 - 5. Faucets shall be lever operated, push type, touch type, electronically operated. See Fixture Schedule. All faucets shall operate in ass than 5 pounds force and shall not require tight grasping, pinching or twisting of the wrist.
- F. PDI Compliance: Comply with standards established by Plumbing and Drainage Institute pertaining to plumbing fixture supports.
- G. Federal Standards: Convol. wh applicable FS WW-P-541/- Series sections pertaining to plumbing fixtures.
- H. UL Labels: Provide water coolers which have been listed and labeled by Underwriters' Laboratories.
- I. ARI Labers: Provide water coolers which are rated and certified in accordance with applicable Air-Cenditoning and Refrigeration Institute Standards.
- 1.5 SUBMITALS

Α

Submit shop drawings and product data in accordance with Section 220000.

Submit the following:

- Product Data: Submit manufacturer's specifications for plumbing fixtures and trim, including catalog cut of each fixture type and trim item furnished, roughing-in dimensioned drawings, templates for cutting substrates, fixture carriers, and installation instructions.
- 2. Color Selection Data: Submit charts or samples for color selection where applicable.
- 3. Maintenance Data: Submit maintenance data and parts lists for each fixture type and trim item, including instructions for care of finishes. Include this data in maintenance manual.

1.6 WARRANTY/GUARANTEE

A. All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, GENERAL REQUIREMENTS.

1.7 PRODUCT DELIVERY, STORAGE AND HANDLING

- A. Deliver plumbing fixtures individually wrapped in factory-fabricated containers.
- B. Handle plumbing fixtures carefully to prevent breakage, chipping and scoring the fixture finish. Do not install damaged plumbing fixtures; replace and return damaged units to equipment manufacture

PART 2 – PRODUCTS

- 2.1 PLUMBING FIXTURES
 - A. Provide factory-fabricated fixtures of type, style and material indicated. For each type fixture, provide fixture manufacturer's standard trim, carrier, seats, and valves as indicated by their published product information; either as designed and constructed, or as recommended by the manufacturer and as required for a complete installation. Where more than one type is indicated, selection is Installer's option; but, all fixtures of same type must be furnished by single-manufacturer. Where type is not otherwise indicated, provide fixtures complying with governing regulations.

2.2 MATERIALS

- A. Unless otherwise specified, comply with applicable Federa Specification WW-P-541/-Series sections pertaining to plumbing fixtures, fittings, trim, meals and finishes. Comply with the requirements of WW-P-541/-specification relative to quality of ware, glazing, enamel, composition and finish of metals, air gaps, and vacuum breakers, even bough some plumbing fixtures specified in this section are not described in WW-P-541/-.
- B. Provide materials which have been velocied for their surface flatness and smoothness. Exposed surfaces which exhibit pitting, seem marks, roller marks, foundry sand holes, stains, decoloration, or other surface imperfection continuited units are not acceptable.
- C. Where fittings, trim and accessories are exposed or semi-exposed, provide bright chrome-plated or polished stainless sevel units. Provide copper or brass where not exposed.
- D. Stainless Steel Shrets: ANSI/ASTM A-167, Type 302/304, hardest workable temper. Finish: No. 4, bright, directional polish on exposed surfaces.
- E. Steel sheets in Baked Enamel Finish: ANSI/ASTM A-591, coating Class C, galvanized-bonderized.
- F. Steel Shets for Porcelain Enamel Finish: ANSI/ASTM A-424, commercial quality, Type 1.

Vereous China: High quality, free from fire cracks, spots, blisters, pinholes, and specks; glaze exposed surfaces, and test for crazing resistance in accordance with ANSI/ASTM C-554.

Fiberglass: ANSI Z124 smooth surfaced, with color selected by Architect/Engineer.

- Aluminum: ANSI/ASTM B-209/B-221 sheet, plate and extrusions, as indicated; alloy, temper and finish as determined by manufacturer, except 0.40 mil natural anodized finish on exposed work unless another finish is indicated.
- J. Synthetic Stone: High quality free from defects, glaze on exposed surfaces, stain resistant.

ADDITION AND RENOVATIONS

2.3 PLUMBING FITTINGS, TRIM AND ACCESSORIES

- A. Lavatory Protective Shield Covers:
 - 1. Fully molded enclosure "Lav Shields" as manufactured by Zurn or Truebro, Inc., complete with tamper-resistant stainless steel fasteners.
 - 2. Shield enclosure to meet A.D.A. #4.19.4, ANSI A117.1 and BOCA P- 1203.4.
- B. Water Outlets: At locations where water is supplied (by manual, automatic or remote control), provide commercial quality faucets, valves, or dispensing devices, of type and size indicated, and as required to operate as indicated. Include manual shutoff valves and connecting system pipes to permittbutlet servicing without shut- down of water supply piping systems.
 - 1. Vacuum Breakers: Provide with flush valves where required by governing regulations, including locations where water outlets are equipped for hose attachment.
- C. P-traps: Include removable P-traps where drains are indicated for direct connection to drainage system. All traps shall be minimum 17 gauge.
- D. Carriers: Provide cast-iron and/or steel supports for fixtures. Carriers chall be provided for all wall-hung fixtures, and/or the carrier shall be selected to support the fixture independently of the wall. Carriers shall be adjustable type, complete with all fittings are foot supports. Carrier shall be single or double, back-to-back, horizontal offset and vertical mack type. Carrier shall be selected and used as best suited within the pipe chases. Where noted or indicated, stud mount type carriers shall be used and installed within stud wall s 8" and less.
- E. Fixture Bolt Caps: Provide manufacturer's standard exposed fixture bolt caps finished to match fixture finish.
- F. Escutcheons: Where fixture supplies and drains penetrate walls in exposed locations, provide chrome plated sheet steel escutcheons with fraction clips.
- G. Aerators: Provide aerators of types approved by Health Departments having jurisdiction.
- H. Comply with additional teture requirements contained in fixture schedule attached to this section.
- 2.4 FIXTURE LIST
 - A. Refer to the Proming Fixture & Equipment Schedule" as indicated on the drawings.
- 2.5 SHOWER HANDICAP
 - A. The shower modules shall be manufactured by Crane Plumbing/Fiat Products #MSCI-307, RH/LH 36" x 35") and shall be thermoformed from continuous cast acrylic sheet into one piece, seamless units. The units shall be reinforced in the back with fiberglass reinforced polyester resin and have a backside fire rating of "A" and flame spread of 25.

Shower modules shall be pre-drilled and equipped with a model 180AA pressure balanced, single dial lever mixing valve with stops, hand held shower head with swivel fitting, 69" flexible stainless steel hose and in-line vacuum breaker, 10 oz. GSA approved shower curtain with curtain hoods, 2" cast brass chrome plated shower drain, and 1" diameter stainless steel curtain rod.

C. All units shall have anti-skid floor treatment to meet performance requirements of ASTM F-462-78. All units are listed under SBCCI Report No. 8556 and BOCA Report No. 86-40.

- D. Additional back reinforcements shall be suitably located to provide code complying structural integrity for factory attaching of grab bars, curtain rod, and slide guide for hand held flexible hose shower head.
- E. Standard Equipment:
 - 1. Factory Installed:
 - a. Grab Bars: see table for individual models.
 - b. Curtain rod: 1" diameter, stainless steel.
 - c. Shower curtain, 10 oz. with curtain hooks, GSA approved.
 - d. 2" shower drain, chrome plated, cast brass.
 - e. Retractable, compact shower seat with stainless steel frame and teal words
 - 2. Optional Equipment:
 - a. Model 180AA pressure balanced, single dial lever mixing all with integral check stops.
 - b. Hand held shower H-11with swivel fitting, 69" flexible stailless steel hose, and in-line vacuum breaker.
 - c. Slide guide H-12, 24"
 - d. Dome light, ATP-1, 60 watt, recessed.
- F. Color: Color option is required. Final selection by Architect.
- 2.6 SENSOR-OPERATED FLUSH VALVES & FAUCETS
 - A. This Contractor shall furnish and instal complete and operating sensor operational faucets and flush valves where so indicated and roted. The shall include, but not be limited to, transformers and low voltage wiring.
 - B. Electrical power wiring and be provided at the pipe chases and lavatory wash stations as required by the Electrical Contractor. See Electrical Plans. All wiring beyond that point shall be considered control wiring and shall be the complete responsibility of the Plumbing Contractor.
 - C. The Contractor shall have a complete understanding of the sensor operated equipment and system they are installing during the bid phase of the work. This includes all control wiring for the operation and function of the flash valves and faucets.
 - D. The Centractor shall install the system in strict conformance with the manufacturer's written instructions. The installation shall be executed with good workmanship and to be clear of any interference with the user including the installation of lavatory protective shield enclosures.

All faucet installations shall require a mixing valve for single water supply to faucet.

All sensor wall-mounted boxes and/or panels, including setting heights, shall be <u>strictly</u> coordinated with the masonry contractor for both drywall and block wall installations.

G. Flush valve electrical box positioning and support kits shall be provided wherever drywall partitions are installed. The kit shall be specifically designed to accurately position and ease the installation of the electrical box and the plumbing rough-in piping and valve. The kit shall be supplied by the flush valve manufacturer and installed by the Plumbing Contractor.

- H. The manufacturer's representative shall have at least two (2) site visits to verify that equipment and wiring are properly installed.
- I. The Contractor shall coordinate the location of the electrical power junction box with Electrical Contractor including the installation of the transformer unit. The Plumbing Contractor shall provide an adequate sized access panel for maintenance and servicing of the transformer and junction box. The Contractor shall locate the junction box and transformer where best suited either above the ceiling or pipe chase wall.
- 2.7 AVAILABLE MANUFACTURERS
 - A. Subject to compliance with requirements, manufacturers offering fixtures, trim and carriers which may be incorporated in the work include, and are limited to the following:

Water Closets (Wall-Mounted Back Outlet - China)

All water closets on this project shall be maximum 1.6 gallons per flush and shall be of the pressure tank (pneumatic assisted) type with the Water Control International "WCI" System ANSI A112-19-2M and ASSE Standard 1037. Manufacturers shall be limited to the following:

Crane JUING American Standard Kohler China/Enameled Fixtures Kohler American Standard Eljer Crane ТоТо Faucets/Trim (N ensor Operated) Kohler American and Delta oen Elkav Speakman Chicago Faucets/Trim (Sensor-Operated) Sloan Speakman

ТоТо Chicago Flush Valves Sloan "Royal" (optima Series (Sensor-Operated) Coyne & Delany Zurn ТоТо Wall Supplies/Traps McGuire ppin opi **Brass-Craft** Kohler American Standard Sanitary-Dash Teledyne Wolverine Pro-Flo Keeny **Fixture Carriers** Zurn Josam Wade Watts Smith **MIFA** xture Seats Olsonite Sperzel Benke Bemis Church Kohler

American Standard Centoco **Comfort Seat** Mop Receptors Note: All mop receptor installations must include the Fiat combination eyewash and service sink faucet arrangement Stern-Williams Mustee Florestone Water Coolers opinsoni Elkay Haws Halsey-Taylor Oasis Acorn Stainless Steel Sinks Elkay Just Dayton Advanced-Tabco Shower Assemblies & Acorn Symmons Leona owe adley Emergency/Safety Equipment Bradley Haws Western Speakman Guardian

Shower Units

Fiat

Fibersheen

Kohler

American Standard

Acryline

Aqua-Glass

Aquarius

Aqua-Bath

Washer Utility Boxes

Symmons

Guy-Gray

Oatey

Instantaneous Electric Water Heaters

Chronomite

Eemax

- B. Cross Reference Identification:
 - 1. If the Contractor selects a nanufacturer of drainage equipment products other than as identified on the Schedule but is selected hom ne available manufacturers listed above, a cover sheet shall be included with the submission of shop drawings indicating the cross referenced manufacturer and model number.

GRUS

- 2. Shop drawings shall not be reviewed or accepted if not in compliance with this requirement.
- 2.8 HANDICAP DAWTORY INSULATION
 - A. Fully molded "P" trap and angle valve insulation kit Handi-Lav Guard Truebro Model #101, 102 and 105 to suit.
 - B. insulation to meet A.D.A. #4.19.4, ANSI A117.1 and BOCA P- 1203.4.

Severtinguishing ASTM D635 burn characteristics, Thermal conductivity ASTM C177-K value 1.17.

RT – EXECUTION

FIXTURE CONNECTIONS

- A. Connections to plumbing fixtures shall be of the sizes indicated on the "Plumbing Fixture & Equipment Schedule".
- B. The sizes indicated on the Schedule are for drainage and water piping serving an individual fixture; the sizes of the mains and branches shall be as indicated on the drawings.

Tetra Tech 200-15704-17001

FIXTURES – PLUMBING 220140-9

3.2 FIXTURE SETTING HEIGHTS

A. The plumbing fixtures shall be set in accordance with the heights established by the latest edition of codes and ADA requirements.

Note: Height indicated is established as follows:

| Water Closets: | From finished floor to top of seat. | |
|-------------------|---|---------------|
| Urinals: | From finished floor to rim of fixture. | |
| Lavatories & EWC: | From finished floor to rim of fixture. | \mathcal{C} |
| Receptor Fitting: | From finished floor to center of fitting. | \frown |
| Shower: | From finished floor to center of shower head. | \sim |

- B. Refer to Architectural drawings and sections for fixture elevations. Fix and s invarious areas may be set at lower elevations. Confirm <u>all</u> rough-in elevations prior to any Instrumion.
- 3.3 LAVATORY PROTECTIVE SHIELD ENCLOSURES
 - A. Installation shall conform to manufacturer's written instruction
 - B. All items involved with wall-hung lavatory installations shall be roughed-in and installed within the enclosure. This includes the offset "P" trap assembly, the mostatic mixing valve, sensor faucet trim and accessories, electrical outlet. Coordinate all work equired for complete concealment of all devices.
 - C. Protective shield enclosures are required on the milet room's countertop lavatories and are furnished by the Architect. Coordinate all trim and accessories to fit within this enclosure.

3.4 INSPECTION AND PREPARATION

- A. Examine roughing-in work or donestic water and waste piping systems to verify actual locations of piping connections prior to installing fixtures. Also examine floors and substrates, and conditions under which fixture work is to be accomplished. Correct any incorrect locations of piping, and other unsatisfactory conditions the installation of plumbing fixtures. Do not proceed with work until satisfactory conditions have been corrected.
- B. Install plumbing fixtures of types indicated where shown and at indicated heights; in accordance with fixture manufacturer's written instructions, roughing-in drawings, and with recognized industry practices. Ensure that plumbing fixtures comply with requirements and service intended purposes. Comply with applicable requirements of the International Plumbing Code pertaining to installation of plumbing fixtures.

Fasten plumbing fixtures securely to indicated supports or building structure; and ensure that fixtures are level and plumb. Secure plumbing supplies behind or within wall construction so as to be rigid, and not subject to pull or push movement.

CLEAN AND PROTECT

- A. Fixture shall be thoroughly cleaned after completion of installation.
- B. Protect installed fixtures from damage during the remainder of the construction period.

3.6 FIELD QUALITY CONTROL

- A. Upon completion of installation of plumbing fixtures and after units are water pressurized, test fixtures to demonstrate capability and compliance with requirements. When possible, correct malfunctioning units at site, then retest to demonstrate compliance; otherwise, remove and replace with new units and proceed with retesting.
- B. Inspect each installed unit for damage to finish. If feasible, restore and match finish to original a site, otherwise, remove fixture and replace with new unit. Feasibility and match to be jurged by Architect/Engineer. Remove cracked or dented units and replace with new units.

END OF SECTION 220140

Republication

<



SECTION 220150

EQUIPMENT – PLUMBING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. The general provisions of the Contract, including the Conditions of the Contract (General Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the Vor specified in this Section.

1.2 DESCRIPTION OF WORK

- A. This section includes:
 - 1. Plumbing equipment as indicated on drawings and provisions of this section, including schedules and equipment lists associated with either drawings or this section.
 - 2. Types of plumbing equipment required for project include the following:

Recirculating Pumps-Domestic Water Return (110 degrees & 140 degrees)

Thermostatic Mixing Valve

1.3 REFERENCE STANDARDS

- A. Refer to Section 220000 for a general description of requirements applying to this section.
- B. UL and NEMA Compliance: Provide electric potors and electrical components required as part of plumbing equipment, which have been used and labeled by Underwriters' Laboratories and comply with NEMA Standards.
- C. NEC Compliance: Comply with National Electrical Code (ANSI/NFPA 70) as applicable to installation and electrical connections of ancillary electrical components of plumbing equipment.
- D. ASME Relief Valve Stamps: Provide water heaters with safety relief valves bearing ASME valve markings.
- E. AWWA Compliance. Comply with applicable American Water Works Association Standards pertaining to such a ter tanks.
- F. CSA and NSF Looels: Provide water tanks which have been listed and labeled by CSA International and Lati nal Sanitation Foundation.
- G. ASME Code Symbol Stamps: For the following equipment, comply with ASME Boiler & Pressure Vessel Code for construction and stamp with ASME Code Symbol:

Packaged Domestic Water Heater

All packaged equipment shall be independently third party, labeled as a system for its intended use by a nationally recognized testing laboratory (NRTL) in accordance with OSHA Federal Regulations 29CFR 1910.303 and .349 as well as NFPA Pamphlet #70 and NEC Article 90.7.

- 1.4 QUALITY ASSURANCE
 - A. Refer to Section 220010 for a general description of requirements applying to this section.

1.5 SUBMITTALS

- A. Submit shop drawings and product data in accordance with Section 220000.
- B. Submit the following:
 - 1. Product data on all equipment including roughing-in data.
 - 2. Connection diagrams for related piping and specialties.
- 1.6 WARRANTY/GUARANTEE
 - A. All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, GENERAL REQUIREMENTS

PART 2 – PRODUCTS

2.1 EQUIPMENT

- A. Refer to "Plumbing Fixture & Equipment Schedule" for type, numbers, aze and manufacturer of all equipment accessories.
- 2.2 HOT WATER CIRCULATING PUMPS
 - A. Provide and install where indicated on the drawings, domestic hot water circulating pumps complete with controls and piping as shown on the drawings. Each pump shall have a capacity of 5 gallons per minute against a total discharge head of 17 feet.
 - B. Pumps shall be close coupled, centrifugal appendit low lead or lead free bronze with flexible connection to a 1/6 HP, 1750 RPM, 60 cycle, 100 volt, single phase motor.
 - C. Pump shall be controlled by a manual starter, furnished and installed by the Electrical Contractor. Pumps shall run continuously and be were into night setback operations.

PART 3 – EXECUTION

- 3.1 INSTALLATION OF THE CLOSTATIC MIXING VALVE
 - A. Install mixing value in accordance with manufacturer's installation instructions and in compliance with applicable code
 - B. At startup of donestic hot water system, mixing valve outlet temperature shall be checked to insure proper cetting and operation. Following adjustments, if required, the mixing valve, if not performing, check if factory required differential temperature in/out with a minimum of 20° Delta "T" is maintained.

Novide minimum 27" (end of pipe to end of pipe) heat trap in accordance with manufacturer's recommendations and as detailed on the drawing.

The temperature of the water delivered by the mixing valve shall be changed by turning the adjusting screw to the right or clockwise for lower temperature; and to the left or counter clockwise for higher temperatures. Maintain a uniform temperature regardless of temperature of incoming water. To facilitate adjustment, a thermometer shall be placed in the line beyond the Holby Tempering Valve as shown in the diagram and water shall be flowing through the Holby Tempering Valve while adjustment is being made.

- E. Check valves shall be installed on both inlet (hot and cold) to the unit. Include a full size bypass valve arrangement.
- F. The hot water return line shall always be piped through the cold water make-up side of the mixing valve.

END OF SECTION 220150

T to BIDDING PURPO



SECTION 220190

TESTING – PLUMBING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. The general provisions of the Contract, including the Conditions of the Contract (General, Supplementary and other conditions, if any) and Division 1 as appropriate, apply to the work specified in this Section.
- 1.2 DESCRIPTION OF WORK
 - A. Extent of plumbing systems to be tested is indicated on the drawings and by requirements of this section.
 - B. Applications of tests include the following:
 - 1. Interior Piping
 - a. Domestic cold, hot & hot water return piping
 - b. Sanitary and condensate waste drainage piping
 - c. Storm water drainage piping
 - d. Compressed air piping
 - 2. Exterior Piping
 - a. Storm drainage piping
 - D. See Fire Protection Specifications for testing of Fire Protection Systems.
- 1.3 REFERENCE STANDARDS
 - A. Refer to Section 220000 for a eneral description of requirements applying to this section.
- 1.4 QUALITY ASSURANCE
 - A. Refer to Section 22,0010 for a general description of requirements applying to this section.
- 1.5 SUBMITTALS
- A. Submit est reports in accordance with Section 220000.
- 1.6 VARRANTY/GUARANTEE

All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, GENERAL REQUIREMENTS.

- PART 2 PRODUCTS
 - 2.1 PIPE & FITTING REPLACEMENTS
 - A. Refer to Section 220010 for replacement of any defective pipe or fittings. Replacement shall include all required uncovering, excavating, recovering and backfilling.

PART 3 - EXECUTION

3.1 GENERAL

- A. All exterior or interior piping shall be tested and approved before backfilling or concealing. Failure to secure the approval of the Municipal Inspector, Utility Company's Inspector or the Inspector of the Architect/Engineer makes it mandatory for the Contractor to completely expose the piping for testing. All expense involved in the uncovering of the piping for the test and recovering shall be bornely the respective Contractor with no change in Contract.
- B. All equipment, material and labor required for testing a plumbing system or part the equipment by the Plumbing Contractor responsible for installing the work.

3.2 INTERIOR PIPING

A. Drainage Piping:

Rough Plumbing: The piping of all plumbing storm, condensate waster san tary drainage and venting systems shall be tested upon completion of the rough piping installation by water or air and proved watertight. Where required by the code official, the cleanout plugs shall be removed to ascertain if the pressure has reached all parts of the system. Either of the following methods shall be used:

1. Water Test: The water test shall be applied to the dranage system either in its entirety or in sections after rough piping has been installed. If applied to the entire system, all openings in the piping shall be closed, except the highest opening, and he system filled with water to the point of overflow. If the system is tested in sections, each opening shall be plugged except the highest opening of the section under test, and each section shall be filled with water, but a section shall not be tested with less than a 10-foot head of water.

In testing successive sections, at least the upper 10 feet of the next preceding section shall be tested, so that a joint or pipe in the beild ang (except the uppermost 10 feet of the system) shall not have been subjected to a test of less than a 10-foot head of water. The water shall be kept in the system or in the portion undertest for a minimum of 15 minutes before inspection starts. The system shall then being test all points.

2. Air Test: The air testenall be made by attaching an air compressor testing apparatus to an opening, and, after closing all other inlets and outlets to the system, forcing air into the system until there is a gauge pressure of 5 pounds per square inch (5 psi) or a minimum of 10-inch column of mercury. This pressure shall be held without introduction of additional air for a minimum period of 15 minutes.

<u>Pre-autionary Note:</u> The compressibility of air and/or other gases result in tremendous amounts of stored energy, even at lower pressures. Over-pressurizing creates a substantial hazard to personnel and property near the area should a failure occur. Consult with the Plastic Pipe Institute (PPI) for statements and alerts, along with State and local safety offices.

Finished Plumbing: Where required by the code official, after the plumbing fixtures have been set and their traps filled with water, their connections shall be tested and proved gas and water-tight by one of the following test methods.

1. The final test for gas and water-tightness of the completed drainage and vent systems shall be made by a smoke test or other approved method. The test shall be made by filling all traps with water, and then introducing into the system smoke produced by one or more smoke machines. When the smoke appears at stack openings on the roof, the stack openings shall be closed and a

TESTING – PLUMBING 220190-2

pressure equivalent to a 1" water column shall be built and maintained for the period of the inspection.

2. After the plumbing fixtures have been set and their traps filled with water, their connections shall be tested and proven gas and water-tight by plugging the stack openings on the roof and building drain where the drain leaves the building and with air introduced into the system equal to the pressure of a 1-inch water column. This shall be accomplished by the use of a "U" tube or manometer inserted in the trap of a water closet. Such pressure shall remain constant for the period of inspection without the introduction of additional air.

Building sewer test: The building sewer shall be tested by insertion of a test plug al the point of connection with the public sewer or individual sewage disposal system. The building sever shall then be filled with water under a head of not less than 10 feet. The water level at the top of the test head of water shall not drop for at least 15 minutes.

B. Domestic Water Piping: All new, altered, extended or replaced interior water piping installed shall be tested at 100 psig maintaining the pressure for four hours with not more ten 1% drop in pressure. The system shall be filled with water which shall remain in the system until the water and the piping are the same temperature. If water pipe testing is under the julisdiction of the local inspector, his requirements shall be used; however, they shall be not less man specified herein. The tests shall be performed in the presence of the representative of the Archited/Engineer and to his satisfaction.

3.3 STERILIZATION

- A. After final testing for leaks, all new potable water opping i stalled including water service piping, shall be flushed to remove foreign material.
- B. Before placing domestic water systems in service, a qualified service organization shall be engaged, to sterilize the entire building including the exterior water service piping in accordance with the following procedure:
 - 1. Contractor shall provide 37." hore connection somewhere in the main entering the building, or in the Mechanical Room and/or in the meter pit, pump in sufficient sodium hypochlorite to produce a free available colorine residual of not less than 100 PPM.
 - 2. Proceed upstream from the point of chlorine application opening all faucets and taps until chlorine is detected. Close faucets and taps when chlorine is evident. Consult with the local code department for additional concentrations and durations.
 - 3. When chlorin ated water has been brought to every faucet and tap with a minimum concentration of 200 PPM chlorine, retain this water in the system for at least three hours.



Proceed to open all faucets and taps and thoroughly flush all new lines until the chlorine residual in the water is less than 1.0 PPM.

- 6. Obtain representative water samples from the system for analysis by a recognized Bacteriological Laboratory.
- 7. If all samples tested for impurities and organisms are negative, a letter and laboratory reports shall be submitted by the service organization to the contractor, certifying successful completion of the sterilization.

- 8. If any samples tested indicate the presence of harmful impurities and organisms, the entire sterilization procedure shall be repeated.
- 9. Plumbing Contractor shall provide plumbing connections and power for pumping chlorine solution into the system.

<u>Warning: PVC and CPVC Pipe:</u> Do not use a dry granular calcium hypochlorite as a disinfecting material for water purification in potable water piping systems. The introduction of granules or relies of calcium hypochlorite with solvent cements and primers (including their vapors), may result in violent chemical reactions.

C. Available Service Organizations: Subject to compliance with requirements, provide the sterilization service of one of the following:

Water Chem

Arc Company, Inc.

Nova Consultants

Artesian Water Co.

END OF SECTION 220196

SECTION 220191

BALANCING – PLUMBING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. The general provisions of the Contract, including the Conditions of the Contract (General, Supplementary and other conditions, if any) and Division 1 as appropriate, apply to the work specified in this Section.
- 1.2 DESCRIPTION OF WORK
 - A. Extent of plumbing systems to be balanced is indicated on the drawings and burequirements of this section.
 - B. Applications of tests include the following:
 - 1. Interior Piping
 - a. Domestic hot water and hot water return
- 1.3 REFERENCE STANDARDS
 - A. Refer to Section 220000 for a general description of requirements applying to this section.
- 1.4 QUALITY ASSURANCE
- A. Refer to Section 220010 for a general description of requirements applying to this section.
- 1.5 SUBMITTALS
 - A. Submit balancing report in accordance with Section 220000.
- 1.6 WARRANTY/GUARANTER
 - A. All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, GENERAL REQUIREMENTS.

PART 2 – PRODUCTS

- 2.1 PIPE & FIFTING REPLACEMENTS
- A. Refecto Section 220010 for replacement of any defective pipe or fittings. Replacement shall include all required draining of system, removal and replacement and uncovering, recovering.

PART 3 EXECUTION

GENERAL

All new hot water return piping installed or wherever system valves are being replaced, the system shall be tested, balanced and approved before concealing. Failure to secure the approval of the Municipal Inspector, A/E Inspector or the Inspector of the Owner makes it mandatory for the Contractor to completely expose the piping for balancing. All expense involved in the uncovering of the piping for the balancing and recovering shall be borne by the respective Contractor with no change in Contract.

B. All equipment, material and labor required for balancing a plumbing system or part thereof shall be furnished by the Plumbing Contractor responsible for installing the work.

3.2 INTERIOR PIPING

- A. Domestic Hot Water Return System: Upon completion of the testing of the domestic hot water supply and recirculation systems, a final procedure is to be performed to obtain uniform circulation within each hot water loop of the domestic hot water system. At the ends of the hot water mains, or wherever a branch return line connects to the main return line, there shall be three (3) valves: ball valve, check valve and balancing valve. These valves are to be installed in an accessible space at/or above the ceiling or where indicated on the drawings.
- B. Based on an Accu-Flo balancing valve, the use of a differential pressure gauge Morel No.779 shall be used to achieve the greatest accuracy.

END OF SECTION 220191

Republick

SECTION 230200

GENERAL PROVISIONS – HVAC

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. The general provisions of the Contract, including the Conditions of the Contract (General, Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to work of this Section.
- B. This specification or drawing and the design features or resulting construction disclosed are the property of Furlow Associates, Inc., and shall not be reproduced without written permission.
- C. All Mechanical Systems shall be part of and included in all of the followin : 20200 thru 230950.

1.2 WORK INCLUDED

- A. Provide labor, materials, equipment and supervision necessary to install complete operating HVAC Systems, including all work at the site and within the proposed construction areas to accomplish the required work.
- B. Wherever the term "provide" is used, it shall be understood to mean both "furnish" and "install".
- 1.3 REGULATIONS, CODES AND STANDARDS
 - A. Work shall be performed in accordance with latest adopted codes, regulations and ordinances by authorities having jurisdiction. Observe all safety regulations.
 - B. Obtain all permits and inspection certaficates and pay all charges.
 - C. Make or arrange for utility conjuctions for pay all charges.
 - D. Latest editions of any referenced standards shall govern.
- 1.4 RELATED WORK
 - A. Refer to equipment shown or specified in sections of Division 1 thru 14 and 26 that will require Mechanical service and provide such service.
 - B. Refer to work related to HVAC as shown on the following contract drawings:

Architectural & Structural



El ctrical

This Contractor shall coordinate with the work of Division 26 and the Fire Alarm System vendor for locations and mounting of all duct smoke detectors. These devices are shown on the Mechanical Drawings for reference only to show the intent of the work. All locations shall be determined based on approved shop drawings from the Fire Alarm System vendor and the Contractor for the work of Division 26, Electrical.

1.5 COORDINATION

- A. The Mechanical, Plumbing and Electrical Contractors are responsible to coordinate all manufacturer's recommended circuit breakers, starters, disconnects and fuse sizes for all equipment. Submission of a shop drawing will certify that this has been completed. Any necessary changes required will be included as part of this contract.
- B. Mechanical Contractor shall coordinate scheduling, submittals and work of the various sections of the Project Manual to ensure efficient and orderly sequence of installation of independent work compute, with provisions to accommodate items that may be installed at a later time.
- C. Mechanical Contractor shall verify utility requirements and all characteristics of operating equipment are compatible with the building utilities. Coordinate the work of all sections related and required for installing, connection and placing in service of all equipment.
- D. Mechanical Contractor shall coordinate all space requirements, supports and installation of all mechanical, electrical, plumbing and fire protection work, which are indicated diagrammatically on the Drawings. Verify routing of all pipes, ducts, conduits and equipment connections. Maximize accessibility for other work, and service requirements for maintenance and repairs. Develop overall coordination drawing (all trades) and submit for review prior to her herication.
- E. Obtain written confirmation from all related trade Contractor and the Owner or his representative that requirements, conflicts and coordination issues have been discussed and resolved.
- F. Submit coordination drawings to verify access and clearances

1.6 DEFINITIONS

- A. Finished Spaces: Spaces other than mean niced and electrical equipment rooms, furred spaces, pipe and duct shafts, unheated spaces immediately below roof, spaces above ceilings, unexcavated spaces, crawl spaces, and tunnels.
- B. Exposed, Interior Installations, Exposed to view indoors. Examples include finished occupied spaces and mechanical equipment points
- C. Exposed, Exterior Installation. Exposed to view outdoors, or subject to outdoor ambient temperatures and weather conditions. Examples include rooftop locations.
- D. Concealed, Installations: Concealed from view and protected from physical contact by building occupants. Examples include above ceilings and in duct shafts.
- E. Concealed, Exterior Installations: Concealed from view and protected from weather conditions and physical contact by building occupants, but subject to outdoor ambient temperatures. Examples include installation within unheated shelters.

SUBMITTALS

Shop Drawings:

- 1. Shop drawings shall be submitted in accordance with Division 1 of these specifications except where herein modified.
- 2. Shop drawings comprising complete catalog cuts, performance test data for HVAC equipment as required by other sections of Division 23, shall be submitted for review checking. The Contractor shall review these shop drawings for conformance to contract documents prior to submission and affix contractor's signature to each submittal certifying that this review has been done. By

approving and submitting shop drawings, product data, samples and similar materials, the Contractor represents that the Contractor has determined and verified materials, field measurements and field construction data that relates to the work, and has checked and coordinated this information with all of the requirements contained in the contract documents for the work of all trades.

- a. The Contractor and equipment manufacturer shall clearly indentify in all submittals and shan drawings any and all applications standards which require additional work to accomplodate this equipment and provide a complete and operational system as described in the contract documents.
- b. The Contractor shall be completely responsible for any and all additional costs associated with the changes required by this and all other trades.
- 3. Submit a 1/4" scale layout of all mechanical equipment rooms. All equipment and pads shall be to scale of equipment being furnished. Obtain size information of any and all equipment from other trades and indicate on drawings. The drawings shall be fully coordinated with all trades prior to submission. Indicate coil pull areas, filter pull areas, mantenance clearances, and access as applicable.
- 4. All shop drawing submittals shall have the following icentification data, as applicable, contained therein or permanently adhered thereto.
 - a. Project name.
 - b. Project number.
 - c. Sub-contractor's, vendor's and/or manufacturer's name and address.
 - d. Product identification.
 - e. Identification of deviation from contract documents.
 - f. Applicable contract drawings and specification section number.
 - g. Shop drawing tite, drawing number, revision number, and date of drawing and revision.
- 5. Resubmit revise or additional shop drawings as requested.
- 6. Where er scopelrawings or vendor's standard data sheets indicate work to be done "by others", it shall be the esponsibility of the contractor making the submission to identify by name, the contractor who is to do this work. If the contractor named is other than the contractor making the submission, the shop drawing submission must be reviewed by the named contractor and bear his mark of approval, prior to submission to the Architect/Engineer.



Where equipment proposed differs from that shown on the drawings or specified, he shall submit for approval drawings showing the manner in which the layout is affected by the substitution.

The Contractor shall keep one copy of approved shop drawings at the job site,, filed in a suitable metal container. The shop drawings shall be cataloged and kept in good repair, and shall be available for use by the Owner, Architect and Engineer.

9. No equipment shall be ordered, fabricated, etc., before approval of shop drawings.

- B. Contractor is responsible for the shop drawing coordination and interface with the work of other contracts and adjacent work. The relationship of Contractor's work shall be verified as it relates to adjacent and critical features of the work of this and all contracts and materials.
- C. The Contractor shall submit a complete schedule of all shop drawings required for the scope of work covering all materials and equipment listed in all sections of Division 23, Mechanical, including all documents required for contract closeout, Owner instructions and training, and all turnover items at the completion of the work. This schedule shall be submitted for review and approval within thirty, days of contract award and before any subsequent materials are provided for review.
- D. The shop drawings provided by the Contractor will be reviewed only once and resubnittas will be reviewed only once. Any other submittals will be billed to the Contractor at the Ergmer's standard rates.

1.8 SITE INSPECTION

- A. The Contractor shall visit site, inspect, and become aware of all conditions which may effect the work during the estimation phase of his work prior to bid openings. Invisigate utilities, protection requirements for adjacent facilities, storage locations, and access to the construction area.
- B. Submission of a bid will be deemed evidence of having completed with this requirement.
- 1.9 SUBSTITUTIONS
 - A. Whenever a material, article, piece of equipment or system is identified in the following specification or indicated on the drawings by reference to manufactures' or vendors' names, trade names, catalog numbers or the like, it is so identified for the variable of establishing the basis of the Bid.
 - B. Substitution approval must be obtained and included as an addendum item prior to the submission of the bid. An approved substitution shall no be considered as an approval for the Contractor or an equipment vendor to deviate from the writen portion of the specifications unless so stated in the addendum.
 - C. The drawings illustrate and space allocated for equipment and the Contractor shall install the equipment accordingly. If changes are required in the building or arrangement due to substitution of equipment, the Contractor backing the substitution must pay for the necessary modifications.
 - D. The listed equivalent or substituted manufacturers along with the bidding related contractor shall be completely responsible to comply with all requirements on all contract documents. This shall include, but not be imited to all: space requirements, code clearances, type-horsepower-capacities-number and size or services required from other trades including all auxiliary items provided by this Contractor and all other trades, and all manufacturer's specific equipment applications standards and requirements, for approved equipment including that which is basis of design or a substitution. The bidding related contractor and equipment manufacturers shall clearly identify in all submittals and shop drawings any and all applications standards which require additional work to accommodate this equipment and provide a complete and operational system as described in the contract documents. If the bidding contractor or manufacturer does not comply with these requirements then they shall be completely responsible for any and all additional costs associated with the changes required by this and all other trades.
 - E. Where only one brand name or manufacturer is identified, no substitutions are permitted.
 - F. Substitutions:

- 1. Until a date no later than seven (7) days before the date Bids are due, Architect will consider written requests from bidders for substitution of Products. Architect will review requests and will notify Bidders in an Addendum if the requested substitution is acceptable.
- 2. Submit a separate request for each Product, supported with complete data, with drawings and samples as appropriate, including:
 - a. Comparison of the qualities of the proposed substitution with that specified.
 - b. Changes required in other elements of the work because of the substitution.
 - c. Effect on the construction schedule.
 - d. Cost data comparing the proposed substitution with the Product specifie
 - e. Any required license fees or royalties.
 - f. Availability of maintenance service, and source of replacement miterials
- 3. Architect shall be the judge of the acceptability of the proposed ubstitution.
- 4. A request for a substitution constitutes a representation in Bidde
 - a. Has investigated the proposed Product and determined that it is equal to or superior in all respects to that specified.
 - b. Will provide the same warranties or bonds for the substitution as for the Product specified.
 - c. Will coordinate the installation of an accepted substitution into the work, and make such other changes as may be required to make the work complete in all respects.
 - d. Waives all claims for additional cests, under his responsibility, which may subsequently become apparent.

1.10 LUBRICATION

- A. Provide and maintain as required lubrication of any equipment operated prior to acceptance by the Owner. Lubrication shall be as recommended by the equipment manufacturer.
- B. Provide one year's supply of lubricants to Owner at date of acceptance.
- C. Verify that required subrication has taken place prior to any equipment start-up.
- 1.11 EQUIPMENT START-UP
 - A. **Perify proper installation by manufacturer or his representative.**
 - Active General Contractor 2 days prior to actual start-up.
 - Verify proper operation. Obtain signed statement by manufacturer or his representative that equipment is operating within warranty requirements. Submit statement to General Contractor.
- D. Perform field mechanical balancing in accordance with Section 230950: TESTING AND BALANCING OF MECHANICAL SYSTEMS.
 - E. The Mechanical Contractor shall own as part of his work, the following:

Provide one (1) additional drive set, if necessary, to obtain final design balancing requirements. The Mechanical Contractor shall coordinate with Balancing Firm and equipment manufacturer for drive selection, including belts and pulleys.

1.12 OPERATION & MAINTENANCE INSTRUCTIONS

- A. Properly and fully instruct Owner's personnel in the operation and maintenance of all systems and equipment.
 - 1. Contractor to demonstrate all systems to Engineer for verification of operation prior to O instruction period.
 - 2. Provide two (2) 4-hour sessions of training to School District/Owner's Maintenance Star
- B. Insure that the Owner's personnel are familiar with all operations to carry op required ectivities.
- C. Such instruction shall be for each item of equipment and each system are when
- D. Provide report that instruction has taken place. Include in the report the equipment and/or systems instructed, date, contractor, Owner's personnel, vendor, and that a compete operating and maintenance manual has been reviewed.
- E. Manual shall include all instructions on operation, mantenance, repair parts list, lubrication requirements, brochures, catalogue cuts, complete schedule of air filters for each unit type in Excel spreadsheet format, wiring diagrams, piping diagrams, compol sequences, service requirements, names and addresses of vendors, suppliers and emergency contacts. Three manuals shall be provided.
- F. Submit manuals for review prior to operating instruction period. Manuals shall be 8-1/2 x 11" with hard cover, suitably bound.
- G. Provide to the Owner any special tools necessary for operation and routine maintenance of any of the equipment.
- H. Upon completion of the project the bechanical Contractor shall provide a complete set of legible asbuilt drawings for the Owner
- I. Furnish three (3) copies of professionally taped video and three (3) copies of professionally prepared drawings demonstrating the following:
 - Locations of main shut-off valves.
 - Proceedings for equipment start-up and seasonal shut-downs.
 - Procedures for maintenance.

Provide written version of all procedures included in video.

The above should cover all equipment/systems including, but not limited to, the following:

Hot water distribution system.

- Chilled water distribution system.
- Air handlers
- Energy recovery unit
- Fans

- Pumps
- D/X cooling units
- VAV box
- Unit ventilators/fan coil units
- Chillers
- Cooling tower
- ATC System
- 1.13 TOOLS
 - A. All equipment furnished by the Mechanical Contractor which requires special tools or devices other than those normally available to the maintenance or operating staff shall optimished in duplicate to the Owner, sufficiently marked, packed or boxed for staff usage. The tools previded shall be listed by the Mechanical Contractor identified as to their use or the equipment applicable in a written transmittal to the Owner.
- 1.14 CLEANING AND FINISHING
 - A. After equipment start-up and all operating tests have been made and the system pronounced satisfactory, each respective Contractor shall go over meentire project, clean all equipment, etc., installed by him and leave in a clean and working condition. Any surfaces found marred after this final cleaning shall be refinished or replaced as the Contractor at no cost to the Owner.
 - B. Provide for the safety and good condition of al materials and equipment until final acceptance by the Owner. Protect all materials and equipment from damage. Provide adequate and proper storage facilities during the progress of an work Special care shall be taken to provide protection for bearings, open connections, pipe coils pumps, compressors and similar equipment.
 - C. All fixtures, piping, finislees subjects and equipment shall have all grease, adhesive labels and foreign materials removed.
 - D. All piping shall be trained and flushed to remove grease and foreign matter. Pressure regulating assemblies, traps, and similar items shall be thoroughly cleaned. Remove and thoroughly clean and reinstall all neuron screens after the system has been in operation ten (10) days.
 - E. When connections are made to existing systems, the Mechanical Contractor shall do all cleaning and purgress of the existing systems required to restore them to the condition existing prior to the start of work.
 - Chan-up: Remove from the premises, all unused material and debris resulting from the performance of work under this section.
 - 2 PRODUCTS

GENERAL

A. All material and equipment shall be new and of present day manufacture, and shall conform to accepted standards of the trade where such a standard has been established for the particular type of equipment or material.
- B. Whenever equipment or material is referred to in the singular, such as "the fan", it shall be deemed to apply to as many such items as necessary to complete the work.
- 2.2 PRODUCT DELIVERY, STORAGE AND HANDLING
 - A. During loading, transporting and unloading exercise care to prevent damage to material.
 - B. Store all materials in dry enclosures or under protective coverings out of way of work progress
 - C. Material shall not be allowed to be stored directly on ground.
 - D. Deliver in manufacturer's original cartons or on skids.
 - E. Handle and protect so as to prevent damage to product or any surrounding material.

2.3 CONCRETE

A. Concrete shall be in accordance with Section 03300.

2.4 WARRANTY

A. Wherever in the specification sections of this division, reference is made to a specific warranty period, this warranty shall be in addition to and not a limitation of concrete ghts the Owner may have against the Mechanical Contractor under the contract documents.

PART 3 - EXECUTION

- 3.1 **PROTECTION**
 - A. Plug or cap open ends of piping systems, conduit and ductwork.
 - B. Stored materials shall be covered to reven damage by inclement weather, sun, dust or moisture.
 - C. Protect all installed work until accepted to place by the Owner.
 - D. Plates, polished metal escutchenes, thermostats and other finished devices shall not be installed until masonry, tile, and painting gravations are complete unless otherwise protected.
 - E. Protect all work from operations which may cause damage such as hauling, welding, soldering, painting, insulating and covering.

3.2 WORKMANSEIP

- A. Installall werk yeat, trim and plumb with building lines.
- B. Install work in spaces allocated.
 - Outting and patching shall be performed by skilled tradesmen normally employed for the work involved.

This Contractor shall provide a complete weathertight seal to all new systems in the building including the necessary caulking, weather-stripping and insulation.

- 3.3 EQUIPMENT SETTING
 - A. Provide as a minimum, a 4-inch concrete pad beneath all floor-mounted equipment. Install anchor bolts in pour.

- B. Provide as a minimum, spring vibration isolation under any equipment 10 HP and over and rubber in shear vibration isolation on any equipment up to 10 HP. For further specifications and additional requirements, refer to other sections.
- C. Concrete shall be 3,000 psi, 28 day compressive strength in accordance with ACI-613. Reinforce with No. 4 rod 12" on centers both ways or as otherwise detailed.

3.4 FASTENERS, HANGERS AND SUPPORTS

- A. Provide all hangers and supports required to suspend, mount, or hang the work.
- B. Provide all miscellaneous steel angles, channels, beams, clips, brackets and anchors necessary that or support the work. Provide submissions for review.
- C. Install concrete inserts before concrete is poured.
- D. Drilled inserts shall not be loaded more than 1/4 rated capacity.
- E. Power-driven fasteners shall not be allowed for piping larger than 2 ince or equipment. When used they shall not be loaded more than 1/8 rated capacity or 200 pounds.
- F. All hangers, miscellaneous steel, braces and supports shall be gavanized, cadmium plated, or primed steel. Copper tubing shall be supported with copper hanger.
- G. Piping shall be supported from adjustable clevis type hangers with insulation pipe saddles or pipe shields in accordance with piping support spacing able of the drawings. Where hangers are 18" or longer provide lateral bracing at every fourth hanger.
- H. Support vertical piping at floor levels. Piping shall have split rings.
- I. Any lintels required for openings for his work if not indicated on Architectural or Structural drawings shall be provided under this Section
- J. Piping on the roof shall be supported by an engineered, prefabricated hanger system specifically designed for installation on the roof without roof penetrations, flashing or damage to the roofing material. The system shall comist of bases made of high density polypropylene plastic with additives for UV protection, hot dipper galvanized structural steel frames, hangers, fasteners, rods, etc. The system shall be complete and designed to fit the piping installed under actual conditions of service. The system shall be furnished as manufactured by PHP Systems & Design or Anvil International Haydon HaBlock
- 3.5 SLEFVES
 - A. Provide each pipe, duct or conduit passing through a masonry or concrete wall, floor or partition with a sleeve made from standard weight steel pipe for pipe or conduit and No. 12 gauge galvanized steel for ducts, with smooth edges, securely and neatly cemented in place. Provide each pipe, duct or conduit passing through a frame or metal partition with a sleeve made from No. 22 gauge galvanized sheet metal, securely fastened in place.

Be responsible for the proper location and alignment of all sleeves.

2. Provide hydrostatic seals for sleeves passing through outside walls, either above or below grade, or through hydrostatically sealed slabs or floors on grade. Provide fire-rated seals for all sleeves which penetrate fire-rated walls.

- D. Install both piping and sleeve seals so as to maintain integrity of seals with expansion and contraction of piping.
- E. Set floor sleeves flush with floor surface in finished areas, 1" above the finished floor in kitchens, cafeterias, and similar service areas unless such areas are slab-on-grade; 1" above the floor in mechanical rooms, pipe chases, pipe spaces and other unfinished areas, unless otherwise indicated, and flush with the underside of slabs. Extend wall and partition sleeves through and cut flush with each surface unless otherwise indicated or specified.
- F. Select sleeves two pipe sizes larger than any pipe or conduit that is to remain uncovered, tales otherwise required by the sealing method specified. Where pipes are to be covered, provide seeves large enough to allow the covering to pass through the sleeves with sufficient clearance force aing as specified hereinafter. Size sleeves for branch piping from vertical risers large enough to permit vertical expansion at the riser.
- G. Select duct sleeve sizes to suit requirements of fire dampers and sealing methods as specified.
- H. Place sleeves imbedded in concrete floors or walls in the forms before concrete is poured; sleeves shall have integral waterstop flanges, where they are to receive either water igns or hydrostatic seals.
- I. Install sleeves passing through above-grade floors of medianical rooms, toilet rooms, kitchens or similar service areas where liquid leaks or spillover may occur in a watertight manner. Sleeves shall be such that waterproofing membrane can be flashed around and into the sleeve where necessary.
- J. Seal sleeves for pipes or conduit passing through ceiling ar plenum walls or the floor above air tight in a manner similar to that specified for fire-rated leeves.
- K. Hydrostatic Sealing Method: Provide compressible synthetic rubber seals, equivalent to LINK SEAL, manufactured by the Thunderline Corporation, or THRUWALL manufactured by O.Z. Gedney. Install seals in accordance with the manufacturer's recommendations to provide air tightness aboveground and hydrostatic scaling between the calking or other type mastic is not acceptable.
- L. Fire-Rated Sealing Method:
 - 1. Sleeves, openings as descalants shall comply with applicable codes, recommended practices and standards, and manufacturer's instructions. Fire sealants shall have ability to prevent spread of flame, smole or water throughout the penetration and shall pass 3-hour test, UL test ASTM E814 and UL 1479.
 - 2. Products: Clase Corporation CTC PR-855, O. Z. Gedney CRS/CAFS, 3M Electro-Products Division Fatty 303 or Caulk CP25 penetration sealing kits, General Electric Company sealants typ_RTV-850, 6428 or 7403, Hilti FS-one, Thunderline Corporation "Link-Seal Pyro-Pak". Instantion and type of sealant to be used as recommended by the manufacturer.
 - PLA TES

Provide chrome plated plates wherever piping passes into finished area.

Plates shall be securely fastened to piping or building construction.

- C. Floor plates shall cover 1 inch sleeve extension.
- 3.7 OFFSETS, TRANSITIONS, MODIFICATIONS
 - A. Provide all offsets necessary to install the work and to provide clearance for other trades.

- B. Maintain adequate headroom and clearance.
- C. Incidental modifications necessary to the installation of the systems shall be made as necessary and as approved by the Architect.

3.8 RECESSES

- A. Furnish information to the General Contractor as to sizes and locations of recesses required to instrupanels, boxes, and other equipment or devices which are to be recessed in walls.
- B. Make offsets or modifications as required to suit final locations.
- 3.9 LABELING
 - A. All HVAC equipment such as pumps, fans, air handling units, and devices requiring identification for operating procedures shall be provided with permanent black laminated micerta value core labels with 3/8 inch letters.
 - B. This shall also apply to all controllers, remote start/stop pushbuttors are equipment cabinets.
 - C. This shall not apply to individual room thermostats.
 - D. All Mechanical Rooms shall be identified with a permakent placard of red-white-red laminated, commercial grade, plastic construction. Letters shall be minimum one inch high and read in capital letters: WARNING MECHANICAL EQUIPMENT ROOM LIMITED ACCESS. Placard shall be centered on each door leading into the mechanical room a five feet above the floor and attached at each corner with brass screws.
 - E. Refrigeration Machinery Rooms shall be identified with a permanent placard of red-white-red laminated, commercial grade, plastic construction. Letters shall be minimum one inch high for the header to read in capital letters: WARNING-REFRIGERATION EQUIPMENT ROOM LIMITED ACCESS. The following information shall be posted in a similar fashion, minimum half-inch high capital letters, indicating:
 - 1. The name and address of the installer.
 - 2. The refrigerant number and amount of refrigerant in pounds.
 - 3. The lubrical tirentity and amount in pounds or ounces as appropriate.
 - 4. The field test pressure applied to the equipment in psig.

Place d shall be centered on each door leading into the refrigeration equipment room at five feet above the flow and attached at each corner with brass screws.

At all fire damper, smoke damper and combination fire/smoke damper locations, access doors in duc work shall be identified with a permanent placard of red-white-red laminated commercial grade plastic construction, minimum one-half inch high capital letters, reading, "FIRE DAMPER", "SMOKE DAMPER", "FIRE/SMOKE DAMPER" as appropriate for the installation. Attach securely to face of access door with brass screws at each corner, sealed airtight.

3.10 FLASHING AND COUNTERFLASHING

- A. Roof curbs, etc., shall have counterflashing fittings. General Contractor shall provide flashing.
- B. Piping and conduit thru the roof shall be flashed by the General Contractor. Provide counterflashing.

C. Provide curbs with base features required to match roof materials, finishes and configuration; e.g., flat, sloped, raised seam, etc.

3.11 ACCESS

- A. Locate all equipment, valves, devices and controllers which may need service in accessible places.
- B. Where access is not available, access panels shall be provided. Furnish access panels to the Gene Contractor for installation.
- C. Access panels shall be Nailor-Hart Industries, Karp Co., or Controlled Air Manufacturing Linkite with 16-gauge frames and 14-gauge steel door, prime painted.
- D. Maintain access clearances for tube or fan removal, coil pulls, and filter removal
- 3.12 WIRING AND MOTOR CONTROLS
 - A. Packaged equipment shall be furnished with disconnect switches, starters, overloads, factory furnished and wired by the unit manufacturer.
 - B. Roof-mounted exhaust fans, except utility sets, rated less than 1/2 HP at 15 volts, single phase, shall be furnished with disconnect switches, factory furnished and wreakly unit manufacturer.
 - C. Rooftop equipment shall be furnished with starters, disconnect switches, overloads, factory furnished and wired by unit manufacturer.
 - D. This Contractor shall furnish all information and assistance required for the Electrical Contractor to purchase all motor starters that are not specification approximation of the mechanical equipment.
 - E. Control wiring shall be provided under this Division of the work.
 - F. All wiring shall be in accordance with the National Electrical Code and as recommended by the equipment manufacturer.
- 3.13 OPENINGS CUTTING REPAIRING
 - A. This Contractor shall experte with the work to be done under other sections in providing information as to openings required in walls, slabs and footings for all piping, ductwork and equipment, including sleeves where required.
 - B. Any drilling or cutting required for the performance of work under this Section, shall be the responsibility of his Contractor and the cost thereof shall be borne by him.
 - C. Holes in Concrete: Sleeves shall be furnished, accurately located and installed in forms before bouring of concrete. This Contractor shall pay all additional costs for cutting of holes as the result of the incorrect location of sleeves. All holes through existing concrete shall be either core drilled or saw cut. All holes required shall have the approval of the Structural Engineer prior to cutting or drilling.

It shall be the responsibility of this Contractor to ascertain that all chases and openings are properly located.

.14 PAINTING

A. This Contractor shall be responsible for painting required in conjunction with cutting and patching of existing building construction, in areas which are not scheduled for painting under the General Contract. This Contractor shall also be responsible for painting existing equipment, and/or piping, where finish is damaged by new work, in these same areas.

GENERAL PROVISIONS – HVAC 230200-12

Tetra Tech 200-15704-17001

- B. Refer to Division 1 for types of paint, color and finish.
- C. Surfaces subjected to temperatures below 180 degrees F, shall be painted with one coat of rust-resisting paint and one coat of high gloss enamel or sufficient finish coats for complete and uniform cover and high glossy finish.
- D. Surfaces subjected to temperatures above 180 deg. F, shall be painted with one coat of heat-resistant paint and one coat of heat resistant enamel, or sufficient finish coats for complete and uniform cover and high glossy finish.
- E. All painting shall be done in a careful, neat and workmanlike manner, with particular carebong exercised to protect adjacent building and equipment finishes. All surfaces shall be thoroughly cleaned of dirt, rust, scale, dust, grease, oil, debris and sanded, sand blasted or power brushed to properly prepare to provide bond for the paint. Contractor shall be entirely responsible for cleaning and preparing all surfaces. Should evidence appear that the surface was not properly prepared, the Contractor shall remove paint, prepare surface and repaint, as required appointed disting a contractor.
- F. All name plates, data plates that indicate manufacturer, model, size, tapa aty codes or identifying data on equipment painted, shall not be painted, but shall be carefully cut h.
- G. All exposed canvas, insulation jackets and other porous surfaces shall be cleaned and sized with at least two coats of sizing primer before finish coats are applied. Sizing shall completely cover canvas so that canvas will not be noticeable through the final-finish coat.
- H. Equipment factory painted and not damaged shall not be painted, except equipment herein specified to be painted a particular color. Damaged surfaces on factory painted equipment will necessitate the painting by this Contractor.
- I. Grilles, access panels, fan screen, convector, and unit heater enclosures and other prime-coated equipment in finished areas, will be painted by the General Contractor.
- J. Piping, fans, floor-mounted pipe supports, containers, hangers, pump and other equipment surfaces that are insulated or uninsulated snak be finished in accordance with the Owner's color code.
- K. The Mechanical Contractor shall furnish and lay drop clothes in all areas where painters finish work is being done, to protect floors and roofs and all other work from defacement. All temporary protections or coverings remove too early from any part of the work shall be promptly replaced, and any damage from neglect to do so shall be made good at the Mechanical Contractor's expense.
- L. At the end of each day, the Mechanical Contractor shall place in covered metal containers, or destroy, all clubs waste and refuse, which have been used in the application of inflammable paint materials. At the completion of work, all staging, scaffolding, containers and debris shall be removed from oremises, leaving all painting in perfect and clean condition. Upon completion, leave the work clean and free from blemishes. Hardware, tile, marble, and similar material shall be thoroughly cleaned of all paint.

GUARANTEE

All work shall be guaranteed to be free from defects for a period of one year of operation from date of acceptance by the Owner.

B. Guarantee shall be extended on an equal time basis for all non-operational periods due to failure within the guarantee period.

- C. All materials and equipment provided and/or installed under this section of the specifications shall be guaranteed for a period of one year from date of acceptance of the work by the Owner unless otherwise specified in Division 1. Should any trouble develop during this period due to defective materials or faulty workmanship, the Mechanical Contractor shall furnish necessary labor and materials to correct the trouble without any cost to the Owner. Any defective materials or inferior workmanship noticed at time of installation and/or during the guarantee period shall be correct a immediately to the entire satisfaction of the Owner.
- D. In the event of occupancy by the Owner prior to final acceptance of the project, the guarantee lace for equipment placed in operation shall be mutually agreed to by the Mechanical Contractor and the Owner's representative.
- E. Contractor to include an 11 month "walk-thru" of the building systems with representatives of the School District, Architect, Engineer and the Construction Manager. The purpose is to establish a list of corrective work that relates to operational issues, material/installation deficiencies, etc. prior to the expiration of the guarantee period.

3.16 DRAWINGS

- A. The Mechanical Systems are indicated on the Contract Drawings. Certain pertinent information and details required by the Mechanical Work appear on the Architectural, Structural and Electrical Drawings; become familiar with all drawings, and incorporate all pertinent requirements.
- B. Drawings are diagrammatic and indicate the general arrangement of systems and requirements of the work. Do not scale drawings. Exact locations of fixe resend equipment, not specifically shown, shall be obtained before starting work.
- 3.17 TESTING AND BALANCING OF MECHANINAL EQUIPMENT
 - A. Perform field mechanical balancing in accordance with Section 230950: TESTING AND BALANCING OF MECHANICAL AYSTEMS.
 - B. The Mechanical Contractor shall over as part of his work, the following:

Provide one (1) addition bariy) set, if necessary, to obtain final design balancing requirements. The Mechanical Contractor shall coordinate with Balancing Firm and equipment manufacturer for drive selection, including, lets and pulleys.

END OF SECTION 230200



SECTION 230210

BASIC MATERIALS AND METHODS – HVAC

PART 1 – GENERAL

1.1 RELATED DOCUMENTS

- A. The general provisions of the Contract, including the Conditions of the Contract (General, Supplementary and other conditions, if any) and Division 1 as appropriate, apply to the work specified in this section.
- B. Refer to Section 230200 for HVAC General Provisions.
- C. Refer to other sections in Division 23 for materials and methods not specified herei

1.2 DESCRIPTION OF WORK

- A. Included in this Section are the following:
 - 1. Steel Pipe and Fittings
 - 2. Copper Tubing & Fittings
 - 3. Polyvinyl Chloride (PVC) Pipe and Fittings
 - 4. Strainers
 - 5. Thermometers
 - 6. Gauges
 - 7. Test Stations Pressure/Temperal
 - 8. Isolating Fittings
 - 9. Unions
 - 10. Motors
- 1.3 REFERENCE STANDARDS
 - A. Refer to Section 30200 for a general description of requirements applying to this section.
 - B. Install work to neet the requirements of the following:
 - New Castle County Dept. of License and Inspections
 - International Mechanical Code
 - Gas Utility Company
 - . NFPA
 - 5. OSHA
 - 6. ASHRAE

7. Manufacturer's Standardization Society (MSS) of the valve and Fittings Industry, Inc.:

SP-58 Pipe Hangers and Supports Materials, Design and Manufacture.

SP-69 Pipe Hangers and Supports Selection and Application

- C. Appliances and materials governed by UL requirements shall meet such requirements and bear the label.
- 1.4 QUALITY ASSURANCE
 - A. Provide adequate supervision of labor force to assure that all aspects of the specifications are being fulfilled.
 - B. Verify that all work and equipment is installed in accordance with manufacturers warranty requirements.

PART 2 – PRODUCTS

- 2.1 STEEL PIPE AND FITTINGS
 - A. Water Piping:
 - 1. ASTM A53 seamless, Schedule 40.
 - 2. Fittings up to 2 inch shall be 150 lb. malleable ron derewed pattern ASME B16.3. Butt weld, ASME B16.9, same thickness as pipe.
 - 3. Fittings 2-1/2" and larger shall be butt with TEB16.9, same thickness as pipe.
 - 4. Weld-O-Lets and Thread-O-Lets snal be maximum of two sizes smaller than main size; i.e., maximum of a 2-inch Weld-O-Let on a 3-inch pipe.
 - 5. Thread tape shall be teflow ape, 3 mils minimum thickness.
- 2.2 COPPER TUBING & FITTING
 - A. Refrigeration Piping:
 - 1. Copper tubing: Type ACR, hard drawn temper.
 - 2. Fitting Whught-copper, solder joints, ASME B16.22 or ASME B16.26.
 - 3. Joints: Prazed, American Welding Society (AWS) Class BCUP-5 for brazing filler metal.
 - B. Water Fiping:
 - Tubing: Hard drawn seamless ASTM B-88 Type "L" aboveground.
 - 2. Soft seamless ASTM B-88 Type "K" below-ground.
 - B. Joint Material: Brazed joints, low temperature silver-bearing solder.
 - a. Flux shall be non-toxic type and non-corrosive.
 - 4. Fittings: ASME B16.15, B16.18, B16.22, or B16.26.
 - C. Condensate Drain Piping:
 - 1. Pipe: Copper tubing Type DWV.

2. Fittings: Wrought copper solder type drainage fittings, ASME B16.23 or B16.29.

2.3 POLYVINYL CHLORIDE (PVC) PIPE AND FITTINGS

- A. Pipe:
 - 1. ASTM D-1785 Schedule 40, Type 1, Grade 1.
 - 2. ASTM D-2665 (DWV) drain, waste and vent.
- B. Fittings:
 - 1. ASTM D-2466 Schedule 40.
 - 2. ASTM D-2665 DWV.
- C. Solvent Cement: ASTM D-2564 Schedule 40 and DWV.
- D. Uniformity: To insure installation uniformity, all piping components share be of one manufacturer.
- E. Flux shall be non-toxic type and non-corrosive.
- 2.4 STRAINERS (WATER)
 - A. Perforations: .033" pipe size to 2", .057" pipe size 2-1/2" , 1/8" pipe size 6" and up.
 - B. Self-cleaning "Y" type screwed end up to 2 inch with machined seats with blow-off outlet, stainless steel screen, iron body.
 - C. Self-cleaning "Y" type flanged 2-1/2 inch and up, with bolted cover flange, blow-off outlet, 125 psi ANSI, brass screen.
 - D. Duplex basket strainers: Fabricated rom cist fron, ASTM A126-B.
 - 1. 125 lb. ANSI class connections, while screwed ends up to 2 inch, flanged ends 2-1/2 inch and larger.
 - 2. Diverter valve shall be that cylindrical bronze plug, operated by a common sealed shaft and handle. The handle shall over the chamber in use to accommodate servicing the chamber that is contaminated
 - 3. Perforated statuless steel baskets, 0.033 inch diameter holes up to 2 inch; 0.064 inch diameter holes 1-1/2 to 4 inch size units.
 - 4. Shell test pressure shall be equal to 1-1/2 times working pressure; valve seat test pressure shall be equal to rated working pressure.

Accessories shall include the following:

- a. Differential pressure indicator mounted to the strainer body to visually indicate when the delta-P is 10 PSID (adjustable).
- b. Magnetic inserts mounted to the baskets to collect metallic particles.
- c. 100% shutoff with special seals within the changeover mechanism.
- d. Air eliminators to automatically vent air from the basket chamber after cleaning.
- e. Mounting studs and brackets for independent support.

- E. Manufacturer: Muesco, Sarco, Hoffman Specialties, Metraflex, Armstrong, Watson McDaniel.
- 2.5 THERMOMETERS
 - A. Separable socket, inserted into fluid flow, adjustable, hermetically sealed, red or blue indicating fluid, non-toxic, die-cast, baked enamel finish, double strength glass lens, white scale and black graduations_
 - B. Scale: Heating Water 30 deg. to 240 deg. F

Chilled Water - 0 deg. to 100 deg. F

Condenser Water-30 deg. to 150 deg. F

Dual Temperature Water – 30 deg. to 240 deg. F.

- C. Manufacturer: U.S. Gauge, H.O. Trerice, Moeller, Duro, Miljoco Corp., Write Instraments.
- 2.6 GAUGES
 - A. Phosphor bronze bourdon tube, polypropylene case, gasketed glass crystal, aluminum dial, black graduations 4-1/2-inch diameter.
 - B. Range: 0 to 60 psi, 5-pound intervals, 1/2-pound graduations
 - C. Install with bronze gauge cock.
 - D. Manufacturers: Danton, U.S. Gauge, H.O. Trerice, Moeller, Miljoco Corp., Winter Instruments, Weksler Instruments.
- 2.7 TEST STATIONS PRESSURE/TEMPERATURE
 - A. Provide a SISCO 1/4" or 1/2" NPT atting Test Plug) of solid brass at desired indicated locations. Test plug shall be capable of receiving other a pressure or temperature probe 1/8" o.d. Dual seal core shall be neoprene for temperature to 2 0 degrees F. Nordel to 350 degrees F and shall be rated zero leakage from vacuum to 1000 ost. V/T plug to have grooved cap and chain.
 - B. P/T plugs shall be provided with extensions as required by insulation.
 - C. Mechanical Contractor shall also provide the following: pressure gauge adapters with 1/8" o.d. probe, 5" stem pocket testing thermometers for 25° to 125° F (tower and chilled water) for 0° to 220° F (hot water) for 50° 500° F (temperatures above 220° F).
 - D. One (1) Master Test Kit shall be furnished to the Owners. Kit shall contain one (1) 2-1/2" test gauge of suitable range, one (1) Gauge Adapter 1/8" o.d. probe, and 5" stem pocket testing thermometers one (1) ¹⁰ 220° F and one (1) 50° 550° F.

Munufacturer: Sisco P/T Plugs.

ISOLATING FITTINGS

Provide isolating fittings between all sections of dissimilar piping materials or piping and equipment where one material is ferrous and the other is non-ferrous.

- B. Manufacturer: Epco Sales, Inc., or insulated unions by Central Plastic Co.
- 2.9 UNIONS
 - A. Up to and including 2-inch pipe size: Screwed pattern, bronze-to- bronze seat.

- B. Above 2-inch pipe size: Flanged pattern, A.S.A. forged steel, with gaskets, bolts and nuts.
- C. Copper tubing unions shall have sweated type ends. Flanged unions on copper tubing may be soldered connections.
- D. Materials and pressure ratings shall be the same as specified for the respective pipe and fitting system unless otherwise specified.

2.10 MOTORS

- A. All single phase and polyphase motors shall be manufactured to incorporate the later standards.
- B. All single phase and polyphase motors shall have steel frames with ball bearings and copper windings. All motors to have a Class "F" insulation system with a service factor of 1.15.
- C. All motors shall be 1725 RPM, 4 pole design, unless otherwise noted as the drawings, or in the equipment specifications.
- D. Motors installed indoors and not exposed to moisture shall be open, dipproof, Class B temperature rise based on 40 deg. C maximum ambient temperature.
- E. Motors installed outdoors and exposed to moisture shall be otally enclosed, fan cooled, Class B temperature rise based on 40 deg. C maximum ambient temperature.
- F. Based on NEMA Standards, motors shall comply with the following minimum nominal efficiencies at full load.

| Nominal Efficiencies for "NEMA Fremium TM" Induction Motors | | | | | | | | |
|---|--|--------------|----------|-----------------------------|----------|----------|--|--|
| | Rated 600 Volus of Less (Random Wound) | | | | | | | |
| | 0 | pen Drip Pro | 0f | Totally Enclosed Fan-Cooled | | | | |
| HP | 3500 RPM | 1869 K.M | 1200 RPM | 3500 RPM | 1800 RPM | 1200 RPM | | |
| 1 | 82.5 | 85.5 | 77.0 | 82.5 | 85.5 | 77.0 | | |
| 1.5 | 800 | 86.5 | 84.0 | 87.5 | 86.5 | 84.0 | | |
| 2 | 87 | 86.5 | 85.5 | 88.5 | 86.5 | 85.5 | | |
| 3 | \$5 | 89.5 | 85.5 | 89.5 | 89.5 | 86.5 | | |
| 5 | 89.5 | 89.5 | 86.5 | 89.5 | 89.5 | 88.5 | | |
| 7.5 | 90.2 | 91.0 | 88.5 | 91.0 | 91.7 | 89.5 | | |
| 10 | 91.7 | 91.7 | 89.5 | 91.0 | 91.7 | 90.2 | | |
| .5 | 91.7 | 93.0 | 90.2 | 91.7 | 92.4 | 91.0 | | |
| 20 | 92.4 | 93.0 | 91.0 | 91.7 | 93.0 | 91.0 | | |
| 25 | 93.0 | 93.6 | 91.7 | 93.0 | 93.6 | 91.7 | | |
| 30 | 93.6 | 94.1 | 91.7 | 93.0 | 93.6 | 91.7 | | |
| 40 | 94.1 | 94.1 | 92.4 | 94.1 | 94.1 | 92.4 | | |

Motor Characteristics: Refer to Equipment Schedules for specific data.

120/208 Volt System: Motors 1/2HP & Larger - 208V, 3 Phase, 3 Wire

Motors Less than 1/2HP- 120V, 1 Phase, 2 Wire

- H. All motors rated less than 1/2HP shall have thermal protection of the auto-reset type as an integral part of the motor.
- I. All motors rated 1/2HP and larger shall have thermal protection provided by an external device.
- J. Whenever a variable frequency PWM drive is installed to control an AC motor, a maintenance-free, circumferential, conductive micro fiber shaft grounding ring shall be installed on the AC motor drive end to discharge shaft currents to ground. Recommended part: AEGIS SGR[™] Bearing Protection Ring, as made by Electro Static Technology. Install in accordance with the manufacturer's written instructions.

PART 3 – EXECUTION

3.1 PIPING SYSTEMS

- A. All piping to drain to low points. Low points shall be provided with drain alves with hose thread.
- B. All piping shall be arranged to have air vents at high points.
 - 1. Air vents shall be automatic in operation when located in Boile Rooms, Chiller Rooms and Mechanical Equipment Rooms. All air vents shall be provided with a PVC drain line which shall be routed to the nearest floor drain. Several air vents may be tied together.
 - 2. Air vents shall be manual in operation in all other location
 - 3. Air vents on steam shall be piped down to floor with opper drain line.
- C. Do not install trapped lines where water cannot be drained or air can accumulate without being vented.
- D. Piping shall run square with building lines.
- E. Piping shall not be insulated or cover d until tested and until building is enclosed.
- F. Necessary drains, off-sets vent and drips shall be provided for coordination of the work as part of the contract.
- G. Running or close ripples are not permitted.
- H. Piping shall not be instanded over electrical transformers, panels, switchgear, substations, and control panels. No piping shall be installed in elevator machine rooms.
- I. Expored instant piping risers in unfinished spaces shall be covered with 22 gauge galvanized steel sleeves from floor to ceiling. Refer to Section: Insulation & Covering HVAC for additional equirements.

Ahow clearance for expansion and contraction.

Install eccentric piping fittings where change in sizes occurs in piping systems. Tops of pipes shall remain level for hydronic systems. Bottom of pipe shall remain level for steam systems.

- L. Install isolating fittings between sections of ferrous and non-ferrous pipe or connected equipment.
- M. Do not support piping from other piping, conduits or equipment.
- N. Strainers shall be installed on suction of all pumps, inlets of control valves, and where indicated on drawings.

- O. Thermometers and gauges shall be installed where indicated on the drawings, required by equipment specifications and where indicated elsewhere in the specifications.
- P. Flexible connectors shall be provided on suction and discharge piping of all base mounted pumps.
- Q. Unions shall be provided adjacent to all valves, at equipment connections, and where necessary to facilitate dismantling of the piping system.

PUT

- R. Material Requirements for Systems:
 - 1. Heating Hot Water Supply & Return Piping:
 - a. Schedule 40 black steel.
 - b. Type L hard copper.
 - 2. Chilled Water Supply & Return Piping:
 - a. Schedule 40 black steel.
 - b. Type L hard copper.
 - 3. Condenser Water Supply & Return Piping:
 - a. Schedule 40 black steel.
 - 4. Make-up Water: Type L hard copper.
 - 5. Dual Temperature Water Supply & Returned
 - a. Schedule 40 black steel.
 - b. Type L hard copper.
 - 6. AC Condensate Drain (including) umped condensate):
 - a. Type DWV copyer
 - b. Schedule 40 PVC.
 - 7. Refrigerant Piping: Type ACR hard copper.
- 3.2 TAGS, CHART, AND IDENTIFICATION
 - A. See Paragraph. "Labeling" in GENERAL PROVISIONS for equipment labeling.
 - B. Identify ach valve in all systems with black, numbered and stamped 1- 1/2" brass or aluminum tags isstened to valve by brass chain and S-hook.
 - Provide 1/8" scale diagrams showing location, number and service or function of each tagged item.
 - 1. Frame diagrams in approved metal frames with clear acrylic front, hinges, and locks.
 - 2. Secure to wall in Mechanical Room.
 - 3. Provide two additional separate copies permanently covered and bound.
 - a. Include one (1) copy in the Operation and Maintenance Manuals.

D. Piping Identification: Identify piping with Seton "Setmark" or Brimar, semi-rigid plastic, wraparound pipe markers with flow arrows and conforming to ANSI A13.1. Locate marker at each valve, changes in direction, where pipes pass thru barriers and every 25' of horizontal runs. Lettering on background shall be in accordance with the following colors:

| | Legend | Background | Lettering | |
|-----|-------------------------------|------------|-----------|---|
| 1. | Chilled Water Supply | - Green | - White | |
| 2. | Chilled Water Return | - Green | - White | |
| 3. | Cooling Tower Water Supply | - Green | - White | |
| 4. | Cooling Tower Water Return | - Green | - White | |
| 5. | Gas Vent, Refrigerant Vent | - Yellow | - Black | |
| 6. | Heating Water Supply | - Yellow | - Black | |
| 7. | Heating Water Return | - Yellow | - Plick | |
| 8. | Dual Temperature Water Supply | - Yellow | Blyck | |
| 9. | Dual Temperature Water Return | - Yellow | - Black | |
| 10. | Refrigerant Liquid | - Yellow | Dlack | |
| 11. | Refrigerant Gas | - Yellow | - Black |] |

E. Provide color coded 1" diameter markers on ceiling tile grid to indice e system and valve locations.

Hot Water: - Red

Dual Temperature Water: - Red

- F. Manufacturers: Seton "Setmark", Brimar, B-Line M.
- 3.3 WELDING
 - A. All concealed and inaccessible black steel piping shall be welded.
 - B. All black steel piping larger than 1-1N inch may be fusion welded.
 - C. All elbows, tees and branch coune tions shall be made with welding fittings ANSI B16.9.
 - D. Welding shall be in accordance with the ASME Boiler and Pressure Vessel Code Section IX.
 - E. Furnish welder test certificate for review. Certificates of successful qualification by the following organizations shall be acceptable.
 - 1. ASMI Boile and Pressure Vessel Code
 - 2. XNST Code for Pressure Piping
 - . National Certified Pipe Welding Bureau

4. Military Specification MIL-STD-248

SOLDERING/BRAZING

- Connections between copper tubing and copper fittings shall be made with the appropriate filler metal. Flux shall be non-corrosive type as recommended by the manufacturer of the filler metal, and conforming to AWS A5.8.
- B. Tubing shall be cut square and then reamed and deburred. End of tubing and inside of fitting cup shall be cleaned with steel wool and the flux shall be applied to the clean surface before joining. After joining, the excess filler metal shall be wiped off while still plastic.

- C. Silver brazing alloy shall be equal to Easy-Flo by Handy and Harmon or Sta-Brite silver solder and shall be used for joints in:
 - 1. Hot water heating piping
 - 2. Chilled water piping
 - 3. Air conditioning condensate drain piping
 - 4. Dual temperature water piping
 - 5. Cold water fill and make-up piping
 - 6. Condenser water piping
- D. Where the silver brazing is performed in a confined non-ventilated space, a non-pxic, admium-free brazing alloy such as braze 560 by Handy & Harman shall be used.
- E. Refrigerant piping shall be silver brazed using Harris Sil-Fos 15 or equivalent, with nitrogen purge.
- F. Bring joint to solder temperature or brazing temperature in as short a time as possible.
- G. Form continuous solder bead or brazing filler bead around entile circumference of joint.
- H. Wipe excess solder from joint area while solder is still plastic.

Reput

END OF SECTION 2302



SECTION 230215

VALVES

PART 1 – GENERAL

1.1 RELATED DOCUMENTS

- A. The general provisions of the Contract, including the Conditions of the Contract (General, Supplementary and other conditions, if any) and Division 1 as appropriate, apply to the work specified in this section.
- B. Refer to Section 230200 for HVAC General Provisions.
- C. Refer to other sections in Division 23 for materials and methods not specified hele

1.2 DESCRIPTION OF WORK

- A. This Section includes the following:
 - 1. General
 - 2. Condenser Water and Chilled Water Systems
 - 3. Dual Temp and Hot Water Heating Systems
 - 4. Refrigerant Valves and Specialties

1.3 QUALITY ASSURANCE

- A. Provide adequate supervision of labor force to assure that all aspects of the specifications are being fulfilled.
- B. Verify that all work and equipment is installed in accordance with manufacturer's warranty requirements.

PART 2 – PRODUCTS

- 2.1 GENERAL
 - A. All gate and globe valves shall be designed for repacking under pressure when fully opened, and shall be equipped with packing suitable for the intended service. When the valve is fully opened, the back seat shall protect the packing and the stem threads from the fluid. All gate and globe valves shall have a gland follower. The pressure- temperature rating of valves shall be not less than the design criteria applicable to all components of the system.
 - B. Assofar as possible, all valves of the same type shall be of the same manufacture.
 - Valves installed above 7 ft. in Mechanical Rooms shall have chain operators.
 - All valves shall be provided with stem extensions. Valve handle shall be clear of insulation jacket.
 - E. Manufacturers:
 - Stockham
 - Milwaukee
 - Hammond

Apollo

Watts

Walworth

Nibco

2.2

A.

| Jamesbury | | | | | | | | |
|--|------------------------------------|-------------------------------------|--|--|-----------------------------------|-------------------------------|------------------------------|------------------|
| CONDENSER | WATE | R AND (| CHILLED WA | ATER SYS | TEMS | | | |
| Gate Valves - 2 | " and sn | naller: | | | | | |) |
| Valves 2" ar compositior Teflon- imp | nd small n, thread pregnated | er shall b ed or sol d packin | be of Class 12: lder ends, solid g and malleab | 5, body and d disc, copj le handwh | bonnet sha per-silicon eel. | ll be of AST alloy stem, t | 11-B-62 cast rass packing | bronze gland, |
| Recommend | ded Valv | ves: | | | | $\langle \mathcal{L} \rangle$ | | |
| Threaded: | | | Solder: | | | | | |
| Stockham B | 8-100 (R | S) | Stockhar | n B-108 (R | | | | |
| or | | or | | | X | | | |
| Stockham B | 8-103 (N | IRS) | Stockhar | n B-104 (N | J.S) | | | |
| | RS | NRS | | - RSC | NRS | | | |
| Hammond | IB690 | IB609 | Nibco | S 111 | S113 | | | |
| Milwaukee | 148 | 105 | Hammond | • IB691 | IB613 | | | |

Class 150 valves meeting the above specifications may be used where pressure requires: Stockham B-120 threaded 5 or B-124 solder end RS, union bonnet. ena

B. Gate Valves - 2-1/2" and

> Valves 2-1/2 and larger shall be Class 125 iron body, bronze mounted, with body and bonnet STM A-126 Class B cast iron, flanged ends with Teflon-impregnated packing conforming and tw e picking gland assembly. p-pie

ed valves: com en Stockham G-623 OS&Y RS Stockham G-612 NRS OS&Y NRS Hammond IR1140 Nibco F617-0 F639-31

Ball Valves - 3" and Smaller: C.

> Valves 3" and smaller shall be 600 psi CWP, have cast brass bodies, replaceable reinforced Teflon seats, conventional port, blowout proof stems, chrome plated brass ball, and threaded or solder

ends. Provide extended valve handle to accommodate up to 2" of insulation with non-thermal conductive material, insulation plug, cap and protective sleeve.

Recommended valves:

Threaded:

Solder:

Stockham S-216-BR-RT Stockham S-216-BR-RS

NIBCO T-580-70BR NIBCO S-580-70BR-R

Jamesbury II 1100TT

Apollo 70-100 Apollo 70-200

Inline 334

Alternative is Stockham S-217-BR-RT (threaded).

Drain valves, ¹/₂" or ³/₄" shall be 600 psi CWP, with stainless steel true, cast bronze body, 2-piece with cap and chain, full port stainless steel ball and stem, RTFE ball seat, threaded or soldered inlet connection, cap rated for 150 psi.

Recommended valve:

Stockham S-285-BR-R-66-HC.

D. Globe Valves - 2" and Smaller:

Valves 2" and smaller shall be of Chass 25, body and bonnet of ASTM B-62 cast bronze composition, threaded or soldered and copper silicon alloy stem, brass packing gland, Teflon-impregnated packing, and malle ble handwheel.

Recommended valves:

Threaded:

(Teflon Dis

Stockh am I

Nilvo T211B

Solder:

Stockham B-13T

Stockham B-14T

(Teflon Disc)

Stockham B-17

(Bronze Disc)

Jenkins 1200

Nibco S211Y

Hammond IB440

20

Class 150 valves meeting the above specifications may be used where pressure requires:

Stockham B-22 (threaded-Teflon Disc)

Stockham B-29 (threaded-Stainless Steel Disc)

Stockham B-24 (solder-Teflon Disc)

E. Globe Valves - 2-1/2" and Larger:

Valves 2-1/2" and larger shall be Class 125 iron body, bronze mounted with body and bonnet

conforming to ASTM A-126 Class B cast iron, flanged ends, with Teflon-impregnated packing and two-piece packing gland assembly.

Recommended valves:

Stockham G-512 (Bronze disc)

Stockham G-514T (Teflon disc)

Bronze Disc: Comp. Disc.:

Hammond IR116 ----Nibco F718B ----

F. Butterfly Valves - 2-1/2" and Larger: (Chilled Water)

Valves 2-1/2" and larger shall be high performance, bubble-tight, lag water type body, 200 psi CWP, conforming to ASTM A-126 Class B cast iron, drilled and topped, fold replaceable EPDM sleeve, with ductile nickel-plated disc, 410 stainless steel stem, and EPDM O-ring stem seals. Sizes 2 - 6" shall be lever operated and 8 - 24" shall have get operators.

Recommended valves:

| Stockham LG-712-DS3-E | Lever operated |
|-----------------------|---------------------|
| Stockham LG-722-DS3-E | Gear operated |
| Lever: | Gear: |
| Demco NE-150-5215351 | ME-150-5215359-2097 |
| Norris R3020-13SS-1F | R3017-13SS-2K |
| Keystone Fig. 129 | 129 |
| Center Line Series LT | Series LT |
| Grinnell LC8201-1 | LC8202-1 |

Alternative L prabove is Stockham LD-712-DS3-E and LD-722-DS3-E ductile iron body butterfly valves (AS14) -395 Ductile).

VOTE: For dead end service, butterfly valves require flanges both upstream and downstream for non-

Butterfly Valves - 2-1/2" and Larger: (Condenser Water)

Ditto Chilled Water except with aluminum bronze discs.

For dead end service or requiring additional body strength:

Stockham LD-712-BS3-ELever OperatedStockham LD-722-BS3-EGear Operated

Demco NE-150-5214351 NE-150-5214359-2098

Gear:

Lever:

| Norris R3010-43SS-1F | R3010-43SS-2K |
|-----------------------|---------------|
| Keystone 129 | 129 |
| Center Line series LT | Series LT |
| Grinnell LC8201-1 | LC8202-13 |

Note: For dead end service, butterfly valves require flanges both upstream and downstream for proper shutoff and retention.

H. Check Valves - 2" and Smaller:

Valves 2" and smaller shall be of Class 125, threaded or solder ends, body and cips shall be ASTM B-62 cast bronze composition, swing type disc.

Recommended valves:

Threaded:

Solder:

| Stockham B-319 | Stockham B-309 |
|----------------|----------------|
|----------------|----------------|

Hammond IB490 IB941

Milwaukee 509 1509

If composition disc is preferred, specify Stockham B/320B - threaded end, or B-310B - solder end, for Class 125 service.

Class 150 valves meeting the above specifications may be used where pressure requires: Stockham B-321 threaded.

I. Check Valves - 2-1/2 and Larger

Valves 2-1/2" and larger shall be from body, bronze mounted with body and cap conforming to ASTM A-126 Class Prast from, flanged ends, swing type disc.

Recommended valves

Stockham C-931

Hammond R124

<u>OR</u>



Alternative Check Valves (2-1/2" and larger) shall be Class 125/250, iron body, bronze mounted, Wafer Check Valves, with ends designed for flanged type connection, aluminum bronze disc, EPDM seats, 316 stainless steel torsion spring, and hinge pin.

Recommended valves:

Stockham WG-971

Center Line CLC

Mission K12 HMP

Tetra Tech 200-15704-17001

Marlin A125 HZDSF

2.3 DUAL TEMP AND HOT WATER HEATING SYSTEM

A. Gate Valves - 2" and smaller:

Valves 2" and smaller shall be of Class 150 with body and union bonnet of ASTM B-62 carboronze composition, threaded or solder ends, solid disc, copper-silicon stem, brass packing glast Teflon- impregnated packing, and malleable handwheel.

Recommended valves:

| Threaded: | Solder: | \sim |
|------------------------------|------------------------------|-----------------------------|
| Stockham B-120 (RS) | Stockham B-124 | \sim |
| Stockham B-130 (RS) | | |
| Hammond IB629 | IB648 | |
| Nibco T134 | S134 | |
| Class 200 valves meeting the | above specifications have be | d where pressure requires - |

Class 200 valves meeting the above specifications may be used where pressure requires - Stockham B-132 (threaded - RS).

B. Ball Valves - 3" and smaller:

Valves 3" and smaller shall be 600 psi CWP, have east brass bodies, replaceable reinforced Teflon seats, conventional port, blowout proof seans, brome plated brass ball, and threaded or solder ends with extended solder cups. Provide extended valve handle to accommodate up to 2" of insulation with non-thermal conductive material, insulation plug, cap and protective sleeve.

| Recommended valves: | \mathbf{N} |
|----------------------|----------------------|
| Threaded: | solder: |
| Stockham S-216-BI-P1 | Stockham S-216-BR-RS |
| Worcester 41/2RT | |
| Jamesbury II (100TT | |
| Apollo 70-100 | Apollo 70-200 |
| Nb-0 T580-70BR | S580-70BR-R |
| Inline 334 | |

Drain valves, ¹/₂" or ³/₄", shall be 600 psi CWP, with stainless steel trim, cast bronze body, 2-piece with cap and chain, full port stainless-steel ball and stem, RTFE ball seat, threaded or soldered inlet connection, cap rated for 150 psi.

Recommended valve:

Stockham S-285-BR-R-66-HC

C. Gate Valves - 2-1/2" and larger:

Valves 2-1/2" and larger shall be Class 125 iron body, bronze mounted, with body and bonnet

conforming to ASTM A-126 Class B cast iron, flanged ends, with Teflon-impregnated packing and two-piece packing gland assembly.

| | Recommended valves: | | | |
|----|--|---|--|--|
| | Stockham G-623 (OS&Y | () RS | | |
| | Stockham G-612 (NRS) | | | ,5 |
| | | OS&Y | NRS | |
| | Nibco | F618-0 | F639-31 | C |
| | Hammond IR1140 | | | \sim |
| D. | Globe Valves - 2" and smalle | er: | | \sim |
| | Valves 2" and smaller sh copper-silicon alloy ster handwheel. | all be of Class m, brass packi | 150 with body and union boar ng gland, Teflon-impregrat | net of ASTM B-62 bronze, a pecking and malleable |
| | Recommended valves: | | \sim | |
| | Threaded: | Solder | : X | |
| | Stockham B-22 (Teflon | Disc) Stockl | nam B-74 (TAlon Disc) | |
| | Stockham B-29 (Stainles | ss trim) | | |
| | Comp. Disc.: | S.S. Trink | 7 | |
| | Nibco T-235-Y | Mil vauke | 591A | |
| | Class 200 valves meeting | g the above keq | uirements may be used where | e pressure required: |
| | Stockham B-32 (Teflon | di.c) | | |
| | Stockham B-62 (Staples | ss rim) | | |
| E. | Globe Valves - 2 1/2' and L | arger: | | |
| | Valves 2.1/2 and large conforming th ASTM A- and two niese packing g | er shall be Cla 126 Class B ca land assembly. | ss 125 body, bronze mounte st iron, flanged ends, with Te | ed, with body and bonnet flon - impregnated packing |
| | Recommended valves: | | | |
| | Stockham G-512 (bronze | e disc) | | |
| | Stockham G-514 (Teflor | n disc) | | |
| | Hammond IR 116 | | | |
| | Nibco F7181B | | | |
| F. | Butterfly Valves - 2-1/2" and | l Larger: | | |
| | Valves 2-1/2" and larger CWP, conforming to AS | shall be high j STM A-126 C | berformance, bubble-tight, lugass B cast iron, drilled and t | g-wafer type body, 200 psi apped, replaceable EPDM |

Tetra Tech 200-15704-17001

sleeve, with ductile nickel-plated disc, 410 stainless steel stem, and EPDM O-ring stem seals. Sizes 2 - 6" shall have lever operators and 8 - 24" shall have gear operators.

Recommended valves:

| | Stockham LG-712-DS3-E | Lever operated |
|----|--|--|
| | Stockham LG-722-DS3-E | Gear operated |
| | Lever: | Gear: |
| | Demco NE-150-5215351 | NE-150-5215359-2097 |
| | Norris R3010-13SS-1F | R3010-13SS-2K |
| | Keystone Fig. 129 | 129 |
| | Center Line Series LT | Series LT |
| | Grinnell LC8201-1 | LC 8202-1 |
| | TREATED SYSTEM: | |
| | Recommended Valves: | \sim |
| | Stockham LG-712-BS3-E | Lever operate |
| | Stockham LG-722-BS3-E | Gear operated |
| | Alternative for above is Stockh ductile iron. | am Dectro Iron Butterfly Valves conforming to ASTM-A-395 |
| | Ductile Valves: | \mathbf{X} |
| | Stockham LD-512-DS3-E & LI | D-112-BS3-E |
| | (lever operated-wefer | body) |
| | Stockham LD-522-DS3-F or LI | D-512-BS3-E |
| | (gear operated-wafer | body) |
| | Stockham LD-7-2-DS3-E or LI | D-712-BS3-E |
| | (lever operated-lug b | ody) |
| | Stoctham LD-722-DS3-E or LI | D-712-BS3-E |
| | (gear operated-lug bo | dy) |
| | • Note: Butterfly valves in dead proper shutoff and retention. | end service require both upstream and downstream flanges for |
| Ch | eck Valves - 2" and smaller: | |

Valves 2" and smaller shall be Class 150 with bodies and caps of ASTM B-62 bronze composition and threaded ends. Class 150 valves shall have lift-type Buna-N-disc and union caps, and are to be used in lines with globe valves.

Recommended valves:

Stockham B-322-B

Hammond IB948

Milwaukee 510

For backflow prevention in lines with gate valves, Y-pattern valves with swing-type disc are recommended.

For Class 150 Service, threaded ends:

Stockham B-321

For Class 200 Service, threaded ends:

Stockham B-345

Hammond IB949

Nibco T453-BY

H. Check Valves - 2-1/2" and Larger:

Valves 2" and larger shall be iron body, bronze mounted, with body and cap conforming to ASTM A-126 Class B cast iron, flanged ends, and swing-type disc.

Recommended valves:

Stockham G-931

Hammond IR1124

Nibco F918-B

Alternative for the above listed check valves shall be Class 125/250 iron body, bronze mounted, Wafer Check Valve, with ends designed for flanged type connection, aluminum bronze disc, EPDM seats 3 to stainless steel torsion spring, and hinge pin.

OR

Recommended valves:

ocham WG-971

Mission K12 HMP

Center Line CLC Series

Marlin A125 HZDSF

REFRIGERANT VALVES & SPECIALTIES

- A. Service Valves:
 - 1. Globe Shutoff Valves: Forged brass, packed, back seating, winged seal cap, 300 degrees F (149 degrees C) temperature rating, 500 psi working pressure.

Tetra Tech 200-15704-17001

- 2. Check Valves: Forged brass, accessible internal parts, soft synthetic seat, fully guided brass piston and stainless-steel spring, 250 degrees F (121 degrees C) temperature rating, 500 psi working pressure.
- 3. Manufacturers:

Henry Valve Co.

Parker Hannifin Corp., Refrigeration & Air-Conditioning

Sporlan Valve Co.

- B. Solenoid Valves:
 - 2-way Solenoid Valves: Forged brass, designed to conform to ARI 760, pornally closed, teflon valve seat, NEMA 1 solenoid enclosure, 24-volt, 60 Hz., UL-listed, 12" conduit adapter, 250 degrees F (121 degrees C) temperature rating, 400 psi working pressure.
 - 2. Manufacturers:

Alco Controls Div., Emerson Electric Co.

Automatic Switch Co.

Sporland Valve Co.

- C. Specialties:
 - 1. Refrigerant Strainers: Brass shell and en t connections, brazed joints, monel screen, 100 mesh, UL listed, 350 psi working pressure.
 - 2. Moisture-Liquid Indicators: Forged briss, single port, removable cap, polished optical glass, solder connections, UL lister, 200 degrees F (93 degrees C) temperature rating, 500 psi working pressure.
 - 3. Refrigerant Filter-Duery: Steel shell, ceramic fired desiccant core, solder connections, UL listed, 500 psi working pressure.
 - 4. Evaporator Dressure Regulators: Provide corrosion-resistant, spring loaded, stainless steel springs, pressure openeed, evaporator pressure regulator, in size and working pressure indicated, with copper connections.
 - 5. Rariger at Discharge Line Mufflers: Provide discharge line mufflers as recommended by equipment manufacturer for use in service indicated, UL listed.
 - Manufacturers:

Alco Controls Div., Emerson Electric Co.

Henry Valve Co.

Parker-Hannifin corp., Refrigeration & Air Conditioning Div.

Sporlan Valve Co.

PART 3 – EXECUTION

3.1 PIPING SYSTEMS

- A. All piping to drain to low points. Low points shall be provided with drain valves with hose thread.
- B. Valve body construction shall match piping system material.
- C. Install isolating fittings between sections of ferrous and non-ferrous pipe or connected equip te
- D. Valves shall be installed with stems above horizontal.
- E. Valves shall be installed on all sides of equipment and control valves to allow isolation for repair.
- F. Unions shall be provided adjacent to all valves, at equipment connections, and where recessary to facilitate dismantling of the piping system.
- 3.2 TAGS, CHARTS AND IDENTIFICATION
 - A. Identify each valve in all systems in accordance with requirements of Section 230210.

RBIDDIR

END OF SECTION 230215

SECTION 230230

INSULATION & COVERING – HVAC

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. The general provisions of the contract, including the conditions of the contract (General, Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the work specified in this section.
- B. Refer to Section 230200 for HVAC General Provisions
- C. Refer to Section 230210 for HVAC Basic Materials & Methods.

1.2 DESCRIPTION OF WORK

- A. This Section includes insulation and covering provided on the following piping and equipment:
 - 1. Cold Water Make-Up Piping
 - 2. Hot Water Heating Piping
 - 3. Chilled Water Piping
 - 4. Condensate Drain Lines
 - 5. Dual temperature water piping.
 - 6. Refrigerant Piping.
 - 7. Cold Equipment Surface
 - 8. Hot equipment surfaces
 - 9. Exterior Piping
 - 10. Acoustic Duct Liner
 - 11. Reusable V lve Covers
 - 12. Insulated Pite Saddles
- B. Insulation shall be installed on the following duct systems:
 - All supply ductwork.
 - All return ductwork.
 - B. All outside air intake and relief ductwork.
 - 4. All ductwork connected to energy recovery units.
- 1.3 REFERENCE STANDARDS
 - A. Refer to Section 230200 for a general description of requirements applying to this section.

ADDITION AND RENOVATIONS

1.4 QUALITY ASSURANCE

- A. Refer to Section 230210 for a general description of requirements applying to this section.
- B. Install insulation in accordance with manufacturer's recommendations.
- C. Provide adequate supervision of labor force to assure that all aspects of the specifications are being fulfilled.
- 1.5 SUBMITTALS
 - A. Submit shop drawings, installation instructions, and manufacturer's literature of all material specified in accordance with Section 230200.
 - B. Submit fabrication instructions for pipe fitting and valve insulation.
 - C. Submit manufacturer's joining recommendations for butt joints and longitudinal seams.
- 1.6 WARRANTY/GUARANTEE
 - A. All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, General Requirements.

PART 2 – PRODUCTS

- 2.1 PIPE INSULATION MATERIAL
 - A. Fiberglass:
 - 1. Material: Preformed fiberglass bonded with resin to form circular pipe sleeves with factory applied, white all service jacket boated to einforced foil vapor barrier jacketing. The jacket shall have factory applied double pressure-sensitive, self-sealing, adhesive closure and vapor sealing of longitudinal joints. Thermal conclucit inty: 0.24 Btu/Hr./SF/inch at 100 degrees F. Flame spread of 25 and developed smake of 50 pr less.
 - 2. All Valves and Fittings
 - a. Glass fiber insert and pre-molded PVC cover, Proto Corp., Johns Manville Corp. "Zeston" and "Hi 20 temp Inserts" for fittings. Glass fiber or prefabricated elastomeric foam fittings must fill the entire space within the cover completely.
 - b. Factory molded fibrous glass fitting covering for fittings. Coat ends with Fosters 30-36 laghet dhesive
 - c. Vitered sections of pipe covering for valves.

Manufacturers: Johns Manville Corp., Certain-Teed, Owens- Corning, Knauf, Armacell.

Closed Cell:

- 1. Material: Black flexible elastomeric foamed closed cell structure insulation 25/50 rated with a flame spread rating of 25 or less and a smoke developed rating of 50 or less with both a moisture seal and a reinforced elastic foam lap seal closure system.
- 2. Flexible pipe insulation shall be a foamed elastomeric closed cell structure material, with a thermal conductivity of not more than 0.27 Btu/Hr./Sq. Ft./Inch at a mean temperature of 75 degrees F. The insulation shall have an average density of at least 2 pounds per cubic foot, shall

be self-extinguishing, and shall have a water vapor transmission rating of not more than 0.1 perms. Between temperature limits of -40 degrees F and plus 220 degrees F, the insulation shall not indicate any deviation from its original state.

3. Specification Compliance:

ASTM-E-84

ASTM-C-534 Type I – Tubular, Type II – Sheet.

ASTM-D-1056, 2B1 – Tubular, Sheet.

MIL-C-3133B (MIL STD 670B) Grade SBE-3

MIL-P-15S280J, Form T, Form S.

- 4. Manufacturers: Armacell, Nomaco, K-Flex, Aeroflex USA, Inc.
- C. Covering of Pipe Insulation Outdoors:
 - 1. Wrapping: Wrap insulation with embossed 0.016" aluminum jack
 - 2. Fastenings: Cover shall be held in place with soft aluranu h bands on 12" centers.
 - 3. Valves and Fittings: Weatherproof all valves and fittings.
- D. Manufacturers: Johns Manville Corp., Certain-Terd, Owens- Corning, Knauf.
- 2.2 DUCT INSULATION
 - A. Concealed Supply, Return, Relief, and Outside Air Ductwork, and all ductwork connected to energy recovery units: Fiberglass duct wrag bonded with resins, 3/4 pound density, aluminum foil facing reinforced with fiberglass scrim, laminted to Kraft, 2" thick.
 - 1. Thermal Conductivity: 0.21 Btu/Ir./SF/Inch at 75 degrees F. Min. installed "R" value w/25% compression shall be 56.
 - 2. Duct wrap shall be out to stretch-out dimensions as provided in manufacturer's instructions. Remove a 2" proce of insulation from the facing at the end of the piece of insulation to form an overlapping statement tape flap. Install with facing outside so tape flap overlaps insulation and facing a objected. Insulation shall be tightly butted and not compressed excessively at duct corners. Seans shall be stapled 6" on center with outward clinching staples. All seams, tears, punctures and other penetrations of the insulation facing shall be sealed with foil tape or vapor proof mastic. Where rectangular ducts are 24" in width or greater, duct wrap shall be secured to the buttom of the duct with mechanical fasteners; i.e., stick pins spaced 18" on center.

Exposed supply, return, relief, and outside air ductwork, and all ductwork connected to energy recovery units, shall be insulated in finished conditioned spaces, penthouse, mechanical rooms, mezzanine areas, equipment closets, and non-conditioned spaces with 2" thick rigid fiberglass board. Insulation shall be 6 P.C.F. density with a "K" value of 0.25 Btu/Hr./SF/Inch at 75 degrees F. mean temperature and shall be U.L. listed at 25 maximum for flame spread, and 50 maximum for smoke developed. Insulation shall be applied using Graham Pins or Stik-Clips and all seams, edges and breaks shall be sealed with 4" matching tape and sealed with Vicryl CP-10 to match ASJ jacket. Insulation shall be provided with all-service jacket facing.

C. Manufacturers: Johns Manville Corp., Certain-Teed or Owens- Corning, Knauf.

2.3 ACOUSTIC DUCT LINER

- A. Duct liner shall be designed for use as an acoustical insulation to absorb air conditioning noise in sheet metal ducts and plenums operating at velocities up to 6000 fpm and temperatures up to 250 deg. F.
- B. Duct liner shall be a bonded mat of glass fibers coated with an EPA registered biocide and a black pigmented fire-resistant coating on the air stream side or flexible elastomeric closed cell foam made with an EPA approved anti-microbial.
- C. Duct liner shall comply with the requirements of NFPA 90A and 90B. Surface burning character shall comply with UL Standard 723 for 25/50 flame and smoke development.
- D. Duct liner shall comply with the property requirements of ASTM Specification C101, Type 1, or ASTM C1534. Material shall resist fungal and bacterial growth when subjected to ASTM G21 and G22 test methods.
- E. Material thickness, name of manufacturer and type shall be printed on the air stream side of the liner for ease of identification.
- F. Duct liner shall be 2" thick, unless otherwise noted on the drawings.
- G. Manufacturers: Owens Corning QuietR® AcousticR[™] Fact Juner, Certainteed, Evonik Industries Solcoustic, Johns Manville Linacoustic® RC, Armacell.
- 2.4 REUSABLE VALVE COVERS
 - A. All valves, strainers, combination valves, etc. in chilled water and heating hot water systems shall be insulated with a factory fabricated removable and reasable cover. (This product shall not be used for pipe and fittings.)
 - B. Insulation shall be either fiberglass blanked or flexible elastomeric thermal insulation as listed in Paragraph 3.2 of this specification, or predioricated fitting from the supplier. Flame and smoke spread shall be 25/50 per ASTM 84
 - C. Outer jacket shall be made of material equal to Tychem QC, overlap and completely cover the insulation, with seams joined b) tabs made from Velcro or fabric straps per manufacturer's standards.
 - D. Outer jacket shall overlap adjoining sections of pipe insulation, and shall be non-combustible, impermeable to water, and prevent mold, mildew and condensation.
 - E. Installation shall not require the use of any special hand tools.
 - F. Manufacturers: Corick Valve Covers, NoSweat Valve Wraps.

2.5 INSULATED PIPE SADDLES

Institution and facing shall each meet 25/50 flame and smoke ratings per ASTM E-84 on a component basis.

A section of rigid insulation shall be used at all cold pipe hangers or support locations and shall consist of:

- 1. A rigid 3.75 PCF phenolic foam pipe insulation designed to support pipe sizes up to and including 6" iron pipe size.
- 2. A rigid 5 PCF phenolic foam pipe insulation designed to support pipe sizes from 8" to 30" iron pipe size.

- 3. For all hot pipe hanger or support locations, the insert material shall be either rigid calcium silicate per ASTM C303 or perlite silicate per ASTM C303 with all service jacket and laminated to a steel support saddle.
- C. The insulation jacket shall contain a vapor retarding material to provide low moisture vapor permeability and resistance to mold, mildew and fungus growth.
- D. The insulation shall be free of any CFC or HCFC materials.
- The insulation shall have a minimum K-factor of 0.13 at 75 deg. F mean temperature, and E. lap joint with high performance acrylic pressure sensitive adhesive tape.
- F. Integral insulation saddle shall be made of G-90 carbon steel, with full 180 deg Covra flared edges to protect the vapor barrier jacket and insulation, and short rib surface to center the addle inside the hanger and prevent movement.
- G. Preformed insulation shall extend beyond the saddle by a minimum of accommodate a tape joint seal at the butt edges of adjoining insulation sections.

| Minimum product dimensions shall be as follows: | | | | | |
|---|------------|------------|----------|--------|--|
| Nominal pipe | Insulation | Insulation | Saddle | Sadale | |
| size | density | length | length | gauge | |
| (inches) | (PCF) | (inches) | (inches) | | |
| ¹ / ₂ - 3-1/2 | 3.75 | 9 | 6 | 20 | |
| 4 - 6 | 3.75 | 12 | 9 | 18 | |
| 8 - 18 | 5.0 | 18 | 2 | 16 | |
| 20 - 30 | 5.0 | 24 | 10 | 14 | |
| | | | | | |

- Manufacturer: Tru-Balance insulated sade made by Buckaroos, Inc., Aerofix-U as made by I. Aeroflex USA, Inc.

PART 3 - EXECUTION

H.

- 3.1 **INSTALLATION - GEV**
 - ve been tested and meet requirements. Do not install until system, h A.
 - Do not install u til brilding is enclosed. B.
 - Heavy work which may damage insulation shall have been completed in the vicinity of the insulation C. work.
 - non-compressible insulation saddles at all piping hanger locations, and at all piping hanger D. Provid ocations where piping is insulated with flexible closed cell insulation.
 - Option: Provide insulation coupling system as made by Klo-Shure Co.
 - All installations shall be made by skilled craftsmen regularly engaged in this type of work.
 - Insulation shall be continuous thru-wall, ceiling and floors.
 - G. Metal shields, 16 gauge galvanized, shall be installed between hangers and pipe insulation.
 - H. Pipe, ductwork and equipment shall be clean and dry prior to insulating.
 - I. Install all insulation per manufacturer's instructions.

- J. To avoid undue compression of insulation, provide solid core inserts at all supports as recommended by the insulation manufacturer. Provide insulation shields between the insulation jacket and the hanger.
- K. Ductwork treated with internal acoustic duct liner does not require external insulation.
- L. Apply vapor proof mastic as recommended by the insulation manufacturer on all longitudinal and batt joints of sectional pipe insulation. Apply similar mastic to the end of every third length of sectional pipe insulation on all chilled water and dual temperature pipe insulation to prevent the migration of condensation that might occur.
- M. Provide insulation on all piping, equipment, and fixtures that are part of a factory ascendy pa ckage not otherwise insulated by the manufacture of such packaged equipment. Insulation type and thickenss shall comply with all of the requirements of this section.
- PIPE INSULATION TYPES & THICKNESSES 3.2
 - Provide fiberglass insulation of thickness specified on: Ser A.
 - 1. Cold Water Make-Up:

1/2" for piping 1-1/4" and below.

1" for piping 1-1/2" and over

2. Heating Hot Water: (Up to 140°F)

1" for piping 1-1/4" and below

1-1/2" for pipes 1-1/2" and over.

3. Heating Hot Water: (141°F

1-1/2" for piping 1-1/4"

2" for pipes 1-1/2"

4. Chilled Water:

1/2" for piptog ------ and below. Option: Flexible closed cell insulation

1" for piping 1-1/2" and over

- ual Temperature Water: 5.
 - 1-1/2° for piping 1-1/4° and below. Option: Flexible closed cell insulation

2" for piping 1-1/2" and larger

Refrigerant Piping: Interior locations, exposed and concealed for suction lines and hot gas bypass lines, if applicable. (NOTE: Insulate liquid line if metering device is mounted at the condensing unit.) Option: Flexible closed cell insulation

Suction Line:

1/2" for piping 1-1/4" and below

1" for piping 1-1/2" and larger

INSULATION & COVERING - HVAC 230230-6

Tetra Tech 200-15704-17001

Hot Gas Bypass: (Liquid Line)

1" for piping 1-1/4" and below

1-1/2" for piping 1-1/2" and larger

- 7. Freeze protection of outdoor piping (over heat tracing tape): 3" thick insulation, with metarjacket.
 - a. Plumbing: Cold water make-up to cooling tower.
 - b. Equipment drain piping.
- B. Provide flexible closed cell insulation of thickness specified on:
 - 1. Refrigerant Piping: Exterior Locations for suction lines and hot gas by assilines, if applicable. (NOTE: Insulate liquid line if metering device is mounted at the condensing unit.)

Suction Line:

1/2" for piping 1-1/4" and below

1" for piping 1-1/2" and larger

Hot Gas Bypass: (Liquid Line)

1" for piping 1-1/4" and below

1-1/2" for piping 1-1/2" and larger

2. Cold surfaces of refrigeration equipment 3/4" thickness

- 3. 1" thickness for all water piping within terminal unit cabinets.
- 4. $\frac{1}{2}$ " thickness for condensate drain lines.
- 3.3 PIPE COVERING (FOAM) D. ASTIC TYPE)
 - A. All joints and seams shall be scaled with a compatible adhesive. Approved adhesives are as follows:

Armstrong World Industries No. 520

Benjamin Foster

Company No. 85-75 up to 200 degrees F.

Contractor may use Armstrong Self-Seal Armaflex 2000 insulation in lieu of the above wherever 1/2" is specified.

Fitting covers shall be fabricated from the foamed plastic pipe insulation or from sheet insulation of the identical material. The fabrication shall be in accordance with manufacturer's instructions, and all seams mitered joints shall be joined using the adhesives described hereinbefore.

Pipe insulation in concealed spaces shall require no finish coatings.

D. Pipe insulation in all other areas shall receive two coats of finish of color selected by Architect. Approved finishes are as follows:

Armstrong World Industries WB Armaflex Finish

ADDITION AND RENOVATIONS

3.4 EXTERIOR PIPE COVERING

- A. Wrapping: Wrap insulation with embossed 0.016" aluminum jacket, orient seam down.
- B. Fastenings: Cover shall be held in place with soft aluminum bands on 12" centers.
- C. Valves and Fittings:
 - 1. Weatherproof all valves and fittings.
 - 2. Finish: Apply two coats of vapor resistant mastic reinforced with glass fabric over year

3.5 INTERIOR PIPE COVERING

- A. Provide premolded PVC cover on all interior insulated piping exposed in finished spaces. Orient seams up in overhead piping and toward the wall in vertical runs.
- B. Provide factory molded fitting covering for fittings and accessories, realed and held in place by manufacturer's recommended sealing system.
- C. Provide mitered sections of covering for valves.
- 3.6 ACOUSTIC DUCT LINER
 - A. All portions of duct designated on the drawings to receive duct iner shall be completely covered with duct liner, adhered to the sheet metal with a 100% coverage of adhesive complying with ASTM C916.
 - B. Transverse joints shall be neatly butted and there shall be no interruptions or gaps. All transverse joints and all exposed leading edges shall be pared. The black coated surface of the duct liner shall face the airstream.
 - C. Duct liner shall be secured with mechanical asteners which shall compress the duct liner sufficiently to hold it firmly in place.
 - D. Duct liner shall be cut to assure overlapped and compressed longitudinal joints.
 - E. After installation is complete, blow out the duct system prior to operation to remove any cutting scraps and foreign material remaining in the duct.
- 3.7 INSULATED PPE SIDDLES
 - A. Insulated sipe seddres shall be installed at all hangers, rollers or supports in accordance with manufactuar's written instructions.
 - B. All pipeg shall be clean and free of oil, rust and moisture prior to and during support installation.
 - 2. All insulated saddles and accessories shall be stored in a dry area protected from weather before and during installation

Seal adjoining butt edges of pipe insulation with approved mastic and tape to insure continuity of the insulation jacket and vapor barrier, especially on cold piping system installations.

END OF SECTION 230230
SECTION 230410

HEATING GENERATION AUXILIARY EQUIPMENT

PART 1 – GENERAL

- 1.1 RELATED DOCUMENTS
 - A. The general provisions of the Contract, including the Conditions of the Contract (General, Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the work specified in this section.
 - B. Refer to Section 230200 for HVAC General Provisions.
 - C. Refer to Section 230210 for HVAC Basic Materials and Methods.
 - D. Refer to Section 230400 for Heating Generation Equipment.
- 1.2 DESCRIPTION OF WORK
 - A. This Section includes labor, material, and equipment necessary for complete boiler system as specified and shown on the drawings:
 - 1. Gas Vent Pipe & Fittings
 - 2. Miscellaneous Breeching Materials
 - B. Provision for boiler vent and combustion air pipe
 - C. Refer to other Division 23 sections for related you
- 1.3 REFERENCE STANDARDS
 - A. Refer to Section 230200 for a conera description of requirements applying to this section.
- 1.4 QUALITY ASSURANCE
 - A. Refer to Section 230210 for a eneral description of requirements applying to this section.
 - B. Quality Assurance:
 - 1. Manufacturers. Firms regularly engaged in manufacture of equipment types and capacities required, whose products have been in satisfactory use in similar service for not less than 5 years.
 - 2. Regulatory Requirements:
 - a. NFPA Compliance: Install gas-fired cast-iron boilers in accordance with National Fire Protection Association (NFPA) Code 54 "National Fuel Gas Code".
 - b. NFPA 211 Compliance: Heating equipment burning gas, solid or liquid fuels, Section 60.

SUBMITTALS

- A. Submit shop drawings and product data in accordance with Section 230200.
- B. Submit the following:
 - 1. Shop Drawings
 - 2. Product Data

Tetra Tech 200-15704-17001

3. Evidence of specified code or other compliance.

1.6 SUBSTITUTIONS

- A. The listed equivalent or substituted manufacturers along with the bidding related contractor shall be completely responsible to comply with all requirements on all contract documents. This shall include, but not be limited to, space requirements, code clearances, the type, horsepower, capacities, number and size of services required from other trades, including all required ancillary items furnisher and installed by other trades. If the manufacturer or related bidding contractor does not comply with these requirements, this Contractor shall be responsible for any and all additional costs associated with the changes required by other trades.
- 1.7 WARRANTY/GUARANTEE
 - A. All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, GENERAL REQUIREMENTS

PART 2 – PRODUCTS

2.1 GAS VENT PIPE & FITTINGS

- A. The gas vent system shall be so engineered and constructed as to develop a positive flow adequate to exhaust all flue gases to outside atmosphere, without condentation within the vent.
- B. All parts of vent system shall be of Underwriters' Laboratories, Inc., listed Metal-Fab Type CORR/Guard Model CG, double wall gas vent piping, and such piping shall be continuous from the appliance outlets into Metal-Fab vent terminal. Verying System shall be rated at 6" W.C. and tested to 15" W.C. per UL Standard 1738.
- C. The Metal-Fab gas vent piping shallbe installed in full compliance with the terms of its listing, with the manufacturer's installation instructions, and with nationally recognized building codes representing good practice for such installations.
- D. For vent sizes 6" to 12" incide diameter, inner wall thickness shall be 0.015", Type AL29-4C stainless steel.

Outer casing shall 0.018, aluminized steel.

- E. Inner and outer weaks shall be connected by means of spacer clips that maintain concentricity of the annular space and abow differential thermal expansion of the inner and outer walls.
- F. All supports, wall penetration, terminal with miter cut and birdscreen, boiler connector and condensate drain brang shall be included.
- G. All joints shall be sealed using manufacturer's approved sealant. Joints exposed to the weather shall be realed to prevent rainwater from entering the annular space between inner and outer walls.
 - Provide adequate accessibility, head room and dimensions so that all vent connections can be correctly sized, spaced and supported.
- I. Manufacturers: Metal-Fab, Metalbestos, Heat Fab, Inc., American Metal Products, Van-Packer Co.
- 2.2 MISCELLANEOUS BREECHING MATERIALS
 - A. Provide miscellaneous materials and products of types and sizes to comply with breeching requirements including proper connection of equipment.

- B. Provide PVC combustion air intake pipe and accessories:
 - 1. Pipe: ASTM D-1785 Schedule 40, Type 1, Grade 1.
 - 2. Fittings: ASTM D-2466, Schedule 40.
 - 3. Solvent Cement: ASTM D-2564, Schedule 40 and DWV.
 - 4. Uniformity: To ensure installation uniformity, all piping components shall be manufacturer.
 - 5. Flux shall be non-toxic type and non-corrosive.

RBIDE

PART 3 - EXECUTION

3.1 GAS VENT PIPE

A. The vent manufacturer shall warrant the complete system against functional failure due to defects in material and workmanship for 10 years from date of delivery. The system annufacturer shall be responsible for checking the sizing, design, and installation of the system. If any component fails to perform its intended function of exhausting combustion by-products from the boiler equipment, for any reason, within 10 years of shipment, the system support shall, at no expense to the Owner, provide a replacement part or parts FOB jobsite.

END OF SECTION 230/10



SECTION 230450

REFRIGERATION EQUIPMENT – HVAC

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. The general provisions of the contract, including the conditions of the contract (General, Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the vork specified in this section.
- B. Refer to Section 230200 for HVAC General Provisions
- C. Refer to Section 230210 for HVAC Basic Materials & Methods.

1.2 DESCRIPTION OF WORK

- A. This Section includes labor, material, equipment and supervision to for the blowing:
 - 1. Rotary Scroll Water Chiller
 - 2. Cooling Tower (Centrifugal-Forced Draft)
 - 3. Refrigerant Gas Detection System
- B. Provide complete refrigeration system including chillers, cooling towers, underground pre-insulated pre-fabricated piping, aboveground piping and anrequired accessories.
- 1.3 REFERENCE STANDARDS
 - A. Refer to Section 230200 for a general description of requirements applying to this section.
 - B. Comply with applicable provisions of
 - 1. International Mechanical Code
 - 2. ASME Codes for Produce Vessels
 - 3. A.R.I. Capacity Latings
 - 4. NFPA Part phots
 - 5. ASHRAE Stundard 15
 - ANARAE Standard 90.1, Section 6, Table 6.8.1A thru J, minimum equipment efficiency.

1.4 **DUALITY ASSURANCE**

Refer to Section 230210 for a general description of requirements applying to this Section.

Whenever a variable frequency PWM drive is installed to control an AC motor, a maintenance-free, circumferential, conductive micro fiber shaft grounding ring shall be installed on the AC motor drive end to discharge shaft currents to ground. Recommended part: AEGIS SGRTM Bearing Protection Ring, as made by Electro Static Technology. Install in accordance with the manufacturer's written instructions.

C. Verification of Fluid Cooler Performance:

- 1. Manufacturer shall have a thermal performance testing program for water cooling towers certified by the Cooling Technology Institute (CTI) in accordance with CTI Specification Standard STD-201. Manufacturer's performance guarantees or performance bonds shall also be accepted.
- 2. Unit Sound Performance ratings shall tested and certified according to CTI ATC-128 standard. Sound ratings shall not exceed specified ratings.

1.5 SUBMITTALS

- A. Submit shop drawings and product data in accordance with Section 230200.
- B. Submit the following:
 - 1. Shop drawings and product data for all equipment in this section.
 - 2. 1/4'' = 1'-0'' scale layout of all equipment in Mechanical Room.

1.6 SUBSTITUTIONS

- A. The listed equivalent or substituted manufacturers along with the bioding related contractor shall be completely responsible to comply with all requirements on all contract documents. This shall include, but not be limited to, space requirements, code clearances, the type, horsepower, capacities, number and size of services required from other trades, including all equired ancillary items furnished and installed by other trades. If the manufacturer or related bidding contractor does not comply with these requirements, this Contractor shall be responsible for any ord all additional costs associated with the changes required by other trades.
- 1.7 WARRANTY/GUARANTEE
 - A. All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, General Reduirements. In addition, the following special guarantee applies:
 - 1. Manufacturer shall guarantee Ubrefrigeration equipment including parts and labor, for five (5) years from start-up

PART 2 – PRODUCTS

2.1 ROTARY SCRULE THATER CHILLER

A. General: Itstall a shown on the schedules and plans, factory assembled, charged, and tested watercooled scroil compressor chiller as specified herein. Chiller shall be designer, selected, and construct using a rarigerant with flammability rating of "1", as defined by ANSI/ASHRAE STANDARD-34 Number Designation and Safety Classification of Refrigerants. Chiller shall include, but is not limited or a complete system with not less than two refrigerant circuits, scroll compressors, direct expansion type evaporator, water-cooled condenser, refrigerant, lubrication system, interconnecting wiring, safety and operating controls including capacity controller, control center, motor starting components, and special features as specified herein or required for safe, automatic operation.

Compressors shall be hermetic, scroll-type, including:

- 1. Compliant design for axial and radial sealing
- 2. Refrigerant flow through the compressor with 100% suction cooled motor.
- 3. Large suction side free volume and oil sump to provide liquid handling capability.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- 4. Compressor crankcase heaters to provide extra liquid migration protection.
- 5. Annular discharge check valve and reverse vent assembly to provide low pressure drop, silentshutdown and reverse rotation protection.
- 6. Initial Oil charge
- 7. Oil level sightglass
- 8. Vibration isolator mounts for compressors.
- 9. Brazed-type connections for fully hermetic refrigerant circuits.
- 10. Microprocessor controlled, factory installed across-the-line type compressor notor starters.
- C. Each refrigerant circuit shall include: liquid line shutoff valve with charging pot, how side pressure relief device, filter-drier, solenoid valve, discharge service valve, system high pressure relief device, sight glass with moisture indicator, expansion valves, and flexible, clored-cell form insulated suction line.
- D. Evaporator
 - 1. Evaporator shall be a direct expansion shall and tune construction, dual circuit heat exchanger capable of refrigerant working pressure of 400 PSIG and iquid side pressure of 150 PSIG.
 - 2. Evaporator shall be covered with ³/₄", flexible, cosea-cell insulation, thermal conductivity of 0.26 (BTU/HR-Ft2-°F/in.) maximum. Water no cells shall be insulated by Contractor after pipe installation.
 - 3. Heat exchangers shall be ASME pressure vessel code certified.
 - 4. Installing contractor must include accommodations in the chilled water piping to allow proper drainage and venting of the neat exchanger.
 - 5. The water connections shall be fully accessible and grooved to accept ANSI/AWWA C-606 couplings.
- E. Condenser
 - 1. Condenser such be a cleanable thru-table construction with removable heads and integral subcoding. New exchanger shall be capable of a refrigerant side working pressure of 560 PSIG and liquid size pressure of 150 PSIG.
 - The condenser shall be equipped with relief valves and be capable of holding the full refrigerant charge for pumpdown.
 - The water connections shall be fully accessible and grooved to accept ANSI/AWWA C-606 couplings.

Controls

- 1. General: Automatic start, stop, operating, and protection sequences across the range of scheduled conditions and transients.
- 2. Microprocessor Enclosure: NEMA 1 (IP32) powder painted steel cabinet with hinged, latched, and gasket sealed door.

- 3. Microprocessor Control Center:
 - a. Automatic control of compressor start/stop, anti-coincidence and anti-recycle timers, automatic pumpdown on shutdown, evaporator pump, and unit alarm contacts. Automatic reset to normal chiller operation after power failure.
 - b. Remote water temperature reset via a Pulse Width Modulated (PWM) input signal or up to two steps of demand (load) limiting.
 - c. Software stored in non-volatile memory, with programmed setpoints retained in a battery backed regulated time clock (RTC) memory for minimum 5 years.
 - d. Forty-character liquid crystal display, numeric data in English units. Scaled teypad with sections for setpoints, display, print, entry, unit options and clock, and On/ off switch. Display descriptions and membrane keypad graphics shown in English language.
 - e. Programmable setpoints (within Manufacturer limits): Display language; chilled liquid temperature setpoint and range, remote reset temperature range scruaity schedule/holiday for start/stop, manual override for servicing, number for compressors, low liquid temperature cutout, low suction pressure cutout, high discharge pressure cutout, anti-recycle timer (compressor start cycle time), and anti-coincident time) (denay compressor starts).
 - f. Display Data: Return and leaving evaporator liquid temperatures, low leaving liquid temperature cutout setting, English data, surtion cressure cutout setting, each system suction pressure, discharge pressure, liquid temperature reset via a 0-20 VDC input, 2-10 VDC input or a 0-20mA input contact closure, anti-recycle timer status for each compressor, anti-coincident system start timer condition compressor run status, no cooling load condition, day, date, and time, daily start/ston times, holiday status, automatic or manual system lead/lag control, lead system definition, compressor starts/operating hours (each), status of hot gas valves (if supplied), run premisive status, number of compressors running, liquid solenoid valve status, load & us oad timer status, water pump status.
 - g. System Safeties: Challesause individual compressor systems to perform auto shut down; manual reset required offer the third trip in 90 minutes. Includes: high discharge pressure, low suction pressure, high pressure switch, and motor protector. Compressor motor protector shall protect against damage due to high input current or thermal overload of windings.
 - h. Unit safetus: Shall be automatic reset and cause compressors to shut down if low ambient, low leaving chilled liquid temperature, under voltage, and flow switch operation. Contractor shall provide flow switch installation and wiring per chiller manufacturer requirements.
 - alarm Contacts: Low ambient, low leaving chilled liquid temperature, low voltage, low battery, and (per compressor circuit): high discharge pressure, and low suction pressure.
 - BAS/EMS Temperature Reset: Chiller to accept 4 to 20mA, 0 to 10 VDC, or discrete contact closure input to reset the leaving chilled liquid temperature.
 - Pressure Transducers and Readout Capability
 - a. Discharge Pressure Transducers: Permits unit to sense and display discharge pressure.
 - b. Suction Pressure Transducers: Permits unit to sense and display suction pressure.
- 5. Manufacturer shall provide any controls not listed above, necessary for automatic chiller operation. Mechanical Contractor shall provide field control wiring necessary to interface sensors

to the chiller control system.

- G. Power Panels:
 - 1. NEMA 1 (IP32), powder painted steel cabinets with hinged, latched, and gasket sealed outer doors. Provide main power connection(s), control power connections, compressor start contactors, current overloads, and factory wiring.
 - 2. Power supply shall enter unit at a single location, be 3-phase of scheduled voltage, and connect to individual terminal blocks per compressor. Separate disconnecting means and/or external cancher circuit protection (by Contractor) required per applicable local or national codes.
- H. Exposed compressor and control power wiring shall be routed through liquid tight con
- I. Power Supply Connection shall be:
 - 1. Single Point Power Supply: Single point Terminal Block for field connection and interconnecting wiring to the compressors. Separate external protection must be supplied, by others, in the incoming power wiring, which must comply with the National Electic Code and/or local codes.
 - 2. Single Point Disconnect Switch: Single Point Non-Fused Disconnect and lockable external handle (in compliance with Article 440-14 of N.E.C.) can be upplied to isolate the unit power voltage for servicing. Separate external fusing must be supplied, by others, in the incoming power wiring, which much comply with the National Electric (code and/or local codes.
 - 3. Control Power Transformer (Factory Mounted). Converts unit power voltage to 120-1-60 (500 VA capacity). Factory mounting includes arima w and secondary wiring between the transformer and the control panel.
 - 4. Differential Pressure Switch (Feld Mounted): 3-45 PSIG (0.2-3 bar) range with ¹/₄" NPTE pressure connections. Provide with evaporator.
 - 5. Standard (3/4") insulation. Water ozzles shall be insulated by Contractor after pipe installation.
 - 6. ANSI/AWWA C-605 Flange Kit (Field Mounted): ANSI/AWWA C-606 flange adapters included with the water connections on the evaporator and condenser providing raised face flanges for field piping connection.
 - 7. Service Isolation Valves (Factory-Mounted): Service suction (ball type) isolation valves are added to unit per cacut in addition to the standard discharge service valve (Factory-mounted)
 - 8. Hot Gas 2., Pass (Factory Mounted): Permits continuous, stable operation at capacities below the minimum step of unloading to as low as 5% capacity (depending on both the unit and operating conditions) by introducing an artificial load on the evaporator. Hot gas by-pass is installed on only one refrigerant circuit (System #2).

Sound Reduction (Factory-Mounted)

- a. Each compressor is individually enclosed in anacoustic sound blankets.
- 10. Vibration Isolation (Field-Mounted)
 - a. One inch spring isolators, level adjustable, spring and cage type, for mounting under the unit base.
- J. Manufacturers: Carrier, McQuay, Trane, York.

1. Any listed equivalent manufacturer and the Mechanical Contractor shall be completely responsible to comply with all requirements on the contract documents. This shall include, but not be limited to, space requirements, code clearances, the type, horsepower, capacities, number and size of services required from other trades.

2.2 COOLING TOWER (CENTRIFUGAL – FORCED DRAFT)

- A. The Contractor shall provide factory assembled cooling tower of the sectional counterflow blowthrough design.
- B. Tower shall have centrifugal fan assemblies built completely within the pan with all movies parts factory mounted and aligned. Internal baffles shall be provided to permit independent operation of the individual fan section assemblies.
- C. All steel components shall be fabricated from hot-dip G235 galvanized steer with an cut or sheared edges given an added protective coat of zinc rich compound.
- D. Cooling tower shall have the capacity to cool water as scheduled on the drawings. The tower fans shall operate against zero inches W.G. external static pressure.
- E. Pan-Fan Section
 - 1. The combination Pan-Fan section shall consist of havy gauge hot-dip galvanized steel framework, V-shaped self-cleaning pan and remrifugal fans mounted beneath the sloping undersides of the pan. The fans and motors shall be ocated in the dry entering air stream and arranged for ease of maintenance.
 - 2. Standard pan accessories shall include vircular access doors; lift- out, hot-dip galvanized perforated steel strainers; anti-vort king to ffte, valved waste water bleed line and brass make-up valve with plastic float, arranged for easy adjustment.
 - 3. The forward curved centrifigal faits shall be balanced statically and dynamically. Inlet rings on fan housings shall be high efficiency compound curve type. Fan discharge cowls shall extend into the pan to increase anniency and prevent water entrance into the fan housing. Fans shall be mounted on a steer shaft upported by heavy duty, self-aligning ball bearings with cast iron housings and crease horngs. Any required intermediate bearings shall be self-aligning, oil lubricated sleeve type, with split cast iron pillow block housings. Extend lube lines to the exterior of the section enclosure.
- F. Fan Motor and Drive
 - 1. Fur shall be indirectly driven by high efficiency type, 1750 RPM TEFC, ball bearing motors with copper windings and a service factor of at least 1.15. Electric service motor requirements shall be as scheduled on the drawings. Each motor shall be located under the side of the sloping pan for weather protection and shall be mounted on an adjustable base.

V-belt fan drives shall be designed for at least 150% of motor nameplate ratings and all moving parts shall be protected by removable screens and panels.

- Surface Section
 - 1. The heat transfer casing section shall be removable to facilitate rigging. The section will contain wave-formed wet deck surface below a spray-type water distribution system, with removable eliminators at the top.

- 2. The wet deck surface material shall be PVC minimum of 12 mils thick, with leading and trailing edges rolled to a double thickness. The material shall be non- combustible and impervious to rot, decay, fungus, or biological attack.
- H. Water Distribution
 - 1. Water shall be distributed evenly over the fill area through a spray tree consisting of Schedule 10 PVC spray header and removable branches. Branches and plastic spray nozzles shall be held in place with snap-in rubber grommets to allow easy removal of branches or individual nozzles. The header shall be equipped with a valved connection for a pressure gauge.
- I. Eliminators
 - 1. Eliminators shall be fabricated of PVC and be removable in easily handled sections. They shall have a minimum of three directional breaks with a hooked leaving edge.
- J. Capacity Control
 - 1. The tower shall be provided with capacity control based on the following:
 - a. Variable speed motor and variable frequency drive selected by the tower manufacturer for this application, complete with all factory mounted wring and controls. NOTE: VFD shall be provided by Division 26 Electrical.
- K. Discharge Sound Attenuation
 - 1. A sound attenuator designed and constructed by the tower manufacturer shall be provided at the tower air discharge.
 - 2. The materials of construction and mine shall be as described for intake sound attenuation.
- L. Electric Heaters
 - 1. Electric immersion heaters are normally factory installed in the cooling tower basin. The heaters are controlled by a thermostet with the sensing bulb located in the pan water. A low water level control, also factory installed, prevents heater operation unless the heater elements are fully submerged. Immersion becaters should be interlocked with water circulating pump to deenergize heaters wherever the circulating pump is running.
- M. Furnish a field wired, NEMA 4 basin heater control panel. Wiring of heaters and contacts shall be field wired to system control panel.
- N. Tower pranufacturer to furnish sound power levels of tower with required attenuation as part of shop drawing submission.

Re cooling tower shall be as manufactured by Baltimore Aircoil Co., Marley or Evapco.

Any listed equivalent manufacturer and the Mechanical Contractor shall be completely responsible to comply with all requirements on the contract documents. This shall include, but not be limited to, space requirements, code clearances, the type, horsepower, capacities, number and size of services required from other trades.

- 2.3 REFRIGERANT GAS DETECTION SYSTEM
 - A. Refrigerant gas detection system supplier/installer shall be familiar with standard practices of safety and installation for refrigerant gas vapor detection systems and shall provide these systems as a normal

course of business. Acceptable gas detection supplier shall provide a list of the last five (5) similar projects.

- B. System shall meet or exceed the latest ASHRAE 15 requirement and EPA standard 608CFR. System shall incorporate all latest revisions.
- C. System shall be capable of detecting presence of CFC, HCFC, or HFC refrigerant based on actual approved and installed chiller. System shall be capable of indicating, alarming and shutting down equipment as specified in this section and in governing regulations. Oxygen deficiency mentioning shall not be acceptable in lieu of TLV-TWA monitoring for human safety exposure.
- D. Sequential sampling and multi-point monitoring shall be employed where air flow currents and room size prohibit a representative sample from one sensing point. Diluted samples due to vontilation air flow currents shall employ multi-point monitoring techniques strategically bected according to regulation guidelines.
- E. System design considerations shall also be incorporated in leak retection monitoring sensing location(s), for early warning indication to prevent a major loss of refrige ant without alarm, should a leak occur.
- F. Analyzer shall be microprocessor-based and employ inflared (IK) sensor technology. It shall accurately provide sensing down to one part per million (ppe) and shall be compound specific and monitor compound as specified and be calibrated for effigerant as required by approved chiller. Any installed unit can be switched to monitor, at a future date, to another refrigerant type by changing one part and recalibrating. Adjustable three level altern for each point shall be supplied with common alarm output contacts. Provide local digital relication of ppm level for four sample points. Alarms shall be identified by an alarm message indicating the point in alarm and the alarm level. Unit shall have self-diagnostics, and supply a common malfunction output for alarm horn-beacon. Loss of sample flow at either sample or ZEK line will indicate system malfunction.
- G. Four-point sequential sampling system shall be integrated into one analyzer enclosure. Microprocessor shall sequentially sourced required flow valves and communicated output signals to allow monitoring from pathole emote sampling locations. Unit shall read and hold output value of infrared sensor and control the corresponding four-point sequential sampling assembly. Each sampling point shell have adjustable sampling time and adjustable levels of alarm. Sample line capability up to 500 °. The system shall have add-on sample point expansion capability for future expansion.
- H. The system shall nonitor and display accurately within the range of 0 to 25 ppm for refrigerant system and chiller diagnostics, detecting low level refrigerant leaks and deterioration of system efficiency.
- I. Frovide NEMA-rated wall mount enclosure with dust-tight door seal. Auto zero calibration shall be nitiated at one-hour interval (adjustable) or manually at the monitor and shall automatically zero by drawing air from an uncontaminated air source. Include built-in sample pump and differential pressure flow switch for low flow indication. Provide four separate 4-20 mA dc analog outputs and one RS-485 output of refrigerant level(s). Unit shall be insensitive to vibration and shall provide for a continuous sample. Response time shall be twenty (20) seconds or less to ninety-nine (99%) percent of reading. Malfunction relay shall be energized due to flow loss or electrical malfunction.
- J. The system shall be configured to provide relay board, with dry contacts for each channel, to initiate output signal for three level alarms at local panel, interface with both the Building ADT Security System and the Room Ventilation System. An early leak warning alarm shall be set at 10 ppm, regardless of refrigerant type, to prevent large refrigerant loss and provide chiller diagnostics. Other

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

alarm level shall be set at or below the TLV-TWA level of 1000 ppm R-410A.

At the TLV-TWA level, the system shall activate the purge ventilation system and sound a refrigerant leak alarm. Malfunction alarm indication and horn shall be provided. Provide dry alarm contacts for each alarm level.

- K. Installation: Unit shall be factory calibrated. Field calibration is unacceptable at time of installation.
- L. Maintenance & Calibration: No calibration shall be required for a period of one (1) year from date of installation. Zero filter and end of line filters shall be replaced every six months or sooner, based or usage.
- M. Manufacturers: General Analysis Corporation, Gensis International, Inc., Yokarawa Corporation, Thermal Gas Systems, Inc., MSA Co.

PART 3 – EXECUTION

- 3.1 REFRIGERATION EQUIPMENT
- A. All equipment to be installed in accordance with manufacturer's recommendations.
- 3.2 INSTALLATION OF ROTARY SCROLL COMPRESSOR WATER CHILLER
 - A. Install in accordance with manufacturer's instructions.
 - B. Provide for connection to electrical service.
 - C. Provide manufacturer's spring isolators to reduce Vibration transmission.
 - D. Arrange piping for easy dismantling to permit the cleaning.
 - E. Provide piping from chiller relief valve to butdoors. Size as recommended by manufacturer and ANSI/ASHRAE Standard 15.
 - F. Manufacturer's Field Services
 - 1. Manufacturer shall survising factory trained service engineer without additional charge to start the unit(s). Representative shall provide leak testing, evacuation, dehydration, and charging of the unit(s) as require l. Chiller manufacturers shall maintain service capabilities to promptly respond within 24 hours or less to service calls at the site.
 - 2. A star up log shall be furnished by the manufacturer to document the chiller's start-up date and shall be signed by the owner or his authorized representative prior to commissioning the chillers.

The nanufacturer shall furnish complete submittal wiring diagrams of the chiller(s) starter(s) and associated components such as cooling tower, pumps, interlocks, etc. as applicable.

Start-up all units in accordance with manufacturer's start-up instructions. Replace damaged or malfunctioning controls and equipment.

REFRIGERANT GAS DETECTION SYSTEM

- A. Install system in accordance with manufacturer's written instructions.
- B. Install sensing lines, alarm lights and horns as shown on the drawings.
- C. On high level alarm, system shall start emergency ventilation system.

3.4 COOLING TOWERS

- A. Examine proposed route of moving cooling towers into place and verify that it is free of interferences.
- B. Examine elements and surfaces to support cooling tower.
- C. Verify piping and wiring roughing-in locations.

- D. Verify suitability of branch-circuit wiring.
- E. Install cooling tower according to manufacturer's written instructions.
- F. Install cooling tower level and plumb, and fasten to supporting structure with seismic restrait
- G. Maintain recommended clearances for service and maintenance.
- H. Electrical Wiring: Install electrical devices furnished by cooling tower manufacturer that are not factory mounted.
- I. Manufacturer's Field Service: Engage a factory-authorized service representative to inspect fieldassembled components and equipment installation, including piping and electrical connections. Report results in writing.

END OF SECTION 230450

SECTION 230451

VARIABLE REFRIGERANT FLOW SPLIT SYSTEM HEAT RECOVERY WITH SIMULTANEOUS HEATING AND COOLING

PART 1 – GENERAL

- 1.1 RELATED DOCUMENTS
 - A. The general provisions of the contract, including the conditions of the contract (Se Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the specified in this section.
 - B. Refer to Section 230200 for HVAC General Provisions
 - C. Refer to Section 230210 for HVAC Basic Materials & Methods.
- 1.2 DESCRIPTION OF WORK
 - A. This Section includes labor, material, equipment and supervision for the following:
 - 1. Single or Multiple Outdoor Condensing Units
 - 2. Multiple Indoor Ducted Cabinet or Ceiling Cassette HearPump Units.
 - 3. MCU refrigerant control unit(s) for simultaneous distribution of liquid and hot gas refrigerant to fan coil units for simultaneous heating and cooling overation.
 - B. Provide complete refrigeration system inchange ondensing units, cooling units, aboveground refrigerant piping, and all required controls and accessories for a complete and operable system.
 - C. The variable capacity, air conditioning system shall be a split system consisting of ductless evaporators exclusively matched to the outd or concerning unit.
- 1.3 REFERENCE STANDARD
 - A. Refer to Section 23020 for a general description of requirements applying to this section.
 - B. Comply with applicable provisions of:
 - 1. International Aechanical Code
 - 2. ASMIL Code for Pressure Vessels
 - 3. A.P.I. Capacity Ratings
 - . NFPA Pamphlets
 - ASHRAE Standard 15
 - . ASHRAE Standard 90.1, Section 6, Table 6.8.1A thru J, minimum equipment efficiency.
 - 7. ISO 9001 and 14001.
- 1.4 QUALITY ASSURANCE
 - A. Refer to Section 230210 for a general description of requirements applying to this Section.

- B. The units shall be listed by Electrical Laboratories (ETL) and bear the ETL label.
- C. All wiring shall be in accordance with the National Electric Code (NEC).
- D. The system shall be produced in an ISO 9001 and ISO 14001 facility, which are standards set by the International Standard Organization (ISO). The system shall be factory tested for safety and function.
- E. The outdoor unit shall be factory charged for a length of 25 feet of refrigerant with R410A refrigera
- F. A dry air holding charge shall be provided in the evaporator(s).
- G. Provide a letter from the VRF system manufacturer agent certifying completion of Pre-Corso ucuon installation training. Certification shall indicate that manufacturer's field pixing and wiring recommendation for specified projects have been reviewed prior to installation.

1.5 SUBMITTALS

- A. Submit shop drawings and product data in accordance with Section 22 020
- B. Submit the following:
 - 1. Shop drawings and product data for all equipment in this sector.
 - 2. $\frac{1}{4}$ = 1'-0" scale layout of all outdoor equipment adjaces to the building.
- 1.6 SUBSTITUTIONS
 - A. The listed equivalent or substituted manufacturers along with the bidding related contractor shall be completely responsible to comply with all req treme ts on all contract documents. This shall include, but not be limited to, space requirements, cole clearances, the type, horsepower, capacities, number and size of services required from other trales, including all required ancillary items furnished and installed by other trades. If the manufacturer or related bidding contractor does not comply with these requirements, this Contractor shall be responsible for any and all additional costs associated with the changes required by other trades.
- 1.7 WARRANTY/GUARA NT 5
 - A. All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, General Requirements. In addition, the following special guarantee applies:
 - 1. Manufacturer shall warrant all refrigeration equipment including parts and labor, for two (2) years from start-up.
 - Compressors shall be warranted for parts only for an additional four (4) years.

PAPT 2 – PRODUCTS

GENERAL REQUIREMENTS

The system performance shall be in accordance with ARI 210/240 test conditions as shown in the performance table below. System refrigerant piping shall be sized and installed per the manufacturer's piping diagrams and piping procedures.

B. The cooling performance is based on 80°F DB / 67°F WB for the indoor unit and 95°F DB /75°F WB for the outdoor unit and 25 feet of piping.

VARIABLE REFRIGERANT FLOW SPLIT SYSTEM HEAT RECOVERY WITH SIMULTANEOUS HEATING AND COOLING 230451-2

Tetra Tech 200-15704-17001

- C. The operating range in cooling will be $23^{\circ}F DB \sim 115^{\circ}F DB$.
- D. Provide all equipment. Materials, programming, and technical support as required to interface with the DDC system, as defined in Section 230900.

2.2 INDOOR UNIT

- A. General: The indoor unit shall be factory assembled and pre-wired with all necessary electronic and refrigerant controls. Both liquid and suction lines shall be individually insulated between the autoor and indoor units. The unit shall have a self diagnostic function, 3-minute time delay mechanism, and have a factory pre-charge of R410A adequate for 25 feet of total length.
- B. Unit Cabinet Ceiling Recessed Type:
 - 1. The indoor unit shall have a finished cabinet for exposed ductless applications
 - 2. The drain and refrigerant piping shall be concealed above the finisher ceiling for flexible installation from the right side.
 - 3. The cabinet shall be supplied with suspension bracket for securely mounting the cabinet to threaded rod.
 - 4. The cabinet includes a receiver to accept signals from a vared remote controller.
 - 5. The indoor unit shall include a stub duct connection for air distribution to an adjoining zone as shown on the drawings.
 - 6. The indoor unit shall include a factory mered and wired high lift condensate drain pump capable of 29" lift. The condensate panchall be internally trapped.
- C. Unit Cabinet Ceiling Concealed Duced Ty
 - 1. The indoor unit shall have a garanized steel, insulated cabinet for ceiling concealed ducted supply and return air appreciations. The unit shall contain factory fabricated filter housing suitable for a 1" or 2" MERV or epilereable filter element, complete with side access and coverplate with full gasket.
 - 2. The drain and refrigerant piping shall be concealed above the finished ceiling for flexible installation from moright side.
 - 3. The cibinet shall be supplied with suspension bracket for securely mounting the cabinet to threaded rod

The cabinet includes a receiver to accept signals from a wired remote controller.

The indoor unit shall include a high lift condensate drain pump accessory, where applicable, eapable of 29" lift. The condensate pump shall be internally trapped.

- 1. The evaporator fan shall be an assembly with of a direct-driven single motor.
 - 2. The fan shall be statically and dynamically balanced and operate on a motor with permanent lubricated bearings.
 - 3. The indoor fan shall offer a choice of three speeds, High, Medium, and Low.

an

ADDITION AND RENOVATIONS

- 4. The supply air shall be distributed to the space through a 4-way blow wide blade grille.
- 5. The return air shall be returned to the unit through an integral center return air grille.
- E. Filter:
 - 1. The return air filter shall be integral to the unit and shall be replaceable or washable, based on unit configuration and manufacturers standard.
- F. Coil:
 - 1. The evaporator coil shall be an aluminum fin on copper tube heat exchanger.
 - 2. All tube joints shall be brazed with silver alloy.
 - 3. All coils shall be factory pressure tested.
 - 4. A condensate pan shall be provided under the coil with a drain correct
 - 5. The evaporator coil shall be controlled by a factory mounted electronic thermal expansion valve.

G. Control:

- 1. The indoor fan coil unit shall have a wired remote controller capable to operate the system.
- 2. The wired remote controller shall control: n/off operation, operation mode, fan speed, temperature set point and filter alarm.
- 3. The wired remote control shall perform finit angnostic functions which may be system related, indoor unit or outdoor unit related depending on the fault code. Temperature range on the remote control shall be 64°F to 90°F in comer node and 50°F to 86°F in heating mode.
- 4. The indoor unit microprocesses shall have the capability to receive and process commands via return air temperature and indoor coil temperature sensors enabled by commands from the remote control.
- 5. The system shall have automatic restart capability after a power failure has occurred.
- 6. Multiple index fan con units located in the same room shall be controlled in a master/slave configuration value single wired remote controller connected to the master indoor unit and interconnecting wiring between the master and slave units.
- 7. The indeor fun coil unit shall include a field installed relay module for interface to the existing IAS system. The relay module shall be interlocked to the existing DDC controller located in the heating only classroom unit ventilator. The existing unit ventilator DDC controller shall enable/disable the cooling only fan coil unit to provide cooling operation per the programmed schedule and sequence of operation of the building automation system.

I.

Each system shall be networked to a centralized controller with data management service device to provide local schedule and set point control, error history management, and web server browsing from static IP address.

Sound:

1. Indoor unit sound levels shall not exceed: 41 dBA

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- J. MCU Refrigerant Distribution Control Module:
 - 1. The MCU unit shall include refrigerant liquid, suction and hot gas connections to the outdoor unit. All three refrigerant lines shall be insulated from the MCU to the outdoor unit.
 - 2. The MCU unit shall include refrigerant liquid and vapor connections to the indoor units.
 - 3. The MCU unit shall include a 208/230-1-60 power supply connection.
 - 4. The MCU shall include a condensate drain connection.
 - 5. The MCU shall include a control communications connection for network to the indocen neo units and outdoor air-cooled condensing unit(s).
 - 6. The MCU shall include heating and cooling solenoid valves for automatic distribution of liquid refrigerant or hot gas for simultaneous heating and cooling fan coil operation.

2.3 OUTDOOR UNIT

- A. General: The outdoor unit(s) shall be specifically matched to the corresponding indoor unit size(s). The outdoor unit(s) shall be complete factory assembled and pre-wired with all necessary electronic and refrigerant controls.
- B. Unit Cabinet: The cabinet shall be ivory with a finished power coated baked enamel paint.

C. Fan

- 1. The fan shall be a direct drive, propeller type in
- 2. The motor shall be BLDC type with perparently lubricated type bearings and inherent overload protection.
- 3. The fan shall be capable of lagar taxic operation up to 0.31" WC for ducted applications.
- 4. A fan guard is provided on the ordoor unit to prevent contact with fan operation.
- 5. Airflow shall be versical a scharge.
- D. Coil: The outdoor all shall be nonferrous construction with corrugated fin tube.
- E. Compressors:
 - 1. The outdoor unit shall have a minimum two compressors. One compressor shall be a Copeland digital scron compressor with 10% to 100% capacity modulation and one compressor shall be a Copeland fixed scroll compressor. Outdoor unit shall be capable of capacity modulation down to 10% of full load capacity.

The outdoor unit shall have an accumulator.

- The compressor shall have an internal thermal overload.
- 4. The outdoor unit shall operate with a maximum vertical height difference of 164 feet and overall maximum piping length of 3280 feet and maximum 721 feet (equivalent length) from outdoor unit to furthest indoor unit.
- F. Electrical:

Tetra Tech 200-15704-17001

- 1. The outdoor unit shall be powered by 208 volt, 3 phase, 60 hertz power.
- 2. The outdoor shall be controlled by a microprocessor located in the outdoor unit and via signals from the indoor units.
- G. Sound: Outdoor unit sound levels shall not exceed: 64 dBA
- 2.4 DESIGN MAKE: Samsung
- 2.5. MANUFACTURERS: Daikin, Samsung, Mitsubishi (Citi-Multi), LG HVAC, Panasonic.
 - A. Any listed equivalent manufacturer and the Mechanical Contractor shall be completely espendible to comply with all requirements on the contract documents. This shall include, but not be limited to, space requirements, code clearances, the type, horsepower, capacities, number and size of services required from other trades.
 - B. Provide a letter from the VRF system manufacturer agent certifying completion of Pre-Construction installation training. Certification shall indicate that manufacturer's field piping and wiring recommendation for specified projects have been reviewed prior to assultation.

PART 3 – EXECUTION

- 3.1 REFRIGERATION EQUIPMENT
 - A. All equipment shall be installed in accordance with manufacturer's recommendations.
- 3.2 FIELD QUALITY CONTROL
 - A. The system shall be installed under the supervision of a certified manufacturer's representative. Provide a letter from the VRF system projutatuler agent certifying completion of Pre-Construction installation training. Certification shall indicate that manufacturer's field piping and wiring recommendation for specified projects have been reviewed prior to installation.
 - B. Start up all units in accordance with manufacturer's start-up instructions. Replace damaged or malfunctioning controls are equipment.
 - C. Start up service and first year preventative/emergency service shall be provided by the manufacturer's authorized representative.
 - D. Customer operator training shall be provided by the manufacturer's authorized representative.

END OF SECTION 230451

SECTION 230510

WATER TREATMENT (HVAC)

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. The general provision of the contract, including the conditions of the contract (Gene Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the specified in this section.
- B. Refer to Section 230200 for HVAC General Provisions.
- C. Refer to Section 230210 for HVAC Basic Materials & Methods.

1.2 DESCRIPTION OF WORK

- A. This Section includes labor, material, equipment and supervision to provide a complete water treatment system for the following:
 - 1. Cleaning and treatment of circulating HVAC chilled water system.
 - a. Cleaning Compounds.
 - b. Chemical Treatment for Closed Loop Systems,
 - c. Chemical Treatment for Evaporative Coder System

1.3 REFERENCE STANDARDS

- A. Refer to Section 230200 for a general description of requirements applying to this section.
- B. Requirements established within the perions of the Project Manual titled Division 1, General Requirements, are collectively applicable to the work of this section.
- C. Technical Services: Provide the vervices of an experienced water treatment chemical engineer or technical representative to direct flushing, cleaning, pre-treatment, training, debugging, and acceptance testing operations; direct and perform chemical limit control during construction period and monitor system. The a period of 12 months after acceptance, including not less than four service calls and water status reports. Minimum service during construction/start-up shall be 8 hours.
- Field Quality Control and Certified Laboratory Reports: During the one-year guarantee period, the water treatment laboratory shall provide not less than 12 reports based on on-site periodic visits, sample aking and testing, and review with Owner, of water treatment control for the previous period. In addition to field tests, the water treatment laboratory shall provide certified laboratory test reports. These monitoring reports shall assess chemical treatment accuracy, scale formation, fouling and corrosion control, and shall contain instructions for the correction of any out-of-control condition.

Log Forms: Provide one-year supply of preprinted water treatment test log forms.

SUBMITTALS

In accordance with Section 230200 provide the following:

- A. Manufacturer's Literature and Data:
 - 1. Cleaning compounds and procedures.

Tetra Tech 200-15704-17001

- 2. Chemical treatment for closed systems.
- 3. Chemical treatment for steam systems, including installation and operating instructions.
- 4. Chemical treatment for open loop systems.
- B. Water analysis verification.
- C. Materials Safety Data Sheet for all proposed chemical compounds.
- D. Maintenance and operating instructions.

PART 2 – PRODUCTS

- 2.1 CLEANING COMPOUNDS:
 - A. Refer to Section, PIPING SYSTEMS & ACCESSORIES HVAC, PART 3, for fushing and cleaning procedures.
- 2.2 CHEMICAL TREATMENT FOR CLOSED LOOP SYSTEMS:
 - A. Inhibitor: Provide sodium silicate, sodium nitrite/borate, or other approved proprietary compound suitable for make-up quality and make-up rate and which will cause or enhance bacteria/corrosion problems or mechanical seal failure due to excessive total dissolved solids. Shot feed manually. Maintain inhibitor residual as determined by water acatment laboratory, taking into consideration residual and temperature effect on pump mechanical seals
 - B. pH Control: Inhibitor formulation shall include advauate buffer to maintain pH range of 8.0 to 10.0.
 - C. Performance: Protect various wetted, coupled, meterials of construction including ferrous, and red and yellow metals. Maintain system essentially free of scale, corrosion, and fouling. Corrosion rate of following metals shall not exceed specified utils per year penetration; ferrous, 0.5; brass, 0.2; copper, 0.15. Inhibitor shall be stable at equipment skin surface temperatures and bulk water temperatures of, respectively, not less than 250 and 125 degrees Fahrenheit. Heat exchanger fouling and capacity reduction shall not exceed the lowed by fouling factor 0.0005.
 - D. Pot Feeder: By-pass type for chemical treatment schedule 10-gauge heads, 3/4-inch system connections and dange neck opening for chemical addition. Feeder shall be bypass filter feeder, minimum five-gellen, hastalled per detail on the drawings, for chilled water system, and for hot water system.
 - E. Water Analysis: Confirm raw water analysis or provide analysis if none is furnished.

| | Description | Year (Avg.) |
|----|------------------------------|-------------|
| | Silica (SiO ₂) | |
| 40 | Insoluble | |
| | Iron & Aluminum | |
| | Calcium (Ca) | |
| | Magnesium (Mg) | |
| | Sodium & Potassium (Na & K) | |
| | Carbonate (CO ₃) | |
| | | |

Tetra Tech 200-15704-17001

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

ADDITION AND RENOVATIONS

| | Bicarbonate (HCO ₃) |
|-----|---|
| | Sulfate (SO ₄) |
| | Chloride (Cl) |
| | Nitrate (NO ₃) |
| | Turbidity |
| | рН |
| | Residual Chlorine |
| | Total Alkalinity |
| | Non Carbonate Hardness |
| | Total Hardness |
| | Dissolved Solids |
| | Fluorine |
| F. | Conduct performance test to prove capacity and performance of treatment system. |
| | Raw water total hardness, ppm. |
| | Concentration cycles. |
| | Raw water, pH. |
| | System water, pH. |
| | Chemical solution used. |
| | Acid solution used, obe. |
| | Quantity or chemical solution hjected into system per cycle. |
| | Quantity of acidanje ted into system per cycle. |
| | Make up water equired. |
| | Wastero divin requirement. |
| G. | Recommended Conditions |
| | Buffered Nitrite: |
| | a. For temperatures 140°F to 180°F - 1000 ppm as sodium nitrite. |
| 7 | 2. Molybdate: |
| | a. 50 - 100 ppm as molybdate (chilled water). |
| | 3. pH - 7.0 to 10.0 |
| 2.3 | CHEMICAL TREATMENT FOR EVAPORATIVE COOLER SYSTEM |
| А. | Provide chemical feed and control equipment, as specified below, to introduce chemicals into each |

Tetra Tech 200-15704-17001 system only when the system is in operation.

- B. Controller: Provide a microprocessor based controller to control proportional chemical feed and system bleed rate. The controller shall have the following features as a minimum:
 - 1. The controller shall be housed in a corrosion resistant NEMA 4X enclosure.
 - 2. The unit shall have a menu driven keyboard programming capability. Control points shall keyboard executed and displayed.
 - 3. The controller shall provide proportional inhibitor feed based on make-up water flow fate
 - 4. A selectable timing feature shall be provided for operating a solenoid valve for ystem blocd-off.
- C. Inhibitor Feed Pump: a positive displacement chemical feed pump shall be supplied for inhibitor feed. Both stroke length and stroke frequency shall be adjustable. Provide an actisit hon pressure relief valve. The pump shall have the appropriate output capacity, pressure using and material of construction for the application.
- D. Contact Head Water Meter: Provide one electrical contacting head water meter, sized for this application, which will signal the controller and provide propertional in libitor feed. The water meter will be installed on the cooling tower make-up water line with a valved bypass.
- E. Solenoid (Bleed) Valve: An electrically operated solenoid valve of appropriate size shall be supplied for installation on the blowdown line and wired to the controller. The solenoid valve shall be installed with a Y-strainer. If solenoid valve is to be installed outside, it must be rated for outdoor use.

PART 3 - EXECUTION

3.1 INSTALLATION:

- A. Delivery and Storage: Deliver all chemicals in manufacturer's sealed shipping containers. Store in designated space and protect from del terious exposure and hazardous spills.
- B. Install equipment furnished by the chemical treatment supplier and charge systems according to the manufacturer's instructions and as directed by the Technical Representative.
- C. Perform tests and report results.
- D. Instruct owner per onnel in system maintenance and operation.
- 3.2 INSPECTIONS IND MAINTENANCE:
 - A. Furnish complete inspection and maintenance service on water treatment equipment for a period of one year after completion and acceptance of the water treatment equipment installation. This haintenance service shall begin concurrently with the guarantee. Maintenance work shall be performed by skilled personnel directly employed and supervised by the same company that provided the water treatment equipment specified herein.

The maintenance service shall include the following:

- 1. Monthly systematic examination of equipment.
- 2. Cleaning, lubricating, adjusting, repairing and replacing of all parts as necessary to keep the equipment in first-class condition and proper working order.
- 3. Furnishing all lubricant, cleaning materials and parts required.

- 4. The operational system shall be maintained to the manufacturer's standards specified including any changes and/or adjustments required to meet varying conditions.
- 5. Provide 24-hour emergency call-back service which shall consist of promptly responding to calls within two hours for emergency service should a shutdown or emergency trouble develop between regular examinations. Overtime emergency call-back shall be limited to minor adjustments and repairs required to protect the immediate safety of the equipment.
- 6. Service personnel shall report to the owner or his authorized representative upon arrival a d again upon completion of the required work. A copy of the work ticket containing a complete description of the work performed shall be given to the owner.
- 7. The Contractor shall maintain a log in the boiler room. The log shall list the rate and time of all monthly examinations and all trouble calls. Each trouble call shall be fully becabed including the nature of the call, necessary correction performed and/or parts replaced.

END OF SECTION 230510

Republic



SECTION 230600

AIR DISTRIBUTION & ACCESSORIES – HVAC

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. The general provisions of the contract, including the conditions of the contract (General, Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the work specified in this section.
- B. Refer to Section 230200 for HVAC General Provisions
- C. Refer to Section 230210 for HVAC Basic Materials & Methods.
- D. This Contractor shall coordinate with the work of Division 26 and the Fin Ararm System vendor for locations and mounting of all duct smoke detectors. These dedices are shown on the Mechanical Drawings for reference only to show the intent of the work. All locations shall be determined based on approved shop drawings from the Fire Ararm System vendor and the Contractor for the work of Division 26, Electrical. Mount shoke letectors in the supply and return air stream at each unit in accordance with NFPA 72.

1.2 DESCRIPTION OF WORK

- A. This Section includes labor, material, equipment and supervision to provide a complete air distribution system as specified herein and as shown on arawings.
 - 1. Ductwork Single Wall, Square and Recongular
 - 2. Ductwork Single Wall, Spiral Lound
 - 3. Double Wall Round Duct
 - 4. Double Wall Ductwork Square and Rectangular
 - 5. Flexible Air Duct
 - 6. Flexible Concessions
 - 7. Dampers
 - 8. Fire Dumper

 - 9. Air Diffusers, Registers and Grilles
 - 0. Prefabricated Roof Curbs and Equipment Supports
 - 11. VAV Units with Reheat
 - 12. Louvers
 - 13. Duct Access Doors (Interior)
 - 14. Roof-Mounted Intake/Exhaust Ventilators
 - 15. Fabric Air Dispersion Ductwork

ADDITION AND RENOVATIONS

1.3 REFERENCE STANDARDS

- A. Refer to Section 230200 for a general description of requirements applying to this section.
- B. Requirements established within the portions of the Project Manual titled Division 1, General Requirements, are collectively applicable to the work of this section.
- C. IMC (International Mechanical Code).
- D. SMACNA (Sheet Metal and Air Conditioning Contractors National Association, Inc.)
- E. American Society of Heating, Refrigerating and Air Conditioning Engineers' recommendation ASHRAE Guide shall apply to this work.
- F. ARI Standard 885 Standard for Estimating Occupied Sound Levels in the Applications of Air Terminals and Air Outlets.
- G. UL (Underwriter's Laboratories, Inc.)
- H. NFPA 90A shall apply to this work.
- I. State Fire Prevention Regulations.
- 1.4 QUALITY ASSURANCE
 - A. Refer to Section 230210 for a general description or requirements applying to this Section.
- 1.5 SUBMITTALS
 - A. Submit shop drawings and product data in accordance with Section 230200.
 - B. Submit the following:
 - 1. Shop drawings of all snet metal. Indicate all steel, piping, conduit, and Architectural/Structural features to demonstrate complete coordination. Scale shall not be less than ¹/₄".
 - a. Shop drawings that indicate the sizes and lengths of each section of ductwork as well as all system components such as coils, VAV boxes, access doors, dampers, diffusers and register locations. Also indicate the type of joints used and where internal acoustic lining or insulation, if required, will be utilized.
 - b. The location of the duct runs and the air outlets shall be closely coordinated with all other trade, by the sheet metal contractor to avoid interference. The shop drawings shall show the contact surfaces adjacent to the ducts or air outlets and the space assigned for concealment. The drawings shall indicate principal items of equipment, adjacent piping and conduit, etc., the location of which shall be secured from the contractors of other trades.
 - c. Sheet Metal Contractor to include resubmissions of the shop drawings to the Engineer. The resubmissions are to include all corrections to previous submissions.
 - 2. Manufacturer's literature and performance data of all equipment and devices.
 - 3. Samples: Furnish color samples, etc., at request of the Architect.
- 1.6 SUBSTITUTIONS
 - A. The listed equivalent or substituted manufacturers along with the bidding related contractor shall

AIR DISTRIBUTION & ACCESSORIES – HVAC 230600-2

Tetra Tech 200-15704-17001

be completely responsible to comply with all requirements on all contract documents. This shall include, but shall not be limited to space requirements, code clearances, the type, horsepower, capacities, number and size of services required from other trades, including all required ancillary items furnished and installed by other trades. If the manufacturer or related bidding contractor does not comply with these requirements, they shall be responsible for any and all additional costs associated with the changes required by other trades.

- 1.7 WARRANTY GUARANTEE
 - A. All work and materials are subject to the general warranty as described in the General Conduct of the Contract and in Division 1, General Requirements.

PART 2 - PRODUCTS

- 2.1 DUCTWORK (SINGLE WALL, SQUARE AND RECTANGULAR)
 - A. All ductwork shall be fabricated in accordance with SMACNA "AVAL Duct Construction Standards - Metal and Flexible" latest Edition except as described below. The minimum thickness of metal ductwork is 26 gauge. Fabrication requirements shall be based on ductwork subjected to positive or negative pressures of 3" W.G. Ductwork systems shall be sealed to SMACNA "Seal Class "A" Standards. Alternatively, "Ductmate" System 4. can be used in accordance with manufacturer's specifications. Drive slip joints are not permitted.
 - B. Rectangular ducts for 3" W.G. or less, positive or regative pressure shall be per SMACNA Table 1-6. Longitudinal seams shall be Pittsburgh Lock Type 1-1 per SMACNA Figure 1-5. Transverse joints shall be standing seam type T -15 per Figure 1-4
 - 1. In the event that material size is not compatible with duct size and segmenting must be utilized to fabricate duct, use SMACNA Figure 1-5, seam L-4 (Standing Seam).
 - C. Joints:
 - 1. Per SMACNA Transverse spint heinforcement Table 1-12, only joints T -22, T -25a, T -25b and Proprietary slip on flagges will be acceptable.
 - 2. Joints T -25a and T ob that have stress fractures from bending will not be accepted.
 - 3. All joints will have butyl gasket 3/16" thick by 5/8" wide installed per manufacturers installation neuronons.
 - D. Ductwork systems for this standard shall be galvanized sheet steel, commercial quality of lock forming grade conforming to ASTM coating standards A-525 or A-527 with coating of design and G-60. For corrosive or moist conditions, use coating designation G-90.

Where the outer surface of the duct is exposed in finished spaces and is <u>not</u> scheduled for insulation, duct material shall be galvannealed, suitable for field painting by the General Contractor.

The size and configuration of each duct shall be indicated on design drawings. Where thicker sheets or different types of materials are required, they shall be specified on the design drawings or in the project specifications.

- 2.2 DUCTWORK (SINGLE WALL, SPIRAL ROUND)
 - A. Design Pressure: 3"
 - B. Leakage: All ductwork shall meet SMACNA Class "A" leak standards.

- C. Fabrication:
 - 1. Gauges, reinforcing angles, seams, joints, fabrication methods, installation methods and practices, duct reinforcement, fabricated dampers and devices installed in duct system, fittings, etc., shall conform to the latest editions of SMACNA standards for construction in accordance with requirements indicated in these specifications.
 - 2. Minimum metal gauges shall be 26 gauge (.019). Follow SMACNA Table 3-2A for Positive pressure and Table 3-2B for Negative pressure.
 - 3. Where the outer surface of the duct is exposed in finished spaces and is <u>not</u> seneduled for insulation, duct material shall be galvannealed, suitable for field painting by the General Contractor.
- D. Joints:
 - 1. Duct up to 36" diameter Male/Female beaded slip joint similar to SMACNA Figure 3-2, joint RT-1 or RT-5, as long as it meets the criteria for the system design pressure. Fittings shall be undersized to fit into spiral duct. All joints shall be recared with a minimum of 4 screws on each duct section (equally spaced). Seal joint with an approved sealant compound, continuously applied prior to assembly of joint and arter fastering, making certain that the majority of the sealant resides on the interior of the joint
 - 2. In lieu of beaded slip connections or Vanston angle ring connections (the above-mentioned joints), there are proprietary connections that may be used, as long as they meet the pressure criteria set forth in this specification.

2.3 DOUBLE WALL ROUND DUCT

A. Double wall round duct will be constructed of solid inner liner, a 2-inch layer of fiberglass insulation, and an outer pressure shell. Duct will be spiral lock seam construction provided in standard 10 foot lengths. It will be fabricated from galvanized steel meeting ASTM-A527 standards, and in accordance with the following guidelines:

| Inner Diameter | Cater Shell Min. Gauge | | |
|----------------|-------------------------------|-------------|---------------|
| (Inches) | (2-10 inches WG) | Inner Gauge | Fitting Gauge |
| 3-8 | 26 | 26 | 24 |
| 9-12 | 26 | 26 | 24 |
| 13-24 | 24 | 26 | 24 |
| 25-34 | 22 | 26 | 24 |
| 36-12 | 20 | 26 | 22 |
| 44-58 | 18 | 22 | 22 |
| 60-82 | 18 | 22 | 20 |

S

For 2-inch insulation, the outer shell will be 4 inches larger than the inner liner nominal limension. When a perforated liner is specified, perforations will be 3/32 inch in diameter with an overall open area of 23 percent. Insulated duct will have a maximum thermal conductivity (k) factor of 0.27 Btu/Hr./SF/inch at 75 deg. F mean ambient temperature.

- C. Follow SMACNA Table 3-2A for Positive Pressure and Table 3-2B for Negative Pressure.
- D. Joints:
 - 1. Duct up to 16" diameter, Male/Female beaded slip joint similar to SMACNA Figure 3-2, Joint RT-1 or RT-5, as long as it meets the criteria for the system design pressure. Fittings

shall be undersized to fit into spiral duct. All joints shall be secured with a minimum of 4 screws on each duct section (equally spaced). Seal joint with an approved sealant compound, continuously applied prior to assembly of joint and after fastening, making certain that the majority of the sealant resides on the interior of the joint. The inner liner slip connection shall be a minimum of 2" longer than the outer wall slip connection. Sealant is not required on the inner wall slip.

- Duct 17" 60" diameter, Accuflange@ joint as manufactured by Accuduct, Inc. Instabation is to be per manufacturer's recommendations. Companion angle Vanstone with full face, gaskets having bolt holes punched through prior to insertion of bolts may be used in hou of Accuflange. Gasketing to be 1/8" thick. Joint is per SMACNA Figure 3-2, joint KT 2 and RT -2A.
- E. Finish: Where the outer surface of the duct is exposed in finished spaces and is not scheduled for insulation, duct material shall be galvannealed, suitable for field painting by the General Contractor.

Type RT-2 or RT-3 joints are acceptable for concealed work or exposed in Mechanical Rooms. For exposed work in finished spaces, use RT-3 only for a clearer looking joint.

2.4 DOUBLE WALL DUCTWORK - SQUARE AND RETAIL

- A. Provide double wall construction for all roof-mounted ductwork.
- B. Construction (Outer Walls): Aluminum ductwork shall be two (2) B&S gauges heavier than specified for the equivalent width steel ductwork. Bracing, supports and joints shall be as specified for steel ductwork. Inner wall shall be generalized steel.
- C. Joints: Shall be companion angle, gastened and watertight.
- D. Insulation:
 - 1. Fiberglass rigid duct beard bonded with resins, 6 pound density, aluminum foil reinforced facing with fiberglass scrine laganated to kraft, 2" thick.
 - 2. Thermal Conductive, 0.3 Btu/Hr./S.F. per inch at 75 deg. F.
 - 3. Overlap but joints 2" and seal with manufacturer recommended pressure sensitive tape Insulation such be banded with 18 gauge, 1/2" wide bands on 2'-0" centers. In addition, on ductwirk 2-" and wider, support bottoms with mechanical fasteners; i.e., stick pins on 18" center.
 - 4. Musufacturers: Manville International, Inc., Certain-Teed, Owens-Corning.
- 2.5 FLEXIBLE AIR DUCT

Insulated flexible air duct shall be non-metallic. Air duct shall comply with the latest NFPA Bulletin No. 90A and be labeled as Class 1 Air Duct, U.L. Standard No. 181.

Air ducts shall be suitable for working pressure of not less than plus 10.0 and minus 0.5 inches of W.G.

C. Non-metallic air duct shall be two element spiral construction composed of a corrosion resisting metal supporting spiral and a vinyl coated fiberglass base fabric and shall be mechanically interlocked together.

Tetra Tech 200-15704-17001

- D. Insulation shall be 1-1/2" thick fiberglass flexible blanket with vapor barrier outer jacket of polyethylene or reinforced mylar. Maximum thermal conductance of 0.23 Btu/Hr./SF/Inch at 75 deg. F temperature.
- E. Approved manufacturers shall include the Wiremold Company, Flexmaster USA, Owens-Corning, Thermaflex Flex Vent.
- 2.6 FLEXIBLE CONNECTIONS
 - A. Required between ductwork and suction and discharge connection of all fans and air hand
 - B. Material: Woven fiberglass with mounting hardware tested in accordance with UL Science 181, listed and labeled as Class 0 or 1.
 - C. Manufacturer: Ventfabrics, Inc., Durodyne, Dynair, Ductmate Pro Flex.

2.7 DAMPERS

- A. Provide where indicated and required to control flow of air and balance system.
- B. Round dampers shall be single blade, molded synthetic bearings at each end, 20-gauge galvanized steel, adjusting quadrant and locking device. Round dampers that be Ruskin Model MDRS25.
- C. Rectangular and square dampers shall be opposed blade within 16-gauge galvanized steel channel frame with corner brace, 16 gauge galvanized steer blades; molded synthetic bearings and hex steel shafts, exposed or concealed linkage, adjustable qualrant and locking device.
- D. Approved Manufacturers: Ruskin Basis of Desig. Model #MD35, Arrow, Nailor-Hart, Pottorff, Lloyd Industries, Inc., Cesco Products, Leuves & Dampers, United Enertech.

2.8 FIRE DAMPERS

- A. Fabricate in accordance with NPA 01 and UL 555.
 - 1. Ceiling Dampers: Galvaniz d seel, 22-gauge frame and 16-gauge flap, two layers 0.125inch ceramic fiber of top size with locking clip.
 - 2. Horizontal Dampers: Galvanized steel, 22-gauge frame, stainless steel closure spring, and lightweight dea retardant, non-asbestos fabric blanket.
 - 3. Curtain 1 yee Dampers: Galvanized steel with interlocking blades. Provide stainless steel closure springs and latches for closure under air flow conditions. Configure with blades out of air scean except for 1.0-inch pressure class ducts up to 12 inches in height.

Multiple Blade Dampers: 16-gauge galvanized steel frame and blades, oil-impregnated bronze or stainless-steel sleeve bearings and plated steel axles, $1/8 \times \frac{1}{2}$ inch plated steel concealed linkage, stainless steel closure spring, blade stops and lock.

Fusible Links: UL 33, separate at 160 deg. F with adjustable link straps for combination fire/balancing dampers.

- Fire dampers of the applicable rating shall be provided at all locations where ductwork penetrates fire-rated walls, ceilings, or floors. Refer to Architectural Drawings.
- C. Manufacturers: Air Balance, Inc., Ruskin, Louvers & Dampers, Prefco, Phillips-Aire, Metalaire, Pottorff, Lloyd Industries, Inc., Cesco Products, Greenheck, United Enertech.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

2.9 AIR DIFFUSERS, REGISTERS AND GRILLES

- A. Air diffusing terminals shall be provided in duct runs on drawings. The diffusers shall properly and uniformly distribute the design air quantity with no objectionable drafts, while maintaining not more than 50 F. P. M. velocity in the occupied portion of the space.
- B. Ceiling Diffusers:
 - 1. Square Louvered Diffuser Face:
 - a. Square housing, welded steel construction core of square concentric louvers removable at face of diffuser, round duct connection, with borders suitable for lay-in comp tile application.
 - b. Diffuser Patterns: Fixed louver face for 1, 2, 3, or 4 direction air flow accection indicated on drawings. Each diffuser shall be provided with adjustable control solds.
 - c. Finish: Matte white finish.
 - d. Manufacturers: Price Model SMD
- C. Registers & Grilles:
 - 1. Registers and grilles shall be steel construction, fixed single deflection type, with clips and/or flange holes and screws (as required by Architectural finishes) to secure registers to ceiling construction. Face bars shall be inclined 30 legree. Registers and grilles shall be factory primed and painted with a baked-on white example in the state of the state of the state of the state.
 - 2. Ceiling Return Register (CR):
 - a. Ceiling registers shall have a periorated face with 3/16-inch diameter holes on 1/4-inch staggered centers and no less than 51 percent free area. Perforated face shall be aluminum according to the nodel selected. The back pan shall be one piece stamped heavy gauge steel of the sizes and mounting types shown on the plans and outlet schedule.
 - b. The finish shall be #26 white. The finish shall be a baked on anodic acrylic paint, with a pencil hadness of HB to H. Inside of back pan shall be painted flat black.
 - c. Price Mo. el PDDR
 - 3. Return Exhaust and Transfer Grilles (RG, EG & TG):

orilles shall be available parallel to the long dimension of the grille. Construction shall e of steel with a 11/4-inch wide border on all sides. Screw holes shall be countersunk for a neat appearance. Corners shall be welded with full penetration resistance welds.

- . Deflection blades shall be firmly held in place by mullions from behind the grille and fixed to the grille by welding in place. Blade deflection angle shall be available at 35°.
- c. The finish shall be #26 white. The finish shall be a baked on anodic acrylic paint, with a pencil hardness of HB to H.
- d. Price Model 535 FL(EG), 535 FS (RG), 535 (RG/EG)
- 4. Ceiling Return Filter Grille (CR):

- a. Return filter grilles shall be of size and mounting type as shown on the drawings and schedules.
- b. Return grilles shall provide minimum free area of 90%.
- c. Borders shall be constructed of heavy extruded aluminum with countersink holes or frame suitable for ceiling finish in each room.
- d. The four corners shall be interlocked and mechanically staked to form a rigid frame
- e. Aluminum grid core shall have $\frac{1}{2} \times \frac{1}{2} \times \frac{1}{2}$ inch openings
- f. Return grilles shall be provided with a filter frame that will accommodate a standard 1inch thick disposable filter to fit the specified duct size. Filter shall be grille module size minus 4 inches. Filter capacity shall be as scheduled on the drawinger
- g. Return grille finish shall be white powder coat.
- h. Price Model 535FF-TB
- D. Manufacturers: Provide diffusers, registers and grilles of one of the following:

| Anemostat | Price |
|------------|-------------------|
| Carnes Co. | Titus |
| Krueger | Tuttle & Bailey |
| Metalaire | Nailor Industrice |

- 2.10 PREFABRICATED ROOF CURBS AND EQUIPMENT SUPPORTS
 - A. Factory fabricated by the manufacture of the respective roof-mounted equipment when available and capable of meeting the following requirements:
 - 1. Thermally and acoustically invated, rubber isolating pads.
 - 2. Built to suit slope obvoof and type of roofing; i.e. standing metal seam with integral cant strip and flashing extension.
 - 3. 8" to 11" height unless otherwise indicated.
 - 4. Support rails shall be aluminum, or sheet steel, with continuous wood nailer and removable counter lashing.
 - B. Curbs shall be a product of a custom manufacture in the following cases:

Curbs as specified are not available from the respective equipment manufacturer.

Piping or ducts penetrating roof.

- Prefabricated equipment supports are required.
- 4. Step flashing assembly, EPDM for normal use and silicone for pipe temperatures above 200°F stainless steel clamp, suitable for single or multiple pipes.
- C. Pipe supports shall be a product of a custom manufacture equal to Pipe Prop as made by JMB Industries, or Anvil International Haydon H-Block.

D. Manufacturers: Pate, Shipman, Custom Curb, Portals Plus, Lloyd Industries, Inc., PHP Systems/Design.

2.11 VAV UNITS WITH REHEAT

- A. Terminal units designated shall be pressure independent type and of sizes shown on drawings or terminal unit schedule. Units shall have factory catalog performance ratings which conform to CFM, Heating Capacity, Static Pressure discharge and radiated sound power and attenuation designated.
- B. Cabinets shall be constructed of not lighter than 22 gauge, zinc-coated steel with factory applied enamel paint finish. Internal surfaces shall be acoustically and thermally insulated with inch glass fiber material surface-treated to prevent erosion and having U.L.181 approval meeting NFPA 90A.
- C. Heating coil shall be copper tube, aluminum fin. Coil performance data shall be based on test in accordance with AHRI Standard 410.
- D. Air volume damper shall be constructed to prevent air leakage in excess of 2% of rated air quantity at 1" inlet static pressure.
- E. Automatic damper operators shall be factory installed and thoroughly tested for proper performance.
- F. Performance of units shall be based on tests conjuctation accordance with AHRI Standard 880 Certification Program.
- G. Units shall be as manufactured by Trane, Cares, Carrier, Titus, Nailor-Hart Industries, Metalaire, Tuttle & Bailey.
- 2.12 LOUVERS
 - A. All wall louvers for intake and exhault shall be stationary stormproof type.
 - B. Construction shall be of extruled aluminum with 0.081-inch-thick blades and frames and all fastening shall be aluminum of stainless steel.
 - C. An aluminum expanded metal bird screen with frame shall be secured to the rear face of the louver assembly
 - D. Depth of the lower frame shall be 6" unless otherwise indicated.
 - E. Performance Rating:
 - Free Area: Louvers shall have a minimum of 50% free area based on a 48-inch-high by 48-inch-wide size.
 - Air Performance: Not more than 0.08 inch WG static pressure drop at 700 FPM free area intake velocity.
 - 3. Water Penetration: Maximum of 0.01 ounces per square foot of free area at an air flow of 750 feet per minute free area velocity when tested for 15 minutes.
 - F. The surface areas shall be factory Kynar 500 baked finish of color selected by Architect.

- G. Manufacturers: Airstream Products Co., Air Balance, Inc., Carnes Co., Arrow Co., Empco, Pottorff, Cesco, Lloyd Industries, Inc., Ruskin, Louvers & Dampers, United Enertech, NCA Manufacturing, Inc.
- 2.13 DUCT ACCESS DOORS (Interior Locations)
 - A. SMACNA standard duct access doors shall be fabricated with 22-gauge galvanized steel door and frame with double wall construction.
 - 1. Doors shall be fabricated of aluminum when installed in aluminum ductwork and steel to match special duct systems.
 - B. Continuous piano type hinge, same material as door.
 - C. Latches shall be sash type locks equal to Ventlock 100 latches.
 - 1. Doors 16" and under shall have one latch.
 - 2. Doors over 16" shall have two latches.
 - D. Door seals shall be foam gasket material continuously bonded to perhapter of door frame.
 - E. Door insulation shall be 1" thick fiberglass, minimum 1.5 ocf lensity.
 - F. Doors shall be able to withstand 3" W.C. static pressure up to 12" x 12" in size; 2" W.C. above that size.
 - G. Makes: Cesco Products, Karp Co., Nailor-Hat maustries, Pottorff, Lloyd Industries, Inc., Ductmate Industries, Inc.
- 2.14 ROOF-MOUNTED INTAKE/EXHAOS AVANTILATORS
 - A. Heavy gauge aluminum construction.
 - B. Hinged hood.
 - C. Hood underside insulated with N fiberglass.
 - D. Aluminum insect screen.
 - E. Provide 12-inclubign insulated roof curb with blocking to secure equipment, and integral cant to accommodate configuration of roofing material/membrane.
 - F. Manufacturys: Venn Ventilator Co., Carnes Co., Greenheck, Loren-Cook or Acme.
- 2.15 EABRIC AIR DISPERSION DUCTWORK
 - reduct shall be constructed of a coated woven fire-retardant fabric complying with the following physical characteristics:
 - Type: Verona
 - 2. Configuration: Standard: round
 - 3. Fabric Construction: Plain polyester weave.
 - 4. Coating: Porous
 - 5. Weight: 5.2 oz. per square yard
- 6. Permeability: 2 cfm per square foot @ 0.5" WC.
- 7. Color: Standard color as selected by the Architect/Owner
- 8. Warranty: 5 years on products for the fabric system.
- 9. Temperature Range: 0 degrees F to 180 degrees F
- 10. Fire Retardancy: Classified by Underwriters Laboratories in accordance with the 25/50 has spread/smoke developed requirements of NFPA 90-A.
- B. Systems Fabrication Requirements:
 - 1. Air dispersion accomplished by round vent, and consist of open orifices rather han a mesh style vent to reduce maintenance requirements (common to mesh style).
 - 2. Size of and location of vents shall be specified and approved by manufacturer.
 - 3. Inlet connection to metal duct via fabric draw band with uncler patches supplied by manufacturer. Anchor patches shall be secured to metal duct via. zip screw fastener supplied by contractor.
 - 4. Inlet connection includes zipper for easy removal / m interance
 - 5. Lengths to include required zippers as specified by manufacturer.
 - 6. System to include Adjustable Flow Devices to balance turbulence, airflow and distribution as needed. Flow restriction device shall include ability to adjust the airflow resistance from 0.06 0.60 in w.g. static pressure.
 - 7. End cap includes zipper for easy mantenance.
 - 8. Fabric system shall include connectors to accommodate suspension system listed below.
 - 9. Any deviation from a stranght run shall be made using a gored elbow or an efficiency tee. Normal 90 degree elbows are 5 gores and the radius of the elbow is 1.5 times the diameter of the DuctSox.
- C. Design Parameters.
 - 1. Fabric air different shall be designed from 0.25" water gage minimum to 3.1" maximum, with 0.5" at the sum ard.
 - 2. Eabric an affusers shall be limited to design temperatures between 0 degrees F and 180 degrees F.
 - Design CFM, static pressure and diffuser length shall be designed and approved by the manufacturer.
 - Do not use fabric diffusers in concealed locations.
 - 5. Use fabric diffusers only for positive pressure air distribution components of the mechanical ventilation system.
- D. Suspension Hardware:
 - 1. Internal Hoop System: Provide a factory fabricated retention system consisting of an internal 360° hoop system spaced on maximum 5' centers. Each hoop shall be fabricated of lightweight aluminum ring and tubing with negligible effect on airflow static resistance. The

rings located at the inlet and end of run shall include tensioning anchor clips to secure the fabric to the hoop system. Sizes shall include 8" to 36" diameter in 2" increments. The system shall be installed with a one row suspension system located 1.5" above top dead center of the fabric duct system. System attachment shall be either cables or u-track using gliders spaced 12" on center.

E. Manufacturer: Duct Sox by Fabric Air Dispersion Products, Fabric Air Inc., or KE Fibertec

PART 3 - EXECUTION

- 3.1 DUCTWORK
 - A. Dimensions on drawings are inside dimensions. Sheet metal dimensions shall be metaled to suit thickness of acoustic duct lining, if applicable. Ductwork that is lined with acoustic lining is or is not insulated.
 - B. Ducts shall be concealed unless otherwise indicated.
 - C. Changes in direction shall be made with radius bends or turning van
 - D. Supports shall be galvanized steel.
 - E. Locate ceiling air diffusers, registers, and grilles on "Reflected Ceiling Plans". Unless otherwise indicated, locate units in center of acoustical ceiling modules
 - F. Do not install ductwork directly above any electrical equipment.
 - G. Ductwork shall be supported per SMACNA Standards except as follows:
 - 1. Rivet or screw to side of duct when using hat strap hangers. Rivet or screw to bottom of duct when using trapeze hangers.
 - 2. Extend hangers down the side of he duct at least 9"; pass hangers under ducts less than 9" deep.
 - 3. Space hangers not plore that 8 on centers for ducts up to 18" wide and 4' on centers for ducts over 18" wide.
 - 4. Wire hangers are not acceptable.
 - 5. Support destwork from building structure with expansion bolts, rods, steel angles or channels installed to neet existing or new building conditions.
 - 6. Krilling into the roof deck is not permitted.
 - Driving nails into anchors is not permitted.
 - An Flow Control:
 - Major take-offs: Install volume control dampers.
 - 2. Branches: Install volume control dampers in all branches and at tap in branch take-off connections.
 - 3. Elbows: Use unvaned elbows with throat radius equal to width of duct and full heel radius; provide turning vanes where full throat and heel radius are not possible.
 - 4. Transitions: Make transitions in ducts as required by structural or architectural interferences.

- a. Proportion airways to compensate for any obstructions within duct.
- b. Avoid dead ends and abrupt angles.
- c. Do not exceed 15 degrees slope on sides of transitions.

3.2 LOUVERS

- A. Locate and place louver units level, plumb and at indicated alignment with adjacent work.
- B. Use concealed anchorages where possible.
- C. Provide perimeter reveals and openings of uniform width for sealants and joint fillers
- D. Repair damaged finishes so no evidence remains of corrective work. Return items that cannot be refinished in the field to the factory, make required alternations and refinisher the unit or provide new units.
- E. Protect galvanized and non-ferrous metal surfaces from corrosion of galennic action by applying a heavy coating of bituminous paint on surfaces that will be in contact with concrete, masonry or dissimilar metals.
- 3.3 FLEXIBLE AIR DUCT
 - A. When flexible duct is used for final connection between ducemains on branches and diffusers on registers. The maximum length of flexible ductwork shall be 5'-0" in length.
 - B. Flexible ductwork shall be properly hung at the ap conlar in order to prevent eventual wear and damage to the flexible duct.
 - C. The ceiling tile system should not be considered a support on which to lay flexible duct. Refer to SMACNA Standards for proper installation
- 3.4 DUCT SYSTEM LEAK SEALING
 - A. Joints in duct systems at duct heaters, air monitors, fire dampers, sound traps, supply air terminals including air handling acht natures, shall be sealed to prevent air leakage.
 - B. All duct joints are pears in medium pressure and high-pressure duct systems shall be sealed to SMACNA Seal Clar." A" Standards to prevent air leakage.
 - C. In the event then is in excess of 5% air leakage indicated in low pressure duct systems, it shall be the Contractors responsibility to seal the duct system. The amount of sealing necessary shall be that required to obtain the design air quantity at each terminal.
 - D. Duct sealing shall be by means of high velocity duct sealants such as Hardcast and/or Neoprene g skets. Type of sealant and method of application shall conform to recommendations in SMACNA high velocity duct construction standards.

DUCTWORK TESTING

The following ductwork shall be pressure leak tested:

- 1. Supply ductwork
- 2. Return ductwork
- 3. Outside air intake ductwork

- B. All tests shall be conducted in accordance with AABC National Standards.
- C. Ducts to be tested at 100% maximum of static pressure before any duct is insulated externally and concealed in accordance with SMACNA Standards.
- D. Calculate the allowable leakage using leakage factor of 5% of Design Air Flow.
- E. Select a limited section of duct for which the estimated leakage will not exceed capacity of test apparatus.
- F. Connect the blower and flow meter to the duct section and provide temporary openings of the ductwork.
- G. Start the blower motor with the inlet damper closed. Increase pressure until the required level is reached.
- H. Read the flow meter and compare the leakage in cfm. Reading should be % or less of design flow for the duct segment being tested.
- I. If reading is more than 5% of design flow, depressurize duct, reparableaks and retest until 5% or less of design flow is obtained.
- J. Complete test reports and obtain Owner's witness signature
- K. Remove all temporary blanks and seals.
- L. Warning: Do not overpressure duct.

3.6 EQUIPMENT

- A. Test apparatus shall consist of an annow measuring device, flow producing unit, pressure indicating devices and accessories necessary to connect the metering system to the test specimen.
- B. The Contractor conducting texts shall arrange for or provide all temporary services, all test apparatus, all temporary seals and all qualified personnel necessary to conduct the specified testing.
- C. Test apparatus shall be accurate within plus or minus 7.5% at the indicated flow rate and test pressure and shall have calibration data or a certificate signifying manufacture of the meter in conformance with the ASME Requirements for Fluid Meters. Verification of above, to be supplied to Owner opon request.
- D. Pressure differential sensing instruments shall be readable to 0.05" scale division for flow rates below 10 cfm or below 0.5" w.g. differential. For flows greater than 10 cfm scale divisions of 9.1" are oppropriate. U-tube manometers should not be used for reading less than 1" of water.

Incruments must be adjusted to zero reading before pressure is applied.

TEST REPORT

Log the project and system identification data.

- B. Enter the fan CFM, the test pressure, and the leakage class specified by the designer.
- C. Enter an identification for each duct segment to be tested.
- D. Calculate the allowable leakage factor. Enter this number on the report for each test segment.
- E. Conduct and record the field tests. If the sum of the CFM measured is less than or equal to the

AIR DISTRIBUTION & ACCESSORIES – HVAC 230600-14

sum of the allowable leakage, the test is passed. Record the date(s), presence of witnesses and flow meter characteristics.

- F. Maintain a mechanical duct plan of all tested duct segments. Plan to include duct segment identification and dates tested.
- G. Test reports shall be submitted as required by the project documents.
- 3.8 LABELING
 - A. At all fire damper, smoke damper and combination fire/smoke damper locations, access doers in ductwork shall be identified with a permanent placard of red-white-red laminated commercial grade plastic construction, minimum one-half inch high capital letters, reading. FIRE DAMPER", "SMOKE DAMPER", "FIRE/SMOKE DAMPER" as appropriate for the installation. Attach securely to face of access door with brass screws at each corner, sealed airtight.

END OF SECTION 230600

Reputies



SECTION 230605

FANS

PART 1 – GENERAL

1.1 RELATED DOCUMENTS

- A. The general provisions of the contract, including the conditions of the contract (General Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the volume specified in this section.
- B. Refer to Section 230200 for HVAC General Provisions
- C. Refer to Section 230210 for HVAC Basic Materials & Methods.
- 1.2 DESCRIPTION OF WORK
 - A. This Section includes labor, material, equipment and supervision to provide a complete air distribution system as specified herein and as shown on drawings.
 - 1. Centrifugal Belt Drive Cabinet Fan
- 1.3 REFERENCE STANDARDS
 - A. Refer to Section 230200 for a general description of rechrements applying to this section.
 - B. Requirements established within the portions of the roject Manual titled Division 1, General Requirements, are collectively applicable to the way of this section.
 - C. IMC (International Mechanical Code)
 - D. SMACNA (Sheet Metal and Air Conditioning Contractors National Association, Inc.)
 - E. American Society of Heating, K frigerating and Air Conditioning Engineers' recommendations in ASHRAE Guide shall apply to his work.
 - F. UL (Underwriter's Labora ories, Inc.)
 - G. NFPA 90A shall apply to this work.
 - H. State Fire Prevention Regulations.
- 1.4 QUALITY ASSURANCE
 - A. Befer to Section 230210 for a general description of requirements applying to this Section.
- 1.5 SUBMITTALS

Submit shop drawings and product data in accordance with Section 230200.

Submit the following:

- 1. Shop drawings of all sheet metal. Indicate all steel, piping, conduit, and Architectural/Structural features to demonstrate complete coordination. Scale shall not be less than 1/4" = 1'-0".
- 2. Manufacturer's literature and performance data of all equipment and devices.

1.6 SUBSTITUTIONS

A. The listed equivalent or substituted manufacturers along with the bidding related contractor shall be completely responsible to comply with all requirements on all contract documents and as described within the specifications. This shall include, but shall not be limited to space requirements, code clearances, the type, horsepower, capacities, number and size of services required from other trades including all required ancillary items furnished and installed by other trades. If the manufacturer or related bidding contractor does not comply with these requirements, they shall be responsible for any and all additional costs associated with the changes required by other trades.

1.7 WARRANTY/GUARANTEE

A. All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, General Requirements.

PART 2 – PRODUCTS

- 2.1 CENTRIFUGAL BELT DRIVE CABINET FAN
 - A. Fan shall be centrifugal belt driven in-line type. Fan housing shall be on the square design constructed of heavy gauge galvanized steel and shall include square duct moduling collars.
 - B. Fan shall be provided with removable service doors located perpendicular to the motor mounting panel. The service doors must be of sufficient size to permit easy access to all interior components.
 - C. Fan wheel shall be of the aluminum backward inclined, centrifugal type. Wheels shall be dynamically and statically balanced and shall overlap thereput to let venturi for maximum performance.
 - D. Motor and drives shall be isolated from the xir stream. Motors shall be of the heavy-duty type with permanently lubricated, sealed ball bearings. Wheel shaft shall be ground and polished shafting mounted in heavy duty permanently realed pillow block bearings. Drives shall be sized for a minimum of 165% of driven he sepower. Pulleys shall be of the fully machined cast iron type, keyed and securely attached to the wheel and motor shaft. Motor pulleys shall be adjustable for final system balancing.
 - E. Flexible wiring leads shall be provided from the fan motor to an external mounted junction box and disconnect switch permitting access for service without disconnecting the field wiring. All fans shall bear the AMCA Varuned Ratings Seal for both air and sound performance.
 - F. Manufacturers: Jenn Ventilator, Greenheck, Carnes, Loren-Cook, American Coolair/ILG, Breidert, Hartzen.

PART 3 - EXECUTION

- 3.1 **NNS, EQUIPMENT AND ACCESSORIES**
 - Install in accordance with manufacturer's details and instructions.

Power wiring shall be part of the work of Division 26.

- C. Perform field mechanical balancing in accordance with Section 230950: TESTING AND BALANCING OF MECHANICAL SYSTEMS.
- D. Install units in accordance with manufacturer's installation instructions. Install units plumb and level, firmly anchored in locations indicated, and maintain manufacturer's recommended clearances.

Ά.

Q

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

E. The Mechanical Contractor shall own as a part of his work, the following:

Provide one (1) additional drive set, if necessary, to obtain final design balancing requirements. The Mechanical Contractor shall coordinate with Balancing Firm and equipment manufacturer for drive selection, including belts and pulleys.

END OF SECTION 230605

Reput

Tetra Tech 200-15704-17001

<



SECTION 230725

TERMINAL HEATING UNITS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. The general provisions of the contract, including the conditions of the contract (General Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the root specified in this section.
- B. Refer to Section 230200 for HVAC General Provisions
- C. Refer to Section 230210 for HVAC Basic Materials & Methods.
- 1.2 DESCRIPTION OF WORK
 - A. This Section includes work necessary and/or required and materials an equipment for construction of a complete system. Such work includes, but is not limited to the following:
 - 1. Unit Heaters
 - 2. Electric Cabinet Heaters
- 1.3 REFERENCE STANDARDS
 - A. Refer to Section 230200 for a general description of requirements applying to this section.
- 1.4 QUALITY ASSURANCE
 - A. Refer to Section 230210 for a general description of requirements applying to this Section.
- 1.5 SUBMITTALS
 - A. Submit shop drawings in accordance with Section 230200.
 - B. Submit shop drawings addes riptive date for all equipment specified in this section.
- 1.6 SUBSTITUTIONS
 - A. The listed equivalent or substituted manufacturers along with the bidding related contractor shall be completely responsible to comply with all requirements on all contract documents. This shall include, but no time d te, space requirements, code clearances, the type, horsepower, capacities, number and size of vervices required from other trades, including all required ancillary items furnished and installed by other trades. If the manufacturer or related bidding contractor does not comply with these requirements, this Contractor shall be responsible for any and all additional costs associated with the charges required by other trades.

WARRANTY/GUARANTEE

All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, General Requirements.

PART 2 – PRODUCTS

2.1 UNIT HEATERS

A. Propeller type, direct drive, resilient-mounted motor, arranged for horizontal discharge,

double-deflection louvers.

- B. Tested at 400 psig hydrostatic and 200 psig air under water.
- C. Enclosure shall be steel, cleaned, phosphated, primed and finished in baked enamel.
- D. Manufacturers: Airtherm Manufacturing Co., American Air Filter, Embassy Industries, Daikir McQuay, Modine, Rittling, Sterling, Trane, Vulcan.
- 2.2 ELECTRIC CABINET HEATERS
 - A. Provide cabinet heaters including chassis, heating elements, fan and motor designed for ether ressed mounting within 2'x2' ceiling grid, or within wall construction.
 - B. Chassis: Galvanized steel wraparound structural frame with edges flanged.
 - C. Power disconnect switch, 30 AMPs, 600 volts, 3 phase.
 - D. Cabinet: Horizontal recessed model, heavy gauge, four sided overlap from purel with stamped steel louver air openings. Clean cabinet parts, phosphatize and coat with taked-on enamel finish. Color: white.
 - E. Coils: Steel fins, copper brazed, for a permanent bond to low watt density, steel sheathed tubular heating elements.
 - F. Grilles: Intake and outlet grilles shall be integral, tamped 15 deg. Deflection in ceiling trim ring.
 - G. Fans: Provide direct drive, five bladed aluminum.
 - H. Motors: Provide single speed impedance protected, totally enclosed motor with integral overload protection and motor cords to junctical box in unit.
 - I. Provide built-in fan delay control and advanatic thermal cutout.
 - J. Manufacturers: Q-Mark, Beno, PL Markel, Indeeco, Electromode.

PART 3 - EXECUTION

- 3.1 INSPECTION
 - A. Examine areas an conditions under which equipment is to be installed. Do not proceed with work until unsat sfactory conditions have been corrected in manner acceptable to Installer.
- 3.2 INSTALLATION OF UNIT HEATERS
 - A. Install houters in accordance with manufacturer's installation instructions.
 - Uncrate units and inspect for damage. Verify that nameplate data corresponds with unit designation.
 - Hang unit from building substrate.
 - Protect units with protective covers during balance of construction.
- 3.3 INSTALLATION OF ELECTRIC HEATERS
 - A. Install heaters in accordance with manufacturer's installation instructions.
 - B. Uncrate units and inspect for damage. Verify that nameplate data corresponds with unit designation.

- C. Hang unit from building substrate.
- D. Protect units with protective covers during balance of construction.
- E. Perform field mechanical balancing in accordance with Section 15950: TESTING AND BALANCING OF MECHANICAL SYSTEMS.

END OF SECTION 230725

2 BIDDING PU



SECTION 230730

TERMINAL HEATING AND COOLING EQUIPMENT

PART 1 – GENERAL

- 1.1 RELATED DOCUMENTS
 - A. The general provisions of the contract, including the conditions of the contract (General, Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the work specified in this section.
 - B. Refer to Section 230200 for HVAC General Provisions
 - C. Refer to Section 230210 for HVAC Basic Materials & Methods.
- 1.2 DESCRIPTION OF WORK
 - A. This Section includes work necessary and/or required and materials an equipment for construction of a complete system. Such work includes, but is not limited to the following:
 - 1. Unit Ventilators (Heating & Cooling)
- 1.3 REFERENCE STANDARDS
 - A. Refer to Section 230200 for a general description of requirements applying to this section.
 - B. Media type air filters shall comply with U.L. Standard 900
- 1.4 QUALITY ASSURANCE
 - A. Refer to Section 230210 for a general description of requirements applying to this Section.
- 1.5 SUBMITTALS
 - A. Submit shop drawings in accordance with Section 230200.
 - B. Submit shop drawings addes riptive date for all equipment specified in this section.
- 1.6 SUBSTITUTIONS
 - A. The listed equivalent or substituted manufacturers along with the bidding related contractor shall be completely responsible to comply with all requirements on all contract documents. This shall include, but no finned to, space requirements, code clearances, the type, horsepower, capacities, number and size of services required from other trades, including all required ancillary items provided by other trades. In the manufacturer or related bidding contractor does not comply with these requirements, this contractor shall be responsible for any and all additional costs associated with the changes required by other trades.

WARRANTY/GUARANTEE

All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, General Requirements. In addition, the following special guarantee applies:

ADDITION AND RENOVATIONS

PART 2 – PRODUCTS

- 2.1 UNIT VENTILATORS (HEATING & COOLING)
 - A. Provide where shown on the plans and in the schedules a 2-pipe, dual temperature, heating and cooling unit ventilator. The unit shall be acoustically treated to insure quiet operation.
 - B. The unit shall be listed by Underwriters Laboratory and Canadian Standards Association. Units shear the UL and CSA labels.
 - C. The units are to be shipped fully assembled so that only the wall sleeve need be removed durin installation process.
 - D. Cabinet:
 - 1. Cabinet shall be fabricated from galvanized furniture grade steel, and shall have all welded galvanized steel frame with baked enamel exterior finish. Color calcord by Architect from manufacturer's color chart. Top surface shall have textured charce all benze finish. Front access panels and top access door shall be provided with tamper resistant hex socket head threaded fasteners. Opening or removing the unit front panels shall not affect unit operation in order to facilitate testing and servicing. Internal sheet metal parts shall be constructed of galvanized steel.
 - 2. Removable discharge grille shall be constructed of continuous round edged steel bars to provide up to 10 degrees vertical deflection. Adjustable side deflection vanes shall be provided beneath the discharge grille to provide optimum lateral air cistribution.
 - 3. A 1/4" mesh screen shall be provided bereath the discharge grille to protect against objects being dropped through the discharge grille.
 - 4. The interior areas of the unit vertilator shall be insulated for sound attenuation and to provide protection against condensation of poisture on or within the unit. The unit shall be provided with an ultra-quiet sound package confisting of acoustically matched low speed fans, sound barrier insulation material (non-liberglas) adhered to the bottom underside of the unit top panel, sides of the fan section and sound absorbing insulation (non-fiberglass) material applied to the unit front panel.
 - E. Room Air Fans & Motor:
 - 1. The motor and fan assembly shall be constructed to assure maximum quietness and uniformity in air distribution. Fans shall be of aluminum and of double-inlet, forward-curved, centrifugal type. They shall be statically and dynamically balanced. Fan housings shall be constructed of steel. Fan and motor assembly shall be of the direct drive type. Motors shall be ECM-high efficiency motor type designed specifically for unit ventilator operation. Motors shall have an internal thermal overload device (auto-reset).

All components of the fan/motor assembly shall be removable from the front of the unit. The motor fan shaft shall have sleeve type bearings with precision tolerances and shall not require oiling more than once annually.

- Dampers:
 - 1. Each unit shall be provided with separate room air and outdoor air dampers. The room air damper shall be constructed of aluminum and shall be counterbalanced against back pressure to close by wind pressure. The outdoor air damper shall be fabricated from galvanized steel for rigidity and

to inhibit corrosion. Dampers shall be fitted with medium density, closed cell neoprene seals along all sealing edges. Damper bearings shall be made of nylon or other material which does not require lubrication.

- G. Wall Sleeve:
 - 1. The galvanized steel one-piece wall sleeve shall be set in a wall opening and butted up directly against the intake louver. Where it is not possible to butt the wall sleeve against the wall intake louver, the contractor shall fabricate and install in accordance with the manufacturer's instructions a horizontal sheet metal baffle between louver and wall sleeve which shall provide an artight separation between condenser discharge and return air to insure against recirculation of the seated air from the condenser fan discharge. The wall sleeve is to be permanently fastened in place in accordance with the manufacturer's instructions and shall be suitably sealed, callked or grouted by the contractor around the entire perimeter to prevent air leakage.
 - 2. The wall sleeve shall be fitted with an electrical junction box containing a main disconnect switch. All field wiring connections shall be made in this wall sleeve junction box.
- H. Dual Temperature Water Coil:
 - 1. Coil shall be aluminum plate fin and copper tube construction. Coil shall be suitable for 150 psi working pressure. Coil shall be provided with an accessible manual air vent at the high point of the coil and threaded drain plug at the low point of the coil.
 - 2. Coil shall be controlled by a modulating face and bypass damper. Coil shall have a 2-way, 2-position, end-of-cycle valve to shut off water now when heating or cooling is no longer required. Control valve shall be provided by the ATC Contractor.
- I. Factory Installed through Piping: End comparement piping shall consist of a factory piped assembly which shall include union connections at the supply and return coil connections, balancing and shut-off valves and control valve. The piping assembly is shipped loose and the installing contractor must reconnect it to the coil through and braze the remaining connections in accordance with the unit ventilator manufacturer's instructions.
- J. Filter: Provide two (2) set of a " throwaway filters, MERV 8.
- K. Temperature Centrols: Refer to Section 230900 Automatic Temperature Controls DDC
- L. Wall Intak Louver
 - 1. The walkint ke louver shall be supplied by the unit manufacturer and shall be of the heavy gauge (clear anodized) aluminum construction. The louver shall be of the vertical blade type and shall be divided in half horizontally across the intake to separate the condenser return air and condenser discharge air. All louvers shall be 28" high by 2-1/4" thick and suitable for both masonry and panel wall construction. The frame of the louver shall have weep holes along the face of the bottom edge, and a 1/2" square mesh galvanized screen shall be provided on the back of the wall intake louver. Provide 12 gauge lattice grille.

Accessories:

- 1. 6" deep false back for crossover piping.
- 2. 2" high subbase.

- N. Manufacturers: American Air Filter, Carrier, Magic Air, Daikin McQuay, Trane.
 - 1. Any listed equivalent manufacturer and the Mechanical Contractor shall be completely responsible to comply with all requirements on the contract documents. This shall include, but not be limited to, space requirements, code clearances, the type, horsepower, capacities, number and size of services required from other trades.

PART 3 - EXECUTION

- 3.1 INSPECTION
 - A. Examine areas and conditions under which equipment is to be installed. Do not proceed with work until unsatisfactory conditions have been corrected in manner acceptable to Installer.
 - B. Install in accordance with manufacturer's recommendations. Unit and all component sections shall be properly supported and vibration isolated.
- 3.2 INSTALLATION

Y FC

- A. Verify that coils, filters, motors, drives and other components are matched with the proper unit.
- B. Assemble unit components following manufacturer's instruction of bandling, testing and operating. Repair damaged galvanized areas, and paint in accuration with manufacturer's written recommendations.
- C. Vacuum clean interior of units prior to operation.
- D. Repair air leaks from or into casing that can be leaved or felt during normal operation.
- E. Perform field mechanical balancing in accordance with Section 230950: TESTING AND BALANCING OF MECHANICAL SYSTEMS.

ND OF SECTION 230730

SECTION 230760

AIR HANDLING EQUIPMENT

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. The general provisions of the contract, including the conditions of the contract (General Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the root specified in this section.
- B. Refer to Section 230200 for HVAC General Provisions
- C. Refer to Section 230210 for HVAC Basic Materials & Methods.
- 1.2 DESCRIPTION OF WORK
 - A. This Section includes work necessary and/or required and materials an equipment for construction of a complete system. Such work includes, but is not limited to the following:
 - 1. Blower Coil Units
 - 2. Air-to-Air Energy Recovery Unit
- 1.3 REFERENCE STANDARDS
 - A. Refer to Section 230200 for a general description of requirements applying to this section.
 - B. AMCA Standards 210 and 300 for fans.
 - C. ARI Standard 410, ASHRAE Standard 33 or Heating and Cooling Coils.
 - D. ASHRAE Standard 52.2 and U.L. Standard 900 for media type air filters.
 - E. AMCA Standard 511 and 5000 for Air Control Dampers.
 - F. ARI Standard 1060 and VAR AE Standard 84 for Air-to-Air Energy Recovery Equipment.
 - G. ARI Standard 260 and 430 for Air Handling Units.
- 1.4 QUALITY ISODIANCE
 - A. Refer to Section 230210 for a general description of requirements applying to this Section.
- 1.5 SUBMUTALS
 - A. Aubmit shop drawings in accordance with Section 230200.

Submit shop drawings and descriptive date for all equipment specified in this section.

SUBSTITUTIONS

The listed equivalent or substituted manufacturers along with the bidding related contractor shall be completely responsible to comply with all requirements on all contract documents. This shall include, but not limited to, space requirements, code clearances, the type, horsepower, capacities, number and size of services required from other trades, including all required ancillary items provided by other trades. If the manufacturer or related bidding contractor does not comply with these requirements, this Contractor shall be responsible for any and all additional costs associated with the changes required by

other trades.

- 1.7 WARRANTY/GUARANTEE
 - A. All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, General Requirements.
- PART 2 PRODUCTS
- 2.1 BLOWER COIL AIR HANDLING UNIT
 - A. General:
 - 1. Install each unit as shown on the plans in accordance with: The Manufacturer's recommendations, and all applicable national and local codes.
 - 2. UL approved.
 - 3. Covered by a 1-year limited parts warranty on the complete unit.
 - 4. In current production with published literature available to check performance, limitations, specifications, power requirements, dimensions, operation and appearance.
 - B. Unit Enclosure:
 - 1. Exterior panels of 18-gauge steel that have been divised with baked enamel to provide a long-lasting, quality appearance. Casing shall be insulated with one inch (1") 1-1/2 lb. density fiberglass, fire resistant and odorless material.
 - 2. Removable panels to provide easy access to the internal components for maintenance and service.
 - 3. A filter rack with space to accommodate 2" throwaway, pleated filters, 30% standard MERV 6 efficiency, flat arrangement with end-overs.
 - C. Fans shall be SWDI, forward curved, centrifugal blower type equipped with heavy duty adjustable speed V-belt drive. The fan blan shall be supported by heavy duty, permanently sealed ball bearings. Fans shall be dynamically coalarced.
 - D. Blower Motor shall:
 - 1. Be 1755 RW, open drip proof.
 - 2. Have inherent protection, permanently lubricated ball bearings and a service factor of at least 1.15.

Benactory mounted within the insulated cabinet and wired to a junction box, factory set to scheduled voltage.

V-nelt Drive shall:

- Be rated at least 25% above the required HP.
- 2. Permit the blower RPM to be adjusted to meet the exact CFM requirement of the system.
- F. Dual Temperature Water Coils:
 - 1. Main coils shall be two, four or six row chilled or hot water.
 - 2. Coils shall use aluminum fins mechanically bonded to seamless copper tubes, factory tested with 450 psig air under water.

AIR HANDLING EQUIPMENT 230760-2

- 3. Maximum operating conditions shall be 300 psig at 200F.
- 4. Sweat type connections shall be standard.
- 5. Include sloped drain pan, plastic or mastic coated steel, fully drainable, main and auxiliary connections of ³/₄" PVC or threaded pipe.
- G. Manufacturers: Carrier, Daikin McQuay, Trane, York/JCI.
 - 1. Any listed equivalent manufacturer and the Mechanical Contractor shall be completely responsible to comply with all requirements on the contract documents. This shall include but not be limited to, space requirements, code clearances, the type, horsepower, capacities, number and size of services required from other trades.

2.2 PACKAGED AIR-TO-AIR ENERGY RECOVERY UNIT

- A. Factory fabricated and assembled unit consisting of constant volume fairs, motors, and drive assemblies, plenum casing, filters, energy recovery wheel (with mote can drive), access doors and operating controls.
- B. Casing:
 - 1. Casing panels shall consist of dual wall, minimum 18-garge galvanized solid exterior skins and 22-gauge galvanized steel solid interior skins enclosing 2" taick 1.5 pcf fiberglass insulation with a minimum R-value of 10 which meets NFPA 90A and UL181 test standards. All metal-to-metal surfaces exposed to the weather shall be sealed artigbt with maximum leakage not-to-exceed 2% at external static pressure of 3" W.C.
 - 2. Removable panels shall be provided for every wrecovery wheels, and fans. The housing shall be supported by an all-welded epory-painted structural base. Lifting lugs shall be welded to the base. All frame and panel members shall be G90 galvanized steel.
 - 3. Access to all internal devices and sections shall be provided through removable, sealed doors. Access doors shall be constructed of the same materials as the unit casing.
 - 4. The unit's duct connections shall be arranged to require only minor ductwork offsets or transitions to the package beating cooling unit.
- C. Fans:
 - 1. Fan raings are based on tests made in accordance with AMCA Standard 210 and shall bear the AMCA scale. Fans shall be of the centrifugal type, designed with a scroll type housing. Fans shall interporate a wheel, structural steel frame and shaft and bearings in the AMCA Arrangement 3 configuration to form a heavy duty integral unit. All fan wheels shall provide stable flow and high rigidity. The wheels shall be non-overloading type. The blades shall be continuously welded, dieformed backward curved type, designed for maximum efficiency and quiet operation. Impellers shall be statically and dynamically balanced and the complete fan assembly shall be test balanced at the operating speed prior to shipment.
 - 2. Bearings shall be heavy duty, grease lubricated, anti-friction ball, self-aligning, ball-type and selected for minimum average bearing life (AFBMA L-10) in excess of 100,000 hours at the maximum class RPM.
 - 3. Fans shall be mounted on vibration bases. Drives for motors shall be direct with motor potentiometer.

- 4. Motors shall be standard NEMA frame, design B high efficiency.
- D. Total Energy (Enthalpy) Recovery Wheel:
 - 1. The rotor media shall be made of aluminum which is coated to prohibit corrosion. All media surfaces shall be light weight polymer coated with a permanently bonded Silica gel desiccant prior to being formed into the honeycomb media structure to ensure that all surfaces are coated and that adequate latent capacity is provided. Desiccant coatings that must be reapplied over time are not acceptable.
 - 2. Sensible and latent recovery efficiencies shall be clearly documented through a certification program conducted in accordance with ASHRAE 84-1991 and the results shall be preserved in accordance with ARI 1060-2000 Standards. The certification shall have been conducted by the unit manufacturer.
 - 3. Wheel testing to document that the desiccant material utilized does not transfer pollutants typically encountered in the indoor air environment shall be provided. The cross-contamination and performance certification reports shall be provided for as part of the submittals for this project.
 - 4. The media shall be cleanable with low temperature steam but water or light detergent, without degrading the latent recovery. Dry particles up to 650 merons shall pass freely through the media.
 - 5. Rotor System:
 - a. Seals: The rotor shall be supplied with cianeter and perimeter seals which shall not make contact with any rotating surface of the univariant rotor face.
 - b. Rotor Support System: The reconnectia shall be provided in segmented fashion to allow for field erection or replacement of one section at a time without requiring side access. The media shall be rigidly how by actuactural spoke system made of stainless steel.
 - c. Rotor Housing: The path housing shall be a structural framework which limits the deflection of the rotor due to air pressure loss to less than 1/32". The housing shall be made of galvanized steerer provent corrosion. The rotor shall be supported by two pillow block bearings which can be maintained or replaced without the removal of the rotor from its casing or the nedia from its spoke system. Bearings shall be selected for an L-10 life in excess of 30 years
 - 1. Drive System: The rotor shall be driven by a self-adjusting flexible, circumferential belt system. A/C motors shall be utilized. Frost control: timed exhaust panel.
 - ssembled system shall incorporate the complete wheel assembly, seals, drive motor and belts in an insulated cassette frame within a slide-out track.
 - Provide filters for both inlet air streams, outside air and return air..
 - 2. Filters shall be disposable 2" thick, MERV 8. The filter shall be listed by Underwriters' Laboratories as Class 2.
 - 3. Provide a bank of galvanized universal holding frames arranged for upstream access.

Filters:

- F. Electrical: Electrical features shall include single point power feed termination, unit-mounted lockable disconnect, internal circuit breaker type overload protection, starters, 24 VAC control transformer and fusing.
- G. Connections: System field connections shall be limited to:
 - 1. Supply air duct connection from the packaged unit.
 - 2. Return air duct connection to the packaged unit.
 - 3. Field supplied power source.
 - 4. Twisted pair, ATC communications wiring.
- H. Manufacturers: Basis of design, Greenheck. Subject to compliance with requirements, manufacturers offering products which may be incorporated in the work are limited to one of the following:

Greenheck

Loren Cook

Temtrol

Valent

1. Any listed equivalent manufacturer and the Machanical Contractor shall be completely responsible to comply with all requirements on the contract documents. This shall include, but not be limited to, space requirements, code clearances, the type, horsepower, capacities, number and size of services required from other trades.

PART 3 – EXECUTION

- 3.1 INSTALLATION
 - A. Verify that coils, filters, motors, drives and other components are matched with the proper unit.
 - B. Assemble unit components on wing manufacturer's instructions for handling, testing and operation. Repair damaged galvancer areas, and paint in accordance with manufacturer's written recommendations.
 - C. Vacuum clean interior of units prior to operation.
 - D. Repair dir taks from or into casing that can be heard or felt during normal operation.
 - E. Perform field mechanical balancing in accordance with Section 230950: TESTING AND BALANCING OF MECHANICAL SYSTEMS.

The Mechanical Contractor shall own as part of his work, the following:

Provide one (1) additional drive set, if necessary, to obtain final design balancing requirements. The Mechanical Contractor shall coordinate with Balancing Firm and equipment manufacturer for drive selection, including belts and pulleys.

G. Provide certified factory start-up and written report on all units.

END OF SECTION 230760



SECTION 230900

AUTOMATIC TEMPERATURE CONTROL

PART 1 – GENERAL

1.1 RELATED DOCUMENTS

- A. Section 230200 and drawings are hereby made a part of this section as fully as if repeated hereby
- B. The Mechanical Contractor shall coordinate with the work of Division 26 and the Fire Alarm System vendor for locations and mounting of all duct smoke detectors. These devices are shown in the Mechanical Drawings for reference only to show the intent of the work. All locations shall be determined based on approved shop drawings from the Fire Alarm System vendor and the Contractor for the work of Division 26, Electrical.
- 1.2 DESCRIPTION OF WORK
 - A. Provide labor, material and supervision necessary to install a complete cirect digital control system of temperature controls to control all HVAC Systems, associated components and accessories as described herein. The controls to be an extension of existing assum.
- 1.3 SUBMITTALS
 - A. Submit shop drawings and manufacturer's data she is of all equipment.
 - B. Submit manufacturer's certificates of conformance with opplicable codes.
 - C. Furnish point-to-point diagram of automatic to operature control system approval, including heating, ventilating and air conditioning equipment wring diagrams where temperature control connections are required.
 - D. Provide ten (10) copies of subrattal cuta within thirty (30) days of contract award.
 - E. Submittal shall consist of
 - 1. System Architecture how hg all digital and pneumatically actuated devices.
 - 2. Equipment lists of all proposed devices and equipment including data sheets of all products.
 - 3. Valve, langer and well and tap schedules showing size, configuration, capacity and location of all equipment.
 - Lata entry forms for initial parameters. Contractor shall provide English listing of all analog points with columnar blanks for high and low warning limits and high and low alarm limits, and a listing of all fan systems with columnar blanks for beginning and end of occupancy periods; and samples of proposed text for points and messages (for at least two systems of at least 15 points total) including sample 480-character alarm message. All text shall be approved prior to data entry.
 - 5. Wiring and piping interconnection diagrams including panel and device power and sources.
 - 6. Sketches of all graphics.

ADDITION AND RENOVATIONS

1.4 QUALITY ASSURANCE

- A. Insure that all work and equipment is installed in accordance with manufacturer's warranty requirements.
- B. Provide adequate supervision of labor force to assure that all aspects of specifications are being fulfilled.
- C. The system shall be engineered, programmed and installed by personnel trained and regularly employed by the control's manufacturer.
- D. Supplier shall have technical support to promptly respond within 24 hours or less to service all to the site with technical staff, spare parts inventory and test and diagnostic equipment.
- E. Codes and Approvals:
 - 1. The complete system installation shall be in strict accordance with actional and local electrical codes. All devices designed for or used in line voltage applications shall be UL listed.
 - a. All microprocessor based devices shall be UL916 listed
 - b. All electrical environmental control and monitoring devices shall be UL429 and/or UL873 listed.
 - 2. All electronic equipment shall conform to the requirements of FCC regulation Class B, Part 15, Section 15 governing radio frequency electrom gnetic interference and be so labeled.
 - 3. The complete system shall conform to ACSUANHRAE Standard 135-2012, BACNET.
- F. All system components shall be design than built to be fault tolerant.
 - 1. Provide satisfactory operation without damage at 100% above and 85% below rated voltage and at +3 Hertz variation in line frequency.
 - 2. Provide static, transient and sucht circuit protection on all inputs and outputs. Communication lines shall be projected against incorrect wiring, static transients and induced magnetic interference. Bus conjected devices shall be A.C. coupled or equivalent so that any single device failure will not disrupt or halt bus communication.

1.5 ELECTRICAL WRING

A. All electrical wiring, components and accessories in connection with the Automatic Temperature Control System shall be furnished and installed by the control contractor.



Electrical Contractor shall provide all wiring to duct smoke detectors.

Unless stated otherwise in the design documents, the ATC Contractor is responsible for providing control power to all valves, actuators, devices and components within the DDC System regardless of the selected voltage of those devices. This also includes all 120 volt power circuits required for devices, panels and control equipment.

- 3. The ATC Contractor shall be responsible for providing the control interface between terminal unit condensate pumps and their respective units at the required voltage of these devices in order to shut down the terminal unit in the event of high water level in the condensate pump receiver.
- B. Control wiring shall include all wiring necessary to interface with new controls, such as electric relays and transducers, and shall also include electric and electronic devices such as freezestats, electronic

sensors, relays, flow switches and controlled devices such as valve and damper operators, both electric/electronic actuated devices, new and existing. Pilot devices such as ON/OFF switches and thermostats installed in series with line voltage circuits shall be considered to be control wiring.

1.6 AUTOMATIC TEMPERATURE CONTROL

- A. Provide a DDC System of automatic temperature control which shall be as manufactured by Schneider Electric SmartStructure, as installed by Tri-M Group, LLC. The system shall be complete in an respects including labor, materials, equipment and services necessary.
- B. All electrical wiring in connection with the installation of the automatic temperature controls shall be furnished and installed under the direct supervision of the control manufacturer.

PART 2 – PRODUCTS

2.1 TEMPERATURE SENSORS

- A. Solid state room sensors shall be of the wire wound resistance type element. Sensors shall be equipped with visual readout and adjustment. Sensors shall be of the completely solid state type with no moving contacts. Printed circuit board under thermostat cover shall ontain a low mass resistance type setpoint dial and amplifier. Provide test points for mersuing output voltage. Sensors shall be direct or reverse acting as required for the sequence of operation.
- B. Sensors shall provide the application for night setback override.
- C. Sensors shall be mounted at ADA height (48" above floor)

2.2 SMOKE DETECTORS

- A. Duct type ionization smoke detectors many exprisible turnished by the Electrical Contractor and installed by the Mechanical Contractor in the supply and return air stream. The Electrical Contractor shall provide wiring from each detector to the Fire Alasm System panel.
- B. The Electrical Contractor shapprovide an alarm output signal from the FAS panel to the BAS for unit shutdown.
- 2.3 ACTUATORS
 - A. Electronic actuator shall be sized to operate their appropriate dampers and valves with sufficient reserve power to provide smooth modulating action or two-position action as specified.
 - B. Provide integral auxiliary switches for direct coupled actuators to indicate when a desired position is reached or to interface additional controls for a specific sequence.
 - C. Align actuator with drive shaft, provide permanent mark to identify closed position of end device.

SENSOR TRANSMITTERS

Duct and immersion sensors shall have minimum spans as required to meet the temperature requirements. Duct sensors shall have sensing elements of sufficient length and accuracy to measure average duct temperature in each location.

- B. Sensors shall be of corrosion resistant construction, tamperproof, suitable for mounting on a vibrating surface. Exposed capillaries shall be temperature compensated, and armored or installed in protective tubing.
- C. All sensing elements for water pipe mounting shall be of the rod and tube type with linear output and

shall be furnished complete with separable protecting wells filled with heat conductive compound. Sensors shall be factory calibrated and tamperproof. If easily adjustable sensors are provided, they shall be located inside metal enclosures with cylinder lock and key to prevent unauthorized setting.

- D. Safety Devices: Provide the following:
 - 1. Low limit, electric type, with 20' long serpentine element, with manual reset, set for 37°F for "freeze" protection and 55°F for fan discharge application, unless otherwise noted.
 - 2. Air and water duty flow switches: Current sensor type for fan status.
 - 3. Carbon dioxide sensor/transducer suitable for wall or duct mounting.
 - a. Analog output of 4 to 20 milliamps corresponding to 0 to 2000 ppm CO
 - b. ABS plastic housing, suitable for an operating environment of 0 to 12 F/0 to 100% RH, non-condensing.
 - c. Repeatability less than +/- 20 ppm.
 - d. Response time less than 60 seconds.
 - e. Power supply, 24 VAC.
 - f. Make: Vaisala Inc. model GMW 21 (wall)
 - GMD 20 (duit)
 - g. Manufacturers: Air Test Technologue, me., Macurco, Rotronic Instrument Corp., Vaisala, Inc.
 - 4. Liquid Leak Detection System: Kele, Faychem, Trace Tek.
 - a. Mechanical float devices attached to or inserted within the auxiliary pan are not acceptable.
 - b. Sensor shall be achyated when there is at least ¹/₄ inch of water, but no more than ¹/₂ inch of water in the auxiliary pan.
 - c. Equal to kee Model AQS00661 water detector.
- E. HUMIDITY TRANSMITTERS
 - 1. Units shall be suitable for duct, wall (room) or outdoor mounting. Unit shall be two-wire transmitter utilizing bulk polymer resistance change or thin film capacitance change humidity sensor. Unit shall produce linear continuous output of 4-20 mA for percent relative humidity (%RH). A combination temperature and humidity sensor may be used for zone level monitoring. Sensors shall have the following minimum performance and application criteria:
 - a. Input Range: 0 100% RH
 - b. Accuracy (%RH): +/- 2% (when used for enthalpy calculation, dewpoint calculation or humidifier control) or +/- 3% (monitoring only) between 20-90%RH at 77°F, including hysteresis, linearity, and repeatability.
 - c. Sensor Operating Range: As required by application.
 - d. Long Term Stability: Less than 1% drift per year.

2. Acceptable Manufacturers: Units shall be Vaisala HM Series, General Eastern, Microline, or Hy-Cal HT Series.

2.5 CONTROL VALVES

- A. Valves shall be rated for a minimum of 150 percent (150%) of system operating pressure at the valve location but not less than 125 psig.
- B. 2" and Smaller: Valves shall be bronze body with screwed or flared connections.
- C. 2-1/2" and Larger: Valves shall be bronze or iron body, flanged.
- D. Flow characteristics:
 - 1. Three-way valves shall have a linear relation of flow vs. valve position.
 - 2. Two-way valve position vs. flow relation shall be equal percentage for where flow control.
- E. Maximum pressure drop through valve:
 - 1. Modulating water flow control: 1/2 the pressure drop through the apparatus with maximum of 10 feet of water. Two position water valves shall be line size.

2.6 CONTROL DAMPERS

- A. The ATC Sub-contractor shall furnish all the controlled examples of the type and sizes indicated on the drawings for installation by the sheet metal Subcontractor.
- B. All 2-position control dampers shall be parallel of the and sized for minimum pressure drop, at the specified duct size.
- C. All modulating dampers shall be opposed blade and sized for an effective linear air flow control characteristics within the angle or rotation and maximum pressure drops specified. Information shall be provided to the sheet metal subcontractor for determining the proper duct reductions or baffles used.
- D. Damper frames shall not be less than 16-gauge galvanized steel, formed with corner braces for extra strength, with mounting hows for enclosed duct mounting.
- E. All damper blackeshan be of not less than 16-gauge galvanized steel formed for strength and high velocity performance. Blades on all dampers must not be over 8" in width. Blades shall be secured to 1/2" diameter zinc plated axles by zinc plated bolts and nuts. All blade bearings shall be nylon or oilite. Blade suce edges shall be sealed off against spring stainless steel seals. Teflon coated thrust bearings shall be provided at each end of every blade to minimize torque requirements and insure smooth operation. All blade leakage hardware shall be constructed of corrosion resistant, zinc plated steel and brass.

F.

Dampers shall be suitable for operation between -40 and 200 degrees. The control manufacturer shall submit leakage and flow characteristics plus a size schedule for all controlled dampers.

- All blade edges shall have inflatable seal edging that shall be rated for leakage less than 10 cubic feet per minute per square foot of damper area at a differential pressure of 4" of water when the damper is being held by a torque not to exceed 50 inert lbs. Leakage shall not exceed 1/2 of 1% of total flow.
- H. Provide permanent mark or scribe end of drive shaft to align damper with actuator in closed position.

2.7 CONTROL CABINETS

A. Control cabinets shall be constructed of 18-gauge steel with locking hinged door. Unless otherwise specified, all controllers, electric relays, switches and other equipment furnished as part of the control system which are not required to be mounted on mechanical equipment, shall be cabinet mounted. The temperature indicators and switches shall be flush mounted on the door tagged with plastic labele. All electrical devices shall be wired to a numbered terminal strip and all devices shall be completely adjusted and checked for proper operation prior to shipment to job site. All wiring shall be numbered according to the control diagram.

2.8 SEQUENCE OF OPERATION

- A. Host Computer and Operator's Work Station (OWS)
 - 1. The existing host computer and peripheral equipment are located in the Maintenance Office.
 - 2. Coordinate and provide all required capacity and features with the Owner's equipment to accommodate the work of the HVAC systems described herein
 - 3. All control programs and application features shall reside in the OVS
 - 4. Control manufacturer shall provide subsequent levels of control capability to whatever extent necessary to achieve performance required for individua units in their respective local control panels. Coordinate power requirements with locations for dedicated 120VAC circuits.
 - 5. Work with the Owner to establish occupied/unoccupied schedules and setpoints. Enter the schedules and setpoints into the system. Provide the required number of input/output points to achieve the specified sequences of operation and monitoring points.
 - 6. Work with the Owner to determine which points shall be trended and the sampling frequency. Set up the trend logs in the BAS

B.1 Chilled Water System Control:

1. This system shall be activated and controlled by the OWS. Once activated, and subject to a flow switch in the leaving rate line at the evaporator, the chiller shall start and sequence through its factory controls to maintain leaving water temperature at 42°F, adjustable.

The following stems shall be available for display at the OWS:

- a. Graphical display of the chiller, pumps, piping layout and temperature control devices with dynamic display of each status, temperature, etc.
- b. Chilled water discharge and return temperature.
- c. Up to 3 alarms/status points for the chiller.
- d. Chilled water discharge temperature reset.
- e. Chiller on/off command.
- B.2 Refrigerant Monitoring System RMS-1 Control:
 - 1. This system consists of a refrigerant monitor and sensors mounted in the basement Boiler Room. Provide all functions and the following interlocks.
 - 2. Whenever the monitor goes into high-level alarm, activate alarm horn and strobe light at each door

into the boiler room. Provide horn reset switch at each door to allow the horn to be silenced, while strobe light continues to flash until the system is reset.

- 3. Activate existing exhaust fan EF-8 after motor operated exhaust damper is open subject to damper limit switch. Open outside air damper at OAI-2.
- 4. Exhaust fan shall continue to run until the monitoring system resets. Once reset, the exhaust fun shall stop, dampers shall close, and the strobe lights shall turn off.
- 5. Provide alarm and system status at the OWS with remote alarm call out through the build Alarm System.
- C. Cooling Tower Control:
 - 1. This system consists of a single cell, fluid cooler, CT-1, existing pump 1.3 and P-4, and interconnecting piping.
 - a. Interface with controls furnished with the tower for operation of the fan motor, fan motor VFD, and basin sump heater.
 - b. Maintain existing sequence of operation for the condense; water pumps.
 - 2. Upon a rise in condenser water return temperature, the power fan motor shall be energized and ramp up from minimum to full speed as required to maintain 85°F setpoint temperature leaving the tower. The reverse shall occur on a fall in cooling tower loop temperature.
 - 3. Provide a water temperature sensor in the sooning tower sump for monitoring and alarm generation to the system. On a fall insult temperature below the low limit setpoint of 40°F, adjustable, activate the basin sump beautr tomaintain water temperature above freezing. On a continued drop in sump water temperature, provide an alarm to the system.
 - 4. Monitor "voltage available" to be heat trace circuits on all exterior drain and makeup water piping at the cooling tower. Provide an alarm to the system if the outside air temperature is below 35°F, adjustable, and voltage is not available to the heat trace circuits. Coordinate with heat trace system provided as par on the work of Division 26 Electric.
 - 5. Provide water miet and outlet water temperature sensors. Sensors shall be used to monitor and control the non-temperatures, and high/low temperature alarms.
 - 6. Provide control wiring for the sump thermostat and sump heater contactor furnished with the conting tower. Power wiring shall be provided as part of the work of Division 26 Electric.



The following items shall be displayed at the OWS:

- a. Outside air temperature.
- b. Condenser temperature entering and leaving the cooling tower.
- c. Condenser temperature high/low temperature alarms.
- d. Sump temperature and low temperature alarm.
- e. Fans status via VFD alarm/status contact and frequency feedback signal.

- f. Heat trace circuit voltage status/alarm.
- g. Diagram showing the layout of the equipment with major components and dynamic temperatures shown where temperature sensors exist in the system.
- D. VRF Heat Pump System Control
 - 1. The sequence that follows is for the VRF system heat pump units HP-1 through HP-5 and condensing unit ACC-1 shown on the drawings for the administration area offices on the first floor. Refer to the drawings for exact locations and zoning.
 - 2. Mount and wire the VRF System Controller, located as directed by the Owner, and controlished by the VRF system manufacturer. Extend control wiring from the controller to all he t pump units and condensing unit within the system.
 - 3. Interface each heat pump unit with its respective wall mounted temperature control, as furnished with each unit.
 - 4. The VRF System Controller shall activate the heat pump units according to its programmed occupied/unoccupied schedule as directed by the Owner. Set roomtemperature control for each heat pump unit to maintain occupied space temperature a 70°1 adjustable, in the heating mode, and 75°F, adjustable, in the cooling mode.
 - 5. The system controller shall provide full system overview display, list all rooms connected to the system, monitor all room conditions, and monitor all heat pump unit configuration settings. Interface the system controller with the QWS for all monitoring and alarm capability.
 - 6. For units mounted horizontally above registed ceilings, provide a water detector in the bottom of the auxiliary drain pan located under to chaint. Upon sensing water in the pan, the unit shall stop and the controller shall initiate an alarm to the OWS.
- E. Ventilation System Control
 - 1. The sequence that follows a for unit ERU-1. The system is a 100% outside air, low velocity, constant volume system and provides pre-conditioned ventilation air to individual heat pump units in the VRF System as leveribed in article 'D' above.
 - a. The unicconsists of a factory packaged unit containing supply and exhaust fans and drives, air filters, never wheel and drive, and factory controls.
 - Provide notor operated dampers and actuators for control of outside air and exhaust air for this unit as shown on the drawings for OAI-1 and EAH-1. Interface these dampers with unit operation.



Provide a DDC Controller and required sensing and control devices for field installation and wiring. The equipment manufacturer shall provide the BAS Contractor with wiring diagrams for the equipment. The BAS Contractor shall then provide wiring diagrams to the equipment manufacturer, detailing installation and wiring requirements for the DDC Controls. The BAS wiring diagrams shall detail connections from the DDC Controller to the equipment terminal blocks.

- 3. The following items shall be provided by the equipment manufacturer:
 - a. Motor starters and overload protection.

- b. Control transformers.
- c. Energy wheel motor, speed controller, defrost controller, and rotation failure contact.
- d. Terminal blocks for all wiring connections between equipment and control devices.
- 4. The following items shall be provided by the BAS Contractor:
 - a. DDC Controller
 - b. Discharge air temperature sensor.
 - c. Return air temperature sensor.
 - d. Temperature sensor at exhaust air outlet.
 - e. Current sensor for one phase of power feeding the supply fan.
 - f. Current sensor for one phase of power feeding the exhaust fac.
 - g. Motor operated dampers and actuators for control of outside ir and exhaust air.
- 5. During the scheduled unoccupied mode, the unit shall remain of and all dampers shall be closed. During the scheduled occupied mode, the energy recovery anit shall run continuously. Delay start of the unit until the zone space temperatures have been restored to occupied setpoint for either the heating or cooling mode of operation for the VRF S, tem.
- 6. The supply fan, exhaust fan and energy recovery wheel shall run continuously while the outside air and exhaust air dampers remain open.
 - a. Whenever the return air temperature at the unit is within 5°F, adjustable, of the outside air temperature, the system shall turn off the energy recovery wheel.
- 7. Provide a current sensor on one phase of power feeding the supply fan and the exhaust fan for status indication and alarn generation.
- 8. Interface input from the unit's factory installed energy wheel rotation sensor for monitoring and alarm generation for starts of the energy wheel. Whenever the rotation sensor indicates wheel failure, the unit shall continue to run until it is shut down manually.
- 9. Interface with a common fire alarm input from the fire alarm system. The fire alarm contact shall be provided at the fire alarm panel by the fire alarm contractor. The status of the alarm contact shall be communicated throughout the system. When the fire alarm contact indicates an alarm contaction, the system shall de-energize the unit. When de-energized, the damper motors shall spring return the outside and exhaust air dampers closed. Provide an alarm to indicate fire alarm status.
- 10. The following items shall be displayed at the OWS:
 - a. Global outside air temperature.
 - b. Discharge air temperature and setpoint.
 - c. Return air temperature.
 - d. Exhaust air temperature.
 - e. Fire Alarm System status/alarm.

- f. Supply and exhaust fan operational status/alarm via current sensors.
- g. Commanded status of fans.
- h. Energy recovery wheel commanded status and alarm.
- i. Diagram showing the layout of the equipment with major components and dynamic temperatures shown where temperature sensors exist in the system.
- F. VAV Terminal Unit Control
 - 1. The sequence that follows is typical for terminal units VAV-1 through VAV-5 as the mon the drawings.
 - 2. Provide control voltage transformer 120/24VAC on the junction box provideous part of the work of Division 26 Electric, for control power to all terminal unit controls. Coordinate with location shown on the electrical drawings.
 - 3. Provide wall-mounted temperature sensor, controller, and two-way ontrol valve and actuator for each terminal unit.
 - 4. On a rise in space temperature, modulate the two-way heating coll control valve closed to coll and open the VAV box air damper from minimum open period to full open position to maintain space temperature at 77°F, adjustable.
 - 5. On a fall in space temperature, the air valve shall modulate closed to the minimum position. On a continued fall in temperature, the two-way heating control valve shall modulate open to coil maintain room temperature at 68°F, adjustable.
 - 6. During the unoccupied mode, the space temperature cooling and heating setpoints shall be reset to 80°F and 65°F, adjustable, respectively if the space temperature rises above or falls below these setpoints, the unit shall index to full cooling or heating until the space temperature decreases or increases by 4°F.
 - 7. During the morning chargeover cycle for associated air handling system, air valve on each terminal shall open fully patil space temperatures have recovered. Air valves shall return to vav operation once s tpoint is restored.
 - 8. The following stems shall be displayed at the OWS:
 - Space temperature.

Coil leaving air temperature.

Terminal unit airflow.

Unit Ventilator Control

- . The sequence that follows is typical for units UV-1 and UV-2. Each unit ventilator shall be controlled by an individual DDC controller. Fan speed shall be adjusted through the fan motor ECM speed control.
- 2. Provide a wall mounted space temperature sensor which shall be wired to the DDC controller.
- 3. Provide a two-way control valve and actuator for the dual temp water coil.
- 4. Provide motor actuators for the face & bypass dampers and the outside air/return air dampers.

AUTOMATIC TEMPERATURE CONTROLS 230900-10

- 5. Morning warm-up: Based on the occupancy schedule in the OWS and prior to the switchover to occupied cycle, the fan shall energize, outside air damper shall remain closed until space temperature is restored to the occupied setpoint for either heating or cooling. When occupied temperature is reached, the outside damper shall open to occupied position. Return air damper shall move in unison.
- 6. Occupied Mode: during the programmed occupied mode, the supply fan shall run continuou ly with the outside air damper open to its minimum position. On a rise in temperature above the programmed cooling set-point, 77°F (adjustable), the two-way control valve shall open foll to the coil and the space sensor shall modulate the face and bypass dampers to maintain serpoint. Or a fall in temperature the reverse shall occur.

On a drop in temperature below the programmed heating set-point, 68°F (adjustable), he two-way control valve shall open full to the coil and the space sensor shall modulate the free and bypass dampers to maintain set-point. On a rise in temperature the reverse shall occur.

- 7. Unoccupied Mode: during the programmed un-occupied cooling and deuting modes, the fan and coil control valve shall cycle/modulate to maintain the un-occupied set-points 80°F cooling/60°F heating, adjustable. The outside air damper shall remain closed, return air damper shall fully open, face & bypass dampers shall be fully open to coil face
- 8. Provide a current sensor on one phase of power feeding the supply fan for status indication at the OWS.
- 9. If the discharge temperature fails to rise or fail to a programmed minimum temperature during a call for heating or cooling, low and high compositure alarms shall be activated at the OWS.
- 10. A low limit control (freeze stat) shell be lestabled in the unit. When tripped, the freeze stat shall function to de-energize the supply fan, damper actuators, and water control valve. When de-energized, the damper actuator had spring return the outside air damper closed, the return air damper open, face & bypast dampers to full face; the water coil control valve shall open to the coil. When the freeze-staterips an alarm shall be generated at the OWS.
- 11. The following items shall be displayed at the OWS:
 - a. Space temperature.
 - b. Space temperature set-point.
 - c. Low Space temperature alarm. Ligh Space temperature alarm.
 - e. Aischarge temperature.
 - Global outside air temperature.
 - g. Freeze stat status, normal/alarm.
 - h. Commanded status of fan, off/on.
 - i. Commanded position of control valve, open/closed.
 - j. Commanded position of OA damper, open/closed.
- H. Cafeteria HVAC Control

- 1. This system consists of air handling unit AHU-1 which is equipped with fan/motor, dual temp coil, and filters, and fan EF-1. The system shall be controlled by an individual DDC controller.
- 2. Provide a wall mounted space temperature sensor and return air duct mounted CO2 sensor as shown on the drawings.
- 3. Provide a three-way control valve and actuator for the dual temp water coil.
- 4. Provide motor operated dampers and actuators for control of outside air, return air, and exhaust air in the system.
- 5. Morning warm-up: based on the occupancy schedule in the OWS and prior to the scheduler to occupied cycle, the supply fan in AHU-1 shall run continuously, outside air damperanti exhaust air damper shall remain closed, EF-1 shall remain off, and return air damper fully oper until space temperature is restored to the occupied setpoint for either heating or country.

Once occupied temperature has been restored:

- a. Outside air damper shall remain closed until return air Co2 avel rises to 700 ppm. The outside air damper shall open from the closed to schedyled open position to maintain CO2 level at or below 700 ppm. The return air damper shall open fully and EF-1 shall start and run continuously.
- b. On a continued rise in CO2 level above 200 ppm, activate an alarm at the OWS. On a decrease in CO2 level below 700 ppm, the new se sequence shall occur.
- 6. Occupied Mode: On a rise in space temperature above the programmed cooling set-point, 77°F adjustable, the three-way control varyeshall modulate open to the coil to maintain set-point. On a fall in temperature the reverse shall occur.
 - a. The mixing box economizer equence shall be activated as the first stage of cooling. The DDC Controller shall receive input from the global outside air temperature and humidity sensors to calculate outside air enthalpy. If the outside air enthalpy is at 25 BTU/lb, adjustable, the mixing box dampers shall modulate to maintain the mixed ait temperature setpoint of 55°F, adjustable. The outside air damper shall continue to open up to 100% outside air to satisfy cooling demand. The return damper in the unit shall move in unison to maintain the balance of air in the unit. The outside air damper shall not close below the minimum position during the occupied period.

Once the outside air damper is fully opened, subject to a limit switch on the damper, the exhaust damper shall open fully and exhaust fan EF-1 shall start and run. Once the economizer mode is deactivated, exhaust fan shall stop and damper shall close.

On a decrease in temperature below the programmed heating set-point, 68°F adjustable, the three-way control valve shall modulate open to the coil to maintain set-point. ON a rise in temperature the reverse shall occur.

7. Unoccupied Mode: during the programmed un-occupied cooling and heating modes, the fan and coil control valve shall cycle/modulate to maintain the un-occupied set-points 80°F cooling/60°F heating, adjustable. The outside air damper shall remain closed, return air damper fully open, exhaust air damper closed with EF-1 off.
- 8. Provide a current sensor on one phase of power feeding the supply and exhaust fans for status indication at the OWS.
- 9. If the discharge temperature fails to rise or fall to a programmed minimum temperature during a call for heating or cooling; low and high temperature alarms shall be activated at the OWS.
- 10. A low limit control (freeze stat) shall be installed in the unit. When tripped, the freeze stat shall function to de-energize the supply fan, damper actuators, and water control valve. When de-energized, the damper actuator shall spring return the outside air damper closed, return air damper open, and the water coil control valve shall open full to the coil. Exhaust air damper shall slore and fan EF-1 shall stop if activated at the time of low limit alarm. When the freeze structures, an alarm shall be generated at the OWS.
- 11. Interface with a common "Global" fire alarm input from the fire alarm sorten. The fire alarm contact shall be provided at the fire alarm panel by the Fire Alarm Contactor. The status of the alarm contact shall be communicated throughout the BAS. When the Fire clarm contact indicates an alarm condition, the BAS shall de-energize the all fans, coil control to be, and damper motors. When de-energized, the damper motors shall spring return all air comper closed. Provide an alarm at the OWS to indicate fire alarm status.
- 12. The Mechanical Contractor shall install duct smoke detectors in the supply and return air ducts at the unit as furnished by the FAS Contractor. When wireo to the fire alarm system as required by the FAS Contractor, the duct smoke detectors shall alarm the FAS. In addition, each duct smoke detector shall have a relay module equipped with a set of dry contacts for extension directly to the DDC controller for the unit. ATC shall provide all control wiring from these contacts to the appropriate controller to shut down the unit upper activation of the smoke detectors.
- 13. The following items shall be displayed at the OWS:
 - a. Space temperature.
 - b. Space temperature set-print
 - c. Low Space temperature alarm.
 - d. High Space temper dare alarm.
 - e. Discharge emperature.
 - f. Mixed at temperature and setpoint
 - Clobal outside air temperature, humidity and enthalpy
 - h. Neeze stat status, normal/alarm.
 - Commanded status of fans, off/on.
 - j. Commanded position of control valve, open/closed.
 - k. Commanded position of all dampers, open/closed.
 - 1. Fire Alarm System status/alarm.
- I. Electric Cabinet Heater Control
 - 1. Unit ECH-1 shall be controlled by a wall mounted space temperature sensor. Provide flat plate temperature sensor as shown on the drawings.

- 2. Interface space sensor with control relay furnished with the heater. Sensor shall activate the heater to maintain space temperature at 70°F, adjustable, during the occupied mode and 60°F, adjustable, during the unoccupied mode.
- 3. Factory mounted sensor shall allow the unit fan to continue to run until residual heat has been dissipated from the unit.
- J. Unit Heater Control
 - 1. Provide wall mounted temperature sensor as shown on the drawings.
 - 2. Sensor shall activate unit fan to maintain space temperature at 65°F, adjustable.
 - 3. Provide space temperature as an indication to the system.

PART 3 - EXECUTION

3.1 INSTALLATION

- A. Install system and materials in accordance with manufacturer's instruction, and roughing-in drawings, and details and drawings. Install electrical work and use electrical products complying with requirements of these specifications. Mount controllers at convenient locations and heights.
- B. All wiring shall be properly supported and run in a neat and wo kmanlike manner. All wiring exposed and in equipment rooms shall run parallel to or at right angles to the building structure. All piping and wiring within enclosures shall be neatly bundled and anchored to prevent obstruction to devices and terminals. All wiring shall be in accordance with an local and national codes. Low voltage wiring for space temperature sensors, communication but between terminal units, etc., above accessible ceilings in finished spaces on the floors may be plenum ated cable. Wiring in all other locations shall be installed in EMT conduit. All electronic wring shall be #18 AWG minimum THHN and shielded if required, except standard network (atheatet, LonWorks, etc.) cabling shall be as tested and recommended in lieu of #18 g uge twisted, #22 or #24 gauge is acceptable if used as a part of an engineered structured cabling system. The control manufacturer must submit technical and application documentation demonstrating that this cabling system has been tested and approved for use by the manufacturer probable the control system and the engineered structured cabling system.
- C. Provide all sensing, ontrol, and interlock wiring for the following:

System inputs and outputs

System communications

Systempower

vstem interlocks

Unit controls

The Control Manufacturer shall enter all computer data into the Host computer including all graphics, control programs, initial approved parameters and settings, and English descriptors. The Control Manufacturer shall maintain diskette copies of all data file and application software for reload use in the event of a system crash or memory failure. One copy shall be delivered to the owner during training sessions, and one copy shall be archived in the Control Manufacturer's local software vault.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

3.2 DATA CONTROL (D/C) AND GRAPHICS SUMMARY

- A. All hardware, custom software, application software, graphics, etc., necessary to accomplish the control sequences and display the graphics specified shall be provided as part of this contract. Provide all controllers, inputs, outputs, valves, dampers, actuators and flow meters required to provide the control and graphic data described. Provide software setpoints required for display in logical group and graphics.
- B. Each digital output shall have a software-associated monitored input. Any time the monitor d input does not track its associated command output within a programmable time interval, a command failed" alarm shall be reported.
- C. Where calculated points (such as CFM) are shown, they shall appear in their respective logical groups.
- D. Unless otherwise specified or approved prior to bidding, the primary analog input and the analog output of each DDC loop shall be resident in a single remote panel containing be DDC algorithm, and shall function independent of any primary or UC communication links. Secondary (reset type) analog inputs may be received from the primary network, but approved default in dues and/or procedures shall be substituted in the DDC algorithm for this secondary input if network communications fail or if the secondary input becomes erroneous or invalid.

3.3 ACCEPTANCE

- A. The Control Manufacturer shall completely check cut, catibrate and test all connected hardware and software to ensure that the system performs in accordance with the approved specifications and sequences of operations approved.
- B. Witnessed acceptance demonstration shall a splax and demonstrate each type of data entry to show site specific customizing capability; demonstrate parameter changes; execute digital and analog commands; and demonstrate DDC locp stability via trend of inputs and outputs.

3.4 MANUALS

- A. The following manuals y mbe provided:
 - 1. An Operators Manual had be provided with graphic explanations of keyboard use for all operator functions specified under Operator Training.
- B. Computerized ern touts of all GPC data file including all point processing assignments, physical terminal relationships, scales and offsets, command and alarm limits, etc.
- C. A manual shall be provided including revised as-built documents of all materials required under the paragraph "SUBMITTALS" on this specification.
 - No Operators Manuals, and two As-Built Manuals shall be provided to the owner.

TRAINING

- All training shall be by the BMCS contractor and shall utilize operator's manuals and as-built documentation.
- B. Operator training shall include one (1) four-hour session encompassing modifying text and graphics, sequence of operation review, selection of all displays and reports, use of all specified OWS functions, troubleshooting of sensors (determining bad sensors), and password assignment and modification. The training session shall be conducted at system completion.

3.6 SERVICE GUARANTEE

A. The control system herein specified shall be free from defects in workmanship and material under normal use and service. After completion of the installation, the control manufacturer shall regulate and adjust all thermostats, control valves, motors and other equipment provided under this contract. If within twenty-four (24) months from date of acceptance either for beneficial use of final acceptance, whichever is earlier, any of the equipment herein described is proven to be defective in workmanship or materials, it will be replaced or repaired free of charge. The control manufacturer shall, after acceptance, provide any service incidental to the proper performance of the control system under guarantee outlined above for the period of one year. Normal maintenance of the system or adjustments of components is not to be considered part of the guarantee. The control manufacturer will upon completion of the installation, during the warranty period, make available to the twen, an annual service agreement covering all labor and material required to efficiently maintain the control system.

3.7 FINAL ADJUSTMENT

- A. After completion of installation, adjust thermostats, control valves, motors and similar equipment provided as work of this section.
- B. Final adjustment shall be performed by specially trained personnel in direct employ of installer of primary temperature control system.
- 3.8 EXISTING CONTROLS

1 FC

- A. The bid for control work shall be based on the premise that existing control devices are operational and are not in need of repair or replacement usless otherwise noted.
- B. Contractor shall notify the Owner of an control devices that need to be replaced or repaired that may be noticed by him in the process of initiallation of the new work.

ND OF SECTION 230900

SECTION 230950

TESTING & BALANCING OF MECHANICAL SYSTEMS

PART 1 – GENERAL

1.1 JOB CONDITIONS

- A. Systems shall be completely installed and in continuous operation as required to accomplish the tests.
- B. Heating, ventilating and air conditioning equipment shall be completely installed and in particular operation as required to accomplish the balance work specified.
- C. Adjust and balance shall be performed when outside conditions approximate design conditions indicated for heating and cooling functions.
- D. Make at least two inspections of the mechanical systems during construction to verify that balancing procedures may be accomplished. Report findings to the Architect/Engineer.
- E. Balancing firm shall balance Mechanical System two (2) times. The first time shall be considered a rough balance. Any discrepancy in air flow shall be addressed in the Architect/Engineer. The final balancing will be accomplished after review of rough balance i pote.
- F. The final balancing reports shall be submitted and approved prior to project's being considered complete; i.e., commencement of warranties.
- 1.2 ENGINEER QUALIFICATIONS
 - A. The firm shall be an independent organization having no affiliation with construction contractors, equipment sales or design engineering
 - B. The firm shall specialize in balancing seating, ventilating and air conditioning systems.
 - C. The firm shall show proof of having balanced and tested at least five projects of similar size and scope.
 - D. All field work shall be under the direct supervision of a registered Professional Engineer who is a full-time employee of the balancing firm.
 - E. The firm shall be certified by and a member of the AABC (Associated Air Balance Council), or NEBB (National Favironmental Balancing Bureau).
- 1.3 REPORT
 - A. Data heets:
 - Submit data sheets on each item of testing equipment required.
 - Include name of device, manufacturer's name, model number, latest date of calibration and correction factors.
 - Report Forms:
 - 1. Submit specimen copies of report forms.
 - 2. Forms shall be 8-1/2 x 11 inch paper for loose-leaf binding, with blanks for listing of the required test ratings and for certification of report.
 - 3. Reports shall be on standard forms published by AABC or NEBB.

ADDITION AND RENOVATIONS

PART 2 - PRODUCTS

- 2.1 AIR BALANCE INSTRUMENTS
 - A. Velometer with probes and alnor pitot tube.
 - B. Rotating Vane Anemometer: 4-inch size.
 - C. ASHRAE Standard Pitot Tubes, stainless steel 5/16 inch outside diameter, lengths 18 inches and 36 inches.
 - D. Magnehelic Differential Air Pressure Gauges, 0 to 0.5 inches, 0 to 1.0 inch and 0 to 5.0 inches water pressure ranges, each arranged as a portable unit for use with a standard Pitot tube
 - E. Combination Inclined-Vertical Portable Manometer, range 0 to 5.0 inches water.
- 2.2 WATER BALANCING INSTRUMENTS
 - A. 30 Inch Mercury U-Tube Manometer, 200 psig, with 3 valve bypass assembly and return wells or mercury check valves.
 - B. Inspector's gauge testing set.
 - C. Water Differential Pressure Gauge, 4-1/2 inch dial, 0 to 100 si range.
 - D. Pressure gauge measurement points, quick connect courtings, 1/4 inch psi.
- 2.3 SYSTEM PERFORMANCE MEASURING INSTRUMENTS
 - A. Insertion Thermometers, with graduation a 0.3 degrees F for air and 0.1 degrees F for water.
 - B. Sling Psychrometer.

PART 3 - EXECUTION

- 3.1 GENERAL REQUIREMEN
 - A. Arrange and pay for all
 - B. Notify Architect/angineer at least three working days in advance of test.
 - C. Tests to be reformed prior to insulation, covering or concealment.
 - D. Provide signed report of completion of test with signature of witnesses. Report shall indicate:
 - Sy em Tested
 - Date

Specified test requirements and actual testing results

The balancing firm shall report to and review the work required with the Architect/Engineer before beginning field balance work. The balancing firm shall make at least two inspections of the air systems during construction and shall report his findings in writing to the Architect/Engineer.

F. The balancing firm shall cooperate with the Architect/Engineer and the Mechanical Contractor to effect smooth coordination of the balancing work with the job schedule.

- G. The balancing firm shall be responsible for getting the various systems into proper operation. They shall enlist the aid of the equipment suppliers and Mechanical Contractor as may be required to effect proper operation consistent with the contract plans and specifications.
- H. When the balancing firm cannot balance a belt-driven piece of equipment with the supplied belts and sheaves, inform the Mechanical Contractor that the Mechanical Contractor shall provide additional sheaves as spelled out in other Division 23 Sections.
- 3.2 CIRCULATING WATER SYSTEM TEST
 - A. All piping tests shall be applied not only to piping, but also to all devices and equipment connected thereto with the exception of control valves, boilers or any other equipment which may be damaged by the test pressure. All valves shall be full open.
 - B. Test at 100 psi hydrostatic pressure for 6 hours:
 - 1. Record pressures each hour
 - 2. Repair all leaks
 - 3. Retest until 6 hours can be completed with no leaks or loss of pressure.
 - C. After completion of successful test, strainers shall be cleaned then system shall be backflushed and strainers cleaned again.
- 3.3 DUCTWORK TESTING
 - A. Witness testing conducted by the Mechanical Contractor per Section 230600, PART 3: EXECUTION.
- 3.4 BALANCING PROCEDURE
 - A. Air System Balance:
 - 1. With the fan supply system set to handle normal minimum outdoor air, the balancing firm shall perform the following tests and compile the following information:
 - Air Handling Equips int
 - a. Design Conditions:
 - (1) CTM Supply Air
 - (2) Static Pressure
 - (3) CFM Fresh Air
 - (4) Fan RPM
 - . Installed Equipment:
 - (1) Manufacturer
 - (2) Size/Model Number
 - (3) Motor HP, Voltage, Phase, Full Load Amperes
 - c. Field Test:
 - (1) Fan Speed

QC.

- (2) No Load Operating Amperes
- (3) Fan Motor Operating Amperes
- (4) Calculated BHP
- d. Test for Total Air:
 - (1) Size of discharge, return air and outside air ducts.
 - (2) Number and locations of Velocity Readings taken.
 - (3) Duct Average Velocity
 - (4) Total CFM
 - (5) Outside Air CFM
 - (6) Return Air CFM
- e. Individual Outlets (Diffusers, Registers and/or Grilles);
 - (1) Identify each outlet or inlet as to location and great notion system
 - (2) Outlet, manufacture and type
 - (3) Outlet size
 - (4) Outlet free area, core area, or neck area
 - (5) Required FPM and test velocity found for each outlet.
 - (6) Required CFM and test results for each outlet
- 2. After completion of tests, a justment and balancing under minimum fresh air conditions, set the system for 100% fresh at a Repeat the total CFM tests to check field versus design conditions. The results under 100% fresh air cycle shall agree with conditions found under "minimum fresh air operation" before are system is considered to be in balance. Adjustments of the proper dampers shall be made achieve balance.
- 3. Testing and adiasting of individual outlets shall be performed under procedures recommended by the manufactures of the outlets. All outlets shall be set for air pattern required and all main supplyair and return air dampers to be adjusted and set for design CFM indicated. Any required changes mair patterns, settings, etc., necessary for achieving correct air balance, shall be provided by his Contractor. Total CFM of all outlets shall agree with total CFM of all branches and the grand total shall agree with the air volume for the fan(s).

Valer Balance:

Water balance shall include heating water, chilled water and condenser water systems. The balancing agency shall perform the following tests, compile data and submit reports.

- 2. Pumps:
 - a. Design Data
 - (1) GPM, head
 - (2) RPM, BHP

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- b. Installed Equipment
 - (1) Manufacturer, Size
 - (2) Type Drive
 - (3) Motor HP, Volts, Cycles and Phase
 - (4) Full Load Amperes
- c. Field Test
 - (1) Discharge Pressures: Full flow & no flow
 - (2) Suction Pressures: Full flow & no flow
 - (3) Operating Head and GPM
 - (4) No Load Amperes (where possible)
 - (5) Full Flow Amperes, No Flow Amperes
 - (6) Calculated BHP
- 3. Heating and/or Cooling Elements Including Loop Water o all terminal Units:
 - a. Design Data:
 - (1) MBH Specified, GPM Specified
 - (2) Entering Water Temperature (A
 - (3) Entering Air Temperature (EA
 - (4) Water Temperature (Drop (DTW)
 - (5) Element Type Specific
 - b. Field Test:
 - (1) Iderafy each element as to location
 - Recursed water temperature drop corrected for item (3) above
 - Actual entering air and water conditions (temperature and GPM)
 - 4) Adjust element until required temperature drop is obtained

h addition to the above work, the Balancing Firm shall check the operation of all automatic temperature control equipment; verify all thermostat, aquastat, etc., set-points and operations; and enlist the aid of the Mechanical Contractor and the Control Subcontractor to make necessary adjustments where required.

END OF SECTION 230950



SECTION 260000

GENERAL PROVISIONS – ELECTRICAL

PART 1 – GENERAL

1.1 RELATED DOCUMENTS

- A. The general provisions of the Contract, including the Conditions of the Contract (General, Supplementary and other conditions, if any) and Division 1 as appropriate, apply to the work of this Section.
- B. The specification or drawing and the design features or resulting construction disclosed are the property of Furlow Associates, Inc., and shall not be reproduced without written permission.

1.2 DESCRIPTION OF WORK

- A. Provide all materials, equipment, labor, services and all appurtenances required to completely install and satisfactorily operate the various systems. The items listed below are for general guidance only and do not necessarily include the entire requirements for the project.
 - 1. Coordination with other trades
 - 2. Interior feeders
 - 3. Lighting and power panels
 - 4. Lighting branch wiring
 - 5. Power wiring
 - 6. Lighting fixtures and lamps
 - 7. Wiring devices
 - 8. Connections for electrically operated equipment
 - 9. Lightning protection system
 - 10. Related work as herein described or otherwise defined under the heading "Related Work".
- B. Wherever the term provide" is used, it shall be understood to mean both "furnish" and "install".
- 1.3 RELATED WOKK
 - A. Equipment specified in sections of Divisions 1 thru 23 that require electric power supply.
 - B. Work related to this trade as defined on the following contract drawings:
 - Architectural/Structural
 - HVAC

Plumbing

- 1.4 SITE CONDITIONS
 - A. Attention of all bidders is called to the necessity for a careful inspection of the site, its present condition and encumbrances, the extent of the work, the protection to be afforded to adjacent

properties or structure, availability of utilities, the extent and nature of the material required to be excavated and the amount of fill and removal. He shall also determine local or site limitations which will affect construction.

1.5 PERMITS, INSPECTIONS AND ORDINANCES

- A. All work shall be executed and inspected in accordance with local and state ordinances, rules and regulations and the requirements of public utilities having jurisdiction. The contractor shall secure and pay for all permits, inspections and connections required.
- B. The Electrical Contractor shall furnish a certificate of inspection to the Owner at the completion.
- C. Requirements of the following organization shall be considered minimum:
 - 1. National Electrical Code
 - 2. National Electrical Safety Code
 - 3. OSHA
 - 4. Local City and County Codes
- D. Reference to technical societies, trade organizations and governmental agencies are in accordance with the following:
 - 1. ANSI American National Standards Institute
 - 2. ASTM American Society for Testing Materials
 - 3. IEEE Institute of Electrical and Electronics Engineers, Inc.
 - 4. NEC National Electrical Code
 - 5. NEMA National Electrical Manufacturer's Association
 - 6. NFPA National Free Projection Association
 - 7. MSS Manufacturer's Standardization Society
 - 8. IES Illumin, ing Engineers Society
 - 9. ETL Engineering Testing Laboratories
 - 10. Et Al-Electronic Industries Association
 - 1. OSHA Occupational Safety and Health Administration
 - 12. Federal Specifications
 - 13. UL Underwriters Laboratories, Inc.
 - QUALITY ASSURANCE
- A. Provide adequate supervision of labor force to assure that all aspects of the contract documents are fulfilled.
- B. Contractor to provide manufacturer's written certification that the following equipment has been installed and will operate correctly and in accordance with the manufacturer's warranty requirements.

Fire Alarm and Detection System

- C. Testing:
 - 1. After completion of the work, the entire wiring system shall test entirely free from grounds, short circuits, opens, overloads and improper voltage.
 - 2. The grounding system shall be tested for a resistance of 25 ohms or less.
 - 3. Perform testing as follows: Arrange and pay for all tests, provide all equipment, hat ands and labor to perform test. Notify Engineer and Owner three (3) working days before tests are to be made. Conduct tests in the presence of the Engineer or authorized represent tive. Repeat tests after defects are corrected.
- D. Special Engineering Services: In the instance of complex specialized electric power and signaling systems, and other similar systems, the installation and final connectons of these systems shall be made by and/or under the supervision of a competent installation and service engineer who shall be a representative of the respective equipment manufacturer. Any and all expenses of these installation and service engineers shall be borne by this Contractor.

1.7 COORDINATION

- A. As a requirement of this project, the Electrical Contactor shall furnish coordination for his equipment and layouts with other subcontractors furnishing equipment and services for Divisions 1 thru 23. Any and all contractors who install their equipment or funcish services prior to coordination, any contractor who changes their equipment or services after coordination has occurred, without notifying associated subcontractors, shall be held responsible for making all required changes with no additional cost to the Owner. Or delay in construction times This coordination will include conduit layout to allow access to equipment for maintenance.
- B. The Mechanical, Plumbing and Executeal Contractors are responsible to coordinate all manufacturer's recommended circuit breakers, surfers, disconnects and fuse sizes for all equipment. Submission of a shop drawing will certify that this has been completed.
- C. The drawings and specifications reflect the type, number and size of services required for the equipment the design is based upon. Should the supplying subcontractor elect to furnish an alternate piece of equipment requiring difference services and/or space conditions, he shall inform the subcontractor furnishing those services and be held responsible to pay for all required changes as part of this contract.

1.8 JUBMINTALS

Shop Drawings:

Shop drawings shall be submitted in accordance with Division 1 of these specifications except where herein modified.

2. Shop drawings comprising complete catalog cuts, performance test data for electrical equipment as required by other sections of Division 26 shall be submitted for review checking. The Contractor shall review these shop drawings for conformance to contract documents prior to submission and affix contractor's signature to each submittal certifying that this review has been done. By approving and submitting shop drawings, product data, wiring diagrams and similar materials, the Electrical Contractor represents that he and/or his subcontractor has determined and

Tetra Tech 200-15704-17001

verified materials, field measurements and field construction data that relates to the work, and has checked and coordinated this information with all of the Divisions 1 thru 23 subcontractors.

- 3. All shop drawing submittals shall have the following identification data, as applicable, contained therein or permanently adhered thereto:
 - a. Project name
 - b. Project number
 - c. Sub-Contractor's, Vendor's and/or manufacturer's name and address.
 - d. Product identification.
 - e. Identification of deviation from the contract documents.
 - f. Applicable contract drawings and specification section number.
 - g. Shop drawing title, drawing number, revision number, and date of drawing and revision.
 - h. Resubmit revised or additional shop drawings as requested.
 - i. Wherever shop drawings or vendor's standard data speets indicate work to be done "by others", it shall be the responsibility of the Contractor making the submission to identify by name, the Contractor who is to do this work. If the Contractor named is other than the Contractor making the submission, the ship drawing submission must be reviewed by the named Contractor and bear his mark of approval, prior to submission to the Architect/Engineer.
 - j. Where equipment proposed differs bonn that shown on the drawings or specified, he shall submit for approval drawings showing the manner in which the layout is affected by the substitution.
 - k. The Contractor shall keep one copy of approved shop drawings at the job site, filed in a suitable metal container. The shop drawings shall be cataloged and kept in good repair, and shall be available for the by the Owner, Architect and Engineer.
 - 1. No equip net shall be ordered, fabricated, etc., before approval of shop drawings.
- 1.9 SUBSTITUTION
 - A. Whenever material, article, piece of equipment or system is identified in the following specification or indicated on the drawings by reference to manufacturers' or vendors' names, trade names, catalog numbers or the like, it is so identified for the purpose of establishing the basis of the Bid.

Substitution approval must be obtained and included as an addendum item prior to the submission of the bid. An approved substitution shall not be considered as an approval for the contractor or an equipment vendor to deviate from the written portion of the specifications unless so stated in the addendum.

- 2. The drawings illustrate the space allocated for equipment and the Contractor shall install the equipment accordingly. If changes are required in the building or arrangement due to substitution of equipment, the Contractor making the substitution must pay for the necessary modifications.
- D. The listed equivalent or substituted manufacturers along with the bidding related contractor shall be completely responsible to comply with all requirements on all contract documents. This shall include,

but shall not be limited to space requirements, code clearances, the type, horsepower, capacities, number and size of services required from other trades, including all required ancillary items furnished and installed by other trades. If the manufacturer or related bidding contractor does not comply with these requirements, then they shall be responsible for any and all additional costs associated with the changes required by other trades.

- 1.10 LUBRICATION
 - A. Furnish, install and maintain all required lubrication of any equipment operated prior to accept and the Owner. Lubrication shall be as recommended by the equipment manufacturer.
 - B. Provide one year's supply of lubricants to Owner at date of acceptance.
 - C. Verify that required lubrication has taken place prior to any equipment start-up
- 1.11 ADJUSTMENT & CLEANING
 - A. Adjust and clean equipment to be placed in proper operation condition.
- 1.12 EQUIPMENT START-UP
 - A. Verify proper installation by manufacturer or his represented w
 - B. Advise General Contractor 2 days prior to actual start-up.
 - C. Verify proper operation. Obtain signed statement by manufacturer or his representative that equipment is operating within warranty requirement. Shomit statement to General Contractor.
- 1.13 OPERATION AND MAINTENANCE INSTRUCTIONS
 - A. Properly and fully instruct Owner's rersonal in the operation and maintenance of all systems and equipment.
 - B. Insure that the Owner's personnel are amiliar with all operations to carry on required activities.
 - C. Such instruction shall be to leach item of equipment and each system as a whole.
 - D. Provide report that instruction has taken place. Include in the report the equipment and/or systems instructed, date, contractor, Owner's personnel, vendor, and that a complete operating and maintenance manual has been eviewed.
 - E. Manual stall include all instructions on operation, maintenance, repair parts list, lubrication requirements brochures, catalogue cuts, wiring diagrams, piping diagrams, control sequences, service requirements, names and addresses of vendors, suppliers and emergency contacts. Three manuals anall be provided.
 - Summit manuals for review prior to operating instruction period. Manuals shall be $8-1/2 \ge 11$ " with hard cover, suitably bound.

Training

1. Electrical Contractor shall be responsible for coordination of Owner training. Factory employed technician(s) shall provide training, including demonstration and education on the system capabilities, operation and maintenance. Training sessions shall be minimum 4 hours (maximum 8 hours), and shall be provided for each shift of workers. Scheduled training shall be coordinated at least two (2) weeks in advance with the Owner and the Commissioning Agent.

Tetra Tech 200-15704-17001

- 2. Video Documentation: Furnish three (3) copies of a professionally taped video and three (3) copies of professionally prepared drawings demonstrating the following:
 - Emergency Generator System (Alternate)
 - Clock and Speaker System
 - VFD's
 - MDF/IDF
 - Stage Dimming Rack
- 1.14 TOOLS
 - A. All equipment furnished by the Contractor which requires special tools or devices other than those normally available to the maintenance or operating staff shall be furnished in highlicate to the Owner, sufficiently marked, packed or boxed for staff usage. The tools provided shall be listed by the Contractor identified as to their use or the equipment applicable in atyritien transmittal to the Owner.

1.15 CLEANING AND FINISHING

A. After equipment start-up and all operating tests have been made and the system pronounced satisfactory, each respective Contractor shall go over the entire project, clean all equipment, etc., installed by him and leave in a clean and working condition. Any surfaces found marred after this final cleaning shall be refinished or replaced by each Contractor at no cost to the Owner.

1.16 OPERATING AND MAINTENANCE MALLA

A. Three complete sets of instructions certaining the manufacturer's operating and maintenance instructions for each piece of equipment shall be furnished to the Architect. Each set shall be furnished before the contract is completed. The following identification shall be inscribed on the covers: the words "OPERATING ANI) MAINTENANCE INSTRUCTIONS", the name and location of the building, the name of the Contractor and the name of the Architect and Engineer. Flysheet shall be placed before instructions covering each subject. The instruction sheets shall be approximately 8-1/2 by 11 inches, with angle neets of drawings folded in. The instructions shall include, but shall not be limited to, the following:

Approved wiring and control diagrams, with data to explain the detailed operation and control of each componen.

A correct sequence describing start-up, operation and shutdown.

perature and maintenance instructions for each piece of equipment, including lubrication estructions.

Manufacturer's bulletins, cuts and descriptive data.

Parts lists and recommended spare parts.

SERVICE INTERRUPTION

A. All service interruptions to the electric or related systems, whether during regular working hours or at any other time, must be coordinated with the Owner. All such interruptions shall be so scheduled and planned as to require a minimum of time and shall occur only during a mutually satisfactory period.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

1.18 INTERPRETATION OF SYSTEMS

A. The interpretation of the Architect will be final in the event there is a lack of understanding of the full scope or requirements of the systems under this contract.

1.19 LAYOUTS

A. On small scale drawings, i.e., 1/8" - 1'-0", the approximate location of the electrical branch circuit items such as receptacle, telephone, grounding and equipment outlets are shown to indicate their existence. The exact location of these items and their related raceways are governed by structural conditions, coordination with the work of other trades and the Architect's final decision. By arsepting a contract, the Contractor agrees to install the work in accordance with the above statement and within the contract price.

PART 2 – PRODUCTS

2.1 MATERIAL

- A. All material shall be new and of good quality. Material shall conform to 11 accepted trade standards, codes, ordinances, regulations, or requirements governing same and shall be approved before being installed.
- B. The Architect reserves the right to require the Contractors to ubmit samples of any or all articles or materials to be used on the project.
- C. Where any device or equipment is herein referred to in the singular number, such as "the panel", this reference shall be deemed to apply to as manufactorial levices or equipment as are required to complete the installation as shown on the drawings or specified.
- D. All materials and equipment used on the work shall comply with the standards of recognized authorities such as UL, NEMA, IEFE, STL JES and EIA in every instance where such standards have been established for the particular type of materials to be installed.
- E. All similar pieces of equipment ormaterials of the same type or classification used for the same purpose shall be of the same manufacturer.
- F. All manufactured expipment shall have factory applied finishes.
- 2.2 CONCRETE
 - A. Concrete shall be in accordance with Section 03300, or ACI-613. Designer choice if 03300 is not used.
 - B. The 28-bay minimum compressive strength shall be 3000 psi.

2.3 WARRANTY

Wherever in the specification sections of this division, reference is made to a specific warranty period, this warranty shall be in addition to and not a limitation of other rights the Owner may have against the Contractor under the contract documents.

PART 3 – EXECUTION

3.1 INSPECTION

A. Prior to performing the work, examine areas and conditions; check and verify all dimensions, under which the work is to be installed and notify the Architect in writing of conditions and dimensions

detrimental to the proper and timely completion of the work. Do not proceed until authorization is given by the Architect.

3.2 LAYING OUT WORK

- A. The Contractor is responsible for the accuracy of all lines, elevations, and measurements, grading and utilities and must exercise proper precaution to verify figures shown on drawings before laying cut work and will be held responsible for any error resulting from his failure to exercise such precation.
- 3.3 WORKMANSHIP
 - A. Install all work neat, trim, parallel and plumb with building lines in accordance with standard trade practice acceptable to the Architect.
- 3.4 PRODUCT DELIVERY, STORAGE AND HANDLING
 - A. Protect all equipment and materials from damage during transportation storage and installation.

3.5 **PROTECTION**

A. Protect all work, equipment and materials during construction to the time of acceptance by the Owner.

Arrange and design the protection to prevent damage from infiltration or dust, debris, moisture, chemicals and water. Cap or plug electrical raceways.

- B. Protect all surfaces against damage from welding, curing, ourning, or similar construction functions. This protection shall be accomplished by calculate errations, covering and shielding. Special care is directed to exposed finished masonry, meal or wood surfaces and painted surfaces. Corrective measures required shall be accomplished by the trade which made the original installation when and as directed by the Architect at the expense of the Contractor.
- C. Cover and protect all lighting extures as may be necessary until completion of the work. Replace damaged fixtures or damaged fixture parts as directed by the Architect at no cost to the Owner.
- D. Do not install devices, boushed metal fittings or parts until adjoining tile or masonry work is completed.
- E. Maintain and replace protective covering when so directed by the Architect until the work is ready for acceptance
- 3.6 CUTTING PATCHING
 - A. Eurnish information to the General Contractor as to sizes and locations of recesses required to install banel boxes and other equipment or devices. If the information is late or incorrect, this Contractor shall, at his own expense, have the trade which originally installed the work do the required cutting and patching.

Perform all cutting of concrete or other material for passage of raceways as required to install the work.

- Close all such openings around raceways with material as specified under the heading "SEALING".
- D. Install concealed work in place for the mason to wall-in as he carries up the walls; otherwise, this Contractor will be responsible as stated in the first paragraph.

3.7 SEALING

- A. Where raceways pass through fire-rated walls and floors, seal opening with RTV foam.
- B. Seal raceways entering the building to conform to the requirements of the NEC.
- 3.8 OFFSETS AND MODIFICATIONS
 - A. Furnish and install all offsets necessary to install the work and to provide clearance for the work other trades.
 - B. Maintain adequate clearance as directed by the Architect/Engineer.
 - C. Incidental modifications necessary to the installation shall be made as necessary and at the direction and/or approval of the Architect.

3.9 SLEEVES

- A. Furnish and install sleeves for all raceways passing through floors and wells. Sleeves shall be Schedule 40 galvanized steel pipe and shall extend 1" above finished loor surface. Where sleeves are set in interior walls, they shall finish flush with the wall.
- B. Furnish and install watertight sleeves for all raceways extending through foundation walls into crawl spaces, mechanical rooms or basement areas from building exterior or from unexcavated areas to building interior. Sleeve shall consist of extra heavy pipesleeve with anchor flange. Space between raceway and the sleeve shall be sealed with modular wall and casing seal similar to Thunderline Corporation "Link-Seal",, Metraseal or approved substitute. Install seal in strict accordance with the manufacturer's recommendations.

3.10 EXCAVATION

- A. The excavation shall be of the open even basehold and to the depths and widths as may be necessary. The Contractor shall do all excevation required in connection with his work. Bottoms of trenches shall be excavated to a uniform grade. All materials excavated shall be deposited on the side of the trenches and beyond thereach or slides. Excavated material shall not be piled where it will interfere with traffic.
- B. No conduits shall be bedded directly on rock. They shall be cushioned by a 6-inch layer of crushed stone or gravel of elected grade, of size to pass through a 3/4" mesh sieve. Not less than 30% shall be fine which will pass through a 3/8" mesh sieve.
- C. Where excavation is required through tree root areas, roots shall be saw cut, treated with pruning paint and covered with burlap. Burlap shall be wet and shall be protected and maintained in a moist condition during entire period of exposure. Backfill shall be carefully placed and hand-tamped to a minimum of 6" above roots.

Bidder shall base his estimate upon the presumption that all excavation required in the performance of this Contract will be earth. If rock is encountered, Contractor will be reimbursed for the additional work required to remove same based upon the unit cost established in the proposal.

E. All detached boulders or loose stone not exceeding 1 cubic yard, all topsoil, sand, gravel, clay, rubbish, walls or other subgrade construction, and all other materials of every name and nature which can be removed without breaking up with pneumatic breakers shall be considered earth excavation.

- F. All rocks, attached boulders, boulders exceeding 1 cubic yard, walls or other subgrade construction and materials which cannot be removed without breaking up with pneumatic equipment shall be considered rock excavation.
- G. Before commencing any rock excavation for which extra compensation is to be paid, a rock contour drawing shall be prepared by the Contractor and checked by the Architect. The width shall be based on 2'-0". This rock contour drawing and width allowance will be used to compute the quantity of rock for which the Contractor will be reimbursed at the unit price established.

3.11 SHORING AND PUMPING

- A. The Contractor shall provide all shoring, bracing or sheet piling necessary to maintain the bank of his excavation and shall take out same as the work progresses and filling in has been accomplished. Shoring shall be in accordance with OSHA Standards.
- B. The arrangement of shoring must be such as to prevent any movement of the trench banks and consequent strains on the conduits. Shoring shall be provided to prevent da mage to work installed by other trades.
- C. The Contractor shall do all pumping required to keep his excavations free of water. The water shall be conveyed in piping or watertight troughs a sufficient distance that n will flow from the site and not affect other work being performed.

3.12 BACKFILLING

- A. After work in trenches has been completed, they scale called with good, clean, fine earth in 8" layers and shall be pneumatically tamped before the next eyer of material has been filled in. The backfill shall be free of excavated rock, cinders, stores, brickbats or other debris.
- B. Wherever rock is removed, the Contractor shall secure and fill select clean earth to a minimum depth of 3'-0" above the top of the contract caless otherwise indicated, no rock shall be deposited in the trench fill. This clean earth fill shall be procured other than from the site unless permission for earth borrow from the site is granted by the Architect. If site borrow is permitted, the topsoil removal, relocation and finished aracters will be accomplished as directed by the Architect.
- C. Under no circumstances shill excavated material be left where it will interfere with the Owner's or other Contractor's or erations.
- D. All earth and ober materials taken from the trenches and not required for backfilling shall be deposited where firected, or removed from the premises as directed by the Architect.
- E. Any how removed from the excavation shall be removed from the project site by the Contractor.
- F. Frenches which pass under wall footings or within 18" of column footings shall be backfilled with clean concrete. To secure adequate foundation support, the method and depositing of the concrete fill shall be as directed by the Architect. To prevent the concrete from adhering to the conduits, necessary conduit protection shall be applied.

FOUNDATIONS FOR EQUIPMENT/HOUSEKEEPING PADS

- A. Provide all foundations for equipment installed under this specification Division and/or as indicated on plans.
- B. Construct concrete foundations on structural floor slabs or on grade in the manner or as required by the approved shop drawing details of the manufacturer or the utility company.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- C. Provide and install concrete.
- D. Metal reinforcement shall be deformed steel bars or cold drawn steel wire, or fabricated forms of these materials as required.
- E. Furnish anchors of size and number noted, with bottom plates and sleeves.
- F. Forms shall conform to the shape, lines, grades, and dimensions of the concrete, required by the approved shop drawing details of the equipment manufacturers, or approved on the Contractor's Equipment room layouts. They shall be sufficiently tight to prevent leakage of mortar and shall be braced or tied together to maintain position and shape. Forms shall be moved in such napper acto insure the complete safety of the structure.
- G. All exposed corners or edges shall be chamfered. All burrs, fins, irregularities of formin, or spillage shall be removed and the surface float or trowel finished to a smooth, straight surface.
- H. Housekeeping Pads: Provide 4" thick, and size as required by approved shop drawings, concrete pad for all equipment installed on floor. Pad shall be steel reinforced with all ages and surfaces finished as described above. When installing over existing concrete, surface of existing pad shall be prepped using a bushing tool to rough in entire surface. Whether pouring over new or existing concrete, provide U-shaped rebar anchors set in epoxy to secure padao pid.

3.14 ITEMS RECESSED IN MASONRY CONSTRUCTION

- A. Wherever boxes, electric panels, equipment, devices, access panels, and similar items of electrical construction are installed in exposed masonry construction, the Contractor shall utilize and submit for approval items of such size, height, and arrangement to conform to the corresponding masonry unit. The Contractor shall include as part of this contract, the necessary offsets, adjustments and relocations necessary to conform with the instructions of the Architect as to the final location of the equipment item in the exposed masonry.
- B. As part of his contract and before the jurchase of the items hereinbefore mentioned, the Contractor shall notify the Architect of such nonfications in the building arrangement that will be necessary to accommodate the proposed equipment.
- 3.15 ROOF FLASHINGS
 - A. All conduit extending unough roofs shall be provided with watertight flashing and counterflashing as hereinafter described
 - B. Furnish and install standard counterflashing fittings on the conduit or properly designed clamped counterflashing with caulking as directed by the Architect/Engineer.
- 3.16 PAINTING
 - Refinish all factory applied finishes that have been damaged to match the original finish as directed by the Architect.

Prime coat all steel furnished under this Division with material and methods as described in another Section under the heading "PAINTING".

- 3.17 EQUIPMENT CONNECTIONS
 - A. Provide required wiring, raceways and final connections for all equipment provided by this Division and Divisions 1 thru 23.

- B. Make final connections in accordance with wiring diagrams obtained from equipment manufacturer.
- C. Rough-in in accordance with approved shop drawings from the manufacturer or supplier of the equipment. Rough-in prior to shop drawing approval will be subject to change without adjustment to contract cost.

3.18 BALANCING

A. The system of feeder and branch circuits for power and lighting shall be connected to panel busses in such a manner as to electrically balance the connected load as close as is practicable. Should the Owner disclose any unfavorable conditions reacting on the service, this Contractor shall make such changes as may be suggested to balance the load.

3.19 GUARANTEE

- A. All work shall be guaranteed to be free from defects for a period of one year of operation from date of acceptance by the Owner unless otherwise specified in Division 1.
- B. Guarantee shall be extended on an equal time basis for all non- operational periods due to failure within the guarantee period.
- C. Contractor to include an 11 month "walk-thru" of the building system with representatives of the School District, Architect, Engineer and the Construction Ma ager. The purpose is to establish a list of corrective work that relates to operational issues anaterial/installation deficiencies.

END OF SECTION 260000

, FOR BID

SECTION 260055

ELECTRICAL IDENTIFICATION

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. This section is a Division 26 Basic Materials and Methods Section, and is part of each Division 26. Section making reference to electrical identification specified herein.
- 1.2 DESCRIPTION OF WORK
 - A. Types of electrical identification specified in this section include the following:

Cable conductor identification.

Operational instructions and warnings.

Danger signs.

Equipment/system identification signs.

PART 2 – PRODUCTS

- 2.1 MANUFACTURERS
 - A. Subject to compliance with requirements, provide products of one of the following (for each type of marker):

W. H. Brady Co.

Ideal Industries, Inc.

Seton Name Plate Co.

3M Electrical Products

- 2.2 ELECTRICAL IDENTIFICATION MATERIALS
 - A. Provide manufacture is standard products of categories and types required for each application. Where more than single type is specified for an application, selection is Installer's option, but provide single selection for each application.
- 2.3 COLOR CODED PLASTIC TAPE
 - A. Provide manufacturer's standard vinyl tape not less than 7 mils thick by 3/4" wide.

Cours: Unless otherwise indicated or required by governing regulations, provide tape color as indicated in Paragraph 3.2.B.

Tape shall be of Type 3M Scotch 35 for color coding, Scotch Super 33+ for splices and Tem Flex 1700 for general use.

- 4 CABLE/CONDUCTOR IDENTIFICATION BANDS
- A. Provide manufacturer's standard vinyl cloth, self-adhesive cable/conductor markers of wrap-around type; either pre-numbered, plastic-coated type, or write-on type with clear plastic, self-adhesive cover flap; numbered to show circuit identification.

Tetra Tech 200-15704-17001

ADDITION AND RENOVATIONS

2.5 BAKED ENAMEL DANGER SIGNS

A. Provide manufacturer's standard "DANGER" signs of baked enamel finish on 20-gage steel; of standard red, black and white graphics; 14" x 10" size except where 10"x 7" is the largest size which can be applied where needed, and except where larger size is needed for adequate vision; with recognized standard explanation wording (as examples: HIGH VOLTAGE, KEEP AWAY, BURIEP CABLE, DO NOT TOUCH SWITCH).

2.6 ENGRAVED PLASTIC-LAMINATE SIGNS

- A. Provide engraved stock melamine plastic laminate, in sizes and thicknesses indicated, engraved with engraver's standard letter style of sizes and wording indicated, punched for mechanical factening except where adhesive mounting is necessary because of substrate.
- B. Thickness: 1/16" for units up to 20 sq. in. or 8" length; 1/8" for larger unit.
- C. Fasteners: Self-tapping stainless steel screws, except contact-type perparent achesive where screws cannot or should not penetrate substrate.

2.7 LETTERING AND GRAPHICS

A. Coordinate names, abbreviations and other designations used in electrical identification work, with corresponding designations shown, specified or scheduled. Provide numbers, lettering and wording as indicated or, if not otherwise indicated, as recommended by manufacturers or as required for proper identification and operation/maintenance of electrical systems and equipment.

PART 3 - EXECUTION

- 3.1 APPLICATION AND INSTALLATION
 - A. Coordination: Where identification is to be applied to surfaces which require finish, install identification after completion of panting.
 - B. Regulations: Comply with you many regulations and requests of governing authorities for identification of electrical work.
- 3.2 CABLE/CONDUCTOR IDENTIFICATION
 - A. Apply cable/conductor identification on each cable and conductor in each box/enclosure/cabinet where wires of more than one circuit or communication/signal system are present. Match identification with marking system used in panelboards, shop drawings, contract documents, and similar previously established identification for project electrical work.
 - B. Conductor Color Coding:

All conductors used in all systems shall have insulation that is inherently colored. All conductors of a system performing the same function shall be colored alike throughout the project.

Equipment Grounding Conductors:

- a. Standard and/or general feeders or circuits shall be green.
- b. Isolated feeders or circuits shall be green with yellow stripe.
- 3. On larger conductors, where colored insulation is not available, colored tape adhesive vinyl bands 3/4" width may be installed 6" maximum from the end of the conductors. Where passing through pull boxes without splice, each conductor shall be banded.

- 4. Power system conductor colors shall be as follows:
 - a. 120/208 Volt System

Phase A - Black

Phase B - Red

Phase C - Blue

Neutral - White or Gray

3.3 DANGER SIGNS

- A. In addition to installation of danger signs required by governing regulations and authorities, install appropriate danger signs at locations indicated and at locations subsequently identified by Installer of electrical work as constituting similar dangers for persons in or about project.
- B. High Voltage: Install danger signs wherever it is possible, under any vircumstances, for persons to come into contact with electrical power voltages higher than 10-20 olts.

3.4 EQUIPMENT/SYSTEM IDENTIFICATION

- A. Install engraved, plastic laminate sign on each major unit of electrical equipment in building, including central or master unit of each electrical system including communication/signal systems, unless unit is specified with its own self-explanatory identification or signal system. Except as otherwise indicated, provide single line of text, 1/2" high lettering on 1-1/2" high sign (2" high where 2 lines are required), white lettering in black held. Provide text matching terminology and numbering of the contract documents and shop crawing. Provide signs for each unit of the following categories of electrical work:
 - 1. Panelboards, electrical carine's and enclosures.
 - 2. Access panel/doors of elevical facilities.
 - 3. Major electrical witchgtar, main and feeder circuit breakers and/or disconnects..
 - 4. Power transfer equipment.
 - 5. Fire Alarm Mister Station and Annunciator.
 - 6. Paging and Intercommunication Systems
- B. Install signs at locations for best convenience of viewing without interference with operation and maintenance of equipment. Secure to substrate with fasteners, except use adhesive where fasteners should not or cannot penetrate the substrate.

JUNCTION AND PULL BOX IDENTIFICATION

- Emergency Systems: Each junction and pull box cover shall be painted orange. Use black indelible liquid marker to label "EMERG." in 3/8" letters minimum.
- B. Fire Alarm System: Each junction and pull box cover shall be painted red. Use black indelible liquid marker to label "F.A." in 3/8" letters minimum.

C. Feeders Shown on Single Line Diagram: Each junction and pull box shall be marked with black indelible liquid marker with the assigned feeder number "FDR #38" in 3/8" letters minimum.

END OF SECTION 260055

ELECTRICAL IDENTIFICATION 260055-4

Tetra Tech 200-15704-17001

SECTION 260110

RACEWAYS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. The general provisions of the Contract, including the Conditions of the Contract (General Supplementary and other conditions, if any) and Division 1 as appropriate, apply to the Vor specified in this Section.
- B. Refer to Section 260000 for General Provisions Electrical.

1.2 DESCRIPTION OF WORK

A. Types of raceways in this section include the following:

Rigid metal conduit

Intermediate metal conduit

Electrical metallic tubing.

Polyvinyl chloride conduit (Exterior Underground Only)

Flexible metal conduit.

Liquid-tight flexible metal conduit.

Surface raceway.

Wireways.

- 1.3 REFERENCE STANDARDS
 - A. Refer to Section 260000 rol a general description of requirements applying to this Section.
- 1.4 QUALITY ASSURANCE
 - A. Refer to Section 260000 for a general description of requirements applying to this Section.
- 1.5 WARRANTY/OUARANTEE
 - A. All work and materials are subject to the general warranty as described in the General Conditions of the Contract and in Division 1, GENERAL REQUIREMENTS.
- 1.6 COORDINATION

2

The drawings and details there upon are scheme and/or diagrammatic in nature, and indicate the need and intent of the design. These are to be used for general guidance only. It shall be the responsibility of the Electrical Contractor to coordinate, with other Division Subcontractors, the installation of all raceways, raceway supports, junction boxes and required fittings. This coordination will include conduit layout to allow access to equipment for maintenance.

B. This coordination shall be carried out prior to actual installation; this shall be done to eliminate the possibility of conflicts between trades on items such as access, clearances and maintenance issues that may arise after completion of construction.

Tetra Tech 200-15704-17001

RACEWAYS 260110-1

C. Should the coordination not be carried out prior to installation, and a conflict exists, the installing contractor shall remove and reinstall the equipment as required to clear the conflict at no additional cost to the Owner and no delay in project completion.

PART 2 – PRODUCTS

2.1 MATERIALS AND EQUIPMENT

- A. <u>Rigid Metal Conduit</u>:
 - 1. Raceway: Full weight, heavy wall rigid steel with zinc coating conforming to AN
 - 2. Fittings: Cast malleable iron fittings with threaded hubs, insulated throat and zinc protective coating.
 - 3. Subject to compliance with requirements, provide products of one of the following

Allied Tube and Conduit Corporation

LTV Steel Tubular Products Co.

Wheatland Tube

- B. Intermediate Metal Conduit:
 - 1. Raceway: Light weight, rigid steel, hot dipped gabanized manufactured in accordance with UL1242.
 - 2. Fittings: Cast malleable iron fittings with threaded hubs, insulated throat and zinc protective coating.
 - 3. Subject to compliance with requirements, provide products of one of the following:

Allied Tube and Conduit Corporation

LTV Steel Tubular Poluci

Wheatland Tube

- C. <u>Electrical Metal ac Tabing</u>:
 - 1. Raceway: Dight weight, thin wall, rigid steel, hot dipped galvanized manufactured in accordance with ANSI O30.3.
 - 2. Putings: Raintight, insulated throat, compression type with zinc protective coating.
 - Subject to compliance with requirements, provide products of one of the following:

Allied Tube and Conduit Corp.

LTV Steel Tubular Products Co.

Wheatland Tube Co.

- D. <u>Polyvinyl Chloride Conduit</u>:
 - 1. Raceway: Heavy wall, rigid non-metallic, schedule 40 with bell type end, designed for above ground exposed applications, direct earth burial, and concrete encasement.

RACEWAYS 260110-2

Tetra Tech 200-15704-17001

- 2. Fittings: Polyvinyl chloride, heavy duty, glue type, designed for Schedule 40 application.
- 3. Subject to compliance with requirements, provide products of one of the following:

Allied Tube & Conduit

Carlon

Queen City Plastics, Inc.

Scepter Electric Systems

- E. <u>Flexible Metal Conduit</u>:
 - 1. Raceway: Construct of single strip, flexible, continuous, interlocked, and double-wrapped steel, galvanized inside and outside.
 - 2. Fittings: Steel, insulated throat, with zinc protective coating.
 - 3. Subject to compliance with requirements, provide products of one of the following:

AFC

Alflex Corp.

Electri-Flex Company

- F. Liquid-Tight Flexible Metal Conduit:
 - 1. Raceway: Construct of single raip, lexible, continuous, interlocked, and double-wrapped, galvanized inside and outside, coa with liquid-tight jacket of flexible polyvinyl chloride.
 - 2. Fittings: Steel, water and oltight, insulated throat, with zinc protective coating.
 - 3. Subject to compliance with equirements, provide products of one of the following:

AFC

Alflex Corp

Electri Flex Company

G. Surface Race



Dual Service Raceway: Two-piece wireway, base to be .054" galvanized steel, cover to be .040" galvanized steel. Complete unit shall have a cross sectional area of 7.2 square inches. Finish shall be color as selected by Architect. Wireway to be similar to Wiremold Series 4000, Cat #V4000B and V4000C.

Fittings: Wireway shall be provided with a complete line of, but shall not necessarily be limited to, couplings, offsets, elbows, adapters, hold-down clips, end-caps and other components and accessories as needed for a completed system.

3. Subject to compliance with requirements, provide products of one of the following:

Wiremold Co.

Walker, Butler Manufacturing Co.

Hubbell

- H. <u>Wireways:</u>
 - 1. Furnish electrical wireways of the type, size, and style for each service indicated. Wireway shall be a complete assembly including but not necessarily limited to, couplings, offsets, elbower adapters, hold-down clips, end-caps and other components and accessories as needed for a complete system.
 - 2. System shall fulfill wiring requirements as indicated in contract documents, and shall ompy with applicable portions of Article 362 of the National Electrical Code.
 - 3. Subject to compliance with requirements, provide products of one of the rollowin

Circle AW Products Co.

The EMF Company, Inc.

Hoffman Engineering Company

Square "D" Company

I. The above items shall include the statement "Approved Equal and/or "Approved Substitute". This statement requires that the product or item be in compliance with the written intent of this specification and the submission meets the requirements of Section 260000.

PART 3 - EXECUTION

- 3.1 INSTALLATION OF ELECTRICAL RACEWAYS
 - A. Install electrical raceways in accordance with manufacturer's written instructions, applicable requirements of NEC and NECA "Standard of Installation", and complying with recognized industry practices.
 - B. Coordinate with other work as becessary to interface installation of electrical raceways, wireways and required components
 - C. Raceways used a redistribution, feeders, or branch circuits shall be a minimum size of 3/4" or equal equivalent cross sectional area. Raceways used for control and signal shall be a minimum size of 1/2" or equal equivalent cross-sectional area.
 - D. All raceways shall be concealed within the building construction, where indicated on the floor plans surface faceway shall be installed. Should it be impossible or impracticable to install a raceway concealed and surface raceway is not indicated, the Contractor shall consult with the Architect or Engineer for approval prior to installation.

All raceways installed in ceiling cavities and exposed within mechanical spaces shall be run parallel with building lines and installed level and square at the proper elevation/height.

- F. Complete the installation of electrical raceways before starting the installation of cables/wires within the raceway.
- G. Furnish and install one (1) nylon or fiberglass pull cord in each empty raceway. Each empty raceway shall be cleaned, capped, and tagged as to its termination location.

- H. Install liquid-tight flexible metal conduit for connections to motors and for other electrical equipment when subject to movement and vibration, and also where subjected to one or more of the following conditions:
 - 1. Exterior locations.
 - 2. Moist or humid atmosphere when condensation can be expected to accumulate.
 - 3. Corrosive atmosphere.
 - 4. Subjected to water spray.
 - 5. Subjected to dripping oil, grease or water.
- I. Install Electrical Metallic Tubing for building interior electrical work except:
 - 1. Underground
 - 2. In gravel, cinder, concrete or other sub-base floor construction,
 - 3. Horizontal runs in concrete floor slabs.
 - 4. Where exposed to the elements.
 - 5. In masonry construction below finished grade.
 - 6. Vertically in poured concrete walls.
- J. Refer to Section 260000 for excavation, shoring and pumping, concrete and backfilling requirements.
- K. Where and whenever possible, install herizontal electrical raceways as tight to building construction as possible and above water, drain and steampiping. A separation of at least six (6) inches shall be maintained between electrical corduits and not water and steam piping.
- L. In accordance with NEC requirements install Rigid or Intermediate Metal Conduit where Electrical Metallic Tubing is not permitted
- M. In all instances where recessed type panelboards are installed, furnish and install one (1) one inch raceway for each two (2) future circuits for which "space" or "spare" provisions have been made in the panelboard. These accuracy shall extend between the panelboard cabinet and a convenient location above an access panel or a removable tile ceiling construction and capped.

3.2 CLEANING

A. Upon completion of installation of raceways, inspect interiors of raceways; remove burrs, dirt and construction debris.

END OF SECTION 260110



SECTION 260120

WIRES AND CABLES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. This section is a Division 26 Basic Materials and Methods section and is part of each Division 26 Section making reference to wires and cables specified herein.
- 1.2 DESCRIPTION OF WORK
 - A. Electrical wire and electrical cable work is indicated by drawings and specifications.
 - B. Types of wire, cable and connectors in this section include, but not limited to the following:

Copper conductors.

Tap type connectors.

Split-bolt connectors.

- C. Refer to other sections of Division 26 for, but not limiter to, raceways, connections used in conjunction with wire and cable work.
- D. Applications for wire, cable and connectors required for project are as follows unless otherwise indicated:
 - 1. Power Distribution Circuitry.
 - 2. Appliance and Equipment Circu
 - 3. Motor Branch Circuitry.
 - 4. Control Circuitry.
 - 5. Signal/Communicator Cilcuitry.

PART 2 – PRODUCTS

- 2.1 MANUFACTURIES
 - A. <u>Wire and Cable</u>

Anaco da Wire and Cable Co.

dvance Wire and Cable, Inc.

American

Cerro Wire and Cable Co.

Electrical Conductors, Inc.

General Cable Corp.

Hitemp Wires, Inc.

Rome Cable Corp.

Tetra Tech 200-15704-17001

POSt

- Southwire Company
- Triangle PWC,, Inc.
- The Okonite Co.
- General Electric Co.
- Connectors
- Burndy Corp.
- Eagle Electric Mfg. Co., Inc.
- Gould, Inc.
- Ideal Industries, Inc
- Joslyn Mfg. and Supply Co.
- O-Z/Gedney Co.
- Pyle National Co.
- Thomas and Betts Co.
- 2.2 WIRE, CABLE AND CONNECTIONS
 - A. Except as otherwise indicated, provide wire, cable and connectors of manufacturer's standard materials, as indicated by published product information; designed and constructed as recommended by manufacturer, and as required for the installation. Minimum wire and cable size is #12 AWG for power and branch circuits and #14 AWC for control and signal/communication circuits unless otherwise indicated.
 - B. Wire: Provide factory fabricated wire of sizes, ratings, materials and types indicated for each service. Where not indicated, provide proper selection as determined by Installer to comply with project's installation requirement and NEC standards. Select from the following types, materials, conductor configurations, insulation and coverings:
 - UL Type: THHN
 - UL Type: IW
 - EL Type, THWN
 - UL Type: TF
 - UL Type: XHHW
 - UL Type: MC (Metal Clad)
 - Material: Copper
 - Conductors: Solid (AWG 14 to AWG 10 only).
 - Conductors: Concentric-lay-stranded (standard flexibility)

Outer Covering: Nylon

Outer Covering: Thermoplastic

C. Connectors: Provide factory fabricated metal connectors of sizes, ratings, materials, types and classes as required for each service. Where not indicated, provide proper selection as determined by Installer to comply with installation requirements and NEC standards. Select from the following types, classes, kinds and styles.

Type: Pressure

Type: Crimp

Type: Threaded

Class: Insulated

Class: Non-insulated

- Kind: Copper (for CU to Cu connection).
- Style: Butt connection
- Style: Elbow connection
- Style: Combined "T" and straight connection
- Style: "T" connection.
- Style: Split-bolt parallel connection
- Style: Tap connection

Style: Pigtail connection

PART 3 – EXECUTION

- 3.1 INSTALLATION
 - A. Install electrical cables, wires and connectors, in compliance with manufacturer's written instructions, applicable requirements of NEC and NECA's "Standard of Installation", and in accordance with recognized and try practices.
 - B. Coordinate cable and wire installation work with electrical raceway and equipment installation work, as necessary for proper interface. Pull conductors together where more than one is being installed in a aceway. Use pulling compound or lubricate, where necessary; compound must not deteriorate conductor or insulation. Use pulling means including fish tape, cable or rope which cannot damage raceway. Rope must be used as pulling means when pulling wires or cables into plastic conduit and duct. Keep conductor splices to a minimum and install in junction boxes only. No splices shall be permitted within conduit. Install splices and tapes which have mechanical strength and insulation rating equivalent or better than conductor. Use splice and tape connectors which are compatible with conductor material.
- 3.2 FIELD QUALITY CONTROL
 - A. Prior to energization, test cable and wire for continuity of circuitry and also for short circuits. Correct malfunctions when detected.

Tetra Tech 200-15704-17001

WIRES AND CABLES 260120-3

B. Subsequent to wire and cable hook-ups, energize circuitry and demonstrate functioning in accordance with requirements.

END OF SECTION 260120

Repurse

WIRES AND CABLES 260120-4

Tetra Tech 200-15704-17001
ELECTRICAL BOXES & FITTINGS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. This section is a Division 26 Basic Materials and Methods section, and is a part of each Division 26, section making reference to electrical wiring boxes and fittings specified herein.

1.2 DESCRIPTION OF WORK

A. Types of electrical boxes and fittings in this section include the following:

Outlet boxes.

Junction boxes.

Pull boxes.

Conduit bodies.

Bushings.

Locknuts.

Knockout closures.

PART 2 – PRODUCTS

- 2.1 INTERIOR METALLIC OUTLET POXES
 - A. Provide galvanized flat rolled sheet techinterior outlet non-gangable wiring boxes, of types, shapes and sizes, including box depths, to suit each respective location and installation; construct with stamped knockouts in back anoside and with threaded screw holes with corrosion-resistant screws for securing box covers and wiring devices.
 - B. Provide outlet box accessories as required for each installation, including mounting brackets, wallboard hangers, extension rings, fixture studs, cable clamps and metal straps for supporting outlet boxes, which are compatible with outlet boxes being used and fulfilling requirements of individual wiring situations. Choice of accessories is Installer's option.
 - C. Manuacturer. subject to compliance with requirements, provide interior outlet boxes of one of the follow.g:

Appleton Electric Co.

Bell Electric/Square D Co.

Pass and Seymour, Inc.

RACO, Inc.

Steel City/Midland-Ross Corp.

- 2.2 WEATHERPROOF OUTLET BOXES
 - A. Provide corrosion resistant cast-metal weatherproof outlet wiring boxes, of types, shapes and sizes,

including depth of boxes, with threaded conduit ends, cast-metal face plates with spring-hinged waterproof caps suitably configured for each application, including face plate gaskets and corrosion-resistant fasteners.

B. Manufacturer: Subject to compliance with requirements, provide weatherproof outlet boxes of one of the following:

Arrow-Hart Div., Crouse-Hinds Co.

Bell Electric/Square D Co.

Harvey Hubbell, Inc.

O-Z/Gedney Co.

Slater Electric Co.

- C. Refer to Section 260140 WIRING DEVICES for exterior receptacle outet baxes.
- 2.3 JUNCTION PULL BOXES
 - A. Provide galvanized code-gauge sheet steel junction and pull oxxex with screw-on covers; of types, shapes and sizes, to suit each respective location and instalation; with welded seams and equipped with stainless steel nuts, bolts, screws and washers.
 - B. Manufacturers: Subject to compliance with requirements, provide junction and pull boxes of one of the following:

Adalet-PLM Div., Scott and Fetzer Co.

Appleton Electric Co.

Arrow-Hart Div., Crouse-Hinds

Bell Electric/Square D Co.

GTE Corporation

Keystone Columbia, Inc

O-Z/Gedney Co.

Slater Electric C

Sprin, City Elect. Mfg. Co.

Springenty Liceu ning. C

2.4 CONDUIT BODIES

Provide galvanized cast-metal conduit bodies, of types, shapes, and sizes, to suit respective locations and installation, construct with threaded-conduit-entrance ends, removable covers, and corrosion-resistant screws.

Manufacturers: Subject to compliance with requirements, provide conduit bodies of one of the following:

Appleton Electric Co.

Crouse-Hinds Co.

Gould, Inc.

Killark Electric Mfg. Co.

O-Z/Gedney Co.

Spring City Electrical Mfg. Co.

2.5 BUSHINGS, KNOCKOUT CLOSURES AND LOCKNUTS

- A. Provide corrosion-resistant punched-steel box knockout closures, conduit locknuts and instated malleable iron conduit bushings, offset connectors, of types and sizes to suit respectively s and installation.
- B. Manufacturers: Subject to compliance with requirements, provide bushinge knockout closures, locknuts and connectors of one of the following:

GR'

Appleton Electric Co.

Burndy Corp.

Crouse-Hinds Co.

Gould, Inc.

O-Z/Gedney Co.

RACO, Inc.

Steel City/Midland-Ross Corp.

Thomas and Betts Co., Inc.

PART 3 – EXECUTION

- 3.1 INSTALLATION OF ELECTRICAL BOXES AND FITTINGS
 - A. Install electrical boxes and fittings, complying with manufacturer's written instructions, applicable requirements of NEC and NACA's "Standard of Installation", and in compliance with recognized industry practices to ensure that products fulfill requirements.
 - B. Coordinate installation of electrical boxes and fittings with wire/cable and raceway installation work.
 - C. Provide weather proof outlets for interior and exterior locations exposed to weather or moisture.
 - D. Provide knockout closures to cap unused knockout holes where blanks have been removed.
 - E. Astall boxes and conduit bodies in those locations to ensure ready accessibility of electrical wiring.

Avoid using round boxes where conduit must enter box through side of box, which would result in difficult and insecure connections when fastened with locknut or bushing on rounded surface.

- Fasten boxes rigidly to substrates or structural surfaces to which attached, or solidly embed electrical boxes in concrete or masonry.
- H. Provide electrical connections for installed boxes.
- I. Pull boxes and junction boxes shall be furnished and installed in all conduit runs at intervals not exceeding 100 feet maximum.

F.

J. Identify each circuit in all pull boxes and junction boxes whether the box contains one or more circuits.

END OF SECTION 260135

*§*C Repland $\langle \langle \cdot \rangle$

WIRING DEVICES

PART 1 - GENERAL

1.1 DESCRIPTION OF WORK

- A. The extent of wiring device work is indicated by drawings, schedules and specifications. Viring devices are defined as single discrete units of the electrical distribution system which are intended to carry but not utilize electric energy.
- B. Types of electrical wiring devices in this section include the following:

Receptacles.

Switches.

Device plates.

Energy Control Devices

- 1.2 SUBMITTALS
 - A. Product Data: Submit manufacturer's data on electrical wiring devices.

PART 2 - PRODUCTS

- 2.1 ACCEPTABLE MANUFACTURERS
 - A. Manufacturers: Subject to compliance with requirements, provide products of one of the following (for each type of wiring device):

Legrand Co. Hubbell, Inc.

Leviton Mfg. Co.

Lutron Electronics Op., Inc.

Cooper Wiring Square D to.

Eaton Corp.

Niemens

Wattstopper

Sensor Switch

FABRICATED WIRING DEVICES

A. Provide factory fabricated wiring devices, in types, styles, colors, and electrical ratings for applications indicated and complying with NEMA Standards Pub. No. WD 1. Where types and grades are not indicated, provide proper selection as determined by Installer to fulfill wiring requirements, and complying with NEC and NEMA Standards for wiring devices. Color selection to be verified by

Tetra Tech 200-15704-17001

WIRING DEVICES 260140-1

Contractor with Architect/Engineer.

2.3 RECEPTACLES

- A. All simplex receptacles shall be extra heavy duty, 20 amperes, 125 volts, 2 pole, 3 wire grounding, with green hexagonal equipment ground screw, with metal plaster ears, side wiring, NEMA configuration 5-20R unless otherwise indicated. Hubbell Cat. #HBL5361, or approved substitute
- B. All duplex receptacles shall be extra heavy duty, 20 amperes, 125 volts, 2 pole, 3 wire grounding type with green hexagonal equipment ground screw, with metal plaster ears, side wiring NZMA configuration 5-20R unless otherwise indicated. Hubbell Cat. #HBL5362, #HBL8300 or approved substitute.
- C. Special Purpose Receptacles: Provide polarized grounding type special purpose receptacles of the required amperage and voltage ratings, extra heavy duty. Device shall include a green hexagonal equipment ground screw.
- D. All ground fault receptacles shall be extra heavy duty duplex, tamper resistant, 20 amperes, 125 volts, 2 pole, 3 wire grounding type with green hexagonal equipment ground screw, integral ground fault circuit interrupter, UL rated Class A, Group 1, with metal plaster ears, side wiring, NEMA Configuration 5-20R, self-testing with red and green LED indic for fights. Device shall include solid state ground-fault sensing and signalling, with a 5 milliampere ground fault trip level, plus or minus 1 milliampere. Hubbell Cat. #GFR5362SG, or approved substitute.
 - 1. Whether indicated or not on the floor plans, the Electrical Contractor shall furnish and install GFI protected devices in kitchen areas on countertops near sinks, water coolers, refrigerators, on rooftop equipment, on exterior walls; one is indicated by the N.E.C., it shall be the discretion of the Electrical Contractor to provide CEI receptacles or GFI circuit breaker. Receptacles protected by GFI circuit breakers shall be remanently labeled on the faceplate as GFCI.

2.4 USB CHARGING DEVICES

A. All USB charging devices with recorracles shall be extra heavy duty, 20 amperes, 125 volts, 2 pole, 3 wire grounding, with green exagonal equipment ground screw, with metal plaster ears, side wiring, NEMA configuration 5-21R, with teo charging outlet ports, 5VDC, 3.8 amp charging capacity, Hubbell Cat. #USb20X2.

2.5 SWITCHES

A. Toggle Switch: Brovide extra heavy duty, industrial series flush toggle, 1 pole, 2 pole, 3-way, 4-way AC oniet cwhen rated 20 amperes @ 120/277 volts with green hexagonal equipment ground screw, metal paster earls, and side wired screw terminals. Similar to Hubbell Series HBL Series or approved substitute.

Key Switch: Provide extra heavy duty, industrial, 1 pole, 2 pole, 3-way, 4-way barrel key locking switch rated at 20 AMPs @ 120/277 volts with green grounding screw, metal plaster ears and side wired screw terminals. The tumbler shall be a six-point cylinder type. All project keyed switches to be keyed alike. Similar to Hubbell 122*RKL series.

5 DEVICE PLATES

A. Provide switch and receptacle outlet wall plates for wiring devices, of types, sizes, and with ganging and cut outs required by the devices being installed. Construct with metal screws for securing plates to devices; screw heads colored to match finish of plates; plates colored to match wiring devices to which

WIRING DEVICES 260140-2

attached. <u>All emergency receptacles to have red coverplates.</u> Provide device plates possessing the following additional construction features: **Receptacle outlet plates to be permanently marked with panel designation and circuit number on back side of plate.**

- 1. Metal Plates to be stainless steel of non-corrosive and non-magnetic 302 alloy, .032" nominal thickness. Plates shall have brushed satin finish.
- 2. Non-Metallic Plates to be a thermoplastic, virtually indestructible, molded polycarbonate material offering resistance to impact, scratches, discoloration and be self-extinguishing. Plates shall have no-line smooth finish.
- B. Weatherproof device plates shall have spring-hinged waterproof cap suitably configured for each application, including face plate gaskets and corrosion-resistant fasteners. Boxes and devices shall be recessed, weatherproof with smoke gray opaque in-use covers. Intermatic Cate#WP1000(H)GRC.
- C. Existing mechanical spaces where concealed work is impractical, such as he sonry pr block walls, Provide 4" square boxes, surface mounted, with ½" deep surface mounted device plates consisting of same material for devices indicated on plans, whether single or double g. og. Use of plaster flange and standard cover plate will not be acceptable.
- 2.7 ENERGY CONTROL DEVICES (Occupancy Sensors)
 - A. Line Voltage:
 - Combination wall switch and sensor shall be Dual reannology Passive Infrared and Ultrasonic, designed for single gang outlet box installation, with a coverage of 180° for a maximum of 400 square feet. Device shall be suitable for 120077 dual voltage operation, and have vandal resistant, hard sensor lens. Device shall be similar to Sensor Switch Cat. No. WSD-PDT or Wattstopper DW-100 Series, DW 103 Series for multi-way, DW-200 for dual relay, DW-203 for multi-way dual relay, or approved substrute.
 - 2. Ceiling sensor shall be Dual Technology Passive Infrared and Ultrasonic 360° coverage, 1200 square feet maximum. Self Contained Relay Device shall be suitable for 120/277 Dual Voltage operation. Device shall be similar to Sensor Switch Cat. No. CMR-PDT, Wattstopper DT-355 or approved substitute.
 - B. Low Voltage:
 - 1. Ceiling module i sensor shall be Dual Technology Passive Infrared and Ultrasonic with 360° coverage up to 20 feet. Device accepts 12 to 24 volt AC or DC. Device shall be similar to Sensor Switch Car. No. CM-PDT or approved substitute.

Senser power pack shall be a low voltage power supply with an input of either 120 volts or 277 volts AC and an output of 24 volts DC @ 150 mA. Device shall contain a 20 AMP isolated load control relay. When relay is used, power supply output shall be reduced to 24 volts DC @ 114 mA. Device shall be similar to Sensor Switch PP-20 or approved substitute.

Photocontrol

- 1. Provide epoxy conformal coated cadmium sulphide photocell with Lexan impact and vandal resistant enclosure. Dome and base to be ultrasonically welded. Photocell shall respond to the light spectrum near to that of a human eye. Housing shall mount to ¹/₂" conduit and have 180° swivel.
- 2. Photocell shall have on/off time delay, on at 1 to 5 FC, off at 3 to 15 FC. Tool free adjustment.

Tetra Tech 200-15704-17001

WIRING DEVICES 260140-3

Unit shall fail in the ON position.

- 3. Unit shall operate from -40°F to 140°F, with a minimum 5-year warranty.
- 4. Provide Tork 2001 series or approved equivalent.

PART 3 – EXECUTION

3.1 INSTALLATION OF WIRING AND CONTROL DEVICES

- A. Install wiring devices as indicated, in compliance with manufacturer's written instructions applicable requirements of NEC and NECA's "Standard of Installation", and in accordance with recognized industry practices to fulfill project requirements.
- B. Coordinate with other work, including painting, electrical box and wiring work as recessary to interface installation of wiring devices with other work.
- C. Install wiring devices only in electrical boxes which are clean, free from blilding materials, dirt and debris.
- D. Provide electrical connections for wiring and control devices.
- E. Delay installation of all wiring and control devices until wring work is completed.
- F. Isolated Ground Receptacle Devices shall be connected to the system ground by way of an insulated ground conductor color coded green with a yellow stripe.
- 3.2 PROTECTION OF WALL PLATES AND PECEPTACLES
 - A. At time of Substantial Completion, replace bost items which have been damaged, including those burned and scorched by faulty plugs

3.3 GROUNDING

- A. Provide electrically continuous in ht grounding connections for wiring and control devices.
- 3.4 TESTING AND COMPUSITING
 - A. Prior to energizing circuitry, test wiring devices for electrical continuity and proper polarity connections. After corgizing circuitry, test wiring devices to demonstrate compliance with requirements.
 - B. After energying vircuitry, the Electrical Contractor shall test and adjust all control devices to provide optimum operation and performance.

C. All areas where energy control devices are specified shall be verified for full coverage and accurate operation. If any area is determined by the Owner, Architect, or Engineer to have inadequate coverage or operation, Contractor shall provide additional energy control devices to remedy the coverage or operation issue. For bidding purposed, own 5 extra devices fully installed. After successful commissioning, uninstalled devices shall be handed over to the Owner for spare devices. Device types shall be as required for commissioning, or as selected by Owner for space devices as applicable.

END OF SECTION 260140

MOTOR STARTERS

PART 1 - GENERAL

1.1 DESCRIPTION OF WORK

- A. Extent of motor starter work is indicated by drawings, schedules and specifications.
- B. Refer to sections of other divisions of these specifications for driven equipment specified motor starters. Motor starters for such equipment are the work of this section.
- C. Types of motor starters in this section include the following:

Manual.

Magnetic Full Voltage, Non-Reversing.

Combination Disconnect Switch and Magnetic Starter.

Adjustable Frequency Drive (AFD)

- 1.2 SUBMITTALS
- A. Product Data: Submit manufacturer's data on motor starters and accessories.
- 1.3 COORDINATION
 - A. The drawings and details there upon are scheme and/or diagrammatic in nature, and indicate the need and intent of the design. These are to be used for general guidance only. It shall be the responsibility of the Electrical Contractor to coordinate with other Division subcontractors, the installation of all motor starters, the need for contracted evenes including the wiring and conduit, to and from the device.
 - B. This coordination shall be carried out prior to actual installation. This shall be done to eliminate the possibility of conflicts between races on items such as access, clearances and maintenance issues that may arise after completion or epordination.
 - C. During the coordination phase of the project, the Electrical Contractor shall consult with Division 1 thru 23 subcontractors with regard to base design equipment characteristics. Any differences from the electrical plans the specifications shall be considered a change. The trade's contractor making the change at to additional cost to the Owner or delay in project completion shall handle these additional costs

PART 2 PRODUCTS

2.1

ACCEPTABLE MANUFACTURERS

Manufacturers: Subject to compliance with requirements, provide products of one of the following (for each type and rating of motor starter):

Allen-Bradley Co.

Cutler Hammer Products

Furnas Electric Co.

Square D Co.

Siemens

2.2 MOTOR STARTERS

- A. Provide motor starters and ancillary components; of types, sizes, ratings and electrical characteristics indicated which comply with manufacturer's standard materials, design and construction in accordance with published product information, and as required for complete installations.
- B. Fractional HP Manual Motor Starters: Provide manual, single phase, fractional HP motor starters for each motor rated less than 1/2 HP, of types, ratings and electrical characteristics indicated. Equip unit with thermal overload relay for protection of 120 volt AC motors. Provide starters with quiper make, quick-break, trip free toggle mechanisms, selector switches for hand-off-automatic control: nount starter in NEMA Type 1 or Type 4 enclosure as indicated or required by the NEC.
- C. Magnetic Motor Starter: Provide magnetic full voltage, non-reversing starter for each motor rated 1/2 HP and more of types, ratings and electrical characteristics indicated; equip with solid state overload relays, control transformers with 120V secondary, with one secondary fuse and one grounded secondary lead, two normally open and two normally closed auxiliary o ntacts, hand-off- automatic selector switch, red and green pilot lights wired and mounted through iron of the enclosure. Mount starter in NEMA Type 1 or Type 4 enclosure as required by the NEC.
- D. Combination Disconnect Switch Magnetic Starter: Provide Ent-voltage, non-reversing, combination non-fused disconnect switch and magnetic starter for each motor rated 1/2 horsepower and more, of types, ratings and electrical characteristics indicated; equip with solid state overload relays, control transformer with 120 volt secondary, one secondary five and one grounded secondary lead, two normally open and two normally closed auxiliary contacts, hand-off- automatic switch, red and green pilot lights wired and mounted through the form of the enclosure. Mount starter in NEMA Type 1 or Type 4 enclosure as required by the National Electrical Code (NEC).
- E. Three (3) phase, full voltage, non-exercing magnetic motor starters, horsepower rating with minimum NEMA size #0 shall be as follows:

| | NEMA | continuous | Maximum Horsepower 208 Volt 480 Volt | |
|----------|------------|------------|---|-------|
| | Size | Tring | | |
| | \bigcirc | 18 AMPs | 3HP | 5HP |
| | | 27 AMPs | 7-1/2HP | 10HP |
| ~ | 2 | 45 AMPs | 10HP | 25HP |
| \cap | 3 | 90 AMPs | 25HP | 50HP |
| LV | 4 | 135 AMPs | 40HP | 100HP |
| | 5 | 270 AMPs | 75HP | 200HP |

Motor full-load current shall not exceed continuous ampere rating of starter.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- F. Adjustable Frequency Drive (AFD): The AFD shall convert the input AC main power to an adjustable frequency and voltage. The output frequency and voltage of the AFD shall be adjustable to maintain a constant voltage/hertz ratio throughout the operating range. The AFD shall be designed to operate from a 480 volt, three phase, 60 hertz main supply that is within +10% or -10% of nominal line voltage. The AFD control technique shall employ the pulse width modulated (PWM) technology.
 - 1. Ratings:
 - a. The AFD shall be capable of supplying 120% of rated full load current for one mout maximum ambient temperature.
 - b. Unit shall be rated for installation in a power system capable of delivering up to 65,000 RMS symmetrical amperes.
 - c. Minimum power factor shall be .95 throughout the entire speed range
 - d. The AFD efficiency shall be 98% at full speed.
 - 2. Adjustments:
 - a. The acceleration and deceleration ramp rates shall be equivable from 1 to 60 seconds.
 - b. The overload trip shall be adjustable from 0 to 100% of rated output current.
 - c. The current limit shall be adjustable from 60 to 120% of rated output current to maximize starting torque.
 - d. Voltage boost shall be adjustable from 190 h 400% of nominal voltage/hertz ratio at 1 hertz tapering to 100% at 20 Hertz.
 - e. The drive shall provide a control for adjusting the minimum frequency setting up to 45 Hertz and a maximum operating frequency adjustable over a range of 40 to 60 Hertz.
 - 3. Protection:
 - a. A non-adjustable in trantaneous overcurrent trip shall be set to 250% of rated output current.
 - b. AFD protection shall be accomplished with fuseless electronic protective circuits, to protect from the following conditions:
 - short-circuit at AFD output.
 - Cround fault at AFD output.
 - Open circuit at AFD output.
 - Input undervoltage.
 - DC bus overvoltage.
 - Loss of input phase.
 - AC line switching transients.
 - Instantaneous overload.
 - Sustained overload exceeding 100% of rated current.
 - Overtemperature.



- 4. Control:
 - a. All the following operator controls shall be mounted to the front panel which is integral to the AFD:
 - Manual speed potentiometer.
 - Hand-Off-Auto (HOA) switch. The AFD shall accept an input signal of 4 to 20 Ma, DC as an automatic speed reference signal when the AFD is in the automatic mode of operation. The manual speed potentiometer shall control the AFD when the switch is in the manual mode.
 - The AFD shall be furnished with an isolated follower with a setpoint control of 4 to 20 Ma. DC with PI control from an isolated ground signal.
- 5. Operator Interface:
 - a. The AFD shall be furnished with an alphanumeric display and ke mad to allow the operator access to drive modes, parameters and status conditions.
 - b. Operator control and setup functions shall include the following
 - (1) Frequency setpoint
 - (2) Acceleration/Deceleration time
 - (3) Minimum/Maximum Output Frequence
 - (4) Proportional Gain
 - (5) Integral Gain
 - (6) Setpoint
 - (7) Drive Reset
 - (8) Elapsed Time
 - (9) Enable II (Setpoint) Control
 - (10) Autor reference Source Select

Operating status information will consist of the following:

- 1) Frequency Output
- (2) Output Current
- (3) Output Voltage
- (4) Accel/Decel Ramp Time
- (5) Forward/Reverse Direction
- (6) Hand/Auto Local Indicator
- (7) Elapsed Time

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

The diagnostic and fault conditions available via the operator interface will include the following:

- (1) Output Frequency
- (2) Output Current
- (3) Output Voltage
- (4) Shutdown Reference Status
- (5) Jog Status
- (6) Mode of Operation
- (7) Input Signal Levels
- (8) Faults
- (9) Overload Timer Activated
- (10)Motor Current Limit
- 6. Enclosure:
 - a. The enclosure shall be NEMA Type 1 win a dead front and back construction with all components and load, line and control termination fully front accessible. The enclosure shall be self-ventilated and have provisions for top and bottom entry of conduit and wire.

PART 3 - EXECUTION

3.1 INSTALLATION OF MOTOR STARTER

- A. Install motor starters in accordance with manufacture's written instructions, applicable requirements of NEC, NEMA Standards, and NECA's Standard of Installation", and in compliance with recognized industry practices to ensure that products fulfill requirements.
- B. The Electrical Contractor hall consult and cooperate with the Control Contractor in assisting him in making control connections to the automatic position of the selector switch and to the auxiliary contacts.
- C. Motor Data: Before installing wiring for motors and starters, the Electrical Contractor shall consult the respective parties furnishing the equipment and obtain from them all data necessary to properly connect the apparatus, and for selection of thermal overload relays in accordance with motor pameplate. Any variance in loads or electrical characteristics from the contract drawings should be reported to the Engineer before proceeding with the work.

When packaged equipment is furnished, all unit starters shall be furnished, mounted and wired by the installing contractor. The Electrical Contractor shall furnish and install a disconnect switch, as specified in Section 260170, and wire between unit's main terminal block and the disconnect switch.

- . When packaged rooftop equipment is furnished, the unit disconnect switch and all starters shall be furnished, mounted and wired by the installing contractor. The Electrical Contractor shall wire between the line side of the disconnect switch and the building system.
- F. Provide connections for motor starters.

ADDITION AND RENOVATIONS

3.2 ADJUST AND CLEAN

- A. Inspect operating mechanisms for malfunctioning and where necessary adjust units for free mechanical movement.
- B. Touch-up scratched or marred surfaces to match original finish.
- 3.3 FIELD QUALITY CONTROL
 - A. Subsequent to wire/cable hookup, energize motor starters and demonstrate functioning of equipment in accordance with requirements.

END OF SECTION 260155

PANELBOARDS

PART 1 - GENERAL

1.1 DESCRIPTION OF WORK

- A. Extent of panelboard load-center and enclosure work, including cabinets and cutout boxes, is indicated, by drawings and schedules.
- B. Types of panelboards and enclosures in this section include the following:

Lighting and Appliance Panelboards.

Distribution Panelboards.

1.2 SUBMITTALS

- A. Product Data: Submit manufacturer's data including specifications, restallation instructions and general recommendations, for each type of panelboard required. Include data substantiating that units comply with requirements.
- B. Shop Drawings: Submit dimensioned drawings of panelbox ds and enclosures showing layouts of enclosures and required individual panelboard devices, including by not necessarily limited to, circuit breakers, contactors, and accessories, including wing dagrams of contactors.

1.3 COORDINATION

- A. The drawings are scheme and/or diagrammatics in nature, and indicate the need and intent of the design. These are to be used for general guidance only. It shall be the responsibility of the Electrical Contractor to coordinate, with other Desision Subcontractors, the installation of all raceways, raceway supports, junction boxes and re dired öttings. This coordination will include conduit layout to allow access to equipment for maintenance
- B. This coordination shall be can jee out prior to actual installation; this shall be done to eliminate the possibility of conflicts bet reer trades on items such as access, clearances and maintenance issues that may arise after completion of construction.
- C. Should the coordination not be carried out prior to installation, and a conflict exists, the installing contractor shall remove and reinstall the equipment as required to clear the conflict at no additional cost to the swine and no delay in project completion.

PART 2 – PRODUCTS

2.1 CCEPTABLE MANUFACTURERS

Manufacturers: Subject to compliance with requirements, provide products of one of the following (for each type of panelboard and enclosure):

Cutler Hammer, Inc. (Eaton)

Square D Company

Siemens

2.2 PANELBOARDS

- A. General:
 - 1. Panelboards shall comply with the following industry standards:
 - a. UL Listing/Approval
 - b. UL Standards:

Panelboards - UL67

Cabinet & Boxes - UL50

- c. National Electric Code
- d. NEMA Standard -PBI
- 2. Interiors:



- a. All interiors shall be completely factory assembled. They shall be to designed that switching and protective devices can be replaced without distubing adjacent units and without removing the main bus connectors, so that circuits may be changed without machining, drilling and tapping.
- b. Branch circuits shall be arranged using double row construction. A nameplate shall be provided listing panel type and rating.
- c. Unless otherwise noted, full size instance seutral bars shall be included. Bus bar taps for panels with single pole branches shall be arranged for sequence phasing of the branch circuit devices. Neutral bussing shall have a satable lug for each outgoing feeder requiring a neutral connection. A ground bus while included in all panelboards.
- 3. Boxes: Boxes shall be a minimum 20 inches wide and manufactured from galvanized steel. Provide minimum gutter space in accordance with the National Electric Code.
- 4. Trim:
 - a. Switching divice handles shall be accessible. Panel access doors shall not uncover any live parts. Doors shall have flush type cylinder lock and catch except doors over 48" in height shall have auxiliary fastenings top and bottom of door in addition to the flush type cylinder lock and catch. Panelboard trim clamps shall be of the indicating type. Upon removal of sprews behind door, the panel interiors become service accessible via piano hinged trim front.



- . Anel access door hinges shall be concealed. All locks shall be keyed alike; directory frame shall be welded metal and having a transparent cover shall be furnished with each door.
- c. All exterior and interior steel surfaces of the trim shall be properly cleaned, primed with a rust inhibiting phosphatized coating and finish with a gray ANSI 61 paint. Trims for flush panels shall overlap the box for a least 3/4 inch all around. Surface trims shall have the same width and height as the box. Trims shall be mountable by a screwdriver and without the need for special tools.
- 5. Main Bus and Branch Circuits: All main bus bars shall be full size aluminum, sized in accordance with U.L. standards to limit the temperature rise on any current carrying part to a maximum of 50 degrees C above an ambient of 40 degrees C maximum.

PANELBOARDS 260160-2

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- B. Distribution Panelboards:
 - 1. Panels shall be provided with molded case circuit breakers tested and U.L. labeled per U.L. 489.
 - 2. Circuit breakers 100 ampere through 400 ampere frame sizes shall be thermal-magnetic trip with inverse time current characteristics.
 - 3. Where multiple pole circuit breakers are indicated, provide with common trip so overload on one pole will trip all poles simultaneously. Molded case circuit breakers shall have a minimum 22,000 symmetrical RMS interrupting capacity at 240 volts.
- C. Lighting and Appliance Panelboards:
 - 1. Provide switching and protective devices in quantities, ratings, types indicated, with anti-turn solderless pressure type lug connectors approved for copper conductors. Consult breakers shall be the bolt-on, molded case, thermal magnetic type, with toggle handles that indicate when tripped. Where multiple pole circuit breakers are indicated, provide with common trip so overload on one pole will trip all poles simultaneously.
 - 2. Panelboards for use at 240 volts AC maximum shall incorporate circuit breakers as shown rated at 10,000 A.I.C. symmetrical at 240 volts.
 - 3. Panelboards for use at 480/277 volts AC maximum shakincorporate circuit breakers as shown rated at 14,000 A.I.C. symmetrical at 480 volts
- D. Split Bus Panelboards: Provide dead-front safety type split bus panelboards as indicated in contract documents; switching and protective devices in quantities, ratings, types shown; with anti-turn solderless pressure type lug connectors approved for copper conductors; construct unit for connecting feeders; equip with aluminum bus bars are mumul full-sized neutral bar, equipment ground bus, with bolt-on molded-case thermal magnetic circuit breaker types for each circuit, with toggle handles that indicate when tripped; where manyle pore breakers are indicated, provide with common trip so overload on one pole will trip al poles simultaneously, and provide panelboards fabricated by same manufacturer as enclosures, and which mate properly with enclosures. Provide in panel mechanically held magnetic contactors with contacts suitable for service indicated of sizes and ratings shown.

PART 3 – EXECUTION

3.1 INSTALLATION OF PANELBOARDS

- A. Install panel boards and enclosures where indicated in contract documents and, in accordance with the equipagent manufacturer's written instructions, applicable requirements of NEC and NECA's "Standard of Installation", and in compliance with recognized industry practices to ensure that products fulfill requirements.
 - Anchor enclosures firmly to walls and structural surfaces, ensuring that they are permanently and mechanically secure.
 - Provide all required electrical and grounding connections within the panelboards and enclosures.
- D. The Electrical Contractor shall furnish and install on the door within each enclosure, a circuit labeling identification system for all electrical panelboards. The system must satisfy the NEC Article No. 110-22. The directories shall be typed, NOT handwritten. Directories shall indicate room numbers as indicated on contract documents and room numbers as physically labeled in the field.

ADDITION AND RENOVATIONS

- E. The Electrical Contractor shall provide directories compiled using a software program that is Windows compatible. Program shall handle multiple panels, calculate panel electrical loads from user supplied data, maintain a history of repairs and upgrades by circuit, and be capable of printing panel directories and summaries. Verify compatibility with Owner's operating system.
- F. Provide two discs to owner containing software and project panel directories and summaries.

Republication

END OF SECTION 260160

PANELBOARDS 260160-4

<

MOTOR AND CIRCUIT DISCONNECTS

PART 1 - GENERAL

1.1 DESCRIPTION OF WORK

- A. Extent of motor and circuit disconnect switch work is indicated by drawings and schedules.
- B. Types of motor and circuit disconnect switches in this section include the following:

Equipment disconnects.

Appliance disconnects.

Motor-circuit disconnects.

1.2 SUBMITTALS

- A. Product Data: Submit manufacturer's data including specifications installation instructions and general recommendations, for each type of motor and circuit disconnect switch required.
- 1.3 COORDINATION
 - A. The drawings are scheme and/or diagrammatic in nature, and indicate the need and intent of the design. These are to be used for general guidance only. It hall be the responsibility of the Electrical Contractor to coordinate, with other Division Subconductors, the installation of all motor and circuit disconnect switches, supporting hardware, including wiring and conduit, to and from the equipment. This coordination will include conduit layor to ellow access to equipment for maintenance.
 - B. This coordination shall be carried outprior to actual installation; this shall be done to eliminate the possibility of conflicts between trades on items such as access, clearances and maintenance issues that may arise after completion of construction.
 - C. Should the coordination no be carried out prior to installation, and a conflict exists, the installing contractor shall remove and reinstall the equipment as required to clear the conflict at no additional cost to the Owner and no delay in project completion.

PART 2 – PRODUCTS

- 2.1 ACCEPT. BLE MANUFACTURERS
 - A. Manufacturer: Subject to compliance with requirements, provide products of one of the following (for each type of switch):

Cutler-Hammer, Inc. (Eaton)

Square D Company

Siemens

- .2 FABRICATED SWITCHES
- A. Safety Switches: Safety switches shall be of sizes noted on the drawings, fusible or non-fusible and contained in a general purpose enclosure. All switches shall be type HD and have quick-make, quick-break operation. All switches shall be of proper horsepower rating as applicable and have dual interlocks designed to interlock the switch box door with the switch operating mechanism. Unit shall

be provided with a suitable means of interlock release. An arrangement shall be provided for locking the operating handle in the "ON" or "OFF" position. Safety switches shall have the proper type metal enclosure, i.e., standard, weatherproof, etc., to suit their specific location as required by the National Electrical Code.

- B. Fuses: Provide fuses for safety switches, as recommended by switch manufacturer, of classes, type and ratings needed to fulfill electrical requirements for service indicated.
- C. When packaged rooftop equipment is furnished, the unit disconnect switch shall be furnish mounted and wired by the installing contractor.
- D. When rooftop exhaust fans rated less than 1/2 HP at 120 volts, single phase, are furnished, except utility sets, the unit disconnect switch shall be furnished, mounted and wired by the installing contractor.

PART 3 - EXECUTION

- 3.1 INSTALLATION OF MOTOR AND CIRCUIT DISCONNECT SWITCH
 - A. Install motor and circuit disconnect switches where indicated, complying with manufacturer's written instructions, applicable requirements of NEC, NEMA, and NECA's "Standard of Installation", and in accordance with recognized industry practices to ensure that products fulfill requirements.
 - B. Install disconnect switches used with motor-driven apprances, and motors and controllers within sight of controller position unless otherwise indicated.

TION 260170

C. Provide electrical connections for motor and gircul disconnect switches.

END OF SE

J.F.ORB

GROUNDING

PART 1 - GENERAL

- 1.1 DESCRIPTION OF WORK
 - A. Types of grounding in this section include the following:

Grounding:

Underground metal piping.

Underground metal water piping.

Grounding rods.

Enclosures.

Systems.

Equipment.

Building Structural Steel (Bonding)

PART 2 – PRODUCTS

2.1 GROUNDING

A. Except as otherwise indicated, provide each electrical grounding system indicated, with assembly of materials including, but not necessarily limited to, cables/wires, connectors, terminals (solderless lugs), and other items and accessories meded for complete installation. Where materials or components are not otherwise indicated, comply with NEC, NEMA, and established industry standards for applications indicated.

211

B. Provide conduit, tube, auch sabe and fittings complying with Division 26 Basic Materials and Methods section, "Raceways" in accordance with the following listing:

Rigid steel conduit.

Electrical metalic libing.

Flexible meal conduit.

Liquid-ight flexible metal conduit.

igid metal conduit fittings.

EMT fittings.

Flexible metal conduit fittings.

Liquid-tight flexible metal conduit fittings.

- 2.2 ELECTRICAL GROUNDING CONDUCTORS
 - A. Unless otherwise indicated, furnish a green insulated equipment grounding conductor for all feeders and branch circuits, matching power supply wiring materials and sized according to NEC.

Tetra Tech 200-15704-17001

GROUNDING 260452-1

2.3 BONDING PLATES, CONNECTIONS, TERMINALS & CLAMPS

A. Provide electrical bonding plates, connectors, terminals and clamps as recommended by bonding plate, connector, terminal and clamp manufacturers for applications.

2.4 GROUND RODS & PLATES

A. Ground Rods: Steel with copper welded exterior, 3/4" dia. x 10'.

PART 3 – EXECUTION

- 3.1 INSTALLATION OF GROUNDING SYSTEMS
 - A. Install electrical grounding systems in accordance with manufacturer's written instructions and with recognized industry practices to ensure grounding complies with requirements. Comply with requirements of NEC, NESC, NEMA and UL standards for installation of grounding systems.
 - B. Coordinate with other electrical work as necessary to interface installation of grounding system with other work.
 - C. Clamp cable connections to ground rods.
 - D. Install bonding jumpers with ground clamps on water meter piping to electrically bypass water meter.
 - E. Install clamp-on connectors only on thoroughly cleaned metal contact surfaces, to ensure electrical conductivity and circuit integrity.
- 3.2 FIELD QUALITY CONTROL
 - A. Upon completion of installation of electrical grounding system, test ground resistance with ground resistance tester. Where tests show resistance-to-ground is over 25 ohms, take appropriate action to reduce resistance to 25 ohms or less by driving additional ground rods and/or by chemically treating soil encircling ground rods with sodium coloride, calcium chloride, copper sulphate, or magnesium. Then retest to demonstrate compliance.

END OF SECTION 260452

1 Fr

DISTRIBUTION CIRCUITS

PART 1 – GENERAL

1.1 DESCRIPTION OF WORK

- A. Distribution circuit work is indicated by drawings and schedules.
- B. The distribution circuits shall include furnishing and installing a complete wire and cond between distribution panelboards and branch circuit panelboards.
- Types of equipment to be furnished and installed in this section include the following C.

Rigid Metal Conduit

Intermediate Metal Conduit (IMC)

JOIN CPI Electrical Metallic Tubing (EMT)

PVC (Below Slab Only)

Wires and Cables

Junction Boxes

Pull Boxes

Conduit Bodies

Bushings

Locknuts

Supporting Devices

PART 2 - PRODUCTS

- 2.1 DISTRIBUTION CRCU
 - Furnish and instal each distribution circuit indicated, with assembly of materials, including but not necessarily limited to, conduit, wire, pull boxes, junction boxes and other items and accessories A. needed for a complete installation. Where materials or components are not otherwise indicated, comery with NEC, NEMA and established industry standards for applications indicated.

PART - EXECUTION

INTALLATION OF DISTRIBUTION CIRCUITS

Install distribution circuits complying with equipment manufacturer's written instructions, applicable requirements of NEC, NEMA, and NECA's "Standard of Installation", and in accordance with recognized industry practices.

B. Multiple circuits within a single raceway shall not be permitted under this section.

END OF SECTION 260470



FEEDER CIRCUITS

PART 1 – GENERAL

1.1 DESCRIPTION OF WORK

- A. Feeder circuit work is indicated by drawings and schedules.
- B. The feeder circuits shall include furnishing and installing a complete wire and conduit system distribution panelboards and major 3 phase loads, between power panels and 3 phase bot
- C. Types of equipment to be furnished and installed in this section include the following
 - **Rigid Metal Conduit**

Electrical Metallic Tubing (EMT)

John Cpi Intermediate Metal Conduit (IMC)

Wires and Cables

Junction Boxes

Pull Boxes

Conduit Bodies

Bushings

Locknuts

Supporting Devices

- PART 2 PRODUCTS
- 2.1 FEEDER CIRCUITS
 - Furnish and installench feeder circuit with assembly of materials, including but not necessarily limited A. boxes, junction boxes and other items and accessories needed for a complete to, conduit, wire pr installation ben materials or components are not otherwise indicated, comply with NEC, NEMA and established industry standards for applications indicated.

PART 3 - EXECUTION

INSTALLATION OF FEEDER CIRCUITS 3.1

Insell feeder circuits, complying with equipment manufacturer's written instructions, applicable requirements of NEC, NEMA and NECA's "Standard of Installation", and in accordance with recognized industry practices.

Multiple circuits within a single raceway shall not be permitted under this section.

END OF SECTION 260471



BRANCH CIRCUITS

PART 1 – GENERAL

1.1 DESCRIPTION OF WORK

- A. Branch circuit work is indicated by drawings.
- Β. The branch circuits shall include furnishing and installing a complete wire and conduit or between panelboards and lighting fixtures, receptacles, fractional horsepower motors, and se phase loads.
- Types of equipment to be furnished and installed in this section include the C.

Rigid Raceways - See Section 260110

Electrical Metallic Tubing (EMT)

DINGRI MC (Metal Clad) (Concealed Work only)

Wires and Cables

Junction Boxes

Pull Boxes

Conduit Bodies

Bushings

Locknuts

Supporting Devices

PART 2 – PRODUCTS

ent:

- 2.1 **BRANCH CIRCUITS**
 - Furnish each brunch souit with an assembly of materials, including but not necessarily limited to, A. , why pull boxes, junction boxes and other items and accessories needed for a complete conduit, wir . Where materials or components are not otherwise indicated, comply with NEC, NEMA installation ablis ed industry standards for applications indicated. and es

2.2 NIENCE BRANCH CIRCUITS CON

- The intent of this portion of the specifications is to describe the requirements of a convenience circuit as it applies to 120-volt receptacles.
- 2. All convenience branch circuits may consist of more than one 120 volt receptacle.
- B. Convenience Circuit - General: A circuit consisting of a phase and neutral conductor, which may share its neutral with other phase conductors provided that the neutral conductor does not become overloaded due to circuit phase relationship. This type of circuit shall also include an equipment grounding conductor as described under the grounding section of the specifications.

C. Convenience Circuit - Dedicated: A circuit consisting of a phase and neutral conductor which DOES NOT share conductors with any other circuits. This type of circuit shall also include an equipment grounding conductor as described under the grounding section of the specifications.

PART 3 - EXECUTION

- 3.1 INSTALLATION OF BRANCH CIRCUITS
 - A. Install branch circuits, complying with equipment manufacturer's written instructions, applicable requirements of NEC, NEMA, and NECA's "Standard of Installation", and in accordance with recognized industry practices.
 - B. Multiple circuits within a single raceway or cable shall be permitted under this section. I shall be the responsibility of the Electrical Contractor to assure that the neutral conductors do not become overloaded due to circuit phase relationship, and isolated grounds not become voiled or compromised due to miswiring or wrong connections.
 - C. The Electrical Contractor may elect to use metal clad cable in lieu of electrical metallic tubing (EMT) in wall cavities, and/or above tile or dry wall ceilings. In all areas of exposed construction, electrical metallic tubing (EMT) shall be installed.

END OF SECTION 26047

OR BIDDIN

LIGHTNING PROTECTION SYSTEMS

PART 1 – GENERAL

1.1 DESCRIPTION OF WORK

- A. Lightning protection system work is indicated by specifications.
- B. Types of lightning protection system equipment and components specified in this section include following:

Air terminals.

Conductors.

Connectors.

Cable Splicers.

Ground rods.

Rod clamps.

Bonding plates.

- 1.2 QUALITY ASSURANCE
 - A. ANSI/NFPA Compliance: Comply with NEC and NFPA No. 780, "Lightning Protection Code", as applicable to materials and installation of lightning protection components, and wiring.

PU

- B. UL Compliance: Comply with UL V6. "Lightning Protection Components" pertaining to design, materials and sizing of lightning protection components and devices. Provide components and devices which are UL listed and labeled.
- 1.3 SUBMITTALS
 - A. Product Data: Submit manufacturer's data on lightning protection equipment, components and devices.
 - B. Shop Drawings. Submit dimensioned layout drawings of all lightning protection system equipment, components and revices including conductor sizing, routing and connections.
 - C. UL Certificate: Provide Owner with UL Master Label for overall system which shall be suitable for fastening to building for display. Comply with UL 96A, "Installation Requirements for Lightning Rotection Systems."

– PRODUCTS

ACCEPTABLE MANUFACTURERS

. Manufacturer: Subject to compliance with requirements, provide lightning protection and components of one of the following manufacturers and their representative:

East Coast Lightning Equipment

Erico Lightning Protection

Heary Bros., Inc.

National Lightning Protection Corp.

Denver, Colorado 80216

Thompson Lightning Protection, Inc.

- 2.2 LIGHTNING PROTECTION SYSTEM COMPONENTS
 - A. Provide lightning protection system equipment and components, of types, sizes, and ratings for service indicated, which comply with manufacturer's standard materials, design and construction in accordance with published product information, and as required for a complete installation. Where type or material is not otherwise indicated, comply with NFPA 780 and UL 96 Standards.

2.3 ANCILLARY COMPONENTS

- A. Air Terminal: Copper with low solid bronze base; 3/8 inch diameter extending 10 inches above the object they protect.
- B. Conductors: Copper cable; 28 strand, 16 gauge; 220 lb. per 1000 feet; structural steel columns may be substituted for the down conductors.
- C. Connectors: Solid bronze cable connector with stainless stee pressure screws.
- D. Ground Rod: Copper clad, 5/8 inch diameter x 10 jeet
- E. Rod Clamp: Bronze ground rod clamp with stainless bolts.
- F. Bonding Plate: Cast bronze flat metal bonding plate with mild steel welding plate 1/4"x4"x4" for attachment to steel columns or beams. Steel plate to be electrically welded to the steel column or beam. Plate shall be furnished with four stainless steel bolts and nuts to attached bonding plate welding plate.

PART 3 – EXECUTION

- 3.1 INSTALLATION OF LIGHT LING PROTECTION SYSTEMS
 - A. Install lightning protection systems, in accordance with equipment manufacturer's written instructions, in compliance with applicable requirements of NFPA 780 and with UL 96A lightning protection standards, o ensure that lightning protection system complies with requirements.
 - B. Interconnece metals as required by the Underwriters' Laboratories code such as cold water pipe, sewer, etc., with lead coated copper strap type pipe bonding clamps.
 - C. Concealed down conductors within building construction where applicable.
 - Coordinate with roofing work, as necessary to interface the installation of the lightning protection system.

Install conductors with direct paths from air terminals to ground connections. Do not use metal casings of structure/equipment as a ground path. Avoid sharp bends and narrow loops.

- 3.2 TESTING
 - A. Upon completion of installation of lightning protection system, test resistance-to-ground level. Where tests show resistance-to-ground is over 25 ohms, the Contractor shall take appropriate action to reduce

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

resistance to 25 ohms or less by driving additional ground rods and/or treating soil in the proximity to the ground rod with sodium chloride, calcium chloride, copper sulphate, or magnesium. Then retest to demonstrate compliance with requirements.

END OF SECTION 260601

Republication

Tetra Tech 200-15704-17001

<



EMERGENCY GENERATOR SYSTEMS

PART 1 - GENERAL

1.1 DESCRIPTION OF WORK

- A. Extent of Emergency Generator System work is indicated by drawings, schedules and specific dop
- B. Types of Emergency Generator System equipment required for project include the follo

Diesel engine-driven generators.

Weatherproof Enclosure

Fuel tank and piping

Battery Charger

Generator Support Panel

- 1.2 SUBMITTALS
 - A. Product Data: Submit manufacturer's data on engine driven electric generator systems and components.
 - B. Shop Drawings: Submit dimensioned drawings of angine driven generator units and accessories, including but not limited to the following:
 - 1. System schematic diagram showing all proving and wiring interconnections, sizes and quantities.
 - 2. Installation fact sheet giving fuel, coolart, lubricating oil, exhaust, ventilation, and other pertinent requirements.
 - 3. Complete piping, conduit, electric power and control schematics, and flow diagrams.
 - 4. Engine generator antractosure elevations (1/10th scale or larger) showing the locations, size, and dimensions of ell required Owner interfaces to the package.
 - 5. Ladder type chematic electrical diagrams with legend identifying all devices on diagrams.
 - 6. Factory certified horsepower and fuel consumption data.

PART 2 – PRODUCTS

2.1 ACCEPTABLE MANUFACTURERS

Manufacturer: Subject to compliance with requirements, furnish Emergency generator systems of one of the following:

Caterpillar, Inc.

MTU Onsite Energy

Generac Industrial Power

Kohler Co.

Cummins Power Generation

ADDITION AND RENOVATIONS

2.2 ENGINE GENERATOR UNIT

- A. The following specifications are designed and written around the characteristics of a Detroit Diesel Engine Generator and represent the minimum requirements for all other listed manufacturers.
- B. Diesel Engine Generator: Furnish an alternating current generator unit as indicated in contract documents, with a standby rating of 250KW/313KVA @ 277/480 volt, 3 phase, 4 wire, 60 Hertz, 0.8 percent power factor. Furnish all the following components, accessories and construction features as required for a complete and satisfactory operating system. 3 phase, 4 wire, 60 Hz, 0.8 percent power factor, 750 KW, diesel engine. All accessories and the following components and construction features as required for a complete and satisfactory operating system.

2.3 ENGINE

- A. The engine shall be six (6) cylinder, 2 or 4 cycle, water cooled, turbo-charged, at er-cooled, with not less than 552 cubic inch displacement. Engine speed shall be governed by a g ar driven governor to maintain generator frequency within 0.5% Hertz from no load to 100% rated load.
- B. A 24-volt D.C., negative ground electric starting system consisting of minimum of 625 AMPs cranking current at an ambient temperature of 0 deg. F, A 65 AMP battely alternator, battery racks and a complete set of battery cables.
- C. Batteries shall be selected and furnished to comply with NFF1 110, Level 1 starting requirements. Each battery (two required) shall be 12 volt, maintenance free, lead-calcium hybrid type with sealed cells. The batteries shall be commissioned according to buttery manufacturer's instructions.

2.4 ALTERNATOR

- A. The alternator shall be four (4) pole, rotating held, self-ventilated, dripproof construction. Class "H" insulation system per NEMA MG1-166, with standard 130 deg. C temperature rise at standby power rating.
- B. Furnish skewed rotor and 2/3 pitc, wildings to smooth voltage wave form, minimize field heating and voltage harmonics.
- C. Rotor shall be a dynamically balanced assembly, with a single bearing and direct coupled to engine by a flexible drive disc. Furnish full amortisseur (damper) windings to help minimize voltage deviations and heating effects and er unbalanced load conditions.
- D. The voltage regulator shall be solid state and furnish torque-matched underfrequency compensation to optimize motor starting performance and to assist the engine during transient load conditions. Voltage regulation from no load to full load shall be +/- 2%.
- E. The exciter shall be a permanent magnet, field rotating, brushless armature and shall power the main alternator field windings through shaft mounted, three (3) phase, full wave silicon diode rectifiers. Semi-conductor surge suppressors shall protect the diodes from transient overvoltages induced by load surges.

The shunt excitation system shall derive its power from the main output of the alternator.

- 2.5 CONTROL PANEL
- A. Furnish a unit-mounted, automatic start, level 1 control panel, with suitable vibration isolators. Panel shall consist of, but not be limited to, the following:

Equipment

PJ

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- 1. Errorproof wiring harness for electrical connections.
- 2. Lamp test switch.
- 3. Cyclic cranking
- 4. Overcrank and starter unmesh protection.
- 5. Two-wire remote start/stop terminals.
- 6. AC interlock to prevent starter re-engagement with engine running.
- 7. Overspeed detection.
- 8. Voltage-adjust Rheostat +/- 5%.
- 9. Run OFF/RESET Auto switch.
- 10. Emergency stop pushbutton.
- 11. Low coolant level detection.
- 12. D.C. circuit protection.
- 13. Panel lamps (2).
- 14. Cool-down timer (5 minutes)
- 15. Alarm horn and silencing switch.

Instruments

- 1. A.C. Voltmeter, 3-1/2", 2% full scale accuracy
- 2. A.C. Ammeter, 3-1/2", 2% full so le accuracy.
- 3. A.C. Frequency meter 3-102", 0.5% full scale accuracy.
- 4. D.C. Voltmeter.
- 5. Engine water ter perature.
- 6. Engine on temperature.
- 7. Running time meter.
 - Phase selector switch, seven (7) position.
- dicator Lamps
 - Overcrank
- Low oil pressure
- 3. High engine temperature
- 4. Overspeed
- 5. Emergency stop
- 6. Not-In-Auto

- 7. System ready
- 8. Low battery volts
- 9. Battery charge fault
- 10. Low fuel.
- 11. Prealarm high engine temperature
- 12. Prealarm low oil pressure.
- 13. Low water temperature.
- 14. Auxiliary alarm
- 15. Auxiliary prealarm
- 16. Air damper.
- B. Furnish and install all required control wiring, fuses, fuse blocks, terbina blocks, nameplates, fault contacts, auxiliary contacts, and metering current transformers
- 2.6 COOLING SYSTEM
 - A. Engine shall have a unit-mounted radiator with engine-driven cooling fan. The radiator shall be sized to adequately cool the engine under full load conditions as outlined, in a 125 degrees F ambient temperature and have adequate capacity for additional heat radiated by engine. A fan and radiator guard shall be included.
 - B. Anti-freeze shall be a 50% mixture or envice glycol and water and shall contain a suitable rust inhibiting agent and be installed in the cooling system. The unit shall be furnished with, as a minimum total replacement, an additional apply 0500% ethylene glycol and water coolant mixture.
- 2.7 EXHAUST SYSTEM
 - A. Furnish one (1) critical grade stencer, with a side inlet and end outlet configuration. Inlet and outlet shall be NPT thread.
 - B. Silencer shall include an ondensate drain plug, and be mounted on the enclosure roof and piped to the engine by prease of a stainless-steel exhaust flexible piping.
 - C. Silencer outlet and shall have a 90-deg. exhaust pipe extension terminating vertically, with a counterbalanced rain cap.

2.8 ENGINE HEATERS AND ACCESSORIES

Coclant heater shall be a 2500 watt, 240 volt, single phase thermostatically controlled device. The heater shall be furnished, installed and wired at the factory. Furnish and install a low water temperature alarm contact to close when water temperature falls below 50 deg. F. Interconnect the alarm contact device to the proper alarm terminals in the generator control panel and remote alarm annunciator.

B. Lube oil heater shall be a 150 watts, 120 volts, single phase thermostatically controlled device. The heater shall be furnished, installed and wired at the factory.
C. Battery heaters shall be a thermostatically controlled, low wattage pad type device, suitable for operation on a 120 volt, single phase circuit.

2.9 BATTERY CHARGER

- A. Charger shall be a fully automatic, SCR, float/equalize battery charger. The 24 volt, 10 AMP, silicon controlled rectifier shall be a constant voltage, current limiting charger designed to be permanently connected for float/equalize charging of lead acid starting batteries. The charger shall furnish automatic "Fload-to-Equalize" operation with individual potentiometer adjustments, and shall charge a minimum of 12 lead-acid maintenance free battery cells.
- B. Charger shall be furnished with an oversized transformer and heatsink to allow for constant current charging at 10 AMPs, up to the equalize voltage settings.
- C. The charger shall be furnished in a NEMA 1, general purpose enclosure, with the following equipment, components and features:
 - 1. DC voltmeter
 - 2. DC ammeter
 - 3. ON/OFF power switch
 - 4. Input and output fuse protection and terminal blocks
 - 5. Operational monitors shall provide visual output as well as individual Form C relay contacts for the following:
 - a. Battery Charger Fault: N.O. contacts close on loss of A.C. input or loss of D.C. output.
 - b. Low Battery Voltage: N.O. sontacts close on low battery voltage.
 - c. High Battery Voltage: N.O. contacts close on high battery voltage, contacts not used.
- D. The charger shall be a wall-mounter unit suitable for operation on a 120 volt, single phase power source.
- 2.10 FUEL STORAGE TANK AND PIPING
 - A. Furnish a sub-bace rule storage tank assembly consisting of, but not limited to the following:
 - 1. The main fuel tank shall be a minimum of 500 gallons (24 hours of operation at 100% load). Construction shall be all welded, steel atmospheric tank built in accordance with the latest codes and standards bearing the U.L. 142 label, for use with diesel fuel. Tank shall be complete with all fuel gipe connections as required by engine manufacturer, as well as fill inlet, tank gauging assembly, drain, and vent stack piped above enclosure and furnished with a weatherproof, acreened vent cap per NFPA 30.

- . Containment tank shall be all welded, steel atmospheric tank containing all sides and top of main tank and forming a weatherproof, totally enclosed containment of the main tank, fully enclosed against intrusion of debris, rainwater, etc. Containment tank shall be complete with inlets for leak detector switch and drain. All pipe fittings to the main storage tank, with the exception of the drain, shall penetrate the roof of the containment through weatherproof hubs. Drains in the bottom of the main tank and the containment tank shall be lockable ball valves.
- 3. The tank gauging assembly shall be a factory mutual approved microprocessor based, remote

reading tank gauge, utilizing an intrinsically safe float level sensor suitable for #2 diesel fuel. Readout shall be in 0.8" LED characters that display continuously in gallons and flash at overfill and low level. System shall indicate alarm on two (2) hours of fuel remaining and shall be wired to the generator controller and the remote alarm annunciator. System to be similar to preferred -Rimcor Instruments.

4. A leak detection sensing system shall be installed in the bottom of the containment tank and shall be wired to the generator controller and the remote alarm annunciator auxiliary fault indicator and labeled as "TANK RUPTURE" alarm.

2.11 WEATHER RESISTANT OUTDOOR ENCLOSURE

- A. The diesel engine generator and its required accessories shall be furnished with a factory installed, base mounted, maintenance free, pre-painted forest green outdoor enclosure
- B. Enclosure shall be made of heavy gauge aluminum, sound attenuated to reduce generator set noise to 85 DBA @ 23', and shall totally enclose the generator set, its accessories and sub-base fuel oil storage tank.
- C. Design Criteria:
 - 1. Rigidity wind test equal to 115 MPH.
 - 2. Roof load equal to 50 lbs. per sq. ft.
 - 3. Rain test equal to 4" per hour.
 - 4. Dimensions: Normal 18' long x 7' wide X 9 m
- D. Enclosure shall consist of a roof, underfance, two (2) side walls, and two (2) end walls, of prepainted aluminum construction and floor.
 - 1. Roof: One piece cambered roof theet of .040" thick aluminum with 1/8" extruded aluminum recessed side and end raise
 - 2. Roof bows: Extruct data inum "I" beams spaced with roof reinforced to carry silencer load.
 - 3. Side and End wills: Panels shall be .040" thick aluminum sheet, mill-prepainted, riveted 3" on center.
 - 4. Floor ind Unterframe: Enclosure will have two (2) "I" beam longitudinal skids with fabricated steel class members on 12" centers. The diesel generator set is mounted through vibration isolators to steel tapping plates. A full steel floor shall be provided.

Door Frames: Welded aluminum frame consisting of extruded alloy 1/8"x4-1/2"x1-1/2", riveted to side panels.

Enclosure Accessories:

- 1. Four (4) steel lift rings welded to the underframe.
- Louvers: Motorized intake and gravity discharge louvers shall be all aluminum construction riveted into aluminized steel frame forming a rigid, water-resistant assembly. Louvers shall be properly sized to allow sufficient engine combustion and radiator cooling air flow with a 0.5" H/2/0 maximum restriction. Birdscreen shall be provided on inlet and exhaust openings.
- 3. Air Plenums: Furnish vertical intake and discharge 90 deg air plenums for intake and cooling air.

EMERGENCY GENERATOR SYSTEMS 260612-6

Tetra Tech 200-15704-17001

- 4. Insulation: Furnish 3" acoustic insulation on walls and ceiling, line with perforated metal lining.
- 5. Access Doors: Furnish four (4) access doors, two (2) on each side of the enclosure, each 38" wide x 80" high with padlockable handles, for servicing and operation of generator set and accessories.
- 6. Exhaust Hardware: The enclosure shall be furnished with silencer supports, brackets, rain collars and rain shields.
- F. The Weatherproof enclosure shall be complete with, but not limited to the following equipment and components:
 - 1. Two (2) duplex ground fault interrupting weatherproof receptacles, one (1) mounted areas side of the unit, on the inside of the enclosure.
 - 2. Two (2) 100 watt surface-mounted, weatherproof industrial globe with case guard axtures. Fixtures to be switched from the instrument door.
 - 3. Generator support panel "GP" shall be a weatherproof, 12 pole, 300 / MP, 120/240 volt, single phase, 3 wire + ground loadcenter. Loadcenter shall consist of the following circuits:
 - a. Circuit No. 1: 20A/1P C.B. Battery Charger
 - b. Circuit No. 2: 20A/1P C.B. Enclosure Lighting
 - c. Circuit No. 3: 20A/1P C.B. Enclosure Receptocles
 - d. Circuit No. 4: 20A/1P C.B. Lube Oil Near
 - e. Circuit No. 5: 20A/1P C.B. Battery Heater Pads
 - f. Circuit No. 6: 20A/1P C.B. Fuel Gauge
 - g. Circuit No. 7: 20A/1P C.B. Dyver Operator
 - h. Circuit No. 8: 20A/22 C.B. Engine Coolant Heater
 - i. Circuits 9, 11 and 12 shall be spaces.
- G. Furnish and instal all boxes, conduit and wire required for a complete and operating enclosure.
- 2.12 VIBRATION ISC ATORS
 - A. Provide rubber-it-shear vibration isolators for mounting between engine-generator skid and the enclosure. The isolators shall be 95% efficient and sized in accordance with equipment manufacturer's requirements.

2.13 **CERTIFICATION**

This Contractor shall consult and cooperate with the factory authorized dealer in making arrangements for a load bank of proper size to certify this unit's power rating, stability, voltage and frequency regulation for 25%, 50%, 75% and 100% load over a four (4) hour period, with a one (1) hour period for each load increment.

B. This Contractor shall provide certification, testing and maintenance in accordance with NEC Article 700-4. These records and reports shall be placed in a looseleaf binder and turned over to the Owner for his continued use.

Tetra Tech 200-15704-17001

PART 3 – EXECUTION

3.1 INSTALLATION OF ENGINE GENERATOR SYSTEM

- A. Install emergency engine generator sets as indicated in contract documents, and in accordance with the equipment manufacturer's written instructions, Division 26 Section 260000 under the listing "SPECIAL ENGINEERING SERVICES', and with recognized industry practices, to ensure that engine generator sets fulfill requirements. Comply with NFPA standards pertaining to installation or emergency engine generator systems and accessories.
- B. Coordinate with other work, including fuel supply, piping and accessories as necessar to in installation of emergency generator system work with other work.
- C. Connect fuel piping to emergency generator equipment and comply with manufacturer's instructions where not otherwise indicated.
- D. Perform emergency generator lubrication, equipment startup as specifical in Division 26, Section 260000 under the listing "LUBRICATION' and "EQUIPMENT START OF
- E. Instruct owner's personnel in the operation and maintenance of Emergency Generator System as specified in Division 26 Section 260000 under the listing "Open tion and Maintenance Instructions".
- F. After completion of the installation, testing and instruction, the Contractor shall leave the site with a minimum of 500 gallons of diesel fuel in the main tank.

3.2 GROUNDING

A. Provide equipment grounding connections, sufficiently tight to assure a permanent and effective ground, for system components as indicated in contract documents.

3.3 TESTING

A. Upon completion of installation of engine generator system and after building circuitry has been energized with normal power source test engine generator to demonstrate emergency capability and compliance with requirements. Where possible, field correct malfunctioning units, then retest to demonstrate compliance.

END OF SECTION 260612

SECTION 260731

WIRELESS CLOCK SYSTEM

PART 1 - GENERAL

- 1.1 GENERAL REQUIREMENTS & SCOPE
 - A. Furnish and install a complete new Wireless Clock System using the Primex OneVue[™] plat
 - B. Furnish and install all system devices, accessories, and materials in accordance with the specifications and drawings to provide a complete and operating Wireless Clock System
 - C. All bids shall be based on the equipment as specified herein. The model designation are that of Primex. The specifying authority must approve any alternate Wireless Clock System.
 - D. System shall include the System Devices below:
 - 1. Bridge with Ethernet/Power over Ethernet (PoE) and Blue both a low energy wireless technology
 - 2. Analog Clocks with Bluetooth® low energy wireless technology
- 1.2 RELATED SECTIONS
 - A. Division 26 "Electrical"
 - B. Division 26 Section "Common Work Results for Electrical"
- 1.3 REFERENCES
 - A. System devices specified shall meet or exceed the requirements of the following.
 - 1. Federal Communications Division (FCC); Part 15 Code of Federal Regulations.
 - 2. National Fire Protection Association (NFPA); NFPA 70E-[2012], Standard for Electrical safety in the Workplace
 - 3. Institute of Electrical and Electronics Engineers (IEEE); IEEE 802.3af-[1998], Standard for Information Technology Telecommunications and Information Exchange Between Systems.
 - 4. Bluetooth? wreless technology standard 4.1
- 1.4 DEFINITIONS
 - A. This section provides commonly used terms within this specification.
 - TSA: Technical Support Agreement
 - AWS: Amazon Web Services
 - 3. TLS: Transport Layer Security
 - 4. HTTPS: Hypertext Transfer Protocol Secure
 - 5. Bluetooth[®] low energy a wireless technology protocol which can be used for communication between embedded devices and smart devices.
 - 6. SNTP Simple Network Time Protocol.

7. PoE - Power over Ethernet

1.5 SYSTEM DESCRIPTION

- A. General Specifications
 - 1. System shall provide synchronized time by way of system devices and a cloud-based system software, hosted by the Manufacturer, that allows Owner to manage and monitor system devices.
 - 2. System can be scaled from a single building to a network of buildings, or an enter rise s across many time zones, providing traceable of accuracy, data and performance.
 - 3. System shall consist of system clock devices enabled with Bluetooth lov energy wireless technology.
 - 4. System Bluetooth clocks shall not be required to have a direct comparison to the Owner's existing Power over Ethernet (PoE) or 802.11 wireless network.
 - 5. System devices enabled with 802.11, Power over Ethernet (Poh) technology shall be required to have a direct connection to the Owner's existing network
 - 6. System Bluetooth clocks shall form a system Bluetooth wireless network by way of the proprietary Bluetooth Network Protocol designed and leveloped by the Manufacturer, that allows system Bluetooth clocks to form a Bluetooth wireless technology network to allow a communication path to send clock data to the system Bridge devices.
 - 7. System Bridge(s) shall receive data from even Bluetooth clocks and send the data to the system software. Bridge shall download settings from the system software and send settings to the system Bluetooth clocks.
 - 8. System shall not require the instanction of any on-site system hardware or software, with the exception of the specified system devices.
- B. System Software specification.
 - 1. System software shall be accessed from a web browser and does not require the installation of software or other third-party plug-ins or applications.
 - 2. System users chall be able to access the system software from a web browser on a mobile or tablet revice or computer with an internet connection.
 - 3. System software shall provide an online device configuration procedure that allows Owner supplied settings to be installed on system 802.11 Wi-Fi or Non-DHCP network devices prior to showent from the Manufacturer.

4. System software shall log NTP accuracy.

- System software shall monitor and display the operating status of system devices.
- 6. System software shall provide an automated reporting method to notify system users of the operating status of system devices that may warrant corrective action.
- 7. System software shall allow authorized system users to create reports, view archived reports, and set the system to automate report notifications sent to system users.
- 8. System software shall maintain and store remote, redundant data backups.

- C. System Bluetooth wireless network specifications
 - 1. System Bluetooth wireless network shall be able to adjust and synchronize system Bluetooth clocks to specified Time Zone Offset and DST rules; time zone settings are managed within the system software and sent to the system Bluetooth clocks over the system Bluetooth wireless network.
 - 2. System Bluetooth wireless network protocol shall form a tree-type topology; where clock data transfer is up or down in a tree structure topology; a path with highest Bluetooth Wireless Signal level quality is chosen at any time.
 - 3. System Bluetooth wireless network shall have the capability to distinguish the system Bluetooth devices with Bluetooth wireless technology by the system devices' unque Device ID and/or Network ID; allowing only system Bluetooth devices to authenticate to the system Bluetooth wireless network.
 - 4. At a system set daily time interval, the system Bluetooth clocks shall advertise their Bluetooth signal to form a system Bluetooth wireless network
 - 5. System Bluetooth wireless network shall be self-forming, self-heating, and self-organizing.
 - a. Self-healing: in the event of a system Blueteeth clock hardware failure or loss of Bluetooth wireless signal, a connection previously handled by it is rerouted to another clock within the system Bluetooth wireless network.
 - b. Self-forming: a system Bluetooth wireless betwork is automatically formed once daily at a system set time to allow system Bluetooth clocks to advertise their status to form the system Bluetooth wireless network
 - c. Self-organizing: the System Bluet oth wireless network protocol automatically connects system Bluetooth clocks, that are within Bluetooth wireless range, to form a data transmission path within the system Bluetooth wireless network.
- D. Bridge specifications
 - 1. Bridge shall be equipped with Bluetooth enabled gateway; which allows the device to send and receive communication to and from system Bluetooth clocks.
 - 2. Bridge device shall connect to the Owner's existing wired Ethernet, PoE, or 802.11 wireless network to send its device data and system Bluetooth clock data to the system software and download its device settings and clock settings from the system software.



- Bigge shall send settings to system Bluetooth clocks within the system Bluetooth wireless network. Each system Bluetooth clock is identified by a unique Device ID allowing each clock to receive its unique Device ID settings. Device ID settings shall be managed in the system software. Bluetooth Analog Clock Device ID settings include its Time Zone Offset setting and DST rules. Bluetooth Digital Clock/Timer Device ID settings include its Time Zone Offset setting and DST rules and display settings.
- 4. Bridge shall listen and receive system Bluetooth clock data advertised in the system Bluetooth wireless network. During its 8-hour deployment mode, a Bridge shall send a new Bluetooth clock Device IDs to the system software within 30 minutes of receiving it. If not in 8-hour deployment mode, new Bluetooth clock data is sent the system software within 24 hours.

- 5. Bridge shall obtain Coordinated Universal Time (UTC) derived from a Network Time Protocol (NTP) Server (either internal or external); up to three designated NTP Servers may be specified to ensure continuity of time synchronization. The Bridge shall send obtained UTC time received from its NTP Server to the system Bluetooth clocks and the clocks shall synchronize the received UTC time to their Time Zone Offset setting and DST rules.
- 6. Bridge devices shall be primarily powered by Power over Ethernet. Battery-power shall be used only as a backup power source for a relatively short period of time.
- 7. Bridge devices that connect to a Wi-Fi or Non-DHCP network can be preconfigured by the Manufacturer with Owner provided settings before shipment to Owner facility of configured locally at the device by the Owner on-site.
- 8. Bridge devices shall have a LCD screen that displays connection status, and indicate when in an error or alarm state.
- 9. Bridge devices shall have LED indicators, located on the front of the divice, that provide a visual indicator of its current status and operating state.
- 10. Bridge device shall have the ability to store configuration date for up to 1400 Bluetooth clocks in its local device memory; to avoid loss of data if a network connectivity issue prevents data to be sent to the system software.
- 11. Check-in interval setting shall be set by the system. Check-in Interval is the frequency the Bridge device is scheduled establish a network connection to Owners network to send data to and download data from the system software
- 12. Unresponsive timeout setting shall be see by the system. Unresponsive timeout is defined as the amount of time a device can go without a connection to the system software; when this time is exceeded, the system set the device to a warning state.
- E. System Clocks enabled with Pluetoch low energy wireless technology specifications (System Bluetooth clocks)
 - 1. Clocks shall not equivalent and a configuration by an end user to establish a connection to a system Blaetooth wireless network, nor does an end user need to know about the intricacies of Bluetooth networking. The entire Bluetooth network infrastructure is transparent to the end user.
 - 2. Clock shall be equipped with a Bluetooth low energy wireless technology radio component; the above system Bluetooth clocks to establish wireless Bluetooth connections to form a system Bluetooth wireless network. Each clock shall be a node within the system Bluetooth wireless network.



4. Clocks shall wake-up once a day, at a system defined time, to form and build a system Bluetooth wireless network; allowing each system clock to send and/or forward its status data to the Bridge and receive setting updates sent by the Bridge. Each clock connects to another clock based on the strongest Bluetooth signal. The Bridge stores and sends clock status data to the system software once a day. When a bridge is in a 8-hour deployment mode, clock data is sent every 30 minutes to the system software.

- 5. Upon first-power up at its installation location, a new system Bluetooth clock shall go through a self-discovery initiation process; the clock shall continuously search for a system clock to receive its time and daily connection schedule.
- 6. Clocks shall not be required to be in plain line of sight or directly connected to a system Bridge device and shall act as independent nodes within the system Bluetooth wireless network.
- 7. If a clock's Bluetooth wireless signal connection is interrupted or down, the other clocks within the Bluetooth wireless range shall be able to transmit data to the system Bridge device through other Bluetooth clocks within the system Bluetooth wireless network. Clock data moves through the system Bluetooth wireless network communication path until the data reaches a system Bridge device.
- 8. Clocks shall be available, at predefined time intervals set by the system to connect to new Bluetooth clocks that are attempting to connect to the system Bluetooth wireless network; allowing new clocks to receive and synchronize their time from an existing system Bluetooth clock.
- 9. Clock device firmware shall perform diagnostics on battery life, time accuracy, and connection strength. The Bridge devices shall transmit the clock diagnostic data to the system software.
- 10. Clocks shall operate with a free running accuracy of 45 seconds per day, and will continue to operate in the absence of receiving the broadcasted UTC time from the Bridge.
- 11. Clocks shall be fully portable, capable or occurrence at any time.
- 12. Analog clocks shall report gross mechanical failures by way of automatically performing a daily midnight hand verification chech; which if this check shall fail for three consecutive days, the clock shall report a non-position failure status, resulting in a clock warning state within the system software.
- F. Encryption and Authentication specifications
 - 1. User software access sersions between the web browser and the system software shall be encrypted by the HTTPS protocol.
 - 2. The network communication of system devices enabled with 802.11, Ethernet, or PoE Technology shall be secure and encrypted using the Transport Layer Security (TLS) encryption protocol and Secure Hypertext Transfer Protocol (HTTPS) authentication.
- G. System Administration specifications

Software interface shall allow the Owner's system admin user(s) to manage the system components, including: system device settings, reports, system-wide user password complexity settings and user session timeout setting to align with Owner information security policies and procedures, manage system users and grant user access to system data and features, activate and deactivate system users, and view user log in history.

- 2. System software shall allow each system user to manage their own system profile, including their log in email address, password, and contact settings.
- 3. System software shall allow system device settings to be user-defined to meet Owner requirements.

- 4. System software shall allow devices that use a 802.11 or Non-DHCP network connection to be assigned to a primary and alternate network for failover purposes. Network settings are managed within the system software, allowing remote management to migrate devices from one network to another.
- 5. System software shall allow user-defined reporting; system shall store and present system historical data in the form of system reports. User-defined data shall include the system devices included in a report, the frequency a report is system generated, and a specific range of data included in a report. System reports shall be displayed in the system software electronically within the interface allow a system user to print and download reports. System shall allow report data to be restricted based on the role(s) assigned to a system userport lie.

1.6 REGULATORY REQUIREMENTS

- A. Equipment and components furnished shall be of the Manufacturer latest rood
- B. System devices shall be installed in compliance with local and tate authorities having jurisdiction.
- C. Electrical Components, Devices, and Accessories: Listed and labeled per NFPA 70 by qualified testing agency.
- D. Regulatory Requirements: System design and installation shall comply with the following: National Electric Code (NEC)
 - 1. Underwriters Laboratory (UL) standards
 - 2. Local codes and regulations

1.7 SUBMITTALS

- A. Product Data: Submit complete analog data for each system device and components, describing physical characteristics and me hod of installation.
- B. Shop Drawings: Showing the following. 1. Diagram of proposed system showing communication pathway and schedule of introvidual system device installation locations. 2. Indicate integration with the Owner's network include a line diagram of network relationships. 3. Show system device power requirements.
- C. Samples: Sponit the specified system device model(s) for approval. Approved sample(s) shall be tagged and shall be installed at location directed.
- D. Man facturer Instructions: Submit complete installation, set-up and maintenance instructions electronically.
- E. Aformation submittal: Manufacturer Sample Warranty

Information submittal: Manufacturer Technical Support Agreement (TSA)

SUBSTITUTIONS

- A. Proposed substitutions, to be considered, shall be manufactured of equivalent materials that meet or exceed specified requirements of this Section.
- B. Proposed substitutions shall be identified not less than 10 days prior to bid date.

1.9 QUALITY ASSURANCE

- A. Manufacturer Qualifications: Manufacturer of wireless and Ethernet connected system with a minimum of ten years record of satisfactory manufacturing and support of systems comparable to basis of specified system design.
- 1.10 DELIVERY STORAGE AND HANDLING
 - A. Deliver all components to the site in the Manufacturer original packaging.
 - B. Packaging shall contain Manufacturer name and address, product identification number, and related information.
 - C. Store equipment in finished building and in unopened packaging until ready for installation.

1.11 PROJECT SITE CONDITIONS

- A. System design is integrated with Owner's existing network, including 800,11 wireless, wired Ethernet, or PoE network; limited to devices equipped with vieness or Ethernet (PoE) technology.
- B. Conductors and Cables: Comply with requirements of Davisio. 2^o Sections "Common Work Results for Communications" and "Communications Horizontal Cabling.
- C. Signal and Control Circuits: Manufacturer recommended stranded, single conductors, or twistedpair cables.
- D. Data Circuits: Category 5 minimum, twisted pair vable

1.12 WARRANTY

- A. Manufacturer shall provide a one-year limited warranty for Education Series Analog Clocks equipped with Bluetooth® low nervy vieless technology.
- B. Manufacturer shall provide a wo year limited warranty for Non-Education Series Clocks equipped with Bluetooth a pow energy wireless technology.
- C. Manufacturer shall provide a two-year limited warranty for Bridge devices equipped with Bluetooth® low enegy wireless technology.
- D. Manufacturer shar provide a two-year limited warranty for Clock devices equipped with Power over Ethernet (PdE) technology.
- E. Man facturer shall offer an extended warranty on system devices.

1.13 **JOFTWARE MAINTENANCE**

Manufacturer shall offer an annual Technical Support Agreement (TSA); agreement shall be inclusive to system software access, phone/email technical support, software maintenance and revisions, and firmware revisions.

All system updates, enhancements and maintenance are performed per agreed upon TSA.

PART 2 – PRODUCTS

- 2.1 SECTION INCLUDES
 - A. The system and equipment is specified as described in this section.

- B. All bids shall be based on the equipment as specified herein. The model designations are that of Primex. The specifying authority must approve an alternate system.
- 2.2 MANUFACTURER
 - A. System shall be manufactured by:

US: Primex, 965 Wells Street, Lake Geneva, WI 53147 | Phone: (800) 537-0464 | Fax: (262) 248-0061 | Email: info@primexwireless.com | www.primexwireless.com

- 2.3 SYSTEM SOFTWARE
 - A. Basis of Design Software Product: Primex OneVue™ Wireless Clock System.
 - B. System Software Platform: Cloud-based software platform that resides on Amaron Web Services (AWS) and is accessed via the internet.
 - C. System stores and monitors system devices operating conditions.
 - D. All system device and system settings are managed within the system software.
- 2.4 SYSTEM DEVICES & ACCESSORIES
 - A. Analog Clocks
 - 1. Clocks shall meet the following specifications:
 - a. Clocks (single sided) shall be wall mounted
 - b. Additional colors, finishes, and dial a ces are available from Manufacturer.
 - c. Clock faces can be customized by Manufacturer to display organization name or logo as specified.
 - d. Clock frames and lenses are of durable thermoplastic.
 - e. Clocks shall have a tanger proof/theft resistant clock-lock mounting slots.
 - f. A dual-mount kin is available from the Manufacturer that combines two single clocks to create a dua sided clock.
 - g. Clocke shall be capable of automatically adjusting for Daylight Saving Time.
 - h. Electric (AC) models will include a cord with plug. The power plug may be removed and cord cut to length for hardwired (pigtail) installation. Pigtail installation requires a 120V~ power line in a junction box installed by a licensed electrician.
- NO 4
- If power is interrupted, the clock will stop until power resumes. Upon resumption of power, the clock will self-correct to the current time.
- . Battery-operated analog clocks shall have a 5-year nominal battery life.
- k. Installer will furnish clock batteries in accordance with Manufacturer instructions.
- 1. Power over Ethernet (PoE) models shall have an USB port which can be used to supply power via external supply and allow clock to operate using conventional Ethernet connection without PoE.
- m. SUPPLY MODELS Analog Clocks

- (1) Education Series
 - (a) Technology: Bluetooth® low energy wireless technology
 - (b) Size: [12.5"] [16"]
 - (c) Color: Black
 - (d) Power: Battery

B. Bridge

- 1. Bridge shall meet the following specifications.
 - a. Enclosure: ABS plastic | Dimension: 4.7" H x 3.7" W x 1.3" D (11.93cm) 9.39cm x 3.30cm) | Weight: 0.3 lb (136 gram) with 2 AA batteries
 - b. Display: Liquid crystal display (LCD), dimension: 0.75 in. Un N8 in. W (1.90 cm x 3.50 cm) | LED Status Indicator: green, yellow, red
 - c. Mounting: Keyhole slot with lock down screw holes in bask panel for wall mount; or surface mount with supplied dual-lock adhesive mounting strips.
 - d. Configuration: configured from system software or locally at device with supplied device configuration software
 - e. Backup Battery-power: 3.0v Primex Lithum/Iron Disulfide Battery Pack or two standalone 1.5v Lithium AA batteries. Use of Jkaline batteries is not recommended.
 - f. Local memory storage capacity: configuration data for up to 1400 Bluetooth clocks.
 - g. Environment: Operating Temperature: 32 °F to 122 °F (0 °C to 50 °C), indoor use only | Storage Temperature: -4 °E to 140 °F (-20 °C to 60 °C)
 - h. Certifications: FCC, CE, and C compliant
 - i. SUPPLY MODELS Bidge
 - (1) Bridge
 - (a) Termology: [802.11 Wireless] [Ethernet (PoE) Technology]

(b) Iower: [AC] [Power over Ethernet (PoE)]

- . ccessomes
 - a. System shall include the accessories below:
 - (1) Analog Clock Wire Clock Guard
 - (a) Description
 - i. 16" Clocks (40.64cm): 18" square x 3"D (45.72cm square x 7.6cm D)

PART 3 – EXECUTION

3.1 EXAMINATION

A. Examine conditions with the Installer present for compliance with requirements and other conditions affecting the performance of the system and the system devices.

Tetra Tech 200-15704-17001

- B. Do not proceed until unsatisfactory conditions have been corrected.
- C. Verify that construction is complete in spaces to receive equipment and that rooms are clean and dry.

3.2 INSTALLATION

- A. General: Install system devices in accordance with applicable codes.
- B. Install system devices in accordance with Manufacturer written instructions.
- C. Provide all system equipment necessary for a complete and operable system.
- D. Comply with requirements of Division 27 Sections "Common Work Secures for Communications" and "Communications Horizontal Cabling."
- E. Cables: Install cables in raceways and cable trays except within consoles, cabinets, and desks [and except in accessible ceiling spaces and framed partitions where exposed wiring is allowed by Owner]. Install plenum cable where required. Conceal cable installation where possible.

3.3 FIELD INSPECTION

- A. Inspection: Make observations to verify that system devices and components are properly labeled.
- B. Prior to final acceptance, inspect each system device and component, adjust as required, and replace parts which are found defective.
- C. At completion of system device installation and prior to final acceptance, turn on the equipment; ensure that all equipment is operating propeny, and that the system software and all system devices and components are functioning.

3.4 SERVICES

- A. Manufacturer system software user guides and system device installation guides shall be provided electronically.
- B. Commissioning General: A vide system commissioning in accordance with Manufacturer written recommendations Perform operational testing to verify compliance with requirements. Adjust as required.
- C. Services shell include specified level of commissioning services.
 - 1. On site commissioning service: system deployment training, including system set up, validation of device preconfiguration, system functionality, verification of device network connections, and device installation training.

3.5 **CLEANING**

Prior to final acceptance, clean exposed surfaces of devices, using cleaning methods recommended by Manufacturer.

DEMONSTRATION

- A. Initial Demonstration and Training: provide demonstration and training for Owner facility staff that is responsible to perform system software administration.
- B. Demonstrate maintenance procedures for system devices.

C. Demonstrate the system software components and features, including monitoring and management of system devices.

3.7 **PROTECTION**

A. Protect finished installation until final acceptance of the project.

3.8 TESTING

A. All system devices must be tested at their operational installation location under non operational conditions.

END OF SECTION 260731

Republication

<



SECTION 260851

HEATING TERMINALS

PART 1 - GENERAL

1.1 DESCRIPTION OF WORK

- A. Extent of heating terminal work is indicated by drawings, schedules and specifications.
- B. Types of electrical heating terminals in this section include the following:
 - 1. Self-regulating cable
 - 2. Constant wattage cable
 - 3. Monitoring controller and sensor
- C. The heating terminal work shall be designed and installed to provide f eeze protection for all exterior and interior piping including, but not limited to, supply, return, fit, drain and equalizing lines on chillers and cooling towers, fire protection systems where indicated or subjected to freezing and rain water conductors where indicated.

1.2 SUBMITTALS

- A. Product Data: Submit manufacturer's data on heating terminals, including dimensions, capacities, ratings, performance characteristics, gauges and finishes of materials, installation and wiring instructions.
- B. Shop Drawings: Submit assembly type shop brawings showing unit dimensions, construction details, and field electrical connection details.

PART 2 - PRODUCTS

2.1 ELECTRIC HEAT TRACING CABLE

- A. The heat tracing cable scall be either self-regulating or constant wattage type. Cable installation shall be designed to maintain 40 deg. F in a -20 deg. F ambient temperature with a 25 MPH wind factor and a 20% safety factor. Jystem shall be designed to operate at the voltage indicated on the plans.
- B. Heat loss design shall be calculated on a nominal 1-1/2" fiberglass insulation thickness. Pipe sizes and lengths shall be taken from the Mechanical, Plumbing and Fire Protection drawings.
- C. Heat race system shall be complete with, but shall not be limited to, controller, sensors, trace cable, wiring, connectors, junction boxes, mounting brackets, supports and fastenings as required in the quantities and lengths to suit the required installations.

Monitoring Controller: Furnish a microprocessor base controller with the following characteristics:

- 1. Supply Voltage: 120/277 Vac, +/- 10%, 50/60 Hz, internal power less than five watts.
- 2. Environmental Temp.: -40°F to 125°F maximum operating temperature range; 0% 95% R.H. @ 40°C non-condensing
- 3. Microprocessor: Non-volatile memory; no data loss on power outage.
- 4. Load Current: 30 AMPs maximum

Tetra Tech 200-15704-17001

| 5. | Control: | Double pole solid state switching | | | |
|----|---|--|--|---------------------------------|--|
| 6. | Temp. Sensor Input: | 100 ohm platinum RTD, 3 wire, 20 ohm maximum, lead wire compensation, $a = .00385$ ohms/ohm/deg. C. | | | |
| 7. | Outputs: | Unit alarm output configurable as "open on alarm" or "close on alarm". AC alarm triac: isolated solid-state triac, SPST, 0.5 AMP max. @ 12 277 volts AC. | | | |
| 8. | Indicators: | LED: Program Mode, Actual Temperature, Control Temperature, Heater Current, Power ON, Heater ON, Alarm Conditions/Deogramming Parameters. Digital Display: Actual Temperature, Control Temperature, Heater Current, Programming Parameter Values, Alarm Values. | | | |
| 9. | Setpoints: | Temperature: | | | |
| | | Units: | Deg. F or Deg. | | |
| | | Control Range: | -40 degar to 299 deg. F or OFF | | |
| | | Low – temp. alarm: | -40 deg. F to 999 | -40 deg. F to 999 deg. F or OFF | |
| | | High – temp. alarm: | | deg. F or OFF | |
| | | Ground Fault: | 3 | | |
| | | Alarm Range: 20 to 100 mA | | | |
| | | Trip Range | 20 to 100 mA or | OFF | |
| | | Low allarm Range: | 0.0 to 30.0 AMPs or OFF 1.0 to 30.0 AMPs or OFF | | |
| | () | Nigh Alarm Range: | | | |
| | Δ^{\vee} | Power Limit: 1.0 to 30.0 AN | | s or OFF | |
| | $\sqrt{O^{X^{-}}}$ | Auto Cycle Time: | 0.5 to 24.0 hours or OFF | | |
| 10 | . Averm Conditions | Low Temperature | Low Current | Memory Failure | |
| | | High Temperature | High Current | Ground Fault Trip | |
| | • | Ground Fault | Sensor failure | SCR Failure | |
| | . Heat Trace System Diagnostic Test. | System cycles heating cable regularly (settable period from 0.5 to 24.0 hours) and confirms proper system operation. Minimum process temperature, maximum process temperature, maximum ground-fault current, maximum heating cable current. | | | |
| 12 | . Stored parameters (measured): | | | | |

- 13. Enclosure: NEMA 4X.
- 14. Additional Features: Soft start, On/Off or proportional control, power limiting, password protection.
- 15. Manufacturer: Controller shall be similar to Raychem Corporation's Digitrace 920/920 or approved equal.
- E. Heat Trace Sensor: Furnish a 3-wire, platinum resistance temperature detector (R1C) with the following parameters:
 - 1) Type: 100 ohm platinum.
 - 2) Accuracy: $100 \text{ ohm} \pm 0.25 \text{ ohm} @ 0 \text{ d}$
 - 3) Alpha: 0.00385 ohm/ohm/deg. C
 - 4) Temperature Measurement Range: RTD 300: 300 deg. 1 (150 deg. C.)
 - 5) Exposure Temperature Limits: RTD 300: 300 leg F. (150 deg. C.)
 - 6) Manufacturer: Sensor shall be similar to Raychan Corporation's RTD 300 or approved equal.
- F. Communications: RS-485 interface to be provided for MODBYS Communications to BAS System.
- G. Power Wiring: Provide required circuit breaters and wiring to local panel for required quantity of heat trace circuits.

2.2 MANUFACTURERS

- A. Subject to compliance with requirements, furnish a heat trace cable system of one of the following:
 - 1. Raychem Corporation
 - 2. Delta Therm Corporatio
 - 3. Thermon Manufacturing Company
 - 4. Approved substitute
- PART 3 EXECUNO

Α

3.1 INSTALATION OF HEATING TERMINALS

Install heat trace cable system and controls as required, in accordance with equipment manufacturer's written instructions, and with recognized industry practices, to ensure that the heat trace system equipment fulfills project requirements. Comply with applicable installation requirements of the NEC and NECA's "Standard of Installation".

- B. Install electric heat trace cable system after the piping tests are complete and before the piping insulation is applied. The Electrical Contractor shall consult, cooperate and coordinate with both the Mechanical Contractor and the Insulation Contractor in completing the required work.
- C. Install heat trace labels every 15'.

Tetra Tech 200-15704-17001

D. Megger test before and after insulation is complete. Tests to be documented. Tests to be witnessed by the Construction Manager and Insulation Contractor.

3.2 GROUNDING

A. Provide equipment grounding connections, sufficiently tight to assure a permanent and effective ground, for heating terminals and controls as indicated in contract documents.

3.3 TESTING

A. Upon completion of installation of heating terminals and controls and after building circulary has been energized; test heating terminals and all control functions to demonstrate capability and compliance with requirements. Where possible, field correct malfunctioning units, then retest to demonstrate compliance.

END OF SECTION 260851

Republic

SECTION 31 10 00 - SITE CLEARING

PART 1 GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supprementation Conditions and Division 1 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. This Section includes the following:
 - 1. Removing existing trees, shrubs, groundcover, plants, and grass.
 - 2. Clearing and grubbing.
 - 3. Stripping and stockpiling topsoil.
 - 4. Removing above- and below-grade site in provenents.
 - 5. Disconnecting, capping or sealing, and acondoning site utilities in place removing site utilities.
 - 6. Temporary erosion and sedimentation control measures.
- B. Related Sections include the following:
 - 1. Division 2 Section "Aerthwork" for soil materials, excavating, backfilling, and site grading.
 - 2. Division 2 Section "Lawns and Grasses" for finish grading including preparing and placing planting oil places and testing of topsoil material.
- 1.3 DEFINITIONS
 - A. Topsoil: Natural or cultivated surface-soil layer containing organic matter and sand, silt, and clay particles; friable, pervious, and black or a darker shade of brown, gray, or red than underlying subsoil; reasonably free of subsoil, clay lumps, gravel, and other objects more than 2 inches in diameter; and free of subsoil and weeds, roots, toxic materials, or other nonsoil naterials.

Tree Protection Zone: Area surrounding individual trees or groups of trees to be protected during construction, and defined by the drip line of individual trees or the perimeter drip line of groups of trees, unless otherwise indicated.

1.4 MATERIAL OWNERSHIP

A. Except for stripped topsoil or other materials indicated to remain Owner's property, cleared materials shall become Contractor's property and shall be removed from Project site.

1.5 SUBMITTALS (NOT USED)

1.6 PROJECT CONDITIONS

- A. Traffic: Minimize interference with adjoining roads, streets, walks, and other adjacent occupied or used facilities during site-clearing operations.
 - 1. Do not close or obstruct streets, walks, or other adjacent occupied or used facilities without written permission from the University of Delaware
 - 2. Provide alternate routes around closed or obstructed traffic w
- B. Utility Locator Service: Notify Miss Utility a minimum of 3 Vays prior to performing any land disturbing activities. Contact Miss Utility of Delmarva -800-282-8555
- C. Do not commence site clearing operations until emporary erosion and sedimentation control measures are in place.

PART 2 - PRODUCTS (Not Applicable)

2.1 SOIL MATERIALS

- A. Satisfactory Soil Materials, Requirements for satisfactory soil materials are specified in Division 2 Section "Earth york"
 - 1. Obtain approved barrow soil materials off-site when satisfactory soil materials are not available or site.

PART 3 - EXECUTION

3.1 **PREPARATION**

Protect and maintain benchmarks and survey control points from disturbance during construction.

Locate and clearly flag trees and vegetation to remain or to be relocated.

- C. Protect existing site improvements to remain from damage during construction.
 - 1. Restore damaged improvements to their original condition, as acceptable to Owner.

3.2 TREE PROTECTION

- A. Erect and maintain temporary fencing around tree protection zones before starting site clearing. Remove fence when construction is complete.
 - 1. Do not store construction materials, debris, or excavated material within fenced arg
 - 2. Do not permit vehicles, equipment, or foot traffic within fenced area.
 - 3. Maintain fenced area free of weeds and trash.
- B. Do not excavate within tree protection zones, unless otherwise indicated.
- C. Where excavation for new construction is required within tree protection zoles, hand clear and excavate to minimize damage to root systems. Use narrow-tine spedia forks, comb soil to expose roots, and cleanly cut roots as close to excavation as possible.
 - 1. Cover exposed roots with burlap and water regularly.
 - 2. Temporarily support and protect roots from damage and they are permanently redirected and covered with soil.
 - 3. Coat cut faces of roots more than 1-1/2 inches in dameter with an emulsified asphalt or other approved coating formulated for use on damaged plant tissues.
 - 4. Backfill with soil as soon as possible.
- D. Repair or replace trees and vegetation in neutro to remain that are damaged by construction operations, in a manner approved by Architect
 - 1. Replace trees that cannot be reported and restored to full-growth status, as determined by Architect.

3.3 UTILITIES

1.

- A. Locate, identity, deconnect, and seal or cap off utilities indicated to be removed.
 - 1. Arrange with utility companies to shut off indicated utilities.
 - 2. Owner will arrange to shut off indicated utilities when requested by Contractor.
- B. Existing Utilities: Do not interrupt utilities serving facilities occupied by Owner or others unless permitted under the following conditions and then only after arranging to provide temporary utility services according to requirements indicated:
 - Notify the Owner not less than two days in advance of proposed utility interruptions.
 - 2. Do not proceed with utility interruptions without the University's written permission.
- C. Excavate for and remove underground utilities indicated to be removed.
- D. Removal of underground utilities is included in Division 2 Sections covering site utilities.

3.4 CLEARING AND GRUBBING

- A. Remove obstructions, trees, shrubs, grass, and other vegetation to permit installation of new construction.
 - 1. Do not remove trees, shrubs, and other vegetation indicated to remain or to be relocated
 - 2. Cut minor roots and branches of trees indicated to remain in a clean and careful manner where such roots and branches obstruct installation of new construction.
 - 3. Grind stumps and remove roots, obstructions, and debris extending to a dept inches below exposed subgrade.
 - 4. Use only hand methods for grubbing within tree protection zone.
 - 5. Chip removed tree branches and dispose of off-site.
- B. Fill depressions caused by clearing and grubbing operations with scalafactory soil material unless further excavation or earthwork is indicated.
 - 1. Place fill material in horizontal layers not exceeding a lowe depth of 8 inches, and compact each layer to a density equal to adjacent original ground.

3.5 TOPSOIL STRIPPING

- A. Remove grass before stripping topsoil.
- B. Strip topsoil to whatever depths are encountered in a manner to prevent intermingling with underlying subsoil or other waste materials.
 - 1. Remove subsoil and non-solumate als from topsoil, including trash, debris, weeds, roots, and other waste materials.
- C. Stockpile topsoil materials ways from edge of excavations without intermixing with subsoil. Grade and shape stockpile to grain surface water. Cover to prevent windblown dust.
 - 1. Do not stocipile topsoil within tree protection zones.
 - 2. Dispose of mess topsoil as specified for waste material disposal.
 - 3. Sto hpile urplus topsoil to allow for re-spreading deeper topsoil.
- 3.6 SITE MPROVEMENTS

Remove existing above- and below-grade improvements as indicated and as necessary to incilitate new construction.

Remove slabs, paving, curbs, gutters, and aggregate base as indicated.

- 1. Unless existing full-depth joints coincide with line of demolition, neatly saw-cut length of existing pavement to remain before removing existing pavement. Saw-cut faces vertically.
- 2. Paint cut ends of steel reinforcement in concrete to remain to prevent corrosion.

- 3.7 DISPOSAL
 - A. Disposal: Remove surplus soil material, unsuitable topsoil, obstructions, demolished materials, and waste materials including trash and debris, and legally dispose of them off Owner's property.

Republic

1. Separate recyclable materials produced during site clearing from other non-recyclable materials. Store or stockpile without intermixing with other materials and transport them to recycling facilities.

END OF SECTION 31 10 00

SECTION 31 20 00 - EARTHMOVING

PART 1 **GENERAL**

1.1 **RELATED DOCUMENTS**

Drawings and general provisions of the Contract, including General and Suppreme A. Conditions and Division 1 Specification Sections, apply to this Section.

1.2 **SUMMARY**

- This Section includes the following: A.
 - 1. Preparing subgrades for pavements, walkways, and building and structural pads.
 - 2. Excavating and backfilling for retaining walls and structur
 - Subbase course for concrete walks. 3.
 - Subbase course for asphalt paving. 4.
 - Subsurface drainage backfill for walls and trenche 5.
 - Excavating and backfilling for utility trench 6.
 - Excavating and backfilling trenches for buried mechanical and electrical utilities and pits 7. for buried utility structures.

Related Sections include the following: B.

- 1. Division 1 Section Allowarces for quantity allowance provisions related to unit-price rock excavation and authorized additional excavation.
- 2. Division 1 Section Unit Prices for unit-price rock excavation and authorized additional
- excavation provision. Division 2 Section Site Clearing for, site stripping, grubbing, stripping and stockpiling 3. topsoil, and remeal of above- and below-grade improvements and utilities.
- Division Section Erosion and Sediment Control, for temporary erosion and sediment 4. control
- 5. Section Lawns and Grasses for finish grading, including preparing and Di placing typsoil and planting soil for lawns.
- Division 3 Section Cast-in-Place Concrete for granular course if placed over vapor 6. etarder and beneath the slab-on-grade.
 - vivisions 2, 15, and 16 Sections for installing underground mechanical and electrical utilities and buried mechanical and electrical structures.

DEFINITIONS

Backfill: Soil materials used to fill an excavation, including the backfill of retaining walls, footings and foundations.

- 1. Initial Backfill: Backfill placed beside and over pipe in a trench, including haunches to support sides of pipe.
- 2. Final Backfill: Backfill placed over initial backfill to fill a trench.
- B. Base Course: Course placed between the subbase course and hot-mix asphalt paving.

- C. Bedding Course: Course placed over the excavated subgrade in a trench before laying pipe.
- D. Borrow Soil: Satisfactory soil imported from off-site for use as fill or backfill. See section 2.1 for the requirements of borrow soil.
- E. Drainage Course: Course supporting the slab-on-grade that also minimizes upward capillary flow of pore water.
- F. Excavation: Removal of material encountered above subgrade elevations and dimensions indicated.
 - 1. Authorized Additional Excavation: Excavation below subgrade elevations or beyond indicated lines and dimensions as directed by Architect.
 - 2. Unauthorized Excavation: Excavation below subgrade elevations or beyond indicated lines and dimensions without direction by Architect. Unauthorized excavation, as well as remedial work directed by Architect, shall be without additional compensation.
- G. Fill: Soil materials used to raise existing grades.
- H. Rock: Rock material in beds, ledges, unstratified masses conglomerate deposits, and boulders of rock material 3/4 cu. yd. or more in volume that exceed a standard penetration resistance of 100 blows/2 inches when tested by an independent grotechnical testing agency, according to ASTM D 1586.
- I. Structures: Buildings, footings, foundations, retaining walls, slabs, tanks, curbs, mechanical and electrical appurtenances, or other man-hade stationary features constructed above or below the ground surface.
- J. Subbase Course: Course priced between the subgrade and base course for hot-mix asphalt pavement, or course placed between the subgrade and a cement concrete pavement or a cement concrete or hot-mix appha wark.
- K. Subgrade: Surface or elevation remaining after completing excavation, or top surface of a fill or backfill immediate phelow subbase, drainage fill, or topsoil materials.
- L. Utilities: On-life underground pipes, conduits, ducts, and cables, as well as underground services within ouildings.

1.4 SUBMITTALS

omply with Division 1 Section Submittal Procedures.

For Approval

- 1. Product Data: For the following:
 - a. Geotextile support fabric.
 - b. Geotextile filter fabric.
- C. For Information:
 - 1. Material Test Reports: From a qualified testing agency indicating and interpreting test results for compliance of the following with requirements indicated:

a. Classification according to ASTM D 2487 of each on-site and borrow soil material proposed for fill and backfill.

b. Laboratory compaction curve according to ASTM D 698 for each on-site and borrow soil material proposed for fill and backfill.

1.5 PROJECT/ SITE CONDITIONS

- A. Test borings and other exploratory operations may be made by the Contractor at no cost Owner.
- B. Existing Utilities: Locate existing underground utilities by hand excavation in areas of work. If utilities are to remain in place, provide adequate means of support and protection during earthwork operations.
 - 1. Should uncharted, or incorrectly charted, piping or other utilities be uncountered during excavation, consult utility owner immediately for directions. Cooperate with Owner and utility companies in keeping respective services and facilities in operation. Repair damaged utilities to satisfaction of utility owner. Do not breat utility connections without providing temporary services, as acceptable to Engineer.
 - 2. Do not interrupt existing utilities serving facilities occupied and used by Owner or others, during occupied hours, except when permitted in writing by Engineer and then only after acceptable temporary utility services have been provided.
 - a. Provide, to Architect, a minimum of 8 hour notice to proceed before interrupting any utility.
 - 3. Demolish and completely remove from site any existing underground utilities indicated to be removed. Coordinate with utility companies for shut-off of services if lines are active.
- C. Use of Explosives: The use of explosives is not permitted.
- D. Protection of Persons and Property: Barricade open excavations occurring as part of this work and post with warning lights.
 - 1. Operate working lights as recommended by authorities having jurisdiction.
 - 2. Proceeds the ctures, utilities, sidewalks, pavements, and other facilities from damage caused by settlement, lateral movement, undermining, washout, and other hazards created by certhwork operations.

1.6 QUALTY ASSURANCE



Regulatory Requirements: Comply with applicable regulatory requirements of municipality and utility companies.

Under pavements, prepared subgrade must be proof rolled to a visually firm and stable condition and to a minimum of 95 percent maximum dry density as determined by a standard proctor test (ASTM: D698) with a minimum 10-ton smooth-wheeled roller in presence of a qualified Geotechnical Engineer or Technician prior to the placing of any base material.

C. A qualified Geotechnical Engineer or Technician, paid for by the Owner, shall monitor all site preparing, grading, and fill construction. The Engineer/Technician should observe and document the site preparation, existing fill material removal, and fill/backfill construction work,

and make appropriate field tests, as necessary, to verify that acceptable fill materials are being used and that construction is being performed in accordance with applicable plans, specifications and acceptable construction practices.

PART 2 - PRODUCTS

- 2.1 SOIL MATERIALS
 - A. General: Provide borrow soil materials when sufficient satisfactory soil materials are not available from excavations. Note: No warranty is made regarding the amount of on site soils material suitable for use as fill or backfill of any kind.
 - B. Satisfactory Soils: Satisfactory soils for use as fill are defined below:
 - 1. Structural fill: Fill under building slabs on grade and foundations, drives and parking, and behind retaining walls (where applicable): Where available, on site soil materials may be used where these on site soils conform to the requirements of the Delaware Department of Transportation Type G Select borrow. Where sufficient on site soils materials are not available, off site borrow conforming to the requirements of the Delaware Department of Transportation Type G select borrow shall be provided.
 - 2. Fill under walkways: Where available, on site foil materials may be used where these on site soils conform to the requirements of the Delaware Department of Transportation Type C select borrow. When sufficient on-site soils materials are not available, off site borrow conforming to the requirements of the Delaware Department of Transportation Type C select borrow shan be provided.
 - 3. Fill under unimproved grass or landscape areas: Any on site soil material free of organic material and rocks or lunge larger than 2 inches.
 - C. Backfill:
 - 1. Structural beckfill For all backfill under building slabs on grade and foundations, drives and packing, and behind retaining walls (where applicable): Where available, on site soil miterials may be used where these on site soils conform to the requirements of the Deraware Department of Transportation Type G Select borrow. Where sufficient on ite soils materials are not available, off site borrow conforming to the requirements of be Delaware Department of Transportation Type G select borrow shall be provided.



- 3. Backfill for all utility trenches under unimproved, grass areas: Any on site soil material free of organic material and rocks or clumps larger than 2 inches.
- D. Fill: All fill shall be comprised of satisfactory soil material as defined above, section 2.01.B.

- E. Subbase: Shall be Delaware Department of Transportation Type A CR-1 graded aggregate, DelDOT specification section 821.03.
- F. Bedding: Shall be Delaware Dept. of Transportation #57 stone.
- G. Drainage Fill Course: Shall be Delaware Dept. of Transportation #57 stone.
- 2.2 GEOTEXTILES
 - A. Geotextile Filter Fabric: Nonwoven geotextile, specifically manufactured as a chanage and separation geotextile; made from polypropylene staple fibers; and with the following tunnimum properties determined according to ASTM D 4759 and referenced standard test methods:
 - 1. Grab Tensile Strength: 120 lb; ASTM D 4632
 - 2. Tear Strength: 60 lb; ASTM D4533
 - 3. Puncture Resistance: 70 lb: ASTM D 4833.
 - 4. Water Flow Rate: 135 gpm per sq. ft.: ASTM D 4491
 - 5. Apparent Opening Size: No. 70: ASTM D 4751
 - B. Geotextile Support Fabric: Woven geotextiles, specifically manufactured for use as a support and separation geotextiles, comprised of UV stabilized porypropylene slit film; and with the following minimum properties according to ASTAD 4759 and referenced standard test methods:
 - 1. Grab Tensile Strength: 315 lb; AST (DA
 - 2. Tear Strength: 120 lb; ASTM D 433
 - 3. Puncture Resistance: 120 lb: AS IVI 24833.
 - 4. Apparent Opening Size: No. 40; ASTM D 4751.

2.3 GEOGRID REINFORCEMENT

A. For use in unsuitable coil creat, as directed by the Architect, shall be an integrally formed grid structure manufactured of a stress resistant polypropylene material. The geogrid shall accept applied force in the by positive mechanical interlock with compacted soil or construction fill materials. The geogrid shall possess sufficient flexural stiffness to enable efficient installation over weak or we

PART 3 EXECUTION

REPARATION

Protect structures, utilities, sidewalks, pavements, and other facilities from damage caused by settlement, lateral movement, undermining, washout, and other hazards created by earthwork operations.

- B. Preparation of subgrade for earthwork operations including removal of vegetation, topsoil, debris, obstructions, and deleterious materials from ground surface is specified in Division 2 Section "Site Clearing."
- C. Protect and maintain erosion and sedimentation controls, which are specified in Division 2 Section "Site Clearing," during earthwork operations.

D. Provide protective insulating materials to protect subgrades and foundation soils against freezing temperatures or frost.

3.2 DEWATERING

- A. Prevent surface water and ground water from entering excavations, from ponding on prepared subgrades, and from flooding Project site and surrounding area. The bidder is responsible for complete dewatering for all site work including stormwater management and erosion control facilities. All dewatering methods must be approved by DNREC and must be per an DNACC ESC Handbook, latest revision effective July 1, 2005. The bidder should thoroughly ramitarize themselves with existing on-site soils and groundwater conditions and review the deotechnical Report.
- B. Protect subgrades from softening, undermining, washout, and damage by rain or water accumulation.
 - 1. Reroute surface water runoff away from excavated areas. Do not allow water to accumulate in excavations. Do not use excavated tranches as temporary drainage ditches.
 - 2. Install a dewatering system to keep subgrades day and convey ground water away from excavations. Maintain until dewatering is no longer required.
- 3.3 EXPLOSIVES
 - A. Explosives: Do not use explosives.
- 3.4 SITE STRIPPING AND PREPARATION IN STRUCTURAL AREAS
 - A. At the start of construction all existing topsoil, roots, asphalt and existing fill material should be stripped and removed in their entirety from the proposed building, parking, and drive areas extending to a minimum lateral distance of 10' beyond outer edge of paved areas and 15' beyond outside edge of blidding footings. Any existing fill material that may be encountered in previously disturbed at as (e.g., utility trenches), as well as any existing utility lines, should also be removed matheir entirety from the building area.
 - B. After completing removal of the above, and upon reaching grade (in cut areas) and/or subgrade (in fill aleas), he entire area should be proof-rolled with a minimum 10-ton smooth-wheeled rolle or obser approved equipment. The purpose of the proof-rolling is to densify the exposed grade/dbgrade areas, which have been loosened or disturbed during the stripping/grading operation. In addition, the proof-rolling will expose any localized soft areas not encountered during the test boring program. In subgrade areas to receive structural fill, the exposed subgrade areas should be compacted to a <u>visually</u> firm and stable condition; compacted subgrade must be inspected and approved by the Geotechnical Engineer. This subgrade compaction effort will enable any structural fill to be placed and compacted at the required densities. Any localized soft and/or excessively wet subgrade areas encountered during this program, which cannot be adequately stabilized by drying and compacting, should be undercut and replaced with properly compacted structural fill (DelDOT Type G) or other suitable materials as directed by Geotechnical Engineer.

3.5 EXCAVATION, GENERAL

- A. Unclassified Excavation: Excavate to subgrade elevations regardless of the character of surface and subsurface conditions encountered. Unclassified excavated materials may include rock, soil materials, and obstructions. No changes in the Contract Sum or the Contract Time will be authorized for rock excavation or removal of obstructions.
 - 1. If excavated materials intended for fill and backfill include unsatisfactory soil materials and rock, replace with satisfactory soil materials.
- 3.6 EXCAVATION FOR RETAINING WALLS AND OTHER STRUCTURES (if appreciable
 - A. Excavate to indicated elevations and dimensions within a tolerance of plus ormines 1 mch. If applicable, extend excavations a sufficient distance from structures for placing and removing concrete formwork, for installing services and other construction, and for insteaded.
 - 1. Excavations for Footings and Foundations: Do not discurb borom of excavation. Excavate by hand to final grade just before placing connecte reinforcement. Trim bottoms to required lines and grades to leave solid base to receive other work.
 - 2. Excavation for Underground Tanks, Basins, and Mechanical or Electrical Utility Structures: Excavate to elevations and dimensions indicated within a tolerance of plus or minus 1 inch. Do not disturb bottom of excavations intended as bearing surfaces.

3.7 EXCAVATION FOR WALKS AND PAVEMENTS

A. Excavate surfaces under walks and paventers to indicated lines, cross sections, elevations, and subgrades.

3.8 EXCAVATION FOR UTILITY TRENCIES

- A. Excavate trenches to indicate gradients, lines, depths, and elevations.
- B. Excavate trenches to mit rem widths to provide the following clearance on each side of pipe or conduit.
 - 1. Clearance: 2 inches each side of pipe or conduit.
- C. Trench lottom: Excavate trenches deeper than bottom of pipe elevation to allow for bedding course (bedding course thickness and material shall be as specified on the drawings and/or specs and as ecommended by manufacturer). Hand excavate for bell of pipe.

.9 SUBGRADE INSPECTION

Notify Architect/Geotechnical Engineer when excavations have reached required subgrade.

If Geotechnical Engineer determines that unsatisfactory soil is present, continue excavation and replace with compacted backfill or fill material as directed.

- C. Proof-roll subgrade below pavements, walkways and structures with heavy pneumatic-tired equipment to identify soft pockets and areas of excess yielding. Do not proof-roll wet or saturated, or frozen subgrades.
 - 1. Completely proof-roll subgrade in one direction.

- 2. Proof-roll with a minimum 10 ton vibratory roller or a fully loaded tandem dump truck in the presence of a qualified soils technician.
- 3. Excavate soft spots, unsatisfactory soils, and areas of excessive pumping or rutting, as determined by Geotechnical Engineer, and replace with compacted backfill or fill as directed.
- D. Reconstruct subgrades damaged by freezing temperatures, frost, rain, accumulated water, dust/drought conditions, or construction activities, as directed by Geotechnical Engineer.

3.10 STORAGE OF SOIL MATERIALS

- A. Stockpile borrow soil materials and excavated satisfactory soil materials without intermixing. Place, grade, and shape stockpiles to drain surface water. Cover to prevent windoiown dust.
 - 1. Stockpile soil materials away from edge of excavations. De not store within drip line of remaining trees.

3.11 BACKFILL

- A. Place and compact backfill in excavations promptly, but he before completing the following:
 - 1. Construction below finish grade including found tions and retaining walls.
 - 2. Surveying locations of underground utilities for Record Documents.
 - 3. Testing and inspecting underground stilling
 - 4. Removing concrete formwork.
 - 5. Removing trash and debris.
 - 6. Removing temporary shoring and tracing, and sheeting.
 - 7. Installing permanent or temporar horizontal bracing on horizontally supported walls.
- B. Place backfill on subgrades free of mud, frost, snow, or ice.
- 3.12 UTILITY TRENCH BACKVILL
 - A. Place backfillion subgrades free of mud, frost, snow, or ice.
 - B. Place and compact bedding course on trench bottoms and where indicated. Shape bedding course to provide continuous support for bells, joints, and barrels of pipes and for joints, fittings and bodies of conduits.

Place and compact initial backfill, material as defined in section 2.01.C to a height of 12 inches over the utility pipe or conduit.

- 1. Carefully compact initial backfill under pipe haunches and compact evenly up on both sides and along the full length of utility piping or conduit to avoid damage or displacement of piping or conduit. Coordinate backfilling with utilities testing.
- D. Place and compact final backfill of material specified in section 2.01.C to final subgrade elevation.
- 3.13 STRUCTURAL LOAD BEARING FILL/BACKFILL

- A. Plow, scarify, bench, or break up sloped surfaces steeper than 1 vertical to 4 horizontal so fill material will bond with existing material.
- B. The load bearing fill material should be placed in horizontal thin lifts with a loose thickness no greater than 8 inches. For top 1-foot of pavement subgrade, each thin lift of fill material should be compacted to 98% maximum dry density, as determined by the Standard Proctor T st (ASTM D-698). For fills below 1 foot of pavement subgrade and for structural fill below building slab and footings/foundations, each thin lift of fill material should be compacted to 95% maximum dry density, as determined by the Standard Proctor Test (ASTM D-698). Structural fill placement, as defined above, shall extend to a minimum lateral distance of 10' beyond outer edge of paved areas and 15' beyond outside edge of building footings.
- C. Place soil material on subgrades free of mud, frost, snow, ice, or other deterious materials.

3.14 SOIL MOISTURE CONTROL

- A. Uniformly moisten or aerate subgrade and each subsequent fill or tackfill soil layer before compaction to within 2 percent of optimum moisture content.
 - 1. Do not place backfill or fill soil material on surfaces that are muddy, frozen, or contain frost or ice.
 - 2. Remove and replace, or scarify and air dry otherwise satisfactory soil material that exceeds optimum moisture content by 2 percent and is too wet to compact to specified dry unit weight.

3.15 COMPACTION OF SOIL BACKFELDS AND FILLS

- A. Place backfill and fill soil materias in layers not more than 8 inches in loose depth for material compacted by heavy compaction equipment, and not more than 4 inches in loose depth for material compacted by hand opened tampers.
- B. Place backfill and fill s al materials evenly on all sides of structures to required elevations, and uniformly along the full length of each structure.
- C. Compact coll materials to not less than the following percentages of maximum dry unit weight according to A. TM D 698:
- onder structures, building slabs, steps, and fills below 1-foot of pavement subgrade, carify and recompact top 12 inches of existing subgrade and each layer of backfill or fill soil material at 95 percent.
 - For top 1-foot of pavement subgrade, each thin lift of fill material should be compacted to 98% maximum dry density.
- . Under walkways, scarify and recompact top 6 inches (below subgrade and compact each layer of backfill or fill soil material at 95 percent.
- 4. Under lawn or unpaved areas, scarify and recompact top 6 inches below subgrade and compact each layer of backfill or fill soil material at 85 percent.
- 5. For utility trenches, compact each layer of backfill soil material at 95 percent.
- 3.16 GRADING

- A. General: Uniformly grade areas to a smooth surface, free of irregular surface changes. Comply with compaction requirements and grade to cross sections, lines, and elevations indicated.
 - 1. Provide a smooth transition between adjacent existing grades and new grades.
 - 2. Cut out soft spots, fill low spots, and trim high spots to comply with required surfage tolerances.
- B. Site Grading: Slope grades to direct water away from buildings and to prevent ponding subgrades to required elevations within the following tolerances:
 - 1. Lawn or Unpaved Areas: Plus or minus 1 inch.
 - 2. Walks: Plus or minus 1 inch.
 - 3. Pavements: Plus or minus 1/2 inch.

3.17 SUBBASE AND BASE COURSES

- Place subbase and base course on subgrades free of mud, frost, sno A.
- On prepared subgrade, place subbase and base course and r pavements, courts, and walks as B. follows:
 - Where shown on the plans, install separation get textile on prepared subgrade according 1. to manufacturer's written instructions, over upping sides and ends.
 - 2. Shape subbase course to required c rn e. vations and cross-slope grades.
 - Place subbase course 6 inches or has a compacted thickness in a single layer. 3.
 - Place subbase course that exceeds V inches in compacted thickness in layers of equal thickness, with no compacted layer more than 6 inches thick or less than 3 inches thick. 4.
 - Compact subbase course a optimum moisture content to required grades, lines, cross 5. sections, and thickness to not less than 95 percent of maximum dry unit weight according to ASTM D 698
- FIELD QUALITY CO 3.18
 - Testing Agenty: Awner shall engage a qualified independent geotechnical engineering testing A. for field quality-control testing. agency to
 - Testing a ency to inspect and test subgrades and each fill or backfill layer. Proceed with B. subsecrent earthwork only after test results for previously completed work complies with requirements.



opting Subgrade: At footing subgrades, at least one test of each soil stratum will be performed to verify design bearing capacities. Subsequent verification and approval of other footing subgrades may be based on a visual comparison of subgrade with tested subgrade when approved by Architect.

D. Testing agency will test compaction of soils in place according to ASTM D 1556, ASTM D 2167, ASTM D 2922, and ASTM D 2937, as applicable. Tests will be performed at the following locations and frequencies:

- 1. Paved Areas and Areas beneath athletic courts: At subgrade and at each compacted fill and backfill layer, at least 1 test for every 2000 sq. ft. or less of paved area or building slab, but in no case fewer than 3 tests.
- 2. Structural Fill and Backfill: At each compacted backfill layer, at least 1 test for each 2000 square feet, but no fewer than 2 tests.
- 3. Trench Backfill: At each compacted initial and final backfill layer, at least 1 test for each 150 feet or less of trench length, but no fewer than 2 tests.
- E. When testing agency reports that subgrades, fills, or backfills have not achieved degree of compaction specified, scarify and moisten or aerate, or remove and replace son to depth required; recompact and retest until specified compaction is obtained.

3.19 PROTECTION

- A. Protecting Graded Areas: Protect newly graded areas from traffic freezing, and erosion. Keep free of trash and debris.
- B. Repair and reestablish grades to specified tolerances where completed or partially completed surfaces become eroded, rutted, settled, or where they lose compaction due to subsequent construction operations or weather conditions.
 - 1. Scarify or remove and replace soil material to death as directed by Architect; reshape and recompact.
- C. Where settling occurs before Project correction period elapses, remove finished surfacing, backfill with additional soil material, compact, and reconstruct surfacing.
 - 1. Restore appearance, quanty, and condition of finished surfacing to match adjacent work, and eliminate evidence of restoration to greatest extent possible.

3.20 DISPOSAL OF SURFLU' AND WASTE MATERIALS

A. Disposal: Remove surplus satisfactory soil and waste material, including unsatisfactory soil, trash, and debits, red legally dispose of it off Owner's property.




SECTION 312319 - DEWATERING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplem Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section includes construction dewatering.
- B. Related Sections:
 - 1. Division 01 Section "Construction Progress Documentation" for recording preexisting conditions and dewatering system progress.
 - 2. Division 31 Section "Earth Moving" for excertaining, tackfilling, site grading, and for site utilities.
 - 3. Division 31 Section "Excavation Support and Protection" for shoring, bracing, and sheet piling of excavations.
 - 4. Division 33 Section "Subdrainage" for permanent foundation wall, underfloor, and footing drainage.

1.3 PERFORMANCE REQUIREMENTS

- A. Dewatering Performance Design, furnish, install, test, operate, monitor, and maintain dewatering system of scaficient scope, size, and capacity to control hydrostatic pressures and to lower, control, remove, and dispose of ground water and permit excavation and construction to proceed on dre, strole subgrades.
 - 1. Delegated Design: Design dewatering system, including comprehensive engineering analysis oy a qualified professional engineer, using performance requirements and design fiteria indicated.
 - Continuously monitor and maintain dewatering operations to ensure erosion control, stability of excavations and constructed slopes, that excavation does not flood, and that damage to subgrades and permanent structures is prevented.
 - Prevent surface water from entering excavations by grading, dikes, or other means.
 - . Accomplish dewatering without damaging existing buildings, structures, and site improvements adjacent to excavation.
 - 5. Remove dewatering system when no longer required for construction.

1.4 SUBMITTALS

- A. Shop Drawings: For dewatering system. Show arrangement, locations, and details of wells and well points; locations of risers, headers, filters, pumps, power units, and discharge lines; and means of discharge, control of sediment, and disposal of water.
 - 1. Include layouts of piezometers and flow-measuring devices for monitoring performance of dewatering system.
 - 2. Include a written plan for dewatering operations including control procedure adopted if dewatering problems arise.
- B. Delegated-Design Submittal: For dewatering system indicated to comply with performance requirements and design criteria, including analysis data signed and sealed by the qualified professional engineer responsible for their preparation.
- C. Qualification Data: For qualified Installer, land surveyor, and professional engineer.
- D. Field quality-control reports.
- E. Other Informational Submittals:
 - 1. Photographs: Show existing conditions of aujoining construction and site improvements that might be misconstrued as damage caused ov lewatering operations.

1.5 QUALITY ASSURANCE

- A. Installer Qualifications: An experienced installer that has specialized in design of dewatering systems and dewatering work
- B. Regulatory Requirements: Couply with governing EPA notification regulations before beginning dewatering. Couply with hauling and disposal regulations of authorities having jurisdiction.
- C. Preinstallation Conference: Conduct conference at Project site (Smyrna High School).
 - 1. Review nethods and procedures related to dewatering including, but not limited to, the following:



Inspection and discussion of condition of site to be dewatered including coordination with temporary erosion control measures and temporary controls and protections.

- b. Geotechnical report.
- c. Proposed site clearing and excavations.
- d. Existing utilities and subsurface conditions.
- e. Coordination for interruption, shutoff, capping, and continuation of utility services.
 - . Construction schedule. Verify availability of Installer's personnel, equipment, and facilities needed to make progress and avoid delays.
- g. Testing and monitoring of dewatering system.

1.6 PROJECT CONDITIONS

- A. Interruption of Existing Utilities: Do not interrupt any utility serving facilities occupied by Owner or others unless permitted under the following conditions and then only after arranging to provide temporary utility according to requirements indicated:
 - 1. Notify Architect no fewer than (2) days in advance of proposed interruption of utility
 - 2. Do not proceed with interruption of utility without Architect's and Owner's writtenpermission.
- B. Project-Site Information: A geotechnical report has been prepared for this Project and is available for information only. The opinions expressed in this report are those of geotechnical engineer and represent interpretations of subsoil conditions, tests, and results of analyses conducted by geotechnical engineer. Owner will not be responsible for interpretations or conclusions drawn from this data.
 - 1. Make additional test borings and conduct other exploritory operations necessary for dewatering.
 - 2. The geotechnical report is referenced elsewhere in the Project Manual.
- C. Survey Work: Engage a qualified land surveyor or processional engineer to survey adjacent existing buildings, structures, and site improvements, establishing exact elevations at fixed points to act as benchmarks. Clearly identify be chinary and record existing elevations.
 - 1. During dewatering, regularly restrict benchmarks, maintaining an accurate log of surveyed elevations for comparison with original elevations. Promptly notify Architect if changes in elevations occur or hicnecks, sags, or other damage is evident in adjacent construction.

PART 2 - PRODUCTS (Not Used

PART 3 - EXECUTION

- 3.1 PREPARATIO
 - A. Prosec structures, utilities, sidewalks, pavements, and other facilities from damage caused by settlement, lateral movement, undermining, washout, and other hazards created by dewatering operations.
 - Prevent surface water and subsurface or ground water from entering excavations, from ponding on prepared subgrades, and from flooding site and surrounding area.
 - 2. Protect subgrades and foundation soils from softening and damage by rain or water accumulation.
 - B. Install dewatering system to ensure minimum interference with roads, streets, walks, and other adjacent occupied and used facilities.

- 1. Do not close or obstruct streets, walks, or other adjacent occupied or used facilities without permission from Owner and authorities having jurisdiction. Provide alternate routes around closed or obstructed traffic ways if required by authorities having jurisdiction.
- C. Provide temporary grading to facilitate dewatering and control of surface water.
- D. Monitor dewatering systems continuously.
- E. Promptly repair damages to adjacent facilities caused by dewatering.
- F. Protect and maintain temporary erosion and sedimentation controls, which are openfied in Division 01 Section "Temporary Facilities and Controls" Division 31 Section "Site Clearing" during dewatering operations.

3.2 INSTALLATION

- A. Install dewatering system utilizing wells, well points, or eighter methods complete with pump equipment, standby power and pumps, filter material gradient, valves, appurtenances, water disposal, and surface-water controls.
 - 1. Space well points or wells at intervals required to provide sufficient dewatering.
 - 2. Use filters or other means to prevent purpose of fine sands or silts from the subsurface.
- B. Before excavating below ground-water level, place system into operation to lower water to specified levels. Operate system continuously until drains, sewers, and structures have been constructed and fill materials have been placed or until dewatering is no longer required.
- C. Provide an adequate system to lower and control ground water to permit excavation, construction of structures, and placement of fill materials on dry subgrades. Install sufficient dewatering equipment to that water-bearing strata above and below bottom of foundations, drains, sewers, and other excavations.
 - 1. Do not permit open-sump pumping that leads to loss of fines, soil piping, subgrade softening, and slope instability.
- D. Reduce hydrostatic head in water-bearing strata below subgrade elevations of foundations, drans sewers, and other excavations.



Maintain piezometric water level to the minimum DNREC required depth below surface of excavation.

Dispose of water removed by dewatering in a manner that avoids endangering public health, property, and portions of work under construction or completed. Dispose of water and sediment in a manner that avoids inconvenience to others. Provide sumps, sedimentation tanks, and other flow-control devices as required by authorities having jurisdiction.

F. Provide standby equipment on site, installed and available for immediate operation, to maintain dewatering on continuous basis if any part of system becomes inadequate or fails. If dewatering

requirements are not satisfied due to inadequacy or failure of dewatering system, restore damaged structures and foundation soils at no additional expense to Owner.

- 1. Remove dewatering system from Project site on completion of dewatering. Plug or fill well holes with sand or cut off and cap wells a minimum of 36 inches below overlying construction.
- G. Damages: Promptly repair damages to adjacent facilities caused by dewatering operation

3.3 FIELD QUALITY CONTROL

- A. Observation Wells: Provide, take measurements, and maintain at least the maximum number of observation wells or piezometers indicated; additional observation wells r ay be required by authorities having jurisdiction.
 - 1. Observe and record daily elevation of ground water and piezometric water levels in observation wells.
 - 2. Repair or replace, within 24 hours, observation wells hat become inactive, damaged, or destroyed. In areas where observation wells are not functioning properly, suspend construction activities until reliable observations can be made. Add or remove water from observation-well risers to demonstrate that observation wells are functioning properly.
 - 3. Fill observation wells, remove piezometers, perfill holes when dewatering is completed.
- B. Provide continual observation to ensure that subsurface soils are not being removed by the dewatering operation.

END OF SECTION 31 23 19



SECTION 311216 - ASPHALT PAVING RESTORATION

- PART 1 GENERAL
- 1.01 RELATED DOCUMENTS
 - A. The general provisions of the Contract, including the Conditions of the Contract (General Supplementary and other Conditions, if any) and Division 1 as appropriate, apply to the Vork specified in this Section.
- 1.02 SECTION INCLUDES
 - A. Refer to Scope Information Sheets for this contract bound in the Project Menual under Section 01010, SUMMARY OF WORK. The Scope Information Sheets describe generally the work included in each contract, but the work is not necessarily limited to that described.
 - B. Provisions for hot-mixed asphalt paving over prepared subhase
 - C. Proof rolling of prepared subbase.
 - D. Traffic Paint
 - E. Contractor shall coordinate all construction documents; where conflicts arise between these specifications and the design drawings, the hore stringent shall apply. The Contractor is advised to contact Tetra Tech with any questions prior to resolving any conflicts, or modifying any of the original design.
- 1.03 RELATED SECTIONS
 - A. Prepared subbase is specified in another Division 2, Section "Earthwork".
 - B. Saw-cutting of elses of existing pavement is specified in site-clearing section.
- 1.04 SUBMITTALS
 - A. Material Sertificates signed by material producer and Contractor, certifying that each material itera complies with or exceeds specified requirements.

ROJECT/SITE CONDITIONS

Weather Limitations: Apply prime and tack coats when ambient temperature is above 50 deg F (10 deg C) and when temperature has not been below 35 deg F (1 deg C) for 12 hours immediately prior to application. Do not apply when base is wet or contains an excess of moisture.

- B. Construct hot-mixed asphalt surface course when atmospheric temperature is above 40 deg F (4 deg C) and when base is dry. Base course may be placed when air temperature is above 30 deg F (minus 1 deg C) and rising.
- C. Grade Control: Establish and maintain required lines and elevations.

1.06 MATERIALS

- A. General: Use locally available materials and gradations that exhibit a satisfactory record of previous installations.
- B. Coarse Aggregate: Sound, angular crushed stone, crushed gravel, or properly cured crushed blast furnace slag, complying with ASTM D 692-88.
- C. Fine Aggregate: Sharp-edged natural sand or sand prepared from stone, properly cured bl furnace slag, gravel, or combinations thereof, complying with ASTM D 1073.
- D. Asphalt Cement: ASTM D 3381 for viscosity-graded material; ASTM D 940 for penetration-graded material.
- E. Prime Coat: Cut-back asphalt type, ASTM D 2027; MC-30, MC 70, at MC-250.
- F. Tack Coat: Emulsified asphalt; ASTM D 977.
- G. Subbase Reinforcement Fabric. Structural Geogrid BX
- 1.07 ASPHALT-AGGREGATE MIXTURE
 - A. Provide plant-mixed, hot-laid asphalt-aggregate mixture complying with ASTM D 3515 and applicable Delaware Department of Transportation standards.

1.08 TRAFFIC PAINT

- A. Traffic Paint: Apply traffic paint for striping and other markings with mechanical equipment to produce uniform straight edges. Apply at manufacturer's recommended rates to provide a 15-mil minimum wet film the knews. All traffic striping is to conform to Delaware Department of Transportation standard specifications.
- PART 2 EXECUTION

2.01 SURFACE PREPARATION

- A. General: Remove loose material from compacted subbase surface immediately before applying prime coat.
 - Proof-roll prepared subbase surface to check for unstable areas and areas requiring additional compaction.
 - Notify Engineer of unsatisfactory conditions. Do not begin paving work until deficient subbase areas have been corrected and are ready to receive paving.
- D. Prime Coat: Apply at rate of 0.20 to 0.50 gal. per sq. yd., over compacted subgrade. Apply material to penetrate and seal, but not flood, surface. Cure and dry as long as necessary to attain penetration and evaporation of volatile.

- E. Tack Coat: Apply to contact surfaces of previously constructed asphalt or Portland cement concrete and surfaces abutting or projecting into hot-mixed asphalt pavement. Distribute at rate of 0.05 to 0.15 gal. per sq. yd. of surface.
- F. Allow to dry until at proper condition to receive paving.
- G. Exercise care in applying bituminous materials to avoid smearing of adjoining concrete surfaces. Remove and clean damaged surfaces.
- 2.02 PLACING MIX
 - A. General: Place hot-mixed asphalt mixture on prepared surface, spread, and strike of Spr mixture at minimum temperature of 225 deg F (107 deg C). Place areas inaccessible to equipment by hand. Place each course to required grade, cross-section, and compacted thickness.
 - B. Paver Placing: Place in strips not less than 10 feet wide, unless otherwise acceptable to Engineer. After first strip has been placed and rolled, place succeeding strips and extend rolling to overlap previous strips. Complete the base course for a section before placing surface course.
 - C. Immediately correct surface irregularities in finish course behind paver. Remove excess material forming high spots with shovel or lute.
 - D. Joints: Make joints between old and new parameters, or between successive days' work, to ensure continuous bond between adjoining work. Construct joints to have same texture, density, and smoothness as other sectors or hot-mixed asphalt course. Clean contact surfaces and apply tack coat.
- 2.03 ROLLING
 - A. General: Begin rolling when mixture will bear roller weight without excessive displacement.
 - B. Compact mixture with hownand tampers or vibrating plate compactors in areas inaccessible to rollers.
 - C. Breakdown Roling: Accomplish breakdown or initial rolling immediately following rolling of joint, and outside edge. Check surface after breakdown rolling and repair displaced areas by loosning and filling, if required, with hot material.

Second Rolling: Follow breakdown rolling as soon as possible, while mixture is hot. Continue second rolling until mixture has been evenly compacted.

Finish Rolling: Perform finish rolling while mixture is still warm enough for removal of roller marks. Continue rolling until roller marks are eliminated and course has attained 95 percent laboratory density.

F. Patching: Remove and replace paving areas mixed with foreign materials and defective areas. Cut out such areas and fill with fresh, hot-mixed asphalt. Compact by rolling to specified surface density and smoothness.

- G. Protection: After final rolling, do not permit vehicular traffic on pavement until it has cooled and hardened.
- H. Erect barricades to protect paving from traffic until mixture has cooled enough not to become marked.

2.04 FIELD QUALITY CONTROL

- A. General: Test in-place hot-mix asphalt courses for compliance with requirements for michness and surface smoothness. Repair or remove and replace unacceptable paving a directed by Engineer.
- B. Thickness: In-place compacted thickness tested in accordance with ASTNO 3549 will not be acceptable if exceeding following allowable variations:
 - 1. Base Course: Plus or minus $\frac{1}{2}$ inch.
 - 2. Surface Course: Plus or minus 1/4 inch.
- C. Surface Smoothness: Test finished surface of each hot-mixed asphalt course for smoothness, using 10-foot straightedge applied parallel with and at right angles to centerline of paved area. Surfaces will not be acceptable if exceeding the following tolerances for smoothness:
 - 1. Base Course Surface: 1/4 inch.
 - 2. Wearing Course Surface: 3/16 inc.
- D. Check surface areas at intervals as directed by Engineer.

END OF SECTION 32 12 16

SECTION 31 10 00

SITE CLEARING

PART 1 GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Capplementary Conditions and Division 1 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. This Section includes the following:
 - 1. Removing existing trees, shrubs, grounderver, pants, and grass.
 - 2. Clearing and grubbing.
 - 3. Stripping and stockpiling topsoil.
 - 4. Removing above- and below-grace she improvements.
 - 5. Disconnecting, capping or sealing, and abandoning site utilities in place removing site utilities.
 - 6. Temporary erosion and ectime tation control measures.
- B. Related Sections include the following:
 - 1. Division 2 Section "Larthwork" for soil materials, excavating, backfilling, and site grading.
 - 2. Division 2 vection "Lawns and Grasses" for finish grading including preparing and placing pointing soil mixes and testing of topsoil material.

1.3 DECINITIONS



Topsoil! Natural or cultivated surface-soil layer containing organic matter and sand, silt, and hay particles; friable, pervious, and black or a darker shade of brown, gray, or red than underlying subsoil; reasonably free of subsoil, clay lumps, gravel, and other objects more than 2 inches in diameter; and free of subsoil and weeds, roots, toxic materials, or other nonsoil materials.

B. Tree Protection Zone: Area surrounding individual trees or groups of trees to be protected during construction, and defined by the drip line of individual trees or the perimeter drip line of groups of trees, unless otherwise indicated.

Tetra Tech 200-15704-17001

SITE CLEARING 31 10 00 - 1

1.4 MATERIAL OWNERSHIP

A. Except for stripped topsoil or other materials indicated to remain Owner's property, cleared materials shall become Contractor's property and shall be removed from Project site.

1.5 SUBMITTALS (NOT USED)

1.6 PROJECT CONDITIONS

- A. Traffic: Minimize interference with adjoining roads, streets, walks, and other adjacent or used facilities during site-clearing operations.
 - 1. Do not close or obstruct streets, walks, or other adjacent occupied or used facilities without written permission from the University of Delaware
 - 2. Provide alternate routes around closed or obstructed traffic w
- B. Utility Locator Service: Notify Miss Utility a minimum of 3 lays prior to performing any land disturbing activities. Contact Miss Utility of Delmarva -800-282-8555
- C. Do not commence site clearing operations until temporary erosion and sedimentation control measures are in place.

PART 2 - PRODUCTS (Not Applicable)

2.1 SOIL MATERIALS

- A. Satisfactory Soil Materials, Requirements for satisfactory soil materials are specified in Division 2 Section "Earth york"
 - 1. Obtain approved burrow soil materials off-site when satisfactory soil materials are not available or site.

PART 3 - EXECUTION

3.1 **PREPARATION**

Protect and maintain benchmarks and survey control points from disturbance during construction.

Locate and clearly flag trees and vegetation to remain or to be relocated.

- C. Protect existing site improvements to remain from damage during construction.
 - 1. Restore damaged improvements to their original condition, as acceptable to Owner.

SITE CLEARING 311000 - 2

3.2 TREE PROTECTION

- A. Erect and maintain temporary fencing around tree protection zones before starting site clearing. Remove fence when construction is complete.
 - 1. Do not store construction materials, debris, or excavated material within fenced are
 - 2. Do not permit vehicles, equipment, or foot traffic within fenced area.
 - 3. Maintain fenced area free of weeds and trash.
- B. Do not excavate within tree protection zones, unless otherwise indicated.
- C. Where excavation for new construction is required within tree protection zoles, hand clear and excavate to minimize damage to root systems. Use narrow-tine spather forks, comb soil to expose roots, and cleanly cut roots as close to excavation as possible.
 - 1. Cover exposed roots with burlap and water regularly.
 - 2. Temporarily support and protect roots from damage and they are permanently redirected and covered with soil.
 - 3. Coat cut faces of roots more than 1-1/2 inches in dameter with an emulsified asphalt or other approved coating formulated for use in damaged plant tissues.
 - 4. Backfill with soil as soon as possible.
- D. Repair or replace trees and vegetation in figure to remain that are damaged by construction operations, in a manner approved by Architect
 - 1. Replace trees that cannot be reported and restored to full-growth status, as determined by Architect.

3.3 UTILITIES

1.

- A. Locate, identity, deconnect, and seal or cap off utilities indicated to be removed.
 - 1. A range with utility companies to shut off indicated utilities.
 - 2. Owner will arrange to shut off indicated utilities when requested by Contractor.
- B. Existing Utilities: Do not interrupt utilities serving facilities occupied by Owner or others unless permitted under the following conditions and then only after arranging to provide temporary utility services according to requirements indicated:
 - Notify the Owner not less than two days in advance of proposed utility interruptions.
 - 2. Do not proceed with utility interruptions without the University's written permission.
- C. Excavate for and remove underground utilities indicated to be removed.
- D. Removal of underground utilities is included in Division 2 Sections covering site utilities.

Tetra Tech 200-15704-17001

3.4 CLEARING AND GRUBBING

- A. Remove obstructions, trees, shrubs, grass, and other vegetation to permit installation of new construction.
 - 1. Do not remove trees, shrubs, and other vegetation indicated to remain or to be relocated
 - 2. Cut minor roots and branches of trees indicated to remain in a clean and careful manner where such roots and branches obstruct installation of new construction.
 - 3. Grind stumps and remove roots, obstructions, and debris extending to a dept inches below exposed subgrade.
 - 4. Use only hand methods for grubbing within tree protection zone.
 - 5. Chip removed tree branches and dispose of off-site.
- B. Fill depressions caused by clearing and grubbing operations with scalafactory soil material unless further excavation or earthwork is indicated.
 - 1. Place fill material in horizontal layers not exceeding a love depth of 8 inches, and compact each layer to a density equal to adjacent original ground.

3.5 TOPSOIL STRIPPING

- A. Remove grass before stripping topsoil.
- B. Strip topsoil to whatever depths are encountered in a manner to prevent intermingling with underlying subsoil or other waste materials.
 - 1. Remove subsoil and non-solumate als from topsoil, including trash, debris, weeds, roots, and other waste materials.
- C. Stockpile topsoil materials ways from edge of excavations without intermixing with subsoil. Grade and shape stockpile to grain surface water. Cover to prevent windblown dust.
 - 1. Do not stocpile topsoil within tree protection zones.
 - 2. Dispose of meess topsoil as specified for waste material disposal.
 - 3. Sto apile surplus topsoil to allow for re-spreading deeper topsoil.
- 3.6 SITE IMPROVEMENTS

Remove existing above- and below-grade improvements as indicated and as necessary to incilitate new construction.

Remove slabs, paving, curbs, gutters, and aggregate base as indicated.

- 1. Unless existing full-depth joints coincide with line of demolition, neatly saw-cut length of existing pavement to remain before removing existing pavement. Saw-cut faces vertically.
- 2. Paint cut ends of steel reinforcement in concrete to remain to prevent corrosion.

SITE CLEARING 311000 - 4 Tetra Tech 200-15704-17001

3.7 DISPOSAL

A. Disposal: Remove surplus soil material, unsuitable topsoil, obstructions, demolished materials, and waste materials including trash and debris, and legally dispose of them off Owner's property.

1. Separate recyclable materials produced during site clearing from other non-recyclable materials. Store or stockpile without intermixing with other materials and transport them to recycling facilities.

END OF SECTION 31 10 00

Tetra Tech 200-15704-17001

SITE CLEARING 31 10 00 - 5



SECTION 322210 - EROSION AND SEDIMENT CONTROL

PART 1 GENERAL

1.01 RELATED DOCUMENTS

Drawings and general provisions of contract, including General and Supplementary Conditions and Division 1 and 2 Specification sections apply to work of this section.

- B. State of Delaware Sediment and Stormwater Regulations, latest edition
- C. State of Delaware Erosion and Sediment Control Handbook, latest revision

1.2 SUMMARY

- A. This Section includes the following:
 - 1. Temporary erosion control devices require rating construction.
- B. Related Sections include the following:
 - 1. Division 2 Section *Site Clearing* for site stripping, grubbing, stripping, and stockpiling topsoil and removal of above and below ground improvements.
 - 2. Division 2 Section *Earthwork* for excavation, grading, fill placement, and back-fill of excavations with specified materials.
- 1.3 DEFINITIONS (NOT USED)
- 1.4 SUBMITTALS
 - A. Comply with Division 1 Section Submittal Procedures
 - B. For Approval:
 - Product Data: For the following:
 - . Filter cloth and pre-fabricated silt fence.
 - b. geotextile fabrics
 - For Information:
 - 1. Materials test report: From a qualified testing agency indicating and interpreting test results for compliance of the following with requirements indicated:
 - a. Classification according to ASTM of all stone and aggregate for use in erosion control devices such as stabilized construction entrances, rock check dams, temporary sediment traps.

1.5 QUALITY ASSURANCE

- A. Geotechnical Testing Agency Qualifications: An independent testing agency qualified according to ASTM E 329 to conduct soil materials and rock-definition testing, as documented according to ASTM D 3740 and ASTM E 548.
- B. General Intent: In accordance with State of Delaware, Section 1, Title 7, Delaware Code Chapter 40 "Erosion and Sediment Control Act," erosion and sediment control will be minimized during construction of the project by temporary and/or permanent stabilization by seeding and other controls to limit erosion. All erosion and sediment controls shall be constructed and installed as detailed on drawings and as per DNREC's Delaware Ero aon & Sodiment Control Handbook, latest revision. As part of the base bid contract scope of work Contractor is responsible for installing additional Erosion & Sediment Control (ASC) measures as directed by Certified Construction Reviewer CCR and/or Engineer as construction progresses. The actual ESC measures shown on the plan are the minimum required to start work and install the proposed site improvements. As construction progresses additional ESC items may be required (i.e. dewatering bag, stone check dams, silt tubes, etc.) to comply with the State Sediment and Stormwater Regulations; these ESC items must be netalled as directed by CCR and/or Engineer at no additional cost to owner. Contractoris also responsible for maintaining all ESC measures throughout duration of construction incruding to-seeding disturbed areas and repairing ESC items as directed by CCR and Engineer.

1.6 PROJECT CONDITIONS (NOT USED)

PART 2 - PRODUCTS

- 2.1 SILT FENCE (must conform to DNREC ESC Handbook Standards, latest revision):
- 2.2 STABILIZED CONSTRUCTION ENTRANCE (must conform to DNREC ESC Handbook Standards, latest revision)
 - A. Stone: Use 2" stone or recycled concrete equivalent
 - B. Filter Fabric: Woven or non-woven fabric consisting only of continuous chain polymeric filaments or variable polyester. The fabric shall be inert to commonly encountered chemicals and hydroca oons, and be mildew and rot resistant. Fabric shall be Trevira Spunbond 1135, Mirafi 600X or approved equivalent.

PART 7 - EXECUTION

ISTALL EROSION CONTROL MEASURES in accordance with the latest issue of Delaware osion Control Handbook and as shown on drawings.

SILT FENCE: Install in accordance with the latest issue of Delaware Erosion Control Handbook and as shown on drawings. Maximum allowable slope length shall be per Delaware Erosion Control Handbook Standards and specifications for silt fence.

- A. Filter cloth to be fastened securely to fence posts with wire ties or staples.
 - 1. Stake size, installation, spacing, and fabric overlap shall conform to the DNREC ESC Handbook details and specifications.

B. Maintenance shall be performed as needed and material removed when "bulges" develop in the silt fence.

3.3 STABILIZED CONSTRUCTION ENTRANCE (if required):

- A. Length: As required, but not less than 50 feet.
- B. Thickness: Not less than 6 inches.
- C. Width: 10 foot minimum, but not less than the full width at points where incress or egress occurs.
- D. Surface Water: All surface water flowing or diverted toward construction entrances shall be piped across the entrance. If piping is impractical, a mountable bern with 5:1 slopes will be permitted.
- E. Maintenance: The entrance shall be maintained in a condition, which will prevent tracking or flowing of sediment onto public rights-of-way. This pay require periodic top dressing with additional stone as conditions demand and repair and/cr cleanout of any measures used to trap sediment. All sediment spilled, dropped, washed or tracked onto public rights-of-way must be removed immediately.
- F. Washing: Vehicle wheels shall be cleaned to remove sediment prior to entrance onto public rights -of-way. When washing is required, it shall be done on an area stabilized with stone and which drains into an approved sediment-trapping device.

END OF SECTION 32 22 10



SECTION 33 25 10 - WATER DISTRIBUTION

PART 1 GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplen Conditions and Division 1 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. This Section includes water-distribution piping and specialties outside the building for the following:
 - 1. Combined water service and fire-service mains.

1.3 DEFINITIONS

- A. Combined Water Service and Fire-Service Main: Exterior water piping for both domestic-water and fire-suppression piping.
- B. Fire-Service Main: Exterior fire-suppression-value piping.
- C. Fire-Suppression-Water Piping: Interior Greesuppression-water piping.
- D. Water-Distribution Piping: Interior domestic-water piping.
- E. Water Service: Exterior dom.stic-water piping.
- F. The following are industry abbreviations for plastic materials:
 - 1. PE: Po vet vlene plastic.
 - 2. DIP: Ducile Iron Pipe.

1.4 SUBMITTAL

- . For approval:
 - Pipe and Fittings.
 - Gate Valves and accessories.
 - 3. Tapping Sleeves and Valves.
 - 4. Fire Hydrants.
- 1.5 QUALITY ASSURANCE
 - A. Product Options: Drawings indicate size, profiles, and dimensional requirements of piping and specialties and are based on the specific system indicated. Refer to Division 1 Section "Product Requirements."

- B. Regulatory Requirements:
 - 1. Comply with requirements of utility company supplying water. Include tapping of water mains and backflow prevention.
 - 2. Comply with standards of authorities having jurisdiction for potable-water-service piping including materials, installation, testing, and disinfection.
 - 3. Comply with standards of authorities having jurisdiction for fire-suppression waterservice piping, including materials, hose threads, installation, and testing.
- C. Piping materials shall bear label, stamp, or other markings of specified testing agend

1.6 DELIVERY, STORAGE, AND HANDLING

- A. Preparation for Transport: Prepare valves, including fire hydrants, according to the following:
 - 1. Ensure that valves are dry and internally protected against rus and corrosion.
 - 2. Protect valves against damage to threaded ends and flange faces.
 - 3. Set valves in best position for handling. Set valves closed to revent rattling.
- B. During Storage: Use precautions for valves, including fire hydrants, according to the following:
 - 1. Do not remove end protectors unless necessary for inspection; then reinstall for storage.
 - 2. Protect from weather. Store indoors and maintain temperature higher than ambient dewpoint temperature. Support off the second or pavement in watertight enclosures when outdoor storage is necessary.
- C. Handling: Use sling to handle values and fire hydrants if size requires handling by crane or lift. Rig values to avoid damage to exposed parts. Do not use handwheels or stems as lifting or rigging points.
- D. Deliver piping with factor applied end caps. Maintain end caps through shipping, storage, and handling to prevent pipeend damage and to prevent entrance of dirt, debris, and moisture.
- E. Protect stored sipiling from moisture and dirt. Elevate above grade.
- F. Protect franges, fittings, and specialties from moisture and dirt.
- 1.7 PROJECT CONDITIONS

Existing Utilities: Do not interrupt utilities serving facilities occupied by Owner or others unless permitted under the following conditions and then only after arranging to provide temporary utility services according to requirements indicated:

- 1. Notify Architect, City of Newark not less than seven days in advance of proposed utility interruptions.
- 2. Do not proceed with utility interruptions without Architect and the Town of Smyrna's written permission.

1.8 COORDINATION

A. Coordinate connection to water main with utility company.

PART 2 - PRODUCTS

2.1 MANUFACTURERS

- A. In other Part 2 articles where subparagraph titles below introduce lists, the following requirements apply for product selection:
 - Available Products: Subject to compliance with requirements products that may be incorporated into the Work include, but are not limited to, the products specified.
 Products: Subject to compliance with requirements, provide one of the products
 - 2. Products: Subject to compliance with requirements, provide one of the products specified.
 - 3. Available Manufacturers: Subject to compliance with requirements, manufacturers offering products that may be incorporated into the Work include, but are not limited to, the manufacturers specified.
 - 4. Manufacturers: Subject to compliance with requirements, provide products by the manufacturers specified.

2.2 PIPING MATERIALS

A. Refer to Part 3 "Piping Applications" Axicle for applications of pipe, tube, fitting, and joining materials.

2.3 DUCTILE-IRON PIPE AND TITNING

- A. Mechanical-Joint, Ductile-In nive: C-900, with mechanical-joint, bell- and plain-spigot end.
 - 1. Mechanical-Joint Ductile-Iron Fittings: C-900, ductile- or gray-iron standard pattern or AWWA (2013, ductle-iron compact pattern.
 - a. Glands, Gaskets, and Bolts: C-900, ductile- or gray-iron glands, rubber gaskets, and steel bolts.
- B. Pust-on-Joint, Ductile-Iron Pipe: Class 52, with push-on-joint, bell- and plain-spigot end unless prooved or flanged ends are indicated.
 - Push-on-Joint, Ductile-Iron Fittings: Class 52, ductile- or gray-iron standard pattern or AWWA C153, ductile-iron compact pattern.
 - a. Gaskets: Class 52, rubber.
- .4 GATE VALVES
 - A. AWWA, Cast-Iron Gate Valves:

- 1. Manufacturers:
 - a. American Cast Iron Pipe Co.; American Flow Control Div.
 - b. American Cast Iron Pipe Co.; Waterous Co. Subsidiary.
 - c. Mueller Co.; Water Products Div.
 - d. McWane, Inc.; Clow Valve Co. Div.
 - e. McWane, Inc.; Kennedy Valve Div.
 - f. McWane, Inc.; Tyler Pipe; Utilities Div.
 - g. NIBCO INC.
 - h. United States Pipe and Foundry Company.
- 2. Non-rising Stem Gate Valves 3 Inches (80 mm) and Larger: AWVA C509, resilient seated; bronze stem, iron body and bonnet with epoxy coated interior surfaces, iron disc with epoxy coating and replaceable steel reinforced rubber stat double O-Ring seal stuffing box, 2" operating nut, open left, 200-psig (138) kPa) working pressure, mechanical joint ends. Mueller A-2370-20, open left or approved equal.
- 3. Valve Boxes: Cast-iron box having top section and cover with lettering "WATER," bottom section with base of size to fit over valve and barrel approximately 5 inches (124 mm) in diameter, and adjustable cast iron extension of length required for depth of bury of valve. Mueller Type H10360, or approved equal.
- 4. Provide one steel tee-handle operating whench. Wrench shall have tee handle with one pointed end, stem of length to operate deepest valve, and socket-fitting valve-operating nut.

2.5 GATE VALVE ACCESSOR ES AND SPECIALTIES

- A. Tapping-Sleeve Assemblies. Comply with MSS SP-60. Include sleeve and valve compatible with drilling machine
 - 1. Manufacturers:

g.

- American Cast Iron Pipe Co.; Waterous Co. Subsidiary.
- Meller Co.; Water Products Div.
- International Piping Services Company.
- McWane, Inc.; Clow Valve Co. Div. (Oskaloosa).
- McWane, Inc.; Kennedy Valve Div.
- f. McWane, Inc.; M & H Valve Company Div.
 - United States Pipe and Foundry Company.
- 2. Tapping Sleeve: Tapping sleeves shall be a Mueller mechanical joint tapping sleeve model H-615, or approved equal. The sleeve shall be a 2 piece bolted sleeve with flanged outlet for new branch connection. The outlet flange dimensions and drilling shall comply with ANSI B16.1, class 125. Sleeve shall be iron body with 3/4" NPT test plug and shall have a maximum working pressure of 200 psig.

3. Tapping Valve: Tapping valves shall be a Mueller Double Disc tapping valve model H-667, or approved equal. The valve shall meet or exceed all applicable requirements of ANSI/AWWA C500 standards. The inlet flange shall comply with ANSI B16.1, class 125 drilling, the mechanical joint outlet shall comply with ANSI/AWWA C114 Standards. The valve shall have a bronze mounted iron body, be non-rising stem and have an O-ring sealed stuffing box. The disk assembly shall be a double disc parallel seat with four point wedging mechanism. A 2" square wrench nut shall be provided The valve shall open left.

2.6 FREESTANDING FIRE HYDRANTS

- A. Dry-Barrel Fire Hydrants: Shall comply with ANSI/AWWA C502, shall be any top design with O-ring sealed oil reservoir. Hydrant shall have a compression type resincalve that closes with pressure, shall have a 5-1/4" main valve opening three way (two lose hozzles and one pumper nozzle), shall have a 200 psig maximum working pressure and 400 psig maximum test pressure, the hydrant shall be Mueller Super Centurion A-423 or approved equal (**must comply Water Supplier and Fire Marshal standards**).
 - 1. Manufacturers:
 - a. American AVK Co.; Valves & Fittings Di
 - b. American Cast Iron Pipe Co.; American flow Control Div.
 - c. American Cast Iron Pipe Co. was us Co. Subsidiary.
 - d. American Foundry Group, hc.
 - e. Mueller Co.; Water Product Dw.
 - f. McWane, Inc.; Clow Valve Co. Div.
 - g. McWane, Inc.; Lennely Valve Div.
 - h. United States Pip, and Foundry Company.

PART 3 - EXECUTION

- 3.1 EARTHWORK
 - A. Refer to Division 2 Section "Earthwork" for excavating, trenching, and backfilling.
- 3.2 PIPING APPLICATIONS

General: Use pipe, fittings, and joining methods for piping systems according to the following applications.

Transition couplings and special fittings with pressure ratings at least equal to piping pressure rating may be used in applications below, unless otherwise indicated.

- C. Do not use flanges, unions, or keyed couplings for underground piping.
- D. Flanges, unions, keyed couplings, and special fittings may be used, instead of joints indicated, on aboveground piping and piping in vaults.

E. Underground Water-Service Piping: Use the following piping materials for each size range:

3.3 JOINT CONSTRUCTION

- See Division 2 Section "Utility Materials" for basic piping joint construction. A.
- B. Make pipe joints according to the following:
 - Ductile-Iron Piping, Gasketed Joints for Water-Service Piping: AWW 1. AWWA M41.

PIPING SYSTEMS - COMMON REQUIREMENTS 3.4

- See Division 2 Section "Utility Materials" for piping-system common A. ements.
- 3.5 PIPING INSTALLATION
 - Water-Main Connection: Arrange with utility company for ta, of s ze and in location indicated A. in water main.
 - Water-Main Connection: Tap water main according to requirements of water utility company B. and of size and in location indicated.
 - C. Make connections to City of Newark ains with tapping machine according to the following:
 - Install tapping sleeve and tapping valve according to MSS SP-60. 1.
 - 2.
 - Install tapping sleeve or pipe to be tapped. Position flanged outlet for gate valve. Use tapping machine comp tible with valve and tapping sleeve; cut hole in main. Remove tapping machine and connect water-service piping. 3.
 - Install gate value of the tapping sleeve. Comply with MSS SP-60. Install valve with stem 4. pointing up and with valve box.
 - Install ductile ror giping according to AWWA C600 and AWWA M41. D.
 - ion with depth of cover over top at least 30 inches. 1.
 - Install pixing by tunneling, jacking, or combination of both, under streets and other obstructions E. that samot be disturbed.

ANCHORAGE INSTALLATION

- Install anchorages for tees, plugs and caps, bends, crosses, valves, and hydrant branches. Include anchorages for the following piping systems:
- 1. Gasketed-Joint, Ductile-Iron, Water-Service Piping: According to AWWA C600.
- Β. Apply full coat of asphalt or other acceptable corrosion-resistant material to surfaces of installed ferrous anchorage devices.

3.7 VALVE INSTALLATION

A. AWWA Gate Valves: Comply with AWWA C600 and AWWA M44. Install each underground valve with stem pointing up and with valve box.

3.8 FIRE HYDRANT INSTALLATION

- A. General: Install each fire hydrant with separate gate valve in supply pipe, and restrained joints or thrust blocks, and support in upright position.
- B. AWWA-Type Fire Hydrants: Comply with AWWA M17.

3.9 CONNECTIONS

- A. Piping installation requirements are specified in other Division 2 sections. Drawings indicate general arrangement of piping and specialties.
- B. See Division 2 Section "Utility Materials" for piping connections to valves and equipment.
- C. Connect water-distribution piping to Town of Smyrna water main. Use tapping sleeve and tapping valve.
- 3.10 FIELD QUALITY CONTROL
 - A. Piping Tests: Conduct piping tests before joints are covered and after thrust blocks have hardened sufficiently. Fill pipeline 24 hours before testing and apply test pressure to stabilize system. Use only potable water.
 - B. Hydrostatic Tests: Test at no less t an 1-1/2 times working pressure for 2 hours.
 - 1. Increase pressure in 50-psig increments and inspect each joint between increments. Hold at test pressure in 1 hour; decrease to 0 psig. Slowly increase again to test pressure and hold for 1 more hoar. Maximum allowable leakage is 2 quarts per hour per 100 joints. Remark leaving joints with new materials and repeat test until leakage is within allowed limits.
 - C. Prepare reports of testing activities.
- 3.11 CLEANING

ean and disinfect water-distribution piping as follows:

- Purge new water-distribution piping systems and parts of existing systems that have been altered, extended, or repaired before use.
- 2. Use purging and disinfecting procedure prescribed by authorities having jurisdiction or, if method is not prescribed by authorities having jurisdiction, use procedure described in AWWA C651 or as described below:

- a. Fill system or part of system with water/chlorine solution containing at least 50 ppm of chlorine; isolate and allow to stand for 24 hours.
- b. Drain system or part of system of previous solution and refill with water/chlorine solution containing at least 200 ppm of chlorine; isolate and allow to stand for 3 hours.
- c. After standing time, flush system with clean, potable water until no ch remains in water coming from system.
- d. Submit water samples in sterile bottles to authorities having jurisdiction procedure if biological examination shows evidence of contamination.
- B. Prepare reports of purging and disinfecting activities.
- C. After completing drinking fountain installation, inspect unit. Remove faint splatters and other spots, dirt, and debris. Repair damaged finish to match original finish
- D. Clean drinking fountains, on completion of installation, according to manufacturer's written instructions.

END OF SECTION 33 25 10

SECTION 33 41 00- STORM DRAINAGE AND SANITARY SEWER

PART 1 GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Suppler Conditions and Division 1 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. This Section includes gravity-flow storm drainage and sanitary sewer outside the building, with the following components:
 - 1. PVC storm and sanitary pipe and fittings
 - 2. RCP storm pipe and flared end section
 - 3. Cleanouts.
 - 4. Drains.
 - 5. Pre-cast concrete catch basins and manhole.
 - 6. Pre-cast concrete pond outlet structure
- 1.3 DEFINITIONS
 - A. RCP: Reinforced concrete pipe
 - B. PVC: Polyvinyl chloride plaste.
- 1.4 PERFORMANCE REQUIREMENTS
 - A. Gravity-Flow, pipe joins shall be at least silt-tight, unless otherwise indicated.

1.5 SUBMITTAK

- A. Product Data: For the following:
 - **ICP** pipe and fittings.
 - Cleanouts
 - Pre-cast pond outlet structure
 - Reinforced concrete flared end sections
 - Storm and Sanitary structures (manholes, manhole seals, catch basins, etc.)

Shop Drawings: For the following:

- 1. Manholes: Include plans, elevations, sections, details, and frames and covers. Include design calculations, and concrete design-mix report for cast-in-place manholes.
- 2. Catch Basins: Include plans, elevations, sections, details, and frames, covers, and grates.
- 3. Stormwater Pond Outlet Structure: Include plans, elevations, sections, details, frames and covers, design calculations, and concrete design-mix report.

- 4. Diversion Structure (DS-1): Include plans, elevations, sections, details, and frames, covers, and grates.
- 5. HDPE to Concrete Connection: Provide ADS specifications or approved equal (as per drawing detail).
- C. Coordination Drawings: Show pipe sizes, locations, and elevations. Show other piping in same trench and clearances from storm drainage system piping. Indicate interface and spatial relationship between manholes, piping, and proximate structures.
- D. Field quality-control test reports.

1.6 DELIVERY, STORAGE, AND HANDLING

- A. Do not store plastic manholes, pipe, and fittings in direct sunlight.
- B. Protect pipe, pipe fittings, and seals from dirt and damage.
- C. Handle manholes according to manufacturer's written rigging histructions.
- D. Handle catch basin and pond outlet structure according to manufacturer's written rigging instructions.
- 1.7 PROJECT CONDITIONS
 - A. Interruption of Existing Storm Drainage Service: Do not interrupt service to facilities occupied by Owner or others unless permittee under the following conditions and then only after arranging to provide temporary service according to requirements indicated:
 - 1. Notify Architect and Gwner to fewer than two days in advance of proposed interruption of service.
 - 2. Do not proceed with interruption of service without Architect's and/or Owner's written permission.

PART 2 - PRODUCTS

- 2.1 MANUFACTURERS
 - In other Part 2 articles where titles below introduce lists, the following requirements apply to product selection:
 - Available Manufacturers: Subject to compliance with requirements, manufacturers offering products that may be incorporated into the Work include, but are not limited to, manufacturers specified.
 - 2. Manufacturers: Subject to compliance with requirements, provide products by one of the manufacturers specified.

2.2 PIPING MATERIALS

A. Refer to Part 3 "Piping Applications" Article for applications of pipe, fitting, and joining materials.

2.3 PE PIPE AND FITTINGS

- A. Corrugated PE Pipe and Fittings: AASHTO M 294M, Type S, with smooth water coupling joints.
 - 1. Silttight Couplings: PE sleeve with ASTM D 1056, Type 2, Class 4, Grade 2 gasket material that mates with pipe and fittings.
 - 2. Soiltight Couplings: AASHTO M 294M, corrugated, matching pipe and mungs.

2.4 PVC PIPE AND FITTINGS

- A. PVC Sewer Pipe and Fittings, 4" to 15" diameter: ASTM D 3034, SDR-26 with bell-and-spigot ends for gasketed joints with ASTM F 477, elastomeric seeks
- B. PVC Sewer Pipe and Fittings, 18" diameter and Larger ASTM F 679, with bell-and-spigot ends for gasketed joints with ASTM F 477, elastomeric seals.

2.5 CONCRETE PIPE AND FITTINGS

A. Reinforced-Concrete Sewer Pipe (RCP) and Fittings: ASTM C 76, with bell-and-spigot ends and gasketed joints with ASTM C 445, ubber gaskets.

2.6 NONPRESSURE-TYPE PIPE COUNTINGS

- A. Comply with ASTM C1103. destomeric, sleeve-type, reducing or transition coupling, for joining underground concressive piping. Include ends of same sizes as piping to be joined, and corrosion-resistant-meta tension band and tightening mechanism on each end.
- B. Sleeve Materials:

2.

- 1. For Concrete Pipes: ASTM C 443, rubber.
 - For Plasac Pipes: ASTM F 477, elastomeric seal or ASTM D 5926, PVC.
- 3. For Dissimilar Pipes: ASTM D 5926, PVC or other material compatible with pipe paterials being joined.

LEANOUTS

Gray-Iron Cleanouts (paved areas): ASME A112.36.2M, round, gray-iron housing with clamping device and round, secured, scoriated, gray-iron cover. Include gray-iron ferrule with inside calk or spigot connection and countersunk, tapered-thread, brass closure plug.

- 1. Manufacturers:
 - a. Josam Company.
 - b. MIFAB Manufacturing, Inc.

- 2. Top-Loading Classification(s): Heavy duty.
- 3. Sewer Pipe Fitting and Riser to Cleanout: ASTM A 74, Service class, cast-iron soil pipe and fittings.
- B. PVC Cleanouts (lawn areas): Schedule-40 PVC body with PVC threaded plug. Include PVC sewer pipe fitting and riser to cleanout of same material as sewer piping.
 - 1. Manufacturers:
 - a. IPS Corporation.
 - b. NDS Inc.

2.8 DRAINAGE STRUCTURES

- A. Standard Precast Concrete Manholes, and Catch Basins: ASTMC 432, precast, reinforced concrete, of depth indicated, with provision for sealant joints.
 - 1. Diameter: 48 inches minimum, unless otherwise indicated.
 - 2. Ballast: Increase thickness of precast concrete sections or and concrete to base section, as required to prevent flotation.
 - 3. Base Section: 6-inch minimum thickness for floorslab and 4-inch minimum thickness for walls and base riser section, and having separate base slab or base section with integral floor.
 - 4. Riser Sections: 4-inch minimum thickness, and lengths to provide depth indicated.
 - 5. Top Section: Eccentric-cone type artists concentric-cone or flat-slab-top type is indicated. Top of cone of size that matches grade rings.
 - 6. Joint Sealant: ASTM C 990 often en or butyl rubber.
 - 7. Resilient Pipe Connectors: ASTMC 923, cast or fitted into manhole walls, for each pipe connection.
 - 8. Steps: Individual FRP steps wide enough to allow worker to place both feet on 1 step and designed to provert lateral slippage off of step. Cast or anchor steps into sidewalls at 12- to 16-inch atervals. Omit steps if total depth from floor of manhole to finished grade is less than 48 incres
 - 9. Adjusting Rongs: Interlocking rings with level or sloped edge in thickness and diameter matching namele frame and cover. Include sealant recommended by ring manufacturer.
 - 10. Grade kings: Reinforced-concrete rings, 6- to 9-inch total thickness, to match diameter of manhole frame and cover.
 - 11. Protective Coating: Plant-applied, SSPC-Paint 16, coal-tar 10-mil minimum thickness applied to exterior and interior surfaces.
 - Manhole Frames and Covers: Ferrous; 24-inch ID by 7- to 9-inch riser with 4-inchminimum width flange and 26-inch-diameter cover. Include indented top design with lettering cast into cover, using wording equivalent to "STORM SEWER."

CONCRETE

12.

- General: Cast-in-place concrete according to ACI 318/318R, ACI 350R, and the following:
 - 1. Cement: ASTM C 150, Type II.
 - 2. Fine Aggregate: ASTM C 33, sand.
 - 3. Coarse Aggregate: ASTM C 33, crushed gravel.
 - 4. Water: Potable.

- B. Portland Cement Design Mix: 4000 psi minimum, with 0.45 maximum water-cementitious materials ratio.
 - 1. Reinforcement Fabric: ASTM A 185, steel, welded wire fabric, plain.
 - 2. Reinforcement Bars: ASTM A 615/A 615M, Grade 60, deformed steel.
- C. Ballast and Pipe Supports: Portland cement design mix, 3000 psi minimum, w maximum water-cementitious materials ratio.
 - 1. Reinforcement Fabric: ASTM A 185, steel, welded wire fabric, plain.
 - 2. Reinforcement Bars: ASTM A 615/A 615M, Grade 60, deformed steel

2.10 PIPE OUTLETS

- A. Head Walls (N/A): Cast-in-place reinforced concrete, with aprox and tapered sides.
- B. Riprap Stilling Basins: Place as per DNREC standards and drawing detail with rip-rap size and thickness as specified on drawings.
- C. Filter Stone: According to NSSGA's "Quarrier stone for Erosion and Sediment Control," No. FS-2, No. 4 screen opening, average-size gr ded stone.

2.11 STORMWATER INLETS

- A. Gutter Inlets: Made with horizontal gutter openings as per DelDOT standards. Include heavyduty frames and Type 3 Bicycle Safe graes.
- B. Combination Inlets: Made who vertical curb and horizontal gutter openings as per DelDOT standards. Include heavy buty stames and Type 3 Bicycle Safe grates.
- C. Frames and Grates: Heavy duty frames and Type 3 Bicycle Safe grates.
- 2.12 STORMWATE: DETENTION OUTLET STRUCTURE
 - A. Pre-Cast Concrete, Stormwater Detention Structures: Submit shop drawings for concrete structure, reinforcing design, slab, trash rack, orifice restrictor plate, and orifice protector

ART 3 - EXECUTION

EARTHWORK

- Excavation, trenching, and backfilling are specified in Division 2 Section "Earthwork."
- 3.2 PIPING APPLICATIONS
 - A. Pipe couplings and special pipe fittings with pressure ratings at least equal to piping rating may be used in applications below, unless otherwise indicated.

- 1. Use nonpressure-type flexible couplings where required to join gravity-flow, nonpressure sewer piping, unless otherwise indicated.
 - a. Flexible or rigid couplings for same or minor difference OD pipes.
 - b. Unshielded, increaser/reducer-pattern, flexible or rigid couplings for pipes with different OD.
 - c. Ring-type flexible couplings for piping of different sizes where annular space between smaller piping's OD and larger piping's ID permits installation.
- 2. Use pressure-type pipe couplings for force-main joints.
- B. Special Pipe Fittings: Use for pipe expansion and deflection. Pipe coupling, and special pipe fittings with pressure ratings at least equal to piping rating may be user in upplications below, unless otherwise indicated.
- C. Gravity-Flow, Nonpressure Sewer Piping: Use the following pipe interials for each size range:
 - 1. Reinforced-concrete sewer pipe and fittings, gasketer and gasketed joints.
 - 2. PVC sewer pipe and fittings, gaskets, and gaskets (NPS 4 and NPS 6).
 - 3. PVC sewer pipe and fittings, gaskets, and gasketed sints (NPS 8 to NPS 12).

3.3 PIPING INSTALLATION

- A. General Locations and Arrangements: Discourse and details indicate general location and arrangement of underground storm drain ce spiping. Location and arrangement of piping layout take design considerations into account, install piping as indicated, to extent practical. Where specific installation is not indicated, follow piping manufacturer's written instructions.
- B. Install piping beginning at low point, true to grades and alignment indicated with unbroken continuity of invert. Place cell adds of piping facing upstream. Install gaskets, seals, sleeves, and couplings according comanufacturer's written instructions for use of lubricants, cements, and other installation requirements.
- C. Install manhoes for manges in direction unless fittings are indicated. Use fittings for branch connections unless direct tap into existing sewer is indicated.
- D. Instal proper ize increasers, reducers, and couplings where different sizes or materials of pipes and strangs are connected. Reducing size of piping in direction of flow is prohibited.

Tunneling: Install pipe under streets or other obstructions that cannot be disturbed by tunneling, jacking, or a combination of both.

Install gravity-flow, nonpressure drainage piping according to the following:

- 1. Install piping pitched down in direction of flow, at minimum slope of 0.005 ft/ft, unless otherwise indicated on plans.
- 2. Install PE corrugated sewer piping according to CPPA's "Recommended Installation Practices for Corrugated Polyethylene Pipe and Fittings."
- 3. Install PVC sewer piping according to ASTM D 2321 and ASTM F 1668.

4. Install reinforced-concrete sewer piping according to ASTM C 1479 and ACPA's "Concrete Pipe Installation Manual."

3.4 PIPE JOINT CONSTRUCTION

- A. Basic pipe joint construction is specified in Division 2 Section "Piped Utilities Basic Materials" and Methods." Where specific joint construction is not indicated, follow piping manufacturer's written instructions.
- B. Join gravity-flow, nonpressure drainage piping according to the following:
 - 1. Join corrugated PE piping according to CPPA 100 and the following
 - a. Use silttight couplings for Type 1, silttight joints.
 - b. Use soiltight couplings for Type 2, soiltight joints.
 - 2. Join PVC sewer piping according to ASTM D 2321 and ASTM D 3034 for elastomericseal joints or ASTM D 3034 for elastomeric gasket joints
 - 3. Join reinforced-concrete sewer piping according to /CPA's "Concrete Pipe Installation Manual" for rubber-gasket joints.
 - 4. Join dissimilar pipe materials with nonpressure-type flexible or rigid couplings.
- C. Join dissimilar pipe materials with pressure-type couplings.

3.5 CLEANOUT INSTALLATION

- A. Install cleanouts and riser extensions from sewer pipes to cleanouts at grade. Use cast-iron soil pipe fittings in sewer pipes at oranghes for cleanouts and cast-iron soil pipe for riser extensions to cleanouts. Install piping socleanouts open in direction of flow in sewer pipe.
 - 1. Use Medium-Duck, top-loading classification cleanouts in **earth or unpaved/paved foot**traffic areas
 - 2. Use Heavy Dety, top-loading classification cleanouts in loading/vehicle-traffic service areas.
- B. Set deaport frames and covers in earth in cast-in-place concrete block, 18 by 18 by 12 inches deep (or as per drawing detail). Set with tops 1 inch above surrounding earth grade.

Set cleanout frames and covers in concrete pavement and roads with tops flush with pavement surface.

MANHOLE INSTALLATION

- General: Install manholes, complete with appurtenances and accessories indicated.
- B. Install precast concrete manhole sections according to ASTM C 891.
- C. Set tops of frames and covers flush with finished surface of manholes that occur in pavements. Set tops 3 inches above finished surface elsewhere, unless otherwise indicated.
CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

3.7 CATCH BASIN INSTALLATION

- A. Construct catch basins to sizes and shapes indicated.
- B. Set frames and grates to elevations indicated.

3.8 DRAIN INSTALLATION

- A. Install type of drains in locations indicated.
 - 1. Use Light-Duty, top-loading classification drains in[earth or unpaved foot-traffic areas.
 - 2. Use Medium-Duty, top-loading classification drains in paved foot traffic areas.
 - 3. Use Heavy-Duty, top-loading classification drains in loading/vebics traffic service areas.
 - 4. Use Extra-Heavy-Duty, top-loading classification drains in [load]] < h sert area>.
- B. Embed drains in 6-inch minimum concrete around bottom and side (see drawing detail).
- C. Fasten grates to drains if indicated.
- D. Set drain frames and covers with tops flush with pavement surface.
- E. Assemble trench sections with flanged joints.
- F. Embed trench sections in 6-inch minimum concrete around bottom and sides (see drawing detail).

3.9 STORM PIPE OUTLET INSTADLATE

- A. Install outlets that spill onto grate with flared end sections that match pipe, where indicated.
- 3.10 CONCRETE PLACENCAN
 - A. Place cast-in-clace concrete according to ACI 318/318R.

3.11 STOPM VATUR INLET AND OUTLET INSTALLATION

A. Construct inlet head walls, aprons, and sides of reinforced concrete, as indicated.

Construct riprap of broken stone, as indicated.

Install outlets that spill onto grade, anchored with concrete, where indicated.

- D. Install outlets that spill onto grade, with flared end sections that match pipe, where indicated.
- E. Construct energy dissipaters at outlets, as indicated.
- 3.12 DRAINAGE SYSTEM INSTALLATION
 - A. Assemble and install components according to manufacturer's written instructions.

STORM DRAINAGE AND SANITARY SEWER

- B. Install with top surfaces of components, except piping, flush with finished surface.
- C. Assemble channel sections to form slope down toward drain outlets. Use sealants, adhesives, fasteners, and other materials recommended by system manufacturer.

3.13 CONNECTIONS

- A. Connect nonpressure, gravity-flow drainage piping in building's storm building drains spec in Division 22 Section "Facility Storm Drainage Piping."
- B. Connect force-main piping to building's storm drainage force mains specified in Division 22 Section "Facility Storm Drainage Piping." Terminate piping where indicated
- C. Make connections to existing piping and underground manholes.
 - 1. Use commercially manufactured wye fittings for piping branch connections. Remove section of existing pipe; install wye fitting into existing piping; and encase entire wye fitting, plus 6-inch overlap, with not less than 6 inches of concrete with 28-day compressive strength of 3000 psi.
 - 2. Make branch connections from side into existing piping, NPS 4 to NPS 20. Remove section of existing pipe, install wye fitting into existing piping, and encase entire wye with not less than 6 inches of concrete with 28 aay compressive strength of 3000 psi.
 - 3. Make branch connections from sole into existing piping, NPS 21 or larger, or to underground manholes and structures by cutting into existing unit and creating an opening large enough to allow 3 tuches of concrete to be packed around entering connection. Cut end of connection pipe passing through pipe or structure wall to conform to shape of an be lust with inside wall unless otherwise indicated. On outside of pipe, manhole, or structure wall, encase entering connection in 6 inches of concrete for minimum length of 12 inches to provide additional support of collar from connection to undisturbed ground.
 - a. Use concrete that will attain a minimum 28-day compressive strength of 3000 psi usless therwise indicated.
 - b. See spoxy-bonding compound as interface between new and existing concrete and pi ing materials.
 - protect existing piping, manholes, and structures to prevent concrete or debris from extering while making tap connections. Remove debris or other extraneous material that may accumulate.

Connect to sediment interceptors specified in Division 22 Section "Sanitary Waste Interceptors."

Pipe couplings, expansion joints, and deflection fittings with pressure ratings at least equal to piping rating may be used in applications below unless otherwise indicated.

1. Use nonpressure-type flexible couplings where required to join gravity-flow, nonpressure sewer piping unless otherwise indicated.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- a. Flexible couplings for same or minor difference OD pipes.
- b. Unshielded, increaser/reducer-pattern, flexible couplings for pipes with different OD.
- c. Ring-type flexible couplings for piping of different sizes where annular space between smaller piping's OD and larger piping's ID permits installation.
- 2. Use pressure-type pipe couplings for force-main joints.

3.14 CLOSING ABANDONED STORM DRAINAGE SYSTEMS

- A. Abandoned Piping: Close open ends of abandoned underground piping indicated to remain in place. Include closures strong enough to withstand hydrostatic and earth pressures that may result after ends of abandoned piping have been closed. Use either proceeding below:
 - 1. Close open ends of piping with at least 8-inch thick, brick max any bulkheads.
 - 2. Close open ends of piping with threaded metal caps, plasts plugs, or other acceptable methods suitable for size and type of material being plosed. Jo not use wood plugs.
- B. Abandoned Manholes and Structures: Excavate around menholes and structures as required and use one procedure below:
 - 1. Remove manhole or structure and close upc, ends of remaining piping.
 - 2. Remove top of manhole or structur edonate at least 36 inches below final grade. Fill to within 12 inches of top with store, rubble, gravel, or compacted dirt. Fill to top with concrete.
- C. Backfill to grade according to Division of Section "Earth Moving."

3.15 IDENTIFICATION

- A. Materials and their instillation are specified in division 2 Section "Earthwork." Arrange for installation of group warning tape directly over piping and at outside edge of underground structures.
 - 1. Use detectable warning tape over nonferrous piping and over edges of underground structures.

3.16 FIELD QUALITY CONTROL

Inspect interior of piping to determine whether line displacement or other damage has occurred. Inspect after approximately 24 inches of backfill is in place, and again at completion of Project.

- 1. Submit separate reports for each system inspection.
- 2. Defects requiring correction include the following:
 - a. Alignment: Less than full diameter of inside of pipe is visible between structures.
 - b. Deflection: Flexible piping with deflection that prevents passage of ball or cylinder of size not less than 92.5 percent of piping diameter.
 - c. Crushed, broken, cracked, or otherwise damaged piping.

CARRCROFT ELEMENTARY SCHOOL WILMINGTON, DE

- d. Infiltration: Water leakage into piping.
- e. Exfiltration: Water leakage from or around piping.
- 3. Replace defective piping using new materials, and repeat inspections until defects are within allowances specified.
- 4. Reinspect and repeat procedure until results are satisfactory.
- 3.17 CLEANING
 - A. Clean interior of piping, flared end sections, and structures of dirt and superfluous materials as directed by CCR and Engineer.

Republic

END OF SECTION 33 41 00

<

